Ca	se 2:18-bk-20151-ER Doc 6669 Filed 00 Main Documerու	2/28/21 Entered 00/28/21 18·3/·/በ Desc Docket #6669 Date Filed: 9/28/2021 רמעדיטוסי			
1 2 3 4	Jason D. Strabo (SBN 246426) MCDERMOTT WILL & EMERY LLP 2049 Century Park East, Suite 3200 Los Angeles, California 90067-3218 Telephone: (310) 788-4125 Facsimile: (310) 277-4730 Email: jstrabo@mwe.com				
5 6 7 8	James W. Kapp, III (admitted <i>pro hac vice</i> ) MCDERMOTT WILL & EMERY LLP 444 West Lake Street, Suite 4000 Chicago, IL 60606-0029 Telephone: (312) 372-2000 Facsimile: (312) 984-7700 E-mail: jkapp@mwe.com				
9	Attorneys for Prime Healthcare Services, Inc.				
10	UNITED STATES BANKRUPTCY COURT				
11		FORNIA – LOS ANGELES DIVISION			
12	In re:	Lead Case No. 2:18-bk-20151-ER			
13	VERITY HEALTH SYSTEM OF, CALIFORNIA, INC., <i>et al.</i> ,	Jointly Administered With: Case No. 2:18-bk-20162-ER			
14	Debtors and Debtors In Possession.	Case No. 2:18-bk-20163-ER Case No. 2:18-bk-20164-ER			
15	Possession.	Case No. 2:18-bk-20165-ER Case No. 2:18-bk-20167-ER			
16	<ul> <li>Affects All Debtors</li> <li>Affects Verity Health System of</li> </ul>	Case No. 2:18-bk-20168-ER Case No. 2:18-bk-20169-ER			
17	California, Inc.  Affects O'Connor Hospital	Case No. 2:18-bk-20171-ER Case No. 2:18-bk-20172-ER			
18	<ul> <li>Affects O'Connor Hospital</li> <li>Affects Saint Louise Regional Hospital</li> </ul>	Case No. 2:18-bk-20173-ER Case No. 2:18-bk-20175-ER			
19	<ul> <li>☑ Affects St. Francis Medical Center</li> <li>□ Affects St. Vincent Medical Center</li> </ul>	Case No. 2:18-bk-20176-ER			
	□ Affects Seton Medical Center	Case No. 2:18-bk-20178-ER Case No. 2:18-bk-20179-ER			
20	<ul> <li>Affects O'Connor Hospital Foundation</li> <li>Affects Saint Louise Regional Hospital</li> </ul>	Case No. 2:18-bk-20180-ER Case No. 2:18-bk-20181-ER			
21	Foundation	Hon. Ernest M. Robles			
22	Affects St. Francis Medical Center of Lynwood Foundation	PRIME HEALTHCARE SERVICES, INC.'S			
23	<ul> <li>Affects St. Vincent Foundation</li> <li>Affects St. Vincent Dialysis Center, Inc.</li> </ul>	REPLY TO POST-EFFECTIVE DATE DEBTORS AND LIQUIDATING			
24	□ Affects Seton Medical Center	TRUSTEE'S MEMORANDUM IN			
25	Foundation <ul> <li>Affects Verity Business Services</li> </ul>	<b>OPPOSITION TO PRIME HEALTHCARE SERVICES, INC.'S MOTION TO ENFORCE</b>			
26	<ul> <li>Affects Verity Medical Foundation</li> <li>Affects Verity Holdings, LLC</li> </ul>	PROVISIONS OF THE ASSET PURCHASE AGREEMENT PERTAINING TO			
27	□ Affects De Paul Ventures, LLC	ACCOUNTS RECEIVABLE ADJUSTMENT; SUPPLEMENTAL			
28					
		1820151210929000000000001			

Ca	se 2:18-bk-20151-ER Doc 6669 Filed 09 Main Document	9/28/21 Ente Page 2 of 5		Desc
1 2 3 4	<ul> <li>Affects De Paul Ventures - San Jose Dialysis, LLC</li> <li>Debtors and Debtors In</li> <li>Possession.</li> </ul>	SUPPORT 7 ANA GOFF SUPPLEME STEVE ALI	FION OF A. JOEL RICH FHEREOF; DECLARAT IN SUPPORT THEREO ENTAL DECLARATION EMAN IN SUPPORT TH ocket Nos. 6645 and 6662	ION OF F; OF
5 6 7 8 9		<u>Hearing Dat</u> Date: Time: Location:	<u>e and Time:</u> October 6, 2021 10:00 a.m. 255 E. Temple St., Courtroom 1568 Los Angeles, CA 90012	
10		J		
11				
12				
13				
14				
15				
16				
17				
18				
19				
20				
21				
22				
23				
24				
25				
26 27				
27 28				
28		2		

Ca	Case 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Main Document Page 3 of 51					
1			TABLE OF CONTENTS			
2	I.	PREL	IMINARY STATEMENT1			
3	II.	STAT	TEMENT OF FACTS SUPPLEMENTING THE MOTION			
4	III.	REPL	.Y7			
5		А.	PRIME'S CALCULATION OF THE FINAL A/R COLLECTED IS ACCURATE			
6			AND CONSISTENT WITH THE CLEAR TERMS OF THE APA7			
7		В.	THE EVIDENCE DEMONSTRATES THAT THE PARTIES EXPRESSLY			
8			AGREED TO EXCLUDE TRAUMA PAYMENTS FROM ACCOUNTS			
9			RECEIVABLE9			
10		C.	PRIME USED GOOD FAITH COMMERCIALLY REASONABLE EFFORTS TO			
11			COLLECT THE ACCOUNTS RECEIVABLE			
12		D.	IF THE COURT FINDS THAT PRIME DID NOT DEVOTE SUFFICIENT			
13			EFFORTS TO COLLECT THE ACCOUNTS RECEIVABLE, THEN AN			
14			EVIDENTIARY HEARING WOULD BE REQUIRED TO DETERMINE			
15			DAMAGES17			
16		E.	PRIME HAS TURNED OVER THE FUNDS TO WHICH SELLERS ARE			
17			ENTITLED			
18						
19						
20						
21						
22						
23						
24						
25						
26						
27 28						
28						

Ca	se 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Main Document Page 4 of 51							
1	TABLE OF AUTHORITIES							
2	Cases							
3	Airborne Freight Corp. v. McPherson, 427 F.2d 1283, 1286 (9th Cir. 1970)11							
4	<i>AMC Tech., LLC v. Cisco Sys., Inc.</i> , No. 11-CV-3403 PSG, 2013 WL 3559807, at *7 (N.D. Cal.							
5	July 11, 2013)							
6 7	Brinderson-Newberg Joint Venture v. Pac. Erectors, Inc., 971 F.2d 272, 279 (9th Cir. 1992)8							
8	<i>In re Hawkeye Ent., LLC</i> , 625 B.R. 745, 752 (Bankr. C.D. Cal. 2021)							
9	In re Motors Liquidation Co., 529 B.R. 510, 565 (Bankr. S.D.N.Y. 2015)							
10	Statutes							
11	Cal. Civ. Code § 16387							
12	Cal. Civ. Code § 16397							
13 14	Cal. Civ. Code § 1641							
14								
16								
17								
18								
19								
20								
21								
22 23								
23 24								
25								
26								
27								
28								

Prime Healthcare Services, Inc. ("Prime") replies (this "Reply") to the Post-Effective 1 1. 2 Date Debtors and Liquidating Trustee's Memorandum in Opposition to Prime Healthcare Services, 3 Inc.'s Motion to Enforce Provisions of the Asset Purchase Agreement Pertaining to Accounts Receivable Adjustment; Declaration of Peter Chadwick [Docket No. 6662] (the "Objection" filed by 4 5 the "Debtors") filed in response to Prime Healthcare Services, Inc.'s Motion to Enforce Provisions 6 of the Asset Purchase Agreement Pertaining to Accounts Receivable Adjustment; Declaration of A. 7 Joel Richlin in Support Thereof; Declaration of Ken Wheeler in Support Thereof; Declaration of Steve Aleman in Support Thereof [Docket No. 6645] (the "Motion")<sup>1</sup>, and in support thereof 8 9 represents as follows:

#### **MEMORANDUM OF POINTS AND AUTHORITIES** I. **PRELIMINARY STATEMENT**

12 2. The APA is crystal clear that trauma payments are excluded from the definition of 13 Accounts Receivable, as trauma payments are specifically defined as Other Receivables. As only 14 Accounts Receivable are subject to the Reconciliation Process in section 1.12(a) of the APA, Other 15 Receivables (and thus trauma payments) cannot be included in the calculation of Final A/R Collected. The Debtors do not seriously dispute that this is what the APA provides, nor can they, since the language is clear.

3. Instead, the Debtors urge the Court to essentially write the term Other Receivables out of the APA altogether (or have it consumed by the term Accounts Receivable) by arguing that the exclusion of trauma payments from the defined term Accounts Receivable would create some unintended gift or windfall to Prime based solely on parol evidence of the outdated derivation of the A/R Target Amount from the Parties' due diligence review prior to executing the APA.<sup>2</sup> Even considering parol evidence, however, this Court will find (as described herein) that the exclusion of trauma payments from Accounts Receivable was heavily negotiated and expressly agreed to by the Debtors. Shockingly, the Debtors failed to inform the Court about this intense negotiation and

10

<sup>26</sup> Capitalized terms used but not defined in this Reply shall have the meanings given to such terms in the Motion. 27

<sup>&</sup>lt;sup>2</sup> Debtors concede that the \$11,974,080 collected pursuant to the Trauma Center Services

Agreement constitute "trauma payments" under the APA. See Objection at p. 5, lines 9-22.

#### Case 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Main Document Page 6 of 51

subsequent agreement, all of which is clearly memorialized in emails and redlined drafts. Far from being a "nullity," the addition of Other Receivables to the APA created an essential exclusion from 3 Accounts Receivable by carving out specified items from the definition of Accounts Receivable and, 4 therefore, the calculation of Final A/R Collected.

5 6

7

8

9

1

2

4. To eliminate a heavily negotiated concept from the APA more than a year after the closing of the SFMC Sale, a concept relied upon by Prime as an inducement to finalize the sale which resulted in \$200 million going to the Debtors' estates, would be a "harsh and unjust result." Prime is before this Court simply asking that it enforce the clear terms of the business deal reached by the Parties and clearly reflected in the APA.

10 5. In addition, the standard governing Prime's collection efforts in section 1.12(e) of the 11 APA is also explicitly clear. This Court has ample evidence before it (including the Wheeler 12 Declaration, the Aleman Declaration, the Supplemental Aleman Declaration, and the Goff 13 Declaration) establishing that Prime used good faith, commercially reasonable best efforts during the 14 Reconciliation Period and that those efforts were at least consistent with, if not better than, the efforts 15 Prime used to collect its other receivables. The Debtors have failed to submit any competent evidence 16 to the contrary and instead rely upon speculation about potential challenges with collection efforts as 17 a basis to accuse Prime of bad faith. See Objection p. 17, lines 7-18. But nothing contained in the 18 single declaration submitted by the Debtors establishes a lack of good faith or entitles Sellers to pay 19 less than they agreed to in the APA. For example, the Debtors' focus on historical collection rates is 20 irrelevant to the determination of whether Prime's collection efforts during the Reconciliation Period 21 were made in good faith and commercially reasonable-which, again, is the governing standard. 22 Moreover, and as discussed herein, the Debtors' analysis of collection rates is misleading and 23 unreliable.

24 6. Finally, Prime is not holding any of the Sellers' QAF V Seller Net Payments or 25 Sellers' share of disproportionate share payments and the Debtors acknowledge such fact in footnote 26 one of their Objection. Section 1.12(d)(ii) of the APA provides that Prime may offset the Sellers' 27 QAF VI Seller Net Payments under certain circumstances. Thus, as there is an ongoing dispute 28 between the Parties regarding an amount greater than the \$24.1 million in QAF VI Seller Net

Payments, it is appropriate under the APA for Prime to withhold such amount until this Court resolves the dispute now before it.

7. For all the foregoing reasons and as set forth in more detail below, the Objection should be overruled and the Motion should be granted.

1

2

3

4

5

6

11

#### II. STATEMENT OF FACTS SUPPLEMENTING THE MOTION

8. Pursuant to the Bid Procedures Order entered by this Court, Prime and Sellers 7 negotiated the APA throughout March and early April of 2020, which coincided with the beginning 8 of the COVID-19 pandemic in the United States. See Supplemental Declaration of A. Joel Richlin 9 (the "Supplemental Richlin Declaration") at ¶ 4. The pandemic created significant uncertainty as to 10 the operations of St. Francis Medical Center ("SFMC") and the valuations on which Prime had relied in pursuing the SFMC Sale. See id. In fact, the uncertainty created by the pandemic was so significant 12 and cast such doubt on Sellers' valuation that Prime seriously considered withdrawing from the bid 13 process and/or lowering the purchase price in its bid. See id. Ultimately, no other bidder sought to 14 acquire the Assets. See Objection  $\P$  6 ("The Debtors did not receive any other 'Qualified Bids' for 15 SFMC under the terms of the Bidding Procedures Order.").

9. To address the uncertainty caused by COVID-19, Prime and Sellers engaged in extensive negotiations to allocate risk between the Parties and to adjust to the new economic environment. See id. at ¶ 5. In particular, Prime sought various items from Sellers in the final week of negotiations that would assuage Prime's concerns and provide it the comfort necessary to close the SFMC Sale. See id. Prime sought these modifications to the APA in exchange for Prime's agreement not to decrease the agreed upon \$200 million purchase price, thereby rendering a benefit to the Debtors and their estates. See id. The negotiated modifications to the APA included the creation of the Accounts Receivable Reconciliation in section 1.12 of the APA and the exclusion of certain items from the definition of the term "Accounts Receivable" in section 1.7(p). See id.

10. The negotiations to address the allocation of risk occurred after the creation of the schedule from which the A/R Target Amount was derived. See id. at ¶ 6. Indeed, this schedule was exchanged during the Parties' due diligence review prior to executing the APA. See id. In allocating 28 the risk in a way to entice Prime to consummate the SFMC Sale, Sellers agreed to a definition of

#### Entered 09/28/21 18:34:40 Case 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Desc Main Document Page 8 of 51

Accounts Receivable disconnected from the line items included in this schedule. See id. Thus, the derivation of the A/R Target Amount is an irrelevant reference point when interpreting the APA's definition of Accounts Receivable. See id.

3

4

5

6

7

8

9

10

11

12

12.

1

2

11. The intense negotiation as to what items constituted, and what should be excluded from, "Accounts Receivable" is clearly demonstrated by contemporaneous emails and redline drafts of the APA exchanged between Prime and Sellers leading up to the Closing Date.<sup>3</sup> The relevant period of negotiation of the APA began on March 26, 2020, when Sellers' counsel provided a draft of the APA to Prime (the "3/26/2020 Redline"), attached hereto as Exhibit 8. See Supplemental Richlin Declaration ¶ 7. In the 3/26/2020 Redline, the term Accounts Receivable did not include any carve outs or exclusions. See 3/26/2020 Redline § 1.7(p). Section 1.7(p) also did not include the term "Other Receivables." Indeed, the term Accounts Receivable in the 3/26/2020 Redline included, among other things, trauma payments and disproportionate share payments. Id. If the APA had not been further negotiated and modified, this version of the APA would have been consistent with the Debtors' interpretation of the APA.

However, Prime was unwilling to accept the language in the 3/26/2020 Redline in light of the financial risk posed by the pandemic and the recent announcements that the Federal Government would be providing significant financial support to hospitals to address the pandemic. See Supplemental Richlin Declaration at  $\P$  8. The amounts and rules related to these government payments were unknown at the time, which created additional uncertainty in the ability of both sides to negotiate over which collections should qualify as Accounts Receivable and be included in the A/R Target Amount. See id. Therefore, Prime rejected the language contained in 3/26/2020 Redline and raised its concerns with Sellers on a phone call on March 31, 2020. See id. Mr. Richlin, Vice President and General Counsel for Prime, memorialized the discussion in a follow-up email (the "COVID-19 Grant Email"), attached hereto as Exhibit 9, to Jim Moloney of Cain Brothers, the investment banking firm representing Sellers. See id. In the COVID-19 Grant Email, Mr. Richlin states that "we need to clarify some of the definitions related to AR and government payments going

<sup>&</sup>lt;sup>3</sup> Such negotiations are described in detail in the Supplemental Richlin Declaration.

#### Entered 09/28/21 18:34:40 Case 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Desc Main Document Page 9 of 51

different ways. This is the point I discussed with Elspeth<sup>4</sup> where Verity excluded additional 1 2 government payments related to COVID in their last turn and we think those are essential to the value 3 of the asset since those payments will not even make the hospital whole. So we are revising that 4 language to make it as clear as possible because the passage of new laws creating these payments 5 needs to be addressed." See COVID-19 Grant Email.

6 13. On April 1, 2020, Prime provided a revised draft APA (the "4/1/2020 Redline"), 7 attached hereto as Exhibit 10, to Sellers by email (the "Revised APA Email"), attached hereto as 8 Exhibit 11. Mr. Richlin expressly notes that the revised APA includes a "revised AR formula to 9 address expected COVID losses and supplemental payments (as discussed on our call)." See Revised 10 APA Email. In fact, the 4/1/2020 Redline now introduces the concept of Other Receivables as 11 separate and apart from Accounts Receivable. See 4/1/2020 Redline § 1.7(p). In particular, the 12 definition of Other Receivables specifically included, among other things, trauma payments, grants related to COVID-19, and disproportionate share payments. See id. Section 1.12 of the 4/1/2020 13 14 Redline remained largely unchanged, reflecting the Parties' understanding that Other Receivables 15 would not be credited towards the Final A/R Collected. See id. at §1.12.

16 14. Later on April 1, 2020, Mr. Richlin had a phone call with Ms. Paul to discuss the 17 4/1/2020 Redline and the changes to the definition of "Accounts Receivable." See Supplemental 18 Richlin Declaration at ¶ 10. During the call, both Mr. Richlin and Ms. Paul agreed to speak with 19 their respective management teams about these issues and reconvene in an effort to negotiate a 20 compromise. See id. Mr. Richlin later sent an email on April 1, 2020 (the "APA Conference Email"), attached hereto as Exhibit 12. As reflected in the email chain, Mr. Richlin and Ms. Paul subsequently 22 arranged a group phone to discuss these issues on April 2, 2020.

23 24 25

21

15. Following the group phone call on April 2, 2020, Ms. Paul sent Mr. Richlin an email (the "Accounts Receivable Email"), attached hereto as Exhibit 13, expressly agreeing to Prime's new definition of "Accounts Receivable" which excluded "trauma payments," in exchange for Prime's agreement to move government payments related to COVID-19 out of the definition of "Other

27 28

<sup>&</sup>lt;sup>4</sup> "Elspeth" refers to Elspeth Paul, the General Counsel for VHS at the time.

Ca	se 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Main Document Page 10 of 51						
1	Receivables" and into the definition of "Accounts Receivable". See Supplemental Richlin						
2	Declaration at ¶ 11.						
3	From: Paul, Elspeth < <u>ElspethPaul@verity.org</u> > Sent: Thursday, April 2, 2020 3:29 PM						
4	To: Joel Richlin (PHMI) < <u>JRichlin@primehealthcare.com</u> > Subject: EXTERNAL:COVID \$						
5	WARNING: This email originated outside the Prime Healthcare email system! Do Not Click links if this user is unknown.						
6	Joel: With your new definition of accounts receivable and receivables, I'm thinking the edit is simply to move where the COVID \$ is referenced:						
7	(p) (i) all accounts and interest thereupon, notes and interest thereupon and other receivables of Sellers, including, without limitation, accounts, notes or other amounts receivable, and all claims, rights, interests and proceeds related thereto, including all accounts and other receivables, , in each case arising from the						
8	rendering of services or provision of goods, products or supplies to inpatients and outpatients at the Hospital, billed and unbilled, recorded and unrecorded (including any such amounts that were written-off by Sellers for any reason), for services, goods, products and supplies provided by Sellers prior to the						
9	Licensure Date whether payable by Medicare, Medi-Cal, Medicaid, or any other payor (including an insurance company), or any health care provider or network (such as a health maintenance organization, preferred provider organization or any other managed care program) or any fiscal intermediary of the foregoing, private pay patients, private insurance or by any other source and all claims, rights, interests and proceeds relating to any grant or governmental						
10	awards directly or indirectly related to COVID-19 (collectively, "Accounts Receivable"); and (ii) all claims, rights, interests and proceeds relating to any grant or governmental awards directly or indirectly related to COVID-19, trauma payments, disproportionate share payments, cost report, claim, EHR or other similar						
11	appeals and Seller Cost Report settlements in each case arising from the rendering of services or provision of goods, products or supplies to inpatients and outpatients at the Hospital (the "Other Receivables" and together with the Accounts Receivable, the "Receivables");						
12	See Supplemental Richlin Declaration at ¶ 11; Accounts Receivable Email.						
13	16. On April 2, 2020, Mr. Richlin emailed Sellers and their counsel (the "APA						
14	Negotiations Email"), attached hereto as Exhibit 14, asking when Sellers would be circulating an						
15	updated redline of the APA based on Mr. Richlin's agreement with Ms. Paul. See APA Negotiations						
16	Email. Sellers' counsel responded later in the chain acknowledging that they were incorporating the						
17	language Mr. Richlin and Ms. Paul agreed to into the APA. See id.						
18	17. Finally, on April 3, 2020, Sellers' counsel provided Prime with a revised redline of						
19	the APA (the " <u>4/3/2020 Redline</u> "), attached hereto as <u>Exhibit 15</u> , by email (the " <u>Final APA Email</u> "),						
20	attached hereto as Exhibit 16, which reflected the agreement between Mr. Richlin and Ms. Paul. See						
21	Final APA Email. The 4/3/2020 Redline reflects that some items, such as COVID-19 grants, were						
22	moved back into the definition of Accounts Receivable. See 4/3/2020 Redline § 1.7(p). Other items,						
23	including trauma payments, remained as "Other Receivables" and are thereby specifically excluded						
24	from the definition of Accounts Receivable. Thus, the numerous calls and correspondence regarding						
25	the items to be included in the definition of Accounts Receivable resulted in the express agreement						
26	to exclude trauma payments from the definition of Accounts Receivable. See Supplemental Richlin						
27	Declaration ¶ 15.						
28							
	6						

18. 1 The thoroughly negotiated definition of Accounts Receivable and the Accounts 2 Receivable true-up procedures were critical in inducing Prime to complete the SFMC Sale. See id. III. REPLY 3 PRIME'S CALCULATION OF THE FINAL A/R COLLECTED IS ACCURATE AND 4 A. CONSISTENT WITH THE CLEAR TERMS OF THE APA. 5 Prime's calculation of the Final A/R Collected complies with the unambiguous terms 19. 6 7 of the APA. Trauma payments are clearly excluded from the definition of Accounts Receivable, as

61 the APA. Trauma payments are clearly excluded from the definition of Accounts Receivable, as
trauma payments are specifically defined as Other Receivables. *See* APA § 1.7(p)(i)-(ii). Only
Accounts Receivable are subject to the Reconciliation Process. *See* APA §1.12(a). Thus, Other
Receivables, such as trauma payments, cannot be included in the Final A/R Collected. *Id*.

11

12

13

14

15

16

20. The APA is clear that trauma payments, as part of Other Receivables, are separate and distinct from Accounts Receivables. *See* APA § 1.7(p)(i)-(ii). There is no ambiguity. *See* Cal. Civ. Code § 1639 ("When a contract is reduced to writing, the intention of the parties is to be ascertained from the writing alone, if possible. . ."); *see also* Cal. Civ. Code § 1638 ("The language of a contract is to govern its interpretation, if the language is clear and explicit, and does not involve an absurdity.").

17 21. In addition to defying the plain language of the APA and ignoring the specific 18 limitation of the term Accounts Receivable created by the term Other Receivables, including trauma 19 payments as Accounts Receivable violates California's rules of contract interpretation. Focusing on the fact that the undefined term "other receivables" is included in the definition of Accounts 20 Receivable, the Debtors argue that Accounts Receivable should essentially consume the term Other 21 22 Receivables and any items (such as trauma payments) included in such definition. See Objection p. 13, lines 3-6. The more specific term "trauma payments," however, is included as part of the carve 23 out from Accounts Receivable. Where a specific list of "Other Receivables" is provided and includes 24 trauma payments, an abstract reference to "other receivables" as a form of Accounts Receivable 25 cannot create ambiguity as to whether trauma payments are also Accounts Receivable. See AMC 26 27 Tech., LLC v. Cisco Sys., Inc., No. 11-CV-3403 PSG, 2013 WL 3559807, at \*7 (N.D. Cal. July 11, 2013) (citing Scudder v. Perce, 159 Cal. 429, 433, 114 P. 571, 573 (1911) ("the familiar rule [is] that 28

#### Case 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Main Document Page 12 of 51

when general and specific provisions of a contract deal with the same subject-matter, the specific
 provisions, if inconsistent with the general provisions, are of controlling force")).

3 22. Ultimately, what the Debtors seek to do is rewrite an unambiguous contract more than 4 a year after the Closing Date because they do not like the outcome of the agreement they reached. 5 Adopting the Debtors' interpretation of the APA requires this Court to violate well-settled principles 6 of contract interpretation. Simply put, trauma payments are excluded from the definition of Accounts 7 Receivable because they are not Accounts Receivable and the ordinary interpretation of this language 8 necessitates that conclusion. See In re Hawkeye Ent., LLC, 625 B.R. 745, 752 (Bankr. C.D. Cal. 9 2021) ("We interpret the intent and scope of the agreement by focusing on the usual and ordinary 10 meaning of the language used. . ." (quoting *Lloyd's Underwriters v. Craig & Rush, Inc.*, 26 Cal. App. 11 4th 1194, 1198 (1994)). The Debtors' interpretation, by their own concession, requires reading 12 "Other Receivables" out of the APA. See Objection p. 14, lines 21-22. To the contrary, however, 13 agreements should be interpreted to give meaning to each of their provisions. See Brinderson-14 Newberg Joint Venture v. Pac. Erectors, Inc., 971 F.2d 272, 279 (9th Cir. 1992) ("It is well settled 15 that a contract should be interpreted so as to give meaning to each of its provisions. . ."); see also Cal. Civ. Code § 1641 ("The whole of a contract is to be taken together, so as to give effect to every 16 17 part, if reasonably practicable, each clause helping to interpret the other."). As a result, the term Other 18 Receivables cannot simply be ignored and must be attributed its reasonable meaning.

Although the Debtors appeal to the alleged unfairness of the transaction,<sup>5</sup> there is 19 23. 20 nothing unfair about enforcing the clear terms of a heavily negotiated contract between sophisticated 21 Parties. Indeed, section 12.14 of the APA asserts that the APA contains the entire understanding 22 between the Parties with respect to the contemplated transaction and supersedes any other 23 agreements, understandings or statements to the contrary. In fact, true unfairness would arise if the 24 Debtors were allowed to now rewrite the provisions of the APA that were critical in inducing Prime 25 to complete the SFMC Sale and provide \$200 million to the Debtors' estates. Such an outcome 26 would deprive Prime of the benefit of the bargain it reached and undermine the reliability of sale

27 28

<sup>5</sup> See Objection p. 14, lines 9-11.

### Case 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Main Document Page 13 of 51

orders in bankruptcy courts. *In re Motors Liquidation Co.*, 529 B.R. 510, 565 (Bankr. S.D.N.Y. 2015) ("But in bankruptcy, the interests inherent in the enforceability of 363 orders (on which the buyers of assets should justifiably be able to rely, and the interests of creditors depending on the maximization of estate value likewise rest) are hugely important. And to the extent that courts can respect and enforce sale orders as written unless there is genuine prejudice, they should do so.").

6

7

8

9

10

11

B.

1

2

3

4

5

### THE EVIDENCE DEMONSTRATES THAT THE PARTIES EXPRESSLY AGREED TO EXCLUDE TRAUMA PAYMENTS FROM ACCOUNTS RECEIVABLE.

24. The evidence before this Court clearly corroborates the plain language of the APA and supports the conclusion that the intention of the Parties was to exclude trauma payments from Accounts Receivable, and as a result, exclude trauma payments from the calculation of Final A/R Collected. Parol evidence should not be used to contradict the plain meaning of the APA.

12 25. What constituted Accounts Receivable, such that it would be considered as part of the Accounts Receivable Reconciliation process, was heavily negotiated during the week preceding the 13 14 execution of the APA. The term was continuously revised, with certain items being added and 15 removed from the definition, right up until the date of execution of the APA. The Parties had calls regarding the items contained in the definition. See Supplemental Richlin Declaration ¶ 15, COVID-16 17 19 Grant Email; Revised APA Email. They negotiated the term by email. See, e.g., Revised APA Email; Accounts Receivable Email. They traded definitions in revised drafts of the APA. See, e.g., 18 19 3/26/2020 Redline; 4/1/2020 Redline; 4/3/2020 Redline. These revisions were critical as each party worked to resolve and address various considerations and concerns, including: (a) the proper 20 allocation of risk between the Parties; (b) the uncertainty created by COVID-19 and the governmental 21 22 response thereto; and (c) the need to induce Prime to close the SFMC Sale after other parties declined 23 to participate in the bidding process. See Supplemental Richlin Declaration ¶ 15.

24 25

26

27

26. Despite the Debtors' claims of "ambiguity," the repeated revision of and discussion regarding the definitions of Accounts Receivable and Other Receivables provides comfort that the APA's language states exactly what the Parties intended: that trauma payments would not be included in the definition of the Accounts Receivable and, therefore, would not be included in the Final A/R

### Case 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Main Document Page 14 of 51

Collected.<sup>6</sup> In each instance, the Parties sought to provide documentation for a transaction they could close. It defies logic that a concept as heavily negotiated as the definition of "Other Receivables" should be ignored or "written out" of the APA, or be considered a "nullity," as urged by the Debtors. *See* Objection p. 14, lines 9-11.

27. Far from being a nullity, the evidence demonstrates that the term "Other Receivables" provided an essential exclusion from the term "Accounts Receivable" and was heavily negotiated by Prime and Sellers. In stark contrast to a "gift",<sup>7</sup> the exclusion of trauma payments from Accounts Receivable was a critical business point that materially impacted Prime's willingness to complete a purchase of a hospital in the midst of a global pandemic that caused the indefinite pause of certain services offered by that hospital and the decline in its financial performance. *See* Supplemental Richlin Declaration at ¶ 16. The initial draft of the APA was drafted prior to the COVID-19 pandemic in the United States. *See id.* As the gravity of the pandemic became obvious, the economic environment in which the Parties initially drafted the APA was drastically different than the one in which the Parties ultimately closed the transaction. *See id.* 

15 28. Completely ignoring the back-and-forth negotiations concerning section 1.7(p) of the 16 APA to discern the Parties' intent, the Debtors rely solely on a schedule exchanged during the Parties' 17 due diligence review prior to executing the APA. According to Sellers, "[t]he derivation of the \$61 18 million A/R Target Amount was a subject of the Parties' due diligence review, and the schedule 19 exchanged during that review" includes a line item for "S-9 COUNTY TRAMA." Declaration of Peter Chadwick (the "Chadwick Declaration") at ¶ 11. However, the scrupulously documented 20 21 evidence of negotiations over the definition of Accounts Receivable shows that Sellers expressly agreed to a definition of Accounts Receivable disconnected from the line items included in this 22 23 schedule. See Supplemental Richlin Declaration at ¶ 6. Indeed, this schedule was drafted during and

24

1

2

3

4

5

6

7

8

9

10

11

12

13

14

failed to mention the various APA redlines reflecting changes to the definition of Account Receivable, some of which were actually provided by Debtors' counsel.

<sup>28</sup> <sup>7</sup> See Objection p. 16, lines 3-5.

 <sup>&</sup>lt;sup>6</sup> Shockingly, despite the Debtors' insistence that extrinsic evidence is necessary to interpret the
 APA, the Debtors completely failed to alert the Court to the numerous calls and emails regarding
 the negotiation of the APA in which the Debtors and their counsel took part. Likewise, the Debtors
 failed to mention the various APA redlines reflecting changes to the definition of Accounts

#### Case 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Main Document Page 15 of 51

1 reflects a drastically different economic environment than the one in which the Parties ultimately 2 closed the transaction. See id. As a result, the derivation of the A/R Target Amount is an irrelevant 3 reference point when interpreting the APA's definition of "Accounts Receivable." See id. Debtors' 4 reliance on this schedule is nothing more than a blatant attempt to somehow wind back several 5 iterations of the APA to enforce a deal to which the Parties never agreed. See id. at ¶¶ 6-8.

29. As the APA is unambiguous, both by its language and the evidence submitted to this Court, parol evidence may not now be used to vary the terms of the contract, as the Debtors attempt. 8 See Airborne Freight Corp. v. McPherson, 427 F.2d 1283, 1286 (9th Cir. 1970). Indeed, the parol evidence only confirms the fact that the heavily negotiated removal of trauma payments from the definition of Accounts Receivable was carefully considered and not a mistake, but valuable consideration that ultimately allowed the Sale to close and \$200 million to benefit the Debtors' estates.

13

С.

6

7

9

10

11

12

14

### PRIME USED GOOD FAITH COMMERCIALLY REASONABLE EFFORTS TO COLLECT THE ACCOUNTS RECEIVABLE.

15 30. The evidence before the Court conclusively demonstrates that Prime complied with 16 its collection obligations under the APA. As set out in the APA, the applicable standard regarding 17 collections is that Prime use "good faith, commercially reasonable best efforts to collect the Accounts 18 Receivable (including at least the efforts used by Purchaser to collect its other receivables)." APA 19 §1.12(e) (emphasis added). The evidence makes clear that Prime's collection efforts during the 20 Reconciliation Period were consistent with industry standards and Prime's general practices for collecting receivables. See Wheeler Declaration ¶¶ 6, 8; see also Aleman Declaration at ¶ 4; see also 21 22 Declaration of Ana Goff (the "Goff Declaration") ¶ 4. Extensive evidence has been submitted 23 detailing the extent of Prime's collection efforts. See generally Wheeler Declaration. On this basis, 24 the Court should find that Prime complied with its collection requirements under the APA. There is 25 no competent evidence to the contrary.

26 31. Indeed, the Debtors have submitted a single declaration from their Chief Financial 27 Officer, Peter Chadwick, speculating about Prime's collection efforts and making unsubstantiated 28 allegations of bad faith based solely on unreliable data reports and a single communication that Mr.

#### Case 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Main Document Page 16 of 51

Chadwick had no role in. *See* Chadwick Declaration at ¶¶ 15-22. Mr. Chadwick's declaration fails to
 adequately address much of the evidence put forth to support Prime's collection efforts.

3 32. For example, the Debtors rely on the Chadwick Declaration to allege that "Prime did 4 not devote significant effort to following up with payors on individual claims as had been historically 5 done." Objection, p. 17, lines 7-8. However, Mr. Chadwick does not even claim to have personal 6 knowledge of the follow-up efforts of Prime's collectors and instead relies solely on summary data 7 from the TRAC system. See Chadwick Declaration at ¶ 16. But Prime previously submitted evidence 8 that "Prime did not need to use the TRAC system and ultimately stopped using the TRAC system 9 altogether after the first 90 days of the Reconciliation Period. [Instead], Prime collectors entered 10 notes directly into Sellers' legacy EMR system, MS4." See Wheeler Declaration at ¶ 5. Debtors 11 simply fail to address the fact that the TRAC data is incomplete and, therefore, unreliable.

Moreover, Debtors also ignore the Wheeler Declaration, which states that "Prime's collectors noted 2,893 follow ups on individual accounts in August; 3,335 individual follow ups in September; and 2,316 individual follow ups in October. These individual follow ups do not include Prime's efforts to work groups of claims at once, which cannot be logged for tracking purposes." *See* Wheeler Declaration at ¶ 7. The Debtors do not dispute these figures nor do they dispute that these efforts are consistent with Prime's collection efforts for other receivables, which, again, is the applicable standard for determining whether Prime's collection efforts were appropriate.

19 34. The Debtors similarly rely on the Chadwick Declaration to allege that Debtors had 20 previously "complained to Prime that, as of December 2020, Prime was holding over \$1.6 million in 21 unbilled claims and had failed to submit adequate documentation to support an additional \$16 million 22 in claims." Objection, p. 15, lines 5-7. However, once again, Mr. Chadwick does not even claim to 23 have personal knowledge of Prime's billing of claims or submission of documentation and instead 24 relies solely on summary data from the legacy EMR, MS4. See Chadwick Declaration at ¶ 18. But 25 Prime previously submitted evidence that "Prime implemented an EMR system conversion from 26 MS4 to Epic upon the Closing Date" and therefore any claims billed out of Epic would not show up 27 in data pulled from MS4. See Wheeler Declaration at ¶ 10. Prime also previously explained that the 28 allegedly unbilled claims appeared to be cross-over claims, which "are those claims that arise from

#### Case 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Main Document Page 17 of 51

patients that underwent treatment in the hospital prior to midnight on the evening of the Closing Date and were discharged from the hospital after the Closing Date." *See id.* Because of the system conversion, "all cross-over claims were billed out of Epic, not MS4" and therefore "they appear in MS4 as having never been billed." *See id.* Prime even provided a data file of these cross-over claims to Sellers for validation. *See id.* Inexplicably, Debtors fail to address this evidence and do not even claim to have performed additional analysis or review of this issue since they first "complained to Prime" back in December of 2020. *See* Chadwick Declaration at ¶ 18.

35. Moreover, even assuming that the Debtors had submitted competent evidence of unbilled claims, which they have not, this still would not mean that Prime failed to exercise good faith, commercially reasonable best efforts. This is because Prime previously submitted undisputed evidence that "Prime diligently tracked any claim that was not yet billed and made commercially reasonable efforts to promptly bill such claims to avoid any timely filing denial. Specifically, the Hospital Billing Supervisor, Yadi Castillo, regularly tracked and circulated information to the business office staff regarding unbilled claims that needed to be billed. Ms. Castillo followed up to ensure that these claims were promptly billed." *See* Wheeler Declaration at ¶ 11. Prime's collectors also "did follow up with ancillary departments to obtain payment on the Accounts Receivable" and "regularly sought support from Prime's clinical, coding, and case management teams to assist with collection efforts." *See id.* at ¶ 12. The Debtors fail to address this detailed evidence.

36. Instead, the Debtors rely on the Chadwick Declaration to speculate and accuse Prime
of bad faith. However, such accusations are wholly unfounded and should not be relied upon by this
Court. Indeed, Mr. Chadwick does not appear to have any direct knowledge of Prime's collection
efforts and instead seems to be relying on second- and third-hand accounts. For example, despite
Mr. Chadwick's descriptions of comments attributed to Ana Goff, Ms. Goff does not even know who
Mr. Chadwick is and has never met or spoken with him. *See* Goff Declaration at ¶ 5. As a result,
Mr. Chadwick has no personal knowledge of any comments attributed to Ms. Goff.

37. Nonetheless, Mr. Chadwick in paragraph 17 of his declaration asserts that
conversations with Ms. Goff "affirmed the deliberate reduction in collection efforts" and that "there
were system issues preventing her team's ordinary-course follow up." This is inaccurate and appears

#### Case 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Main Document Page 18 of 51

to mischaracterize a communication Ms. Goff had with Regina Hernandez, Interim Vice President of Revenue Cycle for Verity Health System, on November 19, 2020, not Mr. Chadwick. *See id.* at ¶ 6.

38. In particular, Ms. Hernandez, on November 19, 2020, inquired as to why cash postings were not appearing in MS4 for the last few days. *See id*. Ms. Goff explained that her team was unable to update the cash postings in MS4 for a few days because of a printer connectivity issue. *See id*. The problem was that there was no direct electronic interface between the electronic remittance advice payment records the hospital was receiving and Sellers' legacy EMR system, MS4. *See id*. Therefore, Ms. Goff's team was required to print out all the payment records and manually scan them into MS4 to reflect the payments. *See id*. The printers in the hospital business office were temporarily offline for a few days and therefore cash postings were delayed in MS4 for a few days. *See id*. The delay in updating MS4 with cash postings, however, in no way impacted actual collections, which were timely pursued. *See id*. Thus, the alleged "systems issues" did not prevent Prime from ordinary-course follow-up on collection of the Accounts Receivable. *See id*.

39. Mr. Chadwick also asserts that Ms. Goff "indicated that collections people had been pulled away from Verity A/R to focus on a Prime audit" and that she "stated that the impact was effectively two collectors working five days a week." See Chadwick Declaration at ¶ 17. Once again, Mr. Chadwick mischaracterizes a communication Ms. Goff had with Ms. Hernandez on November 19, 2020, not Mr. Chadwick. See Goff Declaration at ¶ 7. In particular, Ms. Goff explained to Ms. Hernandez that two of the collectors on site at the St. Francis Medical Center business office had been redirected for a limited period of five working days to assist with an audit. See id. However, this temporary shifting of onsite resources did not impact actual collections because Mr. Wheeler added additional resources from his corporate team located in Ontario, California to maintain staffing levels and prevent any delay in following up on collection of the Accounts Receivable. See id. The Wheeler Declaration previously addressed this point and confirmed that "when the hospital needed additional staff, [Mr. Wheeler] reassigned other business staff from the Prime corporate team to assist." See Wheeler Declaration at ¶ 8.

### Case 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Main Document Page 19 of 51

40. 1 Finally, Mr. Chadwick alleges that Ms. Goff "also stated that she had staffing issues 2 and had insufficient staff to assist her with the legacy accounts." See Chadwick Declaration at ¶ 17. 3 Such assertion is simply false. See Goff Declaration at ¶ 8. Ms. Goff never made such a statement, 4 or any statement at all, to Mr. Chadwick, and she has no recollection of making such a statement to 5 Ms. Hernandez or anyone else. See id. As it is true that certain staff became ill or needed to 6 quarantine due to potential exposure to COVID-19 at various times, which created operational 7 challenges with maintaining staffing levels, it is possible that Ms. Goff discussed the challenges of 8 managing a business office at the height of the COVID-19 pandemic in the fall of 2020. See id. 9 These challenges, however, are no different than those that Ms. Goff has routinely addressed related to managing teams of collectors for the past 30 years. See id. Moreover, when Ms. Goff requested 10 11 additional staffing resources at SFMC, she received support from Mr. Wheeler's corporate collections team located in Ontario, California. See id.

41. Indeed, Ms. Goff was provided ample support in the form of additional staffing from the Prime corporate team to assist with collection of the Accounts Receivable. *See id.* at  $\P$  9. Such additional resources included staff who assisted with drafting clinical and coding appeals throughout the Reconciliation Period. *See id.* Indeed, the staffing devoted to collecting the Accounts Receivable was more than sufficient and at least consistent with industry and Prime's standards. *See id.* 

42. Thus, Mr. Chadwick's assertion that Ms. Goff affirmed Prime's "deliberate reduction in collection efforts" is categorically false. *See id.* at ¶ 10. To the contrary, the entire Prime team including Ms. Goff, the local onsite team at SFMC, and the corporate team in Ontario, California, all worked tirelessly to collect as much as possible from the Accounts Receivable. *See id.* Indeed, the Prime team put in long hours, including overtime, late nights, and weekends throughout the Reconciliation Period (which efforts continue today), to collect as many dollars as possible for SFMC. *See id.* Prime worked just as hard, if not harder, to collect the Accounts Receivable as they continue to work on current collections for the hospital. *See id.* 

43. Rather than providing evidence that Prime did not comply with the collection standard
set forth in the APA, the Debtors instead seek to alter the standard applicable under the APA to
require that Prime's collection rate be consistent with, or exceed, Sellers' historical collection rates.

#### Case 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Main Document Page 20 of 51

1

2

9

22

23

This is an arbitrary benchmark that should not be adopted as it is not consistent with the clear terms of the APA. Instead, the APA is clear that Prime was required to use good faith, commercially 3 reasonable best efforts, including at least the efforts used by Prime to collect its other receivables. 4 See APA §1.12(e). The Debtors fail to present any argument or evidence that Sellers' historic 5 collection rates somehow set the standard for commercial reasonableness or Prime's own collection 6 efforts for other receivables. Indeed, the evidence before this Court demonstrates that Prime "utilized 7 commercially reasonable best efforts to collect the Accounts Receivable and greater efforts than Prime has used to collect its other receivables." See Aleman Declaration at ¶ 4; see also Wheeler 8 Declaration at  $\P$  8; Goff Declaration at  $\P$  10.

10 44. Furthermore, the Debtors' calculation of Prime's collection rate on the Accounts Receivable as being only 78.1% lacks sufficient support or explanation. See Objection, Ex. F. 11 12 According to Mr. Chadwick's Declaration, this calculation is based on "Debtor financials that show accounts receivable outstanding as of the Closing of \$41,915,599 ... compared to Prime's contention 13 14 of having collected \$32,736,688." See Chadwick Declaration at ¶ 19. But Mr. Chadwick's 15 Declaration fails to provide any explanation for the denominator value of \$41,915,599 and instead simply attaches Exhibit F, which is described as a "copy of SFMC's historical vs. Prime's actual A/R 16 17 collections." See id. Exhibit F to the Objection merely asserts that the "Net A/R at 8/13, excluding 18 Trauma" is \$41,915,599, without explanation or any supporting calculation that ties to this figure. 19 See Objection, Ex. F; Supplemental Aleman Declaration at ¶ 4. By contrast, Prime provided a 20 detailed calculation with supporting charts and explanation to show that it actually increased the average collection rate for accounts it collected over Sellers.<sup>8</sup> See Aleman Declaration at ¶¶ 7-8. 21

24 <sup>8</sup> Debtors do not dispute Prime's calculation showing that it increased average collection rates, but instead contend that the calculation should be discounted because "Prime does not explain why a 25 comparison of the first four months of 2020 to the next four months has any relevance." See Objection at p. 20, lines 10-11. However, Prime previously explained that "[t]his is significant 26 because collections in these later months are payments for claims Prime was responsible for following up on and collecting during the Reconciliation Period" and "[e]ven after only a few 27 months of collection efforts by Prime, the average collection rate for August 1 through 13, 2020 is

<sup>28</sup> higher than the average collection rate for January of 2020." See Aleman Declaration at ¶ 8.

45. 1 Regardless, Debtors' calculation of Sellers' historical collection rate as 90.3% is based 2 on a 39-month period ranging from July 2016 to September 2019, which is not directly applicable to 3 the collectability of the Accounts Receivable during the Reconciliation Period in the fall of 2020. 4 See Objection, Ex. E; Supplemental Aleman Declaration at ¶ 5. In fact, the collection rates during 5 the 39-month period relied upon by Debtors ranged from 70% to 98%, demonstrating that a broad 6 range of collection rates existed and are commercially reasonable based on historical experience. See 7 Objection, Ex. E; Supplemental Aleman Declaration at ¶ 5. Thus, even if the Debtors' calculation of Prime's collection rate on the Accounts Receivable was correct, which it is not, this still would 8 9 not show that Prime's collection efforts fell below the standard required in the APA.

# 10

D.

11 12

13

14

16

17

18

19

20

22

23

24

25

26

## IF THE COURT FINDS THAT PRIME DID NOT DEVOTE SUFFICIENT EFFORTS TO COLLECT THE ACCOUNTS RECEIVABLE, THEN AN EVIDENTIARY HEARING WOULD BE REQUIRED TO DETERMINE DAMAGES.

46. As explained above, the Debtors' calculation of historic collection rates is inaccurate and their reliance on this benchmark is inappropriate. As such, the Debtors' conclusion that "if Prime had merely collected the Accounts Receivable at SFMC's historical rates, an additional \$5.1 million 15 would have been collected" is also unreliable and not the appropriate measure of damages, if any. See Objection p. 20, lines 12-14. If this Court were to determine that Prime's collection efforts did not comply with section 1.12(e) of the APA, Prime would request an additional hearing to properly assess and determine any alleged damages and allow the Court an opportunity to address and validate the data provided by each side. Without such a hearing, any damages calculation would be speculative and unsupported by sufficient evidence.

21

E.

# PRIME HAS TURNED OVER THE FUNDS TO WHICH SELLERS ARE ENTITLED.

Finally, Debtors complain about Prime's withholding of certain QAF funds which 47. were Excluded Assets. See Objection p. 18, lines 1-4. But Debtors concede that Prime had already transferred all of the QAF V funds which were Excluded Assets.<sup>9</sup> See Objection p. 2, fn 1; id. p. 9, fn 28; see also Supplemental Aleman Declaration at ¶ 6. With regard to QAF VI funds, Debtors acknowledge that "[t]he APA provides Prime with a limited offset right against Seller's QAF VI

27

<sup>&</sup>lt;sup>9</sup> Debtors also acknowledge that Prime transferred an additional \$1,277,821.50 in disproportionate share payments. See Objection at p. 12, fn 38; see also Supplemental Aleman Declaration at ¶ 6.

### Case 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Main Document Page 22 of 51

Seller Net Payments to the extent the Final A/R Collected is less than the A/R Target Amount." Objection at p.17, lines 19-20; see also APA § 1.12(d)(ii). As of today, Prime is withholding approximately \$24.1 million in QAF VI Seller Net Payments pursuant to its offset rights under the APA because the Final A/R Collect is less than the A/R Target Amount by \$28,263,312. See Supplemental Aleman Declaration at ¶ 7. Given that there is an ongoing dispute between the Parties regarding the amount owed to Prime in which Debtors concede that Prime is entitled to at least \$11.3 million while Prime has presented evidence of entitlement to nearly \$28.3 million, Prime is entitled to withhold the \$24.1 million in QAF VI Seller Net Payments. See Objection at p. 3, lines 5-15. Therefore, Prime has already turned over the funds to which Sellers are entitled. See Supplemental Aleman Declaration at ¶¶ 6-7. 

## Case 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Main Document Page 23 of 51

1	WHEREFORE, Prime respectfully requests that the Bankruptcy Court enter an order (i)
2	determining that the Final A/R Collected Amount was \$32,736,688; (ii) requiring Sellers to pay the
3	\$28,263,312 difference between what Prime paid for the Accounts Receivable and the Final A/R
4	Collected within 10 business days from the entry of such order; (iii) authorizing Prime to offset the
5	Offset Amount against Sellers QAF VI Seller Net Payments in the event that Sellers fail to pay such
6	amount within 10 business days from the entry of such order; and (iv) declaring Prime the prevailing
7	party entitled to recover its court costs and reasonable attorneys' fees under Section 12.12 of the
8	APA.
9	Dated: September 28, 2021
10	Los Angeles, California
11	McDermott Will & Emery, LLP
12	By: <u>/s/ Jason D. Strabo</u> Jason D. Strabo (SBN 246426)
13	2049 Century Park East, Suite 3200 Los Angeles, California 90067-3218
14	Telephone: (310) 788-4125 Facsimile: (310) 277-4730
15	E-mail: jstrabo@mwe.com
16	- and –
17	James W. Kapp, III (admitted <i>pro hac vice</i> ) 444 West Lake Street, Suite 4000
18	Chicago, IL 60606-0029 Telephone: (312) 372-2000
19	Facsimile: (312) 984-7700 E-mail: jkapp@mwe.com
20	
21	Attorneys for Prime Healthcare Services, Inc.
22	
23	
24	
25	
26	
27	
28	19

Case 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Main Document Page 24 of 51

# **PROOF OF SERVICE OF DOCUMENT**

I am over the age of 18 and not a party to this bankruptcy case or adversary proceeding. My business address is: Jason D. Strabo, McDermott Will & Emery LLP

2049 Century Park East, Suite 3200

Los Angeles, California 90067-3206

A true and correct copy of the foregoing document entitled (*specify*): <u>Prime Healthcare Services</u>, Inc.'s Reply to Post-Effective Date Debtors and Liquidating Trustee's Memorandum in Opposition to Prime Healthcare Services, Inc.'s Motion to Enforce Provisions of the Asset Purchase Agreement Pertaining to Accounts Receivable Adjustment; Supplemental Declaration of A. Joel Richlin in Support Thereof; Declaration of Ana Goff in Support Thereof; Supplemental Declaration of Steve Aleman in Support Thereof

will be served or was served (a) on the judge in chambers in the form and manner required by LBR 5005-2(d); and (b) in the manner stated below:

1. <u>TO BE SERVED BY THE COURT VIA NOTICE OF ELECTRONIC FILING (NEF)</u>: Pursuant to controlling General Orders and LBR, the foregoing document will be served by the court via NEF and hyperlink to the document. On (*date*) <u>9/28/2021</u>, I checked the CM/ECF docket for this bankruptcy case or adversary proceeding and determined that the following persons are on the Electronic Mail Notice List to receive NEF transmission at the email addresses stated below:

Service information continued on attached page

#### 2. SERVED BY UNITED STATES MAIL:

On *(date)*, I served the following persons and/or entities at the last known addresses in this bankruptcy case or adversary proceeding by placing a true and correct copy thereof in a sealed envelope in the United States mail, first class, postage prepaid, and addressed as follows. Listing the judge here constitutes a declaration that mailing to the judge <u>will be completed</u> no later than 24 hours after the document is filed.

Service information continued on attached page

#### 3. SERVED BY PERSONAL DELIVERY, OVERNIGHT MAIL, FACSIMILE TRANSMISSION OR EMAIL (state method

for each person or entity served): Pursuant to F.R.Civ.P. 5 and/or controlling LBR, on (*date*) <u>9/28/2021</u>, I served the following persons and/or entities by personal delivery, overnight mail service, or (for those who consented in writing to such service method), by facsimile transmission and/or email as follows. Listing the judge here constitutes a declaration that personal delivery on, or overnight mail to, the judge <u>will be completed</u> no later than 24 hours after the document is filed.

The Honorable Ernest M. Robles (overnight delivery) United States Bankruptcy Court, Edward R. Roybal Federal Building 255 E. Temple Street, Suite 1560 Los Angeles, California 90012

Samuel R. Maizel (via e-mail) Tania M. Moyron (via e-mail) Roger K. Heidenreich (via e-mail) Stephen J. O'Brien (via e-mail) DENTONS US LLP 601 South Figueroa Street, Suite 2500 Los Angeles, California 90017-5704

This form is mandatory. It has been approved for use by the United States Bankruptcy Court for the Central District of California.

Case 2:18-bk-20151-ER	Doc 6669	Filed 09/28/21		Entered 09/28/21 18:34:40	Desc
	Main Doc	ument	Page 2	25 of 51	

samuel.maizel@dentons.com tania.moyron@dentons.com roger.heidenreich@dentons.com stephen.obrien@dentons.com Counsel to Post-Effective Date Debtors and Liquidating Trustee

Alvin Mar Hatty K. Yip Office of the United States Trustee 915 Wilshire Blvd., Suite 1850 Los Angeles, California 90017 alvin.mar@usdoj.gov hatty.yip@usdoj.gov

Service information continued on attached page

I declare under penalty of perjury under the laws of the United States that the foregoing is true and correct.

9/28/2021Jason Strabo/s/ Jason StraboDatePrinted NameSignature

This form is mandatory. It has been approved for use by the United States Bankruptcy Court for the Central District of California.

# Mailing Information for Case 2:18-bk-20151-ER

# **Electronic Mail Notice List**

The following is the list of **parties** who are currently on the list to receive email notice/service for this case.

- Alexandra Achamallah aachamallah@milbank.com, rliubicic@milbank.com
- Melinda Alonzo ml7829@att.com
- LakeForestBankruptcy@jubileebk.net, Anerio V Altman lakeforestpacer@gmail.com
- ramkraut@foxrothschild.com • Robert N Amkraut
- Kyra E Andrassy kandrassy@swelawfirm.com, lgarrett@swelawfirm.com;gcruz@swelawfirm.com;jchung@swelawfirm.com
- Armando V Arballo aarballo@bgrfirm.com, aporcellino@bgrfirm.com
- saron@wrslawyers.com, eweiman@wrslawyers.com Simon Aron
- Lauren T Attard lattard@bakerlaw.com, agrosso@bakerlaw.com
- allison@claimsrecoveryllc.com Allison R Axenrod
- kbanner@greenbergglusker.com, Keith Patrick Banner sharper@greenbergglusker.com;calendar@greenbergglusker.com
- steve@jrgattorneys.com, anushree@jrgattorneys.com • Stephan A Barber
- Cristina E Bautista cristina.bautista@kattenlaw.com, ecf.lax.docket@kattenlaw.com
- William C Beall will@beallandburkhardt.com, carissa@beallandburkhardt.com
- James Cornell Behrens jbehrens@milbank.com, gbray@milbank.com;mshinderman@milbank.com;dodonnell@milbank.com;jb rewster@milbank.com;JWeber@milbank.com
- jbeiswenger@omm.com, jacob-beiswenger-• Jacob Beiswenger 5566@ecf.pacerpro.com;swarren@omm.com
- Bruce Bennett bbennett@jonesday.com
- Peter J Benvenutti pbenvenutti@kellerbenvenutti.com, pjbenven74@yahoo.com
- lberkoff@moritthock.com, hmay@moritthock.com Leslie A Berkoff
- sberman@slk-law.com, • Steven M Berman awit@shumaker.com;bgasaway@shumaker.com
- efile@sfblaw.com • Stephen F Biegenzahn
- tbisconti@bienertkatzman.com, Anthony Bisconti 4579179420@filings.docketbird.com;chowland@bienertkatzman.com
- seb@blakeleyllp.com, ecf@blakeleyllp.com • Scott E Blakelev

- Karl E Block kblock@loeb.com, jvazquez@loeb.com;ladocket@loeb.com;kblock@ecf.courtdrive.com
- Matthew I Bobb mbobb@HuntonAK.com, lgomez@HuntonAK.com;amckenzie@huntonak.com
- J Scott Bovitz bovitz@bovitz-spitzer.com
- **Dustin P Branch** branchd@ballardspahr.com, carolod@ballardspahr.com;hubenb@ballardspahr.com
- Michael D Breslauer mbreslauer@swsslaw.com, wyones@swsslaw.com;mbreslauer@ecf.courtdrive.com;wyones@ecf.courtdriv e.com
- Kayla D Britton kayla.britton@faegredrinker.com
- Chane Buck cbuck@jonesday.com
- Lori A Butler butler.lori@pbgc.gov, efile@pbgc.gov
- Howard Camhi hcamhi@mrllp.com, bankruptcy@mrllp.com;camhihr98234@notify.bestcase.com;wgenovese@mrll p.com;jkissinger@mrllp.com
- Kathryn M.S. Catherwood kcatherwood@grsm.com, mbrookman@grsm.com;sdurazo@grsm.com;jswanson@grams.com
- Jeffrey D Cawdrey jcawdrey@grsm.com, madeyemo@gordonrees.com;sdurazo@grsm.com
- Barry A Chatz barry.chatz@saul.com, jurate.medziak@saul.com
- Shirley Cho scho@pszjlaw.com
- Shawn M Christianson cmcintire@buchalter.com, schristianson@buchalter.com
- Louis J. Cisz lcisz@nixonpeabody.com, jzic@nixonpeabody.com
- Jerome S Cohen jsc@jscbklaw.com
- Leslie A Cohen leslie@lesliecohenlaw.com, jaime@lesliecohenlaw.com;olivia@lesliecohenlaw.com
- Marcus Colabianchi mcolabianchi@duanemorris.com
- Claudia Coleman ccoleman@marshackhays.com, ccoleman@ecf.courtdrive.com
- **Derrick F Coleman** derrick@colemanfrost.com, melissa@colemanfrost.com;marissa@colemanfrost.com
- Kevin Collins kevin.collins@btlaw.com, Tabitha.davis@btlaw.com
- Joseph Corrigan Bankruptcy2@ironmountain.com
- David N Crapo dcrapo@gibbonslaw.com, elrosen@gibbonslaw.com
- Mariam Danielyan md@danielyanlawoffice.com, danielyan.mar@gmail.com
- Brian L Davidoff bdavidoff@greenbergglusker.com, calendar@greenbergglusker.com;jking@greenbergglusker.com
- Aaron Davis aaron.davis@bryancave.com, kat.flaherty@bryancave.com

- Lauren A Deeb lauren.deeb@nelsonmullins.com, maria.domingo@nelsonmullins.com
- Daniel Denny ddenny@milbank.com
- Kerry L Duffy kduffy@bzbm.com, cchou@bzbm.com
- Anthony Dutra adutra@hansonbridgett.com
- Kevin M Eckhardt legal@distressed-debt-investing.com
- Lei Lei Wang Ekvall lekvall@swelawfirm.com, lgarrett@swelawfirm.com;gcruz@swelawfirm.com;jchung@swelawfirm.com
- David K Eldan david.eldan@doj.ca.gov, cynthia.gomez@doj.ca.gov
- Andy J Epstein taxcpaesq@gmail.com
- Richard W Esterkin richard.esterkin@morganlewis.com
- Christine R Etheridge christine.etheridge@ikonfin.com
- John D Faucher jdf@johndfaucher.com, FaucherECF@gmail.com
- **M Douglas Flahaut** flahaut.douglas@arentfox.com
- Michael G Fletcher mfletcher@frandzel.com, sking@frandzel.com
- Alan W Forsley alan.forsley@flpllp.com, awf@fkllawfirm.com,awf@fllawyers.net,addy.flores@flpllp.com
- Joseph D Frank jfrank@fgllp.com, mmatlock@fgllp.com;csmith@fgllp.com;jkleinman@fgllp.com;csucic@fgllp.c om
- William B Freeman bill.freeman@kattenlaw.com, nicole.jones@kattenlaw.com,ecf.lax.docket@kattenlaw.com
- Eric J Fromme efromme@tocounsel.com, stena@tocounsel.com
- Amir Gamliel amir-gamliel-9554@ecf.pacerpro.com, cmallahi@perkinscoie.com;DocketLA@perkinscoie.com
- Norma V Garcia ngarciaguillen@garciarainey.com
- Jeffrey K Garfinkle jgarfinkle@buchalter.com, docket@buchalter.com;dcyrankowski@buchalter.com
- Thomas M Geher tmg@jmbm.com, bt@jmbm.com;fc3@jmbm.com;tmg@ecf.inforuptcy.com
- Evelina Gentry evelina.gentry@akerman.com, rob.diwa@akerman.com
- Lawrence B Gill lgill@nelsonhardiman.com, lcompton@nelsonhardiman.com;rpatel@nelsonhardiman.com;rrange@nelsonh ardiman.com
- Bernard R Given bgiven@loeb.com, mortiz@loeb.com;ladocket@loeb.com;bgiven@ecf.courtdrive.com
- Paul R. Glassman pglassman@sycr.com
- Aaron Matthew Gober-Sims aarongobersims@paulhastings.com
- Matthew A Gold courts@argopartners.net
- Eric D Goldberg eric.goldberg@dlapiper.com, eric-goldberg-1103@ecf.pacerpro.com

- Marshall F Goldberg mgoldberg@glassgoldberg.com, jbailey@glassgoldberg.com
- Eric S Goldstein egoldstein@goodwin.com, bankruptcy@goodwin.com
- Richard H Golubow rgolubow@wghlawyers.com, jmartinez@wghlawyers.com;mweinberg@wghlawyers.com
- Vincent A Gorski vgorski@thegorskifirm.com, ATTY VGORSKI@trustesolutions.com
- Arnold L Graff agraff@wrightlegal.net, bkudgeneralupdates@wrightlegal.net,jcraig@wrightlegal.net
- Barbara R Gross barbara@bgross.law, luz@bgross.law
- David M. Guess guessd@gtlaw.com
- Anna Gumport agumport@sidley.com, laefilingnotice@sidley.com;annagumport-6608@ecf.pacerpro.com
- Ajay K Gupta ag@socal.law, ajay@guptalc.com
- Mary H Haas maryhaas@dwt.com
- Chad V Haes chaes@marshackhays.com, chaes@ecf.courtdrive.com;cmendoza@ecf.courtdrive.com;cmendoza@marsha ckhays.com;kfrederick@ecf.courtdrive.com
- Craig N Haring charing@blankrome.com
- Melissa T Harris harris.melissa@pbgc.gov, efile@pbgc.gov
- Brian T Harvey bharvey@buchalter.com, IFS filing@buchalter.com;dbodkin@buchalter.com
- James A Hayes jhayes@zinserhayes.com, jhayes@ecf.courtdrive.com
- Michael S Held mheld@jw.com
- Lawrence J Hilton lhilton@onellp.com, lthomas@onellp.com,info@onellp.com,lhyska@onellp.com,nlichtenberger@o nellp.com,alooper@onellp.com,jburr@onellp.com
- Robert M Hirsh rhirsh@lowenstein.com
- Florice Hoffman fhoffman@socal.rr.com, floricehoffman@gmail.com
- Lee F Hoffman leehoffmanjd@gmail.com, lee@fademlaw.com
- Marshall J Hogan mhogan@swlaw.com, knestuk@swlaw.com
- Michael Hogue hoguem@gtlaw.com, SFOLitDock@gtlaw.com;navarrom@gtlaw.com
- Matthew B Holbrook mholbrook@sheppardmullin.com, amartin@sheppardmullin.com
- **David I Horowitz** david.horowitz@kirkland.com, keith.catuara@kirkland.com;terry.ellis@kirkland.com;elsa.banuelos@kirkland. com;ivon.granados@kirkland.com
- Virginia Hoyt scif.legal.bk@scif.com
- Brian D Huben hubenb@ballardspahr.com, carolod@ballardspahr.com;rev\_jarushewskyj@ballardspahr.com

- Joan Huh joan.huh@cdtfa.ca.gov
- **Carol A Igoe** cigoe@calnurses.org, ttschneaux@calnurses.org
- Benjamin Ikuta bikuta@hml.law
- William E Ireland wireland@hbblaw.com, edocs@hbblaw.com
- Peter L Isola PIsola@hinshawlaw.com, CKingsley@hinshawlaw.com
- Lawrence A Jacobson laj@cohenandjacobson.com
- Eamon Jafari jafari@barringtonlegal.com, gould@barringtonlegal.com
- John Mark Jennings johnmark.jennings@kutakrock.com, mary.clark@kutakrock.com
- Monique D Jewett-Brewster mjb@hopkinscarley.com, eamaro@hopkinscarley.com
- Crystal Johnson M46380@ATT.COM
- Jeff D Kahane jkahane@duanemorris.com, dmartinez@duanemorris.com
- Steven J Kahn skahn@pszyjw.com
- Cameo M Kaisler salembier.cameo@pbgc.gov, efile@pbgc.gov
- Ivan L Kallick ikallick@manatt.com, ihernandez@manatt.com
- Ori Katz okatz@sheppardmullin.com, lsegura@sheppardmullin.com
- Steven J. Katzman SKatzman@bienertkatzman.com, admin@bienertkatzman.com;chowland@bienertkatzman.com;4579179420@fil ings.docketbird.com
- Sweeney Kelly kelly@ksgklaw.com
- Gerald P Kennedy gerald.kennedy@procopio.com, kristina.terlaga@procopio.com;calendaring@procopio.com;efilebank@procopio.com
- Payam Khodadadi pkhodadadi@mcguirewoods.com, dkiker@mcguirewoods.com
- Christopher M Kiernan chris@talkovlaw.com, talkovlaw@ecf.courtdrive.com
- Christian T Kim ckim@dumas-law.com, ckim@ecf.inforuptcy.com
- Jane Kim jkim@kellerbenvenutti.com
- Monica Y Kim myk@lnbyb.com, myk@ecf.inforuptcy.com
- Alan M Kindred akindred@leechtishman.com, alankindred@hotmail.com;dtomko@leechtishman.com;challer@leechtishman. com
- Benjamin R King bking@loeb.com, karnote@loeb.com;ladocket@loeb.com;bking@ecf.courtdrive.com
- Meredith King kingm@higgslaw.com, malavarj@higgslaw.com
- Gary E Klausner gek@lnbyb.com
- David A Klein david.klein@kirkland.com
- Nicholas A Koffroth nkoffroth@foxrothschild.com
- Jolene E Kramer jkramer@unioncounsel.net, jkramer@unioncounsel.net

- Chet Kronenberg , justin.andrei@stblaw.com,lwashienko@stblaw.com
- Chet Kronenberg ckronenberg@stblaw.com, justin.andrei@stblaw.com,lwashienko@stblaw.com
- David S Kupetz dkupetz@sulmeyerlaw.com, dperez@sulmeyerlaw.com;dperez@ecf.courtdrive.com;dkupetz@ecf.courtdriv e.com
- Jeffrey S Kwong jsk@lnbyb.com, jsk@ecf.inforuptcy.com
- Tarifa Laddon tarifa.laddon@faegredrinker.com, Lorena.lazheztter@faegredrinker.com
- Robert S Lampl advocate45@aol.com, rlisarobinsonr@aol.com
- Ian Landsberg ilandsberg@sklarkirsh.com, lskaist@sklarkirsh.com;yalarcon@sklarkirsh.com;mmadden@sklarkirsh.com;il andsberg@ecf.inforuptcy.com;kfrazier@sklarkirsh.com;cbullock@sklarkirsh.c om
- Richard A Lapping richard@lappinglegal.com
- Paul J Laurin plaurin@btlaw.com,
- slmoore@btlaw.com;jboustani@btlaw.com;jose.barajas@btlaw.com
- Dan Lawton dlawton@klinedinstlaw.com, lparrish@klinedinstlaw.com
- Marc Y Lazo mlazo@kllawgroup.com, rcantillo@kllawgroup.com
- Nathaniel M Leeds nathaniel@mitchelllawsf.com, sam@mitchelllawsf.com
- Paul J Leeds leedsp@higgslaw.com, begaym@higgslaw.com
- David E Lemke david.lemke@wallerlaw.com, chris.cronk@wallerlaw.com;Melissa.jones@wallerlaw.com;cathy.thomas@wal lerlaw.com
- Danielle R Leneck dleneck@jonesday.com, eburnside@jonesday.com;cjlovrien@jonesday.com;joelnavarro@jonesday.com ;cbaker@jonesday.com;atrossum@jonesday.com;mmelvin@jonesday.com;cali forniadockets@jonesday.com
- Lisa Lenherr llenherr@wendel.com, bankruptcy@wendel.com
- Matthew A Lesnick matt@lesnickprince.com, matt@ecf.inforuptcy.com;jmack@lesnickprince.com
- Elan S Levey elan.levey@usdoj.gov, julie.morales@usdoj.gov
- Alexander J Lewicki alewicki@diemerwei.com
- Wendy A Loo wendy.loo@lacity.org
- Kerri A Lyman klyman@steptoe.com, #-FirmPSDocketing@Steptoe.com;nmorneault@Steptoe.com;mhernandez@stept oe.com;aodonnell@steptoe.com
- **Tracy L Mainguy** bankruptcycourtnotices@unioncounsel.net, tmainguy@unioncounsel.net
- **Samuel R Maizel** samuel.maizel@dentons.com, alicia.aguilar@dentons.com;docket.general.lit.LOS@dentons.com;tania.moyro

n@dentons.com; kathryn.howard@dentons.com; joan.mack@dentons.com; derry .kalve@dentons.com

- Lloyd S Mann lmann@mannzarpas.com
- Alvin Mar alvin.mar@usdoj.gov, dare.law@usdoj.gov
- Craig G Margulies Craig@MarguliesFaithlaw.com, Vicky@MarguliesFaithlaw.com;Helen@MarguliesFaithlaw.com;Angela@MarguliesFaithlaw.com
- Robert S Marticello Rmarticello@swelawfirm.com, gcruz@swelawfirm.com;lgarrett@swelawfirm.com;jchung@swelawfirm.com
- **Daniel J McCarthy** dmccarthy@hillfarrer.com, spadilla@hillfarrer.com;nchacon@hfbllp.com
- Kevin Meek kmeek@robinskaplan.com, kevinmeek32@gmail.com;kmeek@ecf.inforuptcy.com
- Hutchison B Meltzer hutchison.meltzer@doj.ca.gov, Alicia.Berry@doj.ca.gov
- Maria Ann Milano mmilano@riddellwilliams.com
- Angela Z Miller amiller@phillipslytle.com, styrone@phillipslytle.com
- Christopher Minier becky@ringstadlaw.com, arlene@ringstadlaw.com
- John A Moe john.moe@dentons.com, glenda.spratt@dentons.com;derry.kalve@dentons.com
- Trey A Monsour tmonsour@foxrothschild.com
- Susan I Montgomery susan@simontgomerylaw.com, assistant@simontgomerylaw.com;simontgomerylawecf.com@gmail.com;mont gomerysr71631@notify.bestcase.com
- Monserrat Morales Monsi@MarguliesFaithLaw.com, Vicky@MarguliesFaithLaw.com;Helen@marguliesfaithlaw.com;Angela@Mar guliesFaithlaw.com
- Kevin H Morse kmorse@clarkhill.com, blambert@clarkhill.com
- Marianne S Mortimer mmartin@jmbm.com
- Tania M Moyron tania.moyron@dentons.com, malka.zeefe@dentons.com;kathryn.howard@dentons.com;derry.kalve@denton s.com;glenda.spratt@dentons.com
- Randall P Mroczynski randym@cookseylaw.com
- Aviram Edward Muhtar aviram.muhtar@lewisbrisbois.com, Nancy.Jasso@lewisbrisbois.com
- Patrick J Mulligan pmulligan@bressler.com
- Benjamin Nachimson ben.nachimson@wnlawyers.com, ben.nachimson@wnlawyers.com
- Alan I Nahmias anahmias@mbnlawyers.com, jdale@mbnlawyers.com
- Akop J Nalbandyan jnalbandyan@LNtriallawyers.com, cbautista@LNtriallawyers.com

- Jennifer L Nassiri jennifernassiri@quinnemanuel.com, bdelacruz@sheppardmullin.com
- Andrew Nazar anazar@polsinelli.com, tbackus@polsinelli.com;docket@polsinelli.com
- Charles E Nelson nelsonc@ballardspahr.com, wassweilerw@ballardspahr.com
- Sheila Gropper Nelson shedoesbklaw@aol.com
- Mark A Neubauer mneubauer@carltonfields.com, mlrodriguez@carltonfields.com;smcloughlin@carltonfields.com;ecfla@carlton fields.com
- Fred Neufeld fneufeld@sycr.com, tingman@sycr.com
- Bryan L Ngo bngo@fortislaw.com, BNgo@bluecapitallaw.com;SPicariello@fortislaw.com;JNguyen@fortislaw.co m;JNguyen@bluecapitallaw.com
- Abigail V O'Brient avobrient@mintz.com, docketing@mintz.com;DEHashimoto@mintz.com;nleali@mintz.com;ABLevin @mintz.com
- John R OKeefe jokeefe@metzlewis.com, slohr@metzlewis.com
- Matthew J Olson olson.matt@dorsey.com, stell.laura@dorsey.com
- Scott H Olson solson@vedderprice.com, scott-olson-2161@ecf.pacerpro.com,ecfsfdocket@vedderprice.com,nortega@vedderprice.c om
- Giovanni Orantes go@gobklaw.com, gorantes@oranteslaw.com,cmh@gobklaw.com,gobklaw@gmail.com,go@ecf.inforuptcy.com;ora ntesgr89122@notify.bestcase.com
- Keith C Owens kowens@foxrothschild.com, khoang@foxrothschild.com
- **R Gibson Pagter** gibson@ppilawyers.com, ecf@ppilawyers.com;pagterrr51779@notify.bestcase.com
- Brian A Paino bpaino@mcglinchey.com, irvineECF@mcglinchey.com
- Joshua K Partington jpartington@swlaw.com, idelgado@swlaw.com
- Paul J Pascuzzi ppascuzzi@ffwplaw.com, docket@ffwplaw.com
- Valerie Bantner Peo vbantnerpeo@buchalter.com
- Lisa M Peters lisa.peters@kutakrock.com, marybeth.brukner@kutakrock.com
- Christopher J Petersen cjpetersen@blankrome.com, gsolis@blankrome.com
- Mark D Plevin mplevin@crowell.com, cromo@crowell.com
- Steven G. Polard steven.polard@ropers.com, calendarlao@ropers.com;keiko.kakiuchi@ropers.com;tiffany.sterling@ropers.com;anth ony.arriola@ropers.com
- David M Powlen david.powlen@btlaw.com, pgroff@btlaw.com

- Christopher E Prince cprince@lesnickprince.com, jmack@lesnickprince.com;cprince@ecf.courtdrive.com;jnavarro@lesnickprinc e.com
- **Douglas B Provencher** dbp@provlaw.com
- Lori L Purkey bareham@purkeyandassociates.com
- Uzzi O Raanan uraanan@DanningGill.com, DanningGill@gmail.com;uraanan@ecf.inforuptcy.com
- William M Rathbone wrathbone@grsm.com, sdurazo@grsm.com
- Jason M Reed Jason.Reed@Maslon.com
- Nathaniel W Reinhardt nreinhardt@bienertkatzman.com, 4579179420@filings.docketbird.com
- Jeffrey M. Reisner jreisner@steptoe.com, #-FirmPSDocketing@Steptoe.com;klyman@steptoe.com;nmorneault@Steptoe.com
- Kenneth R Reynolds krrlaw@sbcglobal.net
- Michael B Reynolds mreynolds@swlaw.com, kcollins@swlaw.com
- Richard J Reynolds rreynolds@bwslaw.com, rjrnef@bwslaw.com,fcabezas@bwslaw.com,jgomez@bwslaw.com;dwetters@ bwslaw.com;bantle@bwslaw.com
- J. Alexandra Rhim arhim@hrhlaw.com
- Emily P Rich erich@unioncounsel.net, bankruptcycourtnotices@unioncounsel.net
- Robert A Rich , candonian@huntonak.com
- Robert A Rich rrich2@huntonak.com, candonian@huntonak.com
- Lesley A Riis lriis@dpmclaw.com
- Debra Riley driley@allenmatkins.com
- Jason E Rios jrios@ffwplaw.com, docket@ffwplaw.com
- Julie H Rome-Banks julie@bindermalter.com
- Mary H Rose mrose@buchalter.com
- **Douglas B Rosner** drosner@goulstonstorrs.com
- Terrel Ross tross@trcmllc.com
- Gregory A Rougeau grougeau@brlawsf.com
- Megan A Rowe mrowe@dsrhealthlaw.com, lwestoby@dsrhealthlaw.com
- Gregory M Salvato gsalvato@salvatoboufadel.com, calendar@salvatolawoffices.com;jboufadel@salvatoboufadel.com;gsalvato@ec f.inforuptcy.com
- Ryan M Salzman ryan.salzman@faegredrinker.com, susan.carlson@faegredrinker.com
- Natalie K Sanders natalie.sanders@bakerbotts.com, anna.tercero@bakerbotts.com
- Scott A Schiff sas@soukup-schiff.com

- Daren M Schlecter daren@schlecterlaw.com, assistant@schlecterlaw.com
- Nathan A Schultz nschultzesq@gmail.com
- Mark A Serlin ms@swllplaw.com, sondra@swllplaw.com
- Isra Shah ishah@rwglaw.com
- Seth B Shapiro seth.shapiro@usdoj.gov
- David B Shemano dshemano@shemanolaw.com
- Joseph Shickich jshickich@riddellwilliams.com
- Mark Shinderman mshinderman@milbank.com, dmuhrez@milbank.com;dlbatie@milbank.com
- Rachelle Singer rsinger@smithsilbar.com, rsinger@smithsilbar.com
- Kyrsten Skogstad kskogstad@calnurses.org, rcraven@calnurses.org
- Andrew Edward Smyth office@smythlo.com
- **Michael St James** ecf@stjames-law.com, stjames.michaelr101488@notify.bestcase.com
- **Peter T Steinberg** mr.aloha@sbcglobal.net
- Joseph L Steinfeld jsteinfeld@askllp.com, lmiskowiec@askllp.com;mudem@askllp.com;bmcgrath@askllp.com;kcasteel @askllp.com
- Andrew Still astill@swlaw.com, kcollins@swlaw.com
- Rachel P Stoian stoian.rachel@dorsey.com, stell.laura@dorsey.com
- Jason D Strabo jstrabo@mwe.com, cgreer@mwe.com
- Sabrina L Streusand Streusand@slollp.com
- Ralph J Swanson ralph.swanson@berliner.com, sabina.hall@berliner.com
- Michael A Sweet msweet@foxrothschild.com, swillis@foxrothschild.com;pbasa@foxrothschild.com
- James M Toma james.toma@doj.ca.gov, teresa.depaz@doj.ca.gov
- **Gary F Torrell** gtorrell@health-law.com
- Levi Reuben Uku levireuben@gmail.com
- Gary D Underdahl , lmiskowiec@askllp.com
- Gary D Underdahl gunderdahl@askllp.com, lmiskowiec@askllp.com
- United States Trustee (LA) ustpregion16.la.ecf@usdoj.gov
- Cecelia Valentine cecelia.valentine@nlrb.gov
- Jason Wallach jwallach@ghplaw.com, g33404@notify.cincompass.com
- Kenneth K Wang kenneth.wang@doj.ca.gov, Jennifer.Kim@doj.ca.gov;Stacy.McKellar@doj.ca.gov;yesenia.caro@doj.ca.go v
- **Phillip K Wang** phillip.wang@rimonlaw.com, tad.prizant@rimonlaw.com;docketing@rimonlaw.com
- Ashley R Wedding awedding@fabozzimillerlaw.com, dsandoval@fabozzimillerlaw.com
- Noah Weingarten nweingarten@loeb.com

- Sharon Z. Weiss sharon.weiss@bclplaw.com, raul.morales@bclplaw.com,REC\_KM\_ECF\_SMO@bclplaw.com
- Joseph M Welch jwelch@buchalter.com, dcyrankowski@buchalter.com;docket@buchalter.com
- Adam G Wentland awentland@tocounsel.com, lkwon@tocounsel.com
- Steven Werth swerth@sulmeyerlaw.com, cblair@sulmeyerlaw.com;mviramontes@sulmeyerlaw.com;dperez@sulmeyerl aw.com;swerth@ecf.inforuptcy.com
- Cranston J Williams cwilliams3@semprautilities.com
- Latonia Williams lwilliams@goodwin.com, bankruptcy@goodwin.com
- Michael S Winsten mike@winsten.com
- **Rebecca J Winthrop** rebecca.winthrop@nortonrosefulbright.com, diana.cardenas@nortonrosefulbright.com
- Jeffrey C Wisler jwisler@connollygallagher.com, dperkins@connollygallagher.com
- Neal L Wolf nwolf@hansonbridgett.com, lchappell@hansonbridgett.com
- Kirsten A Worley WORLEYK@HIGGSLAW.COM, MALAVARJ@HIGGSLAW.COM
- Claire K Wu claire.wu@pillsburylaw.com, irene.hooper@pillsburylaw.com;docket@pillsburylaw.com
- Steven D Wyllie steven.wyllie@nlrb.gov
- Hatty K Yip hatty.yip@usdoj.gov, hatty.k.yip@usdoj.gov
- Bennett G Young byoung@jmbm.com, jb8@jmbm.com
- Andrew J Ziaja aziaja@leonardcarder.com, sgroff@leonardcarder.com;msimons@leonardcarder.com;lbadar@leonardcarde r.com
- Rose Zimmerman rzimmerman@dalycity.org

# SUPPLEMENTAL DECLARATION OF A. JOEL RICHLIN

Desc

I, A. Joel Richlin, declare as follows:

I am over 18 years of age. I have personal knowledge of the facts set forth below and, 1. 4 if called to testify, would and could competently testify thereto.

1

2

3

2. I am the Vice President and General Counsel for Prime Healthcare Services, Inc. ("Prime"). I am licensed to practice law in the State of California and before this Court. I have served as in-house counsel for Prime since April of 2017 and previously served as Deputy General Counsel and Chief Litigation Counsel. I have led all legal aspects of Prime's acquisition of St. Francis Medical Center since Prime's initial bid in the fall of 2018, which was rejected in favor of a different buyer. I was deeply involved in the negotiation of the APA.

I submit this declaration (the "Declaration") in support of Prime Healthcare Services, 3. Inc.'s Reply to Post-Effective Date Debtors and Liquidating Trustee's Memorandum in Opposition to Prime Healthcare Services, Inc.'s Motion to Enforce Provisions of the Asset Purchase Agreement Pertaining to Accounts Receivable Adjustment filed contemporaneously herewith (the "Reply"). Unless otherwise indicated, all capitalized but undefined terms herein shall have the same meanings ascribed to them in the Reply.

4. Pursuant to the Bid Procedures Order entered by this Court, Prime and Sellers negotiated the APA throughout March and early April of 2020, which coincided with the beginning of the COVID-19 pandemic in the United States. The pandemic created significant uncertainty as to the operations of SFMC and the valuations on which Prime had relied in pursuing the SFMC Sale. In fact, the uncertainty created by the pandemic was so significant and cast such doubt on Sellers' valuation that Prime seriously considered withdrawing from the bid process and/or lowering the purchase price in its bid. Ultimately, no other bidder sought to acquire the Assets, likely because of this uncertainty.

5. To address the uncertainty caused by the pandemic, Prime and Sellers engaged in extensive negotiations to allocate risk between the parties and to adjust to the new economic

SUPPLEMENTAL DECLARATION OF A. JOEL RICHLIN IN SUPPORT OF PRIME HEALTHCARE SERVICES, INC.'S REPLY TO POST-EFFECTIVE DATE DEBTORS AND LIQUIDATING TRUSTEE'S MEMORANDUM IN OPPOSITION TO PRIME HEALTHCARE SERVICES, INC.'S MOTION TO ENFORCE PROVISION OF THE ASSET PURCHASE AGREEMENT PERTAINING TO ACCOUNTS RECEIVABLE ADJUSTMENT

#### Case 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Main Document Page 38 of 51

environment. In particular, Prime sought various items from Sellers in the final week of negotiations 2 that would assuage Prime's concerns and provide it the comfort necessary to close the SFMC Sale. 3 Prime sought these modifications to the APA in exchange for Prime's agreement not to decrease the 4 agreed upon \$200 million purchase price, thereby rendering a benefit to the Debtors and their estates. 5 These negotiated modifications to the APA included the creation of the Accounts Receivable 6 Reconciliation in section 1.12 of the APA and the exclusion of certain items from the definition of 7 the term "Accounts Receivable" in section 1.7(p).

8 6. The negotiations to address the allocation of risk occurred after the creation of the 9 schedule from which the A/R Target Amount was derived. Indeed, this schedule was exchanged 10 during the Parties' due diligence review prior to executing the APA. In allocating the risk in a way 11 to entice Prime to consummate the SFMC Sale, Sellers agreed to a definition of Accounts Receivable 12 disconnected from the line items included in this schedule. This schedule was drafted during and 13 reflects a drastically different economic environment than the one in which the Parties ultimately 14 closed the transaction. Thus, the derivation of the A/R Target Amount is an irrelevant reference point 15 when interpreting the APA's definition of "Accounts Receivable."

7. 16 I believe the intense negotiation as to what items constituted, and what should be 17 excluded from, "Accounts Receivable" is clearly documented by contemporaneous emails and 18 redline drafts of the APA exchanged between Prime and Sellers, which were either sent by me or on which I was copied. The relevant period of negotiation begins on March 26, 2020, when Sellers' counsel sent Prime a redline draft of the APA (the "3/26/2020 Redline"), which added the Accounts Receivable Reconciliation process in Section 1.12 for the first time. A true and correct copy of the 3/26/2020 Redline is attached to the Reply as <u>Exhibit 8</u>. This version of the APA was favorable to Sellers because section 1.7(p) broadly defined all collections as "Accounts Receivable" and credited all these amounts toward the A/R Target Amount:

2 SUPPLEMENTAL DECLARATION OF A. JOEL RICHLIN IN SUPPORT OF PRIME HEALTHCARE SERVICES, INC.'S REPLY TO POST-EFFECTIVE DATE DEBTORS AND LIQUIDATING TRUSTEE'S MEMORANDUM IN OPPOSITION TO PRIME HEALTHCARE

VICES, INC.'S MOTION TO ENFORCE PROVISIONS OF THE ASSET PURCHASE

CCOUNTS RECEIVABLE ADJUSTMENT

28

Case 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Main Document Page 39 of 51

(p) all accounts and interest thereupon, notes and interest thereupon and other receivables of Sellers, including, without limitation, accounts, notes or other amounts receivable, and all claims, rights, interests and proceeds related thereto, including all accounts and other receivables, quality assurance fee payments, trauma payments, disproportionate share payments,

9

US\_Active\114188900114483166\V-64

cost report, claim, EHR or other similar appeals and Seller Cost Report settlements, in each case arising from the rendering of services or provision of goods, products or supplies to inpatients and outpatients at the Hospital, billed and unbilled, recorded and unrecorded (including any such amounts that were written-off by Sellers for any reason), for services, goods, products and supplies provided by Sellers prior to the Licensure Date whether payable by Medicare, Medi-Cal, Medicaid, or any other payor (including an insurance company), or any health care provider or network (such as a health maintenance organization, preferred provider organization or any other managed care program) or any fiscal intermediary of the foregoing, private pay patients, private insurance or by any other source (collectively, "Accounts Receivable");

8. Prime was unwilling to accept this language in light of the financial risk posed by the pandemic and recent announcements that the Federal Government would be providing significant financial support to hospitals to address the pandemic. The amounts and rules related to these government payments were unknown at the time, which created additional uncertainty in the ability of both sides to negotiate over which collections should qualify as Accounts Receivable and be included in the A/R Target Amount. Therefore, Prime rejected the language contained in the 3/26/2020 Redline and I personally raised Prime's concerns with Sellers on a phone call on March 31, 2020. I memorialized the discussion in a follow-up email to Jim Moloney of Cain Brothers, the investment banking firm representing Sellers. A true and correct copy of my March 31, 2020 email to Mr. Maloney (the "<u>COVID-19 Grant Email</u>") is attached to the Reply as Exhibit 9. In the COVID-19 Grant Email with the subject line "Call to Discuss COVID 19 between Prime and Verity," I stated to Mr. Maloney as follows: "As discussed on the call, we need to clarify some of the definitions related to AR and government payments going different ways. This is the point I discussed with

SUPPLEMENTAL DECLARATION OF A. JOEL RICHLIN IN SUPPORT OF PRIME HEALTHCARE SERVICES, INC.'S REPLY TO POST-EFFECTIVE DATE DEBTORS AND LIQUIDATING TRUSTEE'S MEMORANDUM IN OPPOSITION TO PRIME HEALTHCARE

SERVICES, INC.'S MOTION TO ENFORCE PROVISIONS OF THE ASSET PURCHASE AGREEMENT PERTAINING TO ACCOUNTS RECEIVABLE ADJUSTMENT

#### Case 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Page 40 of 51 Main Document

1

2

3

4

18

19

20

21

22

23

24

25

26

27

28

Elspeth where Verity excluded additional government payments related to COVID in their last turn and we think those are essential to the value of the asset since those payments will not even make the hospital whole. So we are revising that language to make it as clear as possible because the passage of new laws creating these payments needs to be addressed."

5 9. Following the March 31, 2020 call, I sent Sellers a revised redline of the APA on April 1, 2020 (the "<u>4/1/2020 Redline</u>") with a cover email (the "<u>Revised APA Email</u>") specifically 6 7 calling their attention to changes to the definition of Accounts Receivable. True and correct copies 8 of the 4/1/2020 Redline and the Revised APA Email are attached to the Reply as Exhibit 10 and 9 Exhibit 11, respectively. Indeed, I stated in my email to Sellers, "The key changes I will call to your attention are: (1) no longer pre-paying for QAF (Seller still gets to keep QAF so no change in net 10 value to Seller, but reduced cash consideration at close); (2) revised AR formula to address expected 11 12 COVID losses and supplemental payments (as discussed on our call); (3) compromise language on Material Adverse Effect; and (4) certain bid protections such as minimum overbid and bid 13 14 increments." (emphasis added). This version of the APA removed "trauma payments" from the 15 definition of "Accounts Receivable" in section 1.7(p) and created a new category of collections called "Other Receivables" which included both trauma payments and government payments related to 16 17

COVID-19, and would not count toward meeting the A/R Target Amount:

(p) (i) all accounts and interest thereupon, notes and interest thereupon and other receivables of Sellers, including, without limitation, accounts, notes or other amounts receivable, and all claims, rights, interests and proceeds related thereto, including all accounts and other receivables, quality assurance fee payments, trauma payments, disproportionate share payments, cost report, claim, EHR or other similar appeals and Seller Cost Report settlements, in each case arising from the rendering of services or provision of goods, products or supplies to inpatients and outpatients at the Hospital, billed and unbilled, recorded and unrecorded (including any such amounts that were written-off by Sellers for any reason), for services, goods, products and supplies provided by Sellers prior to the Licensure Date whether payable by Medicare, Medi-Cal, Medicaid, or any other payor (including an insurance company), or any health care provider or network (such as a health maintenance organization, preferred provider organization or any other managed care program) or any fiscal intermediary of the foregoing, private pay patients, private insurance or by any other source (collectively, "Accounts Receivable"); and (ii) all claims, rights, interests and proceeds relating to any grant or governmental awards directly or indirectly related to COVID-19, trauma payments, disproportionate share payments, cost report, claim, EHR or other similar appeals and Seller Cost Report settlements in each case arising from the rendering of services or provision of goods, products or supplies to inpatients and outpatients at the Hospital (the "Other Receivables" and together with the Accounts Receivable, the "Receivables");

4 SUPPLEMENTAL DECLARATION OF A. JOEL RICHLIN IN SUPPORT OF PRIME HEALTHCARE SERVICES, INC.'S REPLY TO POST-EFFECTIVE DATE DEBTORS AND LIQUIDATING TRUSTEE'S MEMORANDUM IN OPPOSITION TO PRIME HEALTHCARE SERVICES, INC.'S MOTION TO ENFORCE PROVISIONS OF THE ASSET PURCHASE AGRÉEMENT PERTAINING TO ACCOUNTS RECEIVABLE ADJUSTMENT

# Case 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Main Document Page 41 of 51

1 10. Later in the day on April 1, 2020, I had a phone call with Sellers' General Counsel, 2 Elspeth Paul, to discuss the 4/1/2020 Redline and the changes to the definition of "Accounts" 3 Receivable" in Section 1.7(p). During the call, both Ms. Paul and myself agreed to speak with our 4 respective management teams about these issues and reconvene in an effort to negotiate a 5 compromise. I later sent an email in the evening of April 1, 2020 (the "APA Conference Email") to Ms. Paul stating as follows: "Elspeth, I spoke to management and have feedback on our discussion. 6 7 Let me know once you have feedback on your end and can talk. I think we need to have a discussion 8 about AR where I need Steve to join and we probably need BRG on your end, or whoever you deem 9 best." A true and correct copy of 4/1/2020 Email is attached to the Reply as Exhibit 12. As reflected later in the email chain, Ms. Paul and I subsequently arranged a group phone call to discuss these 10 11 issues on April 2, 2020.

12 11. Following the group phone call on April 2, 2020, Ms. Paul sent me an email (the 13 "<u>Accounts Receivable Email</u>") expressly agreeing to Prime's new definition of "Accounts 14 Receivable" in section 1.7(p) of the APA which excluded "trauma payments," in exchange for 15 Prime's agreement to move government payments related to COVID-19 out of the definition of 16 "Other Receivables" and into the definition of "Accounts Receivable." A true and correct copy of 17 the Accounts Receivable Email is attached to the Reply as <u>Exhibit 12</u>. Ms. Paul included in her email 18 a snipped image of the language proposed by Sellers for clarity:

 19
 From: Paul, Elspeth <<u>ElspethPaul@verity.org</u>>

 Sent: Thursday, April 2, 2020 3:29 PM

 To: Joel Richlin (PHMI) <<u>JRichlin@primehealthcare.com</u>>

 Subject: EXTERNAL:COVID \$

21

22

28

WARNING: This email originated outside the Prime Healthcare email system! Do Not Click links if this user is unknown.

Joel: With your new definition of accounts receivable and receivables, I'm thinking the edit is simply to move where the COVID \$ is referenced:

(p) (i) all accounts and interest thereupon, notes and interest thereupon and other receivables of Sellers, including, without limitation, accounts, notes or other 23 amounts receivable, and all claims, rights, interests and proceeds related thereto, including all accounts and other receivables, , in each case arising from the rendering of services or provision of goods, products or supplies to inpatients and outpatients at the Hospital, billed and unbilled, recorded and unrecorded 24 (including any such amounts that were written-off by Sellers for any reason), for services, goods, products and supplies provided by Sellers prior to the Licensure Date whether payable by Medicare, Medi-Cal, Medicaid, or any other payor (including an insurance company), or any health care provider or network (such as a health maintenance organization, preferred provider organization or any other managed care program) or any fiscal intermediary of the 25 foregoing, private pay patients, private insurance or by any other source and all claims, rights, interests and proceeds relating to any grant or governmental awards directly or indirectly related to COVID-19 (collectively, "Accounts Receivable"); and (ii) all claims, rights, interests and proceeds relating to any grant or governmental awards directly or indirectly related to COVID-19, trauma payments, disproportionate share payments, cost report, claim, EHR or other similar 26 appeals and Seller Cost Report settlements in each case arising from the rendering of services or provision of goods, products or supplies to inpatients and outpatients at the Hospital (the "Other Receivables" and together with the Accounts Receivable, the "Receivables"); 27

5 SUPPLEMENTAL DECLARATION OF A. JOEL RICHLIN IN SUPPORT OF PRIME HEALTHCARE SERVICES, INC.'S REPLY TO POST-EFFECTIVE DATE DEBTORS AND LIQUIDATING TRUSTEE'S MEMORANDUM IN OPPOSITION TO PRIME HEALTHCARE SERVICES, INC.'S MOTION TO ENFORCE PROVISIONS OF THE ASSET PURCHASE AGREEMENT PERTAINING TO ACCOUNTS RECEIVABLE ADJUSTMENT

## Case 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Main Document Page 42 of 51

1 As shown in the image provided by Ms. Paul, "trauma payments" remain excluded from the 2 definition of "Accounts Receivable," while "governmental awards directly or indirectly related to 3 COVID-19" were moved out of the definition of "Other Receivables" and into the definition of 4 "Accounts Receivable." In the same email chain, which is encompassed within the email attached 5 to the Reply as Exhibit 12, I replied to Ms. Paul's proposed language for section 1.7(p) of the APA 6 stating that Prime would agree if Sellers would also agree to a slight modification of Section 7 1.12(d)(i) to protect Prime from owing Sellers a refund if the A/R Target Amount were met and 8 exceeded based purely on government payments related to COVID-19. Specifically, the language I 9 proposed to add to section 1.12(d)(i) meant that Sellers would get credit for any government 10 payments related to COVID-19 for purposes of meeting the A/R Target Amount, but Prime would 11 be able to remove any such government payments from the calculation of amounts that Prime might 12 owe back to Sellers if the A/R Target Amount were exceeded. Ms. Paul agreed to my proposal later 13 that day, stating, "Looks good. Adding it to the APA."

14 12. On April 2, 2020, I emailed Sellers and their counsel (the "<u>APA Negotiations Email</u>")
15 asking when they would be circulating an updated redline of the APA based on my agreement with
16 Ms. Paul. A true and correct copy of the APA Negotiations Email is attached to the Reply as Exhibit
17 14.

18 13. Specifically, I stated, "My understanding is that Dentons is working on a redline that 19 reflects the parties' negotiations and agreements today. I reviewed the AR language with Elspeth and 20 that should be good to go." (emphasis added). I did not receive a response to my email that same 21 day, so on the morning of April 3, 2020, I replied to my prior email asking for an update from Sellers 22 based on concern that we would miss the deadline set forth in the Bid Procedure Order. Counsel for 23 Sellers, Tania M. Moyron, promptly replied to my April 3, 2020 inquiry stating "Thanks, Joel. 24 Dentons has been brought up to speed on the discussions that occurred with you and others through 25 late yesterday. We are working with our team and Elspeth to incorporate those points into the APA." 26 (emphasis added).

27

28

6 SUPPLEMENTAL DECLARATION OF A. JOEL RICHLIN IN SUPPORT OF PRIME HEALTHCARE SERVICES, INC.'S REPLY TO POST-EFFECTIVE DATE DEBTORS AND LIQUIDATING TRUSTEE'S MEMORANDUM IN OPPOSITION TO PRIME HEALTHCARE SERVICES, INC.'S MOTION TO ENFORCE PROVISIONS OF THE ASSET PURCHASE AGREEMENT PERTAINING TO ACCOUNTS RECEIVABLE ADJUSTMENT Case 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Main Document Page 43 of 51

1 14. Shortly thereafter on April 3, 2020, counsel for Sellers, R. Matthew Garms, circulated 2 a revised redline of the APA (the "<u>4/3/2020 Redline</u>") by email (the "<u>Final APA Email</u>"), in an email 3 on which I was copied, which reflected the agreement between Ms. Paul and myself on the definition 4 of Accounts Receivable in section 1.7(p) and the protection for Prime in section 1.12(d)(i) against 5 including governmental payments related to COVID-19 toward any amount that Prime might have 6 to refund to Sellers if the A/R Target Amount were exceeded:

(i) all accounts and interest thereupon, notes and interest thereupon and (p) other receivables of Sellers, including, without limitation, accounts, notes or other amounts receivable, and all claims, rights, interests and proceeds related thereto, including all accounts and other receivables, in each case arising from the rendering of services or provision of goods, products or supplies to inpatients and outpatients at the Hospital, billed and unbilled, recorded and unrecorded (including any such amounts that were written-off by Sellers for any reason), for services, goods, products and supplies provided by Sellers prior to the Licensure Date whether payable by Medicare, Medi-Cal, Medicaid, or any other payor (including an insurance company), or any health care provider or network (such as a health maintenance organization, preferred provider organization or any other managed care program) or any fiscal intermediary of the foregoing, private pay patients, private insurance or by any other source (collectively, "Accounts-Receivable"); and (ii)and all claims, rights, interests and proceeds relating to any grant or governmental awards directly or indirectly related to COVID-19,19 (collectively, "Accounts Receivable"); and (ii) trauma payments, disproportionate share payments (subject to Section 1.8(c)), California Health Foundation & Trust payments, cost report, claim, EHR or other similar appeals and Seller Cost Report settlements in each case arising from the rendering of services or provision of goods, products or supplies to inpatients and outpatients at the Hospital (the "Other Receivables" and together with the Accounts Receivable, the "Receivables");

True and correct copies of the 4/3/2020 Redline and the Final APA Email are attached to the Reply as Exhibit 15 and Exhibit 16, respectively. As shown in the 4/3/2020 Redline, "trauma payments" are excluded from the definition of "Accounts Receivable" and instead are defined as "Other Receivables" which do not count toward the Final A/R Collected. However, government payments related to COVID-19 are included in the definition of "Accounts Receivable" and count toward the Final A/R Collected. This is the final version of sections 1.7(p) and 1.12(d)(i) that the parties ultimately agreed to and signed.

15. As set forth above and memorialized in the emails between the parties, this thoroughly negotiated definition of Accounts Receivable and the Accounts Receivable Reconciliation

7 SUPPLEMENTAL DECLARATION OF A. JOEL RICHLIN IN SUPPORT OF PRIME HEALTHCARE SERVICES, INC.'S REPLY TO POST-EFFECTIVE DATE DEBTORS AND LIQUIDATING TRUSTEE'S MEMORANDUM IN OPPOSITION TO PRIME HEALTHCARE SERVICES, INC.'S MOTION TO ENFORCE PROVISIONS OF THE ASSET PURCHASE

AGRÉEMENT PERTAINING TO ACCOUNTS RECEIVABLE ADJUSTMENT

27

28

25

26

7

8

9

10

11

12

13

14

15

16

#### Case 2:18-bk-20151-ER Doc 6669 Entered 09/28/21 18:34:40 Filed 09/28/21 Desc Main Document Page 44 of 51

procedures were critical in inducing Prime to complete the SFMC Sale. The parties had calls regarding the items contained in the definition of Accounts Receivable and expressly agreed to exclude trauma payments. These revisions were critical as each party worked to resolve and address various considerations and concerns, including: (a) the proper allocation of risk between the parties: (b) the uncertainty created by COVID-19 and the governmental response thereto; and (c) the need to induce Prime to close the SFMC Sale with a \$200 million purchase price after other parties declined to participate in the bidding process.

The exclusion of trauma payments from Accounts Receivable was a critical business 16. point that materially impacted Prime's willingness to complete a \$200 million purchase of a hospital in the midst of an unprecedented global pandemic that caused the indefinite pause of certain services offered by that hospital and the decline in its financial performance. The initial draft of the APA was drafted prior to the COVID-19 pandemic in the United States. As the gravity of the pandemic became obvious, the economic environment in which the parties initially drafted the APA was drastically different than the one in which the parties ultimately closed the transaction.

I declare under penalty of perjury under the laws of the United States of America that the foregoing is true and correct. Executed September 28, 2021, at Ontario, California.

ARATION OF A. JOEL RICHLIN IN SUPPORT INC.'S REPLY TO POST-EFFECTIVE DATE D

ORANDUM IN OPPOSITION

GRÉEMENT PERTAINING TO ACCOUNTS RECEIVABLE ADJUSTMEN

OTION TO ENFORCE

.. JOEL RICHLIN

JOEL RICHLIN IN SUPPORT OF PRIME

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

1

2

3

4

26

27

28

# **DECLARATION OF ANA GOFF**

I, Ana Goff, declare as follows:

1. I am over 18 years of age. I have personal knowledge of the facts set forth below and, if called to testify, would and could competently testify thereto.

5 2. I am the Director of Patient Financial Services for St. Francis Medical Center and have served in this role since September 15, 2020. Prior to joining St. Francis Medical Center, I 6 7 held several other hospital business office leadership positions at leading health systems for over 8 30 years. Specifically, I worked at Tenet Healthcare in various business office roles for 9 approximately 25 years. My final position with Tenet Healthcare was as Director of Patient 10 Financial Services for Chapman Medical Center in Orange County, California. In that role, I was responsible for leading the business office and was in charge of all billing and collection for the 11 12 hospital. After leaving Tenet Healthcare, I went to work for AHMC Healthcare as Director of 13 Patient Financial Services for Anaheim Regional Medical Center for approximately 7 years. In 14 that role, I was responsible for leading the business office and was in charge of all billing and 15 collection for the hospital. After leaving AHMC Healthcare, I went to work for Riverside 16 University Health System as Executive Director for Revenue Cycle for approximately 2 years. In 17 that role, I was responsible for leading the business office and was in charge of all billing and 18 collection for all of the health system's hospitals and clinics. I also had a leadership role in revenue 19 cycle processes for the health system. I have substantial experience collecting accounts receivable 20 and leading hospital business offices.

3. I submit this declaration (the "<u>Declaration</u>") in support of *Prime Healthcare Services, Inc.'s Reply in Support of Motion To Enforce Provisions Of The Asset Purchase Agreement Pertaining To Accounts Receivable Adjustment* filed contemporaneously herewith (the
"<u>Reply</u>"). Unless otherwise indicated, all capitalized but undefined terms herein shall have the
same meanings ascribed to them in the Reply.

1 DECLARATION OF ANA GOFF IN SUPPORT OF PRIME HEALTHCARE SERVICES, INC.'S REPLY TO POST-EFFECTIVE DATE DEBTORS AND LIQUIDATING TRUSTEE'S MEMORANDUM IN OPPOSITION TO PRIME HEALTHCARE SERVICES, INC.'S MOTION TO ENFORCE PROVISIONS OF THE ASSET PURCHASE AGREEMENT PERTAINING TO ACCOUNTS RECEIVABLE ADJUSTMENT

## Case 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Main Document Page 46 of 51

4. In my role as Director of Patient Financial Services for St. Francis Medical Center, 1 2 I am responsible for leading the business office at St. Francis Medical Center with the goal of 3 collecting on accounts receivable for the hospital. Since joining St. Francis Medical Center, I 4 worked on the transition of business office services from Sellers to Prime as part of the acquisition 5 of St. Francis Medical Center. I also helped lead Prime's collection efforts for the Accounts Receivable. My direct supervisor is Prime's Regional Vice President of Patient Financial Services, 6 7 Ken Wheeler, and I worked under his direction with regard to the transition of business office services and the collection of the Accounts Receivable. I have first-hand knowledgeable of 8 9 Prime's efforts to collect on the Accounts Receivable. Based on my experience in the industry, I am also knowledgeable about standard industry practices regarding the collection of accounts 10 11 receivable, including commonly used software programs and electronic medical record ("EMR") 12 systems. Based on my knowledge of Prime's efforts to collect the Accounts Receivable and prior 13 experience in the industry, I can state with confidence that since my first day at St. Francis Medical 14 Center and continuing to today, Prime has utilized commercially reasonable best efforts to collect 15 the Accounts Receivable and greater efforts than I have seen at other health systems. These efforts 16 are described in detail below and are commercially reasonable based on the number of full-time 17 employees and managers consistently dedicated to the collection of the Accounts Receivable, as 18 well as the additional support provided by Prime's clinical, coding, and case management teams.

19 5. I reviewed the Declaration of Peter Chadwick filed in support of Sellers' Objection 20 alleging that Prime did not fulfill its obligations under the APA with respect to collecting the 21 Accounts Receivable. In paragraph 17 of his declaration, Mr. Chadwick claims that I made specific statements about Prime's efforts to collect the Accounts Receivable. As an initial matter, 22 23 I do not know who Mr. Chadwick is and to the best of my knowledge have never met or spoken 24 with him. I did have regular communication with Regina Hernandez, Interim Vice President of 25 Revenue Cycle for VHS, from the time I started working at St. Francis Medical Center and continuing through the end of November 2020. I also interacted during this period with Sellers' 26

DECLARATION OF ANA GOFF IN SUPPORT OF PRIME HEALTHCARE SERVICES, INC.'S REPLY TO POST-EFFECTIVE DATE DEBTORS AND LIQUIDATING TRUSTEE'S MEMORANDUM IN OPPOSITION TO PRIME HEALTHCARE SERVICES, INC.'S MOTION TO ENFORCE PROVISIONS OF THE ASSET PURCHASE AGREEMENT PERTAINING TO ACCOUNTS RECEIVABLE ADJUSTMENT

27

#### Case 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Main Document Page 47 of 51

Senior Director of IT Application Services, Brenda Buchas, when I needed assistance with various systems or software platforms maintained by Sellers.

3 6. In paragraph 17 of his declaration, Mr. Chadwick states that "conversations with 4 Prime's head of collections affirmed the deliberate reduction in collection efforts." Specifically, 5 Mr. Chadwick claims that I "stated on November 17, 2020 that there were systems issues 6 preventing [my] team's ordinary-course follow up, without explaining what type of issues." This 7 is not accurate and mischaracterizes a communication I had with Ms. Hernandez on November 19, 8 2020, not Mr. Chadwick. On November 19, 2020, Ms. Hernandez reached out to me to ask why 9 cash postings were not appearing in MS4 for the last few days. I responded to explain that my 10 team was unable to update the cash postings in MS4 for a few days because of a printer connectivity issue. There was no direct electronic interface between the electronic remittance 11 12 advice payment records the hospital was receiving and Sellers' legacy EMR system, MS4. 13 Therefore, my team was required to print out all of these payment records and manually scan them 14 into MS4 to reflect the payments. The printers in the hospital business office were temporarily 15 offline for a few days and therefore cash postings were delayed in MS4 for a few days. However, 16 this delay in updating MS4 with cash postings in no way impacted actual collections, which were 17 timely pursued. Therefore, the alleged "systems issues" that I discussed with Ms. Hernandez on 18 November 19, 2020 did not prevent my team's ordinary-course follow up on collection of the 19 Accounts Receivable.

7. 20 Mr. Chadwick next states in paragraph 17 that I "indicated that collections people 21 had been pulled away from Verity A/R to focus on a Prime audit" and that I "stated that the impact 22 was effectively two collectors working five days a week." Once again, this appears to be a 23 mischaracterization of a communication I had with Ms. Hernandez on November 19, 2020, not 24 Mr. Chadwick. On November 19, 2020, I explained to Ms. Hernandez that 2 of the collectors on 25 site at the St. Francis Medical Center business office had been redirected for a limited period of 5 26 working days to assist with an audit. However, this temporary shifting of onsite resources did not

DECLARATION OF ANA GOFF IN SUPPORT OF PRIME HEALTHCARE SERVICES, NC.'S REPLY TO POST-EFFECTIVE DATE DEBTORS AND LIQUIDATING TRUSTEE'S INC.<sup>3</sup> MEMORANDUM IN OPPOSITION TO PRIME HEALTHCARE SERVICES, INC MOTION TO ENFORCE PROVISIONS OF THE ASSET PURCHASE AGREEMENT PERTAINING TO ACCOUNTS RECEIVABLE ADJUSTMENT

27

#### Case 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Main Document Page 48 of 51

impact the actual collections because Mr. Wheeler added additional resources from his corporate team located in Ontario, California to maintain staffing levels and prevent any delay in following up on collection of the Accounts Receivable.

4 8. Finally, Mr. Chadwick states in paragraph 17 that I "also stated that [I] had staffing 5 issues and had insufficient staff to assist [myself] with the legacy accounts." This assertion by Mr. 6 Chadwick is simply false. I never said this, or anything, to Mr. Chadwick, and I have no 7 recollection of making such a statement to Ms. Hernandez or anyone else. The only statement 8 remotely similar to that contained in Mr. Chadwick's declaration that I might have made is that I 9 may have talked about the challenges of managing a business office at the height of the COVID-10 19 pandemic in the fall of 2020. It is true that certain staff in the business office became ill or needed to quarantine due to potential exposure to COVID-19 at various times and this created 11 12 operational challenges with maintaining staffing levels. However, these challenges are no 13 different than I have routinely addressed related to managing teams of collectors for the past 30 14 years and I was grateful that when I needed additional staffing resources at St. Francis Medical 15 Center, I was able to get support from Mr. Wheeler's corporate collections team located in Ontario, California. 16

17 9. Overall, I was provided ample support in the form of additional staffing from the 18 Prime corporate team to assist with collection of the Accounts Receivable. This includes staff who 19 assisted with drafting clinical and coding appeals throughout the Reconciliation Period. Based on 20 30 years of experience in the industry, I believe the staffing devoted to collecting the Accounts 21 Receivable was more than sufficient and at least consistent with industry standards.

22 10. Accordingly, Mr. Chadwick's assertion in paragraph 17 that I made statements which "affirmed" Prime's "deliberate reduction in collection efforts" is categorically false. 23 24 Neither myself, nor anyone at Prime ever deliberately delayed or impaired in any way the 25 collection of the Accounts Receivable. To the contrary, the entire Prime team including myself, 26 the local onsite team at St. Francis Medical Center, and the corporate team in Ontario, California,

DECLARATION OF ANA GOFF IN SUPPORT OF PRIME HEALTHCARE SERVICES, IC.'S REPLY TO POST-EFFECTIVE DATE DEBTORS AND LIQUIDATING TRUSTEE'S INC.<sup>3</sup> MEMORANDUM IN OPPOSITION TO PRIME HEALTHCARE SERVICES, INC MOTION TO ENFORCE PROVISIONS OF THE ASSET PURCHASE AGREEMENT PERTAINING TO ACCOUNTS RECEIVABLE ADJUSTMENT

27

#### Case 2:18-bk-20151-ER Doc 6669 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Main Document Page 49 of 51

all worked tirelessly to collect as much as possible from the Accounts Receivable. We put in long hours including overtime, late nights, and weekends throughout the Reconciliation Period and continuing today to bring in as many dollars as possible for St. Francis Medical Center. The Prime team worked just as hard if not harder on the Accounts Receivable as they continue to work on current collections for the hospital.

I declare under penalty of perjury under the laws of the United States of America that the foregoing is true and correct. Executed September 27, 2021, at Lynwood, California.

ANA GO

DECLARATION OF ANA GOFF IN SUPPORT OF PRIME HEALTHCARE SERVICES, INC.'S REPLY TO POST-EFFECTIVE DATE DEBTORS AND LIQUIDATING TRUSTEE'S MEMORANDUM IN OPPOSITION TO PRIME HEALTHCARE SERVICES, INC.'S MOTION TO ENFORCE PROVISIONS OF THE ASSET PURCHASE AGREEMENT PERTAINING TO ACCOUNTS RECEIVABLE ADJUSTMENT

# **SUPPLEMENTAL DECLARATION OF STEVE ALEMAN**

I, Steve Aleman, declare as follows:

1. I am over 18 years of age. I have personal knowledge of the facts set forth below and, if called to testify, would and could competently testify thereto.

2. I am the Chief Financial Officer for Prime Healthcare Services, Inc. ("<u>Prime</u>"). I joined Prime in the Winter of 2019 and previously served as Vice President of Acquisitions. I have led all financial aspects of Prime's acquisition of St. Francis Medical Center since early 2020 and was deeply involved in the negotiation of the APA. I led the financial diligence process for the acquisition that occurred in 2020 and subsequently led all financial reporting and analysis for the acquisition.

3. I submit this declaration (the "<u>Declaration</u>") in support of *Prime Healthcare Services, Inc.'s Reply to Post-Effective Date Debtors and Liquidating Trustee's Memorandum in Opposition to Prime Healthcare Services, Inc.'s Motion to Enforce Provisions of the Asset Purchase Agreement Pertaining to Accounts Receivable Adjustment* filed contemporaneously herewith (the "<u>Reply</u>"). Unless otherwise indicated, all capitalized but undefined terms herein shall have the same meanings ascribed to them in the Reply.

4. I have reviewed Debtors' Objection as well as the attached exhibits and declaration of Peter Chadwick. Based on my review of these documents, and Exhibit F in particular, there is no explanation or supporting calculation for the \$41,915,599 figure described merely as "Net A/R at 8/13, excluding Trauma." I have spent several hours trying to back into this figure, but have not found any calculation or explanation contained in Debtors' Objection which would explain the derivation of this amount.

5. Regardless, Debtors' calculation of Sellers' historical collection rate as 90.3% is based on a 39-month period ranging from July 2016 to September 2019, which is not directly applicable to the collectability of the Accounts Receivable during the Reconciliation Period in the fall of 2020. In fact, the collection rates during the 39-month period relied upon by Debtors ranged from 70% to

SUPPLEMENTAL DECLARATION OF STEVE ALEMAN IN SUPPORT OF PRIME HEALTHCARE SERVICES, INC.'S REPLY TO POST-EFFECTIVE DATE DEBTORS AND LIQUIDATING TRUSTEE'S MEMORANDUM IN OPPOSITION TO PRIME HEALTHCARE SERVICES, INC.'S MOTION TO ENFORCE PROVISIONS OF THE ASSET PURCHASE AGREEMENT PERTAINING TO ACCOUNTS RECEIVABLE ADJUSTMENT

27

1 98%, demonstrating that a broad range of collection rates existed and are commercially reasonable 2 based on historical experience.

6. Prime is not withholding any disproportionate share payments or QAF V payments to which Sellers are entitled. In fact, on September 21, 2021, Prime wired to Sellers \$6,987,948,56 on account of received QAF V Seller Net Payments and \$1,277,821.50 on account of received disproportionate share payments relating to dates of service prior to the Effective Time. Based on my review of Prime's financial records, I have determined that these are the complete amounts currently owed to Sellers.

7. Prime is currently holding \$24,100,380 in QAF VI Seller Net Payments as an offset because the Final A/R Collect is less than the A/R Target Amount by \$28,263,312.

I declare under penalty of perjury under the laws of the United States of America that the foregoing is true and correct. Executed September 28, 2021, at Ontario, California.

STEVE ALEMAN

11 12 13 14 15 16 17 18 19 20 21 22 23 24 25 26 27 SUPPLEMENTAL DECLARATION OF STEVE ALEMAN IN SUPPORT PRIME HEALTHCARE SERVICES, INC.'S REPLY TO POST-EFFECTIVE DATE DEBTORS AND LIQUIDATING TRUSTEE'S MEMORANDUM IN OPPOSITION TO PRIME HEALTHCARE SERVICES, INC.'S MOTION TO ENFORCE PROVISIONS OF THE ASSET PURCHASE AGREEMENT PERTAINING TO ACCOUNTS RECEIVABLE ADJUSTMENT 28

3

4

5

6

7

8

9

# ASSET PURCHASE AGREEMENT

# By and Among

# ST. FRANCIS MEDICAL CENTER, VERITY HOLDINGS, LLC, and VERITY HEALTH SYSTEM OF CALIFORNIA, INC.

(as "Sellers")

and

# PRIME HEALTHCARE SERVICES, INC.

(as "Purchaser")

Dated \_\_\_\_\_, 2020

The Sellers will consider all proposals, bids and submissions. This form of<br/>Asset Purchase Agreement, including all exhibits, is merely provided as an<br/>example of the structure of a transaction as an aid to potential bidders.Bidders should consider their own structures, needs and best offers as part of<br/>their proposals. This Asset Purchase Agreement is not an offer capable of<br/>acceptance.

# Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 2 of 111

# **INDEX OF SCHEDULES & EXHIBITS**

## Schedules

Schedule <u>1.4.3</u> <u>1.4(c)</u>	Owned Real Property
Schedule 1.7(b)	Licenses
Schedule 1.7(c)(ii)	Leased Real Property
Schedule 1.7(c)(iii)	Tenant Leases
Schedule 1.7(d)	Assigned Contracts and Assigned Leases
Schedule 1.8(cc)	Certain Other Excluded Assets
Schedule 1.9(i)	Other Obligations
Schedule 2.4	Seller's Knowledge of Breach or Default
Schedule 2.5	Brokers and Finders (Sellers)
Schedule 2.6	Compliance (Sellers)
Schedule 2.7	Required Consents (Sellers)
Schedule 2.8	Legal Proceedings (Sellers)
Schedule 2.9(b)	Environmental Matters (Sellers)
Schedule 2.11	Missing Material Licenses (Sellers)
Schedule 2.13	Non-Bankruptcy Court Legal Proceedings (Sellers)
Schedule 2.19	List of Excluded Individuals (Seller)
Schedule 3.4	No Violation (Purchaser)
Schedule 3.7	Legal Proceedings (Purchaser)
Schedule 5.3	"Hospital Employees"
Schedule 11.3(b)	Tax Allocation Schedule

# Exhibits

No.	Description	Status
Exhibit 1.2	Escrow Agreement	to come
Exhibit 1.3(a)	Leaseback Agreement	Attached
Exhibit 1.3(b)	Interim Management Agreement	Attached
Exhibit	Bill of Sale	to come
<u>1.4.1<u>1.4(a)</u></u>		
Exhibit	Real Estate Assignment Agreements	to come
<u>1.4.21.4(b)</u>		
Exhibit	Quitclaim Deeds	to come
<u>1.4.31.4(c)</u>		
Exhibit	Transfer Agreement	to come
<u>1.4.41.4(d)</u>		
Exhibit	Transition Services Agreement	Attached
<u>1.4.61.4(f)</u>		
Exhibit	Accepted Conditions (Attorney General Conditions)	to come
<u>5.9.35.9(c)</u>		

Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 3 of 111

# TABLE OF CONTENTS

(continued)

Page

L	1

::	

US\_Active\114188900\V-6

US\_Active\114483166\V-14

#### ASSET PURCHASE AGREEMENT

This Asset Purchase Agreement (the "Agreement") is made and entered into as of \_\_\_\_\_\_, 2020 (the "Signing Date") by and among St. Francis Medical Center, a California nonprofit public benefit corporation ("SFMC"), Verity Holdings, LLC ("Verity Holdings"), a California limited liability company and Verity Health System of California, Inc., a California nonprofit public benefit corporation ("VHS" and, together with SFMC and Verity Holdings, the "Sellers" and each individually a "SellersSeller") and Prime Healthcare Services, Inc., a Delaware corporation ("Purchaser" and, together with the Sellers, the "Parties" and each individually a "Party").

#### RECITALS

A. SFMC engages in the business of operating a hospital known as St. Francis Medical Center ("St. Francis") in the City of Lynwood, California, including the hospital pharmacy, laboratory and emergency department as well as through the medical office buildings and clinics owned or operated by SFMC (with St. Francis, collectively referred to herein, as the "Hospital").

B. VHS is the sole member of SFMC and Verity Holdings. Verity Holdings owns and rents certain properties on or near the Hospital campus.

C. Purchaser desires to purchase from Sellers, and Sellers desire to sell to Purchaser, the assets described in <u>Section 1.7</u> below for the consideration and upon the terms and conditions contained in this Agreement.

D. On August 28, 2018, Sellers filed voluntary petitions for relief under Chapter 11 of Title 11 of the United States Code (the "**Bankruptcy Code**") within the United States Bankruptcy Court for the Central District of California, Los Angeles Division, before the Honorable Ernest M. Robles (the "**Bankruptcy Court**"), which isare jointly administered with itstheir affiliates under Case No. Case No. 18-20151 (the "**Bankruptcy Cases**").

E. On February 10, 2020, Sellers filed the Debtors' Notice of Motion and Motion for the Entry of (I) an Order (1) Approving Form of Asset Purchase Agreement; (2) Approving Auction Sale Format, Bidding Procedures and Stalking Horse Bid Protections; (3) Approving Form of Notice to be Provided to Interested Parties; (4) Scheduling a Court Hearing to Consider Approval of the Sale to the Highest Bidder; and (5) Approving Procedures Related to the Assumption of Certain Executory Contracts and Unexpired Leases; and (II) an Order (A) Authorizing the Sale of Property Free and Clear of All Claims, Liens and Encumbrances; Memorandum Of Points And Authorities In Support Thereof [Docket No. \_\_\_] (the "Bidding Procedures") Motion") which was approved by an order of the Bankruptcy Court (the "Bidding Procedures Order") [Docket No. 4165] and pursuant thereto the parties Parties have entered into this Agreement.

F. The <u>partiesParties</u> intend to effectuate the transactions contemplated by this Agreement through a sale of the Assets pursuant to the Bidding Procedures and as approved by the Bankruptcy Court pursuant to <u>SectionSections</u> 363 and 365 of the Bankruptcy Code.

#### AGREEMENT

NOW, THEREFORE, in consideration of the foregoing premises and the mutual promises and covenants contained in this Agreement, and for their mutual reliance and incorporating into this Agreement the above recitals, the <u>parties</u> hereto agree as follows:

## ARTICLE 1 SALE AND TRANSFER OF ASSETS; CONSIDERATION; CLOSING

#### 1.1 <u>Purchase Price</u>.

(a) Subject to the terms and conditions of this Agreement, the aggregate consideration shall be as follows:

(i) Cash payment to Sellers of Two Hundred Million Dollars (\$200,000,000) ("**Base Price**");

(ii) Cash payment of <u>One Hundred Fourteen Million Dollars</u> (\$114,000,000) (the "the Closing QAF Payment Amount" (defined below in Section 1.1(c));

(iii) Cash payment of Fifty-Two Million Dollars  $($52,000,000)^{4}$  (the "A/R <u>Target</u> Amount") as consideration for the Accounts Receivable transferred at Closing (subject to adjustment in Section 1.12);

(iv) Cash payment for Sellers' payroll liabilities at Closing (the "**Payroll Amount**") (which as of October 31, 2019, had an aggregate value of approximately Seven Million Dollars (\$7,000,000));

(v) Cash payment for accrued vacation and other paid time-off of Sellers' employees at Closing (the "**PTO Amount**") (which as of October 31, 2019, had an aggregate value of approximately Eleven Million Dollars (\$11,000,000<del>));</del>

(vi) Purchaser's commitment to invest Thirty-Five Million Dollars (\$35,000,000) in estimated deferred capital expenditures for St. Francisover the five (5) year period immediately after the Closing Date;(vii) Purchaser's commitment to remediate Sellers' estimated NPC-3 seismicliability (which is estimated to cost approximately Twelve Million Dollars-(\$12,000,000)); and

(vi) (viii) An amount equal to the Cure Costs (defined below) associated with outstanding liabilities of Sellers under any Assigned Leases and/or Assigned Contracts; and

<sup>&</sup>lt;sup>1</sup> Note to Sellers: The fixed A/R Amount is subject to mutual agreement between Sellers and Purchaser. Purchaser is also willing to consider alternative A/R valuation approaches, including a post Closing reconciliation mechanism.

(vii) (ix) An amount determined in accordance with <u>Section 1.6</u>.

(b) At the Closing, Purchaser shall pay to Sellers, by wire transfer of immediately available funds to the accounts specified by Sellers to Purchaser in writing, an aggregate amount equal to the Base Price, *plus* the <u>Closing</u> QAF Payment Amount, *plus* the A/R<sub>\_</sub> <u>Target</u> Amount, *plus* the Cure Cost Amount, *plus* the PTO Amount, *plus* the Payroll Amount (collectively, the "**Purchase Price**"), *minus* the Deposit (defined below).

For purposes of this Agreement, the "QAF Program" means the (c) California Department of Health Care Services ("DHCS") Hospital Quality Assurance Fee Programs V ("QAF V") and VI ("QAF VI"). During the period prior to the Closing, Sellers shall pay or permit DHCS to offset from amounts owed to St. Francis any fees due and owing prior to the Closing under QAF V and QAF VI (such payments or offsets shall be included in the formula described in Section 1.8(b)), and Sellers shall be entitled to retain all payments received under QAF V and QAF VI. At Closing, to the extent not yet received by Sellers, Purchaser shall pay Sellers an aggregate amount in respect of OAF V and OAF VI (the "Closing OAF Payment Amount-equal to One Hundred Fourteen Million Dollars (\$114,000,000). The OAF Payment-Amount is comprised of (i) Forty-Six Million Dollars (\$46,000,000) as consideration for the OAF V payments due to Sellers on or around March 2020, (ii) Twenty-Seven Million Dollars-(\$27,000,000'') equal to: (i) (A) Twenty-Eight Million Dollars (\$28,000,000) as consideration for the QAF V payments due to Sellers on or around November 2020, and (iii) Forty-One Million Dollars (\$41,000,000) (scheduled by DHCS to be paid to Sellers in November 2020 and May 2021, plus (B) any fees paid by Sellers or offset by DHCS in respect of OAF V relating to St. Francis between the Signing Date and the Closing; plus (ii) an amount equal to (A) the QAF VI Seller Net Payments (defined below), multiplied by (B) fifty percent (50%), minus (C) the dollar value of all payments actually received by Sellers in respect of QAF VI relating to St. Francis prior to the Closing, plus (D) the dollar value of all fees paid by Sellers or offset by DHCS in respect of QAF VI relating to St. Francis prior to the Closing (the amount determined pursuant to this clause (ii), the "QAF VI Partial Payment") as a partial up-front payment for the QAF VI payments that otherwise will be payable to Sellers as described in Section 1.8(b) hereofSeller Net Payments. After the Closing, Purchaser shall pay to Sellers, promptly following receipt thereof (but in any event within ten (10) business days of such receipt), all QAF VI payments received relating to St. Francis until Purchaser has paid to Sellers an amount, when added to the OAF VI Partial Payment, equal to the QAF VI Seller Net Payments.

- **.**
- (d) Separate from, and in addition to, the Purchase Price, Purchaser commits

<u>to:</u>

(i) invest Thirty-Five Million Dollars (\$35,000,000) in estimated deferred capital expenditures for St. Francis over the five (5) year period immediately after the Closing Date; and

(ii) remediate Sellers' estimated NPC-3 seismic compliance responsibilities (which is estimated to cost approximately Twelve Million Dollars (\$12,000,000)).

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 7 of 111

1.2 Deposit. Purchaser has deposited an amount equal to ten percent (10%) of the Purchase Price (the "Initial Deposit") by wire transfer to Chicago Title Insurance Company ("Escrow Agent") pursuant to that certain Escrow Agreement attached hereto as Exhibit 1.2. Within two (2) business day after the date of the conclusion of the auction for the Assets, if any, if Purchaser is the Prevailing Bidder (as defined below) and as a result of such auction has opted to increase the Purchase Price (the "Auction Purchase Price"), Purchaser shall deposit an additional amount with the Escrow Agent such that the total amount deposited with the Escrow Agent is equal to ten percent (10%) of the Auction Purchase Price (the Initial Deposit and any additional amount deposited with the Escrow Agent. All fees of the Escrow Agent shall be paid by Purchaser. The Deposit shall be non-refundable in all events, except in the event the Closing does not occur due to Purchaser's termination of the Agreement pursuant to Sections 9.1 (a), (c), (d), (f), (g), (h) or (i) hereof. Upon Closing, the Deposit will be credited against the Purchase Price.

1.3 <u>Closing Date</u>. The consummation of the transactions contemplated by this Agreement (the "Closing") shall take place at the offices of Dentons US LLP, 601 South Figueroa St., Suite 2500, Los Angeles, CA 90017-5704 (the day on which Closing actually occurs, the "Closing Date") within three (3) business days following the satisfaction or waiver of the conditions set forth in ARTICLE 7 and ARTICLE 8, and other than those conditions that by their nature are to be satisfied at Closing but subject to fulfillment or waiver of those conditions. The Closing shall be deemed to occur and to be effective as of 12:00 a.m. pacific time on the day immediately after the Closing Date (the "Effective Time"). Purchaser and Sellers agree that because the change of ownership and regulatory approval process may take an extended period of time, at the Effective Time, the Assets (less any Assets constituting drugs or pharmacy assets, the Medicare/Medi-Cal Agreements, the Lockboxes, the Transferred Managed Care Agreements and the Transferred Private Payor Agreements (as such terms are defined below)) will be sold to Purchaser and immediately leased back to Sellers (substantially in the form of the Leaseback Agreement attached hereto as Exhibit 1.3(a), the "Leaseback Agreement"), with a concurrent management arrangement (substantially in the form of the Interim Management Agreement attached hereto as Exhibit 1.3(b), the "IMA"). On the effective date that Purchaser obtains a general acute care hospital license from the California Department of Public Health ("CDPH") and a hospital pharmacy permit from the California State Board of Pharmacy ("BOP") (i) the Leaseback Agreement and IMA will terminate and (ii) the drugs and pharmacy assets, the Medicare/Medi-Cal Agreements, the Lockboxes, the Transferred Managed Care Agreements and the Transferred Private Payor Agreements will be transferred to Purchaser (without payment of any additional Purchase Price) (the "Licensure Date"). For the avoidance of any doubt, the Licensure Date shall be the date the Purchaser's hospital license and pharmacy permit are effective, even if they are not actually issued until some timea later date. The Licensure Date may be determined based on oral assurances from CDPH and the BOP.

1.4 <u>Items to be Delivered by Sellers at Closing</u>. At or before the Closing, Sellers shall deliver, or cause to be delivered, to Purchaser the following:

(a) 1.4.1 a Bill of Sale substantially in the form of Exhibit 1.4.1 (a) attached hereto (the "**Bill of Sale**"), duly executed by each Seller;

(b) 1.4.2—Real Estate Assignment Agreements (the "**Real Estate** Assignments") in the form of Exhibit 1.4.21.4(b) attached hereto with respect to (i) the Leased Real Property, and (ii) the Tenant Leases, each duly executed by the applicable Sellers;

(c) 1.4.3 Quitclaim Deeds in the form of Exhibit 1.4.31.4(c) attached hereto with respect to the real property listed in Schedule 1.4.3, 1.4(c), together with all plant, buildings, structures, installments, improvements, fixtures, betterments, additions and constructions in progress situated thereon (the "Owned Real Property") duly executed by SFMC and Verity Holdings as applicable;

(d) 1.4.4 an Assigned Contract Transfer Agreement (the "**Transfer Agreement**") in the form of Exhibit 1.4.41.4(d) attached hereto, duly executed by the applicable Sellers;

(e) <u>1.4.5</u>-evidence of payment of all Cure Costs;

(f) 1.4.6 the Transition Services Agreement (the "**Transition Services Agreement**") in form attached hereto as <u>Exhibit 1.4.6, 1.4 (f)</u>, duly executed by the applicable Sellers;

(g) 1.4.7-the Leaseback Agreement and IMA, duly executed by SFMC;

(h) 1.4.8-favorable certificates of good standing, of each Seller, issued by the State of California, dated no earlier than a date which is fifteen (15) calendar days prior to the Closing Date;

(i) 1.4.9 a duly executed certificate of an officer of each Seller certifying to Purchaser (i) the incumbency of the officers of such Seller on the Signing Date and on the Closing Date and bearing the authentic signatures of all such officers who shall execute this Agreement and any additional documents contemplated by this Agreement and (ii) the due adoption and text of the resolutions or consents of the Board of Directors of such Seller authorizing (I) the transfer of the Assets and transfer of the Transferred Obligations by such Seller to Purchaser and (II) the due execution, delivery and performance of this Agreement and all additional documents contemplated by this Agreement, and that such resolutions have not been amended or rescinded and remain in full force and effect on the Closing Date;

(j) 1.4.10 a certified copy of the Sale Order (as defined below);

(k) 1.4.11-non-foreign affidavits of SFMC and Verity Holdings, as applicable, dated as of the Closing Date, in the form of Exhibit 1.4.111.4(k); and

(1) 1.4.12 Any such other instruments, certificates, consents or other documents which Purchaser and Sellers mutually deem reasonably necessary to carry out the transactions contemplated by this Agreement and to comply with the terms hereof.

1.5 <u>Items to be Delivered by Purchaser at Closing</u>. At or before the Closing, Purchaser shall deliver or cause to be delivered to Sellers the following: (a) <u>1.5.1</u>-payment of the Purchase Price, minus the Deposit;

(b) 1.5.2-a duly executed certificate of the Secretary of Purchaser certifying to Sellers (a) the incumbency of the officers of Purchaser on the Signing Date and on the Closing Date and bearing the authentic signatures of all such officers who shall execute this Agreement and any additional documents contemplated by this Agreement and (b) the due adoption and text of the resolutions of the Board of Directors of Purchaser authorizing the execution, delivery and performance of this Agreement and all additional documents contemplated by this Agreement, and that such resolutions have not been amended or rescinded and remain in full force and effect on the Closing Date;

(c) 1.5.3-favorable certificate of good standing, of Purchaser, issued by the California Secretary of State dated no earlier than a date which is fifteen (15) calendar days prior to the Closing Date;

- (d) <u>1.5.4</u> the Transition Services Agreement, duly executed by Purchaser;
- (e) <u>1.5.5</u>-the Leaseback Agreement and IMA, duly executed by Purchaser;
- (f)  $\frac{1.5.6}{1.5.6}$  the Bill of Sale, duly executed by Purchaser;
- (g) <u>1.5.7</u> the Real Estate Assignment(s), duly executed by Purchaser;
- (h) 1.5.8 the Transfer Agreement, <u>duly executed by Purchaser</u>;

(i) Preliminary Change of Ownership Report(s) (BOE-502-A) with respect to the Owned Real Property, duly executed by Purchaser; and

(j) 1.5.9 any such other instruments, certificates, consents or other documents which Purchaser and Sellers mutually deem reasonably necessary to carry out the transactions contemplated by this Agreement and to comply with the terms hereof.

1.6 <u>Prorations and Utilities</u>. All items of income and expense listed below with respect to the Assets shall be prorated in accordance with the principles and the rules for the specific items set forth hereafter:

(a)  $\frac{1.6.1}{1.6.1}$  All transfer, conveyance, sales, use, stamp, similar state and local taxes arising from the sale of the Assets hereunder shall be the responsibility of, and allocated to, Purchaser.

(b) 1.6.2 Other than the Utility Deposits (defined below), which are governed by Section 1.8(n), and other than with respect to Cure Costs, the following costs and expenses shall be prorated based upon the payment period (*i.e.*, calendar or other tax fiscal year) to which the same are attributable: (i) all real estate and personal property lease payments, real estate and personal property taxes, real estate assessments and other similar charges against real estate, (ii) power and utility charges, (iii) payments made by Sellers for information technology software or services for periods after the Effective Time, (iv) payments made by Sellers in association with Licenses or dues paid to government and non-governmental agencies for calendar year 2020, and

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 10 of 111

(v) other similar costs for items or services which continue past the Effective Time (collectively, the "**Prorated Charges**"). Sellers shall pay at or prior to the Closing (or Purchaser shall receive credit for) any unpaid Prorated Charges attributable to periods or portions thereof occurring prior to the Effective Time, and Purchaser shall be responsible for or, to the extent previously paid by Sellers, pay to Sellers at the Closing all Prorated Charges attributable to periods or portions thereof occurring from and after the Effective Time. In the event that as of the Closing Date the actual tax bills for the tax year or years in question are not available and the amount of taxes to be prorated as aforesaid cannot be ascertained, then rates, millages and assessed valuation of the previous year, with known changes, shall be used. The <u>partiesParties</u> agree that if the real estate and personal property tax prorations are made based upon the taxes for the preceding tax period, the prorations shall be re-prorated after the Closing. As to power and utility charges, "final readings," if any, shall be paid by Purchaser.

(c) 1.6.3-Sellers shall be entitled to all rents and other payments under Tenant Leases accruing for the period prior to the Effective Time ("Pre Effective Time Lease Amounts"), and Purchaser shall be entitled to all rents and other payments under tenant leases accruing for the period after the Effective Time ("Post Effective Time Lease Amounts" and together with the Pre Effective Time Lease Amounts, the "Lease Amounts"). All Lease Amounts that are collected prior to the Closing shall be prorated as of the Closing in accordance with the immediately preceding sentence. All Lease Amounts that are accrued but unpaid as of the Closing (including, without limitation, rents and other payments accrued prior to the Closing but payable in arrears after the Closing) (collectively, the "Unpaid Amounts") shall belong to Sellers, and Purchaser shall, upon receipt of said rents and other payments, receive the same in trust for Sellers and shall promptly remit any of such amounts to Sellers within ten (10) days after Purchaser's receipt of same.

(d) 1.6.4 If Purchaser requests that Sellers transfer electronic medical records in a specific electronic format due to the Sellers' electronic medical record ("EMR") system and Purchaser's EMR system not being interoperable, then Purchaser shall reimburse, on the Closing Date, all amounts paid or to be paid by Sellers to transfer electronic medical records to Purchaser in such different electronic format.<sup>2</sup>

(e) 1.6.5—All prorations and payments to be made under the foregoing provisions shall be agreed upon by Purchaser and Sellers prior to the Closing and shall be binding upon the parties Parties; provided, however, with respect to the Unpaid Amounts, in the event any proration, apportionment or computation shall prove to be incorrect for any reason, then either Sellers or Purchaser shall be entitled to an adjustment to correct the same, provided that said partyParty makes written demand on the partyParty from whom it is entitled to such adjustment within fifteen (15) calendar days after the erroneous payment or computation was made, or such later time as may be required, in the exercise of due diligence, to obtain the necessary information for proration.

(f)  $\frac{1.6.6}{1.6.6}$  This <u>Section 1.6</u> shall survive Closing.

<sup>&</sup>lt;sup>2</sup> <u>NTD</u>: While we have left this provision in the document, we believe it will be inapplicable to this transaction. Purchaser proposes to assume the EMR at Closing.

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 11 of 111

1.7 <u>Transfer of Sellers Assets</u>. On the Closing Date, subject to the terms and conditions of this Agreement and except as otherwise contemplated in <u>Section 1.3</u>, Sellers shall sell, assign, transfer, convey and deliver to Purchaser, free and clear of all interests, including but not limited to all liens, privileges, pledges, security interests, rights of first refusal, options, defects in title and encumbrances other than the Permitted Exceptions (defined below), and Purchaser shall acquire, all of Sellers' right, title and interest in and to only the following assets and properties, as such assets shall exist on the Closing Date, to the extent not included among the Excluded Assets, such transfer being deemed to be effective at the Effective Time (the "Assets"):

(a) all of the tangible personal property owned by Sellers, or to the extent assignable toor transferable by Sellers, and used by Sellers in the operation of the Hospital, including equipment, furniture, fixtures, machinery, vehicles, office furnishings and leasehold improvements (the "**Personal Property**");

(b) subject to <u>Section 1.7(e)</u>, all of Sellers' rights, to the extent assignable or transferable, to all licenses, permits, approvals, certificates of exemption, franchises, accreditations and registrations and other governmental licenses, permits or approvals issued to Sellers for use in the operation of the Hospital (the "**Licenses**"), including, without limitation, the Licenses set forth on <u>Schedule 1.7(b)</u>, except to the extent Purchaser elects, in its discretion, not to take assignment of any such Licenses;

(c) all of Sellers' interest in and to the Owned Real Property and all of Sellers' interest, to the extent assignable or transferable and that have been designated by Purchaser, in and to all of the following (the "Assigned Leases"): (i) the real property leases for all real property leased by Sellers and set forth on <u>Schedule 1.7(c)(i)</u> (the "Leased Real Property"), and (ii) the real property leased or subleased by Sellers to a third party and set forth on <u>Schedule 1.7(c)(ii)</u> (the "Tenant Leases");

(d) all of Sellers' interest in, and all of Sellers' obligations due under from and after the Effective Time, to the extent assignable or transferable, all contracts and agreements (including, but not limited to, purchase orders) that have been designated by Purchaser as an Assigned Contract, pursuant to Section 1.11 and appearing on Schedule 1.7(d);

(e) so long as the conditions of <u>Section 8.8</u> are satisfied prior to the Closing-Date, all of Sellers' interest in, from and after the Licensure Date, to the extent assignable or transferable, the Hospital's Medicare Provider Agreement (and provider number) and the Hospital's Medi-Cal Provider Agreement (and provider number) (collectively, the "Medicare/Medi-Cal Agreements");

(f) all of Sellers' interest in, and all of Sellers' obligations due under, from and after the Licensure Date, to the extent assignable or transferable, in and to any of the Hospital's managed care, pre-paid, capitated or other full-risk health plan agreements (collectively, the "**Managed Care Agreements**") that have been designated by Purchaser as an Assigned Contract pursuant to <u>Section 1.11</u> (to the extent so designated, the "**Transferred Managed Care Agreements**");

I

### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 12 of 111

(g) subject to <u>Section 4.6</u>, all of Sellers' interest in, and all of Sellers' obligations due under, from and after the Licensure Date, to the extent assignable or transferable, in and to any of the Hospital's services, participation or provider agreements with private health plans, insurers or other third party payors (collectively, the "**Private Payor Agreements**") that have been designated by Purchaser as an Assigned Contract pursuant to <u>Section 1.11</u> (to the extent so designated, the "**Transferred Private Payor Agreements**"), *provided that*, Private Payor Agreements shall not include any "risk-sharing" agreements with independent physician associations;

(h) to the extent assignable or transferable, all inventories of supplies, drugs, food, janitorial and office supplies and other disposables and consumables (i) located at the Hospital or (ii) used in the operation of the Hospital (the "**Inventory**") except as set forth in <u>Sections 1.3</u> and <u>1.8(i)</u>;

(i) other than Utility Deposits, all prepaid rentals, deposits, prepayments and similar amounts relating to the Assigned Contracts and/or the Assigned Leases, which were made with respect to the operation of the Hospital (the "**Prepaids**");

(j) to the extent assignable or transferable, all of the following that are not proprietary to Sellers and/or owned by or proprietary to Sellers' affiliates: operating manuals, files and computer software with respect to the operation of the Hospital, including, without limitation, all patient records, medical records, employee records, billing records, financial records, equipment records, construction plans and specifications, and medical and administrative libraries; *provided, however*, that any electronic medical records may be transferred in paper or "pdf" if Sellers' EMR system and Purchaser's EMR system are not interoperable;

(k) to the extent assignable or transferable, all rights in all warranties of any manufacturer or vendor in connection with the Personal Property;

(1) all right, title and interest in and to the name "St. Francis Medical Center," including any associated Hospital trademarks, service marks, trade names, logos and domain names but excluding the domain <u>https://stfrancis.verity.org</u> and content therein;

(m) all goodwill of the Hospital evidenced by the Assets;

(n) to the extent transferable or assignable, Sellers' right or interest in the telephone and facsimile numbers used with respect to the operation of the Hospital;

(o) to the extent assignable or transferable, Sellers' lock box account(s) associated with Medicare or Medi-Cal fee-for-service receivables (the "Lockboxes") on or after the Licensure Date;

(p) all accounts and interest thereupon, notes and interest thereupon and other receivables of Sellers, including, without limitation, accounts, notes or other amounts receivable, and all claims, rights, interests and proceeds related thereto, including all accounts and other receivables, quality assurance fee payments, trauma payments, disproportionate share payments,

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 13 of 111

cost report, claim, EHR or other similar appeals and Seller Cost Report settlements, in each case arising from the rendering of services or provision of goods, products or supplies to inpatients and outpatients at the Hospital, billed and unbilled, recorded and unrecorded (including any such amounts that were written-off by Sellers for any reason), for services, goods, products and supplies provided by Sellers prior to the Licensure Date whether payable by Medicare, Medi-Cal, Medicaid, or any other payor (including an insurance company), or any health care provider or network (such as a health maintenance organization, preferred provider organization or any other managed care program) or any fiscal intermediary of the foregoing, private pay patients, private insurance or by any other source (collectively, "Accounts Receivable");

(q) all documents, records, correspondence, work papers and other documents, other than patient records, primarily relating to the Accounts Receivable;

(r) all QAF payments due to the Hospital from the State of California or any of its administrative entities or other entitles, including without limitations, Medi-Cal managed care plans, received on and after the Licensure Date and attributable to QAF V, QAF VI and any subsequent QAF Program, regardless of the state fiscal year for which the payments are made in reference to and regardless of the state fiscal year for which the data was derived to calculate eligibility for such payments, other than the QAF payments received prior to Closing or specifically excluded pursuant to Section 1.8(b) as QAF VI Seller Net Payments;

(s) <u>except as set forth in Section 1.8(1) (*i.e.*, certain causes of action), all claims, causes of action, choses in action, rights of recovery, rights of set off and rights of recoupment of Sellers against third parties related to or associated with the physical condition of any of the Assets; and</u>

(t) to the extent assignable or transferable, any other assets owned by Sellers (which are not otherwise specifically described above in this <u>Section 1.7</u>) that are used primarily in the operation of the Hospital.

As used herein, the term "**Permitted Exceptions**" means (i) the Transferred Obligations; (ii) liens for taxes not yet due and payable: (iii) easements, rights of way, zoning ordinances and other similar encumbrances affecting real property; (iv) imperfections of title or encumbrances identified in the Title Commitments (defined below) other than those specifically identified in <u>Schedule 1.7(I)</u>; and (iv) other imperfections of title or encumbrances, if any, which are not monetary in nature and that are not, individually or in the aggregate, material to the business of the Hospital.

1.8 <u>Excluded Assets</u>. Notwithstanding anything to the contrary in <u>Section 1.7</u>, Sellers shall retain all interests, rights and other assets owned directly or indirectly by it (or any of Sellers' affiliates) which are not among the Assets, including, without limitation, the following interests, rights and other assets of Sellers (collectively, the "**Excluded Assets**"):

(a) cash, cash equivalents and short-term investments;

(b) <u>all QAF payments received prior to Closing under any QAF program and,</u> <u>in addition, with respect to the QAF VI Program, the Excluded Assets shall include the "QAF VI</u>

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 14 of 111

Seller Net Payments" which shall be, with respect to the Hospital, an amount equal to the product of: (A) all payments received or to be received by the Hospital under the QAF VI Program minus all payments already made, required to be made in the future or to be offset by the government with respect to the QAF VI Program, including payments to the California Health Foundation & Trust, multiplied by (B) the Closing Date Percentage. The "Closing Date Percentage" shall mean the percentage derived by dividing the total number of days between July 1, 2019 and the Effective Time, by 915. For avoidance of doubt, the QAF VI Seller Net Payments to be remitted to Sellers once received will be reduced by the QAF VI Partial Payment;915:

(c) all Disproportionate Share Hospital Payments ("**DSH**") received on or after the Effective Time but calculated based on data from periods prior to the Effective Time (whether received before or after the Effective Time and whether paid to Sellers or Purchaser);

(d) all Sellers Plans (defined below) and the assets of all Sellers Plans and any asset that would revert to the employer upon the termination of any Sellers Plan, including, without limitation, any assets representing a surplus or overfunding of any Sellers Plan;

(e) all contracts that are not Assigned Contracts and all risk sharing agreements with independent physician associations ("IPAs");

(f) any Private Payor Agreement that is not a Transferred Private Payor Agreement and any Managed Care Agreement that is not a Transferred Managed Care Agreement;

(g) all collective bargaining agreements or other arrangements with unions representing Sellers' employees;

(h) all leases that have not been designated as Assigned Leases, and all Contracts that have not been designated as Assigned Contracts and all rents, deposits, prepayments, and similar amounts relating thereto;

(i) the portions of Inventory, Prepaids, and other assets disposed of, expended or canceled, as the case may be, by Sellers after the Signing Date and prior to the Effective Time in the ordinary course of business;

(i) assets owned by vendors of services or goods to the Hospital;

(k) all of Sellers' organizational or corporate record books, minute books and tax records;

(1) Except as set forth in Section 1.7(ik) or 1.7(s) hereof (*i.e.*, rights under warranties and physical condition claims), all claims, counterclaims and causes of action of Sellers or Sellers' bankruptcy estate (including parties acting for or on behalf of Sellers' bankruptcy estate, including, but not limited to, the official committee of unsecured creditors appointed in the Bankruptcy Case), including, without limitation, (A) causes of action arising out of any claims and causes of action under chapter 5 of the Bankruptcy Code ("Avoidance").

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 15 of 111

<u>Claims</u>"), (B) any claims, counterclaims and causes of action under applicable non-bankruptcy law (including claims, counterclaims and causes of action against any health plan or other third party payors related to services provided prior to the Effective Time), and (C) any rights to challenge liens asserted against property of the Sellers' bankruptcy estate (including, but not limited to, liens attaching to the Purchase Price paid to the Sellers), and the proceeds from any of the foregoing; *provided, however*, that Purchaser shall acquire and be deemed to release and waive as of the Effective Time avoidance claimsall Avoidance Claims against counterparties to Assigned Contracts and Assigned Leases solely to the extent such avoidance claims arise from, or are in connection with, executory contracts and unexpired leases assigned by the Sellers to Purchaser pursuant to Section 1.11 hereof;

(m) Except as set forth in <u>Section 1.7(s)</u>, all insurance policies and contracts and coverages obtained by Sellers or listing Sellers as insured party, a beneficiary or loss payee, including prepaid insurance premiums, and all rights to insurance proceeds under any of the foregoing, and all subrogation proceeds related to any insurance benefits arising from or relating to (i) Assets prior to the Licensure Date and/or (ii) Excluded Assets whether prior to or after the Licensure Date;

(n) all deposits made with any entity that provides utilities to the Hospital (the "Utility Deposits");

(o) all unclaimed property of any third party as of the Effective Time, including, without limitation, property which is subject to applicable escheat laws;

(p) all bank accounts of Sellers, other than the Lockboxes as set forth in <u>Section 1.7(o);</u>

(q) all writings and other items that are protected from discovery by the attorney-client privilege, the attorney work product doctrine or any other cognizable privilege or protection;

(r) the rights of Sellers to receive mail and other communications with respect to Excluded Assets or Excluded Liabilities;

(s) all tax refunds and tax assets of Sellers;

(t) all documents, records, operating manuals and film pertaining to the Hospital that the parties Parties agree that Sellers is required by law to retain;

(u) all patient records and medical records which are not part of any electronic medical record software transferred to Purchaser and are not required by law (including Section 351 of the Bankruptcy Code) to be maintained by Purchaser as of the Effective Time;

(v) any rights or documents relating to any Excluded Liability or other Excluded Asset;

(w) all deposits or other prepaid charges and expenses paid in connection with or relating to any other Excluded Assets and/or Excluded Liabilities;

### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 16 of 111

(x) any rights or remedies provided to Sellers under this Agreement and each other document executed in connection with the Closing and the actions necessary to complete the sale of the Hospital pursuant to this Agreement;

(y) any rights or remedies, including deposits, against any individual or entity arising pursuant to (including in connection with Sellers' termination of) or relating to that certain Asset Purchase Agreement dated January 8, 2019 between, *inter alia*, Sellers and Strategic Global Management, Inc.;

(z) any (i) personnel files for employees of Sellers who are not hired by Purchaser; (ii) all documents, records, correspondence (including with respect to any employees), work papers, patient records or other books and records that Sellers is required by Law to retain; *provided, however*, that except as prohibited by Law and subject to <u>ARTICLE 5</u>, Purchaser shall have the right to make copies of any portions of such retained books and records that relate to the business of the Hospital as conducted before the Closing or that relate to any of the Assets; (iii) documents which Sellers is not permitted to transfer pursuant to any contractual obligation owed to any third party; (iv) documents primarily related to any Excluded Assets; and (v) documents necessary to prepare tax returns (Purchaser shall be entitled to a copy of such documents). With respect to documents necessary to prepare cost reports, Purchaser shall receive the original document and Sellers shall be entitled to retain or request a copy of such documents;

(aa) any and all rights and entitlements of Sellers in respect of that certain Settlement Agreement, executed as of April 29, 2019, by and between, on the one hand, Premier, Inc., Premier Services, LLC ("**Premier GP**"), Premier Healthcare Alliance, L.P. ("**Premier LP**"), Premier Healthcare Solutions, Inc. ("**PHSI**") and each of Premier, Inc.'s other subsidiaries (collectively and including Premier GP, Premier LP and PHSI, "**Premier**"), and on the other hand, <del>Verity Health System of California, Inc., formerly known as Daughters of Charity Health System ("VHS")<u>VHS</u>, as approved by the Bankruptcy Court by order entered on May 29, 2019 [Docket No. 2461], including but not limited to the right to convert and exchange partnership interests arising under that certain Amended and Restated Limited Partnership Agreement, effective as of October 1, 2013, as amended, by and among Premier LP, Premier GP and the limited partners of Premier LP party thereto (including VHS);</del>

(bb) any grant or governmental awards received, or allocable to patient services performed, prior to the Effective Time; and

(cc) (bb) any assets identified in <u>Schedule 1.8(bbcc</u>).

For the avoidance of doubt, Purchaser is not acquiring any asset owned by any affiliate of Sellers.

1.9 <u>Transferred Obligations</u>. Purchaser is not assuming any liabilities of Sellers. Instead, on and after the Closing Date, Purchaser shall be responsible for and agrees to discharge, perform and satisfy fully, on and after the Effective Time, the following liabilities and obligations (collectively, the "**Transferred Obligations**"):

(a) the Assigned Contracts, after Sellers pay the Cure Costs from the proceeds of the Purchase Price;

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 17 of 111

(b) the Assigned Leases, after Sellers pay the Cure Costs from the proceeds of the Purchase Price;

(c) all liabilities and obligations arising out of or relating to any act, omission, event or occurrence connected with the use, ownership or operation by Purchaser of the Hospital or any of the Assets on or after the Effective Time;

(d) all liabilities and obligations related to the Hired Employees arising on or following the Effective Time (which for avoidance of doubt, does not include any duties, obligations or liabilities arising from or related to employment-related documentation required to be maintained by such Seller prior to the Effective Time, including but not limited to, documentation of I-9 compliance for the Hired EmployeesSellers' employees and any alternative work schedule compliance duties, obligations or liabilities that relate to the Hired EmployeesSellers' employees);

(e) all unpaid real and personal property taxes, if any, that are attributable to the Assets after the Effective Time, subject to the prorations provided in <u>Section 1.6;</u>

(f) all liabilities and obligations relating to utilities being furnished to the Assets, subject to the prorations provided in <u>Section 1.6;</u>

(g) any documentary, sales and transfer tax liabilities of Sellers incurred as a result of the consummation of the transaction contemplated by this Agreement;

- (h) all liabilities or obligations provided for in <u>Section 5.3</u>; and
- (i) any other obligations and liabilities identified in <u>Schedule 1.9(i)</u>.

1.10 <u>Excluded Liabilities</u>. Purchaser shall have those duties, obligations and liabilities set forth in this Agreement, the IMA, the Leaseback Agreement, the Transition Services Agreement, the Bill of Sale, the Transfer Agreement and the Real Estate Assignment(s) and shall be responsible for the Transferred Obligations. However, Purchaser is not assuming any liabilities of Sellers related to the Assets, <u>Hired EmployeesSellers' employees</u> or the Hospitals, and to the maximum extent permitted by law shall not be deemed a successor to Sellers or their estates by reason of any theory of law or equity with respect to any claims or liens against Sellers or the Assets (the "**Excluded Liabilities**").

1.11 Designation of Assigned Contracts and Assigned Leases. Each Seller will assign to Purchaser the Assigned Leases and such other contracts and leases as are subject to evaluation by Purchaser for assumption or rejection (collectively "**Evaluated Contracts**"). At the later of (i) 5:00 p.m. pacific time on the day that is seven (7) calendar days prior to the date of the auction for the Assets and (ii) the date the Purchaser submits its bid for the Assets: (A) Purchaser shall notify Seller in writing signed and dated by Purchaser of which Evaluated Contracts are to be assigned to Purchaser (the "Assigned Contracts") and (B) Purchaser shall notify Seller in writing signed and dated by Purchaser of which Evaluated Contracts are to be rejected by Seller (the "**Rejected Contracts**"); provided, that Purchaser shall have the right to designate any contracts on the Assigned Contract list as a Rejected Contract on or prior to 5:00 p.m. pacific

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 18 of 111

time on the day that is thirty (30) days prior to Closing Date, provided further thatand Sellers shall have the absolute right to remove any Evaluated Contract from the list of Assigned Contracts in order to preserve avoidance claims; provided, however, that notwithstanding anything to the contrary, and to enable Sellers to comply with the terms of the Bidding Procedures Order with respect to UnitedHealthcare Insurance Company ("UnitedHealthcare"), Purchaser shall irrevocably designate all UnitedHealthcare agreements as Assigned Contracts or Rejected Contracts by the date that is two (2) calendars days after the Bankruptcy Court enters the Sale Order. The final list of Assigned Contracts at Closing will appear on Schedule 1.7(d).

# <u>1.12</u> <u>Accounts Receivable Reconciliation.</u>

(a) Sellers and Purchaser have mutually agreed that the target Accounts Receivable amount that will be transferred from Sellers to Purchaser as of the Effective Time is the A/R Target Amount. In connection with the Account Receivable reconciliation process, Purchaser shall collect the Accounts Receivable during the one hundred thirty-five (135) day period immediately following the Closing Date (collectively, the "Final A/R Collected"), and within the thirty (30) day period thereafter, Purchaser shall provide Sellers, in good faith, a schedule which provides an accounting of the Final A/R Collected (the "A/R Accounting Schedule"), together with reasonably detailed schedules and data supporting such accounting.

(b) After receipt of the A/R Accounting Schedule, Sellers shall have sixty (60) days to review the A/R Accounting Schedule and the Final A/R Collected as proposed by Purchaser, together with the work papers used in the preparation thereof, and have their representatives and advisors review such A/R Accounting Schedule and proposed Final A/R Collected. In connection with the review of the A/R Accounting Schedule and proposed Final A/R Collected, Purchaser shall give, and shall cause its representatives and advisors to give, to Sellers and their representatives and advisors reasonable access, upon reasonable prior notice, to the books, records and other materials and the personnel of, and work papers prepared by or for, Purchaser and its representatives and advisors, including to such historical financial information relating to the Sellers and the Accounts Receivable as Sellers or their representatives or advisors may request, in each case, in order to permit the timely and complete review of the A/R Accounting Schedule and proposed Final A/R Collected in accordance with this Section 1.12(b) and so long as such access does not unreasonably interfere with the operations of the Purchaser.

(c) If Sellers disagree with the A/R Accounting Schedule and/or proposed Final A/R Collected, they shall notify Purchaser in writing within sixty (60) days after the date on which Purchaser delivers such A/R Accounting Schedule and proposed Final A/R Collected to Sellers, which shall include the items as to which they disagree and their calculation of such disputed amounts with reasonable supporting detail (the "Statement of Objections"). Purchaser and Sellers shall reasonably cooperate to resolve any such disagreements. If Purchaser and Seller are unable to resolve all such disagreements on or before the date which is thirty (30) days following notification by Sellers of any such disagreements, such disagreements shall be submitted to the Bankruptcy Court for resolution. The A/R Accounting Schedule and Final A/R Collected shall become final on the earlier of (i) failure by Sellers to deliver a Statement of Objections within the time period required by this section; (ii) mutual written agreement by Sellers and Purchaser; or (iii) a determination by the Bankruptcy Court.

### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 19 of 111

(d) Once the A/R Accounting Schedule and Final A/R Collected have become final in accordance with Section 1.12(c), the following shall occur:

(i) if the Final A/R Collected is more than the A/R Target Amount, then such excess amount shall be paid by Purchaser to Sellers within ten (10) business days of Purchaser's delivery of the accounting of the Final A/R Collected to Sellers;

(ii) if the Final A/R Collected is less than the A/R Target Amount, then such deficit amount shall be paid by Sellers to Purchaser within ten (10) business days of Purchaser's delivery of the accounting of the Final A/R Collected to Sellers; or

(iii) if the Final A/R Collected is equal to the A/R Target Amount, then no adjusting payments in respect of the Accounts Receivable shall be required by either Purchaser or Sellers.

(e) During the one hundred thirty-five (135) day period immediately following the Closing Date Purchaser shall (i) use good faith, commercially reasonable best efforts to collect the Accounts Receivable (including at least the efforts used by Purchaser to collect its other receivables) within the one hundred thirty-five (135) day period immediately following the Closing Date; (ii) not take any action for the purpose of or which would be reasonably likely to result in any of the Accounts Receivable not being collected in a timely manner and within the one hundred thirty-five (135) day period immediately following the Closing Date; and (iii) provide Sellers with weekly written updates on its collection of the Accounts Receivable.

<u>1.13</u> <u>1.12</u> <u>Disclaimer of Warranties; Release.</u>

THE ASSETS TRANSFERRED TO PURCHASER WILL BE SOLD BY (a) SELLERS AND PURCHASED BY PURCHASER IN THEIR PHYSICAL CONDITION AT THE EFFECTIVE TIME, "AS IS, WHERE IS AND WITH ALL FAULTS AND NONCOMPLIANCE WITH LAWS" WITH NO WARRANTIES, INCLUDING, WITHOUT LIMITATION. THE WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE, SUITABILITY, USAGE, WORKMANSHIP, QUALITY, PHYSICAL CONDITION, OR VALUE (OTHER THAN EXPRESSLY SET FORTH IN ARTICLE 2), AND ANY AND ALL SUCH OTHER REPRESENTATIONS AND WARRANTIES ARE HEREBY EXPRESSLY DISCLAIMED, AND WITH RESPECT TO THE LEASED REAL PROPERTY WITH NO WARRANTY OF HABITABILITY OR FITNESS FOR HABITATION, INCLUDING, WITHOUT LIMITATION, THE LAND, THE BUILDINGS AND THE IMPROVEMENTS. ALL OF THE PROPERTIES, ASSETS, RIGHTS, LICENSES, PERMITS, PRIVILEGES, LIABILITIES, AND OBLIGATIONS OF SELLERS INCLUDED IN THE ASSETS AND THE TRANSFERRED OBLIGATIONS ARE BEING ACQUIRED OR RECEIVED "AS IS, WHERE IS" ON THE CLOSING DATE AND IN THEIR PRESENT CONDITION, WITH ALL FAULTS. ALL OF THE TANGIBLE ASSETS SHALL BE FURTHER SUBJECT TO NORMAL WEAR AND TEAR AND NORMAL AND CUSTOMARY USE OF THE INVENTORY AND SUPPLIES IN THE ORDINARY COURSE OF BUSINESS UP TO THE EFFECTIVE TIME.

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 20 of 111

(b) Purchaser acknowledges that Purchaser has examined, reviewed and inspected all matters which in Purchaser's judgment bear upon the Purchase Price, the Assets, the Sellers, the Hospital, the businesses of the Hospital and their value and suitability for Purchaser's purposes and, except for Sellers representations and warranties in <u>ARTICLE 2</u>, is relying solely on Purchaser's own examination, review and inspection of the Assets and Transferred Obligations. Purchaser hereby releases Sellers and their affiliates from all responsibility and liability regarding the condition, valuation, salability or utility of the businesses of the Hospital or the Assets, or their suitability for any purpose whatsoever. Purchaser further acknowledges that the representations and warranties of Sellers contained in <u>ARTICLE 2</u> of this Agreement are the sole and exclusive representations and warranties made by Sellers to Purchaser (including with respect to the Hospital, the Assets and the Transferred Obligations) and shall expire, and be of no further force or effect at the Closing.

# ARTICLE 2 REPRESENTATIONS AND WARRANTIES OF SELLERS

Each Seller hereby represents, warrants and covenants to Purchaser, severally (and not jointly) with respect to such Seller that the following matters are true and correct as of the Signing Date, except as would not have a Material Adverse Effect (as defined below) upon the Assets, and except as disclosed in the disclosure schedule, as may be amended pursuant to the terms of this Agreement (the "**Disclosure Schedule**"), provided that the representations and warranties set forth in <u>Sections 2.1</u> (Authorization), <u>2.2</u> (Binding Agreement), <u>2.3</u> (Organization and Good Standing; No Violation), <u>2.7</u> (Compliance with Legal Requirements), <u>2.8</u> (Required Consents), <u>2.10</u> (Title) and <u>2.13</u> (Legal Proceedings) (the "**Sale Order Date Representations**") shall also be made as of immediately prior to the entry of the Sale Order (the "**Sale Order Date"**):

The term "**Material Adverse Effect**" means any event, change or occurrence that, individually or in the aggregate with other events, changes or occurrences, has had or would reasonably be expected to have, a material adverse effect on the <u>Sellers'</u> financial condition, operation or Assets in excess of Fifty Million Dollars (\$50,000,000)<del>;</del> provided, that the term <u>Material Adverse Effect shall not include any event, change or occurrence resulting from: (a) the SARS-CoV-2 virus or mutations therefrom or in connection with the disease COVID-19; (b) changes to the U.S. economy, the global economy, in each case, as a whole, or the Hospital industry (except to the extent such changes disproportionately affect the Business); or (c) general economic, regulatory or political conditions or changes in the geographic regions in which the Hospital operates, except to the extent any such event described in (b), (c) or (d) disproportionately affects or impacts the Sellers' financial condition, operation or Assets as compared to other similarly situated businesses in the industry.</u>

2.1 <u>Authorization</u>. Such Seller has all necessary corporate power and authority to enter into this Agreement and, subject to Bankruptcy Court approval, to carry out the transactions contemplated hereby.

2.2 <u>Binding Agreement</u>. This Agreement has been duly and validly executed and delivered by Sellers and, assuming due and valid execution by Purchaser, this Agreement constitutes a valid and binding obligation of Sellers enforceable in accordance with its terms

### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 21 of 111

subject to (a) applicable bankruptcy, reorganization, insolvency, moratorium and other laws affecting creditors' rights generally from time to time in effect (including, without limitation approval of the Bankruptcy Court) and (b) limitations on the enforcement of equitable remedies.

# 2.3 <u>Organization and Good Standing; No Violation</u>.

(a) Each of SFMC and VHS is a non-profit corporation duly organized, validly existing and in good standing under the laws of the State of California. Verity Holdings is a limited liability company duly organized, validly existing and in good standing under the laws of the State of California. Each Seller has all necessary power and authority to own, operate and lease its properties and to carry on its businesses as now conducted.

(b) Neither the execution and delivery by such Seller of this Agreement nor the consummation of the transactions contemplated hereby by such Seller nor compliance with any of the material provisions hereof by such Seller, will violate, conflict with or result in a breach of any material provision of such Seller's articles of incorporation or bylaws or any other organizational documents of such Seller.

2.4 <u>Contracts</u>. Except as set forth in <u>Schedule 2.4</u>, upon entry of the Sale Order and payment of the Cure Costs, to Sellers' knowledge, Sellers are not in breach or default of the Assigned Contracts or Assigned Leases. No provision of this <u>Section 2.4</u> shall apply to any failure to obtain consents to the assignment of the Assigned Contracts and/or Assigned Leases from third parties in connection with the assignment of such Assigned Contracts and/or Assigned Leases to Purchaser (the "**Contract and Lease Consents**").

2.5 <u>Brokers and Finders</u>. Except as set forth on <u>Schedule 2.5</u>, neither such Seller nor any affiliate thereof, nor any officer or director thereof, have engaged or incurred any liability to any finder, broker or agent in connection with the transactions contemplated hereunder.

2.6 <u>Non-Contravention</u>. Neither the execution and delivery by Sellers of this Agreement and each Ancillary Agreement nor performance of any of the material provisions hereof by Sellers, will violate, conflict with or result in a breach of any material provisions of the articles of incorporation or bylaws of Sellers.

2.7 <u>Compliance with Legal Requirements</u>. Except as set forth in <u>Schedule 2.7</u>, to the knowledge of Sellers: each Seller, with respect to the operation of the Hospital, is in material compliance with all applicable laws, statutes, ordinances, orders, rules, regulations, licenses, certificates, judgments or decrees of all judicial or governmental authorities (federal, state, local, foreign or otherwise) (collectively, "Legal Requirements"). Except as set forth in <u>Schedule 2.7</u>, to the knowledge of Sellers, none of the Sellers, with respect to the operation of the Hospital, has been charged in writing with or been given written notice of any material violation or any obligation to take material remedial action under, any applicable Legal Requirements.

2.8 <u>Required Consents</u>. Except as set forth in <u>Schedule 2.8</u>, and other than in connection with any Licenses, any provider agreements (including any such agreements with a governmental authority) and the Attorney General, Sellers are not a party to or bound by, nor are any of the Assets subject to, any mortgage, or any material lien, deed of trust, material lease, or

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 22 of 111

material contract or any material order, judgment or decree which, after giving effect to the Sale Order (a) will require the consent of any third party to the execution of this Agreement or (b) will require the consent of any third party to consummate the transactions contemplated by this Agreement.

## 2.9 <u>Environmental Matters</u>

(a) Sellers have provided Purchaser with the Phase I Environmental Site Assessments in Seller's electronic data room.

(b) Except as disclosed in <u>Schedule 2.9(b)</u>, to the knowledge of Sellers, the operations of the Hospital are not in material violation of any applicable limitations, restrictions, conditions, standards, prohibitions, requirements and obligations of Environmental Laws and related orders of any court or any other governmental authority.

(c) For the purposes of this Section, the term "Environmental Laws" shall mean all state, federal or local laws, ordinances, codes or regulations relating to Hazardous Substances or to the protection of the environment, including, without limitation, laws and regulations relating to the storage, treatment and disposal of medical and biological waste. For purposes of this Agreement, the term "Hazardous Substances" shall mean (i) any hazardous or toxic waste, substance, or material defined as such in (or for the purposes of) any Environmental Laws, (ii) asbestos-containing material, (iii) medical and biological waste, (iv) polychlorinated biphenyls, (v) petroleum products, including gasoline, fuel oil, crude oil and other various constituents of such products, and (vi) any other chemicals, materials or substances, exposure to which is prohibited, limited or regulated by any Environmental Laws.

2.10 <u>Title</u>. Sellers have delivered at their own expense for all Owned Real Property (i) preliminary title reports issued by Chicago Title <u>Insurance Company</u> (the "**Title Commitments**") and (ii) copies of or access to all material underlying title documents listed on the Title Commitments (the "Underlying Title Documents" and collectively with the Title <u>Commitments, the "Title Documents"</u>)...

2.11 Certain Other Representations with Respect to the Hospitals.

(a) Except as set forth in <u>Schedule 2.11</u>, all Licenses which are material and necessary to the operation of the Hospital by SFMC are valid and in good standing and SFMC is in compliance with the terms and conditions of all such Licenses in all material respects, in each case except where the failure to be valid and in good standing or in compliance would not have a Material Adverse Effect on the Assets. Except as set forth in <u>Schedule 2.11</u>, as of the Closing Date, SFMC will have all material Licenses required under Legal Requirements to operate the Hospital as presently operated by SFMC, except where the failure to have any such License would not have a Material Adverse Effect on the Assets. To the knowledge of Sellers, no loss of any License is pending or threatened.

(b) SFMC is certified for participation in the Medicare, Medi-Cal and TRICARE programs and any other federal or state health care reimbursement programs in which SFMC participates, and has current and valid provider agreements with each such program,

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 23 of 111

except where the failure to be so certified or have such provider agreements would not have a Material Adverse Effect.

(c) SFMC has not been excluded from Medicare, Medi-Cal or any federal or state health care reimbursement program, and, to the knowledge of Sellers, there is no pending or threatened exclusion action by a governmental authority against SFMC.

## 2.12 <u>Financial Statements</u>.

(a) Sellers have provided to Purchaser: (i) the unaudited balance sheets of the Sellers as of June 30, 2018; (ii) unaudited income statements of the Sellers for the twelve-month periods ended June 30, 2018; (iii) the audited consolidated income statements of Sellers for the years ended 2016 and 2017; and (iv) the unaudited consolidated balance sheet of Sellers as of June 30, 2018 (collectively, the "Historical Financial Statements").

(b) the income statements contained in the Historical Financial Statements present, fairly in all material respects the results of the operations of the Sellers as of and for the periods covered therein and the balance sheets contained in the Historical Financial Statements (i) are true, complete and correct in all material respects; (ii) present, fairly in all material respects the financial condition of the Sellers as of the dates indicated thereon; and (iii) to the extent prepared by an independent certified public accounting firm, have been prepared in accordance with generally accepted accounting principles consistently applied throughout the periods covered, except as disclosed therein.

Legal Proceedings. Except as set forth on Schedule 2.13, and except for any and 2.13 all cases and/or pleadings filed or to be filed in the Bankruptcy Court, which shall be available through Sellers' claims and noticing agent's website at http://www.kccllc.com/VERITYHEALTH/, to the knowledge of Sellers, there are no material claims, proceedings or investigations pending or threatened with respect to the ownership of the Assets or the operation of the Hospital or the Hospital by Sellers before any governmental authority. Except as set forth on Schedule 2.13, and other than any action or proceeding brought in the Bankruptcy Court, to the knowledge of Sellers, Sellers are not subject to any government order with respect to the ownership or operation by SFMC of the Hospital and SFMC is in substantial compliance with respect to any such government order.

2.14 <u>Employee Benefits</u>. Sellers have provided Purchaser with a list of (i) each pension, profit sharing, bonus, deferred compensation, or other retirement plan or arrangement of Seller with respect to the operation of the Hospital, whether oral or written, which constitutes an "employee pension benefit plan" as defined in Section 3(2) of ERISA, (ii) each medical, health, disability, insurance or other plan or arrangement of Seller with respect to the operation of the Hospital, whether oral or written, which constitutes an "employee welfare benefit plan" as defined in Section 3(1) of ERISA, and (iii) each other employee benefit or perquisite provided by Seller with respect to the operation of the Hospital, in which any employee of Seller participates in his/her capacity as such (collectively, the "Seller Plans").

2.15 <u>Personnel</u>. Sellers have provided Purchaser with a complete list (as of the date set forth therein) of names, positions and current annual salaries or wage rates and scheduled bonus,

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 24 of 111

and the accrued paid time off pay of all employees of Sellers (including employees of the Hospital and employees of VHS and Verity Holdings) immediately prior to [February 29, 2020], whether such employees are full time employees, part-time employees, on short-term or long-term disability or on leave of absence pursuant to Sellers's policies, the Family and Medical Leave Act of 1993 or other similar Legal Requirements (the "**Hospital Employees**") and indicating whether the Hospital Employee is full- time or part-time. Sellers shall provide an updated list to Purchaser no later than five (5) Business Days before the date scheduled for the Closing.

2.16 <u>Insurance</u>. Sellers have provided Purchaser with a list of all material insurance maintained by Sellers with respect to the Assets, as of the Signing Date.

2.17 <u>Accounts Receivable</u>. To the knowledge of Sellers, all Accounts Receivable included in the Assets at Closing result from the bona fide provision of products or services in the ordinary course of business. All proceeds of Sellers' Accounts Receivable are currently deposited, either electronically or manually, into those bank accounts provided to Purchaser in Seller's electronic data room.

2.18 <u>Payor Contracts</u>. To the knowledge of Sellers, and subject to Section 365 of the Bankruptcy Code, Sellers have provided Purchaser with a complete list of all written contracts with private third-party payors including insurance companies and HMOs. Sellers have provided Purchaser with a true and correct copy of all material Payor Contracts.

2.19 Excluded Individuals. Except as set forth on Schedule 2.19, to the knowledge of Sellers: neither Sellers, Hospital nor any director, officer or employee of Sellers or Hospital (a) was, is or is proposed to be, suspended, excluded from participation in, or sanctioned under, any federal or state health care program (including, without limitation, Medicare and Medicaid) (an "**Excluded Individual**"); (b) has been convicted of any criminal offense related to the delivery of any medical or health care services or supplies, or related to the neglect or abuse of patients; (c) has failed to maintain its current License to provide the services required to be provided by it to or on behalf of Sellers and Hospital; or (d) is unable to obtain or maintain liability insurance consistent with commercially reasonable industry practices.

2.20 <u>Seller Knowledge</u>. References in this Agreement to "Sellers' knowledge" or "the knowledge of Sellers" means the actual knowledge of the Chief Executive Officer and Chief Financial Officer of the applicable Seller, without independent research. No constructive or imputed knowledge shall be attributed to any such individual by virtue of any position held, relationship to any other Person or for any other reason.\_\_

## ARTICLE 3 REPRESENTATIONS AND WARRANTIES OF PURCHASER

As an inducement to Sellers to enter into this Agreement and to consummate the transactions contemplated by this Agreement, Purchaser hereby represents and warrants to Sellers as to the following matters as of the Signing Date and as of the Closing Date:

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 25 of 111

3.1 <u>Authorization</u>. Purchaser has full power and authority to enter into this Agreement and has full power and authority to perform its obligations hereunder and to carry out the transactions contemplated hereby. No additional internal consents are required in order for Purchaser to perform its obligations and agreements hereunder.

3.2 <u>Binding Agreement</u>. This Agreement has been duly and validly executed and delivered by Purchaser and, assuming due and valid execution by Sellers, this Agreement constitutes a valid and binding obligation of Purchaser enforceable in accordance with its terms subject to (a) applicable bankruptcy, reorganization, insolvency, moratorium and other laws affecting creditors' rights generally from time to time in effect, and (b) limitations on the enforcement of equitable remedies.

3.3 <u>Organization and Good Standing</u>. Purchaser is a corporation duly organized, validly existing and in good standing under the laws of the State of California, is or will be duly authorized to transact business in the State of California, and has full power and authority to own, operate and lease its properties and to carry on its business as now conducted.

3.4 <u>No Violation</u>. Except as set forth in <u>Schedule 3.4</u>, neither the execution and delivery by Purchaser of this Agreement nor the consummation of the transactions contemplated hereby nor compliance with any of the material provisions hereof by Purchaser will (a) violate, conflict with or result in a breach of any material provision of the Articles of Incorporation, Bylaws or other organizational documents of Purchaser or any contract, lease or other instrument by which Purchaser is bound; (b) require any approval or consent of, or filing with, any governmental agency or authority, (c) violate any law, rule, regulation, or ordinance to which Purchaser is or may be subject, <u>or (d) violate any judgment</u>, order or decree of any court or other governmental agency or authority to which Purchaser is subject.

3.5 <u>Brokers and Finders</u>. Neither Purchaser nor any affiliate thereof nor any officer or director thereof has engaged any finder or broker in connection with the transactions contemplated hereunder.

3.6 <u>Representations of Sellers</u>. Purchaser acknowledges that it is purchasing the Assets on an "AS IS, WHERE IS" basis (as more particularly described in <u>Section 1.12</u>), and that Purchaser is not relying on any covenant, representation, warranty (expressed or implied, oral or otherwise) made by or on behalf of Sellers, or any other provision of this Agreement, in entering into and performing under this Agreement. Purchaser further acknowledges that Sellers are not making any covenants, representations or warranties herein relating to the Assets or the operation of the Hospital on or after the Effective Time.

3.7 <u>Legal Proceedings</u>. Except as described on <u>Schedule 3.7</u>, there are no claims, proceedings or investigations pending or, to the best knowledge of Purchaser, threatened relating to or affecting Purchaser or any affiliate of Purchaser before any court or governmental body (whether judicial, executive or administrative) in which an adverse determination would adversely affect Purchaser's ability to consummate the transactions contemplated hereby. Neither Purchaser nor any affiliate of Purchaser is subject to any judgment, order, decree or other governmental restriction specifically (as distinct from generically) applicable to Purchaser or any

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 26 of 111

affiliate of Purchaser which would adversely affect Purchaser's ability to consummate the transactions contemplated hereby.

No Knowledge of Sellers' Breach. Neither Purchaser nor any of its affiliates has 3.8 knowledge of any breach of any covenant, representation or warranty by Sellers or of any condition or circumstance that would give Purchaser a right to terminate this Agreement pursuant to Section 9.1(c). If information comes to Purchaser's attention on or before the Closing Date (whether through Sellers or otherwise and whether before or after the Signing Date) which indicates that Sellers have breached any of their covenants, representations, warranties or any other provision or condition under this Agreement, then the effect shall be as if the covenants, representations and warranties or any other provision or condition of this Agreement had been modified in accordance with the actual state of facts existing prior to the Effective Time such that there will be no breach under Sellers' covenants, representations and warranties or any other provision or condition of this Agreement in relation to such information; provided, further, that Purchaser must immediately notify Sellers if any such breach comes to its attention on or before the Closing Date, and Purchaser's failure to so notify Sellers shall constitute a waiver by Purchaser of Sellers' breach, if any, of this Agreement or any ancillary agreements entered into pursuant to this Agreement. Upon written request of Sellers, Purchaser shall promptly confirm and remake this representation in writing.

3.9 <u>Ability to Perform.</u> Purchaser has the ability to obtain funds and at the Closing shall have cash in amounts necessary to consummate the transactions contemplated by this Agreement by means of cash, credit facilities or otherwise. Upon the earlier of ten (10) days after satisfaction of <u>Section 8.3</u> or five (5) days prior to the Closing Date, Purchaser shall have delivered to Sellers true, correct and complete copies of (i) executed commitment letter(s) from one or more banks or other lending institutions or sources (the "**Debt Commitment Letter**"), pursuant to which, and subject to the terms and conditions thereof, the lender parties thereto have committed to lend the amounts set forth therein to Purchaser for the purpose of funding the transactions contemplated by this Agreement and (ii) any executed equity commitment letter(s) (the "**Equity Commitment Letter**" and, together with the Debt Commitment Letter, the "**Financing Commitments**"). The Financing Commitments provide sufficient funds to consummate the transactions contemplated by this Agreement.

3.10 Investigation. Purchaser has been afforded reasonable access to, and has been provided adequate time to review, the books, records, information, operations, facilities and personnel of Sellers and the Hospital for purposes of conducting a due diligence investigation of Sellers and the Hospital. Purchaser has conducted a reasonable due diligence investigation of Sellers and the Hospital and has received satisfactory answers to all inquiries it has made respecting Sellers and the Hospital and has received all information it considers necessary to make an informed business evaluation of Sellers and the Hospital. In connection with its due diligence investigation of Sellers and the Hospital, purchaser has not relied upon any books, records, information, operations, facilities and personnel provided by Sellers, including in making its determination to enter into this Agreement and/or consummate the transactions contemplated hereby. Purchaser has completed all of its due diligence of Sellers and the Hospital and this Agreement is not subject to any further due diligence of Sellers and the Hospital by Purchaser.

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 27 of 111

3.11 <u>Purchaser Knowledge</u>. References in this Agreement to "Purchaser's knowledge" or "the knowledge of Purchaser" means the actual knowledge of the Chief Executive Officer and Chief Financial Officer of Purchaser, without independent research. No constructive or imputed knowledge shall be attributed to any such individual by virtue of any position held, relationship to any other Person or for any other reason.

### ARTICLE 4 COVENANTS OF SELLERS

### 4.1 Access and Information; Inspections.

4.1.1 From the Signing Date through the Effective Time, (a) Sellers shall <u>(a)</u> afford to the officers and agents of Purchaser (which shall include accountants, attorneys, bankers and other consultants and authorized agents of Purchaser) reasonable access during normal business hours at Sellers' corporate headquarters in Los Angeles, California to, and the right to inspect, the books, accounts, records and all other relevant documents and information with respect to the assets, liabilities and business of the Hospital and the plant and property of the Hospital at the Hospital and (b) Sellers shall furnish Purchaser with such additional financial and operating data and other information in Sellers' possession as to the businesses and properties of the Hospital as Purchaser or its representatives may from time to time reasonably request; provided, however, that Sellers are not obligated to disclose information which is proprietary to Sellers and would not be essential to the ongoing operation of the Hospital by Purchaser; provided, further, that all disclosures of information shall be consistent with the confidentiality agreements and any other non-disclosure agreements entered into (or to be entered into) among Purchaser, its representatives and Sellers or their representatives. Purchaser's right of access and inspection shall be exercised in such a manner as not to interfere unreasonably with the operations of Sellers or the Hospital.

(b) 4.1.2 Notwithstanding anything contained herein, Sellers shall not be required to provide Purchaser or its representatives or agents access to or disclose information where such access or disclosure would violate the rights of its patients, jeopardize the attorney-client or similar privilege with respect to such information or contravene any law, judgment, fiduciary duty or contract entered into prior to or on the date of this Agreement with respect to such information.

#### 4.2 <u>Cooperation</u>.

(a) 4.2.1 The parties Parties shall reasonably cooperate with each other and their respective authorized representatives and attorneys\_in: (a) in all efforts to obtain all consents, approvals, authorizations, clearances and licenses required to carry out the transactions contemplated by this Agreement (including, without limitation, those of governmental and regulatory authorities), (b) in the preparation of any document or other material which may be required by any governmental agency as a predicate to or result of the transactions contemplated in this Agreement, and (c) in Purchaser's efforts to effectuate the assignment of Assigned Contracts to Purchaser as of the Closing Date.

(b) 4.2.2 Except as may be otherwise requested by Sellers in order to comply

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 28 of 111

with applicable law or regulatory guidance, notwithstanding anything contained herein, other than Bankruptcy Court orders and authorizations, it shall be Purchaser's sole responsibility (including payment of any fees, expenses, filing costs or other amounts) to obtain the Contract and Lease Consents, as well as all governmental consents, approvals, assignments, authorizations, clearances and licenses required to (x) carry out the transactions contemplated by this Agreement and/or (y) transfer any of the Assets, including any Licenses, Transferred Managed Care Agreements and/or Transferred Private Payor Agreements, not including the application required to be submitted to the California Attorney General and the transfer of the Medicare Provider Agreement and the Medi-Cal Provider Agreement, which are dealt with in Sections 5.9.35.9(c) and 4.5 hereto respectively. Upon request, Sellers shall provide Seller-specific information that may be needed by Purchaser to obtain such Contract and Lease Consents and such governmental consents, approvals, assignments, authorizations, clearances and licenses.

(c) 4.2.3 Notwithstanding any provision to the contrary contained in this Agreement, Sellers shall not be obligated to obtain the approval or consent to the assignment, to Purchaser, of any Assigned Contracts or Assigned Leases, from any party to any of the Assigned Contracts or Assigned Leases even if any such contract or lease states that it is not assignable without such party's consent.

4.3 <u>Sellers' Efforts to Close</u>. Sellers shall use their reasonable commercial efforts to satisfy all of the conditions precedent set forth in <u>ARTICLE 7</u> and <u>ARTICLE 8</u> to their or Purchaser's obligations under this Agreement to the extent that Sellers' action or inaction can control or materially influence the satisfaction of such conditions; *provided, however*, that Sellers shall not be required to pay or commit to pay any amount to (or incur any obligation in favor of) any person (other than any filing or application fees to the extent Sellers are responsible therefor under this Agreement).

4.4 <u>Termination Cost Reports</u>. Sellers shall file all Medicare, Medi-Cal and any other termination cost reports required to be filed as a result of the consummation of (a) the transfer of the Assets to Purchaser and (b) the transactions contemplated by this Agreement, provided that Purchaser shall fund reasonable costs and expenses of preparation, filing and audit of such reports. Purchaser shall permit Sellers access to all Hospital books and records to prepare such reports and shall assist Sellers in the process of preparing, filing, and reviewing the termination cost reports. All such termination cost reports shall be filed by Sellers in a manner that is consistent with current laws, rules and regulations. Sellers shall be responsible for filing governmental cost reports for the period [\_\_\_\_\_\_]all periods through the day immediately preceding the Licensure Date. Purchaser shall be responsible for their own cost report filings relating to the Hospital beginning on the Licensure Date.

4.5 <u>Medicare and Medi-Cal Enrollment</u>. Sellers shall make commercially reasonable efforts to enter into settlement agreements with the Centers for Medicare and Medicaid Services ("CMS"), with respect to the Medicare Provider Agreement, and the California Department of Health Care Services ("DHCS"), with respect to the Medi-Cal Provider Agreement or, alternatively, obtain Bankruptcy Court rulings that the provider agreements for Medicare Provider Agreement and/or the Medi-Cal Provider Agreement may be transferred without the consent of CMS or DHCS, as applicable, and without successor liability, to enable such

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 29 of 111

agreements to be assigned to Purchaser. Between the Closing Date and the Licensure Date, Purchaser may bill and collect for patient services under Seller's health plan agreements, pursuant to the terms of the IMA and Leaseback Agreement.

4.6 Transferred Private Payor Agreements. For purposes of this Section 4.6, the following terms shall have the meanings set forth herein: (a) "Payor" means the non-debtor counter-party to a Transferred Private Payor Agreement; (b) "Overpayment" means an overpayment made by a Payor to SFMC on account of a Contracted Payment; (c) "Contracted Payment" means the contractual reimbursement due from a Payor to SFMC under a Transferred Private Payor Agreement for covered services rendered by Hospital under such agreement with a date of service on or prior to the Effective Time. Purchaser agrees that, as a condition to the assignment of any Transferred Private Payor Agreement on the Licensure Date, it shall honor any defenses to the payment of, and shall permit recoupment against, an Account Receivable due from Payor to Purchaser based on an Overpayment under such agreement, provided that, the determination and allowance of any Overpayment shall remain subject to the terms and conditions of the relevant Transferred Private Payor Agreement. Without limiting the scope of the foregoing sentence, Purchaser acknowledges and agrees that (a) the Sale Order shall authorize a Payor to continue to exercise its defenses to the payment to Purchaser of an Account Receivable based on an Overpayment, and (b) Overpayments do not constitute Cure Costs under this Agreement. This Section 4.6 shall satisfy the requirements under Sections 365(b) and (f) of the Bankruptcy Code. Following the Licensure Date, pursuant to Section 365(k) of the Bankruptcy Code, the SFMC shall be relieved and released from any obligation to any Payor on account of any Overpayment or otherwise.

4.7 <u>Hospital Operations</u>. From the date of the Sale Order until the Closing, Sellers shall, with respect to the operation of the Hospital, use commercially reasonable efforts (in each case subject to actions relating to and impacts arising from the SARS-CoV-2 virus or mutations therefrom or in connection with the disease COVID-19) to:

(a) 4.7.1 without regard to negative financial impacts or any Material Adverse Effect, carry on Sellers' operation of the Hospital consistent with past practice, but subject to the Bankruptcy Cases and Sellers' obligations and actions in connection therewith;

(b) 4.7.2 without regard to Material Adverse Effect, maintain in effect the insurance coverages with respect to the Assets;

(c) 4.7.3-without regard to Material Adverse Effect, perform Sellers' material obligations under all Assigned Contracts with respect to the Assets in compliance with the Bankruptcy Code;

(d) 4.7.4 with respect to material deficiencies, if any, cited by any governmental authority or accreditation body in the most recent surveys conducted by each, develop and implement a plan of correction (without regard to Material Adverse Effect) that is reasonably acceptable to such governmental authority or such accreditation body, but excluding any deficiencies cited (i) by the Attorney General or (ii) with respect to seismic or OSHPD-related requirements; *provided, however*, that Sellers may appeal any material deficiency citations under applicable laws if a plan of correction is not accepted by the applicable

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 30 of 111

governmental authority or accreditation body which appeal shall satisfy Sellers obligations under this subsection;

(e) 4.7.5 following entry of the Sale Order, permit and allow reasonable access by Purchaser and its representatives to make offers of post-Closing employment to any of Sellers' personnel and to establish relationships with physicians, medical staff and others having business relations with Sellers, provided, that any written materials shall be approved by Sellers prior to being sent, and provided further that such actions by Purchaser do not unreasonably interfere with Sellers' operation of the Hospital;

(f) 4.7.6-timely file or cause to be filed all material reports, notices and tax returns required to be filed and pay all required taxes as they come due, unless such failure would not have a Material Adverse Effect; and

(g) 4.7.7 without regard to Material Adverse Effect, maintain all existing material approvals, permits and environmental permits relating to the Hospital.

4.8 <u>Interim Collections</u>. Purchaser may bill and collect for patient services rendered by Hospital between the Closing Date and the Licensure Date under Seller's Private Payor Agreements and Transferred Managed Care Agreements, pursuant to the terms of the IMA and Leaseback Agreement.

Contract With Unions. Representatives of Sellers who are parties to collective 4.9 bargaining agreements and Purchaser shall meet and confer from time to time as reasonably requested by either partyParty to discuss strategic business options and alternative approaches in negotiating each collective bargaining agreement. The applicable Sellers and Purchaser shall each participate in all union negotiations related to any specific collective bargaining agreement. Promptly following the Signing Date, applicable Sellers shall use commercially reasonable efforts to initiate discussions with Purchaser and conduct discussions to renegotiate each collective bargaining agreement currently in effect with each applicable union. The applicable Sellers will not unreasonably withhold, condition or delay approval or implementation of any successfully renegotiated collective bargaining agreement. The parties Parties recognize that Seller's failure to conclude a successor collective bargaining agreement shall not be a breach of Sellers' obligation under this Agreement, provided that if the unions refuse to negotiate, or otherwise are not timely, reasonable or realistic in negotiating modified or new collective bargaining agreements during the period between the Signing Date and the Closing Date, Sellers and Purchaser will jointly consider, and negotiate mutually in good faith, alternative approaches that may be available and/or necessary to reduce Sellers' labor cost structure, including, but not limited to, seeking to reject the existing collective bargaining agreement(s).

4.10 <u>Consulting Services</u>. <u>1</u> <u>Subject to compliance with applicable Legal</u> <u>Requirements and after entry of the Sale Order if the Purchaser is the Prevailing Bidder:</u>

(a) Purchaser shall provide consulting services to the Hospital for the period prior to the Closing Date-: The

<sup>&</sup>lt;sup>1</sup> To be discussed - subject to review by Verity and BRG.

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 31 of 111

(b) the consulting services provided by Purchaser may include, but not be limited to, reviewing and advising SFMC regarding accounting and financial records, contracting, billing and collection activities, compliance with law, any of the plans or actions proposed in the following sentence, and other functions:

(c) Sellers shall not unreasonably delay or deny approval or any recommendation for anyconsider and implement, at no cost to Sellers, any mutually reasonably agreed upon plans and actions proposed by Purchaser to (a) assist the Hospital in meeting any seismic compliance deadlines, (b) stabilize and improve the operations of the Hospital or (c) develop and implement turnaround plans for the long-term viability of the Hospital...; and

(d) Purchaser shall have reasonable access to the business office and records of the Hospital, which shall include reasonable access to the Hospital's chief executive officer, chief financial officer, chief operations officer or other equivalent personnel of the Hospital reasonably necessary for Purchaser to perform the aforementioned consulting services.

Notwithstanding the foregoing, <u>VHS</u>, <u>SFMC</u> and <u>its board</u><u>their boards</u> of directors shall at all times retain ultimate control and governance over the assets and operation of the Hospital. <u>Neither VHS nor</u> <u>SFMC</u> shall<u>not</u> delegate to Purchaser any of the powers, duties and responsibilities required to own or operate the Hospital or those that are retained by <u>VHS</u> and/or <u>SFMC</u> under law (including all certificates and licenses issued under authority of law for ownership or operation of the Hospital). For the period of time during which such consulting <u>services are provided</u>, <u>Hospital and VHS shall have the non-exclusive</u>, <u>royalty-free</u>, <u>unlimited license to use any implemented Purchaser-owned intellectual property</u>.

## ARTICLE 5 COVENANTS OF PURCHASER

5.1 <u>Purchaser's Efforts to Close</u>. Purchaser shall use its reasonable commercial efforts to satisfy all of the conditions precedent set forth in <u>ARTICLE 7</u> and <u>ARTICLE 8</u> to its or Sellers' obligations under this Agreement to the extent that Purchaser's action or inaction can control or materially influence the satisfaction of such conditions.

5.2 <u>Consents</u>. Purchaser shall be entitled, but not obligated, to seek to obtain the Contract and Lease Consents. Purchaser shall be entitled, but not obligated, to solicit and seek to obtain estoppel certificates from any third party to any Leased Real Property. Purchaser's failure to obtain any or all of the Contract and Lease Consents or estoppel certificates as of the Closing Date shall not be a condition precedent to either <u>partyParty</u>'s obligation to consummate the Closing and perform all transactions contemplated by this Agreement.

5.3 <u>Certain Employee Matters</u>.

(a) Purchaser agrees to make offers of employment, effective as of the Effective Time, to substantially all persons (whether such persons are full time employees, part-time employees, on short-term or long-term disability or on leave of absence, military leave or workers compensation leave) who, immediately prior to the Effective Time are: (i) employees of SFMC; or (ii) employed by another Seller or affiliate and are listed on <u>Schedule 5.3</u>

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 32 of 111

(collectively, the "**Hospital Employees**"). For the avoidance of doubt, the Hospital Employees shall not include any employees of Verity Health System of California, Inc.<u>VHS</u> or any other affiliate of Sellers unless such individual is listed on <u>Schedule 5.3</u>. Any of the Hospital Employees who accept an offer of employment with Purchaser as of or after the Effective Time shall be referred to in this Agreement as the "**Hired Employees**." All employees who are Hired Employees shall cease to be employees of Sellers or its affiliates as of the Closing Date.

(b) After the Closing Date, Purchaser's human resources department will give reasonable assistance to Sellers and their affiliates with respect to Sellers' and Sellers' affiliates' post-Closing administration of Sellers' and Sellers' affiliates' pre-Closing employee benefit plans for the Hospital Employees. Within five (5) days after the Closing Date, Purchaser shall provide to Sellers a list of all the Hospital Employees who were offered employment by Purchaser but refused such employment along with a list of all Hired Employees (which such list Purchaser shall periodically update).

(c) With respect to any collective bargaining agreements or labor contract with respect to any union employees, Purchaser shall comply with the applicable laws, or to the extent applicable, Bankruptcy Court orders relating to collective bargaining agreements or labor contracts.

(d) The provisions of this <u>Section 5.3</u> are solely for the benefit of the <u>parties to</u> <u>this AgreementParties</u>, and no employee or former employee or any other individual associated therewith or any employee benefit plan or trustee thereof shall be regarded for any purpose as a third party beneficiary of this Agreement, and nothing herein shall be construed as an amendment to any employee benefit plan for any purpose.

5.4 <u>Waiver of Bulk Sales Law Compliance</u>. Purchaser hereby waives compliance by Sellers with the requirements, if any, of Article 6 of the Uniform Commercial Code as in force in any state in which the Assets are located and all other laws applicable to bulk sales and transfers.

5.5 <u>Conduct Pending Closing</u>. Prior to consummation of the transactions contemplated hereby or the termination or expiration of this Agreement pursuant to its terms, unless Sellers shall otherwise consent in writing, Purchaser shall not take any action or fail or omit to take any action which would cause any of Purchaser's representations and warranties set forth in <u>ARTICLE 4</u> to be inaccurate or untrue during the term of this Agreement and as of the Closing.

5.6 <u>Resale Certificate</u>. Purchaser agrees to furnish to Sellers any resale certificate or certificates or other similar documents reasonably requested by Sellers to comply with or obtain an exemption from pertinent excise, sales and use tax laws.

5.7 <u>Cure Costs</u>. On or about the Closing Date, Sellers (from the proceeds of the Purchase Price) shall pay an amount equal to the Cure Costs to each counter party to an Assigned Contract and Assigned Lease so that each such Assigned Contract and Assigned Lease may be assumed by Sellers and assigned to Purchaser in accordance with the provisions of Section 365 of the Bankruptcy Code. For purposes of this Agreement, "**Cure Costs**", means all amounts that must be paid and all obligations that otherwise must be satisfied, including pursuant to Sections

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 33 of 111

365(b)(1)(A) and (B) of the Bankruptcy Code in connection with the assumption and/or assignment of the Assigned Contracts and Assigned Leases to Purchaser as provided herein.

5.8 <u>Operating Covenant</u>. Purchaser shall act in good faith in fulfilling its obligations under this Agreement.

### 5.9 <u>Governmental Approvals</u>.

(a) <u>5.9.1 Best Efforts.</u> Purchaser (a) shall use its best efforts to secure, as promptly as possible after the Signing Date, all consents, approvals (or exemptions therefrom), authorizations, clearances and licenses required to be obtained from governmental and regulatory authorities in order to carry out the transactions contemplated by this Agreement and to cause all of its covenants and agreements to be performed, satisfied and fulfilled, and (b) will provide such other information and communications to governmental and regulatory authorities as Sellers or such authorities may reasonably request. Purchaser is responsible for all filings with and requests to governmental authorities necessary to enable Purchaser to operate the Hospital at and after the Licensure Date. Purchaser acknowledges that Sellers may independently contact governmental and regulatory authorities as part of this process.

(b) <u>5.9.2 Change of Ownership Applications</u>. Purchaser shall, promptly, but no later than ten (10) business days after the entry of the Sale Order, or sooner if required by applicable governmental or regulatory authorities, file all applications, licensing packages and other documents with all applicable governmental and regulatory authorities which are necessary for the operation of a hospital and the consummation of the transactions hereunder, including the hospital license change of ownership application with CDPH, the hospital pharmacy change of ownership applications.

5.9.3 Attorney General Application. Purchaser and Sellers shall, (c)promptly, but no later than five (5) business days after the date of the Sale Order, file with the California Attorney General (the "Attorney General"), the application and report forms required for the transactions contemplated hereby and any supplemental information that may be reasonably requested in connection therewith pursuant to Title 11, Division 1, Chapter 15, Section 999.5 of the California Code of Regulations ("Section 999.5"), which application and report forms and supplemental information will comply in all material respects with the requirements of such regulations and shall state that Purchaser agrees to close the transactions contemplated by this Agreement so long as any conditions imposed by the Attorney General are substantially consistent with not materially more burdensome than the conditions attached hereto as Exhibit 5.9.35.9(c) (the "Accepted Conditions"). Purchaser shall pay all fees (including, without limitation, attorneys' fees) required of Purchaser or Sellers with respect to the preparation and submission of the application, hearings, expert reports, Attorney General attorney review time, reports and other requirements of the Attorney General under Section 999.5. Each of Purchaser and Sellers shall furnish to the other such information and assistance as the other shall reasonably requires in connection with the preparation and submission to, or agency proceedings by, the Attorney General, and each of Purchaser and Sellers shall keep the other promptly apprised of any communications with, and inquires or requests for information from, the Attorney General. Purchaser shall take such action as may be required by the Attorney

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 34 of 111

General in order to resolve with the minimum practicable delay any objections the Attorney General may have to the transactions contemplated by this Agreement under Section 999.5. For any late submission of the Attorney General application, at Seller's election, either the Purchase Price shall be increased by an amount equal to One Thousand Dollars (\$1,000) per day or Sellers may terminate this Agreement.

5.9.4 HSR Filing. Purchaser and Sellers shall, promptly, but no later than (d)five (5) business days after the date of the Sale Order, file with the Federal Trade Commission and the Department of Justice the notification and report forms required for the transactions contemplated hereby and any supplemental information that may be reasonably requested in connection therewith pursuant to the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended (the "HSR Act"), which notification and report forms and supplemental information will comply in all material respects with the requirements of the HSR Act. Purchaser shall pay all filing fees required of Purchaser or Sellers with respect to the notification, report and other requirements of the HSR Act. Each of Purchaser and Sellers shall furnish to the other such information and assistance as the other shall reasonably requires in connection with the preparation and submission to, or agency proceedings by, any governmental authority under the HSR Act, and each of Purchaser and Sellers shall keep the other promptly apprised of any communications with, and inquires or requests for information from, such governmental Purchaser shall take such action (including divestitures or hold separate authorities. arrangements) as may be required by any governmental authority in order to resolve with the minimum practicable delay any objections such governmental authorities may have to the transactions contemplated by this Agreement under the HSR Act. For any late submission of the application under the HSR Act, at Seller's election, either the Purchase Price shall be increased by an amount equal to One Thousand Dollars (\$1,000) per day or Sellers may terminate this Agreement.

### ARTICLE 6 BANKRUPTCY COURT APPROVAL

### 6.1 Bankruptcy Court Approval.

Sellers and Purchaser acknowledge that this Agreement has been solicited (a) in conformity with the Bidding Procedures approved by the Bankruptcy Court on February 2020 [Docket No. ] (the "Bid Procedures Order"), and that the sale of the Assets and the assignment of the Assigned Contracts and Assigned Leases remain subject to Bankruptcy Court approval. Purchaser further acknowledges that this Agreement and the transactions contemplated hereby are subject to Sellers' right and ability to consider higher or better competing bids with respect to the Assets subject to the terms of the Bid Procedures Order. Pursuant to the Bid Procedures Order, Purchaser shall, if its bid is determined to be the second highest bid, serve as a back-up bidder (the "Back-up Bidder") and, subject to the provisions of the Bid Procedures Order with respect to the retention of the Deposit, keep Purchaser's bid to consummate the transactions contemplated by this Agreement on the terms and conditions set forth in this Agreement (as the same may be modified with the written consent of Purchaser in any auction under the Bid Procedures Order) open and irrevocable until the earlier of (i) (a) 5:00 p.m. pacific time on the date which is thirty (30) days after the entry by the Bankruptcy Court of the Sale Order; or (b) 5:00 p.m. pacific time on May 15,31, 2020 (the "Outside Back-up Date"), or (ii)

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 35 of 111

the date of closing of an alternative transaction with the bidder who prevails at the auction (the "**Prevailing Bidder**"). Following the entry of the Sale Order and prior to the Outside Back-up Date, if the Prevailing Bidder breaches or fails to perform its obligations under the terms and conditions of its respective transaction documents and Sellers terminate such agreement with the Prevailing Bidder, the Back-up Bidder will be deemed to have the new prevailing bid, and the Bankruptcy Court order approving the sale to the Prevailing Bidder shall provide that Sellers will be authorized, without further order of the Bankruptcy Court, to consummate the transactions contemplated by this Agreement on the terms and conditions set forth in this Agreement (as the same may be modified with the written consent of Purchaser in the auction) with the Back-up Bidder.

(b) Purchaser further acknowledges that this Agreement is subject to termination up to the time of entry of the Sale Order, except to the extent otherwise provided in the Bid Procedures Order, in the event Sellers in the exercise of their fiduciary duties determine that they have received a higher and better offer for Sellers' Assets and the Bankruptcy Court has authorized the Sellers to accept and implement the terms of such offer in accordance with the Bankruptcy Code.

(c) Sellers shall, at any hearing to consider approval of this Agreement (the "**Sale Hearing**"), exercise reasonable efforts to obtain a "Sale Order" approving this Agreement subject to the exercise of their fiduciary duties to consider and accept a higher and better offer for Sellers' assets in accordance with the Bankruptcy Code. For purposes of this Agreement, the term "**Sale Order**" shall mean an order of the Bankruptcy Court in form and substance reasonably acceptable to Purchaser, authorizing the sale of the Assets pursuant to Section 363(b) of the Bankruptcy Code (including the Sellers' assumption and assignment to Purchaser of the Assigned Contracts and Assigned Leases pursuant to Section 365 of the Bankruptcy Code) on the terms and conditions set forth herein, free and clear of all Liens (other than Permitted LiensExceptions) and Excluded Liabilities, including, for the avoidance of doubt, any successor liability, to the maximum extent permitted by the Bankruptcy Code.

(d) Sellers agree, subject to the exercise of their fiduciary duties, to seek a Bankruptcy Court determination that Purchaser is a good faith purchaser within the meaning of Section 363(m) of the Bankruptcy Code and in good faith to file such declarations and other evidence as may be required to support a determination.

(e) Sellers shall seek an order from the Bankruptcy Court retaining jurisdiction over all matters relating to claims against Sellers, whether or not arising in connection with this Agreement, solely in the Bankruptcy Court.

6.2 <u>Appeal of Sale Order</u>. In the event an appeal is taken or a stay pending appeal is requested from the Sale Order, Sellers shall immediately notify Purchaser of such appeal or stay request and shall provide to Purchaser promptly a copy of the related notice of appeal or order of stay. Sellers shall also provide Purchaser with written notice of any motion or application filed in connection with any appeal from either of such orders. In the event of an appeal of the Sale Order, Sellers shall be primarily responsible for drafting pleadings and attending hearings as necessary to defend against the appeal.

6.3 <u>Bidding Procedures</u>. Sellers and Purchaser shall comply with the terms of the Bid Procedures Order. Sellers shall sign this Agreement as and when permitted pursuant to the Bid Procedures Order and the Sale Order. Notwithstanding anything to the contrary herein or in the Bid Procedures Order, the ability of the parties to alter the Sale Order or this Agreement from the forms thereof approved by Purchaser shall terminate upon approval of this Agreement as the Winning Bid or Back Up Bid, absent the express written consent of Purchaser. In the event of any material alteration of the terms of this Agreement or the Sale Order by any party, which determination shall be made by Purchaser in its reasonable discretion, Purchaser shall have no obligation to proceed with the transaction.

## ARTICLE 7 CONDITIONS PRECEDENT TO OBLIGATIONS OF SELLERS

Sellers' obligation to sell the Assets and to close the transactions as contemplated by this Agreement shall be subject to the satisfaction of each of the following conditions on or prior to the Closing Date unless specifically waived in writing by Sellers in whole or in part at or prior to the Closing:

7.1 <u>Signing and Delivery of Instruments</u>. Purchaser shall have executed and delivered all documents, instruments and certificates required to be executed and delivered pursuant to the provisions of this Agreement.

7.2 <u>No Restraints</u>. No temporary restraining order, preliminary or permanent injunction or other order preventing the consummation of the transactions contemplated in this Agreement shall have been issued by any court of competent jurisdiction or any other governmental body and remain in effect on the Closing Date.

7.3 <u>Governmental Submissions</u>. Purchaser shall have submitted the hospital license change of ownership application to CDPH and the hospital pharmacy change of ownership application to the BOP.

7.4 <u>Attorney General Approval</u>. The Attorney General shall have approved the transactions contemplated by this Agreement, or the Debtors shall have obtained a Bankruptcy Court order or orders providing for the transfer of the Debtors' assets on conditions no less favorable which are not materially more burdensome than those forth in Exhibit 5.9.3.5.9(c).

7.5 <u>Bankruptcy Court Approval</u>. The Bankruptcy Court shall have entered the Sale Order.

7.6 <u>HSR Act</u>. The applicable waiting period under the HSR Act shall have expired or been earlier terminated.

# ARTICLE 8 CONDITIONS PRECEDENT TO OBLIGATIONS OF PURCHASER

Purchaser's obligation to purchase the Assets and to close the transactions contemplated by this Agreement shall be subject to the satisfaction of each of the following conditions on or

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 37 of 111

prior to the Closing Date unless specifically waived in writing by Purchaser in whole or in part at or prior to the Closing.

8.1 <u>Signing and Delivery of Instruments</u>. Sellers shall have executed and delivered all documents, instruments and certificates required to be executed and delivered pursuant to the provisions of this Agreement.

8.2 <u>No Restraints</u>. No temporary restraining order, preliminary or permanent injunction or other order preventing the consummation of the transactions contemplated in this Agreement shall have been issued by any court of competent jurisdiction or any other governmental body and remain in effect on the Closing Date.

8.3 Attorney General Approval. The Attorney General shall have approved the transactions contemplated by this Agreement, or the Debtors shall have obtained a Bankruptcy Court order or orders providing for the transfer of the Debtors' assets on conditions no lessfavorable which are not materially more burdensome than those forth in Exhibit 5.9.3.5.9(c). In the event the Attorney General imposes conditions on the transactions contemplated by this Agreement, or on Purchaser in connection therewith which are materially differentmore burdensome than the Purchaser Approved Conditions set forth on Exhibit 5.9.35.9(c) (the "Additional Conditions"), Sellers shall have the opportunity to file a motion with the Bankruptcy Court seeking the entry of an order ("Supplemental Sale Order") finding that the Additional Conditions are an "interest in property" for purposes of 11 U.S.C. § 363(f), and that the Assets can be sold free and clear of the Additional Conditions without the imposition of any other conditions, which would adversely affect the Purchaser. For purposes of this Section 8.3, Additional Conditions which individually or collectively impose a direct or indirect cost to Purchaser of Five Million Dollars (\$5,000,000), or more, shall be conclusively deemed to be "materially differentmore burdensome." If Sellers determine not to seek such Supplemental Sale Order, or fail to obtain such Supplemental Sale Order within sixty (60) days of the Attorney General's imposition of Additional Conditions, Purchaser shall be entitled to terminate this Agreement and receive the return of its Deposit. Upon the entry of such an order from the Bankruptcy Court, and so long as such order remains-final and in full force and effect, Purchaser shall be required to consummate the transactions contemplated by this Agreement. If Sellers do not obtain such an order, or such order does not relieve the Additional Conditions to the satisfaction of Purchaser or such order does not remain in full force and effect, Purchaser shall not be required to consummate the transactions contemplated by this Agreement.

8.4 <u>Bankruptcy Court Approval</u>. The Bankruptcy Court shall have entered the Sale Order and made a finding that Purchaser is a "good faith" purchaser, and the Sale Order shall become final unless such condition has been waived by Purchaser.

8.5 <u>HSR Act</u>. The applicable waiting period under the HSR Act shall have expired or been earlier terminated.

8.6 <u>Medicare and Medi-Cal Provider Agreement</u>. Following consultation with Purchaser, Sellers shall have obtained agreements with CMS and DHCS or a final-court order by a court having proper jurisdiction shall have been entered with respect to the transfer of the Medicare<u>/Provider Agreement and/or the Medi-Cal AgreementsProvider Agreement</u>, such that

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 38 of 111

(a) all liabilities and obligations under the Medicare/Medi-Cal Agreements are fully satisfied, discharged, and released with regard to any claims under the Medicare/Medi-Cal Agreements, whether known or unknown, that CMS or DHCS has against the Sellers or Purchaser for liabilities and obligations arising under the Medicare/Medi-Cal Agreements before the Effective Time, and (b) such the Medicare/Medi-Cal Agreements will be transferred to Purchaser as of the Effective Time free and clear of such pre-Closing liabilities and obligations; provided, however, that Purchaser acknowledges and agrees that it may be treated by CMS and DHCS as the successor to the quality history associated with the relevant Medicare/Medi-Cal Agreements assigned and, for purposes of survey and certification issues associated with such quality history, Purchaser may be treated as if it is the relevant Seller and no change of ownership occurred.

8.7 Collective Bargaining Agreements and Labor Contracts. Purchaser agrees to close the transactions contemplated by this Agreement, so long as (a) after good faith negotiations with the applicable St. Francis labor unions, such labor unions agree to either (i) modify the St. Francis collective bargaining agreements to be substantially consistent with the Purchaser's existing new collective bargaining agreements with each such respective labor union, and settle all liabilities under the existing agreements, or (ii) enter into new collective bargaining agreements that are substantially consistent with the Purchaser's existing collective bargainingagreements with each such respective labor union, and settle all liabilities under the existing agreements, or (b) the Bankruptcy Court takes action to reject any such collective bargaining agreement that is not substantially consistent or where the union has not agreed to settle all liabilities under the agreement. In the event Purchaser and the respective St. Francis labor unionare unable to agree to a modified or new collective bargaining agreement that is substantially consistent with the Purchaser's existing collective bargaining agreement with the respective laborunion, as well as a settlement of all liabilities under the existing agreements, and the Bankruptcy-Court does not reject such collective bargaining agreement pursuant to a final order, then Purchaser shall not be required to consummate the transactions contemplated by this Agreement.

# ARTICLE 9 TERMINATION

- 9.1 <u>Termination</u>. This Agreement may be terminated at any time prior to Closing:
  - (a) by the mutual written consent of the <u>parties</u><u>Parties</u>;

(b) by Sellers if a material breach of this Agreement has been committed by Purchaser and such breach has not been (i) waived in writing by Sellers or (ii) cured by Purchaser to the reasonable satisfaction of Sellers within fifteen (15) business days after Sellers provide Purchaser of a written notice which describes the nature of such breach and acceptable cure actions; *provided*, *however*, Sellers shall not be permitted to terminate this Agreement pursuant to this Section 9.1(b) if Sellers are also in material breach of this Agreement;

(c) by Purchaser if Purchaser has complied with <u>Section 3.8</u> and a material breach of this Agreement has been committed by Sellers, which material breach has resulted, or would more likely than not result, in a Material Adverse Effect on the Assets taken as a whole, and such breach has not been (i) waived in writing by Purchaser or (ii) cured by Sellers to the reasonable satisfaction of Purchaser within fifteen (15) business days after Purchaser provides

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 39 of 111

Sellers of a written notice which describes the nature of such breach and acceptable cure actions; *provided, however*, Purchaser shall not be permitted to terminate this Agreement pursuant to this Section 9.1(c) if Purchaser is also in material breach of this Agreement;

(d) by Purchaser if satisfaction of any condition in <u>ARTICLE 8</u> is or becomes impossible and Purchaser has not waived such condition in writing (provided that the failure to satisfy the applicable condition or conditions has occurred by reason other than (i) through the failure of Purchaser to comply with its obligations under this Agreement or (ii) Sellers' failure to provide their closing deliveries on the Closing Date as a result of Purchaser not being ready, willing and able to close the transaction on the Closing Date);

(e) by Sellers if satisfaction of any such condition in <u>ARTICLE 7</u> is or becomes impossible and Sellers have not waived such condition in writing (provided that the failure to satisfy the applicable condition or conditions has occurred by reason other than (i) through the failure of Sellers to comply with their obligations under this Agreement or (ii) Purchaser's failure to provide its closing deliveries on the Closing Date as a result of Sellers not being ready, willing and able to close the transaction on the Closing Date);

(f) by either Purchaser or Sellers if the Bankruptcy Court enters an order dismissing the Bankruptcy Case prior to the sale closing or fails to approve the sale of the Assets to Purchaser;

(g) by Sellers in the event Sellers in the exercise of their fiduciary duties determine that they have received a higher and better offer for Sellers' assets and the Bankruptcy Court has authorized the Sellers to accept and implement the terms of such offer in accordance with the Bankruptcy Code;

(h) by Purchaser in the event that its Back-Up Bidder status has expired in accordance with the Bid Procedures Order and the terms of this Agreement; and

(i) by either Purchaser or Sellers if the Closing has not occurred (other than through the failure of any <u>partyParty</u> seeking to terminate this Agreement to comply fully with its obligations under this Agreement) on or before <u>December 31,September 1</u>, 2020 (the "**Termination Date**")<del>,</del>; provided, that the Termination Date shall be extended in the event the conditions set forth in Section 7.4 and Section 8.3 have not been satisfied (but the conditions to Closing in Article 7 and Article 8 have otherwise been satisfied other than such conditions that are to be satisfied by payments and deliveries to be made at the Closing) to the earlier of (A) ten (10) business days after the satisfaction of the conditions set forth in Section 7.4 and Section 8.3 or (B) December 31, 2020.

9.2 <u>Termination Consequences</u>. If this Agreement is terminated pursuant to <u>Section</u> <u>9.1</u>: (a) all further obligations of the <u>partiesParties</u> under this Agreement shall terminate, other than Purchaser's right to receive a return on the Deposit in accordance with <u>Section 1.2</u> and a break-up fee in accordance with the Bidding Procedures, and provided that the provisions of <u>ARTICLE 11</u> shall survive, and (b) each <u>partyParty</u> shall pay the costs and expenses incurred by it in connection with this Agreement; *provided, however*, that in the case of any termination based on <u>Section 9.1(b)</u> or <u>Section 9.1(c)</u>, the consequences of such termination shall be

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 40 of 111

determined in accordance with <u>ARTICLE 11</u> hereof. Each <u>partyParty</u> acknowledges that the agreements contained in this <u>Section 9.2</u> are an integral part of the transactions contemplated by this Agreement, that without these agreements such <u>partyParty</u> would not have entered into this Agreement.

## ARTICLE 10 POST-CLOSING MATTERS

Excluded Assets. Subject to Section 10.2 hereof, any Excluded Asset (or 10.1 proceeds thereof) (a) pursuant to the terms of this Agreement, (b) as otherwise determined by the partiesParties' mutual written agreement or (c) absent such agreement, as determined by adjudication by the Bankruptcy Court, which comes into the possession, custody or control of Purchaser (or its respective successors-in-interest, assigns or affiliates) shall, within sixtyten (6010) business days following receipt, be transferred, assigned or conveyed by Purchaser (and its respective successors-in-interest, assigns and affiliates) to Sellers without imposing any charge to Sellers for Purchaser's transfer, storage, handling or holding of same on and after the Effective Time. Purchaser (and its respective successors-in-interest, assigns and affiliates) shall have neither the right to offset amounts payable to Sellers under this Section 10.1 against, nor the right to contest its obligation to transfer, assign and convey to Sellers because of, outstanding claims, liabilities or obligations asserted by Purchaser against Sellers. If Purchaser does not remit any monies included in the Excluded Assets (or proceeds thereof) to Sellers in accordance with the first sentence of this Section 10.1, such withheld funds shall bear interest at the Prime Rate in effect on the calendar day upon which such payment was required to be made to Sellers (the "Excluded Asset Due Date") plus five percent (5%) (or the maximum rate allowed by law, whichever is less), such interest accruing on each calendar day after the Excluded Asset Due Date until payment of the Excluded Assets and all interest thereon is made to Sellers.

### 10.2 <u>Preservation and Access to Records After the Closing</u>.

(a) From the Licensure Date until seven (7) years after the Licensure Date or such longer period as required by law (the "Document Retention Period"), Purchaser shall keep and preserve all medical records, patient records, medical staff records and other books and records which are among the Assets as of the Effective Time, but excluding any records which are among the Excluded Assets. Purchaser will afford to the representatives of Sellers, any of their affiliates, the Official Committee of the Unsecured Creditors of the Sellers, Sellers' estate representative or any liquidating trustee of the Sellers' bankruptcy estate ("Seller Parties"), including their counsel and accountants, full and complete access to, and copies (including, without limitation, color laser copies) of, such records with respect to time periods prior to the Licensure Date (including, without limitation, access to records of patients treated at the Hospital prior to the Licensure Date) during normal business hours after the Licensure Date, to the extent reasonably needed by any Seller Party for any lawful purpose. Purchaser acknowledges that, as a result of entering into this Agreement and operating the Hospital, it will gain access to patient records and other information which are subject to rules and regulations concerning Purchaser shall abide by any such rules and regulations relating to the confidentiality. confidential information it acquires. Purchaser shall maintain the patient and medical staff records at the Hospital in accordance with applicable law and the requirements of relevant insurance carriers. After the expiration of the Document Retention Period, if Purchaser intends

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 41 of 111

to destroy or otherwise dispose of any of the documents described in this <u>Section 10.2(a)</u>, Purchaser shall provide written notice to Sellers of Purchaser's intention no later than forty-five (45) calendar days prior to the date of such intended destruction or disposal. Any of the Seller Parties shall have the right, at its sole cost, to take possession of such documents during such forty-five (45) calendar day period. If any of the Seller Parties <u>doesdo</u> not take possession of such documents during such forty-five (45) calendar day period, Purchaser shall be free to destroy or otherwise dispose of such documentation upon the expiration of such forty-five (45) calendar day period.

(b) Provided that Purchaser shall not incur any out of pocket costs, Purchaser shall give full cooperation to the Seller Parties and their insurance carriers in connection with the administration of Sellers' estate, including, without limitation, in connection with all claims, actions, causes of action or audits relating to the Excluded Assets, Excluded Liabilities or pre-Closing operation of the Sellers or the Hospital that any Seller Party may elect to pursue, dispute or defend, in respect of events occurring prior to the Licensure Date with respect to the operation of the Hospital. Such cooperation shall include, without limitation, making the Hired Employees available for interviews, depositions, hearings and trials and other assistance in connection with the administration of Sellers' estate and such cooperation shall also include making all of its employees available to assist in the securing and giving of evidence and in obtaining the presence and cooperation of witnesses (all of which shall be done without payment of any fees or expenses to Purchaser or to such employees. In addition, Sellers and Sellers' affiliates shall be entitled to remove from the Hospital originals of any such records, but only for purposes of pending litigation involving the persons to whom such records refer, as certified in writing prior to removal by counsel retained by Sellers or any of Sellers' affiliates in connection with such litigation. Any records so removed from the Hospital shall be promptly returned to Purchaser following Sellers' or its applicable affiliate's use of such records.

(c) In connection with (i) the transition of the Hospital pursuant to the transaction contemplated by this Agreement, (ii) Sellers' rights to the Excluded Assets, (iii) any claim, audit, or proceeding, including, without limitation, any tax claim, audit, or proceeding and (iv) the Sellers' obligations under the Excluded Liabilities, Purchaser shall after the Effective Time give Sellers access during normal business hours to Purchaser's books, personnel, accounts and records and all other relevant documents and information with respect to the assets, liabilities and business of the Hospital as representatives of Sellers and Sellers' affiliates may from time to time reasonably request, all in such manner as not to unreasonably interfere with the operations of the Hospital.

(d) Purchaser and its representatives shall be given access by Sellers during normal business hours to the extent reasonably needed by Purchaser for business purposes to all documents, records, correspondence, work papers and other documents retained by Sellers pertaining to any of the Assets prior to the Effective Time (excluding confidential employee information, privileged materials and patient records), all in such manner as to not interfere unreasonably with Sellers. Such documents and other materials shall be, at Sellers' option, either (i) copied by Sellers for Purchaser at Purchaser's expense, or (ii) removed by Purchaser from the premises, copied by Purchaser and promptly returned to Sellers.

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 42 of 111

(e) Purchaser shall comply with, and be solely responsible for, all obligations under the Standards for Privacy of Individually Identifiable Health Information (45 CFR Parts 160 and 164) promulgated pursuant to the Health Insurance Portability and Accountability Act of 1996 and any and all corresponding California state law requirements with respect to the operation of the Hospital on and after the Licensure Date.

(f) Purchaser shall cooperate with Sellers, on a timely basis and as reasonably requested by Sellers, in connection with the provision of all data of the Hospital and other information required by Sellers for reporting purposes, including without limitation QAF or DSH reporting.

(g) To the maximum extent permitted by law, if any Person requests or demands, by subpoena or otherwise, any documents relating to the Excluded Liabilities or Excluded Assets, including without limitation, documents relating to the operations of any of the Hospital or any of the Hospital's committees prior to the Licensure Date, prior to any disclosure of such documents, Purchaser shall notify Sellers and shall provide Sellers with the opportunity to object to, and otherwise coordinate with respect to, such request or demand.

10.3 Provision of Benefits of Certain Contracts. Notwithstanding anything contained herein to the contrary, this Agreement shall not constitute an agreement to assign any Assigned Contract or Assigned Lease, if, notwithstanding the provisions of Sections 363 and 365 of the Bankruptcy Code, an attempted assignment thereof, without the consent of the third party thereto, would constitute a breach thereof or in any way negatively affect the rights of Sellers or Purchaser, as the assignee of such Assigned Contract or Assigned Lease, as the case may be, thereunder. If, notwithstanding the provisions of Sections 363 and 365 of the Bankruptcy Code, such consent or approval is required but not obtained, Sellers will cooperate with Purchaser in any reasonable arrangement designed to both (a) provide Purchaser with the benefits of or under any such Assigned Contract or Assigned Lease. Further, notwithstanding anything contained in this Agreement to the contrary, this Agreement shall not constitute an agreement to assign any Account Receivable the assignment of which is either prohibited by law or by the terms of any contract with a payor without the consent of such payor.

General Cooperation and Turnover Obligations. 10.4 The parties Parties shall cooperate to ensure that any and all payments that constitute "Excluded Assets" shall be paid to and received by Sellers, with any payments that constitute "Assets" transferred to Purchaser pursuant to Section 1.7 or that otherwise arise from services rendered by Purchaser on or after the Licensure Date be paid to and received by Purchaser. In this regard, for a period of  $\frac{12}{12}$ yearyears after the Effective Date ("Turnover Period"), the parties Parties shall, within fiveten (510) business days of receipt, copy and send to the other party Party copies (either in hard copy or via electronic file) of all remittance advices for all deposits to all Lockboxes or other bank accounts for the Accounts Receivables, from whatever payor or source of funds, that are received on and after the Effective Time. In the event that payments that constitute a transferred Asset are deposited to a bank account of Sellers which is not automatically swept or transferred to Purchaser, then Sellers, within fiveten (510) days of notice of the receipt of such payments shall turnover and pay Purchaser said funds. In the event that a deposit representing payment of any Excluded Assets is received by Purchaser, then Purchaser, within  $\frac{\text{fiveten}}{10}$  days of notice of

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 43 of 111

the receipt of funds representing any Excluded Assets, shall turnover and pay Sellers such funds. Each <u>partyParty</u> shall have the right, within three (3) months after the expiration of the Turnover Period, to audit by an independent and competent auditor, at the requesting <u>partyParty</u>'s sole expense, of the bank records and remittance advices of the other <u>partyParty</u>. Thereafter, upon the findings of the auditor that there has either been an overpayment or and underpayment of funds due, the <u>partyParty</u> owning funds shall, within <u>fiveten</u> (510) business days, make a payment of such funds to whom they are owed.

10.5 <u>Closing of Financials</u>. Provided that Purchaser shall not incur any out of pocket costs, Purchaser shall cause the individual acting as the chief financial officer of the Hospital after the Effective Time (the "**Post-Effective Time CFO**") to cooperate with Sellers' representatives in order to complete the standardized closing of Sellers' financial records through the Licensure Date including, without limitation, the closing of general ledger account reconciliations (collectively, the "**Closing of Financials**"). Purchaser shall cause the Post-Effective Time CFO to use their good faith efforts to cooperate with Sellers' representatives in order to complete the Closing of Financials by no later than the date which is sixty (60) calendar days after the Closing Date. The Post-Effective Time CFO and other appropriate personnel shall be reasonably available to Sellers for a period of no less than one hundred eighty (180) calendar days after the Closing Date to assist Sellers in the completion of Sellers' post-Closing audit, such assistance not to interfere unreasonably with such Post-Effective Time CFO's other duties.

10.6 <u>Medical Staff</u>. To ensure continuity of care in the community, Purchaser agrees that the Hospital's medical staff members in good standing as of the Licensure Date shall maintain medical staff privileges at the Hospital as of the Licensure Date. On and after the Licensure Date, the medical staff will be subject to the Hospital's Medical Staff Bylaws then currently in effect, provided that such Bylaws are in compliance with all applicable laws and regulations and contain customary obligations.

10.7 <u>Shared Intangible Assets</u>. In the event and to the extent that certain intangible Assets transferred by Sellers have been used to operate businesses of Verity or Verity Holdings or their affiliates ("**Shared Intangible Assets**") and such Shared Intangible Assets continue to be used by Verity or Verity Holdings or their affiliates to operate such businesses after Closing, Verity and Verity Holdings retain the rights to continue to use such Assets notwithstanding their sale to Purchaser. Purchaser shall reasonably cooperate with Verity and Verity Holdings and their affiliates to give effect to such rights and shall provide Verity and Verity Holdings and their affiliates such documentation, records and information and reasonable access to such systems as necessary for Verity and Verity Holdings and their affiliates to continue to reasonably interfere with the operations of the Hospitals; *provided, however*, Purchaser shall not be required to incur any out-of-pocket costs in association therewith unless reimbursed by Verity and Verity Holdings and their affiliates.

## ARTICLE 11 DEFAULT, TAXES AND COST REPORTS

11.1 <u>Purchaser Default</u>. If the Agreement is terminated pursuant to <u>Section 9.1(b)</u>, Sellers shall be entitled to retain the Deposit, and Sellers may, in addition thereto, pursue any

### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 44 of 111

rights or remedies that Sellers may have under this Agreement or applicable law, including the right to sue for damages or specific performance.

11.2 <u>Sellers Default</u>. If the Agreement is terminated pursuant to <u>Section 9.1(c)</u>, Purchaser shall have the right to demand and receive a refund of the Deposit, and Purchaser may, in addition thereto, pursue any rights or remedies that Purchaser may have under applicable law, including the right to sue for damages or specific performance.

### 11.3 <u>Tax Matters; Allocation of Purchase Price</u>.

(a) After the Licensure Date, the <u>partiesParties</u> shall cooperate fully with each other and shall make available to each other, as reasonably requested, all information, records or documents relating to tax liabilities or potential tax liabilities attributable to Sellers with respect to the operation of the Hospital for all periods prior to the Licensure Date and shall preserve all such information, records and documents at least until the expiration of any applicable statute of limitations or extensions thereof. The <u>partiesParties</u> shall also make available to each other to the extent reasonably required, and at the reasonable cost of the requesting <u>partyParty</u> (for out-of-pocket costs and documents in connection with tax matters and as Sellers reasonably may request in connection with the completion of any post-Closing audits of the Hospital.

(b) Solely for purposes of tax reporting, <u>Schedule 11.3</u> sets forth the allocation of the Purchase Price (including any liabilities that are considered to be an increase to the Purchase Price for United States federal income <u>Taxtax</u> purposes) among the Assets in accordance with Section 1060 of the Code and the Treasury Regulations promulgated thereunder (the "Allocation Schedule"). The Allocation Schedule shall be final and binding upon Sellers and Purchaser with respect to matters relating to required tax reporting by each such <u>partyParty</u>. The <u>partiesParties</u> shall refrain from taking any position that is inconsistent with the Allocation Schedule with respect to tax reporting.

# 11.4 Cost Report Matters.

(a) Consistent with <u>Section 4.4</u>, Sellers shall, at Purchaser's expense, prepare and timely file all cost reports relating to the periods ending prior to the Licensure Date or required as a result of the consummation of the transactions described in this Agreement, including, without limitation, those relating to Medicare, Medicaid, and other third party payors which settle on a cost report basis (the "Seller Cost Reports").

(b) Upon reasonable notice and during normal business office hours, Purchaser will cooperate reasonably with Sellers in regard to Sellers' preparation and filing of the Seller Cost Reports. Such cooperation shall include, at no cost to Sellers, obtaining access to files at the Hospital and Purchaser's provision to Sellers of data and statistics, and the coordination with Sellers pursuant to reasonable notice of Medicare and Medicaid exit conferences or meetings. Sellers shall have no obligations after the Licensure Date with respect to Seller Cost Reports except for preparation and filing thereof.

### ARTICLE 12 MISCELLANEOUS PROVISIONS

12.1 <u>Further Assurances and Cooperation</u>. Each <u>partyParty</u> shall execute, acknowledge and deliver to the other <u>partyParty</u> any and all other assignments, consents, approvals, conveyances, assurances, documents and instruments reasonably requested by such <u>partyPartyParty</u> at any time and shall take any and all other actions reasonably requested by such <u>partyPartyParty</u> at any time for the purpose of consummating the transactions hereunder and fulfilling such <u>partyPartyParty</u>'s obligations hereunder. After consummation of the transactions contemplated in this Agreement, the <u>partiesParties</u> agree to cooperate with each other and take such further actions as may be necessary or appropriate to effectuate, carry out and comply with all of the terms of this Agreement, the documents referred to in this Agreement and the transactions contemplated hereby.

12.2 <u>Successors and Assigns</u>. All of the terms and provisions of this Agreement shall be binding upon and shall inure to the benefit of and be enforceable by the respective successors and assigns of the <u>partiesParties</u> hereto; *provided*, *however*, that no <u>partyParty</u> hereto may assign any of its rights or delegate any of its duties under this Agreement without the prior written consent of the other <u>partiesParties</u> which consent shall not be unreasonably withheld or delayed, except that Purchaser may, without the prior written consent of Sellers, assign all or any portion of its rights under this Agreement to one or more of its affiliates prior to the Closing Date so long as such affiliate was described and was part of the Attorney General application under Section 5.9(c) and provided that any such assignment shall not relieve Purchaser of or reduce Purchaser's obligations under this Agreement.

12.3 <u>Governing Law; Venue</u>. This Agreement shall be construed, performed, and enforced in accordance with, and governed by, the laws of the State of California (without giving effect to the principles of conflicts of laws thereof), except to the extent that the laws of such State are superseded by the Bankruptcy Code or other applicable federal law. For so long as Sellers are subject to the jurisdiction of the Bankruptcy Court, the <u>partiesParties</u> irrevocably elect, as the sole judicial forum for the adjudication of any matters arising under or in connection with the Agreement, and consent to the exclusive jurisdiction of, the Bankruptcy Court. The <u>partiesParties</u> hereby consent to the jurisdiction of such court and waive their right to challenge any proceeding involving or relating to this Agreement on the basis of lack of jurisdiction over the Person or forum non conveniens.

12.4 <u>Amendments</u>. This Agreement may not be amended other than by written instrument signed by the <u>parties heretoParties</u>.

12.5 <u>Exhibits, Schedules and Disclosure Schedule</u>. The Disclosure Schedule and all exhibits and schedules referred to in this Agreement shall be attached hereto and are incorporated by reference herein. From the Signing Date until the Closing, the <u>partiesParties</u> agree that Sellers may update the Disclosure Schedule as necessary upon written notice to Purchaser, and the applicable representation and warranty shall thereafter be deemed amended for all purposes by such updated Disclosure Schedule. Notwithstanding the foregoing, should any exhibit or schedule not be completed and attached hereto as of the Signing Date, Sellers and Purchaser shall promptly negotiate in good faith any such exhibit or schedule, which exhibit or schedule must be

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 46 of 111

acceptable to each of Sellers and Purchaser in their reasonable discretion prior to being attached hereto. Any matter disclosed in this Agreement or in the Disclosure Schedule with reference to any Section of this Agreement shall be deemed a disclosure in respect of all sections to which such disclosure may apply. The headings, if any, of the individual sections of the Disclosure Schedule are provided for convenience only and are not intended to affect the construction or interpretation of this Agreement. The Disclosure Schedule is arranged in sections and paragraphs corresponding to the numbered and lettered sections and paragraphs of this Agreement merely for convenience, and the disclosure of an item in one section of the Disclosure Schedule as an exception to a particular obligation, representation or warranty shall be deemed adequately disclosed as an exception with respect to all other obligations, representations or warranties, notwithstanding the presence or absence of an appropriate section of the Disclosure Schedule with respect to such other obligation, representations or warranties or an appropriate cross reference thereto.

12.6 <u>Notices</u>. Any notice, demand, letter or other communication required, permitted, or desired to be given hereunder shall be deemed effectively given when either personally delivered, or when received by telegraphic or other electronic means (including facsimile) or overnight courier, or five (5) calendar days after being deposited in the United States mail, with postage prepaid thereon, certified or registered mail, return receipt requested, addressed as follows:

If to Sellers:	Verity Health System of California, Inc. 601 South Figueroa St., Suite 4050 Los Angeles, CA 90017-5704 Attention: Chief Executive Officer
With a copies to: (which copies shall not constitute notice)	Dentons US LLP 601 South Figueroa St., Suite 2500 Los Angeles, CA 90017-5704 Attention: Tania Moyron, Esq. Telephone: 213-243-6101
If to Purchaser:	Prime Healthcare Services, Inc. 3480 East Guasti Road, 2 <sup>nd</sup> Floor Ontario, California 91761 Attention: General Counsel Facsimile: 909-235-4316
With a copy to: (which copies shall not constitute notice)	McDermott Will & Emery LLP 2049 Century Park East, Suite 3200 Los Angeles, California 90067 Attention: Jeffrey Reisner, Esq. Facsimile: 310-277-4730

or at such other address as one <u>partyParty</u> may designate by notice hereunder to the other <u>partiesParties</u>.

12.7 <u>Headings</u>. The section and other headings contained in this Agreement and in the Disclosure Schedule, exhibits and schedules to this Agreement are included for the purpose of convenient reference only and shall not restrict, amplify, modify or otherwise affect in any way the meaning or interpretation of this Agreement or the Disclosure Schedule, exhibits and schedules hereto.

12.8 <u>Publicity</u>. Prior to the Closing Date, Sellers and Purchaser shall consult with each other as to the form and substance of any press release or other public disclosure materially related to this Agreement or any other transaction contemplated hereby and each shall have the right to review and comment on the other's press releases at least forty-eight (48) hours prior to issuance; *provided, however*, that nothing in this <u>Section 12.8</u> shall be deemed to prohibit either Sellers or Purchaser from making any disclosure that its counsel deems necessary or advisable in order to satisfy either <u>partyParty</u>'s disclosure obligations imposed by law subject to reasonable prior notice to the other <u>partyParty</u> thereof.

12.9 <u>Fair Meaning</u>. This Agreement shall be construed according to its fair meaning and as if prepared by all <u>parties heretoParties</u>.

12.10 Gender and Number; Construction; Affiliates. All references to the neuter gender shall include the feminine or masculine gender and vice versa, where applicable, and all references to the singular shall include the plural and vice versa, where applicable. Unless otherwise expressly provided, the word "including" followed by a listing does not limit the preceding words or terms and shall mean "including, without limitation." Any reference in this Agreement to an "affiliate" shall mean any Person directly or indirectly controlling, controlled by or under common control with a second Person. The term "control" (including the terms "controlled by" and "under common control with") means the possession, direct or indirect, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by contract or otherwise. A "Person" shall mean any natural person, partnership, corporation, limited liability company, association, trust or other legal entity.

12.11 <u>Third Party Beneficiary</u>. None of the provisions contained in this Agreement are intended by the <u>partiesParties</u>, nor shall they be deemed, to confer any benefit on any person not a party to this Agreement, except for the <u>partiesParties</u>' successors and permitted assigns, and except for any liquidating trustee or plan administrator for Sellers' estate.

12.12 Expenses and Attorneys' Fees. Except as otherwise provided in this Agreement, each partyParty shall bear and pay its own costs and expenses relating to the preparation of this Agreement and to the transactions contemplated by, or the performance of or compliance with any condition or covenant set forth in, this Agreement, including without limitation, the disbursements and fees of their respective attorneys, accountants, advisors, agents and other representatives, incidental to the preparation and carrying out of this Agreement, whether or not the transactions contemplated hereby are consummated. The partiesParties expressly agree that all sales, transfer, documentary transfer and similar taxes, fees, surcharges and the like in connection with the sale of the Assets shall be borne by Purchaser. If any action is brought by any partyParty to enforce any provision of this Agreement, the prevailing partyParty shall be entitled to recover its court costs and reasonable attorneys' fees.

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 48 of 111

12.13 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same Agreement, binding on all of the <u>partiesParties</u> hereto. The <u>partiesParties</u> agree that facsimile copies of signatures shall be deemed originals for all purposes hereof and that a <u>partyParty</u> may produce such copies, without the need to produce original signatures, to prove the existence of this Agreement in any proceeding brought hereunder.

12.14 <u>Entire Agreement</u>. This Agreement, the Disclosure Schedule, the exhibits and schedules, and the documents referred to in this Agreement contain the entire understanding between the <u>partiesParties</u> with respect to the transactions contemplated hereby and supersede all prior or contemporaneous agreements, understandings, representations and statements, oral or written, between the <u>partiesParties</u> on the subject matter hereof (the "**Superseded Agreements**"), which Superseded Agreements shall be of no further force or effect.

12.15 <u>No Waiver</u>. Any term, covenant or condition of this Agreement may be waived at any time by the <u>partyParty</u> which is entitled to the benefit thereof but only by a written notice signed by the <u>partyParty</u> expressly waiving such term or condition. The subsequent acceptance of performance hereunder by a <u>partyParty</u> shall not be deemed to be a waiver of any preceding breach by any other <u>partyParty</u> of any term, covenant or condition of this Agreement, other than the failure of such other <u>partyParty</u> to perform the particular duties so accepted, regardless of the accepting <u>partyParty</u>'s knowledge of such preceding breach at the time of acceptance of such performance. The waiver of any term, covenant or condition shall not be construed as a waiver of any other term, covenant or condition of this Agreement.

12.16 <u>Severability</u>. If any term, provision, condition or covenant of this Agreement or the application thereof to any <u>partyParty</u> or circumstance shall be held to be invalid or unenforceable to any extent in any jurisdiction, then the remainder of this Agreement and the application of such term, provision, condition or covenant in any other jurisdiction or to persons or circumstances other than those as to whom or which it is held to be invalid or unenforceable, shall not be affected thereby, and each term, provision, condition and covenant of this Agreement shall be valid and enforceable to the fullest extent permitted by law.

12.17 <u>Time is of the Essence</u>. Time is of the essence for all dates and time periods set forth in this Agreement and each performance called for in this Agreement.

12.18 WAIVER OF JURY TRIAL. TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT, OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (B) EACH PARTY Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 49 of 111

UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (C) EACH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (D) EACH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 12.18. THIS IS AN AFFIRMATIVE WAIVER OF THE PARTIES' RIGHTS TO A JURY TRIAL UNDER CALIFORNIA LAW, Cal. C. Civ. Pro. Sec 631. BY SIGNING BELOW ON THE SIGNATURE LINES, EACH PARTY IS EXPLICITLY WAIVING JURY TRIAL AND AUTHORIZING ANY AND ALL PARTIES TO FILE THIS WAIVER WITH ANY COURT AS THE WAIVER REQUIRED UNDER Cal. C. Civ. Proc. Sec. 631(f)(2).

[REMAINDER OF PAGE IS BLANK]

# Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 50 of 111

**IN WITNESS WHEREOF,** this Agreement has been entered into as of the day and year first above written.

# **PURCHASER:**

# Prime Healthcare Services, Inc.

Signature:	
Print Name:_	
Title:	
Date:	

## **SELLERS:**

# Verity Health System of California, Inc.

Signature:	
Print Name:	
Title:	
Date:	

## Verity Holdings, LLC

Signature:	
Print Name:_	
Title:	
Date:	

# **St. Francis Medical Center**

Signature:	
Print Name:_	
Title:	
Date:	

Description	Owner	Address	APN
Hospital Campus (including Health Services Pavilion, Progressive Care Unit, Family Life Center, Power Plant and Parking Structure #1)	St. Francis Medical Center	3630 East Imperial Highway, Lynwood, CA 90262	6173-021-008
Huntington Park MOB	St. Francis Medical Center	2700 East Slauson Avenue, Huntington Park, CA 90255	6320-006-069
Maywood MOB	St. Francis Medical Center	5953 South Atlantic Boulevard 5, Maywood, CA 90270	6313-013-028
Parking Lot (St. Francis Medical Center)	Verity Holdings, LLC	3633 Martin Luther King Jr., Boulevard, Lynwood, CA 90262	6173-015-047
Ministry Services Building (St. Francis Medical Center)	Verity Holdings, LLC	3663 Martin Luther King Jr., Boulevard, Lynwood, CA 90262	6173-019-022
Parking Lot (St. Francis Medical Center)	Verity Holdings, LLC	3621 Martin Luther King Jr., Boulevard, Lynwood, CA 90262	6173-015-048
Five Unit Apartment (St. Francis Medical Center)	Verity Holdings, LLC	3570 Brenton Avenue, Lynwood, CA 90057	6191-016-008

# <u>Schedule 1.4.31.4(c)</u> St. Francis Medical Center

# Exhibit 1.3(a) to APA

### Form of Leaseback Agreement

See attached.

### SALE LEASEBACK AGREEMENT

[Note: This agreement is drafted for 3 Purchaser parties: Purchaser/Parent Company, Hospital Newco and Manager. This structure is not required.]

This Sale Leaseback Agreement (the "Leaseback Agreement") is made and entered into as of \_\_\_\_\_\_, 2020 (the "Signing Date"), by and among St. Francis Medical Center, a California nonprofit public benefit corporation (the "Corporation") on the one hand, and ("Parent Company") and

("Hospital Newco," and collectively with Parent Company, the "Parent Company Parties") on the other hand. The Corporation, Parent Company, and Hospital Newco may be referred to herein individually as a "Party," and collectively as the "partiesParties."

### RECITALS

**A.** On August 31, 2018, the Corporation and certain of its affiliates each filed a voluntary petition for relief (collectively, the "**Bankruptcy Cases**") under chapter 11 of title 11 of the United State Code (the "**Bankruptcy Code**").

**B.** The Bankruptcy Cases are jointly administered under Lead Case No. 18-20151 and are currently pending in the Bankruptcy Court for the Central District of California in Los Angeles (the "**Bankruptcy Court**").

C. Parent Company and the Corporation have entered into that certain Asset Purchase Agreement, dated of even date herewith (the "APA"), which provides for the sale of the assets of the Corporation and related assets (collectively, the "Assets"). Capitalized terms not defined in this Leaseback Agreement shall have the meanings given to such terms in the APA.

**D.** The Corporation operates St. Francis Medical Center as a 384-bed general acute care hospital (the "**Hospital**"), with its primary location at 3630 East Hospital Newco Highway, Lynwood, California 90262, and such other locations where the Hospital's services are provided (collectively, the "**Hospital Premises**").

**E.** Parent Company designated its affiliate, Hospital Newco, as the owner of the Corporation's Assets purchased by Parent Company under the APA (the "**Designation**"), and its affiliate \_\_\_\_\_\_ (the "**Manager**") as the operator of the Hospital.

**F.** Pursuant to Section 1.3 of the APA, the Corporation, Parent Company, and the Manager, are entering into that certain Interim Management Agreement ("IMA") of even date herewith, and commencing at the Effective Time, to enable the Manager to manage the day-to-day operations of the Hospital following the Closing until the Manager is issued the Licenses necessary to operate the Hospital (for the avoidance of doubt, that date will occur when the Manager is issued both a license to operate the Hospital as an acute care hospital by the California Department of Public Health, and a permit to operate a hospital-based pharmacy by the California Board of Pharmacy (collectively, the "New Licenses")).

A. Pursuant to the APA, at the Effective Time, Parent Company will purchase the Corporation's Assets, except for the Excluded Assets (as set forth in the APA). Also pursuant to the APA, that portion of the Assets constituting drugs, dangerous devices, pharmacy systems, or other pharmacy assets (the "**Pharmacy Assets**"), the Medicare/Medi-Cal Agreements, the

3

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 54 of 111

Lockboxes, the Transferred Managed Care Agreements and the Transferred Private Payor Agreements (collectively, the "Licensure Date Assets") shall transfer to Parent Company (or its affiliate) as of the Licensure Date.

**B.** Immediately following the Closing, and until the Manager obtains the New Licenses, the Parent Company Parties desire to lease back or license to the Corporation all of the then-acquired Corporation's Assets used in the operation of the Hospital, and the Corporation desires to so lease or license such Corporation's Assets from the Parent Company Parties on the terms and conditions set forth herein. The Corporation's Assets shall exclude the Licensure Date Assets, which the Corporation shall own until the New Licenses are issued.

### AGREEMENT

NOW, THEREFORE, in consideration of the covenants and conditions contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the <u>partiesParties</u> hereby agree as follows:

1. <u>Definitions</u>. Any capitalized term appearing herein that is not defined shall have the same definition ascribed to it under the IMA or the APA (as designated when such term first appears herein).

## 2. <u>Description of The Leased or Licensed Assets</u>.

**2.1** Leased or Licensed Assets. The Parent Company Parties hereby lease or license to the Corporation, and the Corporation leases or licenses from the Parent Company Parties, all of the Corporation's Assets used in the operation of the Hospital, except for the Licensure Date Assets, which the Corporation shall continue to own until the Licensure Date (as defined in the APA) (collectively, the "Leased or Licensed Assets"). Specifically, the Leased or Licensed Assets shall include: (a) the Hospital Premises (which shall be leased); (b) tangible personal property, including, but not limited to, fixtures, furnishings, hard copy medical and financial records and equipment (including, but not limited to, hardware to operate and run the electronic health record systems, hospital operating systems, laboratory information systems, and financial reporting systems) (which shall be leased); and (c) intangible intellectual property saved or embodied in the electronic health record systems, hospital operating systems, laboratory information systems, laboratory information systems, and financial reporting systems (which shall be licensed for use by the Corporation).

### 2.2 <u>Management of Pharmacy and Use of Pharmacy Assets</u>.

(a) The Corporation shall at all times during the term of this Leaseback Agreement be the owner of the Pharmacy Assets. The <u>partiesParties</u> acknowledge, however, that under the IMA, the Manager will be managing the Corporation's pharmacy during the Management Period (as defined in the IMA), and the Corporation therefore grants the Manager (i) access to and authority to use the drugs, dangerous drug delivery devices, or other tangible pharmacy assets, and (ii) a license to use the intangible intellectual property saved or embodied in the pharmacy systems, in each case to the extent necessary for the Manager to fulfill its obligations under the IMA.

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 55 of 111

(b) Pursuant to the IMA, the Corporation shall be responsible for purchasing drugs and dangerous devices identified by the Manager as necessary for the operations of the Hospital, and the Manager shall reimburse the Corporation for all costs and expenses incurred for such purchases.

**3.** <u>Term; Termination</u>. This Leaseback Agreement shall have a term coextensive with the Management Period (as defined in the IMA) and shall automatically terminate upon the termination of the IMA.

4. <u>Payments by the Parent Company Parties</u>.

**4.1** <u>Utilities</u>. The Parent Company Parties shall pay all utilities and services supplied to the Hospital during the term hereof, including but not limited to water, gas, air conditioning, heat, light, power, telephone service, and waste removal services.

**4.2** <u>Taxes</u>. The Parent Company Parties shall pay all taxes, assessments, and levies of any kind or nature whatsoever, including real property taxes, personal property taxes, income taxes, employment taxes, and sales or use taxes, that are taxed, assessed, levied, invoiced or imposed upon or against the Leased or Licensed Assets, the Hospital, and/or the Corporation, after the Effective Time.

**4.3** Insurance. The Parent Company Parties shall pay for all insurance coverages, including premiums, deductibles, stop-loss, and any other insurance covering the Leased or Licensed Assets, the Licensure Date Assets, the Hospital, and the Corporation during the term hereof. The Parent Company Parties covenant and agree that the Leased or Licensed Assets, the Licensure Date Assets, the Hospital, and the Corporation are covered as of the date hereof and will be covered at all times by general liability, fire, theft, business interruption, cyber, professional liability, directors and officers insurance, employment practices liability, terrorism, workers' compensation & employers' liability, directors and officers, fiduciary, crime, punitive damages excess liability, physical damage, property liability, automobile, storage tank, helipad and non-owned aviation, sexual misconduct and molestation, medical provider professional liability, and provider capitation stop loss (managed care excess loss) insurance. All such insurance shall name the Parent Company Parties and the Corporation as insureds as their respective interests may appear.

**4.4** <u>Repairs and Maintenance: Alterations</u>. During the term of this Leaseback Agreement, the Parent Company Parties shall pay all costs of repairing (including replacement of) and maintaining the Leased or Licensed Assets and Hospital and every part thereof in good and sanitary order, condition and repair during the term hereof, reasonable wear and tear excepted, <u>excluding, however, anyincluding, without limitation, all costs of all</u> repairs<u>or</u>, replacements related to seismic orand maintenance required by any applicable governmental law, <u>statute</u>, ordinance, rule or regulation, including the California Office of Statewide Health Planning and Development ("**OSHPD**")-related requirements; *provided, however*, that Parent-Company Parties may begin to make preparations for such seismic or OSHPD related repairs and maintenance that will be undertaken after the Management Period (as defined in the IMA). The Corporation shall not make any alterations or changes to the Leased or Licensed Assets without

### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 56 of 111

prior written approval of the Parent Company Parties, which may be given or withheld in the Parent Company Parties' sole discretion.

**4.5** <u>Payment</u>. Nothing in this <u>Section 4</u> shall in any way limit, reduce, or otherwise affect Parent Company's payment obligations under the IMA or the APA.

5. <u>Use</u>. The Leased or Licensed Assets shall be used for the operation of the Hospital, subject to the terms of the APA and the IMA.

# 6. <u>Risk of Loss</u>.

**6.1** The Corporation assumes no risks or liability for damage to or injury occurring to the Leased or Licensed Assets or Hospital during the term of this Leaseback Agreement by any means whatsoever, including fire, storm, earthquake, vandalism, strike, accident or any other casualty (collectively, "Casualty"), and the Parent Company Parties shall have all right, title, and interest in and to the proceeds of any insurance it obtained and paid for covering such Casualty.

**6.2** If, during the term of this Leaseback Agreement, action is initiated to take the Hospital Premises or any portion thereof by eminent domain <u>or condemnation</u> proceedings<u>exercise of state authority under an executive order</u> or by deed in lieu thereof (collectively, "**Condemnation**"), the Parent Company Parties, and not the Corporation, shall have all right, title, and interest in and to the award from the Condemnation.

**6.3** In the event of a Casualty or Condemnation, neither the Parent Company Parties nor the Corporation may terminate this Leaseback Agreement.

7. <u>Continued Access</u>. Following termination of this Leaseback Agreement, and until the entry of final decrees closing the Bankruptcy Cases, the Patient Care Ombudsman, appointed by the United States Trustee pursuant to Bankruptcy Code § 333 and approved by the Bankruptcy Court [*see* Docket No. 430], shall have continuing access to the Leased or Licensed Assets and related personnel during normal business hours and upon at least one (1) business day's prior written notice to the Parent Company Parties, for the purpose of winding down the Corporation's affairs, in connection with any litigation or adversary proceedings, and to effectuate the chapter 11 plan as approved by the Bankruptcy Court.

# 8. <u>Miscellaneous</u>.

**8.1** <u>Further Assurances</u>. Each of the <u>parties Parties</u> hereto agrees to execute and deliver any and all further agreements, documents or instruments necessary to effectuate this Leaseback Agreement and the transactions referred to herein or contemplated hereby or reasonably requested by the other Party to perfect or evidence their rights hereunder.

**8.2** <u>Notices and Demands</u>. All notices and demands, requests, consents, approvals, and other similar communications under this Leaseback Agreement shall be in writing and shall be sent by personal delivery or by either (a) United States certified or registered mail,

return receipt requested, postage prepaid, or (b) Federal Express or similar generally recognized overnight carrier regularly providing proof of delivery, addressed as follows:

If to the Corporation:	St. Francis Medical Center c/o Verity Health System of California, Inc. 601 South Figueroa Street, Suite 4050 Los Angeles, CA 90017-5704 Attention: Chief Executive Officer
With copies to: (which copy shall not constitute notice)	Tania Moyron, Esq. Dentons US LLP 601 South Figueroa St., Suite 2500 Los Angeles, CA 90017-5704 and

Hope Levy-Biehl, Esq. Davis Wright Tremaine LLP 865 S Figueroa St, Los Angeles, CA 90017

If to the Parent Company Parties:

With copies to: (which copies shall not constitute notice)

Any notice so given by mail shall be deemed to have been given as of the date of delivery (whether accepted or refused) established by U.S. Post Office return receipt or the overnight carrier's proof of delivery, as the case may be, whether accepted or refused. Any such notice not so given shall be deemed given upon receipt of the same by the Party to whom the same is to be given. Any Party hereto may designate a different address for itself by notice to the other parties in accordance with this Section 8.2.

**8.3** <u>Payment of Expenses</u>. Except for the Parent Company Parties' obligation to be responsible for certain costs, fees, and expenses as set forth elsewhere in this Leaseback Agreement, each Party hereto shall bear its own legal, accounting, and other expenses incurred in connection with the preparation and negotiation of this Leaseback Agreement and the consummation of the transactions contemplated hereby, whether or not the transaction is consummated.

**8.4** <u>Rent</u>. The Corporation has prepaid the sum of One Thousand Dollars (\$1,000.00), the receipt of which is hereby acknowledged by the Parent Company Parties, and the Corporation shall not be required to pay the Parent Company Parties any additional rent under this Leaseback Agreement.

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 58 of 111

**8.5** Entire Agreement; Amendment; Waiver. This Leaseback Agreement, the IMA, and those provisions of the APA expressly identified in this Leaseback Agreement contain the entire agreement among the <u>partiesParties</u> with respect to the subject matter hereof and supersede all prior agreements, arrangements, and understandings relating to the subject matter hereof and thereof. There are no written or oral agreements, understandings, representations, or warranties among the <u>partiesParties</u> other than those set forth in this Leaseback Agreement, the IMA, and those provisions of the APA expressly identified in this Leaseback Agreement. Nothing in this Leaseback Agreement modifies or shall be construed as modifying any orders entered by the Bankruptcy Court. This Leaseback Agreement may not be modified or amended except in writing signed by the <u>parties heretoParties</u>. No waiver of any term, provision or condition of this Leaseback Agreement in any one or more instances, shall be deemed to be or be construed as a further or continuing waiver of any such term, provision or condition of this Leaseback Agreement. No failure to act shall be construed as a waiver of any term, provision, condition or rights granted hereunder.

**8.6** <u>Assignment</u>. Neither this Leaseback Agreement nor the rights, duties or obligations arising hereunder shall be assignable or delegable by the Corporation or the Parent Company Parties without the prior written consent of the other <u>partiesParties</u>, which may be granted, denied or conditioned in such Party's absolute discretion except that the Parent Company Parties may assign this Leaseback Agreement in connection with any permitted assignment under the IMA. Subject to the foregoing, this Leaseback Agreement shall be binding upon, and inure to the benefit of, the respective successors and assigns of the <u>parties</u>.

**8.7** <u>Joint Venture; No Third Party Beneficiaries</u>. Nothing contained herein shall be construed as forming a joint venture or partnership among the <u>parties heretoParties</u> with respect to the subject matter hereof. The <u>parties heretoParties</u> do not intend that any third party shall have any rights under this Leaseback Agreement.

**8.8** <u>Captions</u>. The section headings contained herein are for convenience only and shall not be considered or referred to in resolving questions of interpretation.

**8.9** <u>Governing Law</u>. This Leaseback Agreement shall be governed by and construed in accordance with the laws of the State of California without regard to its conflicts of laws principles or decisions, except to the extent that the laws of such State are superseded by the Bankruptcy Code or other applicable federal law.

**8.10** Jurisdiction. The parties Parties agree that the Bankruptcy Court shall have exclusive personal and subject matter jurisdiction over any action or proceeding with respect to any claim arising out of or related to this Leaseback Agreement, and any of the documents or transactions contained in or contemplated by this Leaseback Agreement; *provided, however*, that the parties Parties agree that the United States District Court for the Central District of California (together with the Bankruptcy Court, the "Chosen Courts") shall have exclusive jurisdiction over such claim if (i) the Bankruptcy Cases are closed and if the Bankruptcy Cases are not reopened to adjudicate such claim after request by the Party bringing such claim or (ii) the Bankruptcy Court determines that it does not have jurisdiction over such claim. Solely in connection with claims arising under this Leaseback Agreement, or any of the documents or

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 59 of 111

transactions contemplated hereby, the <u>partiesParties</u> (a) irrevocably submit to the exclusive jurisdiction of the Chosen Courts, (b) waive any objection to laying venue in any such action or proceeding in the Chosen Courts, (c) waive any objection that the Chosen Courts are an inconvenient forum or do not have jurisdiction over any Party hereto, and (d) agree that service of process upon such Party in any such action or proceeding shall be effective if notice is given in accordance with <u>Section 8.2</u> hereof

**8.11** <u>Conditions to Effectiveness</u>. The <u>partiesParties</u> acknowledge that this Leaseback Agreement is subject to and contingent upon the occurrence of the Closing under the APA. This Leaseback Agreement shall not be effective, nor shall any Party have any obligations hereunder, unless and until the Closing under the APA has occurred.

**8.12** <u>Cooperation on Regulatory Compliance Matters</u>. The Parent Company Parties understand and acknowledges that the Corporation intends to comply with applicable federal and state laws, regulations, and guidance, as well as the requirements or recommendations of any accrediting agencies. In the event the terms of this Leaseback Agreement need to be amended or supplemented based on guidance from or at the request or direction of a regulator made during the term of this Leaseback Agreement, the Parent Company Parties shall cooperate with such amendment and/or supplement to ensure the Corporation's ability to comply with such guidance, request, recommendation or directive.

**8.13** <u>Transition Services Agreement</u>. Pursuant to Section <u>1.4.61.4(f)</u> and Section <u>1.5.41.5(d)</u> of the APA, the Sellers and Parent Company will enter into a Transition Services Agreement to facilitate the winding down of the Sellers' businesses, the completion of the Bankruptcy Cases, and the dissolution of the Sellers. The <u>parties heretoParties</u> shall cooperate with each other to enable the Corporation and Parent Company to carry out their obligations under, and give effect to the terms of, the Transition Services Agreement.

**8.14** <u>Fair Meaning</u>. This Leaseback Agreement shall be construed according to its fair meaning and as if prepared by all <u>parties heretoParties</u>.

**8.15** <u>Counterparts</u>. This Leaseback Agreement may be executed by one or more of the <u>parties heretoParties</u> on any number of separate counterparts (including by telecopy), and all of said counterparts taken together shall be deemed to constitute one and the same instrument. This Leaseback Agreement may be delivered by facsimile or electronic transmission, including by e-mail as a PDF, and facsimile or PDF copies of executed signature pages, which shall be binding as originals.

8.16 WAIVER OF JURY TRIAL. TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT, OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. EACH PARTY Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 60 of 111

CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (B) EACH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (C) EACH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (D) EACH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 8.16. THIS IS AN AFFIRMATIVE WAIVER OF THE PARTIES' RIGHTS TO A JURY TRIAL UNDER CALIFORNIA LAW, Cal. C. Civ. Pro. Sec 631. BY SIGNING BELOW ON THE SIGNATURE LINES, EACH PARTY IS EXPLICITLY WAIVING JURY TRIAL AND AUTHORIZING ANY AND ALL PARTIES TO FILE THIS WAIVER WITH ANY COURT AS THE WAIVER REQUIRED UNDER Cal. C. Civ. Proc. Sec. 631(f)(2).

[Signature Page Follows.]

# Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 61 of 111

**IN WITNESS WHEREOF,** the <u>partiesParties</u> agree to the foregoing terms of agreement through the execution below by their respective, duly authorized representatives as of the Signing Date.

Parent Company Parties:	Corporation:
NAME	ST. FRANCIS MEDICAL CENTER, a California non-profit public benefit corporation
By:	
Name:	By:
Title:	Name:
	Title:
By:	
Name:	
Title:	
Ву:	
Name:	
Title:	

# Exhibit 1.3(b) to APA

# Form of Interim Management Agreement

See attached.

### **INTERIM MANAGEMENT AGREEMENT**

[Note: This IMA is drafted for 3 Purchaser parties: Purchaser/Parent Company, Hospital Newco and Manager. This IMA is drafted assuming Purchaser is buying hospital's accounts receivable. This structure is not required.]

This Interim Management Agreement (the "Agreement") is made and entered into as of \_\_\_\_\_\_, 2020 (the "Signing Date"), by and among St. Francis Medical Center, a California nonprofit public benefit corporation (the "Corporation") on the one hand, and \_\_\_\_\_\_ ("Parent Company") and \_\_\_\_\_\_ (the "Manager," and collectively with the Corporation and Parent Company, the "parties" Parties" and each individually a "Party") on the other hand.

### RECITALS

**A.** On August 31, 2018, the Corporation and certain of its affiliates each filed a voluntary petition for relief (collectively, the "**Bankruptcy Cases**") under chapter 11 of title 11 of the United State Code (the "**Bankruptcy Code**").

**B.** The Bankruptcy Cases are jointly administered under Lead Case No. 18-20151 and are currently pending in the Bankruptcy Court for the Central District of California in Los Angeles (the "**Bankruptcy Court**").

C. Parent Company, as purchaser, and the Corporation and certain of its affiliates (collectively, the "Sellers" as defined in the APA), as seller, entered into that certain Asset Purchase Agreement, dated of even date herewith (the "APA"), which provides for the sale of certain assets of the Sellers (collectively, the "Assets"). All terms not otherwise defined herein shall have the meaning ascribed to them in the APA.

**D.** The Corporation operates St. Francis Medical Center as a 384-bed general acute care hospital (the "**Hospital**"), with its primary location at 3630 East Hospital Newco Highway, Lynwood, California 90262, and such other locations where the Hospital's services are provided (collectively, the "**Hospital Premises**")

**E.** Pursuant to the APA, Parent Company agreed to purchase the Assets, which include certain assets of the Corporation (the "Corporation's Assets").

F. Parent Company designated its affiliate, \_\_\_\_\_ ("Hospital Newco") as the owner of the Corporation's Assets, and the Manager as the operator of the Hospital.

**G.** Parent Company and the Manager have requested this Agreement to afford the Manager additional time to obtain its general acute care hospital license from the California Department of Public Health ("**CDPH**"), and its hospital pharmacy permit from the California

### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 64 of 111

Board of Pharmacy (together, the "New Licenses," and the date on which the New Licenses are issued is the "Licensure Date").

**H.** Pursuant to the APA, that portion of the Assets constituting drugs, dangerous devices, pharmacy systems, or other pharmacy assets (the "**Pharmacy Assets**"), the Medicare/Medi-Cal Agreements, the Lockboxes, the Transferred Managed Care Agreements and the Transferred Private Payor Agreements (collectively, the "**Licensure Date Assets**") shall transfer to Hospital Newco as of the Licensure Date.

**I.** The Corporation shall maintain a possessory interest in the Hospital and the Hospital Premises, and Parent Company and Hospital Newco on the one hand as lessor, and the Corporation, on the other hand as lessee, are entering into that certain Sale Leaseback Agreement of even date herewith, pursuant to which certain of the Corporation's Assets will be leased back to the Corporation (the "Leaseback Agreement").

J. Until Hospital Newco obtains the New Licenses, the Manager desires to assume the management of the Hospital, including its pharmacy, on behalf of the Corporation, and the Corporation desires to avail itself of such management services, upon the terms and conditions set forth in this Agreement.

## **TERMS OF AGREEMENT**

**NOW, THEREFORE,** in consideration of the promises and the mutual covenants contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the <u>partiesParties</u> hereby agree as follows:

1. Term.

**1.1** The term of this Agreement (the "**Management Period**") shall commence as of the Effective Time (as defined in the APA) and shall continue until the earlier of the Licensure Date or [June 30, 2021]. The parties Parties acknowledge that, during the Management Period, the Corporation shall remain the licensee of the Hospital, and in that capacity, and during such period, shall retain statutory and regulatory authority and responsibility for the Hospital and for oversight of the Manager.

## 2. Acknowledgements and Covenants.

**2.1** As of the Effective Time, Parent Company shall have acquired all of the Corporation's Assets as set forth in the APA (other than the Licensure Date Assets), which shall be leased back to the Corporation pursuant to the terms of the Leaseback Agreement during the term of this Agreement.

**2.2** Pursuant to Section 1.3 of the APA, contemporaneously with the date the New Licenses are issued and the termination of the Leaseback Agreement, ownership of the Licensure Date Assets shall be transferred to Hospital Newco. Parent Company and the Manager acknowledge that this Agreement and the foregoing subsequent transfer of the Licensure Date Assets are made at the request of Parent Company and the Manager, to provide more time for the

### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 65 of 111

Manager to obtain its New Licenses. As such, Parent Company acknowledges, covenants, and agrees that the total Purchase Price (as defined in the APA) shall be paid to the Sellers under the APA as if the Licensure Date Assets were transferred as of the Effective Time.

**2.3** As of the Effective Time, the Corporation shall have ended the employment or engagement of all employees and contractors and, to the extent they received offers of employment or engagement from the Manager (or its affiliates) and accepted such offers, shall have been transferred to the Manager (or its affiliate) as employees or contractors of the Manager (collectively, the "**Hired Employees**").

**2.4** During the Management Period, the Corporation will (a) assign an individual to oversee the operation of the Hospital and serve as the president of the Corporation, as required by the Centers for Medicare & Medicaid Services, CDPH, and applicable California laws and regulations (the "**Responsible Officer**"), and (b) maintain proper oversight by a board of directors.

# **3.** Appointment of Manager.

**3.1** During the Management Period, the Corporation hereby appoints the Manager as the sole and exclusive provider of the Services (defined below) and hereby grants to the Manager the exclusive right to manage the Hospital under the Corporation's Licenses (as defined in the APA) as a general acute care hospital, including without limitation, the right to undertake those certain management responsibilities and permitted activities described in <u>Section 4</u> below. The Manager hereby accepts such appointment for all purposes with respect to the Corporation's rights, duties, and responsibilities under the Licenses for the Hospital, to the fullest extent permitted by law, and agrees, to the fullest extent permitted by law, to provide management services to the Hospital on behalf of the Corporation (the "Services").

**3.2** The Manager's Services hereunder shall include management and operation of the Hospital's pharmacy on behalf of the Hospital, even though the Pharmacy Assets will not be transferred to Hospital Newco pursuant to the APA until the Licensure Date.

**3.3** Upon the Licensure Date, the Manager's Services provided to the Hospital under the Corporation's Licenses shall terminate and, thereafter, Hospital Newco (and its affiliates) will be operating the Hospital as the licensee holding its own New Licenses.

**3.4** During the Management Period, the Manager shall submit claims for services rendered by the Hospital to various governmental and non-governmental entities, patients, and other third parties pursuant to the Corporation's provider agreements, payor contracts, and NPI numbers as set forth in <u>Schedule 3.4</u>, attached hereto (collectively, the "**Corporation's Billing Credentials**"). Because all billing and collecting shall be under the Corporation's Billing Credentials, payments shall be made in the Corporation's name and deposited in the Corporation's bank accounts. The <u>partiesParties</u> acknowledge and agree that during the Management Period, the Corporation's bank accounts and lockboxes shall remain under the Corporation's name and in the Corporation's control. Consistent with <u>Section 4.5(b)</u>, the Manager may transfer the Manager Compensation to its accounts. To the extent the Manager is unable to make such transfers to its own accounts, upon request by the Manager, the Corporation

### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 66 of 111

will, at the Manager's cost, forward to the Manager any payment or reimbursement received during the Management Period.

**3.5** To the extent the Corporation's lenders continue to have liens, security interests, charges, mortgages, or any other encumbrances whatsoever (collectively, "**Encumbrances**") on the Corporation's bank accounts or lockboxes as of the Effective Time, the Corporation, Parent Company, and the Manager shall work cooperatively to facilitate termination and release of such Encumbrances.

**3.6** The Corporation shall retain and, upon the Licensure Date, the Manager shall assume, any contracts necessary for the Corporation to continue to be the holder of the Licenses and to bill for Hospital services during the Management Period, in accordance with the APA. Such contracts shall be as set forth in <u>Schedule 3.6</u>, attached hereto (the "**Retained Contracts**"). Parent Company hereby designates each Retained Contract as an Assigned Contract under Section 1.11 of the APA.

## 4. Management Responsibility.

4.1 During the Management Period, the Manager shall, subject to all applicable legal and regulatory requirements and the Corporation's ultimate oversight and control, have responsibility for the management of the Hospital, and agrees to assume and discharge all responsibilities, duties, liabilities, payments, and obligations in connection with properly maintaining the Hospital in full compliance with all regulations and standards required of a general acute care hospital facility so licensed. In furtherance thereof, the Manager's Services shall include, but not be limited to, the following duties, which duties shall be performed at the Manager's sole cost and expense:

(a) Managing the operations of the Hospital as a general acute care hospital in compliance with all applicable laws, regulations, provider agreements, payor contracts, CDPH requirements for maintenance of the Licenses in good standing, Medicare conditions of participation and requirements for payment with respect to governmental programs, and the requirements for maintenance of the Hospital's accreditations;

(b) Employing and managing the Hired Employees and any other non-clinical and clinical personnel (i) deemed necessary by the Corporation for the operation of the Hospital as a general acute care hospital, or (ii) required by law so as to meet the all applicable labor laws and regulations, and consistent with orders of the Bankruptcy Court;

(c) Ensuring that the Corporation is able to pay itself out of the Hospital Revenues for the costs and expenses set forth in <u>Sections 4.5(b)</u>;

(d) Maintaining and repairing, as needed, the Hospital Premises so as to ensure material compliance with all applicable local, state and federal law, except for any repairsor replacements related to seismic or the California Office of Statewide Health Planning and Development ("OSHPD")-related requirements; *provided, however,* that Manager may begin tomake preparations for such seismic or OSHPD-related repairs and maintenance that will be undertaken after the Management Periodand construction timelines imposed by OSHPD.

### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 67 of 111

(e) Providing security services reasonably necessary to prevent unlawful entry or damage to the Hospital Premises;

(f) Affording the Responsible Officer or his or her designee access, during normal business hours, to the Hospital Premises, the books and records at the Hospital Premises or in Manager's possession, the Hired Employees and any other personnel of Manager or otherwise who are providing services associated with the operation of the Hospital, and such other access and assistance as reasonably requested by the Responsible Officer;

(g) Upon at least one (1) day's prior written notice to the Manager, providing access, during normal business hours, to the Hospital Premises to lessors of equipment at the Hospital Premises, if any, who have been authorized by order of the Bankruptcy Court to remove their equipment from the Hospital Premises, provided that the Manager shall have full power and authority to require that the removal of such equipment by such lessors does not damage the Hospital Premises;

(h) Providing access, during normal business hours, to the Hospital, the Hospital Premises, the Hospital's books and records, electronic health records, financial information systems, operating systems, laboratory systems, the Hired Employees and any other personnel of Manager or otherwise who are providing services associated with the operation of the Hospital, to the Corporation, the Corporation's directors, officers and representatives, and the Corporation's successors in interest, including, but not limited to, any plan administrator, liquidating trustee or similar representatives appointed or approved by the Bankruptcy Court, as well as the Patient Care Ombudsman appointed in the Bankruptcy Cases under Section 333 of the Bankruptcy Code (the "PCO") [see Docket No. 430], for the purpose of winding down the Corporation's affairs, pursuing litigation and adversary proceedings, and to effectuate a plan of liquidation, as approved by the Bankruptcy Court (the "Plan"), provided, however, that such access does not unreasonably disrupt the Hospital's operations;

(i) Maintaining, all licenses, permits consents, approvals, accreditations, and certifications currently held by the Corporation in good standing, in active status, and in compliance with all applicable local, state, and federal laws, including the timely payment of all applicable fees to support or renew these approvals;

(j) Maintaining and obtaining all insurance coverages, from and after the Effective Time, for the Hospital that a prudent hospital operator or owner would maintain, including directors and officers insurance with no less coverage than was maintained for directors and officers just prior to the Effective Time;

(k) Maintaining and obtaining those insurance coverages required under the Leaseback Agreement, for its own account, with the Corporation included as a named insured, and paying all amounts required under the Leaseback Agreement in a timely manner, including rent, utilities, taxes, and insurance premiums;

(1) Opening and forwarding all mail relating to the financial or business affairs of the Corporation to the notice address below;

### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 68 of 111

(m) Periodically reporting to the Corporation (or its designee), either in person or telephonically, the condition of the Hospital and the Hospital Premises;

(n) Coordinating with the governing board and the organized medical staff (each, as established by the Corporation and the Sellers) on the appropriateness and quality of medical care and all medical staff issues requiring governing board oversight;

(o) Paying all costs and expenses in connection with and incidental to ownership of the Corporation's Assets and management and operation of the Hospital hereunder, including but not limited to, all the Hospital operating costs, employee-related costs, and taxes, whether or not identified, described, or referenced in this Agreement;

(p) Cooperating with the Corporation in facilitating termination and release of any Encumbrances on the Corporation's bank accounts and lockboxes; and

(q) Performing such other duties and activities as are reasonably necessary for the Manager to fulfill its responsibilities under this Agreement and the APA.

**4.2 Permitted Manager Activities**. During the Management Period, the Manager may do any of the following, in consultation with the Corporation and subject to the requirements of applicable local, state, and federal law, which activities may be performed by the Manager at the Manager's sole cost and expense:

(a) Make alterations, improvements, and repairs to the interior or exterior of the Hospital Premises, including structural alterations, improvements, and repairs;

(b) Remove and dispose of furniture, fixtures, equipment (other than equipment owned by equipment lessors), and supplies at the Hospital Premises;

(c) Move into and install furniture, fixtures, equipment, and supplies at the Hospital Premises;

(d) Prepare the Hospital for a name change, except that no such name change may take effect, and no signage reflecting such change shall be installed, during the Management Period; and

(e) Perform, or permit to be performed, any other activities at the Hospital Premises that are not inconsistent with operating the Hospital under the Licenses, and receiving and retaining for the Manager's own account all revenues and proceeds of any such activities, to the extent they comprise the Manager Compensation.

# 4.3 **Prohibited Manager Activities.**

(a) Notwithstanding anything to the contrary in this Agreement, the Manager shall have no authority to take and shall not take any action with respect to any Excluded Assets or Excluded Liabilities (as such terms are defined in the APA) of the Corporation.

### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 69 of 111

(b) The Manager's authority to manage and operate the Hospital is limited to those actions that Manager is expressly required or permitted to do hereunder.

(c) The Manager shall not (i) take any action that interferes with the Corporation's transfer of funds to pay itself out of the Hospital Revenues as set forth in <u>Section</u> 4.5(b); or (ii) remove, withdraw, or authorize removal or withdrawal of funds from the Corporation's bank accounts or lockboxes to the extent that the Corporation would be unable to fully pay itself as set forth in <u>Section 4.5(b)</u>.

## 4.4 APA Provisions; Transition Services Agreement.

(a) Nothing herein shall modify the prorations of expenses and utilities set forth in the APA, including specifically as set forth in <u>Section 1.6</u> of the APA.

(b) None of the information accessed, learned or obtained by Parent Company or the Manager or any of their affiliates in the course of performing their duties hereunder may serve as the basis for payment of less than the full Purchase Price or to otherwise assert a claim against the Sellers.

(c) Nothing herein shall modify the transfer of the Assets from the Sellers to Parent Company as contemplated in the APA, including specifically as set forth in <u>Section 1.7</u> of the APA.

(d) Nothing herein shall modify the APA in respect of the exclusion from purchase by Parent Company of the Excluded Assets, including specifically as set forth in <u>Section 1.8</u> of the APA.

(e) Pursuant to Section <u>1.4.61.4(f)</u> and Section <u>1.5.41.5(d)</u> of the APA, the Sellers and Parent Company will be entering into a Transition Services Agreement to facilitate the winding down of the Sellers' businesses, the completion of the Bankruptcy Cases, and the dissolution of the Sellers. The <u>parties heretoParties</u> shall cooperate with each other to enable the Corporation and Parent Company to carry out their obligations under, and give effect to the terms of, the Transition Services Agreement.

## 4.5 Manager Compensation; Distribution of Revenue.

(a) As used in this Agreement, the term "**Hospital Revenues**" shall mean all revenues and reimbursement received by the Corporation after the Effective Time, including, but not limited to, cash, accounts, notes, or other accounts receivable, disproportionate share payments, quality assurance fee payments, Seller Cost Report (as defined in the APA) settlements, and capitation premiums, whether payable by Medicare, Medi-Cal or any other commercial or governmental payor, or any health maintenance organization or any other managed care program or any private pay patients, but only to the extent they are not Excluded Assets (as defined in the APA).

I

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 70 of 111

(b) As full and complete payment for the Manager's Services, the Manager shall be entitled to receive an amount equal to the Hospital Revenues less the following amounts that will be paid to the Corporation by Manager (the "Manager Compensation"):

(i) All costs and expenses incurred by the Corporation for the Corporation's purchase of drugs and dangerous devices that the Manager determines are necessary for the operation of the Hospital;

(ii) All salaries, stipends, costs and expenses associated with the employment or engagement of the board of directors of the Corporation<u>and its parent</u> organization, VHS, the Responsible Officer and any <u>other</u> retained employees or contractors of the Corporation primarily or exclusively providing services necessary for the operation of the Hospital<u>and compliance with applicable laws</u>, if any; and

(iii) All other costs and expenses associated with keeping the Corporation in good standing, maintaining its Licenses and maintaining the Retained Contracts, including, but not limited to, filings with the secretary of state, filing tax returns, board of directors expenses, directors and officers insurance, employment practices liability insurance, compensation, benefits, and such other costs and expenses incurred by a hospital corporation in the normal course of business.

(c) Corporation shall instruct the financial institutions that service its bank accounts or lockboxes to, immediately following the Effective Time and during the Management Period, sweep all Hospital Revenues from the Corporation's bank accounts or lockboxes on a daily basis into an account designated in writing by Manager to Corporation, and Corporation shall not revise such written instructions without obtaining the prior written consent of Manager.

(d) Under no circumstance shall the Manager seek payment for the Manager's Services from the liquidating trustee, the Corporation, any of the Corporation's officers, directors, agents, contractors, personnel, affiliates or subsidiaries, Verity Health System of California, Inc. ("VHS"), or any of VHS' officers, directors, agents, contractors, personnel, affiliates or subsidiaries.

(e) The Corporation shall issue invoices to the Manager on a weekly basis for the items described in Section 4.5(b) (i), (ii) and (iii), with reasonable supporting detail therefor. The Manager shall pay such invoices within ten (10) business days of receipt of such invoices. If the Manager does not remit payment in respect of such invoices in accordance with the immediately preceding sentence, the unpaid amount of such invoices shall bear interest at the Prime Rate in effect on the calendar day upon which such payment was required to be made to the Corporation (the "Invoice Payment Due Date") plus five percent (5%) (or the maximum rate allowed by law, whichever is less), such interest accruing on each calendar day after the Invoice Payment Due Date until payment of such invoices and all interest thereon is made to the Corporation.

## 4.6 Liabilities and Losses.

(a) Parent Company shall be responsible for all the Hospital liabilities and losses incurred or accrued during the Management Period.

(b) Nothing hereunder shall, or is intended to, modify or supersede Parent Company's responsibility for the Transferred Obligations (as defined in the APA) or the Corporation's responsibility for the Excluded Liabilities (as defined in the APA).

(c) Specifically, with respect to any and all quality assurance fee ("QAF") liabilities owed by the Corporation, Parent Company shall pay all QAF fees that become due and owing or are overdue during the Management Period, even though such fees may have been calculated based on data prior to the Effective Time.

(d) If Parent Company (or its affiliate) has designated any health plan capitation agreements to be assigned to Hospital Newco, then Parent Company hereby assumes all liabilities and responsibilities related to managing the members covered by such health plan capitation agreements, including arranging for services from and payment under all related risk pool arrangements, all related arrangements with independent physician associations (IPAs), all downstream provider agreements, and all out-of-network provider arrangements or services, without regard to whether the member services were rendered prior to or after the Effective Time.

**4.7** The Corporation's Ultimate Control. Notwithstanding anything to the contrary in this <u>Section 4</u> or in this Agreement more generally, the Corporation, as holder of the Licenses, shall remain ultimately responsible for the operation of the Hospital, and may, at any time and from time-to-time during the Management Period, take any action (at Parent Company's or the Manager's sole cost and expense) necessary to ensure the Corporation's compliance with applicable laws and regulations, even if such action requires the Corporation to intervene in the Manager's performance of the Manager's duties or permitted activities pursuant to this <u>Section 4</u>.

# 5. Continued Responsibility of the Corporation.

**5.1** During the Management Period, the Corporation shall maintain (at Parent Company's or the Manager's sole cost and expense), and shall not take or voluntarily permit any actions which may adversely affect, the Corporation's corporate existence and its full rights as the licensee under the Licenses. In addition, during the Management Period, the Corporation and its officers shall reasonably cooperate with the Manager (at Parent Company's or the Manager's sole cost and expense) in the Manager's provision of the Services.

**5.2** Notwithstanding the statutory and regulatory authority and responsibility of the Corporation for the continued management of the Hospital during the Management Period, the parties<u>Parties</u> recognize and acknowledge that under this Agreement, the Manager shall, subject to the ultimate oversight by and approval of the Corporation, be responsible for the day-to-day operation and maintenance of the Hospital as a general acute care hospital. In the event that any violation or alleged violation of or non-compliance with any statute or regulation applicable to the operation or maintenance of the Hospital as a general acute care hospital certified by the

I

### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 72 of 111

Medicare and Medi-Cal programs occurs during the Management Period, then without regard to legal or statutory fault on the part of the Manager or of the Corporation, the Manager shall immediately notify the Corporation of such violation or alleged violation or non-compliance and take reasonable efforts to avoid or minimize any related adverse consequences. Parent Company and the Manager shall be responsible for the costs of any penalty, fine or remediation identified during the Management Period arising out of or relating to any act, omission, event or occurrence connected with the operation of the Hospital after the Effective Time, including, without limitation, the cost of engaging third party consultants or experts to help address or resolve the violation, alleged violation or non-compliance, and shall indemnify and hold the Corporation harmless for the same in accordance with <u>Section 9.2</u>. The Corporation retains the right to join the Manager in contesting said violations upon providing the Manager with notice of its intent to do so.

**5.3** The Corporation shall be responsible for purchasing drugs and dangerous devices that the Manager determines are necessary for the operation of the Hospital at the Manager's sole cost and expense, as set forth in Section 4.5(b)(i).

**5.4** The Corporation agrees to execute and deliver to the Manager such documents as the Manager may reasonably request to maintain the hospital license active and in good standing with CDPH and the other Licenses necessary or appropriate to maintain the Hospital as a general acute care hospital and to facilitate the Manager's obtaining of the New Licenses.

6. The Parties' Cooperation with Regulatory Agencies. The Manager shall use its best efforts to obtain the New Licenses as expeditiously as possible. The Manager shall provide updates to the Corporation weekly, and as requested by the Corporation, on the status of the Manager's efforts to obtain the New Licenses. To the extent not already submitted prior to the Effective Time, the Manager shall provide a copy of all such initial and supplementary or amended license applications for the New Licenses to Corporation at least two (2) business days prior to submission. The Manager and the Corporation shall cooperate with each other in the event the Corporation requests revisions to a license application for a New License. The Corporation shall, at Parent Company's or the Manager's cost, reasonably cooperate with the Manager's efforts to obtain the New Licenses, and may communicate and coordinate with licensing agencies as necessary in connection with obtaining the New Licenses. Notwithstanding the foregoing, obtaining all governmental consents, approvals, assignments, authorizations, and clearances necessary to obtain the New Licenses shall be solely Parent Company's and the Manager's (and not the Corporation's) responsibility, including payment of any fees, expenses, filing costs or other amounts related thereto.

# 7. Risk of Loss.

7.1 The Corporation assumes no risks or liability for damage to or injury occurring to the Hospital Premises, Assets or the Hospital during the term of this Agreement by any means whatsoever, including fire, storm, earthquake, vandalism, strike, cyberattack, accident or any other casualty (collectively, "Casualty"), and Parent Company shall have all right, title, and interest in and to the proceeds of any insurance it obtained and paid for covering such Casualty.

I

### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 73 of 111

7.2 If, during the term of this Agreement, action is initiated to take the Hospital Premises or any portion thereof by eminent domain proceedings or by deed in lieu thereof (collectively, "Condemnation"), Parent Company, and not the Corporation, shall have all right, title, and interest in and to the award from the Condemnation.

**7.3** In the event of a Casualty or Condemnation, neither Parent Company nor the Corporation may terminate this Agreement.

8. Continued Access. Following the expiration or termination of this Agreement and until the entry of final decrees closing the Bankruptcy Cases, if expressly authorized or directed by the Bankruptcy Court, and only to the extent so authorized or directed, the PCO shall have continuing access to the Corporation's Assets and related personnel during normal business hours and upon at least one (1) business day's prior written notice to Parent Company or the Manager, for the purpose of winding down the Corporation's affairs, in connection with any litigation or adversary proceedings, and to effectuate the Plan.

## 9. Exculpation; Indemnification.

**9.1** The Corporation and the Corporation's affiliates, members, officers, directors, employees, attorneys, accountants, consultants, agents, representatives, successors and assigns, including the liquidating trustee and responsible officer (collectively the "Corporation Indemnified Parties") shall have no liability in contract, tort or otherwise unless and until a Chosen Court finds in a final, non-appealable judgment that any Damages result solely from a Corporation Indemnified partyParty's gross negligence or willful misconduct.

9.2 Parent Company and the Manager shall promptly and fully keep and hold the Corporation Indemnified Parties forever harmless from, and shall indemnify and defend the Corporation Indemnified Parties from and against, without regard to materiality, any and all obligations, judgments, fines, civil money penalties, sanctions, awards, liabilities, losses, penalties, claims, costs, demands, damages, expenses, liens, and encumbrances, including investigation costs, time spent in depositions and reasonable attorneys' fees and expenses (collectively, "Damages"), whether civil or criminal, direct, indirect or consequential and no matter how arising, in any way related to, connected with, arising or resulting from, or under this Agreement, the APA, the Hired Employees (as defined in the APA), the Manager's performance of the Services, or the operation or management of the Hospital or the Corporation's Assets, in each instance after the Effective Time. Notwithstanding the foregoing, the parties Parties understand that except as otherwise specifically provided for in the APA, Parent Company and the Manager are not, by virtue of this Agreement or any term or provision herein, assuming any claim, liability, expense, debt or other obligation of the Corporation that both relates to the operation or management of the Hospital or the Corporation's Assets prior to the Effective Time and constitutes an Excluded Liability under the APA.

10. HIPAA Compliance. The Manager agrees to take such steps as are necessary to ensure compliance with the Health Insurance Portability and Accountability Act of 1996 (HIPAA), the California Confidentiality of Medical Information Act, and other applicable federal and state privacy laws and regulations (collectively, the "**Privacy Laws**") with respect to the Hospital and its operations, and the Corporation agrees not to take or voluntarily permit any actions which

violate Privacy Laws with respect to the Hospital or its operations. Toward this end, Parent Company, the Manager, and the Corporation agree to execute and deliver that certain Business Associate Agreement, attached hereto as <u>Exhibit A</u> and incorporated by reference herein, upon execution of this Agreement.

11. Further Assurances. Each of the <u>partiesParties</u> hereto agree to execute and deliver any and all further agreements, documents or instruments necessary to effectuate this Agreement and the transactions referred to herein or contemplated hereby or reasonably requested by any other <u>partyParty</u> hereto to perfect or evidence their rights hereunder.

12. Relationship of Parties. In performing their duties and permitted activities under this Agreement, Parent Company, the Manager, and the Corporation shall, at all times be acting and performing as independent contractors. Parent Company, the Manager, and the Corporation are not partners or joint venturers with each other and nothing herein shall be construed as making them partners or joint venturers or imposing upon any of them any liability as partners or joint venturers.

**13.** Notices and Demands. All notices and demands, requests, consents, approvals, and other similar communications under this Agreement shall be in writing and shall be sent by personal delivery or by either (a) United States certified or registered mail, return receipt requested, postage prepaid, or (b) Federal Express or similar generally recognized overnight carrier regularly providing proof of delivery, addressed as follows:

If to the Corporation:	St. Francis Medical Center c/o Verity Health System of California, Inc. 601 South Figueroa Street, Suite 4050 Los Angeles, CA 90017-5704 Attention: Chief Executive Officer
With copies to: (which copy shall not constitute notice)	Tania Moyron, Esq. Dentons US LLP 601 South Figueroa St., Suite 2500 Los Angeles, CA 90017-5704
	and

Hope Levy-Biehl, Esq. Davis Wright Tremaine LLP 865 S Figueroa St, Los Angeles, CA 90017

If to Parent Company and/or the Manager: With copies to: (which copies shall not constitute notice)

Any notice so given by mail shall be deemed to have been given as of the date of delivery (whether accepted or refused) established by U.S. Post Office return receipt or the overnight carrier's proof of delivery, as the case may be, whether accepted or refused. Any such notice not so given shall be deemed given upon receipt of the same by the <u>partyParty</u> to whom the same is to be given. Any <u>partyParty</u> hereto may designate a different address for itself by notice to the other <u>partyParty</u> in accordance with this <u>Section 13</u>.

14. Expenses. Except for Parent Company's and the Manager's obligations to be responsible for certain costs, fees, and expenses as set forth elsewhere in this Agreement, each <u>partyParty</u> to this Agreement shall pay its own expenses in connection with the preparation of this Agreement and the consummation of the transactions contemplated hereby, including the fees of any attorneys, accountants, financial advisors, investment bankers or other professionals engaged by such <u>partyParty</u>.

**15.** Entire Agreement. This Agreement, the Leaseback Agreement, and those provisions of the APA expressly identified in this Agreement, contain the entire agreement among the parties<u>Parties</u> with respect to the subject matter hereof and supersede all prior agreements, arrangements, and understandings relating to the subject matter hereof and thereof. There are no written or oral agreements, understandings, representations, or warranties among the parties<u>Parties</u> other than those set forth in this Agreement, the Leaseback Agreement, and those provisions of the APA expressly identified in this Agreement. Nothing in this Agreement modifies or shall be construed as modifying any orders entered by the Bankruptcy Court.

**16. Amendment**. This Agreement may not be modified, amended, altered or supplemented except by a written agreement executed by all the <u>parties heretoParties</u>.

17. Waiver. Waiver by any <u>partyParty</u> of any breach or failure to comply with any provision of this Agreement by any other <u>partyParty</u> shall not be construed as or constitute, a continuing waiver of such provision, or a waiver of any other breach of, or failure to comply with, any other provision of this Agreement. No waiver of any such breach or failure or of any term or condition of this Agreement shall be effective unless in a written notice signed by the waiving <u>partyParty</u> und delivered, in the manner required for notices generally, to each affected <u>partyParty</u>.

18. Severability. In case any provision of this Agreement shall be found by a court of competent jurisdiction to be invalid, illegal or unenforceable, such provision shall be construed and enforced as if it had been narrowly drawn so as not to be invalid, illegal or unenforceable, and the validity, legality and enforceability of the remaining provisions of this Agreement shall not in any way be affected or impaired thereby.

**19.** Successors and Assigns. This Agreement shall be binding upon, and shall inure to the benefit of, the <u>parties heretoParties</u> and their respective permitted successors and assigns. The

#### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 76 of 111

Corporation shall not be permitted to assign its rights or its obligations under this Agreement without the prior consent of Parent Company and the Manager. The <u>partiesParties</u> further acknowledge and agree that the Manager may subcontract for any of the goods or services required to be provided by the Manager pursuant to this Agreement, and the Manger may assign any of its rights hereunder and/or delegate any of its obligations hereunder, so long as in each case the Manager remains responsible for such subcontracted goods or services and for any of such Manager obligations hereunder.

**20.** Attorneys' Fees. In the event of any litigation or arbitration between the partiesheretoParties arising out of this Agreement, the prevailing partyParty therein shall be allowed to recover from the other partyParty all court costs and reasonable attorneys' fees which shall be fixed by the court or arbitrator.

**21. Headings**. The descriptive headings of sections and subsections of this Agreement are inserted for convenience only and do not constitute a part of this Agreement and shall not affect in any way the meaning or interpretation of this Agreement.

**22.** Governing Law. This Agreement shall be governed by and construed in accordance with the laws of the State of California without regard to its conflicts of laws principles or decisions<u>except to the extent that the laws of such State are superseded by the Bankruptcy Code or other applicable federal law.</u>

23. Jurisdiction. The parties Parties agree that the Bankruptcy Court shall have exclusive personal and subject matter jurisdiction over any action or proceeding with respect to any claim arising out of or related to this Agreement, and any of the documents or transactions contained in or contemplated by this Agreement; provided, however, that the parties Parties agree that the United States District Court for the Central District of California (together with the Bankruptcy Court, the "Chosen Courts") shall have exclusive jurisdiction over such claim if (i) the Bankruptcy Cases are closed and if the Bankruptcy Cases are not reopened to adjudicate such claim after request by the partyParty bringing such claim or (ii) the Bankruptcy Court determines that it does not have jurisdiction over such claim. Solely in connection with claims arising under this Agreement, or any of the documents or transactions contemplated hereby, the parties Parties (a) irrevocably submit to the exclusive jurisdiction of the Chosen Courts, (b) waive any objection to laying venue in any such action or proceeding in the Chosen Courts, (c) waive any objection that the Chosen Courts are an inconvenient forum or do not have jurisdiction over any partyParty hereto, and (d) agree that service of process upon such partyParty in any such action or proceeding shall be effective if notice is given in accordance with Section 13 hereof.

**24.** Commencement. The <u>parties</u> acknowledge that commencement of the Manager's Services under this Agreement is subject to and contingent upon the occurrence of the Closing (as defined in the APA). The Management Period shall not commence, and the <u>partiesParties</u>' obligations during the Management Period shall not commence, unless and until the Closing has occurred.

**25.** Cooperation on Regulatory Compliance Matters. Parent Company and the Manager understand and acknowledge that the Corporation intends to comply with applicable federal and state laws, regulations, and guidance. In the event the terms of this Agreement need to be

amended or supplemented based on guidance from or at the request or direction of a regulator made during the term of this Agreement, Parent Company and the Manager shall cooperate with such amendment and/or supplement to ensure the Corporation's ability to comply with such guidance, request or directive.

**26.** Counterparts. This Agreement may be executed by one or more of the partiesheretoParties on any number of separate counterparts (including by telecopy), and all of said counterparts taken together shall be deemed to constitute one and the same instrument. It may be delivered by facsimile or electronic transmission, including by e-mail as a PDF, and facsimile or PDF copies of executed signature pages, which shall be binding as originals.

27. WAIVER OF JURY TRIAL. TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT, OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (B) EACH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (C) EACH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (D) EACH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 27. THIS IS AN AFFIRMATIVE WAIVER OF THE PARTIES' RIGHTS TO A JURY TRIAL UNDER CALIFORNIA LAW, Cal. C. Civ. Pro. Sec 631. BY SIGNING BELOW ON THE SIGNATURE LINES, EACH PARTY IS EXPLICITLY WAIVING JURY TRIAL AND AUTHORIZING ANY AND ALL PARTIES TO FILE THIS WAIVER WITH ANY COURT AS THE WAIVER REOUIRED UNDER Cal. C. Civ. Proc. Sec. 631(f)(2).

[Signature page follows.]

# Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 78 of 111

**IN WITNESS WHEREOF,** the <u>partiesParties</u> agree to the foregoing terms of agreement through the execution below by their respective, duly authorized representatives as of the Signing Date.

## MANAGER:

# **CORPORATION:**

NAME

By:	
Name:	
Title:	

ST. FRANCIS MEDICAL CENTER,
a California non-profit public benefit
corporation

By:	
Name:	
Title:	

## **PARENT COMPANY:**

NAME

By:	
Name:	
Title:	

# **HOSPITAL NEWCO:**

NAME

By:		
Name:		
Title:		

# **Exhibit A to IMA**

**Business Associate Agreement** 

I

# Schedule 3.4 to IMA

**Corporation's Billing Credentials** 

I

Schedule 3.6 to IMA

**Retained Contracts** 

# Exhibit **<u>1.4.1</u>** to APA

# Form of Bill of Sale

See attached.

# Exhibit **<u>1.4.2</u><u>1.4(b)</u>** to APA

# Form of Real Estate Assignment Agreements

See attached.

# Exhibit <u>1.4.31.4(c)</u> to APA

# Form of Quitclaim Deed

See attached.

# Exhibit **<u>1.4.4</u>** to APA

## Form of Transfer Agreement

See attached.

# Exhibit 1.4.61.4(f) to APA

# Form of Transition Services Agreements

See attached.

#### **TRANSITION SERVICES AGREEMENT**

This Transition Services Agreement (the "Agreement"), is made and entered into as of \_\_\_\_\_\_, 2020 (the "Signing Date"), by and among Verity Health System of California, Inc., a California nonprofit public benefit corporation ("Verity"), Verity Holdings, LLC, a California limited liability company ("Verity Holdings"), St. Francis Medical Center, a California nonprofit public benefit corporation ("St. Francis" and together with Verity and Verity Holdings, collectively the "Sellers" and each individually a "Seller") and \_\_\_\_\_\_

("**Purchaser**"). The Sellers, the Liquidating Trust, and Purchaser may each be referred to herein as a "**party**" and collectively as the "**parties**."

This Agreement is being entered into in connection with the Asset APA, dated of even date herewith among Sellers and Purchaser (the "APA"). Capitalized terms used in this Agreement but not defined herein shall have the meanings given to them in the APA. The <u>partiesParties</u> have agreed to provide or cause one or more of their respective direct or indirect subsidiaries or affiliates to provide certain services as set forth below. The <u>partyParty</u> providing services hereunder (together with those of its direct and indirect subsidiaries and affiliates providing services hereunder) is referred to herein as the "Service Provider" and the <u>partyParty</u> receiving services hereunder (together with those of its direct and indirect subsidiaries and affiliates receiving services hereunder) is referred to herein as the "Service Recipient". The <u>partiesParties</u> agree that this Agreement will be amended to include the Liquidating Trust (as defined below) after its creation, pursuant to the effective date set forth in the plan of liquidation (the "Plan") to be confirmed by the Bankruptcy Court.

#### RECITALS

A. On August 31, 2018, the Sellers and certain of their affiliates each filed a voluntary petition for relief (collectively, the "**Bankruptcy Cases**") under chapter 11 of title 11 of the United State Code (the "**Bankruptcy Code**").

B. The Bankruptcy Cases are jointly administered under Lead Case No. 18-20151 and are currently pending in the Bankruptcy Court for the Central District of California in Los Angeles (the "**Bankruptcy Court**").

C. Purchaser and the Sellers entered into the APA, which provides for the sale of the assets of the Sellers (collectively, the "Assets").

D. Sellers will file a motion with the Bankruptcy Court to approve the APA and upon approval, the Bankruptcy Court will enter a sale order (the "**Sale Order**").

E. Sellers and certain of their affiliates will file a chapter 11 Plan, pursuant to which a liquidating trust (the "**Liquidating Trust**") will be created on the effective date of the Plan after confirmation thereof.

F. The primary purpose of the Liquidating Trust is to, among other things, liquidate and distribute assets, which includes assets of the Sellers not sold to Purchaser pursuant to the APA.

G. Following the Closing Date, the Sellers and the Liquidating Trust need access to certain assets, systems, facilities, equipment, and personnel of Purchaser in connection with the Sellers' wind-down of the Businesses, the completion of the Bankruptcy Cases, and the dissolution of the Sellers, and Purchaser agrees to provide such access and services in connection therewith.

### AGREEMENT

**NOW, THEREFORE,** in consideration of the promises and the mutual covenants contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the <u>partiesParties</u> hereby agree as follows:

## ARTICLE I SERVICES

Section 1.1 <u>Services to be Provided</u>.

(a) <u>Services to be Provided by the Sellers and/or the Liquidating Trust</u>. Subject to the terms and conditions of this Agreement, the Sellers and/or the Liquidating Trust, as the Service Provider, shall provide or cause to be provided to Purchaser, as the Service Recipient, those services in support of the business of Purchaser on such terms (including payment) as may be mutually agreed upon in writing from time to time by the Sellers and/or the Liquidating Trust and Purchaser (the "Seller Services").

(b) <u>Services to be Provided by Purchaser</u>. Subject to the terms and conditions of this Agreement, Purchaser, as the Service Provider, shall provide or cause to be provided to the Sellers and the Liquidating Trust, as the Service Recipient, those services and personnel in support of the business of the Sellers and/or the wind-down and liquidation of the Sellers set forth on <u>Schedule A</u> attached hereto (the "**Purchaser Services**").

Section 1.2 Access to and Use of Assets, Systems, Facilities, and Equipment.

(a) Access. Beginning on the Effective Time and continuing through the termination or expiration of this Agreement in accordance with its terms, Purchaser hereby grants to the Sellers and the Liquidating Trust, reasonable access to and the right to use the hospital assets, systems, facilities, and equipment to the extent reasonably required in connection with the Sellers' wind-down of the Businesses, the completion of the Bankruptcy Cases, and the dissolution of the Sellers. If any third party to any lease, license or other agreement requires Sellers to sign any documents in order for Sellers to be granted access to, and the right to use, any hospital asset, system, facility or equipment under such lease, license or other agreement, Sellers agree to execute and deliver any such documents; provided, that such documents, leases, licenses or other agreements do not require Sellers to pay any money or take any action, not required by the Sale Order; and provided, further, that if Sellers do not sign such requisite documents, leases, licenses or other agreements, then Purchaser has no obligation to provide Sellers with access and

the right to use the associated hospital assets, systems, facilities and equipment provided by such third party; and provided, further, that Purchaser agrees to indemnify, defend and hold harmless each of the Sellers in connection with any obligations under any such documents, leases, licenses or other agreements other than those obligations of Sellers thereunder which are required by the Sale Order.

**(b)** Office Space. In order to enable the provision of the Seller Services, Purchaser hereby grants to Sellers' and their affiliates' employees and any third-party service providers or subcontractors the right to use and occupy the portion of St. Francis Medical Center identified on Schedule B attached hereto (the "Premises") at no cost for use exclusively as office space pursuant to the terms and conditions hereof, and until the termination of this Agreement pursuant to Article IV herein. Sellers shall keep the Premises neat, clean and sanitary and dispose of all garbage. Sellers may terminate its right to occupy the Premises at any time upon five (5) days prior written notice to Purchaser. Upon termination of Sellers' right to occupy the Premises, Sellers shall vacate the Premises in broom-clean condition. Purchaser shall make the Premises available to Sellers 24 hours a day, 7 days a week. Purchaser may access the Premises for purposes of making repairs or improvement or other reasonable purposes upon (except in the case of an emergency) reasonable prior written notice to Sellers; however, Purchaser shall at all times use its commercially reasonable efforts to prevent interference with or interruption of the Sellers' business activities on the Premises. The Sellers use of the Premises shall additionally be subject to the following terms and conditions:

(i) Purchaser shall have the right, on reasonable advance notice (which shall not be less than three (3) business days) to Sellers, to relocate the space covered by the preceding license to similar space in the same area as the original designated Premises, provided the replacement space is reasonably suitable for the same uses as contemplated herein. In the event of any such relocation, references herein to the Premises shall constitute references to such relocated space.

(ii) Sellers shall not make any material alterations, decorations, additions, or improvements in or to the Premises or the Purchaser's other property without the prior written consent of Purchaser, which may be given or withheld in Purchaser's sole discretion.

(iii) Sellers shall not do, or permit their representatives to do, anything that would: (a) unreasonably obstruct or materially interfere with the lawful operation of the Premises or other portions of the Purchaser's property by Purchaser and other tenants of the Purchaser's property; or (b) use the Premises for any illegal or unlawful purpose, including without limitation any violations of applicable laws or regulations relating to or regulating hazardous substances.

(iv) The license, occupancy and access rights granted to Sellers under this Agreement shall be effective notwithstanding any ground lease, mortgage, deed of trust, or any other hypothecation or security placed at any time upon the Premises or any of Purchaser's property of which the Premises is a part, and notwithstanding any change in ownership (including any sale and/or leaseback involving any property of which the Premises is a part).

**Section 1.3** <u>Subcontracting</u>. The Service Provider may, in its sole discretion and without any written notice to the Service Recipient engage one or more parties (including third parties or

I

### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 90 of 111

affiliates of the Service Provider) to provide the Services, whether in whole or in part or subcomponent. Without limiting the preceding, Purchaser will not be required to use the Sellers existing IT and similar systems, but shall have the right to replace IT and other systems, used to provide any of the Purchaser Services provided that the new or replacement systems provide equivalent functionality. Sellers and Purchaser agree to fully cooperate with each other, at Purchaser's sole cost and expense, in connection with any data or other transition steps required in connection moving any of the functions covered by the Purchaser Services to a new vendor or system.

**Section 1.4** <u>Cooperation; Access</u>. Each <u>partyParty</u> shall, at reasonable times under the circumstances, make available to any other <u>partyParty</u> properly authorized personnel for the purpose of consultation and decisions in connection with the Services.

## Section 1.5 Certain Qualifications.

(a) <u>Provided AS-IS</u>. By accepting the use to the Premises and other access rights provided herein, Sellers conclusively shall be deemed, subject to the representations, warranties and obligations expressly set forth in this Agreement, to have (i) accepted the Premises and access to other property of Purchaser "AS IS" and without express or implied warranty; and (ii) is fully aware of and has approved the Premises in its condition existing as of such date.

(b) <u>Limited Representation</u>. Except for the representations, warranties and obligations expressly set forth in this Agreement, Purchaser makes no representations and warranties of any kind, implied or expressed, with respect to the Purchaser Services, including, without limitation, no warranties of merchantability or fitness for a particular purpose, which are specifically disclaimed.

(c) <u>Privacy and Confidentiality Protections</u>. All access provided to Sellers, to the Premises or otherwise as provided herein, shall be permitted only in a manner which protects patient privacy and the confidentiality of patient information under applicable Privacy Laws.

(d) <u>Limitation on Liability</u>. In no event shall any <u>partyParty</u> have any liability under any provision of this Agreement for any punitive, incidental, consequential, special or indirect damages, whether based on statute, contract, tort or otherwise, and whether or not arising from any other <u>partyParty</u>'s sole, joint, or concurrent negligence, strict liability, criminal liability or other fault.

### ARTICLE II LICENSE GRANT

**Section 2.1** <u>License Grant</u>. Beginning on the Effective Time and continuing through the later of the termination or expiration of this Agreement in accordance with its term or the dissolution of the Sellers and their Businesses, the wind-down of the Liquidating Trust and the completion of the Bankruptcy Cases, Purchaser grants to the Sellers and the Liquidating Trust a royalty free, license to use the name "St. Francis Medical Center," including any trademarks, service marks, trademark and service mark registrations and registration applications, trade name registrations, logos, domain names, trade dress, copyrights, copyright registrations, website

### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 91 of 111

content, know-how, trade secrets, related to the use of the preceding name, and the corporate or company names of the Sellers and the name of the Hospital, together with all rights to sue and recover damages for infringement, dilution, misappropriation or other violation or conflict associated with any of the foregoing, in all respects solely as reasonably needed in connection with the wind-down of the Businesses, the completion of the Bankruptcy Cases, and the dissolution of the Sellers. Sellers and the Liquidating Trust shall not assign, transfer or hypothecate any of the rights subject to, and pursuant to, the preceding license and shall not use any such license rights in any manner which could adversely affect Purchaser's intellectual property rights with respect to the items which are the subject of such license.

# ARTICLE III PAYMENT

**Section 3.1** <u>Service Fees</u>. The fee, rate or amount to be charged to the Service Recipient for the Services (the "Service Fees") shall be as set forth on <u>Schedule A</u> hereto for the Purchaser Services and as agreed upon in writing from time to time by Sellers and/or the Liquidating Trust and Purchaser for the Purchaser Services. The Service Provider shall issue invoices on a monthly basis to the Service Recipient for the Services, with a statement of such Services and reasonable supporting detail therefor. The Service Recipient shall pay the undisputed portion of such invoices within thirty (30) days of receipt of such invoices. Any dispute relating to the Services shall be resolved in accordance with the dispute resolution procedures set forth in this Agreement.

# ARTICLE IV TERM AND TERMINATION

Section 4.1 Term. Unless a Service is earlier terminated by the Service Recipient in accordance with Section 4.2(b), the Service Provider shall provide each Service to the Service Recipient for the period beginning at the Effective Time (as defined under the APA) and continuing through and including the applicable dates set forth in: (i) Schedule A (with respect to the Purchaser Services); and (ii) any written agreement between Sellers and/or the Liquidating Trust and Purchaser (with respect to the Seller Services) (the "Applicable Term"); provided, that in the event a particular Service provided by Purchaser hereunder is required by one or more Sellers to allow such Seller(s) to comply with applicable laws, rules or regulations, the Applicable Term shall be extended for so long as such Service is required to allow such Seller(s) to comply with such laws, rules and regulations; provided, however, that notwithstanding any other term herein any partyParty may, in any case, terminate this Agreement by written notice to any other partyParty at any time on and after the end of the 18<sup>th</sup> calendar month following the Licensure Date (as defined in the APA). In addition, upon termination or expiration of this Agreement, the parties Parties may mutually agree in writing to a tail period for any portion of the Services, and the terms of this Agreement shall continue to apply to such ongoing Services until performed in full.

# Section 4.2 <u>Termination</u>.

(a) <u>Termination for Material Breach</u>. This Agreement may be terminated by the Sellers and/or the Liquidating Trust on the one hand, or Purchaser on the other hand, upon

### Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 92 of 111

written notice to any other <u>partyParty</u> in the event that any other <u>partyParty</u> materially breaches any provision of this Agreement and fails to remedy such breach within thirty (30) days of its receipt of such written notice, except to the extent such remedy is prevented or not permitted by the Bankruptcy Court; and provided, further, however, that this Agreement may only be terminated under this <u>Section 4.2(a)</u> following approval of such termination by the Bankruptcy Court.

(b) Early Termination. The Service Recipient may terminate any particular Service early by providing not less than thirty (30) days prior written notice to the Service Provider of its intent to terminate early such particular Service. Any such partial termination notice delivered shall specify in detail (i) the Service or Services to be terminated and (ii) the effective date(s) of such termination (subject to the foregoing sentence). If any such termination of a Service prior to the expiration of its Applicable Term requires any out-of-pocket fee, cost or expense of a similar nature payable to a third party (a "Termination Fee"), the Service Provider shall have no obligation to terminate such Service prior to the expiration of its Applicable Term unless the Service Recipient approves in advance such Termination Fee in writing (a "Termination Fee Approval"). If the Termination Fee Approval is granted, the Termination Fee shall be billed to and payable by the Service Recipient.

(c) <u>Continued Access</u>. Notwithstanding the termination of any particular Services hereunder, Service Recipient shall continue to have the right, upon reasonable advance notice and consistent with the other provisions herein, to reasonable access data and Service Provider, personnel relating to such Services for reasonable purposes (including compliance with applicable law).

**Section 4.3** Effect of Termination. The termination of this Agreement pursuant to <u>Section 4.2</u> shall not relieve any <u>partyParty</u> from liability for any willful and material breach of this Agreement prior to such termination. Further, the provisions of <u>Article II</u>, <u>Article III</u>, this <u>Section 4.3</u>, <u>Article V</u> and <u>Article VI</u> shall survive any termination of this Agreement pursuant to <u>Section 4.2</u>.

# ARTICLE V COORDINATION AND OTHER COVENANTS

**Section 5.1** <u>Coordination</u>. The <u>partiesParties</u> hereby appoint as of the Effective Time the following representatives responsible for coordinating and managing the provision and receipt of the Services, which representatives shall have authority to act on the applicable <u>partyParty</u>'s behalf with respect to matters relating to this Agreement:

Sellers' Representative:

Address:

Telephone:

Facsimile:

Email:

Liquidating Trust Representative:
Address:
Telephone:
Facsimile:
Email:
Purchaser's Representative:
Address:
Telephone:
Facsimile:
Email:

In the event that the Sellers, the Liquidating Trust or Purchaser elect, in their sole discretion, to change their respective representative appointed for purposes of this <u>Section 5.1</u>, the Sellers, the Liquidating Trust or Purchaser (as the case may be) shall deliver written notice thereof to the other promptly following the effective date of appointment of the name and contact information of such new representative. The representatives shall work in good faith to address and attempt to resolve any dispute arising out of or relating to this Agreement or the breach, termination or validity thereof, whether sounding in contract or tort and whether arising during or after termination of this Agreement, including any pricing and other Service related matters.

Section 5.2 General Cooperation and Turnover Obligations. The parties Parties shall cooperate to ensure that any and all payments that constitute Excluded Assets (as defined in the APA), shall be paid to and received by Sellers, with any payments that constitute Assets (as defined in the APA) transferred to Purchaser transferred to Purchaser pursuant to Section 1.7 of the APA or that otherwise arise from services rendered by Purchaser on or after the Licensure Date be paid to and received by Purchaser. In this regard, for a period of onetwo (12) yearyears after the Effective Time ("Turnover Period"), the parties Parties shall, within fiveten (510) business days, copy and send to any other partyParty copies (either in hard copy or via electronic file) of all remittance advices for all deposits to all Lockboxes or other bank accounts for the Accounts Receivables (as defined in the APA), from whatever payor or source of funds, that are received on and after the Effective Time. In the event that Assets are deposited to a bank account of Sellers which is not automatically swept or transferred to Purchaser, then Sellers, within fiveten (510) business days of notice of the receipt of such Assets, shall turnover and pay Purchaser said funds. In the event that a deposit representing payment of any Excluded Assets is received by Purchaser, then Purchaser, within five (5) business days of notice of the receipt of funds representing any Excluded Assets, shall turnover and pay Sellers such funds. Each partyParty, moreover, shall have the right, within three (3) months after the expiration of the

Turnover Period, to audit by an independent and competent auditor, at the requesting <u>partyParty</u>'s sole expense, of the bank records and remittance advices of any other <u>partyParty</u>. Thereafter, upon the findings of the auditor that there has either been an overpayment or an underpayment of funds due, the <u>partyParty</u> owing funds shall, within five (5) business days, make a payment of such funds to the <u>partyParty</u> to whom they are owed.

**Section 5.3** <u>HIPAA Compliance</u>. The <u>partiesParties</u> agrees to take such steps as are necessary to ensure compliance with the Health Insurance Portability and Accountability Act of 1996 ("**HIPAA**"), the California Confidentiality of Medical Information Act, and other applicable federal and state privacy laws and regulations (collectively, the "**Privacy Laws**") with respect to the Hospital and its operations, and the performance by the <u>partiesParties</u> of the services and obligations provide herein.

## Section 5.4 <u>Confidentiality</u>.

During the term of this Agreement and thereafter, the parties heretoParties shall, **(a)** and shall instruct their respective Representatives (as defined below) to, maintain in confidence and not disclose any other partyParty's financial, technical, sales, marketing, development, personnel, and other information, records, or data, including, without limitation, customer lists, supplier lists, trade secrets, designs, product formulations, product specifications or any other proprietary or confidential information, however recorded or preserved, whether written or oral (any such information, "Confidential Information"). Each partyParty hereto shall use the same degree of care, but no less than reasonable care, to protect any other partyParty's Confidential Information as it uses to protect its own Confidential Information of like nature. Unless otherwise authorized in any other agreement between the parties Parties, any partyParty receiving any Confidential Information of any other partyParty (the "Receiving Party") may use Confidential Information only for the purposes of fulfilling its obligations under this Agreement (the "Permitted Purpose"). Any Receiving Party may disclose such Confidential Information only to its directors/managing members, managers, officers, employees, consultants, financial advisors, prospective or existing lenders, counsel, accountants and other agents of such Receiving Party (and, in addition, Sellers may disclose such Confidential Information to the creditors committee in the Bankruptcy Cases and its consultants, financial advisors, counsel, accountants and other agents, the Liquidating Trust and its trustee, consultants, financial advisors, counsel, accountants and other agents and any advisory board members and their consultants, financial advisors, counsel, accountants and other agents) ("Representatives") who have a need to know such information for the Permitted Purpose and who have been advised of the terms of this Section 5.5 and the Receiving Party shall be liable for any breach of these confidentiality provisions by such persons; provided, however, that any Receiving Party may disclose such Confidential Information to the extent such Confidential Information is required to be disclosed by a governmental agency or court order ("Governmental Order"), in which case the Receiving Party shall promptly notify, to the extent possible, the disclosing partyParty (the "Disclosing Party"), and take reasonable steps to assist in contesting such Governmental Order or in protecting the Disclosing Party's rights prior to disclosure, and in which case the Receiving Party shall only disclose such Confidential Information that it is advised by its counsel in writing that it is legally bound to disclose under such Governmental Order.

## Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 95 of 111

(b) Notwithstanding the foregoing, "Confidential Information" shall not include any information that the Receiving Party can demonstrate: (i) was publicly known at the time of disclosure to it, or has become publicly known through no act of the Receiving Party or its Representatives in breach of this <u>Section 5.5</u>; (ii) was rightfully received from a third party without a duty of confidentiality; or (iii) was developed by it independently without any reliance on the Confidential Information.

(c) Upon demand by the Disclosing Party at any time, or upon expiration or termination of this Agreement with respect to any Service, the Receiving Party agrees promptly to return or destroy, at the Receiving Party's option, all Confidential Information. If such Confidential Information is destroyed, an authorized officer of the Receiving Party shall certify to such destruction in writing.

## ARTICLE VI MISCELLANEOUS

Section 6.1 <u>Relationship of the Parties</u>.

(a) The <u>partiesParties</u> and their respective representatives shall be deemed independent contractors for all purposes under this Agreement.

(b) This Agreement shall not be deemed or construed to create the relationship of employer or employee, partnership or any type of joint venture relationship among the <u>parties</u>.

(c) The <u>partiesParties</u> acknowledge and agree that the <u>partiesParties</u> are not providing legal, accounting or tax advice under this Agreement. The <u>partiesParties</u> further acknowledge and agree that no fiduciary or other similar relationship is being created among the <u>partiesParties</u> relating to the Services or otherwise under this Agreement.

(d) Except as expressly set forth herein, neither the Sellers, the Liquidating Trust, nor any of their respective representatives or affiliates, on the one hand, nor Purchaser or any of its representatives or affiliates, on the other hand, shall have the authority to contract for or assume obligations of any nature in the name of any other <u>partyParty</u>, as the case may be, without the other's prior written consent.

**Section 6.2** <u>No Set-off</u>. Neither the Sellers, the Liquidating Trust, nor any of their respective representatives or affiliates, on the one hand, nor Purchaser nor any of its representatives or affiliates, on the other hand, shall have any set-off or other similar rights pursuant to this Agreement.

**Section 6.3** <u>Notices</u>. All notices, requests, claims, demands or other communications hereunder shall be deemed to have been duly given and made if in writing and (a) at the time personally delivered if served by personal delivery upon the <u>partyPartyParty</u> for whom it is intended, (b) at the time received if delivered by registered or certified mail (postage prepaid, return receipt requested) or by a national courier service (delivery of which is confirmed), or (c) upon confirmation if sent by facsimile; in each case to the <u>partyParty</u> at the address set forth in <u>Section</u>

<u>5.1</u>, or such other address as may be designated in writing hereafter, in the same manner, by such <u>partyParty</u>.

**Section 6.4** <u>Amendment, Modification and Waiver</u>. Any provision of this Agreement may be amended, modified or waived if, and only if, such amendment, modification or waiver is in writing and signed, in the case of an amendment, by the <u>partiesParties</u>, or in the case of a waiver, by the <u>partyParty</u> against whom the waiver is to be effective. No failure or delay by any <u>partyParty</u> in exercising any right, power or privilege hereunder shall operate as a waiver thereof nor shall any single or partial exercise thereof preclude any other or further exercise thereof or the exercise of any other right, power or privilege.

**Section 6.5** <u>Assignment</u>. Neither this Agreement nor any of the rights, interests or obligations under it may be directly or indirectly assigned, delegated, hypothecated, sublicensed, subleased or transferred by any of the <u>partiesParties</u>, in whole or in part, to any other person without the prior written consent of the non-assigning <u>partyParty</u>, and any attempted or purported assignment in violation of this <u>Section 6.5</u> will be null and void; provided, that Sellers may assign their rights hereunder to the Liquidating Trust. Subject to the preceding sentence, this Agreement will be binding upon, inure to the benefit of, and be enforceable by the <u>partiesParties</u> and their respective successors and permitted assigns. Notwithstanding the preceding, however, Purchaser may assign its rights or obligations hereunder to its any of its affiliates and may subcontract with third parties to provide any of the Purchaser Services as provided elsewhere in this Agreement. provided that any such assignment or subcontracting shall not relieve Purchaser of or reduce Purchaser's obligations under this Agreement.

**Section 6.6** <u>Entire Agreement</u>. This Agreement constitutes the entire agreement among the <u>partiesParties</u> with respect to the subject matter hereof and supersedes all prior agreements and understandings, oral or written, with respect to such matters.

**Section 6.7** <u>Third Party Beneficiaries</u>. Nothing expressed or implied in this Agreement is intended to confer any rights, benefits, remedies, obligations or liabilities upon any person other than the <u>parties Parties</u> and their respective successors and permitted assigns.

**Section 6.8** Expenses. In the event of litigation arising from or relating to this Agreement, the non-prevailing partyParty as determined by a court of competent jurisdiction in a final non-appealable order shall be liable and pay to the prevailing partyParty the reasonable attorney's fees and expenses incurred in connection with such litigation.

**Section 6.9** <u>Governing Law</u>. This Agreement shall be construed, performed, and enforced in accordance with, and governed by, the laws of the State of California (without giving effect to the principles of conflicts of laws thereof), except to the extent that the laws of such State are superseded by the Bankruptcy Code or other applicable federal law.

Section 6.10 WAIVER OF JURY TRIAL. TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT, OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (B) EACH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (C) EACH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (D) EACH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 6.10. THIS IS AN AFFIRMATIVE WAIVER OF THE PARTIES' RIGHTS TO A JURY TRIAL UNDER CALIFORNIA LAW, Cal. C. Civ. Pro. Sec 631. BY SIGNING BELOW ON THE SIGNATURE LINES, EACH PARTY IS EXPLICITLY WAIVING JURY TRIAL AND AUTHORIZING ANY AND ALL PARTIES TO FILE THIS WAIVER WITH ANY COURT AS THE WAIVER REQUIRED UNDER Cal. C. Civ. Proc. Sec. 631(f)(2).

Section 6.11 <u>Jurisdiction; Mediation</u>. The <u>parties</u> agree that the Bankruptcy Court shall have exclusive personal and subject matter jurisdiction over any action or proceeding with respect to any claim arising out of or related to this Agreement, and any of the documents or transactions contained in or contemplated by this Agreement; provided, however, that the parties Parties agree that the United States District Court for the Central District of California (together with the Bankruptcy Court, the "Chosen Courts") shall have exclusive jurisdiction over such claim if (i) the Bankruptcy Cases are closed and if the Bankruptcy Cases are not reopened to adjudicate such claim after request by the partyParty bringing such claim or (ii) the Bankruptcy Court determines that it does not have jurisdiction over such claim. Solely in connection with claims arising under this Agreement, or any of the documents or transactions contemplated hereby, the parties Parties (a) irrevocably submit to the exclusive jurisdiction of the Chosen Courts, (b) waive any objection to laying venue in any such action or proceeding in the Chosen Courts, (c) waive any objection that the Chosen Courts are an inconvenient forum or do not have jurisdiction over any partyParty hereto, and (d) agree that service of process upon such partyParty in any such action or proceeding shall be effective if notice is given in accordance with Section 6.3 hereof.

**Section 6.12** <u>Mediation</u>. If a dispute arises out of or relates to this contract, or the breach thereof, and if the dispute cannot be settled through negotiation, the <u>partiesParties</u> agree first to try in good faith to settle the dispute by mediation administered by the American Arbitration Association under its Commercial Mediation Procedures before resorting to litigation in accordance with the terms of this Agreement. Any such mediation shall take place in Los Angeles, California and be subject to the governing law as set forth in this Agreement.

**Section 6.13** <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, each of which will be deemed to constitute an original, but all of which shall constitute one and the same agreement, and may be delivered by facsimile or other electronic means intended to preserve the original graphic or pictorial appearance of a document.

## Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 98 of 111

**Section 6.14** <u>Ownership of Intellectual Property</u>. Subject to the provisions of the APA, each of the <u>parties</u> shall retain all right, title and interest in and to their respective intellectual property and data and other proprietary and other assets.

**Section 6.15** <u>Service Standards</u>. Service Provider shall use commercially reasonable efforts to provide the Services to be provided by it in accordance with all applicable laws, rules and regulations and in a professional, competent, workman-like and timely manner.

**Section 6.16** Severability. The provisions of this Agreement shall be deemed severable and the invalidity or unenforceability of any provision shall not affect the validity or enforceability of the other provisions hereof. If any provision of this Agreement, or the application thereof to any circumstance, is found by a court or other governmental authority of competent jurisdiction to be invalid or unenforceable, the remainder of this Agreement and the application of such provision to other circumstances shall not be affected by such invalidity or unenforceability, nor shall such invalidity or unenforceability affect the validity or enforceability of such provision, or the application thereof, in any other jurisdiction. If any provision of this Agreement is so broad as to be unenforceable, the provision shall be interpreted to be only so broad as would be enforceable.

**Section 6.17** Force Majeure. No partyParty shall be liable for nonperformance or late performance of any of its obligations under this Agreement to the extent and for such periods of time as such nonperformance, or late performance is due to reasons outside such party's controlParty's control (excluding any arising from the SARS-CoV-2 virus or mutations therefrom or in connection with the disease COVID-19), including acts of God, war (declared or undeclared), riots, revolutions, fire, floods, explosions, sabotage, nuclear incidents, lightning, weather, earthquakes, storms, sinkholes, epidemics, failure of utilities, (or similar nonperformance or late performance, the Service Provider shall use commercially reasonable efforts to restore or resume the Services within a reasonable period of time.

**Section 6.18** <u>Fair Meaning</u>. This Agreement shall be construed according to its fair meaning and as if prepared by all <u>parties heretoParties</u>.

**IN WITNESS WHEREOF,** the <u>parties heretoParties</u> have executed this Agreement as of the Signing Date.

NAME	VERITY HEALTH SYSTEM OF
By:	CALIFORNIA, INC.
-	By:
Name:	
	Name:
Title:	
	Title:
NAME	
Den	VERITY HOLDINGS, LLC
By:	

## Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 99 of 111

Name:	By:
Title:	Name:
NAME	Title:
Ву:	ST. FRANCIS MEDICAL CENTER
Name:	Ву:
Title:	Name:
	Title:

## Schedule A to TSA

## Services to be Provided by Purchaser to Sellers

I. Phase One: [from closing date to 6-9 months thereafter]

A. Description of Services: Purchaser will assist in activities related to the Debtors' efforts in support of its winding down activities, including accounting for monthly transactions, assisting in materials management and purchasing, facilitating voucher and payment of accounts payable and financial reporting. Purchaser will assist Debtors' in additional winding down activities, including facilitating the Debtors' efforts in claims reconciliation, contracts rejections/assumptions, preparing W2s and 1099s, and preparing tax returns.

- B. Accounting TBD
- C. Accounts Payable / Supply Chain TBD
- D. Human Resources / Payroll TBD

E. Information Technology: Provide application and infrastructure support; maintain licensing; maintain user access and perform required application and operating system upgrades as needed to perform services for all applications. List of applications: - TBD

II. Phase Two: [from end of Phase I to two years thereafter]

A. Description of Services: Purchaser will continue to archive and provide access to select data.

- B. Accounting / Finance TBD
- C. Human Resources / Payroll TBD

D. Information Technology: Provide application and infrastructure support; maintain licensing; maintain user access and perform required application and operating system upgrades as needed to perform services for all applications. List of applications: - TBD

## Schedule B to TSA

Office Space for Use by Sellers

## Exhibit 5.9.35.9(c) to APA

## **Accepted Conditions (Attorney General)**

See attached.

# Conditions to the Sale of St. Francis Medical Center<sup>32</sup><sub>=</sub> and Approval of the Asset Purchase Agreement by and among Verity Health System of California, Inc., Verity Holdings, LLC, St. Francis Medical Center and Prime Healthcare Services, Inc. ("Purchaser")<sup>1</sup>

These Conditions shall be legally binding on Verity Health System of California, Inc., a California nonprofit public benefit corporation, Verity Holdings, LLC, a California limited liability company, St. Francis Medical Center, a California nonprofit public benefit corporation, and St. Francis Medical Center Foundation, a nonprofit public benefit corporation and Purchaser, a Delaware corporation, any other subsidiary, parent, general partner, limited partner, member, affiliate, successor, successor in interest, assignee, or person or entity serving in a similar capacity of any of the above-listed entities including, but not limited to, any entity succeeding thereto as a result of consolidation, affiliation, merger, or acquisition of all or substantially all of the real property or operating assets of St. Francis Medical Center, is located, any and all current and future owners, lessees, licensees, or operators of St. Francis Medical Center, and any and all current and future lessees and owners of the real property on which St. Francis Medical Center is located.

The transaction conditionally approved by the Attorney General consists of \_\_\_\_\_\_.

All the entities listed in Condition I, and any other parties referenced in the above agreements shall fulfill the terms of these agreements or documents and shall notify and obtain the Attorney General's approval in writing of any proposed modification or rescission of any of the terms of these agreements or documents. Such notifications shall be provided at least sixty days prior to their effective date in order to allow the Attorney General to consider whether they affect the factors set forth in Corporations Code section 5917 and obtain the Attorney General's approval.

For ten years from the closing date of the Asset Purchase Agreement, Verity Health System of California, Inc., Verity Holdings, LLC, Purchaser, and all future owners, managers, lessees, licensees, or operators of St. Francis Medical Center shall be required to provide written notice to the Attorney General sixty days prior to entering into any agreement or transaction to do any of the following:

Sell, transfer, lease, exchange, option, convey, manage, or otherwise dispose of St. Francis Medical Center;

Transfer control, responsibility, management, or governance of St. Francis Medical Center. The substitution, merger or addition of a new member or members of the governing body of Purchaser that transfers the control of, responsibility for or governance of St. Francis Medical Center, shall be deemed a transfer for purposes of this Condition. The substitution or addition of one or more members of the governing body of Purchaser, or any arrangement, written or oral, that would transfer voting control of the members of the governing body of Purchaser shall also be deemed a transfer for purposes of this Condition.

US\_Active\114188900\V-6US\_Active\114483166\V-14

<sup>&</sup>lt;sup>32</sup> Throughout this document, the term "St. Francis Medical Center" shall mean the general acute care hospital located at 3630 East Imperial Highway, Lynwood, CA 90262, and any other clinics, laboratories, units, services, or beds included on the license issued to St. Francis Medical Center by the California Department of Public Health, effective January 1, 2019, unless otherwise indicated.

For ten years from the closing date of the Asset Purchase Agreement, St. Francis Medical Center shall be operated and maintained as a licensed general acute care hospital (as defined in California Health and Safety Code Section 1250) and shall maintain and provide 24-hour emergency and trauma medical services at no less than current<sup>43</sup><sub>=</sub> licensure and designation with the same types and/or levels of services, including the following:

46 emergency treatment stations at a minimum;
Designation as a Level II Trauma Center;
Designation as a 5150 Receiving Facility, as defined by the Welfare and Institutions Code, section 5150, for behavioral health patients under involuntary evaluation;
Psychiatric evaluation team;
Designation as an Emergency Department Approved for Pediatrics (EDAP);
Designation as a Paramedic Base Station; and Certification as a Primary Stroke Center.

Purchaser must give one-year advance written notice to the Los Angeles County Emergency Medical Services Agency and the California Department of Public Health if St. Francis Medical Center seeks to reduce trauma or trauma-related care services or stop operating the Level II Trauma Center after ten years from the closing date of the Asset Purchase Agreement.

For at least ten years from the closing date of the Asset Purchase Agreement, St. Francis Medical shall maintain on-call coverage contracts and/or comparable coverage arrangements with physicians that are necessary to retain its qualification as a Level II trauma center. The following on-call coverage contracts and/or comparable coverage arrangements are required to retain St. Francis Medical Center's status as a Level II trauma center:

Neurology; Obstetrics/gynecology; Ophthalmology; Oral or maxillofacial or head and neck; Orthopedic Plastic surgery; Reimplantation/microsurgery capability (this surgical service may be provided through a written transfer agreement); and Urology.

For at least ten years from the closing date of the Asset Purchase Agreement, St. Francis Medical Center shall maintain the following services at current licensure, types, and/or levels of services:

- Cardiac services, including three cardiac catheterization labs and the designation as a STEMI Receiving Center;
- Critical care services, including a minimum of 36 intensive care unit beds or 24 intensive care beds and 12 definitive observation beds;
- Neonatal intensive care services, including a minimum of 29 neonatal intensive care beds, and at minimum, maintaining a Level II NICU;
- Women's health services, including women's imaging services;

 $<sup>\</sup>frac{43}{=}$  The term "current" or "currently" throughout this document means as of January 1, 2019.

Pediatric services, including a designated area with at least five general acute care beds for pediatric services;

Orthopedic and rehabilitation services;

Wound care services;

Behavioral health services, including a minimum of 40 distinct part inpatient acute psychiatric beds; and

Perinatal services, including a minimum of 50 perinatal beds.

Purchaser shall not place all or any portion of its above-listed licensed-bed capacity or services in voluntary suspension or surrender its license for any of these beds or services.

For at least ten years from the closing date of the Asset Purchase Agreement, St. Francis Medical Center shall maintain the same types and/or levels of women's healthcare services currently provided at the location below or a location within three miles of St. Francis Medical Center:

Family Life Center at St. Francis Medical Center, located at 3630 E Imperial Highway, Lynwood, California.

For at least five years from the closing date of the Asset Purchase Agreement, Purchaser shall either: (1) operate clinics (listed below) with the same number of physicians and mid-level provider full-time equivalents in the same or similar alignment structures, or (2) sell the clinics (listed below) with the same number of physician and mid-level provider full-time equivalents and require the purchaser(s) to maintain such services for 5 years from the closing date of the Asset Purchase Agreement and to participate in the Medi-Cal and Medicare programs as required in the conditions herein, or (3) ensure that a third party is operating the clinics (listed below) with the same number of physician and mid-level provider full-time equivalents and require the third party to maintain such services for 5 years from the closing date of the Asset Purchase Agreement and to participate in the Medi-Cal and Medicare programs as required to maintain such services for 5 years from the closing date of the Asset Purchase Agreement and to participate in the Medi-Cal and Medicare programs as required to maintain such services for 5 years from the closing date of the Asset Purchase Agreement and to participate in the Medi-Cal and Medicare programs as required in the conditions herein. For any of these options, each clinic can be moved to a different location within a three-mile radius of each clinic's current location, and St. Francis Medical Center can utilize an alternative structure in providing such services. The following clinic is subject to this condition:

Orthopedic services at 3628 E. Imperial Highway, Suite 300, Lynwood, California.

For ten years from the closing date of the Asset Purchase Agreement, Purchaser shall:

Be certified to participate in the Medi-Cal program at St. Francis Medical Center;

Maintain and have Medi-Cal Managed Care contracts with the below listed Medi-Cal Managed Care Plans to provide the same types and levels of emergency and non-emergency services at St. Francis Medical Center to Medi-Cal beneficiaries (both Traditional Medi-Cal and Medi-Cal Managed Care) as required in these Conditions, on the same terms and conditions as other similarly situated hospitals offering substantially the same services, without any loss, interruption of service or diminution in quality, or gap in contracted hospital coverage, unless the contract is terminated for cause or not extended or renewed by the Medi-Cal Managed Care Plan:

Local Initiative: L.A. Care Health Plan or its successor; and

Commercial Plan: Health Net Community Solutions, Inc. or its successor.

If Purchaser questions whether it is being reimbursed on the same terms and conditions as other similarly situated hospitals offering substantially the same services, it shall notify the Attorney General's Office with at least 120 days' notice prior to taking any action that would effectuate any loss, interruption of service or diminution in quality, or gap in contracted hospital coverage or prior to giving any required notice of taking such action.

Be certified to participate in the Medicare program by maintaining a Medicare Provider Number to provide the same types and levels of emergency and non-emergency services at St. Francis Medical Center to Medicare beneficiaries (both Traditional Medicare and Medicare Managed Care), on the same terms and conditions as other similarly situated hospitals, as required in these Conditions.

For six fiscal years from the closing date of the Asset Purchase Agreement, Purchaser shall provide an annual amount of Charity Care (as defined below) at St. Francis Medical Center equal to or greater than \$8,000,000 (the Minimum Charity Care Amount). For purposes hereof, the term "charity care" shall mean the amount of charity care costs (not charges) incurred by Purchaser in connection with the operation and provision of services at St. Francis Medical Center. The definition and methodology for calculating "charity care" and the methodology for calculating "costs" shall be the same as that used by Office of Statewide Health Planning Development (OSHPD) for annual hospital reporting purposes.<sup>54</sup>

Purchaser's obligation under this Condition shall be prorated on a daily basis if the closing date of the Asset Purchase Agreement is a date other than the first day of Verity Health System of California, Inc.'s fiscal year.

For the second fiscal year and each subsequent fiscal year, the Minimum Charity Care Amount shall be increased (but not decreased) by an amount equal to the Annual Percent increase, if any, in the 12 Months Percent Change: All Items Consumer Price Index for All Urban Consumers in the Los Angeles-Long Beach-Anaheim Average Base Period: 1982-84=100 (CPI-LA, as published by the U.S. Bureau of Labor Statistics).

If the actual amount of charity care provided at St. Francis Medical Center for any fiscal year is less than the Minimum Charity Care Amount (as adjusted pursuant to the above-referenced Consumer Price Index) required for such fiscal year, Purchaser shall pay an amount equal to the deficiency to one or more tax-exempt entities that provide direct healthcare services to residents in the St. Francis Medical Center's service area (31 ZIP codes), as described on page 54 in the Healthcare Impact Report authored by JD Healthcare dated August 16, 2019. (Exhibit 1.) Such payment(s) shall be made within six months following the end of such fiscal year.

For six fiscal years from the closing date of the Asset Purchase Agreement Purchaser shall provide an annual amount of Community Benefit Services at St. Francis Medical Center equal to or greater than \$1,139,301 (the "Minimum Community Benefit Services Amount") exclusive of

<sup>&</sup>lt;sup>54</sup> OSHPD defines charity care by contrasting charity care and bad debt. According to OSHPD, "the determination of what is classified as . . . charity care can be made by establishing whether or not the patient has the ability to pay. The patient's accounts receivable must be written off as bad debt if the patient has the ability but is unwilling to pay off the account."

any funds from grants. For six fiscal years, the following community benefit programs and services shall continue to be offered at its current or equivalent location:

Southern California Crossroads Program; Health Benefit Resource Center; Welcome Baby Program; Healthy Community Initiatives; American Career College access for onsite training; Paramedic Training and Education; and Patient Transportation support.

The planning of, and any subsequent changes to, the community benefit services provided at St. Francis Medical Center shall be decided after consultation with the Local Governing Board of Directors.

Purchaser's obligation under this Condition shall be prorated on a daily basis if the effective date of the Asset Purchase Agreement is a date other than the first day of Verity Health System of California, Inc.'s fiscal year.

For the second fiscal year and each subsequent fiscal year, the Minimum Community Benefit Services Amount shall be increased (but not decreased) by an amount equal to the Annual Percent increase, if any, in the 12 Months Percent Change: All Items Consumer Price Index for All Urban Consumers in the Los Angeles-Long Beach-Anaheim Average Base Period: 198284=100 (CPI-LA, as published by the U.S. Bureau of Labor Statistics).

If the actual amount of community benefit services provided at St. Francis Medical Center for any fiscal year is less than the Minimum Community Benefit Services Amount (as adjusted pursuant to the above-referenced Consumer Price Index) required for such fiscal year, Purchaser shall pay an amount equal to the deficiency to one or more tax-exempt entities that provide community benefit services for residents in St. Francis Medical Center's service area (31 ZIP codes), as defined on as described on page 54 in the Healthcare Impact

Report authored by JD Healthcare dated August 16, 2019. (Exhibit 1.) Such payment(s) shall be made within six months following the end of such fiscal year.

For at least ten years from the closing date of the Asset Purchase Agreement unless otherwise indicated, Purchaser shall maintain its contracts and any amendments and exhibits thereto with the City and/or County of Los Angeles for services, including the following:

- Participation in the Hospital Preparedness Program between the Hospital and Los Angeles County;
- Department of Mental Health Legal Entity Contract between the Hospital and Los Angeles County;

Paramedic Base Hospital Services between the Hospital and Los Angeles County; Radiation Therapy Services between the Hospital and Los Angeles County;

I

- Designation Agreement between the County of Los Angeles Department of Mental Health (LAC-DMH) and the Hospital and approved as a 72-hour Evaluation and Intensive Treatment facility;
- Trauma Center Service Agreement between the Hospital and Los Angeles County; and
- Paramedic Training Institute Students between the Hospital and Los Angeles County.

For at least ten years from the closing date of the Asset Purchase Agreement, Purchaser shall provide to the Los Angeles County Department of Health Services and Los Angeles County of Department of Mental Health information and documents related to staffing assessments, clinical guidelines, services provided, and technology needs for St. Francis Medical Center. The goal is to ensure that Purchaser's decisions or changes in these areas will not be motivated by a desire to move away from serving the Medi-Cal population. Such information and documents will also be provided to the Local Governing Board.

For ten years from the closing date of the Asset Purchase Agreement, Purchaser shall have at St. Francis Medical Center a Local Governing Board of Directors. Purchaser shall consult with the Local Governing Board of Directors prior to making changes to medical services, community benefit programs, making capital expenditures, including making changes to the charity care and collection policies, and making changes to charity care services provided at St. Francis Medical Center. The members of the Local Governing Board of Supervisors, and community representatives form St. Francis Medical center's medical staff, St. Francis Medical Center's Chief of Staff, one member designated by the Los Angeles County Board of Supervisors, and community representatives from St. Francis Medical Center's primary service area (31 ZIP codes), as described on page 54 in the Healthcare Impact Report authored by JD Healthcare dated August 16, 2019 attached hereto as Exhibit 1, including at least one member from a local healthcare advocacy group. Such consultation shall occur at least sixty days prior to the effective date of such changes or actions unless done so on an emergency basis. The Local Governing Board's approval is required of all reports submitted to the Attorney General regarding compliance with these Conditions.

Purchaser shall commit to reserve or expend for St. Francis Medical Center, St. Vincent Medical Center, and Seton Medical Center for capital improvements to the hospitalshospital over the five-year period from the closing of the Asset Purchase Agreement, the amount of capital that remains unexpended from the \$180 million commitment required of Blue Mountain Capital Management, LLC as part of the Attorney General Conditions approved on December 3, 2015 but this amount can be no less than \$5.8 million among the three hospitalsSt. Francis Medical Center and St. Vincent Medical Center. Purchaser understands that such condition has previously been satisfied but will still commit to expend approximately \$35 million on St. Francis Medical Center over the five-year period from the closing of the Asset Purchase Agreement.

Purchaser shall commit the necessary investments required to maintain OSHPD seismic compliance requirements at the Hospital through 2030 under the Alfred E. Alquist Hospital Facilities Seismic Safety Act of 1983, as amended by the California Hospital Facilities Seismic Safety Act, (Health & Safety. Code, § 129675-130070).

## Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 109 of 111

Purchaser shall maintain privileges for current medical staff who are in good standing as of the closing date of the Asset Purchase Agreement. Further, the closing of the Asset Purchase Agreement shall not change the independence of the medical staff.

There shall be no discrimination against lesbian, gay, bisexual, or transgender individuals at St. Francis Medical Center, and no restriction or limitation on providing or making reproductive health services available at St. Francis Medical Center, its medical office buildings, or at any of its facilities. Both of these prohibitions shall be set forth in Purchaser's written policies, adhered to, and strictly enforced.

For eleven fiscal years from the closing date of the Asset Purchase Agreement Purchaser shall submit to the Attorney General, no later than four months after the conclusion of each fiscal year, a report describing in detail compliance with each Condition set forth herein. The Chairman of the Board of Directors of Purchaser shall certify that the report is true, accurate, and complete and provide documentation of the review and approval of the report by the Local Governing Board.

At the request of the Attorney General, all parties listed in Condition I, Verity Health System of California, Inc., Verity Holdings, LLC, Purchaser, and any other parties referenced in the agreements listed in Condition II shall provide such information as is reasonably necessary for the Attorney General to monitor compliance with these Conditions and the terms of the transaction as set forth herein. The Attorney General shall, at the request of a <u>partyParty</u> and to the extent provided by law, keep confidential any information so produced to the extent that such information is a trade secret or is privileged under state or federal law, or if the private interest in maintaining confidentiality clearly outweighs the public interest in disclosure.

Once the Asset Purchase Agreement is closed, and so long as such actions are consistent with the Asset Purchase Agreement, all parties listed in Condition I, and any other parties referenced in the agreements listed in Condition II are deemed to have explicitly and implicitly consented to the applicability and compliance with each and every Condition and to have waived any right to seek judicial relief with respect to each and every Condition.

The Attorney General reserves the right to enforce each and every Condition set forth herein to the fullest extent provided by law. In addition to any legal remedies the Attorney General may have, the Attorney General shall be entitled to specific performance, injunctive relief, and such other equitable remedies as a court may deem appropriate for breach of any of these Conditions. Pursuant to Government Code section 12598, the Attorney General's office shall also be entitled to recover its attorney fees and costs incurred in remedying each and every violation.

## Analysis of the Hospital's Service Area

## Service Area Definition

Based upon the Hospital's CY 2017 inpatient discharges, the Hospital's service area is comprised of 31 ZIP Codes from which 75% of the Hospital's inpatient discharges came from. Approximately 51% of the Hospital's discharges originated from the top eight ZIP Codes, located in Lynwood, South Gate, Los Angeles, Bell, Compton, Bell Gardens, and Huntington Park. In CY 2017, the Hospital's market share in the service area was approximately 11% based on total area discharges.

PATIENT ORIGIN, CY 2017						
			Percentage	Cumulative	Market	Total
ZIP Code	Community	Discharges	of Discharges	Percentage	Share	Discharges
90262	Lynwood	2,490	11.1%	11.1%	38.1%	6,538
90280	South Gate	2,187	9.8%	20.9%	29.0%	7,554
90221	Compton	1,400	6.3%	27.2%	24.1%	5,812
90201	Bell	1,359	6.1%	33.3%	16.3%	8,363
90002	Los Angeles	1,066	4.8%	38.0%	18.4%	5,797
90255	Huntington Park	956	4.3%	42.3%	15.5%	6,172
90059	Los Angeles	948	4.2%	46.6%	17.2%	5,527
90001	Los Angeles	922	4.1%	50.7%	15.6%	5,901
90220	Compton	708	3.2%	53.9%	12.7%	5,554
90222	Compton	700	3.1%	57.0%	18.1%	3.868
90003	Los Angeles	625	2.8%	59.8%	7.6%	8,209
90044	Los Angeles	542	2.4%	62.2%	4.5%	11,994
90723	Paramount	525	2.3%	64.6%	11.7%	4,483
90061	Los Angeles	358	1.6%	66.2%	9.5%	3,764
90650	Norwalk	344	1.5%	67.7%	3.3%	10,373
90270	Maywood	282	1.3%	69.0%	12.2%	2,309
90805	Long Beach	267	1.2%	70.2%	2.7%	9,940
90706	Bellflower	263	1.2%	71.3%	3.6%	7,223
90242	Downey	252	1.1%	72.5%	6.2%	4,038
90241	Downey	224	1.0%	73.5%	6.0%	3,726
90660	Pico Rivera	91	0.4%	73.9%	1.4%	6,608
90240	Downey	69	0.3%	74.2%	3.3%	2,073
90670	Santa Fe Springs	46	0.2%	74.4%	2.7%	1,703
90605	Whittier	44	0.2%	74.6%	1.1%	4,082
90606	Whittier	44	0.2%	74.8%	1.4%	3,244
90703	Cerritos	37	0.2%	74.9%	0.9%	4,026
90604	Whittier	32	0.1%	75.1%	0.9%	3,698
90701	Artesia	31	0.1%	75.2%	1.7%	1,813
90638	La Mirada	30	0.1%	75.4%	0.7%	4,274
90603	Whittier	3	0.0%	75.4%	0.1%	2,152
90639	La Mirada	0	0.0%	75.4%	0.0%	10
Sub-Tot		16,845	75.4%	75.4%	10.5%	160,828
al All Other		5,504	24.6%	100%		·
Grand To	tal	22,349	100%	10070		
Source: OSHPD Discharge Database. CY 2017						

Source: OSHPD Discharge Database, CY 2017

Case 2:18-bk-20151-ER Doc 6669-1 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 8 - 3/26/2020 Redline Page 111 of 111

## Case 2:18-bk-20151-ER Doc 6669-2 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 9 - COVID-19 Grant Email Page 1 of 5

## Esposito, Zach

From:	Joel Richlin (PHMI) <jrichlin@primehealthcare.com></jrichlin@primehealthcare.com>
Sent:	Tuesday, March 31, 2020 9:57 PM
То:	Moloney, Jim; Steve Aleman (PHMI)
Subject:	RE: Call to Discuss COVID 19 between Prime and Verity

Jim,

I'm still working on the tweaks. As discussed on the call, we need to clarify some of the definitions related to AR and government payments going different ways. This is the point I discussed with Elspeth where Verity excluded additional government payments related to COVID in their last turn and we think those are essential to the value of the asset since those payments will not even make the hospital whole. So we are revising that language to make it as clear as possible because the passage of new laws creating these payments needs to be addressed. Realistically, we can likely circulate our redline tomorrow morning, and then Steve and I could get a on a quick call later that day to review it with the team. I suggest we have that touch base call to discuss our redline before the principals call, which could happen either later Wednesday or more realistically Thursday morning.

Joel

## A. Joel Richlin, Esq.

Vice President, Deputy General Counsel, & Chief Litigation Counsel 3480 E. Guasti Road, Ontario, CA 91761 Direct: 909-235-4235 | Facsimile: 909-235-4316 Email: JRichlin@primehealthcare.com





From: Moloney, Jim <JMoloney@cainbrothers.com>
Sent: Tuesday, March 31, 2020 7:40 PM
To: Steve Aleman (PHMI) <SAleman2@primehealthcare.com>; Joel Richlin (PHMI) <JRichlin@primehealthcare.com>
Subject: RE: Call to Discuss COVID 19 between Prime and Verity

WARNING: This email originated outside the Prime Healthcare email system! Do Not Click links if this user is unknown.

Joel,

I meant to copy you on this – the deposit will need to be in cash via a wire transfer. Our team is working on SFMC and preparing for the SVMC court hearing tomorrow. Should we expect a turn of the APA tonight or tomorrow?

JM

Jim Moloney | Managing Director Cain Brothers, a division of KeyBanc Capital Markets O: 415.962.2961 | M: 510.759.7300 | San Francisco jmoloney@cainbrothers.com | www.cainbrothers.com Please note our new address: One California Street Suite 2400 San Francisco, CA 94111

From: Moloney, Jim
Sent: Tuesday, March 31, 2020 7:22 PM
To: 'Steve Aleman (PHMI)' <<u>SAleman2@primehealthcare.com</u>>
Subject: RE: Call to Discuss COVID 19 between Prime and Verity

Steve,

Sorry for the late response but the Bidding Procedures require a deposit via a wire transfer. Let me know if you would like to discuss.

JM

Jim Moloney | Managing Director Cain Brothers, a division of KeyBanc Capital Markets O: 415.962.2961 | M: 510.759.7300 | San Francisco jmoloney@cainbrothers.com | www.cainbrothers.com

Please note our new address: One California Street Suite 2400 San Francisco, CA 94111

From: Steve Aleman (PHMI) <<u>SAleman2@primehealthcare.com</u>>
Sent: Tuesday, March 31, 2020 12:11 PM
To: Moloney, Jim <<u>JMoloney@cainbrothers.com</u>>
Subject: RE: Call to Discuss COVID 19 between Prime and Verity

Jim,

I appreciate the setting up the call this morning, it was very helpful. As we prepare for end of week, is a letter of credit an acceptable form of deposit?

## **Steve Aleman**

Vice President, Investor Relations and Corporate Development 3480 E. Guasti Road, Ontario, CA 91761 Direct: 909-638-0068 | Cell: 805-506-1792 E-mail: <u>SAleman2@primehealthcare.com</u>





#### Case 2:18-bk-20151-ER Doc 6669-2 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 9 - COVID-19 Grant Email Page 3 of 5

----Original Appointment----From: Steve Aleman (PHMI) On Behalf Of Moloney, Jim
Sent: Tuesday, March 31, 2020 7:01 AM
To: Sunitha Reddy (PHS); Michael Bogert (PHMI); Steve Aleman (PHS); Adcock, Rich; Chadwick, Peter (VHS)
Subject: FW: Call to Discuss COVID 19 between Prime and Verity
When: Tuesday, March 31, 2020 9:00 AM-9:45 AM (UTC-08:00) Pacific Time (US & Canada).
Where: Dial-in: (415) 655-0001,,,852892186#

Sent from my Verizon, Samsung Galaxy smartphone

When: Mar 31, 2020 9:00:00 AM Where: Dial-in: (415) 655-0001,,,852892186#

When: Mar 31, 2020 9:00:00 AM Where: Dial-in: (415) 655-0001,,,852892186#

From: Moloney, Jim
Sent: Tuesday, 31 March 2020 02:43:29 UTC
To: Moloney, Jim; Steve Aleman (PHS); Adcock, Rich; Chadwick, Peter (VHS)
Subject: Call to Discuss COVID 19 between Prime and Verity
When: Tuesday, March 31, 2020 4:00 PM-4:45 PM.
Where: Dial-in: (415) 655-0001,,,852892186#

WARNING: This email originated outside the Prime Healthcare email system! Do Not Click links if this user is unknown.

-- Do not delete or change any of the following text. --

## When it's time, join your Webex meeting here.

Meeting number (access code): 852 892 186

## Join meeting

Join by phone Tap to call in from a mobile device (attendees only) +1-415-655-0001 US Toll 1-844-621-3956 US Toll Free Global call-in numbers | Toll-free calling restrictions

Join from a video system or application

## Case 2:18-bk-20151-ER Doc 6669-2 Filed 09/28/21

Entered 09/28/21 18:34:40 Desc Exhibit 9 - COVID-19 Grant Email Page 4 of 5

Dial 852892186@keybank.webex.com

You can also dial <u>173.243.2.68</u> and enter your meeting number.

Join using Microsoft Lync or Microsoft Skype for Business

Dial 852892186.keybank@lync.webex.com

If you are a host, go here to view host information.

Need help? Go to http://help.webex.com

This communication may contain privileged and/or confidential information. It is intended solely for the use of the addressee. No confidentiality or privilege is waived or lost as a result of any transmission errors. If you are not the intended recipient, you are strictly prohibited from disclosing, copying, distributing or using any of this information. If you have received this communication in error, please contact the sender immediately and destroy the material in its entirety, whether electronic or hard copy.

This communication is for informational purposes only, is not an offer, solicitation, recommendation or commitment for any transaction or to buy or sell any security or other financial product, and is not intended as investment advice or as a confirmation of any transaction. Any market price, indicative value, estimate, view, opinion, data or other information herein is not warranted as to completeness or accuracy, is subject to change without notice, and KeyBanc Capital Markets Inc. accepts no liability for its use or to update or keep it current. Any views or opinions are those of the individual sender, not necessarily of KeyBanc Capital Markets Inc.

The sender of this communication is a licensed securities representative employed by or associated with KeyBanc Capital Markets Inc. Member FINRA/SIPC and may also represent KeyBank National Association ("KeyBank N.A."). Securities products and services are offered by KeyBanc Capital Markets Inc. Banking products and services are offered by KeyBank N.A.

## 127 Public Square, Cleveland, OH 44114

If you prefer not to receive future e-mail offers for products or services from Key send an e-mail to DNERequests@key.com with 'No Promotional E-mails' in the SUBJECT line. This email, and any accompanying attachment, is for your information only. The opinions expressed in this email and/or its attachments are those of the author(s) and are not necessarily those of Cain Brothers, a division of KeyBanc Capital Markets Inc. ("Cain Brothers") or its affiliates. This email may contain information that is confidential, privileged or otherwise protected from disclosure and is intended only for the use of the individual or entity to which it is addressed. Any unauthorized dissemination, distribution, copying or use of the contents of this email is strictly prohibited. If you are not an intended recipient or have received this email message in error, please notify the sender immediately and destroy this communication. Please be advised that Cain Brothers, subject to applicable law, may monitor, review and retain email communications transmitted through its networks and systems, and may provide any such communication to regulators, law enforcement, in litigation or as required by law.

## Case 2:18-bk-20151-ER Doc 6669-2 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 9 - COVID-19 Grant Email Page 5 of 5

"Cain Brothers, a division of KeyBanc Capital Markets Inc." is a trade name of KeyBanc Capital Markets Inc. Member NYSE/FINRA/SIPC. KeyBanc Capital Markets Inc. and KeyBank National Association are separate but affiliated companies. Securities products and services are offered by KeyBanc Capital Markets Inc. and its licensed securities representatives.

Securities products and services: Not FDIC Insured – No Bank Guarantee – May Lose Value

Banking products and services are offered by KeyBank National Association Member FDIC and Equal Housing Lender. Credit products are subject to credit approval.

The information contained in this email may be confidential and/or legally privileged, including HIPAA Protected Health Information or information that is proprietary or trade secret. It has been sent for the sole use of the intended recipient(s). If you are not the intended recipient, you are hereby notified that any unauthorized review, use, disclosure, dissemination, distribution, or copying of this communication is strictly prohibited. If you received this communication in error, please inform the sender and destroy all copies of the message. The organization accepts no liability for any damage caused by transmittance of this email. Thank you.

## ASSET PURCHASE AGREEMENT

## By and Among

## ST. FRANCIS MEDICAL CENTER, VERITY HOLDINGS, LLC, and VERITY HEALTH SYSTEM OF CALIFORNIA, INC.

(as "Sellers")

and

## PRIME HEALTHCARE SERVICES, INC.

(as "Purchaser")

Dated \_\_\_\_\_, 2020

## Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 2 of 147

## **INDEX OF SCHEDULES & EXHIBITS**

## Schedules

Schedule 1.4(c)	Owned Real Property
Schedule 1.7(b)	Licenses
Schedule 1.7(c)(ii)	Leased Real Property
Schedule 1.7(c)(iii)	Tenant Leases
Schedule 1.7(d)	Assigned Contracts and Assigned Leases
Schedule 1.8(eebb)	Certain Other Excluded Assets
Schedule 1.9(i)	Other Obligations
Schedule 2.4	Seller's Knowledge of Breach or Default
Schedule 2.5	Brokers and Finders (Sellers)
Schedule 2.6	Compliance (Sellers)
Schedule 2.7	Required Consents (Sellers)
Schedule 2.8	Legal Proceedings (Sellers)
Schedule 2.9(b)	Environmental Matters (Sellers)
Schedule 2.11	Missing Material Licenses (Sellers)
Schedule 2.13	Non-Bankruptcy Court Legal Proceedings (Sellers)
Schedule 2.19	List of Excluded Individuals (Seller)
Schedule 3.4	No Violation (Purchaser)
Schedule 3.7	Legal Proceedings (Purchaser)
Schedule 5.3	"Hospital Employees"
Schedule 11.3(b)	Tax Allocation Schedule

## Exhibits

No.	Description	Status
Exhibit 1.2	Escrow Agreement	to come
Exhibit 1.3(a)	Leaseback Agreement	Attached
Exhibit 1.3(b)	Interim Management Agreement	Attached
Exhibit 1.4(a)	Bill of Sale	to come
Exhibit 1.4(b)	Real Estate Assignment Agreements	to come
Exhibit 1.4(c)	Quitclaim Deeds	to come
Exhibit 1.4(d)	Transfer Agreement	to come
Exhibit 1.4(f)	Transition Services Agreement	Attached
Exhibit 5.9(c)	Accepted Conditions (Attorney General Conditions)	to come

## ASSET PURCHASE AGREEMENT

This Asset Purchase Agreement (the "Agreement") is made and entered into as of \_\_\_\_\_\_\_, 2020 (the "Signing Date") by and among St. Francis Medical Center, a California nonprofit public benefit corporation ("SFMC"), Verity Holdings, LLC ("Verity Holdings"), a California limited liability company and Verity Health System of California, Inc., a California nonprofit public benefit corporation ("VHS" and, together with SFMC and Verity Holdings, the "Sellers" and each individually a "Seller") and Prime Healthcare Services, Inc., a Delaware corporation ("Purchaser" and, together with the Sellers, the "Parties" and each individually a "Party").

## RECITALS

A. SFMC engages in the business of operating a hospital known as St. Francis Medical Center ("St. Francis") in the City of Lynwood, California, including the hospital pharmacy, laboratory and emergency department as well as through the medical office buildings and clinics owned or operated by SFMC (with St. Francis, collectively referred to herein, as the "Hospital").

B. VHS is the sole member of SFMC and Verity Holdings. Verity Holdings owns and rents certain properties on or near the Hospital campus.

C. Purchaser desires to purchase from Sellers, and Sellers desire to sell to Purchaser, the assets described in <u>Section 1.7</u> below for the consideration and upon the terms and conditions contained in this Agreement.

D. On August 28, 2018, Sellers filed voluntary petitions for relief under Chapter 11 of Title 11 of the United States Code (the "**Bankruptcy Code**") in the United States Bankruptcy Court for the Central District of California, Los Angeles Division, before the Honorable Ernest M. Robles (the "**Bankruptcy Court**"), which are jointly administered with their affiliates under Case No. 18-20151 (the "**Bankruptcy Cases**").

E. On February 10, 2020, Sellers filed the Debtors' Notice of Motion and Motion for the Entry of (I) an Order (1) Approving Form of Asset Purchase Agreement; (2) Approving Auction Sale Format, Bidding Procedures and Stalking Horse Bid Protections; (3) Approving Form of Notice to be Provided to Interested Parties; (4) Scheduling a Court Hearing to Consider Approval of the Sale to the Highest Bidder; and (5) Approving Procedures Related to the Assumption of Certain Executory Contracts and Unexpired Leases; and (II) an Order (A) Authorizing the Sale of Property Free and Clear of All Claims, Liens and Encumbrances; Memorandum Of Points And Authorities In Support Thereof [Docket No. \_\_\_4069] (the "Bidding Procedures Motion") which was approved by an order of the Bankruptcy Court (the "Bidding Procedures Order") [Docket No. 4165] and pursuant thereto the Parties have entered into this Agreement.

F. The Parties intend to effectuate the transactions contemplated by this Agreement through a sale of the Assets pursuant to the Bidding Procedures and as approved by the Bankruptcy Court pursuant to Sections 363 and 365 of the Bankruptcy Code.

#### AGREEMENT

NOW, THEREFORE, in consideration of the foregoing premises and the mutual promises and covenants contained in this Agreement, and for their mutual reliance and incorporating into this Agreement the above recitals, the Parties hereto agree as follows:

## ARTICLE 1 SALE AND TRANSFER OF ASSETS; CONSIDERATION; CLOSING

#### 1.1 <u>Purchase Price</u>.

(a) Subject to the terms and conditions of this Agreement, the aggregate consideration shall be as follows:

(i) Cash payment to Sellers of Two Hundred Million Dollars (\$200,000,000) ("**Base Price**");

(ii) Cash payment of the Closing QAF Payment Amount (defined below in Section 1.1(c));Sellers shall retain, as an Excluded Asset, an estimated amount of Twenty-Nine Million Dollars (\$29,000,000) in connection with QAF V payments and Eighty-Three Million Dollars (\$83,000,000) in connection with QAF VI payments (subject to adjustment as described in Section 1.8(b));

(iii) Cash payment of <u>Fifty-TwoSixty-One</u> Million Dollars (\$52,000,00061,000,000) (the "A/R Target Amount") as consideration for the Accounts Receivable transferred at Closing (subject to adjustment in Section 1.12);

(iv) Cash payment for Sellers' payroll liabilities at Closing (the "**Payroll Amount**") (which as of October 31, 2019, had an aggregate value of approximately <u>SevenFive</u> Million Dollars (\$7,000,000<u>5,000,000</u>));

(v) Cash payment for accrued vacation and other paid time-off of Sellers' employees at Closing (the "**PTO Amount**") (which as of October 31, 2019, had an aggregate value of approximately <u>ElevenTen</u> Million Dollars (\$11,000,00010,000,000));-and

(vi) An amount equal to the Cure Costs (defined below) associated with outstanding liabilities of Sellers under any Assigned Leases and/or Assigned Contracts; and

(vii) An amount determined in accordance with <u>Section 1.6</u>.

(b) At the Closing, Purchaser shall pay to Sellers, by wire transfer of immediately available funds to the accounts specified by Sellers to Purchaser in writing, an aggregate amount equal to the Base Price, *plus* the Closing QAF Payment Amount, *plus* the A/R

## Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 5 of 147

Target Amount, *plus* the Cure Cost Amount, *plus* the PTO Amount, *plus* the Payroll Amount (collectively, the "**Purchase Price**"), *minus* the Deposit (defined below).

For purposes of this Agreement, the "QAF Program" means the California (c)Department of Health Care Services ("DHCS") Hospital Quality Assurance Fee Programs V ("QAF V") and VI ("QAF VI"). During the period prior to the Closing, Sellers shall pay or permit DHCS to offset from amounts owed to St. Francis any fees due and owing prior to the Closing under QAF V and QAF VI (such payments or offsets shall be included in the formula described in Section 1.8(b)), and Sellers shall be entitled to retain all payments received under QAF V and QAF VI. At Closing, Purchaser shall pay Sellers an aggregate amount in respect of QAF V and QAF VI (the "Closing QAF Payment Amount") equal to: (i) (A) Twenty-Eight Million Dollars (\$28,000,000) as consideration for the QAF V payments scheduled by DHCS to be paid to Sellers in November 2020 and May 2021, plus (B) any fees paid by Sellers or offset by DHCS in respect of QAF V relating to St. Francis between the Signing Date and the Closing; plus (ii) an amount equal to (A) the QAF VI Seller Net Payments (defined below), multiplied by (B) fifty percent (50%), minus (C) the dollar value of all payments actually received by Sellers in respect of QAF VI relating to St. Francis prior to the Closing, plus (D) the dollar value of all fees paid by Sellers or offset by DHCS in respect of QAF VI relating to St. Francis prior to the Closing (the amount determined pursuant to this clause (ii), the "QAF VI Partial Payment") as a partial up-front payment for the QAF VI Seller Net Payments. After the Closing, Purchaser shall pay to Sellers, promptly following receipt thereof (but in any event within ten (10) business days of such receipt), all QAF VI payments received relating to St. Francis until Purchaser has paid to Sellers an amount, when added to the QAF VI Partial Payment, equal to the QAF VI Seller Net Payments.

(d) Separate from, and in addition to, the Purchase Price, Purchaser commits to:(i) invest Thirty-FiveForty-Seven Million Dollars (\$35,000,00047,000,000) in estimated deferred capital expenditures for St. Francis over the five (5) year period immediately after the Closing Date; and

(ii) remediate Sellers' estimated<u>(including</u> NPC-3 seismic compliance responsibilities (which is estimated to cost approximately Twelve Million Dollars (\$12,000,000)).

1.2 Deposit. Purchaser has deposited an amount equal to ten percent (10%) of the Purchase Price (the "Initial Deposit") by wire transfer to Chicago Title Insurance Company ("Escrow Agent") pursuant to that certain Escrow Agreement attached hereto as Exhibit 1.2. Within two (2) business day after the date of the conclusion of the auction for the Assets, if any, if Purchaser is the Prevailing Bidder (as defined below) and as a result of such auction has opted to increase the Purchase Price (the "Auction Purchase Price"), Purchaser shall deposit an additional amount with the Escrow Agent such that the total amount deposited with the Escrow Agent is equal to ten percent (10%) of the Auction Purchase Price (the Initial Deposit and any additional amount deposited with the Escrow Agent (if any) referred to collectively as the "Deposit") by wire transfer to the Escrow Agent. All fees of the Escrow Agent shall be paid by Purchaser. The Deposit shall be non-refundable in all events, except in the event the Closing does not occur due to Purchaser's termination of the Agreement pursuant to Sections 9.1 (a), (c), (d), (f), (g), (h) or (i) hereof. Upon Closing, the Deposit will be credited against the Purchase Price.

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 6 of 147

1.3 Closing Date. The consummation of the transactions contemplated by this Agreement (the "Closing") shall take place at the offices of Dentons US LLP, 601 South Figueroa St., Suite 2500, Los Angeles, CA 90017-5704 (the day on which Closing actually occurs, the "Closing Date") within three (3) business days following the satisfaction or waiver of the conditions set forth in ARTICLE 7 and ARTICLE 8, and other than those conditions that by their nature are to be satisfied at Closing but subject to fulfillment or waiver of those conditions. The Closing shall be deemed to occur and to be effective as of 12:00 a.m. pacific time on the day immediately after the Closing Date (the "Effective Time"). Purchaser and Sellers agree that because the change of ownership and regulatory approval process may take an extended period of time, at the Effective Time, the Assets (less any Assets constituting drugs or pharmacy assets, the Medicare/Medi-Cal Agreements, the Lockboxes, the Transferred Managed Care Agreements and the Transferred Private Payor Agreements (as such terms are defined below)) will be sold to Purchaser and immediately leased back to Sellers (substantially in the form of the Leaseback Agreement attached hereto as Exhibit 1.3(a), the "Leaseback Agreement"), with a concurrent management arrangement (substantially in the form of the Interim Management Agreement attached hereto as Exhibit 1.3(b), the "IMA"). On the effective date that Purchaser obtains a general acute care hospital license from the California Department of Public Health ("CDPH") and a hospital pharmacy permit from the California State Board of Pharmacy ("BOP") (i) the Leaseback Agreement and IMA will terminate and (ii) the drugs and pharmacy assets, the Medicare/Medi-Cal Agreements, the Lockboxes, the Transferred Managed Care Agreements and the Transferred Private Payor Agreements will be transferred to Purchaser (without payment of any additional Purchase Price) (the "Licensure Date"). For the avoidance of any doubt, the Licensure Date shall be the date the Purchaser's hospital license and pharmacy permit are effective, even if they are not actually issued until a later date. The Licensure Date may be determined based on oral assurances from CDPH and the BOP.

1.4 <u>Items to be Delivered by Sellers at Closing</u>. At or before the Closing, Sellers shall deliver, or cause to be delivered, to Purchaser the following:

(a) a Bill of Sale substantially in the form of <u>Exhibit 1.4(a)</u> attached hereto (the "**Bill of Sale**"), duly executed by each Seller;

(b) Real Estate Assignment Agreements (the "**Real Estate Assignments**") in the form of <u>Exhibit 1.4(b)</u> attached hereto with respect to (i) the Leased Real Property, and (ii) the Tenant Leases, each duly executed by the applicable Sellers;

(c) Quitclaim Deeds in the form of <u>Exhibit 1.4(c)</u> attached hereto with respect to the real property listed in <u>Schedule 1.4(c)</u>, together with all plant, buildings, structures, installments, improvements, fixtures, betterments, additions and constructions in progress situated thereon (the "**Owned Real Property**") duly executed by SFMC and Verity Holdings as applicable;

(d) an Assigned Contract Transfer Agreement (the "**Transfer Agreement**") in the form of Exhibit 1.4(d) attached hereto, duly executed by the applicable Sellers;

(e) evidence of payment of all Cure Costs;

## Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 7 of 147

(f) the Transition Services Agreement (the "**Transition Services Agreement**") in form attached hereto as <u>Exhibit 1.4(f)</u>, duly executed by the applicable Sellers;

(g) the Leaseback Agreement and IMA, duly executed by SFMC;

(h) favorable certificates of good standing, of each Seller, issued by the State of California, dated no earlier than a date which is fifteen (15) calendar days prior to the Closing Date;

(i) a duly executed certificate of an officer of each Seller certifying to Purchaser (i) the incumbency of the officers of such Seller on the Signing Date and on the Closing Date and bearing the authentic signatures of all such officers who shall execute this Agreement and any additional documents contemplated by this Agreement and (ii) the due adoption and text of the resolutions or consents of the Board of Directors of such Seller authorizing (I) the transfer of the Assets and transfer of the Transferred Obligations by such Seller to Purchaser and (II) the due execution, delivery and performance of this Agreement and all additional documents contemplated by this Agreement, and that such resolutions have not been amended or rescinded and remain in full force and effect on the Closing Date;

(j) a certified copy of the Sale Order (as defined below);

(k) non-foreign affidavits of SFMC and Verity Holdings, as applicable, dated as of the Closing Date, in the form of Exhibit 1.4(k); and

(1) Any such other instruments, certificates, consents or other documents which Purchaser and Sellers mutually deem reasonably necessary to carry out the transactions contemplated by this Agreement and to comply with the terms hereof.

1.5 <u>Items to be Delivered by Purchaser at Closing</u>. At or before the Closing, Purchaser shall deliver or cause to be delivered to Sellers the following:

(a) payment of the Purchase Price, minus the Deposit;

(b) a duly executed certificate of the Secretary of Purchaser certifying to Sellers (a) the incumbency of the officers of Purchaser on the Signing Date and on the Closing Date and bearing the authentic signatures of all such officers who shall execute this Agreement and any additional documents contemplated by this Agreement and (b) the due adoption and text of the resolutions of the Board of Directors of Purchaser authorizing the execution, delivery and performance of this Agreement and all additional documents contemplated by this Agreement, and that such resolutions have not been amended or rescinded and remain in full force and effect on the Closing Date;

(c) favorable certificate of good standing, of Purchaser, issued by the California Secretary of State dated no earlier than a date which is fifteen (15) calendar days prior to the Closing Date;

(d) the Transition Services Agreement, duly executed by Purchaser;

- (e) the Leaseback Agreement and IMA, duly executed by Purchaser;
- (f) the Bill of Sale, duly executed by Purchaser;
- (g) the Real Estate Assignment(s), duly executed by Purchaser;
- (h) the Transfer Agreement, duly executed by Purchaser;

(i) Preliminary Change of Ownership Report(s) (BOE-502-A) with respect to the Owned Real Property, duly executed by Purchaser; and

(j) any such other instruments, certificates, consents or other documents which Purchaser and Sellers mutually deem reasonably necessary to carry out the transactions contemplated by this Agreement and to comply with the terms hereof.

1.6 <u>Prorations and Utilities</u>. All items of income and expense listed below with respect to the Assets shall be prorated in accordance with the principles and the rules for the specific items set forth hereafter:

(a) All transfer, conveyance, sales, use, stamp, similar state and local taxes arising from the sale of the Assets hereunder shall be the responsibility of, and allocated to, Purchaser.

Other than the Utility Deposits (defined below), which are governed by (b) Section 1.8(n), and other than with respect to Cure Costs, the following costs and expenses shall be prorated based upon the payment period (*i.e.*, calendar or other tax fiscal year) to which the same are attributable: (i) all real estate and personal property lease payments, real estate and personal property taxes, real estate assessments and other similar charges against real estate, (ii) power and utility charges, (iii) payments made by Sellers for information technology software or services for periods after the Effective Time, (iv) payments made by Sellers in association with Licenses or dues paid to government and non-governmental agencies for calendar year 2020, and (v) other similar costs for items or services which continue past the Effective Time (collectively, the "Prorated Charges"). Sellers shall pay at or prior to the Closing (or Purchaser shall receive credit for) any unpaid Prorated Charges attributable to periods or portions thereof occurring prior to the Effective Time, and Purchaser shall be responsible for or, to the extent previously paid by Sellers, pay to Sellers at the Closing all Prorated Charges attributable to periods or portions thereof occurring from and after the Effective Time. In the event that as of the Closing Date the actual tax bills for the tax year or years in question are not available and the amount of taxes to be prorated as aforesaid cannot be ascertained, then rates, millages and assessed valuation of the previous year, with known changes, shall be used. The Parties agree that if the real estate and personal property tax prorations are made based upon the taxes for the preceding tax period, the prorations shall be re-prorated after the Closing. As to power and utility charges, "final readings" as of the Closing Date shall be ordered from the utilities; the cost of obtaining such "final readings," if any, shall be paid by Purchaser.

(c) Sellers shall be entitled to all rents and other payments under Tenant Leases accruing for the period prior to the Effective Time ("**Pre Effective Time Lease Amounts**"), and

## Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 9 of 147

Purchaser shall be entitled to all rents and other payments under tenant leases accruing for the period after the Effective Time ("**Post Effective Time Lease Amounts**" and together with the Pre Effective Time Lease Amounts, the "**Lease Amounts**"). All Lease Amounts that are collected prior to the Closing shall be prorated as of the Closing in accordance with the immediately preceding sentence. All Lease Amounts that are accrued but unpaid as of the Closing (including, without limitation, rents and other payments accrued prior to the Closing but payable in arrears after the Closing) (collectively, the "**Unpaid Amounts**") shall belong to Sellers, and Purchaser shall, upon receipt of said rents and other payments, receive the same in trust for Sellers and shall promptly remit any of such amounts to Sellers within ten (10) days after Purchaser's receipt of same.

(d) If Purchaser requests that Sellers transfer electronic medical records in a specific electronic format due to the Sellers' electronic medical record ("EMR") system and Purchaser's EMR system not being interoperable, then Purchaser shall reimburse, on the Closing Date, all amounts paid or to be paid by Sellers to transfer electronic medical records to Purchaser in such different electronic format.

(e) All prorations and payments to be made under the foregoing provisions shall be agreed upon by Purchaser and Sellers prior to the Closing and shall be binding upon the Parties; *provided, however*, with respect to the Unpaid Amounts, in the event any proration, apportionment or computation shall prove to be incorrect for any reason, then either Sellers or Purchaser shall be entitled to an adjustment to correct the same, provided that said Party makes written demand on the Party from whom it is entitled to such adjustment within fifteen (15) calendar days after the erroneous payment or computation was made, or such later time as may be required, in the exercise of due diligence, to obtain the necessary information for proration.

(f) This <u>Section 1.6</u> shall survive Closing.

1.7 <u>Transfer of Sellers Assets</u>. On the Closing Date, subject to the terms and conditions of this Agreement and except as otherwise contemplated in <u>Section 1.3</u>, Sellers shall sell, assign, transfer, convey and deliver to Purchaser, free and clear of all interests, including but not limited to all liens, privileges, pledges, security interests, rights of first refusal, options, defects in title and encumbrances other than the Permitted Exceptions (defined below), and Purchaser shall acquire, all of Sellers' right, title and interest in and to only the following assets and properties, as such assets shall exist on the Closing Date, to the extent not included among the Excluded Assets, such transfer being deemed to be effective at the Effective Time (the "Assets"):

(a) all of the tangible personal property owned by Sellers, or to the extent assignable or transferable by Sellers, and used by Sellers in the operation of the Hospital, including equipment, furniture, fixtures, machinery, vehicles, office furnishings and leasehold improvements (the "**Personal Property**");

(b) subject to <u>Section 1.7(e)</u>, all of Sellers' rights, to the extent assignable or transferable, to all licenses, permits, approvals, certificates of exemption, franchises, accreditations and registrations and other governmental licenses, permits or approvals issued to Sellers for use in the operation of the Hospital (the "**Licenses**"), including, without limitation, the

## Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 10 of 147

Licenses set forth on <u>Schedule 1.7(b)</u>, except to the extent Purchaser elects, in its discretion, not to take assignment of any such Licenses;

(c) all of Sellers' interest in and to the Owned Real Property and all of Sellers' interest, to the extent assignable or transferable and that have been designated by Purchaser, in and to all of the following (the "Assigned Leases"): (i) the real property leases for all real property leased by Sellers and set forth on <u>Schedule 1.7(c)(i)</u> (the "Leased Real Property"), and (ii) the real property leased or subleased by Sellers to a third party and set forth on <u>Schedule 1.7(c)(i)</u> (the "Tenant Leases");

(d) all of Sellers' interest in, and all of Sellers' obligations due under, from and after the Effective Time, to the extent assignable or transferable, all contracts and agreements (including, but not limited to, purchase orders) that have been designated by Purchaser as an Assigned Contract, pursuant to <u>Section 1.11</u> and appearing on <u>Schedule 1.7(d)</u>;

(e) all of Sellers' interest in, from and after the Licensure Date, to the extent assignable or transferable, the Hospital's Medicare Provider Agreement (and provider number) and the Hospital's Medi-Cal Provider Agreement (and provider number) (collectively, the "Medicare/Medi-Cal Agreements");

(f) all of Sellers' interest in, and all of Sellers' obligations due under, from and after the Licensure Date, to the extent assignable or transferable, in and to any of the Hospital's managed care, pre-paid, capitated or other full-risk health plan agreements (collectively, the "Managed Care Agreements") that have been designated by Purchaser as an Assigned Contract pursuant to <u>Section 1.11</u> (to the extent so designated, the "Transferred Managed Care Agreements");

(g) subject to <u>Section 4.6</u>, all of Sellers' interest in, and all of Sellers' obligations due under, from and after the Licensure Date, to the extent assignable or transferable, in and to any of the Hospital's services, participation or provider agreements with private health plans, insurers or other third party payors (collectively, the "**Private Payor Agreements**") that have been designated by Purchaser as an Assigned Contract pursuant to <u>Section 1.11</u> (to the extent so designated, the "**Transferred Private Payor Agreements**"), *provided that*, Private Payor Agreements shall not include any "risk-sharing" agreements with independent physician associations;

(h) to the extent assignable or transferable, all inventories of supplies, drugs, food, janitorial and office supplies and other disposables and consumables (i) located at the Hospital or (ii) used in the operation of the Hospital (the "**Inventory**") except as set forth in <u>Sections 1.3</u> and <u>1.8(i)</u>;

(i) other than Utility Deposits, all prepaid rentals, deposits, prepayments and similar amounts relating to the Assigned Contracts and/or the Assigned Leases, which were made with respect to the operation of the Hospital (the "**Prepaids**");

(j) to the extent assignable or transferable, all of the following that are not proprietary to Sellers and/or owned by or proprietary to Sellers' affiliates: operating manuals, files

## Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 11 of 147

and computer software with respect to the operation of the Hospital, including, without limitation, all patient records, medical records, employee records, billing records, financial records, equipment records, construction plans and specifications, and medical and administrative libraries; *provided, however*, that any electronic medical records may be transferred in paper or "pdf" if Sellers' EMR system and Purchaser's EMR system are not interoperable;

(k) to the extent assignable or transferable, all rights in all warranties of any manufacturer or vendor in connection with the Personal Property;

(1) all right, title and interest in and to the name "St. Francis Medical Center," including any associated Hospital trademarks, service marks, trade names, logos and domain names but excluding the domain <u>https://stfrancis.verity.org</u> and content therein;

(m) all goodwill of the Hospital evidenced by the Assets;

(n) to the extent transferable or assignable, Sellers' right or interest in the telephone and facsimile numbers used with respect to the operation of the Hospital;

(o) to the extent assignable or transferable, Sellers' lock box account(s) associated with Medicare or Medi-Cal fee-for-service receivables (the "Lockboxes") on or after the Licensure Date;

(i) all accounts and interest thereupon, notes and interest thereupon and (p) other receivables of Sellers, including, without limitation, accounts, notes or other amounts receivable, and all claims, rights, interests and proceeds related thereto, including all accounts and other receivables, quality assurance fee payments, trauma payments, disproportionate share payments, cost report, claim, EHR or other similar appeals and Seller Cost Report settlements, in each case arising from the rendering of services or provision of goods, products or supplies to inpatients and outpatients at the Hospital, billed and unbilled, recorded and unrecorded (including any such amounts that were written-off by Sellers for any reason), for services, goods, products and supplies provided by Sellers prior to the Licensure Date whether payable by Medicare, Medi-Cal, Medicaid, or any other payor (including an insurance company), or any health care provider or network (such as a health maintenance organization, preferred provider organization or any other managed care program) or any fiscal intermediary of the foregoing, private pay patients, private insurance or by any other source (collectively, "Accounts Receivable"); and (ii) all claims, rights, interests and proceeds relating to any grant or governmental awards directly or indirectly related to COVID-19, trauma payments, disproportionate share payments, cost report, claim, EHR or other similar appeals and Seller Cost Report settlements in each case arising from the rendering of services or provision of goods, products or supplies to inpatients and outpatients at the Hospital (the "Other Receivables" and together with the Accounts Receivable, the "Receivables");

(q) all documents, records, correspondence, work papers and other documents, other than patient records, primarily relating to the <u>Accounts ReceivableReceivables</u>;

(r) all QAF payments due to the Hospital from the State of California or any of its administrative entities or other entitles, including without limitations, Medi-Cal managed care plans, received on and after the Licensure Date and attributable to QAF V, QAF VI and any

## Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 12 of 147

subsequent QAF Program, regardless of the state fiscal year for which the payments are made in reference to and regardless of the state fiscal year for which the data was derived to calculate eligibility for such payments, other than the QAF payments received prior to Closing or specifically excluded pursuant to <u>Section 1.8(b)</u> as QAF VI Seller Net Payments;

(s) except as set forth in <u>Section 1.8(1)</u> (*i.e.*, certain causes of action), all claims, causes of action, choses in action, rights of recovery, rights of set off and rights of recoupment of Sellers against third parties related to or associated with the physical condition of any of the Assets; and

(t) to the extent assignable or transferable, any other assets owned by Sellers (which are not otherwise specifically described above in this <u>Section 1.7</u>) that are used primarily in the operation of the Hospital.

As used herein, the term "**Permitted Exceptions**" means (i) the Transferred Obligations; (ii) liens for taxes not yet due and payable; (iii) easements, rights of way, zoning ordinances and other similar encumbrances affecting real property; (iv) imperfections of title or encumbrances identified in the Title Commitments (defined below) other than those specifically identified in <u>Schedule 1.7(I)</u>; and (iv) other imperfections of title or encumbrances, if any, which are not monetary in nature and that are not, individually or in the aggregate, material to the business of the Hospital.

1.8 <u>Excluded Assets</u>. Notwithstanding anything to the contrary in <u>Section 1.7</u>, Sellers shall retain all interests, rights and other assets owned directly or indirectly by it (or any of Sellers' affiliates) which are not among the Assets, including, without limitation, the following interests, rights and other assets of Sellers (collectively, the "**Excluded Assets**"):

(a) cash, cash equivalents and short-term investments;

(b) all QAF payments received prior to Closing under any QAF program and, in addition, with respect to the QAF VI Program, the Excluded Assets shall include the "QAF VI Seller Net Payments" which shall be, with respect to the Hospital, an amount equal to the product of: (A) all payments received or to be received by the Hospital under the QAF VI Program minus all payments already made, required to be made in the future or to be offset by the government with respect to the QAF VI Program, including payments to the California Health Foundation & Trust, multiplied by (B) the Closing Date Percentage. The "Closing Date Percentage" shall mean the percentage derived by dividing the total number of days between July 1, 2019 and the Effective Time, by 915;

(c) all Disproportionate Share Hospital Payments ("**DSH**") received on or after the Effective Time but calculated based on data from periods prior to the Effective Time (whether received before or after the Effective Time and whether paid to Sellers or Purchaser);

(d) all Sellers Plans (defined below) and the assets of all Sellers Plans and any asset that would revert to the employer upon the termination of any Sellers Plan, including, without limitation, any assets representing a surplus or overfunding of any Sellers Plan;

## Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 13 of 147

(e) all contracts that are not Assigned Contracts and all risk sharing agreements with independent physician associations ("**IPAs**");

(f) any Private Payor Agreement that is not a Transferred Private Payor Agreement and any Managed Care Agreement that is not a Transferred Managed Care Agreement;

(g) all collective bargaining agreements or other arrangements with unions representing Sellers' employees;

(h) all leases that have not been designated as Assigned Leases, and all Contracts that have not been designated as Assigned Contracts and all rents, deposits, prepayments, and similar amounts relating thereto;

(i) the portions of Inventory, Prepaids, and other assets disposed of, expended or canceled, as the case may be, by Sellers after the Signing Date and prior to the Effective Time in the ordinary course of business;

(j) assets owned by vendors of services or goods to the Hospital;

all of Sellers' organizational or corporate record books, minute books and

tax records;

(k)

Except as set forth in Section 1.7(k) or 1.7(s) hereof (*i.e.*, rights under (1)warranties and physical condition claims), all claims, counterclaims and causes of action of Sellers or Sellers' bankruptcy estate (including parties acting for or on behalf of Sellers' bankruptcy estate, including, but not limited to, the official committee of unsecured creditors appointed in the Bankruptcy Case), including, without limitation, (A) causes of action arising out of any claims and causes of action under chapter 5 of the Bankruptcy Code ("Avoidance Claims"), (B) any claims, counterclaims and causes of action under applicable non-bankruptcy law (including claims, counterclaims and causes of action against any health plan or other third party payors related to services provided prior to the Effective Time), and (C) any rights to challenge liens asserted against property of the Sellers' bankruptcy estate (including, but not limited to, liens attaching to the Purchase Price paid to the Sellers), and the proceeds from any of the foregoing; provided, however, that Purchaser shall acquire and be deemed to release and waive as of the Effective Time all Avoidance Claims against counterparties to Assigned Contracts and Assigned Leases solely to the extent such avoidance claims arise from, or are in connection with, executory contracts and unexpired leases assigned by the Sellers to Purchaser pursuant to Section 1.11 hereof;

(m) Except as set forth in <u>Section 1.7(s)</u>, all insurance policies and contracts and coverages obtained by Sellers or listing Sellers as insured party, a beneficiary or loss payee, including prepaid insurance premiums, and all rights to insurance proceeds under any of the foregoing, and all subrogation proceeds related to any insurance benefits arising from or relating to (i) Assets prior to the Licensure Date and/or (ii) Excluded Assets whether prior to or after the Licensure Date;

(n) all deposits made with any entity that provides utilities to the Hospital (the "Utility Deposits");

### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 14 of 147

(o) all unclaimed property of any third party as of the Effective Time, including, without limitation, property which is subject to applicable escheat laws;

(p) all bank accounts of Sellers, other than the Lockboxes as set forth in <u>Section</u> <u>1.7(o);</u>

(q) all writings and other items that are protected from discovery by the attorney-client privilege, the attorney work product doctrine or any other cognizable privilege or protection;

(r) the rights of Sellers to receive mail and other communications with respect to Excluded Assets or Excluded Liabilities;

(s) all tax refunds and tax assets of Sellers;

(t) all documents, records, operating manuals and film pertaining to the Hospital that the Parties agree that Sellers is required by law to retain;

(u) all patient records and medical records which are not part of any electronic medical record software transferred to Purchaser and are not required by law (including Section 351 of the Bankruptcy Code) to be maintained by Purchaser as of the Effective Time;

(v) any rights or documents relating to any Excluded Liability or other Excluded Asset;

(w) all deposits or other prepaid charges and expenses paid in connection with or relating to any other Excluded Assets and/or Excluded Liabilities;

(x) any rights or remedies provided to Sellers under this Agreement and each other document executed in connection with the Closing and the actions necessary to complete the sale of the Hospital pursuant to this Agreement;

(y) any rights or remedies, including deposits, against any individual or entity arising pursuant to (including in connection with Sellers' termination of) or relating to that certain Asset Purchase Agreement dated January 8, 2019 between, *inter alia*, Sellers and Strategic Global Management, Inc.;

(z) any (i) personnel files for employees of Sellers who are not hired by Purchaser; (ii) all documents, records, correspondence (including with respect to any employees), work papers, patient records or other books and records that Sellers is required by Law to retain; *provided, however*, that except as prohibited by Law and subject to <u>ARTICLE 5</u>, Purchaser shall have the right to make copies of any portions of such retained books and records that relate to the business of the Hospital as conducted before the Closing or that relate to any of the Assets; (iii) documents which Sellers is not permitted to transfer pursuant to any contractual obligation owed to any third party; (iv) documents primarily related to any Excluded Assets; and (v) documents necessary to prepare tax returns (Purchaser shall be entitled to a copy of such documents). With respect to documents necessary to prepare cost reports, Purchaser shall receive the original document and Sellers shall be entitled to retain or request a copy of such documents;

(aa) any and all rights and entitlements of Sellers in respect of that certain Settlement Agreement, executed as of April 29, 2019, by and between, on the one hand, Premier, Inc., Premier Services, LLC ("**Premier GP**"), Premier Healthcare Alliance, L.P. ("**Premier LP**"), Premier Healthcare Solutions, Inc. ("**PHSI**") and each of Premier, Inc.'s other subsidiaries (collectively and including Premier GP, Premier LP and PHSI, "**Premier**"), and on the other hand, VHS, as approved by the Bankruptcy Court by order entered on May 29, 2019 [Docket No. 2461], including but not limited to the right to convert and exchange partnership interests arising under that certain Amended and Restated Limited Partnership Agreement, effective as of October 1, 2013, as amended, by and among Premier LP, Premier GP and the limited partners of Premier LP party thereto (including VHS);

(bb) any grant or governmental awards received, or allocable to patient services performed, prior to the Effective Time; and

(bb) (cc)-any assets identified in <u>Schedule 1.8(cebb</u>).

For the avoidance of doubt, Purchaser is not acquiring any asset owned by any affiliate of Sellers.

1.9 <u>Transferred Obligations</u>. Purchaser is not assuming any liabilities of Sellers. Instead, on and after the Closing Date, Purchaser shall be responsible for and agrees to discharge, perform and satisfy fully, on and after the Effective Time, the following liabilities and obligations (collectively, the "**Transferred Obligations**"):

(a) the Assigned Contracts, after Sellers pay the Cure Costs from the proceeds of the Purchase Price;

(b) the Assigned Leases, after Sellers pay the Cure Costs from the proceeds of the Purchase Price;

(c) all liabilities and obligations arising out of or relating to any act, omission, event or occurrence connected with the use, ownership or operation by Purchaser of the Hospital or any of the Assets on or after the Effective Time;

(d) all liabilities and obligations related to the Hired Employees arising on or following the Effective Time (which for avoidance of doubt, does not include any duties, obligations or liabilities arising from or related to employment-related documentation required to be maintained by such Seller prior to the Effective Time, including but not limited to, documentation of I-9 compliance for Sellers' employees and any alternative work schedule compliance duties, obligations or liabilities that relate to Sellers' employees);

(e) all unpaid real and personal property taxes, if any, that are attributable to the Assets after the Effective Time, subject to the prorations provided in <u>Section 1.6;</u>

(f) all liabilities and obligations relating to utilities being furnished to the Assets, subject to the prorations provided in <u>Section 1.6</u>;

(g) any documentary, sales and transfer tax liabilities of Sellers incurred as a result of the consummation of the transaction contemplated by this Agreement;

- (h) all liabilities or obligations provided for in <u>Section 5.3</u>; and
- (i) any other obligations and liabilities identified in <u>Schedule 1.9(i)</u>.

1.10 <u>Excluded Liabilities</u>. Purchaser shall have those duties, obligations and liabilities set forth in this Agreement, the IMA, the Leaseback Agreement, the Transition Services Agreement, the Bill of Sale, the Transfer Agreement and the Real Estate Assignment(s) and shall be responsible for the Transferred Obligations. However, <u>except as expressly set forth herein</u>, Purchaser is not assuming any liabilities of Sellers related to the Assets, Sellers' employees or the Hospitals, is <u>purchasing the Assets free</u> and toclear of the maximum extent permitted by <u>lawEncumbrances except the Permitted Exceptions (defined herein) and</u> shall not be deemed a successor to Sellers or their estates by reason of any theory of law or equity with respect to any claims or liens against Sellers or the Assets (the "Excluded Liabilities").

Designation of Assigned Contracts and Assigned Leases. Each Seller will assign to 1.11 Purchaser the Assigned Leases and such other contracts and leases as are subject to evaluation by Purchaser for assumption or rejection (collectively "Evaluated Contracts"). At the later of (i) 5:00 p.m. pacific time on the day that is seven (7) calendar days prior to the date of the auction for the Assets and (ii) the date the Purchaser submits its bid for the Assets: (A) Purchaser shall notify Seller in writing signed and dated by Purchaser of which Evaluated Contracts are to be assigned to Purchaser (the "Assigned Contracts") and (B) Purchaser shall notify Seller in writing signed and dated by Purchaser of which Evaluated Contracts are to be rejected by Seller (the "Rejected Contracts"); provided, that Purchaser shall have the right to designate any contracts on the Assigned Contract list as a Rejected Contract on or prior to 5:00 p.m. pacific time on the day that is thirty (30) days prior to Closing Date, and Sellers shall have the absolute right to remove any Evaluated Contract from the list of Assigned Contracts in order to preserve avoidance claims; provided, however, that notwithstanding anything to the contrary, and to enable Sellers to comply with the terms of the Bidding Procedures Order with respect to UnitedHealthcare Insurance Company ("UnitedHealthcare"), Purchaser shall irrevocably designate all UnitedHealthcare agreements as Assigned Contracts or Rejected Contracts by the date that is two (2) calendars days after the Bankruptcy Court enters the Sale Order. The final list of Assigned Contracts at Closing will appear on Schedule 1.7(d).

# 1.12 Accounts Receivable Reconciliation.

(a) Sellers and Purchaser have mutually agreed that the target Accounts Receivable amount that will be transferred from Sellers to Purchaser as of the Effective Time is the A/R Target Amount. In connection with the Account Receivable reconciliation process, Purchaser shall collect the Accounts Receivable during the one hundred thirty-five (135) day period immediately following the Closing Date (collectively, the "Final A/R Collected"), and within the thirty (30) day period thereafter, Purchaser shall provide Sellers, in good faith, a schedule which provides an accounting of the Final A/R Collected (the "A/R Accounting Schedule"), together with reasonably detailed schedules and data supporting such accounting.

(b) After receipt of the A/R Accounting Schedule, Sellers shall have sixty (60) days to review the A/R Accounting Schedule and the Final A/R Collected as proposed by Purchaser, together with the work papers used in the preparation thereof, and have their

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 17 of 147

representatives and advisors review such A/R Accounting Schedule and proposed Final A/R Collected. In connection with the review of the A/R Accounting Schedule and proposed Final A/R Collected, Purchaser shall give, and shall cause its representatives and advisors to give, to Sellers and their representatives and advisors reasonable access, upon reasonable prior notice, to the books, records and other materials and the personnel of, and work papers prepared by or for, Purchaser and its representatives and advisors, including to such historical financial information relating to the Sellers and the Accounts Receivable as Sellers or their representatives or advisors may request, in each case, in order to permit the timely and complete review of the A/R Accounting Schedule and proposed Final A/R Collected in accordance with this <u>Section 1.12(b)</u> and so long as such access does not unreasonably interfere with the operations of the Purchaser.

(c) If Sellers disagree with the A/R Accounting Schedule and/or proposed Final A/R Collected, they shall notify Purchaser in writing within sixty (60) days after the date on which Purchaser delivers such A/R Accounting Schedule and proposed Final A/R Collected to Sellers, which shall include the items as to which they disagree and their calculation of such disputed amounts with reasonable supporting detail (the "<u>Statement of Objections</u>"). Purchaser and Sellers shall reasonably cooperate to resolve any such disagreements. If Purchaser and Seller are unable to resolve all such disagreements on or before the date which is thirty (30) days following notification by Sellers of any such disagreements, such disagreements shall be submitted to the Bankruptcy Court for resolution. The A/R Accounting Schedule and Final A/R Collected shall become final on the earlier of (i) failure by Sellers to deliver a Statement of Objections within the time period required by this section; (ii) mutual written agreement by Sellers and Purchaser; or (iii) a determination by the Bankruptcy Court.

(d) Once the A/R Accounting Schedule and Final A/R Collected have become final in accordance with Section 1.12(c), the following shall occur:

(i) if the Final A/R Collected is more than the A/R Target Amount, then such excess amount shall be paid by Purchaser to Sellers within ten (10) business days of Purchaser's delivery of the accounting of the Final A/R Collected to Sellers;

(ii) if the Final A/R Collected is less than the A/R Target Amount, then such deficit amount shall be paid by Sellers to Purchaser within ten (10) business days of Purchaser's delivery of the accounting of the Final A/R Collected to Sellers (and if not paid, Purchaser shall have collateral rights in Seller's QAF VI Seller Net Payments); or

(iii) if the Final A/R Collected is equal to the A/R Target Amount, then no adjusting payments in respect of the Accounts Receivable shall be required by either Purchaser or Sellers.

(e) During the one hundred thirty-five (135) day period immediately following the Closing Date Purchaser shall (i) use good faith, commercially reasonable best efforts to collect the Accounts Receivable (including at least the efforts used by Purchaser to collect its other receivables) within the one hundred thirty-five (135) day period immediately following the Closing Date; (ii) not take any action for the purpose of or which would be reasonably likely to result in any of the Accounts Receivable not being collected in a timely manner and within the one hundred thirty-five (135) day period immediately following the Closing Date; and (iii) provide

Sellers with weekly written updates on its collection of the Accounts Receivable.

## 1.13 Disclaimer of Warranties; Release.

THE ASSETS TRANSFERRED TO PURCHASER WILL BE SOLD BY (a) SELLERS AND PURCHASED BY PURCHASER IN THEIR PHYSICAL CONDITION AT THE EFFECTIVE TIME, "AS IS, WHERE IS AND WITH ALL FAULTS AND NONCOMPLIANCE WITH LAWS" WITH NO WARRANTIES, INCLUDING, WITHOUT LIMITATION, THE WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE, SUITABILITY, USAGE, WORKMANSHIP, QUALITY, PHYSICAL CONDITION, OR VALUE, AND ANY AND ALL SUCH OTHER REPRESENTATIONS AND WARRANTIES ARE HEREBY EXPRESSLY DISCLAIMED. AND WITH RESPECT TO THE LEASED REAL PROPERTY WITH NO WARRANTY OF HABITABILITY OR FITNESS FOR HABITATION, INCLUDING, WITHOUT LIMITATION, THE LAND, THE BUILDINGS AND THE IMPROVEMENTS. ALL OF THE PROPERTIES, RIGHTS. LICENSES. PERMITS, PRIVILEGES, ASSETS. LIABILITIES. AND OBLIGATIONS OF SELLERS INCLUDED IN THE ASSETS AND THE TRANSFERRED OBLIGATIONS ARE BEING ACQUIRED OR RECEIVED "AS IS, WHERE IS" ON THE CLOSING DATE AND IN THEIR PRESENT CONDITION, WITH ALL FAULTS. ALL OF THE TANGIBLE ASSETS SHALL BE FURTHER SUBJECT TO NORMAL WEAR AND TEAR AND NORMAL AND CUSTOMARY USE OF THE INVENTORY AND SUPPLIES IN THE ORDINARY COURSE OF BUSINESS UP TO THE EFFECTIVE TIME.

(b) Purchaser acknowledges that Purchaser has examined, reviewed and inspected all matters which in Purchaser's judgment bear upon the Purchase Price, the Assets, the Sellers, the Hospital, the businesses of the Hospital and their value and suitability for Purchaser's purposes and, except for Sellers representations and warranties in <u>ARTICLE 2</u>, is relying solely on Purchaser's own examination, review and inspection of the Assets and Transferred Obligations. Purchaser hereby releases Sellers and their affiliates from all responsibility and liability regarding the condition, valuation, salability or utility of the businesses of the Hospital or the Assets, or their suitability for any purpose whatsoever. Purchaser further acknowledges that the representations and warranties of Sellers contained in <u>ARTICLE 2</u> of this Agreement are the sole and exclusive representations and warranties made by Sellers to Purchaser (including with respect to the Hospital, the Assets and the Transferred Obligations) and shall expire, and be of no further force or effect at the Closing.

## ARTICLE 2 REPRESENTATIONS AND WARRANTIES OF SELLERS

Each Seller hereby represents, warrants and covenants to Purchaser, severally (and not jointly) with respect to such Seller that the following matters are true and correct as of the Signing Date, except as would not have a Material Adverse Effect (as defined below) upon the Assets, and except as disclosed in the disclosure schedule, as may be amended pursuant to the terms of this Agreement (the "**Disclosure Schedule**"), provided that the representations and warranties set forth in <u>Sections 2.1</u> (Authorization), <u>2.2</u> (Binding Agreement), <u>2.3</u> (Organization and Good Standing; No Violation), <u>2.7</u> (Compliance with Legal Requirements), <u>2.8</u> (Required Consents), <u>2.10</u> (Title)

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 19 of 147

and <u>2.13</u> (Legal Proceedings) (the "**Sale Order Date Representations**") shall also be made as of immediately prior to the entry of the Sale Order (the "**Sale Order Date**"):

The term "**Material Adverse Effect**" means any event, change or occurrence that, individually or in the aggregate with other events, changes or occurrences, has had or would reasonably be expected to have, a material adverse effect on the Sellers' financial condition, operation or Assets in excess of Fifty Million Dollars (\$50,000,000); provided, that the term Material Adverse Effect shall not include any event, change or occurrence resulting from: (a) the SARS-CoV-2 virus or mutations therefrom or in connection with the disease COVID-19; (b) changes to the U.S. economy, the global economy, in each case, as a whole, or the Hospital industry (except to the extent such changes disproportionately affect the Business); or (c) general economic, regulatory or political conditions or changes in the geographic regions in which the Hospital operates, except to the extent any such event described in (b), (c) or (d) disproportionately affects or impacts the Sellers' financial condition, operation or Assets as compared to other similarly situated businesses in the industry if such threshold is exceeded, then in lieu of Purchaser's right of termination, the Parties shall negotiate in good faith to reduce the Purchase Price as may be appropriate and any dispute will be resolved by the Bankruptcy Court as to the determination of any reasonable reduction.

2.1 <u>Authorization</u>. Such Seller has all necessary corporate power and authority to enter into this Agreement and, subject to Bankruptcy Court approval, to carry out the transactions contemplated hereby.

2.2 <u>Binding Agreement</u>. This Agreement has been duly and validly executed and delivered by Sellers and, assuming due and valid execution by Purchaser, this Agreement constitutes a valid and binding obligation of Sellers enforceable in accordance with its terms subject to (a) applicable bankruptcy, reorganization, insolvency, moratorium and other laws affecting creditors' rights generally from time to time in effect (including, without limitation approval of the Bankruptcy Court) and (b) limitations on the enforcement of equitable remedies.

# 2.3 Organization and Good Standing; No Violation.

(a) Each of SFMC and VHS is a non-profit corporation duly organized, validly existing and in good standing under the laws of the State of California. Verity Holdings is a limited liability company duly organized, validly existing and in good standing under the laws of the State of California. Each Seller has all necessary power and authority to own, operate and lease its properties and to carry on its businesses as now conducted.

(b) Neither the execution and delivery by such Seller of this Agreement nor the consummation of the transactions contemplated hereby by such Seller nor compliance with any of the material provisions hereof by such Seller, will violate, conflict with or result in a breach of any material provision of such Seller's articles of incorporation or bylaws or any other organizational documents of such Seller.

2.4 <u>Contracts</u>. Except as set forth in <u>Schedule 2.4</u>, upon entry of the Sale Order and payment of the Cure Costs, to Sellers' knowledge, Sellers are not in breach or default of the Assigned Contracts or Assigned Leases. No provision of this <u>Section 2.4</u> shall apply to any failure

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 20 of 147

to obtain consents to the assignment of the Assigned Contracts and/or Assigned Leases from third parties in connection with the assignment of such Assigned Contracts and/or Assigned Leases to Purchaser (the "**Contract and Lease Consents**").

2.5 <u>Brokers and Finders</u>. Except as set forth on <u>Schedule 2.5</u>, neither such Seller nor any affiliate thereof, nor any officer or director thereof, have engaged or incurred any liability to any finder, broker or agent in connection with the transactions contemplated hereunder.

2.6 <u>Non-Contravention</u>. Neither the execution and delivery by Sellers of this Agreement and each Ancillary Agreement nor performance of any of the material provisions hereof by Sellers, will violate, conflict with or result in a breach of any material provisions of the articles of incorporation or bylaws of Sellers.

2.7 <u>Compliance with Legal Requirements</u>. Except as set forth in <u>Schedule 2.7</u>, to the knowledge of Sellers: each Seller, with respect to the operation of the Hospital, is in material compliance with all applicable laws, statutes, ordinances, orders, rules, regulations, licenses, certificates, judgments or decrees of all judicial or governmental authorities (federal, state, local, foreign or otherwise) (collectively, "**Legal Requirements**"). Except as set forth in <u>Schedule 2.7</u>, to the knowledge of Sellers, none of the Sellers, with respect to the operation of the Hospital, has been charged in writing with or been given written notice of any material violation or any obligation to take material remedial action under, any applicable Legal Requirements.

2.8 <u>Required Consents</u>. Except as set forth in <u>Schedule 2.8</u>, and other than in connection with any Licenses, any provider agreements (including any such agreements with a governmental authority) and the Attorney General, Sellers are not a party to or bound by, nor are any of the Assets subject to, any mortgage, or any material lien, deed of trust, material lease, or material contract or any material order, judgment or decree which, after giving effect to the Sale Order (a) will require the consent of any third party to the execution of this Agreement or (b) will require the consent of any third party to consummate the transactions contemplated by this Agreement.

# 2.9 Environmental Matters

(a) Sellers have provided Purchaser with the Phase I Environmental Site Assessments in Seller's electronic data room.

(b) Except as disclosed in <u>Schedule 2.9(b)</u>, to the knowledge of Sellers, the operations of the Hospital are not in material violation of any applicable limitations, restrictions, conditions, standards, prohibitions, requirements and obligations of Environmental Laws and related orders of any court or any other governmental authority.

(c) For the purposes of this Section, the term "**Environmental Laws**" shall mean all state, federal or local laws, ordinances, codes or regulations relating to Hazardous Substances or to the protection of the environment, including, without limitation, laws and regulations relating to the storage, treatment and disposal of medical and biological waste. For purposes of this Agreement, the term "**Hazardous Substances**" shall mean (i) any hazardous or

### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 21 of 147

toxic waste, substance, or material defined as such in (or for the purposes of) any Environmental Laws, (ii) asbestos-containing material, (iii) medical and biological waste, (iv) polychlorinated biphenyls, (v) petroleum products, including gasoline, fuel oil, crude oil and other various constituents of such products, and (vi) any other chemicals, materials or substances, exposure to which is prohibited, limited or regulated by any Environmental Laws.

2.10 <u>Title</u>. Sellers have delivered at their own expense for all Owned Real Property (i) preliminary title reports issued by Chicago Title Insurance Company (the "**Title Commitments**") and (ii) copies of or access to all material underlying title documents listed on the Title Commitments.

## 2.11 <u>Certain Other Representations with Respect to the Hospitals</u>.

(a) Except as set forth in <u>Schedule 2.11</u>, all Licenses which are material and necessary to the operation of the Hospital by SFMC are valid and in good standing and SFMC is in compliance with the terms and conditions of all such Licenses in all material respects, in each case except where the failure to be valid and in good standing or in compliance would not have a Material Adverse Effect on the Assets. Except as set forth in <u>Schedule 2.11</u>, as of the Closing Date, SFMC will have all material Licenses required under Legal Requirements to operate the Hospital as presently operated by SFMC, except where the failure to have any such License would not have a Material Adverse Effect on the Assets. To the knowledge of Sellers, no loss of any License is pending or threatened.

(b) SFMC is certified for participation in the Medicare, Medi-Cal and TRICARE programs and any other federal or state health care reimbursement programs in which SFMC participates, and has current and valid provider agreements with each such program, except where the failure to be so certified or have such provider agreements would not have a Material Adverse Effect.

(c) SFMC has not been excluded from Medicare, Medi-Cal or any federal or state health care reimbursement program, and, to the knowledge of Sellers, there is no pending or threatened exclusion action by a governmental authority against SFMC.

# 2.12 Financial Statements.

(a) Sellers have provided to Purchaser: (i) the unaudited balance sheets of the Sellers as of June 30, 2018; (ii) unaudited income statements of the Sellers for the twelve-month periods ended June 30, 2018; (iii) the audited consolidated income statements of Sellers for the years ended 2016 and 2017; and (iv) the unaudited consolidated balance sheet of Sellers as of June 30, 2018 (collectively, the "**Historical Financial Statements**").

(b) the income statements contained in the Historical Financial Statements present, fairly in all material respects the results of the operations of the Sellers as of and for the periods covered therein and the balance sheets contained in the Historical Financial Statements (i) are true, complete and correct in all material respects; (ii) present, fairly in all material respects the financial condition of the Sellers as of the dates indicated thereon; and (iii) to the extent prepared

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 22 of 147

by an independent certified public accounting firm, have been prepared in accordance with generally accepted accounting principles consistently applied throughout the periods covered, except as disclosed therein.

Legal Proceedings. Except as set forth on Schedule 2.13, and except for any and all 2.13 cases and/or pleadings filed or to be filed in the Bankruptcy Court, which shall be available Sellers' through claims and noticing agent's website at http://www.kccllc.com/VERITYHEALTH/, to the knowledge of Sellers, there are no material claims, proceedings or investigations pending or threatened with respect to the ownership of the Assets or the operation of the Hospital or the Hospital by Sellers before any governmental authority. Except as set forth on Schedule 2.13, and other than any action or proceeding brought in the Bankruptcy Court, to the knowledge of Sellers, Sellers are not subject to any government order with respect to the ownership or operation by SFMC of the Hospital and SFMC is in substantial compliance with respect to any such government order.

2.14 <u>Employee Benefits</u>. Sellers have provided Purchaser with a list of (i) each pension, profit sharing, bonus, deferred compensation, or other retirement plan or arrangement of Seller with respect to the operation of the Hospital, whether oral or written, which constitutes an "employee pension benefit plan" as defined in Section 3(2) of ERISA, (ii) each medical, health, disability, insurance or other plan or arrangement of Seller with respect to the operation of the Hospital, whether oral or written, which constitutes an "employee welfare benefit plan" as defined in Section 3(1) of ERISA, and (iii) each other employee benefit or perquisite provided by Seller with respect to the operation of the Hospital, in which any employee of Seller participates in his/her capacity as such (collectively, the "Seller Plans").

2.15 <u>Personnel</u>. Sellers have provided Purchaser with a complete list (as of the date set forth therein) of names, positions and current annual salaries or wage rates and scheduled bonus, and the accrued paid time off pay of all employees of Sellers (including employees of the Hospital and employees of VHS and Verity Holdings) immediately prior to [February 29March 31, 2020], whether such employees are full time employees, part-time employees, on short-term or long-term disability or on leave of absence pursuant to Sellers's policies, the Family and Medical Leave Act of 1993 or other similar Legal Requirements (the "Hospital Employees") and indicating whether the Hospital Employee is full- time or part-time. Sellers shall provide an updated list to Purchaser no later than five (5) Business Days before the date scheduled for the Closing.Insurance. Sellers have provided Purchaser with a list of all material insurance maintained by Sellers with respect to the Assets, as of the Signing Date.

2.17 <u>Accounts ReceivableReceivables</u>. To the knowledge of Sellers, all <u>Accounts</u> <u>ReceivableReceivables</u> included in the Assets at Closing result from the bona fide provision of products or services in the ordinary course of business. All proceeds of Sellers' <u>Accounts</u> <u>ReceivableReceivables</u> are currently deposited, either electronically or manually, into those bank accounts provided to Purchaser in Seller's electronic data room.

2.18 <u>Payor Contracts</u>. To the knowledge of Sellers, and subject to Section 365 of the Bankruptcy Code, Sellers have provided Purchaser with a complete list of all written contracts with private third-party payors including insurance companies and HMOs. Sellers have provided Purchaser with a true and correct copy of all material Payor Contracts.

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 23 of 147

2.19 <u>Excluded Individuals</u>. Except as set forth on <u>Schedule 2.19</u>, to the knowledge of Sellers: neither Sellers, Hospital nor any director, officer or employee of Sellers or Hospital (a) was, is or is proposed to be, suspended, excluded from participation in, or sanctioned under, any federal or state health care program (including, without limitation, Medicare and Medicaid) (an "**Excluded Individual**"); (b) has been convicted of any criminal offense related to the delivery of any medical or health care services or supplies, or related to the neglect or abuse of patients; (c) has failed to maintain its current License to provide the services required to be provided by it to or on behalf of Sellers and Hospital; or (d) is unable to obtain or maintain liability insurance consistent with commercially reasonable industry practices.

2.20 <u>Seller Knowledge</u>. References in this Agreement to "Sellers' knowledge" or "the knowledge of Sellers" means the actual knowledge of the Chief Executive Officer and Chief Financial Officer of the applicable Seller, without independent research. No constructive or imputed knowledge shall be attributed to any such individual by virtue of any position held, relationship to any other Person or for any other reason.

### ARTICLE 3 REPRESENTATIONS AND WARRANTIES OF PURCHASER

As an inducement to Sellers to enter into this Agreement and to consummate the transactions contemplated by this Agreement, Purchaser hereby represents and warrants to Sellers as to the following matters as of the Signing Date and as of the Closing Date:

3.1 <u>Authorization</u>. Purchaser has full power and authority to enter into this Agreement and has full power and authority to perform its obligations hereunder and to carry out the transactions contemplated hereby. No additional internal consents are required in order for Purchaser to perform its obligations and agreements hereunder.

3.2 <u>Binding Agreement</u>. This Agreement has been duly and validly executed and delivered by Purchaser and, assuming due and valid execution by Sellers, this Agreement constitutes a valid and binding obligation of Purchaser enforceable in accordance with its terms subject to (a) applicable bankruptcy, reorganization, insolvency, moratorium and other laws affecting creditors' rights generally from time to time in effect, and (b) limitations on the enforcement of equitable remedies.

3.3 <u>Organization and Good Standing</u>. Purchaser is a corporation duly organized, validly existing and in good standing under the laws of the State of California, is or will be duly authorized to transact business in the State of California, and has full power and authority to own, operate and lease its properties and to carry on its business as now conducted.

3.4 <u>No Violation</u>. Except as set forth in <u>Schedule 3.4</u>, neither the execution and delivery by Purchaser of this Agreement nor the consummation of the transactions contemplated hereby nor compliance with any of the material provisions hereof by Purchaser will (a) violate, conflict with or result in a breach of any material provision of the Articles of Incorporation, Bylaws or other organizational documents of Purchaser or any contract, lease or other instrument by which Purchaser is bound; (b) require any approval or consent of, or filing with, any governmental agency or authority, (c) violate any law, rule, regulation, or ordinance to which

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 24 of 147

Purchaser is or may be subject, or (d) violate any judgment, order or decree of any court or other governmental agency or authority to which Purchaser is subject.

3.5 <u>Brokers and Finders</u>. Neither Purchaser nor any affiliate thereof nor any officer or director thereof has engaged any finder or broker in connection with the transactions contemplated hereunder.

3.6 <u>Representations of Sellers</u>. Purchaser acknowledges that it is purchasing the Assets on an "AS IS, WHERE IS" basis (as more particularly described in <u>Section 1.12</u>), and that Purchaser is not relying on any covenant, representation, warranty (expressed or implied, oral or otherwise) made by or on behalf of Sellers, or any other provision of this Agreement, in entering into and performing under this Agreement. Purchaser further acknowledges that Sellers are not making any covenants, representations or warranties herein relating to the Assets or the operation of the Hospital on or after the Effective Time.

3.7 <u>Legal Proceedings</u>. Except as described on <u>Schedule 3.7</u>, there are no claims, proceedings or investigations pending or, to the best knowledge of Purchaser, threatened relating to or affecting Purchaser or any affiliate of Purchaser before any court or governmental body (whether judicial, executive or administrative) in which an adverse determination would adversely affect Purchaser's ability to consummate the transactions contemplated hereby. Neither Purchaser nor any affiliate of Purchaser is subject to any judgment, order, decree or other governmental restriction specifically (as distinct from generically) applicable to Purchaser or any affiliate of Purchaser which would adversely affect Purchaser's ability to consummate the transactions contemplated hereby.

No Knowledge of Sellers' Breach. Neither Purchaser nor any of its affiliates has 3.8 knowledge of any breach of any covenant, representation or warranty by Sellers or of any condition or circumstance that would give Purchaser a right to terminate this Agreement pursuant to Section 9.1(c). If information comes to Purchaser's attention on or before the Closing Date (whether through Sellers or otherwise and whether before or after the Signing Date) which indicates that Sellers have breached any of their covenants, representations, warranties or any other provision or condition under this Agreement, then the effect shall be as if the covenants, representations and warranties or any other provision or condition of this Agreement had been modified in accordance with the actual state of facts existing prior to the Effective Time such that there will be no breach under Sellers' covenants, representations and warranties or any other provision or condition of this Agreement in relation to such information; provided, further, that Purchaser must immediately notify Sellers if any such breach comes to its attention on or before the Closing Date, and Purchaser's failure to so notify Sellers shall constitute a waiver by Purchaser of Sellers' breach, if any, of this Agreement or any ancillary agreements entered into pursuant to this Agreement. Upon written request of Sellers, Purchaser shall promptly confirm and remake this representation in writing.

3.9 <u>Ability to Perform.</u> Purchaser has the ability to obtain funds and at the Closing shall have cash in amounts necessary to consummate the transactions contemplated by this Agreement by means of cash, credit facilities or otherwise. Upon the earlier of ten (10) days after satisfaction of <u>Section 8.3</u> or five (5) days prior to the Closing Date, Purchaser shall have delivered to Sellers true, correct and complete copies of (i) executed commitment letter(s) from one or more

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 25 of 147

banks or other lending institutions or sources (the "**Debt Commitment Letter**"), pursuant to which, and subject to the terms and conditions thereof, the lender parties thereto have committed to lend the amounts set forth therein to Purchaser for the purpose of funding the transactions contemplated by this Agreement and (ii) any executed equity commitment letter(s) (the "**Equity Commitment Letter**" and, together with the Debt Commitment Letter, the "**Financing Commitments**"). The Financing Commitments provide sufficient funds to consummate the transactions contemplated by this Agreement.

3.10 <u>Investigation</u>. Purchaser has been afforded reasonable access to, and has been provided adequate time to review, the books, records, information, operations, facilities and personnel of Sellers and the Hospital for purposes of conducting a due diligence investigation of Sellers and the Hospital. Purchaser has conducted a reasonable due diligence investigation of Sellers and the Hospital and has received satisfactory answers to all inquiries it has made respecting Sellers and the Hospital and has received all information it considers necessary to make an informed business evaluation of Sellers and the Hospital. Purchaser has not relied upon any books, records, information, operations, facilities and personnel provided by Sellers, including in making its determination to enter into this Agreement and/or consummate the transactions contemplated hereby. Purchaser has completed all of its due diligence of Sellers and the Hospital and this Agreement is not subject to any further due diligence of Sellers and the Hospital by Purchaser.

3.11 <u>Purchaser Knowledge</u>. References in this Agreement to "Purchaser's knowledge" or "the knowledge of Purchaser" means the actual knowledge of the Chief Executive Officer and Chief Financial Officer of Purchaser, without independent research. No constructive or imputed knowledge shall be attributed to any such individual by virtue of any position held, relationship to any other Person or for any other reason.

### ARTICLE 4 COVENANTS OF SELLERS

## 4.1 <u>Access and Information; Inspections</u>.

From the Signing Date through the Effective Time, (a) Sellers shall afford (a) to the officers and agents of Purchaser (which shall include accountants, attorneys, bankers and other consultants and authorized agents of Purchaser) reasonable access during normal business hours at Sellers' corporate headquarters in Los Angeles, California to, and the right to inspect, the books, accounts, records and all other relevant documents and information with respect to the assets, liabilities and business of the Hospital and the plant and property of the Hospital at the Hospital and (b) Sellers shall furnish Purchaser with such additional financial and operating data and other information in Sellers' possession as to the businesses and properties of the Hospital as Purchaser or its representatives may from time to time reasonably request; provided, however, that Sellers are not obligated to disclose information which is proprietary to Sellers and would not be essential to the ongoing operation of the Hospital by Purchaser; provided, further, that all disclosures of information shall be consistent with the confidentiality agreements and any other non-disclosure agreements entered into (or to be entered into) among Purchaser, its representatives and Sellers or their representatives. Purchaser's right of access and inspection shall be exercised in such a manner as not to interfere unreasonably with the operations of Sellers or the Hospital.

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 26 of 147

(b) Notwithstanding anything contained herein, Sellers shall not be required to provide Purchaser or its representatives or agents access to or disclose information where such access or disclosure would violate the rights of its patients, jeopardize the attorney-client or similar privilege with respect to such information or contravene any law, judgment, fiduciary duty or contract entered into prior to or on the date of this Agreement with respect to such information.

## 4.2 <u>Cooperation</u>.

(a) The Parties shall reasonably cooperate with each other and their respective authorized representatives and attorneys in: (a) all efforts to obtain all consents, approvals, authorizations, clearances and licenses required to carry out the transactions contemplated by this Agreement (including, without limitation, those of governmental and regulatory authorities), (b) the preparation of any document or other material which may be required by any governmental agency as a predicate to or result of the transactions contemplated in this Agreement, and (c) Purchaser's efforts to effectuate the assignment of Assigned Contracts to Purchaser as of the Closing Date.

(b) Except as may be otherwise requested by Sellers in order to comply with applicable law or regulatory guidance, notwithstanding anything contained herein, other than Bankruptcy Court orders and authorizations, it shall be Purchaser's sole responsibility (including payment of any fees, expenses, filing costs or other amounts) to obtain the Contract and Lease Consents, as well as all governmental consents, approvals, assignments, authorizations, clearances and licenses required to (x) carry out the transactions contemplated by this Agreement and/or (y) transfer any of the Assets, including any Licenses, Transferred Managed Care Agreements and/or Transferred Private Payor Agreements, not including the application required to be submitted to the California Attorney General and the transfer of the Medicare Provider Agreement and the Medi-Cal Provider Agreement, which are dealt with in Sections 5.9(c) and 4.5 hereto respectively. Upon request, Sellers shall provide Seller-specific information that may be needed by Purchaser to obtain such Contract and Lease Consents and such governmental consents, approvals, assignments, authorizations, clearances and licenses.

(c) Notwithstanding any provision to the contrary contained in this Agreement, Sellers shall not be obligated to obtain the approval or consent to the assignment, to Purchaser, of any Assigned Contracts or Assigned Leases, from any party to any of the Assigned Contracts or Assigned Leases even if any such contract or lease states that it is not assignable without such party's consent.

4.3 <u>Sellers' Efforts to Close</u>. Sellers shall use their reasonable commercial efforts to satisfy all of the conditions precedent set forth in <u>ARTICLE 7</u> and <u>ARTICLE 8</u> to their or Purchaser's obligations under this Agreement to the extent that Sellers' action or inaction can control or materially influence the satisfaction of such conditions; *provided, however*, that Sellers shall not be required to pay or commit to pay any amount to (or incur any obligation in favor of) any person (other than any filing or application fees to the extent Sellers are responsible therefor under this Agreement).

4.4 <u>Termination Cost Reports</u>. Sellers shall file all Medicare, Medi-Cal and any other termination cost reports required to be filed as a result of the consummation of (a) the transfer of

the Assets to Purchaser and (b) the transactions contemplated by this Agreement, provided that Purchaser shall fund reasonable costs and expenses of preparation, filing and audit of such reports. Purchaser shall permit Sellers access to all Hospital books and records to prepare such reports and shall assist Sellers in the process of preparing, filing, and reviewing the termination cost reports. All such termination cost reports shall be filed by Sellers in a manner that is consistent with current laws, rules and regulations. Sellers shall be responsible for filing governmental cost reports for all periods through the day immediately preceding the Licensure Date. Purchaser shall be responsible for their own cost report filings relating to the Hospital beginning on the Licensure Date.

4.5 <u>Medicare and Medi-Cal Enrollment</u>. Sellers shall make commercially reasonable efforts to <u>expeditiously</u> enter into settlement agreements with the Centers for Medicare and Medicaid Services ("**CMS**"), with respect to the Medicare Provider Agreement, and the California Department of Health Care Services ("**DHCS**"), with respect to the Medi-Cal Provider Agreement or, alternatively, obtain Bankruptcy Court rulings <u>and Final Orders</u> that the Medicare Provider Agreement and/or the Medi-Cal Provider Agreement may be transferred without the consent of CMS or DHCS, as applicable, <u>free and clear of Encumbrances and Excluded Liabilities</u> and without successor liability, to enable such agreements to be assigned to Purchaser. Between the Closing Date and the Licensure Date, Purchaser may bill and collect for patient services under Seller's health plan agreements, pursuant to the terms of the IMA and Leaseback Agreement.

4.6 Transferred Private Payor Agreements. For purposes of this Section 4.6, the following terms shall have the meanings set forth herein: (a) "Payor" means the non-debtor counter-party to a Transferred Private Payor Agreement; (b) "Overpayment" means an overpayment made by a Payor to SFMC on account of a Contracted Payment; (c) "Contracted Payment" means the contractual reimbursement due from a Payor to SFMC under a Transferred Private Payor Agreement for covered services rendered by Hospital under such agreement with a date of service on or prior to the Effective Time. Purchaser agrees that, as a condition to the assignment of any Transferred Private Payor Agreement on the Licensure Date, it shall honor any defenses to the payment of, and shall permit recoupment against, an Account Receivable due from Payor to Purchaser based on an Overpayment under such agreement, provided that, the determination and allowance of any Overpayment shall remain subject to the terms and conditions of the relevant Transferred Private Payor Agreement. Without limiting the scope of the foregoing sentence, Purchaser acknowledges and agrees that (a) the Sale Order shall authorize a Payor to continue to exercise its defenses to the payment to Purchaser of an Account Receivable based on an Overpayment, and (b) Overpayments do not constitute Cure Costs under this Agreement. This Section 4.6 shall satisfy the requirements under Sections 365(b) and (f) of the Bankruptcy Code. Following the Licensure Date, pursuant to Section 365(k) of the Bankruptcy Code, the SFMC shall be relieved and released from any obligation to any Payor on account of any Overpayment or otherwise.

4.7 <u>Hospital Operations</u>. From the date of the Sale Order until the Closing, Sellers shall, with respect to the operation of the Hospital, use commercially reasonable efforts (in each case subject to actions relating to and impacts arising from the SARS-CoV-2 virus or mutations therefrom or in connection with the disease COVID-19) to:

(a) without regard to negative financial impacts or any Material Adverse Effect, carry on Sellers' operation of the Hospital consistent with past practice, but subject to the

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 28 of 147

Bankruptcy Cases and Sellers' obligations and actions in connection therewith;

(b) without regard to Material Adverse Effect, maintain in effect the insurance coverages with respect to the Assets;

(c) without regard to Material Adverse Effect, perform Sellers' material obligations under all Assigned Contracts with respect to the Assets in compliance with the Bankruptcy Code;

(d) with respect to material deficiencies, if any, cited by any governmental authority or accreditation body in the most recent surveys conducted by each, develop and implement a plan of correction (without regard to Material Adverse Effect) that is reasonably acceptable to such governmental authority or such accreditation body, but excluding any deficiencies cited (i) by the Attorney General or (ii) with respect to seismic or OSHPD-related requirements; *provided, however*, that Sellers may appeal any material deficiency citations under applicable laws if a plan of correction is not accepted by the applicable governmental authority or accreditation body which appeal shall satisfy Sellers obligations under this subsection;

(e) following entry of the Sale Order, permit and allow reasonable access by Purchaser and its representatives to make offers of post-Closing employment to any of Sellers' personnel and to establish relationships with physicians, medical staff and others having business relations with Sellers, provided, that any written materials shall be approved by Sellers prior to being sent, and provided further that such actions by Purchaser do not unreasonably interfere with Sellers' operation of the Hospital;

(f) timely file or cause to be filed all material reports, notices and tax returns required to be filed and pay all required taxes as they come due, unless such failure would not have a Material Adverse Effect; and

(g) without regard to Material Adverse Effect, maintain all existing material approvals, permits and environmental permits relating to the Hospital.

4.8 <u>Interim Collections</u>. Purchaser may bill and collect for patient services rendered by Hospital between the Closing Date and the Licensure Date under Seller's Private Payor Agreements and Transferred Managed Care Agreements, pursuant to the terms of the IMA and Leaseback Agreement.

4.9 <u>Contract With Unions</u>. Representatives of Sellers who are parties to collective bargaining agreements and Purchaser shall meet and confer from time to time as reasonably requested by either Party to discuss strategic business options and alternative approaches in negotiating each collective bargaining agreement. The applicable Sellers and Purchaser shall each participate in all union negotiations related to any specific collective bargaining agreement. Promptly following the Signing Date, applicable Sellers shall use commercially reasonable efforts to initiate discussions with Purchaser and conduct discussions to renegotiate each collective bargaining agreement currently in effect with each applicable union. The applicable Sellers will not unreasonably withhold, condition or delay approval or implementation of any successfully renegotiated collective bargaining agreement. The Parties recognize that Seller's failure to

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 29 of 147

conclude a successor collective bargaining agreement shall not be a breach of Sellers' obligation under this Agreement, provided that if the unions refuse to negotiate, or otherwise are not timely, reasonable or realistic in negotiating modified or new collective bargaining agreements during the period between the Signing Date and the Closing Date, Sellers and Purchaser will jointly consider, and negotiate mutually in good faith, alternative approaches that may be available and/or necessary to reduce Sellers' labor cost structure, including, but not limited to, seeking to reject the existing collective bargaining agreement(s).

4.10 <u>Consulting Services</u>.<sup>+</sup> Subject to compliance with applicable Legal Requirements and after entry of the Sale Order if the Purchaser is the Prevailing Bidder:

(a) Purchaser shall provide consulting services to the Hospital for the period prior to the Closing Date;

(b) the consulting services provided by Purchaser may include, but not be limited to, reviewing and advising SFMC regarding accounting and financial records, contracting, billing and collection activities, compliance with law, any of the plans or actions proposed in the following sentence, and other functions;

(c) Sellers shall consider and implement, at no cost to Sellers, any mutually reasonably agreed upon plans and actions proposed by Purchaser to (a) assist the Hospital in meeting any seismic compliance deadlines, (b) stabilize and improve the operations of the Hospital or (c) develop and implement turnaround plans for the long-term viability of the Hospital; and

(d) Purchaser shall have reasonable access to the business office and records of the Hospital, which shall include reasonable access to the Hospital's chief executive officer, chief financial officer, chief operations officer or other equivalent personnel of the Hospital reasonably necessary for Purchaser to perform the aforementioned consulting services.

Notwithstanding the foregoing, VHS, SFMC and their boards of directors shall at all times retain ultimate control and governance over the assets and operation of the Hospital. Neither VHS nor SFMC shall delegate to Purchaser any of the powers, duties and responsibilities required to own or operate the Hospital or those that are retained by VHS and/or SFMC under law (including all certificates and licenses issued under authority of law for ownership or operation of the Hospital). For the period of time during which such consulting services are provided, Hospital and VHS shall have the non-exclusive, royalty-free, unlimited license to use any implemented Purchaser-owned intellectual property. <u>Purchaser shall not charge Sellers any consulting fee for providing the services described in this Section 4.11</u>.

<u>4.11</u> Cure Costs. On or about the Closing Date, Sellers (from the proceeds of the Purchase Price) shall pay an amount equal to the Cure Costs to each counter party to an Assigned Contract and Assigned Lease so that each such Assigned Contract and Assigned Lease may be assumed by Sellers and assigned to Purchaser in accordance with the provisions of Section 365 of the Bankruptcy Code. For purposes of this Agreement, "Cure Costs", means all amounts that

<sup>&</sup>lt;sup>4</sup> To be discussed subject to review by Verity and BRG.

Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 30 of 147

must be paid and all obligations that otherwise must be satisfied, including pursuant to Sections 365(b)(1)(A) and (B) of the Bankruptcy Code in connection with the assumption and/or assignment of the Assigned Contracts and Assigned Leases to Purchaser as provided herein.

# ARTICLE 5 COVENANTS OF PURCHASER

5.1 <u>Purchaser's Efforts to Close</u>. Purchaser shall use its reasonable commercial efforts to satisfy all of the conditions precedent set forth in <u>ARTICLE 7</u> and <u>ARTICLE 8</u> to its or Sellers' obligations under this Agreement to the extent that Purchaser's action or inaction can control or materially influence the satisfaction of such conditions.

5.2 <u>Consents</u>. Purchaser shall be entitled, but not obligated, to seek to obtain the Contract and Lease Consents. Purchaser shall be entitled, but not obligated, to solicit and seek to obtain estoppel certificates from any third party to any Leased Real Property. Purchaser's failure to obtain any or all of the Contract and Lease Consents or estoppel certificates as of the Closing Date shall not be a condition precedent to either Party's obligation to consummate the Closing and perform all transactions contemplated by this Agreement.

# 5.3 <u>Certain Employee Matters</u>.

(a) Purchaser agrees to make offers of employment, effective as of the Effective Time, to substantially all persons (whether such persons are full time employees, part-time employees, on short-term or long-term disability or on leave of absence, military leave or workers compensation leave) who, immediately prior to the Effective Time are: (i) employees of SFMC; or (ii) employed by another Seller or affiliate and are listed on <u>Schedule 5.3</u> (collectively, the "**Hospital Employees**"). For the avoidance of doubt, the Hospital Employees shall not include any employees of VHS or any other affiliate of Sellers unless such individual is listed on <u>Schedule 5.3</u>. Any of the Hospital Employees who accept an offer of employment with Purchaser as of or after the Effective Time shall be referred to in this Agreement as the "**Hired Employees**." All employees who are Hired Employees shall cease to be employees of Sellers or its affiliates as of the Closing Date.

(b) After the Closing Date, Purchaser's human resources department will give reasonable assistance to Sellers and their affiliates with respect to Sellers' and Sellers' affiliates' post-Closing administration of Sellers' and Sellers' affiliates' pre-Closing employee benefit plans for the Hospital Employees. Within five (5) days after the Closing Date, Purchaser shall provide to Sellers a list of all the Hospital Employees who were offered employment by Purchaser but refused such employment along with a list of all Hired Employees (which such list Purchaser shall periodically update).

(c) With respect to any collective bargaining agreements or labor contract with respect to any union employees, Purchaser shall comply with the applicable laws, or to the extent applicable, Bankruptcy Court orders relating to collective bargaining agreements or labor contracts.

(d) The provisions of this <u>Section 5.3</u> are solely for the benefit of the Parties, and no employee or former employee or any other individual associated therewith or any employee benefit plan or trustee thereof shall be regarded for any purpose as a third party beneficiary of this Agreement, and nothing herein shall be construed as an amendment to any employee benefit plan for any purpose.

5.4 <u>Waiver of Bulk Sales Law Compliance</u>. Purchaser hereby waives compliance by Sellers with the requirements, if any, of Article 6 of the Uniform Commercial Code as in force in any state in which the Assets are located and all other laws applicable to bulk sales and transfers.

5.5 <u>Conduct Pending Closing</u>. Prior to consummation of the transactions contemplated hereby or the termination or expiration of this Agreement pursuant to its terms, unless Sellers shall otherwise consent in writing, Purchaser shall not take any action or fail or omit to take any action which would cause any of Purchaser's representations and warranties set forth in <u>ARTICLE 4</u> to be inaccurate or untrue during the term of this Agreement and as of the Closing.

5.6 <u>Resale Certificate</u>. Purchaser agrees to furnish to Sellers any resale certificate or certificates or other similar documents reasonably requested by Sellers to comply with or obtain an exemption from pertinent excise, sales and use tax laws.

5.7 <u>Cure Costs</u>. On or about the Closing Date, Sellers (from the proceeds of the Purchase Price) shall pay an amount equal to the Cure Costs to each counter party to an Assigned Contract and Assigned Lease so that each such Assigned Contract and Assigned Lease may be assumed by Sellers and assigned to Purchaser in accordance with the provisions of Section 365 of the Bankruptcy Code. For purposes of this Agreement, "**Cure Costs**", means all amounts that must be paid and all obligations that otherwise must be satisfied, including pursuant to Sections 365(b)(1)(A) and (B) of the Bankruptcy Code in connection with the assumption and/or assignment of the Assigned Contracts and Assigned Leases to Purchaser as provided herein.

<u>5.7</u> <u>5.8</u> <u>Operating Covenant</u>. Purchaser shall act in good faith in fulfilling its obligations under this Agreement.

# <u>5.8</u> <u>5.9</u> <u>Governmental Approvals</u>.

(a) <u>Best Efforts</u>. Purchaser (a) shall use its best efforts to secure, as promptly as possible after the Signing Date, all consents, approvals (or exemptions therefrom), authorizations, clearances and licenses required to be obtained from governmental and regulatory authorities in order to carry out the transactions contemplated by this Agreement and to cause all of its covenants and agreements to be performed, satisfied and fulfilled, and (b) will provide such other information and communications to governmental and regulatory authorities as Sellers or such authorities may reasonably request. Purchaser is responsible for all filings with and requests to governmental authorities necessary to enable Purchaser to operate the Hospital at and after the Licensure Date. Purchaser acknowledges that Sellers may independently contact governmental and regulatory authorities as part of this process.

(b) <u>Change of Ownership Applications</u>. Purchaser shall, promptly, but no later than ten (10) business days after the entry of the Sale Order, or sooner if required by applicable

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 32 of 147

governmental or regulatory authorities, file all applications, licensing packages and other documents with all applicable governmental and regulatory authorities which are necessary for the operation of a hospital and the consummation of the transactions hereunder, including the hospital license change of ownership application with CDPH, the hospital pharmacy change of ownership applications.

Attorney General Application. Purchaser and Sellers shall, promptly, but (c) no later than five (5) business days after the date of the Sale Order, file with the California Attorney General (the "Attorney General"), the application and report forms required for the transactions contemplated hereby and any supplemental information that may be reasonably requested in connection therewith pursuant to Title 11, Division 1, Chapter 15, Section 999.5 of the California Code of Regulations ("Section 999.5"), which application and report forms and supplemental information will comply in all material respects with the requirements of such regulations and shall state that Purchaser agrees to close the transactions contemplated by this Agreement so long as any conditions imposed by the Attorney General are not materially more burdensome than the conditions attached hereto as Exhibit 5.9(c) (the "Accepted Conditions"). Purchaser shall pay all fees (including, without limitation, attorneys' fees) required of Purchaser or Sellers with respect to the preparation and submission of the application, hearings, expert reports, Attorney General attorney review time, reports and other requirements of the Attorney General under Section 999.5. Each of Purchaser and Sellers shall furnish to the other such information and assistance as the other shall reasonably requires in connection with the preparation and submission to, or agency proceedings by, the Attorney General, and each of Purchaser and Sellers shall keep the other promptly apprised of any communications with, and inquires or requests for information from, the Attorney General. Purchaser shall take such action as may be required by the Attorney General in order to resolve with the minimum practicable delay any objections the Attorney General may have to the transactions contemplated by this Agreement under Section 999.5. For any late submission of the Attorney General application, at Seller's election, either the Purchase Price shall be increased by an amount equal to One Thousand Dollars (\$1,000) per day or Sellers may terminate this Agreement.

HSR Filing. Purchaser and Sellers shall, promptly, but no later than five (5) (d) business days after the date of the Sale Order, file with the Federal Trade Commission and the Department of Justice the notification and report forms required for the transactions contemplated hereby and any supplemental information that may be reasonably requested in connection therewith pursuant to the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended (the "HSR Act"), which notification and report forms and supplemental information will comply in all material respects with the requirements of the HSR Act. Purchaser shall pay all filing fees required of Purchaser or Sellers with respect to the notification, report and other requirements of the HSR Act. Each of Purchaser and Sellers shall furnish to the other such information and assistance as the other shall reasonably requires in connection with the preparation and submission to, or agency proceedings by, any governmental authority under the HSR Act, and each of Purchaser and Sellers shall keep the other promptly apprised of any communications with, and inquires or requests for information from, such governmental authorities. Purchaser shall take such action (including divestitures or hold separate arrangements) as may be required by any governmental authority in order to resolve with the minimum practicable delay any objections such governmental authorities may have to the transactions contemplated by this Agreement under the HSR Act. For any late submission of the application under the HSR Act, at Seller's election, either the Purchase Price

shall be increased by an amount equal to One Thousand Dollars (\$1,000) per day or Sellers may terminate this Agreement.

## ARTICLE 6 BANKRUPTCY COURT APPROVAL

## 6.1 <u>Bankruptcy Court Approval</u>.

Sellers and Purchaser acknowledge that this Agreement has been solicited (a) in conformity with the Bidding Procedures approved by the Bankruptcy Court on February —26, 2020 [Docket No. —4165] (the "Bid Procedures Order"), and that the sale of the Assets and the assignment of the Assigned Contracts and Assigned Leases remain subject to Bankruptcy Court approval. Purchaser further acknowledges that this Agreement and the transactions contemplated hereby are subject to Sellers' right and ability to consider higher or better competing bids with respect to the Assets subject to the terms of the Bid Procedures Order. Pursuant to the Bid Procedures Order, Purchaser shall, if its bid is determined to be the second highest bid, serve as a back-up bidder (the "Back-up Bidder") and, subject to the provisions of the Bid Procedures Order with respect to the retention of the Deposit, keep Purchaser's bid to consummate the transactions contemplated by this Agreement on the terms and conditions set forth in this Agreement (as the same may be modified with the written consent of Purchaser in any auction under the Bid Procedures Order) open and irrevocable until the earlier of (i) (a) 5:00 p.m. pacific time on the date which is thirty (30) days after the entry by the Bankruptcy Court of the Sale Order; or (b) 5:00 p.m. pacific time on May 31, 2020 (the "Outside Back-up Date"), or (ii) the date of closing of an alternative transaction with the bidder who prevails at the auction (the "Prevailing Bidder"). Following the entry of the Sale Order and prior to the Outside Back-up Date, if the Prevailing Bidder breaches or fails to perform its obligations under the terms and conditions of its respective transaction documents and Sellers terminate such agreement with the Prevailing Bidder, the Back-up Bidder will be deemed to have the new prevailing bid, and the Bankruptcy Court order approving the sale to the Prevailing Bidder shall provide that Sellers will be authorized, without further order of the Bankruptcy Court, to consummate the transactions contemplated by this Agreement on the terms and conditions set forth in this Agreement (as the same may be modified with the written consent of Purchaser in the auction) with the Back-up Bidder.

(b) Purchaser further acknowledges that this Agreement is subject to termination, up to the time of entry of the Sale Order, except to the extent otherwise provided in the Bid Procedures Order, in the event Sellers in the reasonable exercise of their fiduciary duties determine that they have received a higher and better offer for Sellers' Assets and the Bankruptcy Court has authorized the Sellers to accept and implement the terms of such offer in accordance with the Bankruptcy Code.

(c) Sellers shall, at any hearing to consider approval of this Agreement (the "**Sale Hearing**"), exercise reasonable efforts to <u>expeditiously</u> obtain a <u>Final</u> "Sale Order" approving this Agreement subject to the <u>reasonable</u> exercise of their fiduciary duties to consider and accept a higher and better offer for Sellers' assets in accordance with the Bankruptcy Code and <u>this Agreement</u>. For purposes of this Agreement, the term "**Sale Order**" shall mean an order of the Bankruptcy Court in form and substance <u>set forth in Exhibit [] and</u> reasonably acceptable to Purchaser, authorizing the sale of the Assets pursuant to Section 363(b) of the Bankruptcy Code

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 34 of 147

(including the Sellers' assumption and assignment to Purchaser of the Assigned Contracts and Assigned Leases pursuant to Section 365 of the Bankruptcy Code) on the terms and conditions set forth herein, free and clear of all <u>LiensEncumbrances</u> (other than Permitted Exceptions) and Excluded Liabilities, including, for the avoidance of doubt, any successor liability, to the maximum extent permitted by the Bankruptcy Code.

(d) For purposes of this Agreement, "Final Order" means an order, ruling, or judgment of the Bankruptcy Court or any other court of competent jurisdiction as to which the time to appeal, petition for certiorari, or move for reargument or rehearing has expired and as to which no appeal, petition for certiorari, or other proceedings for reargument or rehearing shall then be pending, or as to which any right to appeal, petition for certiorari, reargue, or rehear shall have been waived in writing, in form and substance satisfactory to Purchaser or, in the event that an appeal, writ of certiorari, or reargument or rehearing thereof has been sought, such order of the Bankruptcy Court of other court of competent jurisdiction shall have been determined by the highest court to which such order was appealed, or certiorari, reargument or rehearing shall have been denied and the time to take any further appeal, petition for certiorari, or move for reargument or rehearing shall have expired; provided, however, that the possibility that a motion under Rule 59 or Rule 60 of the Federal Rules of Civil Procedure, or any analogous Bankruptcy Rule or applicable state court rules of civil procedure, may be filed with respect to such order shall not cause such order not to be a Final Order.

(e) (d) Sellers agree, subject to the <u>reasonable</u> exercise of their fiduciary duties, to <u>expeditiously</u> seek a Bankruptcy Court determination that Purchaser is a good faith purchaser within the meaning of Section 363(m) of the Bankruptcy Code and in good faith to file such declarations and other evidence as may be required to support a determination.

(f) (e)-Sellers shall seek <u>expeditiously</u> an order from the Bankruptcy Court retaining jurisdiction over all matters relating to claims against Sellers, whether or not arising in connection with this Agreement, solely in the Bankruptcy Court.

6.2 <u>Appeal of Sale Order</u>. In the event an appeal is taken or a stay pending appeal is requested from the Sale Order, Sellers shall immediately notify Purchaser of such appeal or stay request and shall provide to Purchaser promptly a copy of the related notice of appeal or order of stay. Sellers shall also provide Purchaser with written notice of any motion or application filed in connection with any appeal from either of such orders. In the event of an appeal of the Sale Order, Sellers shall be primarily responsible for drafting pleadings and attending hearings as necessary to defend against the appeal.

6.3 <u>Bidding Procedures</u>. Sellers and Purchaser shall comply with the terms of the Bid Procedures Order. Sellers shall sign this Agreement as and when permitted pursuant to the Bid Procedures Order and the Sale Order. <u>Notwithstanding anything to the contrary herein or in the</u> <u>Bid Procedures Order, once approved the Sale Order shall not be materially modified absent the</u> <u>written consent of Purchaser, unless such modification has no substantive effect on Purchaser. In</u> <u>the event that the Sale Order is modified in contravention of this provision, Purchaser may, at its</u> <u>sole election, terminate the transaction proposed hereby.</u>

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 35 of 147

(a) Provided that there is a Competing Bidder and a competitive sale of the Assets is conducted rather than a sale outright to Purchaser without a further sale process, the initial overbid over the Purchase Price shall be Twenty Million Dollars (\$20,000,000.00) and each subsequent bid shall be in increments of Ten Million Dollars (\$10,000,000.00). Any Competing Bidders must be a Qualified Bidder under the conditions set forth in the Bid Procedures without waiver thereof. Purchaser is irrevocably deemed to be a Qualified Bidder.

(b) The Sellers shall provide to Purchaser, immediately upon receipt/determination, each of the following: (i) notice of the determination of which others have been determined to be Qualified Bidders (as defined in the Bid Procedures Order), (ii) notice of which bidders have been determined to not have submitted Qualified Bids, and (iii) provide to Purchaser copies of all bids.

## ARTICLE 7 CONDITIONS PRECEDENT TO OBLIGATIONS OF SELLERS

Sellers' obligation to sell the Assets and to close the transactions as contemplated by this Agreement shall be subject to the satisfaction of each of the following conditions on or prior to the Closing Date unless specifically waived in writing by Sellers in whole or in part at or prior to the Closing:

7.1 <u>Signing and Delivery of Instruments</u>. Purchaser shall have executed and delivered all documents, instruments and certificates required to be executed and delivered pursuant to the provisions of this Agreement.

7.2 <u>No Restraints</u>. No temporary restraining order, preliminary or permanent injunction or other order preventing the consummation of the transactions contemplated in this Agreement shall have been issued by any court of competent jurisdiction or any other governmental body and remain in effect on the Closing Date.

7.3 <u>Governmental Submissions</u>. Purchaser shall have submitted the hospital license change of ownership application to CDPH and the hospital pharmacy change of ownership application to the BOP.

7.4 <u>Attorney General Approval</u>. The Attorney General shall have approved the transactions contemplated by this Agreement, or the Debtors shall have obtained a Bankruptcy Court order or orders providing for the transfer of the Debtors' assets on conditions which are not materially more burdensome than those forth in <u>Exhibit 5.9(c)</u>.

7.5 <u>Bankruptcy Court Approval</u>. The Bankruptcy Court shall have entered the Sale Order.

7.6 <u>HSR Act</u>. The applicable waiting period under the HSR Act shall have expired or been earlier terminated.

### ARTICLE 8 CONDITIONS PRECEDENT TO OBLIGATIONS OF PURCHASER

Purchaser's obligation to purchase the Assets and to close the transactions contemplated by this Agreement shall be subject to the satisfaction of each of the following conditions on or prior to the Closing Date unless specifically waived in writing by Purchaser in whole or in part at or prior to the Closing.

8.1 <u>Signing and Delivery of Instruments</u>. Sellers shall have executed and delivered all documents, instruments and certificates required to be executed and delivered pursuant to the provisions of this Agreement.

8.2 <u>No Restraints</u>. No temporary restraining order, preliminary or permanent injunction or other order preventing the consummation of the transactions contemplated in this Agreement shall have been issued by any court of competent jurisdiction or any other governmental body and remain in effect on the Closing Date.

8.3 Attorney General Approval. The Attorney General shall have approved the transactions contemplated by this Agreement, or the Debtors shall have obtained a Bankruptey Court order or orders Final Order providing for the transfer of the Debtors' assets on conditions which are not materially more burdensome than those forth in Exhibit 5.9(c). In the event the Attorney General imposes conditions on the transactions contemplated by this Agreement, or on Purchaser in connection therewith which are materially more burdensome than the Purchaser Approved Conditions set forth on Exhibit 5.9(c) (the "Additional Conditions"), Sellers shall have the opportunity to file a motion with the Bankruptcy Court seeking the entry of an order ("Supplemental Sale Order") finding that the Additional Conditions are an "interest in property" for purposes of 11 U.S.C. § 363(f), and that the Assets can be sold free and clear of the Additional Conditions without the imposition of any other conditions, which would adversely affect the Purchaser, and such Order shall have become Final. For purposes of this Section 8.3, Additional Conditions which individually or collectively impose a direct or indirect cost to Purchaser of Five Million Dollars (\$5,000,000), or more, shall be conclusively deemed to be "materially more burdensome." If Sellers determine not to seek such Supplemental Sale Order, or fail to obtain such Final Supplemental Sale Order within sixty (60) days of the Attorney General's imposition of Additional Conditions, Purchaser (at its sole discretion) shall be entitled to terminate this Agreement and receive the return of its Deposit or elect to extend such sixty (60) day period. Upon the entry of such an order from the Bankruptcy Courtbecoming a Final Order, and so long as such order remains in full force and effect, Purchaser shall be required to consummate the transactions contemplated by this Agreement. If Sellers do not obtain such an order, or such order does not relieve the Additional Conditions to the satisfaction of Purchaser or such order does not remain in full force and effectbecome Final within the applicable time period, Purchaser shall not be required to consummate the transactions contemplated by this Agreement.

8.4 <u>Bankruptcy Court Approval</u>. The Bankruptcy Court shall have entered the Sale Order and made a finding that Purchaser is a "good faith" purchaser, and such order shall be in full force and effect and not subject to any stay requested or obtained within the period in which such stay may be so requested or obtained.

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 37 of 147

8.5 <u>HSR Act</u>. The applicable waiting period under the HSR Act shall have expired or been earlier terminated.

Medicare and Medi-Cal Provider Agreement. Following consultation with 8.6 Purchaser, Sellers shall have obtained agreements with CMS and DHCS or a court order Final Order by a court having proper jurisdiction shall have been entered with respect to the transfer of the Medicare Provider Agreement and/or the Medi-Cal Provider Agreement, such that (a) all liabilities-and obligations and Encumbrances under the Medicare/Medi-Cal Agreements are fully satisfied, discharged, and released with regard to any claims under the Medicare/Medi-Cal Agreements, whether known or unknown, that CMS or DHCS has against the Sellers or Purchaser for liabilities and obligations arising under the Medicare/Medi-Cal Agreements before the Effective Time, and (b) the Medicare/Medi-Cal Agreements will be transferred to Purchaser as of the Effective Time free and clear of such pre-Closing liabilities and, obligations and Encumbrances; provided, however, that Purchaser acknowledges and agrees that it may be treated by CMS and DHCS as the successor to the quality history associated with the relevant Medicare/Medi-Cal Agreements assigned and, for purposes of survey and certification issues associated with such quality history, Purchaser may be treated as if it is the relevant Seller and no change of ownership occurred.

Collective Bargaining Agreements and Labor Contracts. Purchaser agrees to close 8.7 the transactions contemplated by this Agreement, so long as (a) after good faith negotiations with the applicable St. Francis labor unions, such labor unions agree to either (i) modify the St. Francis collective bargaining agreements to be substantially consistent with the Purchaser's existing new collective bargaining agreements with each such respective labor union, and settle all liabilities under the existing agreements, or (ii) enter into new collective bargaining agreements that are substantially consistent with the Purchaser's existing collective bargaining agreements with each such respective labor union, and settle all liabilities under the existing agreements, or (b) the Bankruptcy Court takes action to reject any such collective bargaining agreement that is not substantially consistent or where the union has not agreed to settle all liabilities under the agreement. In the event Purchaser and the respective St. Francis labor union are unable to agree to a modified or new collective bargaining agreement that is substantially consistent with the Purchaser's existing collective bargaining agreement with the respective labor union, as well as a settlement of all liabilities under the existing agreements, and the Bankruptcy Court does not reject such collective bargaining agreement pursuant to a final order, then Purchaser shall not be required to consummate the transactions contemplated by this Agreement.

## ARTICLE 9 TERMINATION

- 9.1 <u>Termination</u>. This Agreement may be terminated at any time prior to Closing:
  - (a) by the mutual written consent of the Parties;

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 38 of 147

(b) by Sellers if a material breach of this Agreement has been committed by Purchaser and such breach has not been (i) waived in writing by Sellers or (ii) cured by Purchaser to the reasonable satisfaction of Sellers within fifteen (15) business days after Sellers provide Purchaser of a written notice which describes the nature of such breach and acceptable cure actions; *provided*, *however*, Sellers shall not be permitted to terminate this Agreement pursuant to this Section 9.1(b) if Sellers are also in material breach of this Agreement;

(c) by Purchaser if Purchaser has complied with <u>Section 3.8</u> and a material breach of this Agreement has been committed by Sellers, which material breach has resulted, or would more likely than not result, in a Material Adverse Effect on the Assets taken as a whole, and such breach has not been (i) waived in writing by Purchaser or (ii) cured by Sellers to the reasonable satisfaction of Purchaser within fifteen (15) business days after Purchaser provides Sellers of a written notice which describes the nature of such breach and acceptable cure actions; *provided, however*, Purchaser shall not be permitted to terminate this Agreement pursuant to this Section 9.1(c) if Purchaser is also in material breach of this Agreement;

(d) by Purchaser if satisfaction of any condition in <u>ARTICLE 8</u> is or becomes impossible and Purchaser has not waived such condition in writing (provided that the failure to satisfy the applicable condition or conditions has occurred by reason other than (i) through the failure of Purchaser to comply with its obligations under this Agreement or (ii) Sellers' failure to provide their closing deliveries on the Closing Date as a result of Purchaser not being ready, willing and able to close the transaction on the Closing Date);

(e) by Sellers if satisfaction of any such condition in <u>ARTICLE 7</u> is or becomes impossible and Sellers have not waived such condition in writing (provided that the failure to satisfy the applicable condition or conditions has occurred by reason other than (i) through the failure of Sellers to comply with their obligations under this Agreement or (ii) Purchaser's failure to provide its closing deliveries on the Closing Date as a result of Sellers not being ready, willing and able to close the transaction on the Closing Date);

(f) by either Purchaser or Sellers if the Bankruptcy Court enters an order dismissing the Bankruptcy Case prior to the sale closing or fails to approve the sale of the Assets to Purchaser;

(g) by Sellers in the event Sellers in the exercise of their fiduciary duties determine that they have received a higher and better offer for Sellers' assets and the Bankruptcy Court has authorized the Sellers to accept and implement the terms of such offer in accordance with the Bankruptcy Code;

(h) by Purchaser in the event that its Back-Up Bidder status has expired in accordance with the Bid Procedures Order and/<u>or</u> the terms of this Agreement; and

(i) by either Purchaser or Sellers if the Closing has not occurred (other than through the failure of any Party seeking to terminate this Agreement to comply fully with its obligations under this Agreement) on or before September 1, 2020 (the "**Termination Date**"); <u>provided</u>, that the Termination Date shall be extended in the event the conditions set forth in Section 7.4 and Section 8.3 have not been satisfied (but the conditions to Closing in Article 7 and

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 39 of 147

Article 8 have otherwise been satisfied other than such conditions that are to be satisfied by payments and deliveries to be made at the Closing) to the earlier of (A) ten (10) business days after the satisfaction of the conditions set forth in Section 7.4 and Section 8.3 or (B) December 31, 2020.

9.2 <u>Termination Consequences</u>. If this Agreement is terminated pursuant to <u>Section</u> 9.1: (a) all further obligations of the Parties under this Agreement shall terminate, other than Purchaser's right to receive a return on the Deposit in accordance with <u>Section 1.2</u> and a break-up fee in accordance with the Bidding Procedures, and provided that the provisions of <u>ARTICLE 11</u> shall survive, and (b) each Party shall pay the costs and expenses incurred by it in connection with this Agreement; *provided, however*, that in the case of any termination based on <u>Section 9.1(b)</u> or <u>Section 9.1(c)</u>, the consequences of such termination shall be determined in accordance with <u>ARTICLE 11</u> hereof. Each Party acknowledges that the agreements contained in this <u>Section 9.2</u> are an integral part of the transactions contemplated by this Agreement, that without these agreements such Party would not have entered into this Agreement.

## 9.3 Additional Termination Consequences.

In the event that (i) a bid from a purchaser other than Purchaser for all or (a) any part of the Assets in accordance with the Bid Procedures Order (a "Competing Bid") is consummated, or (ii) this Agreement is terminated for any reason other than a material breach of Purchaser, in consideration for the Purchaser having expended considerable time and expense in connection with this Agreement and the negotiation thereof and the identification and quantification of assets of Sellers, and without the requirement of any notice or demand from Purchaser or any other application to or order of the Bankruptcy Court, (A) the Deposit shall be returned to Purchaser in accordance with the Bid Procedures Order (if not previously returned pursuant to the Bid Procedures Order because Purchaser is neither the Winning Bidder (as defined in the Bid Procedures Order) nor the Back-Up Bidder (as defined in the Bid Procedures Order)) and (B) Sellers shall jointly and severally, pay (or cause to be paid to) Purchaser, in accordance with the terms hereof, a termination payment solely in the event of a successful Competing Bid in an amount equal to two and one-half percent (2.5%) of the Purchase Price (the "Stalking Horse Bidder Protections"). Sellers shall pay such Stalking Horse Bidder Protections in immediately available funds in the event of termination pursuant to Section 9.3(a)(ii), or upon consummation of a Competing Bid pursuant to Section 9.3(a)(i), to such account or accounts as may be specified in writing by Purchaser. The Stalking Horse Bidder Protections shall constitute an allowed administrative expense claim of Sellers' estates under sections 503(b) and 507 of the Bankruptcy Code and notwithstanding the prior entry of any order of the Bankruptcy Court relating to the use of cash collateral, be paid on the second (2nd) business day following the date of consummation of a Competing Bid or termination of this Agreement under Section 9.1(a), (c), (d), (f), (g), (h) or (i) if no material breach by Purchaser of this Agreement has occurred. The Sale Order shall provide, although not required for the enforceability hereof, for payment by Sellers of the Stalking Horse Bidder Protections as and when such amounts are due and payable hereunder. Nothing in this Section 9.3(a) shall relieve Purchaser or Sellers of any liability for a breach of this Agreement prior to the date of termination.

(b) Each of the Parties acknowledges and agrees that the agreements contained in this Section 9.3 are an integral part of the transactions contemplated by this Agreement and that, without these agreements, the other Parties would not enter into this Agreement. Each of the

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 40 of 147

Parties further acknowledges that the payment by Sellers of the Stalking Horse Bidder Protections is not a penalty, but rather liquidated damages in a reasonable amount that will compensate Purchaser, in the circumstances in which such Stalking Horse Bidder Protection is payable, for the efforts and resources expended and the opportunities foregone while negotiating this Agreement and in reliance on this Agreement and on the expectation of the consummation of the transactions contemplated by this Agreement, which amount would otherwise be impossible to calculate with precision. The obligation to pay the Stalking Horse Bidder Protections in accordance with the provisions of this Agreement will (i) be binding upon and enforceable jointly and severally against each Seller and (ii) survive the subsequent termination of this Agreement, solely to the extent permitted by applicable law. The obligation to pay the Stalking Horse Bidder Protections as and when required under this Agreement, are intended to be, and are, binding upon (A) each Seller, (B) any successors or assigns of any Seller and (C) any trustee, examiner or other representative of a Seller's estate (each of (A) through (C), a "Successor") as if such Successor were a Seller hereunder.

## ARTICLE 10 POST-CLOSING MATTERS

10.1 Excluded Assets. Subject to Section 10.2 hereof, any Excluded Asset (or proceeds thereof) (a) pursuant to the terms of this Agreement, (b) as otherwise determined by the Parties' mutual written agreement or (c) absent such agreement, as determined by adjudication by the Bankruptcy Court, which comes into the possession, custody or control of Purchaser (or its respective successors-in-interest, assigns or affiliates) shall, within tenthirty (1030) business days following receipt, be transferred, assigned or conveyed by Purchaser (and its respective successors-in-interest, assigns and affiliates) to Sellers without imposing any charge to Sellers for Purchaser's transfer, storage, handling or holding of same on and after the Effective Time. Purchaser (and its respective successors-in-interest, assigns and affiliates) shall have neither the right to offset amounts payable to Sellers under this Section 10.1 against, nor the right to contest its obligation to transfer, assign and convey to Sellers because of, outstanding claims, liabilities or obligations asserted by Purchaser against Sellers. If Purchaser does not remit any monies included in the Excluded Assets (or proceeds thereof) to Sellers in accordance with the first sentence of this Section 10.1, such withheld funds shall bear interest at the Prime Rate in effect on the calendar day upon which such payment was required to be made to Sellers (the "Excluded Asset Due Date") plus five percent (5%) (or the maximum rate allowed by law, whichever is less), such interest accruing on each calendar day after the Excluded Asset Due Date until payment of the Excluded Assets and all interest thereon is made to Sellers.

## 10.2 Preservation and Access to Records After the Closing.

(a) From the Licensure Date until seven (7) years after the Licensure Date or such longer period as required by law (the "**Document Retention Period**"), Purchaser shall keep and preserve all medical records, patient records, medical staff records and other books and records which are among the Assets as of the Effective Time, but excluding any records which are among the Excluded Assets. Purchaser will afford to the representatives of Sellers, any of their affiliates, the Official Committee of the Unsecured Creditors of the Sellers, Sellers' estate representative or any liquidating trustee of the Sellers' bankruptcy estate ("Seller Parties"), including their counsel and accountants, full and complete access to, and copies (including,

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 41 of 147

without limitation, color laser copies) of, such records with respect to time periods prior to the Licensure Date (including, without limitation, access to records of patients treated at the Hospital prior to the Licensure Date) during normal business hours after the Licensure Date, to the extent reasonably needed by any Seller Party for any lawful purpose. Purchaser acknowledges that, as a result of entering into this Agreement and operating the Hospital, it will gain access to patient records and other information which are subject to rules and regulations concerning confidentiality. Purchaser shall abide by any such rules and regulations relating to the confidential information it acquires. Purchaser shall maintain the patient and medical staff records at the Hospital in accordance with applicable law and the requirements of relevant insurance carriers. After the expiration of the Document Retention Period, if Purchaser intends to destroy or otherwise dispose of any of the documents described in this Section 10.2(a), Purchaser shall provide written notice to Sellers of Purchaser's intention no later than forty-five (45) calendar days prior to the date of such intended destruction or disposal. Any of the Seller Parties shall have the right, at its sole cost, to take possession of such documents during such forty-five (45) calendar day period. If any of the Seller Parties do not take possession of such documents during such forty-five (45) calendar day period, Purchaser shall be free to destroy or otherwise dispose of such documentation upon the expiration of such forty-five (45) calendar day period.

Provided that Purchaser shall not incur any out of pocket costs, Purchaser (b) shall give full cooperation to the Seller Parties and their insurance carriers in connection with the administration of Sellers' estate, including, without limitation, in connection with all claims, actions, causes of action or audits relating to the Excluded Assets, Excluded Liabilities or pre-Closing operation of the Sellers or the Hospital that any Seller Party may elect to pursue, dispute or defend, in respect of events occurring prior to the Licensure Date with respect to the operation of the Hospital. Such cooperation shall include, without limitation, making the Hired Employees available for interviews, depositions, hearings and trials and other assistance in connection with the administration of Sellers' estate and such cooperation shall also include making all of its employees available to assist in the securing and giving of evidence and in obtaining the presence and cooperation of witnesses (all of which shall be done without payment of any fees or expenses to Purchaser or to such employees. In addition, Sellers and Sellers' affiliates shall be entitled to remove from the Hospital originals of any such records, but only for purposes of pending litigation involving the persons to whom such records refer, as certified in writing prior to removal by counsel retained by Sellers or any of Sellers' affiliates in connection with such litigation. Any records so removed from the Hospital shall be promptly returned to Purchaser following Sellers' or its applicable affiliate's use of such records.

(c) In connection with (i) the transition of the Hospital pursuant to the transaction contemplated by this Agreement, (ii) Sellers' rights to the Excluded Assets, (iii) any claim, audit, or proceeding, including, without limitation, any tax claim, audit, or proceeding and (iv) the Sellers' obligations under the Excluded Liabilities, Purchaser shall after the Effective Time give Sellers access during normal business hours to Purchaser's books, personnel, accounts and records and all other relevant documents and information with respect to the assets, liabilities and business of the Hospital as representatives of Sellers and Sellers' affiliates may from time to time reasonably request, all in such manner as not to unreasonably interfere with the operations of the Hospital.

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 42 of 147

(d) Purchaser and its representatives shall be given access by Sellers during normal business hours to the extent reasonably needed by Purchaser for business purposes to all documents, records, correspondence, work papers and other documents retained by Sellers pertaining to any of the Assets prior to the Effective Time (excluding confidential employee information, privileged materials and patient records), all in such manner as to not interfere unreasonably with Sellers. Such documents and other materials shall be, at Sellers' option, either (i) copied by Sellers for Purchaser at Purchaser's expense, or (ii) removed by Purchaser from the premises, copied by Purchaser and promptly returned to Sellers.

(e) Purchaser shall comply with, and be solely responsible for, all obligations under the Standards for Privacy of Individually Identifiable Health Information (45 CFR Parts 160 and 164) promulgated pursuant to the Health Insurance Portability and Accountability Act of 1996 and any and all corresponding California state law requirements with respect to the operation of the Hospital on and after the Licensure Date.

(f) Purchaser shall cooperate with Sellers, on a timely basis and as reasonably requested by Sellers, in connection with the provision of all data of the Hospital and other information required by Sellers for reporting purposes, including without limitation QAF or DSH reporting.

(g) To the maximum extent permitted by law, if any Person requests or demands, by subpoena or otherwise, any documents relating to the Excluded Liabilities or Excluded Assets, including without limitation, documents relating to the operations of any of the Hospital or any of the Hospital's committees prior to the Licensure Date, prior to any disclosure of such documents, Purchaser shall notify Sellers and shall provide Sellers with the opportunity to object to, and otherwise coordinate with respect to, such request or demand.

10.3 Provision of Benefits of Certain Contracts. Notwithstanding anything contained herein to the contrary, this Agreement shall not constitute an agreement to assign any Assigned Contract or Assigned Lease, if, notwithstanding the provisions of Sections 363 and 365 of the Bankruptcy Code, an attempted assignment thereof, without the consent of the third party thereto, would constitute a breach thereof or in any way negatively affect the rights of Sellers or Purchaser, as the assignee of such Assigned Contract or Assigned Lease, as the case may be, thereunder. If, notwithstanding the provisions of Sections 363 and 365 of the Bankruptcy Code, such consent or approval is required but not obtained, Sellers will cooperate with Purchaser in any reasonable arrangement designed to both (a) provide Purchaser with the benefits of or under any such Assigned Contract or Assigned Lease. Further, notwithstanding anything contained in this Agreement to the contrary, this Agreement shall not constitute an agreement to assign any Account Receivable the assignment of which is either prohibited by law or by the terms of any contract with a payor without the consent of such payor.

10.4 <u>General Cooperation and Turnover Obligations</u>. The Parties shall cooperate to ensure that any and all payments that constitute "Excluded Assets" shall be paid to and received by Sellers, with any payments that constitute "Assets" transferred to Purchaser pursuant to <u>Section</u> <u>1.7</u> or that otherwise arise from services rendered by Purchaser on or after the Licensure Date be paid to and received by Purchaser. In this regard, for a period of two (2) years after the Effective

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 43 of 147

Date ("**Turnover Period**"), the Parties shall, within ten (10) business days of receipt, copy and send to the other Party copies (either in hard copy or via electronic file) of all remittance advices for all deposits to all Lockboxes or other bank accounts for the <u>Accounts</u> Receivables, from whatever payor or source of funds, that are received on and after the Effective Time. In the event that payments that constitute a transferred Asset are deposited to a bank account of Sellers which is not automatically swept or transferred to Purchaser, then Sellers, within ten (10) days of notice of the receipt of such payments shall turnover and pay Purchaser said funds. In the event that a deposit representing payment of any Excluded Assets is received by Purchaser, then Purchaser, within ten (10) days of notice of the receipt of funds. Each Party shall have the right, within three (3) months after the expiration of the Turnover Period, to audit by an independent and competent auditor, at the requesting Party's sole expense, of the bank records and remittance advices of the other Party. Thereafter, upon the findings of the auditor that there has either been an overpayment or and underpayment of funds due, the Party owning funds shall, within ten (10) business days, make a payment of such funds to whom they are owed.

10.5 <u>Closing of Financials</u>. Provided that Purchaser shall not incur any out of pocket costs, Purchaser shall cause the individual acting as the chief financial officer of the Hospital after the Effective Time (the "**Post-Effective Time CFO**") to cooperate with Sellers' representatives in order to complete the standardized closing of Sellers' financial records through the Licensure Date including, without limitation, the closing of general ledger account reconciliations (collectively, the "**Closing of Financials**"). Purchaser shall cause the Post-Effective Time CFO to use their good faith efforts to cooperate with Sellers' representatives in order to complete the Alter which is sixty (60) calendar days after the Closing Date. The Post-Effective Time CFO and other appropriate personnel shall be reasonably available to Sellers for a period of no less than one hundred eighty (180) calendar days after the Closing Date to assist Sellers in the completion of Sellers' post-Closing audit, such assistance not to interfere unreasonably with such Post-Effective Time CFO's other duties.

10.6 <u>Medical Staff</u>. To ensure continuity of care in the community, Purchaser agrees that the Hospital's medical staff members in good standing as of the Licensure Date shall maintain medical staff privileges at the Hospital as of the Licensure Date. On and after the Licensure Date, the medical staff will be subject to the Hospital's Medical Staff Bylaws then currently in effect, provided that such Bylaws are in compliance with all applicable laws and regulations and contain customary obligations.

10.7 <u>Shared Intangible Assets</u>. In the event and to the extent that certain intangible Assets transferred by Sellers have been used to operate businesses of Verity or Verity Holdings or their affiliates ("**Shared Intangible Assets**") and such Shared Intangible Assets continue to be used by Verity or Verity Holdings or their affiliates to operate such businesses after Closing, Verity and Verity Holdings retain the rights to continue to use such Assets notwithstanding their sale to Purchaser. Purchaser shall reasonably cooperate with Verity and Verity Holdings and their affiliates to give effect to such rights and shall provide Verity and Verity Holdings and their affiliates such documentation, records and information and reasonable access to such systems as necessary for Verity and Verity Holdings and their affiliates to continue to operate such businesses; all in such manner as not to reasonably interfere with the operations of the Hospitals;

*provided, however*, Purchaser shall not be required to incur any out-of-pocket costs in association therewith unless reimbursed by Verity and Verity Holdings and their affiliates.

## ARTICLE 11 DEFAULT, TAXES AND COST REPORTS

11.1 <u>Purchaser Default</u>. If the Agreement is terminated pursuant to <u>Section 9.1(b)</u>, Sellers shall be entitled to retain the Deposit, and Sellers may, in addition thereto, pursue any rights or remedies that Sellers may have under this Agreement or applicable law, including the right to sue for damages or specific performance.

11.2 <u>Sellers Default</u>. If the Agreement is terminated pursuant to <u>Section 9.1(c)</u>, Purchaser shall have the right to demand and receive a refund of the Deposit, and Purchaser may, in addition thereto, pursue any rights or remedies that Purchaser may have under applicable law, including the right to sue for damages or specific performance.

# 11.3 <u>Tax Matters; Allocation of Purchase Price</u>.

(a) After the Licensure Date, the Parties shall cooperate fully with each other and shall make available to each other, as reasonably requested, all information, records or documents relating to tax liabilities or potential tax liabilities attributable to Sellers with respect to the operation of the Hospital for all periods prior to the Licensure Date and shall preserve all such information, records and documents at least until the expiration of any applicable statute of limitations or extensions thereof. The Parties shall also make available to each other to the extent reasonably required, and at the reasonable cost of the requesting Party (for out-of-pocket costs and documents in connection with tax matters and as Sellers reasonably may request in connection with the completion of any post-Closing audits of the Hospital.

(b) Solely for purposes of tax reporting, <u>Schedule 11.3</u> sets forth the allocation of the Purchase Price (including any liabilities that are considered to be an increase to the Purchase Price for United States federal income tax purposes) among the Assets in accordance with Section 1060 of the Code and the Treasury Regulations promulgated thereunder (the "Allocation Schedule"). The Allocation Schedule shall be final and binding upon Sellers and Purchaser with respect to matters relating to required tax reporting by each such Party. The Parties shall refrain from taking any position that is inconsistent with the Allocation Schedule with respect to tax reporting.

# 11.4 Cost Report Matters.

(a) Consistent with <u>Section 4.4</u>, Sellers shall, at Purchaser's expense, prepare and timely file all cost reports relating to the periods ending prior to the Licensure Date or required as a result of the consummation of the transactions described in this Agreement, including, without limitation, those relating to Medicare, Medicaid, and other third party payors which settle on a cost report basis (the "**Seller Cost Reports**").

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 45 of 147

(b) Upon reasonable notice and during normal business office hours, Purchaser will cooperate reasonably with Sellers in regard to Sellers' preparation and filing of the Seller Cost Reports. Such cooperation shall include, at no cost to Sellers, obtaining access to files at the Hospital and Purchaser's provision to Sellers of data and statistics, and the coordination with Sellers pursuant to reasonable notice of Medicare and Medicaid exit conferences or meetings. Sellers shall have no obligations after the Licensure Date with respect to Seller Cost Reports except for preparation and filing thereof.

### ARTICLE 12 MISCELLANEOUS PROVISIONS

12.1 <u>Further Assurances and Cooperation</u>. Each Party shall execute, acknowledge and deliver to the other Party any and all other assignments, consents, approvals, conveyances, assurances, documents and instruments reasonably requested by such Party at any time and shall take any and all other actions reasonably requested by such Party at any time for the purpose of consummating the transactions hereunder and fulfilling such Party's obligations hereunder. After consummation of the transactions contemplated in this Agreement, the Parties agree to cooperate with each other and take such further actions as may be necessary or appropriate to effectuate, carry out and comply with all of the terms of this Agreement, the documents referred to in this Agreement and the transactions contemplated hereby.

12.2 <u>Successors and Assigns</u>. All of the terms and provisions of this Agreement shall be binding upon and shall inure to the benefit of and be enforceable by the respective successors and assigns of the Parties hereto; *provided, however*, that no Party hereto may assign any of its rights or delegate any of its duties under this Agreement without the prior written consent of the other Parties which consent shall not be unreasonably withheld or delayed, except that Purchaser may, without the prior written consent of Sellers, assign all or any portion of its rights under this Agreement to one or more of its affiliates prior to the Closing Date so long as such affiliate was described and was part of the Attorney General application under Section 5.9(c) and provided that any such assignment shall not relieve Purchaser of or reduce Purchaser's obligations under this Agreement.

12.3 <u>Governing Law; Venue</u>. This Agreement shall be construed, performed, and enforced in accordance with, and governed by, the laws of the State of California (without giving effect to the principles of conflicts of laws thereof), except to the extent that the laws of such State are superseded by the Bankruptcy Code or other applicable federal law. For so long as Sellers are subject to the jurisdiction of the Bankruptcy Court, the Parties irrevocably elect, as the sole judicial forum for the adjudication of any matters arising under or in connection with the Agreement, and consent to the exclusive jurisdiction of, the Bankruptcy Court. The Parties hereby consent to the jurisdiction of such court and waive their right to challenge any proceeding involving or relating to this Agreement on the basis of lack of jurisdiction over the Person or forum non conveniens.

12.4 <u>Amendments</u>. This Agreement may not be amended other than by written instrument signed by the Parties.

12.5 <u>Exhibits, Schedules and Disclosure Schedule</u>. The Disclosure Schedule and all exhibits and schedules referred to in this Agreement shall be attached hereto and are incorporated

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 46 of 147

by reference herein. From the Signing Date until the Closing, the Parties agree that Sellers may update the Disclosure Schedule as necessary upon written notice to Purchaser, and the applicable representation and warranty shall thereafter be deemed amended for all purposes by such updated Disclosure Schedule. Notwithstanding the foregoing, should any exhibit or schedule not be completed and attached hereto as of the Signing Date, Sellers and Purchaser shall promptly negotiate in good faith any such exhibit or schedule, which exhibit or schedule must be acceptable to each of Sellers and Purchaser in their reasonable discretion prior to being attached hereto. Any matter disclosed in this Agreement or in the Disclosure Schedule with reference to any Section of this Agreement shall be deemed a disclosure in respect of all sections to which such disclosure may apply. The headings, if any, of the individual sections of the Disclosure Schedule are provided for convenience only and are not intended to affect the construction or interpretation of this Agreement. The Disclosure Schedule is arranged in sections and paragraphs corresponding to the numbered and lettered sections and paragraphs of this Agreement merely for convenience, and the disclosure of an item in one section of the Disclosure Schedule as an exception to a particular obligation, representation or warranty shall be deemed adequately disclosed as an exception with respect to all other obligations, representations or warranties, notwithstanding the presence or absence of an appropriate section of the Disclosure Schedule with respect to such other obligation, representations or warranties or an appropriate cross reference thereto.

12.6 <u>Notices</u>. Any notice, demand, letter or other communication required, permitted, or desired to be given hereunder shall be deemed effectively given when either personally delivered, or when received by telegraphic or other electronic means (including facsimile) or overnight courier, or five (5) calendar days after being deposited in the United States mail, with postage prepaid thereon, certified or registered mail, return receipt requested, addressed as follows:

If to Sellers:	Verity Health System of California, Inc. 601 South Figueroa St., Suite 4050 Los Angeles, CA 90017-5704 Attention: Chief Executive Officer
With a copies to: (which copies shall not constitute notice)	Dentons US LLP 601 South Figueroa St., Suite 2500 Los Angeles, CA 90017-5704 Attention: Tania Moyron, Esq. Telephone: 213-243-6101
If to Purchaser:	Prime Healthcare Services, Inc. 3480 East Guasti Road, 2 <sup>nd</sup> Floor Ontario, California 91761 Attention: General Counsel Facsimile: 909-235-4316
With a copy to: (which copies shall not constitute notice)	McDermott Will & Emery LLP 2049 Century Park East, Suite 3200 Los Angeles, California 90067 Attention: Jeffrey Reisner, Esq. Facsimile: 310-277-4730

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 47 of 147

or at such other address as one Party may designate by notice hereunder to the other Parties.

12.7 <u>Headings</u>. The section and other headings contained in this Agreement and in the Disclosure Schedule, exhibits and schedules to this Agreement are included for the purpose of convenient reference only and shall not restrict, amplify, modify or otherwise affect in any way the meaning or interpretation of this Agreement or the Disclosure Schedule, exhibits and schedules hereto.

12.8 <u>Publicity</u>. Prior to the Closing Date, Sellers and Purchaser shall consult with each other as to the form and substance of any press release or other public disclosure materially related to this Agreement or any other transaction contemplated hereby and each shall have the right to review and comment on the other's press releases at least forty-eight (48) hours prior to issuance; *provided, however*, that nothing in this <u>Section 12.8</u> shall be deemed to prohibit either Sellers or Purchaser from making any disclosure that its counsel deems necessary or advisable in order to satisfy either Party's disclosure obligations imposed by law subject to reasonable prior notice to the other Party thereof.

12.9 <u>Fair Meaning</u>. This Agreement shall be construed according to its fair meaning and as if prepared by all Parties.

12.10 <u>Gender and Number; Construction; Affiliates</u>. All references to the neuter gender shall include the feminine or masculine gender and vice versa, where applicable, and all references to the singular shall include the plural and vice versa, where applicable. Unless otherwise expressly provided, the word "including" followed by a listing does not limit the preceding words or terms and shall mean "including, without limitation." Any reference in this Agreement to an "affiliate" shall mean any Person directly or indirectly controlling, controlled by or under common control with a second Person. The term "control" (including the terms "controlled by" and "under common control with") means the possession, direct or indirect, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by contract or otherwise. A "Person" shall mean any natural person, partnership, corporation, limited liability company, association, trust or other legal entity.

12.11 <u>Third Party Beneficiary</u>. None of the provisions contained in this Agreement are intended by the Parties, nor shall they be deemed, to confer any benefit on any person not a party to this Agreement, except for the Parties' successors and permitted assigns, and except for any liquidating trustee or plan administrator for Sellers' estate.

12.12 Expenses and Attorneys' Fees. Except as otherwise provided in this Agreement, each Party shall bear and pay its own costs and expenses relating to the preparation of this Agreement and to the transactions contemplated by, or the performance of or compliance with any condition or covenant set forth in, this Agreement, including without limitation, the disbursements and fees of their respective attorneys, accountants, advisors, agents and other representatives, incidental to the preparation and carrying out of this Agreement, whether or not the transactions contemplated hereby are consummated. The Parties expressly agree that all sales, transfer, documentary transfer and similar taxes, fees, surcharges and the like in connection with the sale of the Assets shall be borne by Purchaser. If any action is brought by any Party to enforce any

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 48 of 147

provision of this Agreement, the prevailing Party shall be entitled to recover its court costs and reasonable attorneys' fees.

12.13 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same Agreement, binding on all of the Parties hereto. The Parties agree that facsimile copies of signatures shall be deemed originals for all purposes hereof and that a Party may produce such copies, without the need to produce original signatures, to prove the existence of this Agreement in any proceeding brought hereunder.

12.14 <u>Entire Agreement</u>. This Agreement, the Disclosure Schedule, the exhibits and schedules, and the documents referred to in this Agreement contain the entire understanding between the Parties with respect to the transactions contemplated hereby and supersede all prior or contemporaneous agreements, understandings, representations and statements, oral or written, between the Parties on the subject matter hereof (the "**Superseded Agreements**"), which Superseded Agreements shall be of no further force or effect.

12.15 <u>No Waiver</u>. Any term, covenant or condition of this Agreement may be waived at any time by the Party which is entitled to the benefit thereof but only by a written notice signed by the Party expressly waiving such term or condition. The subsequent acceptance of performance hereunder by a Party shall not be deemed to be a waiver of any preceding breach by any other Party of any term, covenant or condition of this Agreement, other than the failure of such other Party to perform the particular duties so accepted, regardless of the accepting Party's knowledge of such preceding breach at the time of acceptance of such performance. The waiver of any term, covenant or condition shall not be construed as a waiver of any other term, covenant or condition of this Agreement.

12.16 <u>Severability</u>. If any term, provision, condition or covenant of this Agreement or the application thereof to any Party or circumstance shall be held to be invalid or unenforceable to any extent in any jurisdiction, then the remainder of this Agreement and the application of such term, provision, condition or covenant in any other jurisdiction or to persons or circumstances other than those as to whom or which it is held to be invalid or unenforceable, shall not be affected thereby, and each term, provision, condition and covenant of this Agreement shall be valid and enforceable to the fullest extent permitted by law.

12.17 <u>Time is of the Essence</u>. Time is of the essence for all dates and time periods set forth in this Agreement and each performance called for in this Agreement.

12.18 <u>WAIVER OF JURY TRIAL</u>. TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT, OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE, AGENT OR

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 49 of 147

ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (B) EACH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (C) EACH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (D) EACH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 12.18. THIS IS AN AFFIRMATIVE WAIVER OF THE PARTIES' RIGHTS TO A JURY TRIAL UNDER CALIFORNIA LAW, Cal. C. Civ. Pro. Sec 631. BY SIGNING BELOW ON THE SIGNATURE LINES, EACH PARTY IS EXPLICITLY WAIVING JURY TRIAL AND AUTHORIZING ANY AND ALL PARTIES TO FILE THIS WAIVER WITH ANY COURT AS THE WAIVER REQUIRED UNDER Cal. C. Civ. Proc. Sec. 631(f)(2).

[REMAINDER OF PAGE IS BLANK]

# Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 50 of 147

**IN WITNESS WHEREOF,** this Agreement has been entered into as of the day and year first above written.

## **PURCHASER:**

### Prime Healthcare Services, Inc.

Signature:	
Print Name:	
Title:	
Date:	

### **SELLERS:**

# Verity Health System of California, Inc.

Signature:	
Print Name:	
Title:	
Date:	

# Verity Holdings, LLC

Signature:	
Print Name:	
Title:	
Date:	

### **St. Francis Medical Center**

Signature:	
Print Name:	
Title:	
Date:	

Description	Owner	Address	APN
Hospital Campus (including Health Services Pavilion, Progressive Care Unit, Family Life Center, Power Plant and Parking Structure #1)	St. Francis Medical Center 3630 East Imperial Highway, Lynwood, CA 90262		6173-021-008
Huntington Park MOB	St. Francis Medical Center	2700 East Slauson Avenue, Huntington Park, CA 90255	6320-006-069
Maywood MOB	St. Francis Medical Center5953 South Atlantic Boulevard 5, Maywood CA 90270		6313-013-028
Parking Lot (St. Francis Medical Center)	Verity Holdings, LLC	3633 Martin Luther King Jr., Boulevard, Lynwood, CA 90262	6173-015-047
Ministry Services Building (St. Francis Medical Center)	Verity Holdings, LLC	3663 Martin Luther King Jr., Boulevard, Lynwood, CA 90262	6173-019-022
Parking Lot (St. Francis Medical Center)	Verity Holdings, LLC	3621 Martin Luther King Jr., Boulevard, Lynwood, CA 90262	6173-015-048
Five Unit Apartment (St. Francis Medical Center)	Verity Holdings, LLC	3570 Brenton Avenue, Lynwood, CA 90057	6191-016-008

# <u>Schedule 1.4(c)</u> St. Francis Medical Center

# Exhibit 1.3(a) to APA

# Form of Leaseback Agreement

Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 53 of 147

#### SALE LEASEBACK AGREEMENT

[Note: This agreement is drafted for 3 Purchaser parties: Purchaser/Parent Company, Hospital Newco and Manager. This structure is not required.]

This Sale Leaseback Agreement (the "Leaseback Agreement") is made and entered into as of \_\_\_\_\_\_, 2020 (the "Signing Date"), by and among St. Francis Medical Center, a California nonprofit public benefit corporation (the "Corporation") on the one hand, and \_\_\_\_\_\_\_ ("Parent Company") and \_\_\_\_\_\_ ("Hospital Newco," and collectively with Parent Company, the "Parent Company Parties") on the other hand. The Corporation, Parent Company, and Hospital Newco may be referred to herein individually as a "Party," and collectively as the "Parties."

#### RECITALS

**A.** On August 31, 2018, the Corporation and certain of its affiliates each filed a voluntary petition for relief (collectively, the "**Bankruptcy Cases**") under chapter 11 of title 11 of the United State Code (the "**Bankruptcy Code**").

**B.** The Bankruptcy Cases are jointly administered under Lead Case No. 18-20151 and are currently pending in the Bankruptcy Court for the Central District of California in Los Angeles (the "**Bankruptcy Court**").

C. Parent Company and the Corporation have entered into that certain Asset Purchase Agreement, dated of even date herewith (the "APA"), which provides for the sale of the assets of the Corporation and related assets (collectively, the "Assets"). Capitalized terms not defined in this Leaseback Agreement shall have the meanings given to such terms in the APA.

**D.** The Corporation operates St. Francis Medical Center as a 384-bed general acute care hospital (the "**Hospital**"), with its primary location at 3630 East Hospital Newco Highway, Lynwood, California 90262, and such other locations where the Hospital's services are provided (collectively, the "**Hospital Premises**").

E. Parent Company designated its affiliate, Hospital Newco, as the owner of the Corporation's Assets purchased by Parent Company under the APA (the "**Designation**"), and its affiliate \_\_\_\_\_\_ (the "**Manager**") as the operator of the Hospital.

**F.** Pursuant to Section 1.3 of the APA, the Corporation, Parent Company, and the Manager, are entering into that certain Interim Management Agreement ("IMA") of even date herewith, and commencing at the Effective Time, to enable the Manager to manage the day-to-day operations of the Hospital following the Closing until the Manager is issued the Licenses necessary to operate the Hospital (for the avoidance of doubt, that date will occur when the Manager is issued both a license to operate the Hospital as an acute care hospital by the California Department of Public Health, and a permit to operate a hospital-based pharmacy by the California Board of Pharmacy (collectively, the "New Licenses")).

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 54 of 147

A. Pursuant to the APA, at the Effective Time, Parent Company will purchase the Corporation's Assets, except for the Excluded Assets (as set forth in the APA). Also pursuant to the APA, that portion of the Assets constituting drugs, dangerous devices, pharmacy systems, or other pharmacy assets (the "**Pharmacy Assets**"), the Medicare/Medi-Cal Agreements, the Lockboxes, the Transferred Managed Care Agreements and the Transferred Private Payor Agreements (collectively, the "**Licensure Date Assets**") shall transfer to Parent Company (or its affiliate) as of the Licensure Date.

**B.** Immediately following the Closing, and until the Manager obtains the New Licenses, the Parent Company Parties desire to lease back or license to the Corporation all of the then-acquired Corporation's Assets used in the operation of the Hospital, and the Corporation desires to so lease or license such Corporation's Assets from the Parent Company Parties on the terms and conditions set forth herein. The Corporation's Assets shall exclude the Licensure Date Assets, which the Corporation shall own until the New Licenses are issued.

### AGREEMENT

NOW, THEREFORE, in consideration of the covenants and conditions contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby agree as follows:

1. <u>Definitions</u>. Any capitalized term appearing herein that is not defined shall have the same definition ascribed to it under the IMA or the APA (as designated when such term first appears herein).

### 2. <u>Description of The Leased or Licensed Assets</u>.

**2.1** Leased or Licensed Assets. The Parent Company Parties hereby lease or license to the Corporation, and the Corporation leases or licenses from the Parent Company Parties, all of the Corporation's Assets used in the operation of the Hospital, except for the Licensure Date Assets, which the Corporation shall continue to own until the Licensure Date (as defined in the APA) (collectively, the "Leased or Licensed Assets"). Specifically, the Leased or Licensed Assets shall include: (a) the Hospital Premises (which shall be leased); (b) tangible personal property, including, but not limited to, fixtures, furnishings, hard copy medical and financial records and equipment (including, but not limited to, hardware to operate and run the electronic health record systems, hospital operating systems, laboratory information systems, and financial reporting systems) (which shall be leased); and (c) intangible intellectual property saved or embodied in the electronic health record systems, hospital operating systems, laboratory information systems, laboratory information systems, and financial reporting systems (which shall be licensed for use by the Corporation).

### 2.2 <u>Management of Pharmacy and Use of Pharmacy Assets</u>.

(a) The Corporation shall at all times during the term of this Leaseback Agreement be the owner of the Pharmacy Assets. The Parties acknowledge, however, that under the IMA, the Manager will be managing the Corporation's pharmacy during the Management Period (as defined in the IMA), and the Corporation therefore grants the Manager (i) access to

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 55 of 147

and authority to use the drugs, dangerous drug delivery devices, or other tangible pharmacy assets, and (ii) a license to use the intangible intellectual property saved or embodied in the pharmacy systems, in each case to the extent necessary for the Manager to fulfill its obligations under the IMA.

(b) Pursuant to the IMA, the Corporation shall be responsible for purchasing drugs and dangerous devices identified by the Manager as necessary for the operations of the Hospital, and the Manager shall reimburse the Corporation for all costs and expenses incurred for such purchases.

**3.** <u>Term; Termination</u>. This Leaseback Agreement shall have a term coextensive with the Management Period (as defined in the IMA) and shall automatically terminate upon the termination of the IMA.

4. <u>Payments by the Parent Company Parties</u>.

**4.1** <u>Utilities</u>. The Parent Company Parties shall pay all utilities and services supplied to the Hospital during the term hereof, including but not limited to water, gas, air conditioning, heat, light, power, telephone service, and waste removal services.

**4.2** <u>Taxes</u>. The Parent Company Parties shall pay all taxes, assessments, and levies of any kind or nature whatsoever, including real property taxes, personal property taxes, income taxes, employment taxes, and sales or use taxes, that are taxed, assessed, levied, invoiced or imposed upon or against the Leased or Licensed Assets, the Hospital, and/or the Corporation, after the Effective Time.

**4.3** <u>Insurance</u>. The Parent Company Parties shall pay for all insurance coverages, including premiums, deductibles, stop-loss, and any other insurance covering the Leased or Licensed Assets, the Licensure Date Assets, the Hospital, and the Corporation during the term hereof. The Parent Company Parties covenant and agree that the Leased or Licensed Assets, the Licensure Date Assets, the Hospital, and the Corporation are covered as of the date hereof and will be covered at all times by general liability, fire, theft, business interruption, cyber, professional liability, directors and officers insurance, employment practices liability, terrorism, workers' compensation & employers' liability, directors and officers, fiduciary, crime, punitive damages excess liability, physical damage, property liability, automobile, storage tank, helipad and non-owned aviation, sexual misconduct and molestation, medical provider professional liability, and provider capitation stop loss (managed care excess loss) insurance. All such insurance shall name the Parent Company Parties and the Corporation as insureds as their respective interests may appear.

4.4 <u>Repairs and Maintenance: Alterations</u>. During the term of this Leaseback Agreement, the Parent Company Parties shall pay all costs of repairing (including replacement of) and maintaining the Leased or Licensed Assets and Hospital and every part thereof in good and sanitary order, condition and repair during the term hereof, reasonable wear and tear excepted, including, without limitation, all costs of all repairs, replacements and maintenance required by any applicable governmental law, statute, ordinance, rule or regulation, including the California Office of Statewide Health Planning and Development ("**OSHPD**"). The Corporation shall not

### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 56 of 147

make any alterations or changes to the Leased or Licensed Assets without prior written approval of the Parent Company Parties, which may be given or withheld in the Parent Company Parties' sole discretion.

**4.5** <u>Payment</u>. Nothing in this <u>Section 4</u> shall in any way limit, reduce, or otherwise affect Parent Company's payment obligations under the IMA or the APA.

5. <u>Use</u>. The Leased or Licensed Assets shall be used for the operation of the Hospital, subject to the terms of the APA and the IMA.

# 6. <u>Risk of Loss</u>.

**6.1** The Corporation assumes no risks or liability for damage to or injury occurring to the Leased or Licensed Assets or Hospital during the term of this Leaseback Agreement by any means whatsoever, including fire, storm, earthquake, vandalism, strike, accident or any other casualty (collectively, "Casualty"), and the Parent Company Parties shall have all right, title, and interest in and to the proceeds of any insurance it obtained and paid for covering such Casualty.

**6.2** If, during the term of this Leaseback Agreement, action is initiated to take the Hospital Premises or any portion thereof by eminent domain or condemnation proceedings, exercise of state authority under an executive order or by deed in lieu thereof (collectively, "**Condemnation**"), the Parent Company Parties, and not the Corporation, shall have all right, title, and interest in and to the award from the Condemnation.

**6.3** In the event of a Casualty or Condemnation, neither the Parent Company Parties nor the Corporation may terminate this Leaseback Agreement.

7. <u>Continued Access</u>. Following termination of this Leaseback Agreement, and until the entry of final decrees closing the Bankruptcy Cases, the Patient Care Ombudsman, appointed by the United States Trustee pursuant to Bankruptcy Code § 333 and approved by the Bankruptcy Court [*see* Docket No. 430], shall have continuing access to the Leased or Licensed Assets and related personnel during normal business hours and upon at least one (1) business day's prior written notice to the Parent Company Parties, for the purpose of winding down the Corporation's affairs, in connection with any litigation or adversary proceedings, and to effectuate the chapter 11 plan as approved by the Bankruptcy Court.

### 8. <u>Miscellaneous</u>.

**8.1** <u>Further Assurances</u>. Each of the Parties hereto agrees to execute and deliver any and all further agreements, documents or instruments necessary to effectuate this Leaseback Agreement and the transactions referred to herein or contemplated hereby or reasonably requested by the other Party to perfect or evidence their rights hereunder.

**8.2** <u>Notices and Demands</u>. All notices and demands, requests, consents, approvals, and other similar communications under this Leaseback Agreement shall be in writing and shall be sent by personal delivery or by either (a) United States certified or registered mail,

return receipt requested, postage prepaid, or (b) Federal Express or similar generally recognized overnight carrier regularly providing proof of delivery, addressed as follows:

If to the Corporation:	St. Francis Medical Center c/o Verity Health System of California, Inc. 601 South Figueroa Street, Suite 4050 Los Angeles, CA 90017-5704 Attention: Chief Executive Officer
With copies to: (which copy shall not constitute notice)	Tania Moyron, Esq. Dentons US LLP 601 South Figueroa St., Suite 2500 Los Angeles, CA 90017-5704
	and
	Hope Levy-Biehl, Esq. Davis Wright Tremaine LLP

865 S Figueroa St, Los Angeles, CA 90017

If to the Parent Company Parties:

With copies to: (which copies shall not constitute notice)

Any notice so given by mail shall be deemed to have been given as of the date of delivery (whether accepted or refused) established by U.S. Post Office return receipt or the overnight carrier's proof of delivery, as the case may be, whether accepted or refused. Any such notice not so given shall be deemed given upon receipt of the same by the Party to whom the same is to be given. Any Party hereto may designate a different address for itself by notice to the other Parties in accordance with this <u>Section 8.2</u>.

**8.3** <u>Payment of Expenses</u>. Except for the Parent Company Parties' obligation to be responsible for certain costs, fees, and expenses as set forth elsewhere in this Leaseback Agreement, each Party hereto shall bear its own legal, accounting, and other expenses incurred in connection with the preparation and negotiation of this Leaseback Agreement and the consummation of the transactions contemplated hereby, whether or not the transaction is consummated.

**8.4** <u>Rent</u>. The Corporation has prepaid the sum of One Thousand Dollars (\$1,000.00), the receipt of which is hereby acknowledged by the Parent Company Parties, and the Corporation shall not be required to pay the Parent Company Parties any additional rent under this Leaseback Agreement.

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 58 of 147

**8.5** Entire Agreement; Amendment; Waiver. This Leaseback Agreement, the IMA, and those provisions of the APA expressly identified in this Leaseback Agreement contain the entire agreement among the Parties with respect to the subject matter hereof and supersede all prior agreements, arrangements, and understandings relating to the subject matter hereof and thereof. There are no written or oral agreements, understandings, representations, or warranties among the Parties other than those set forth in this Leaseback Agreement. Nothing in this Leaseback Agreement modifies or shall be construed as modifying any orders entered by the Bankruptcy Court. This Leaseback Agreement may not be modified or amended except in writing signed by the Parties. No waiver of any term, provision or condition of this Leaseback Agreement in any one or more instances, shall be deemed to be or be construed as a further or continuing waiver of any such term, provision or condition of this Leaseback Agreement. No failure to act shall be construed as a waiver of any term, provision, condition or rights granted hereunder.

**8.6** <u>Assignment</u>. Neither this Leaseback Agreement nor the rights, duties or obligations arising hereunder shall be assignable or delegable by the Corporation or the Parent Company Parties without the prior written consent of the other Parties, which may be granted, denied or conditioned in such Party's absolute discretion except that the Parent Company Parties may assign this Leaseback Agreement in connection with any permitted assignment under the IMA. Subject to the foregoing, this Leaseback Agreement shall be binding upon, and inure to the benefit of, the respective successors and assigns of the Parties.

**8.7** <u>Joint Venture; No Third Party Beneficiaries</u>. Nothing contained herein shall be construed as forming a joint venture or partnership among the Parties with respect to the subject matter hereof. The Parties do not intend that any third party shall have any rights under this Leaseback Agreement.

**8.8** <u>Captions</u>. The section headings contained herein are for convenience only and shall not be considered or referred to in resolving questions of interpretation.

**8.9** <u>Governing Law</u>. This Leaseback Agreement shall be governed by and construed in accordance with the laws of the State of California without regard to its conflicts of laws principles or decisions, except to the extent that the laws of such State are superseded by the Bankruptcy Code or other applicable federal law.

**8.10** Jurisdiction. The Parties agree that the Bankruptcy Court shall have exclusive personal and subject matter jurisdiction over any action or proceeding with respect to any claim arising out of or related to this Leaseback Agreement, and any of the documents or transactions contained in or contemplated by this Leaseback Agreement; *provided, however*, that the Parties agree that the United States District Court for the Central District of California (together with the Bankruptcy Court, the "**Chosen Courts**") shall have exclusive jurisdiction over such claim if (i) the Bankruptcy Cases are closed and if the Bankruptcy Cases are not reopened to adjudicate such claim after request by the Party bringing such claim or (ii) the Bankruptcy Court determines that it does not have jurisdiction over such claim. Solely in connection with claims arising under this Leaseback Agreement, or any of the documents or transactions contemplated hereby, the Parties (a) irrevocably submit to the exclusive jurisdiction of the Chosen Courts, (b) waive any objection to laying venue in any such action or proceeding in the Chosen Courts, (c)

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 59 of 147

waive any objection that the Chosen Courts are an inconvenient forum or do not have jurisdiction over any Party hereto, and (d) agree that service of process upon such Party in any such action or proceeding shall be effective if notice is given in accordance with <u>Section 8.2</u> hereof

**8.11** <u>Conditions to Effectiveness</u>. The Parties acknowledge that this Leaseback Agreement is subject to and contingent upon the occurrence of the Closing under the APA. This Leaseback Agreement shall not be effective, nor shall any Party have any obligations hereunder, unless and until the Closing under the APA has occurred.

**8.12** <u>Cooperation on Regulatory Compliance Matters</u>. The Parent Company Parties understand and acknowledges that the Corporation intends to comply with applicable federal and state laws, regulations, and guidance, as well as the requirements or recommendations of any accrediting agencies. In the event the terms of this Leaseback Agreement need to be amended or supplemented based on guidance from or at the request or direction of a regulator made during the term of this Leaseback Agreement, the Parent Company Parties shall cooperate with such amendment and/or supplement to ensure the Corporation's ability to comply with such guidance, request, recommendation or directive.

**8.13** <u>Transition Services Agreement</u>. Pursuant to Section 1.4(f) and Section 1.5(d) of the APA, the Sellers and Parent Company will enter into a Transition Services Agreement to facilitate the winding down of the Sellers' businesses, the completion of the Bankruptcy Cases, and the dissolution of the Sellers. The Parties shall cooperate with each other to enable the Corporation and Parent Company to carry out their obligations under, and give effect to the terms of, the Transition Services Agreement.

**8.14** <u>Fair Meaning</u>. This Leaseback Agreement shall be construed according to its fair meaning and as if prepared by all Parties.

**8.15** <u>Counterparts</u>. This Leaseback Agreement may be executed by one or more of the Parties on any number of separate counterparts (including by telecopy), and all of said counterparts taken together shall be deemed to constitute one and the same instrument. This Leaseback Agreement may be delivered by facsimile or electronic transmission, including by e-mail as a PDF, and facsimile or PDF copies of executed signature pages, which shall be binding as originals.

8.16 <u>WAIVER OF JURY TRIAL</u>. TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT, OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (B) EACH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (C) EACH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (D) EACH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 8.16. THIS IS AN AFFIRMATIVE WAIVER OF THE PARTIES' RIGHTS TO A JURY TRIAL UNDER CALIFORNIA LAW, Cal. C. Civ. Pro. Sec 631. BY SIGNING BELOW ON THE SIGNATURE LINES, EACH PARTY IS EXPLICITLY WAIVING JURY TRIAL AND AUTHORIZING ANY AND ALL PARTIES TO FILE THIS WAIVER WITH ANY COURT AS THE WAIVER REQUIRED UNDER Cal. C. Civ. Proc. Sec. 631(f)(2).

[Signature Page Follows.]

# Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 61 of 147

**IN WITNESS WHEREOF,** the Parties agree to the foregoing terms of agreement through the execution below by their respective, duly authorized representatives as of the Signing Date.

Parent Company Parties:	Corporation:
NAME	ST. FRANCIS MEDICAL CENTER, a California non-profit public benefit corporation
Ву:	
Name:	By:
Title:	Name:
	Title:
By:	
Name:	
Title:	
By:	
Name:	
Title:	

# Exhibit 1.3(b) to APA

# Form of Interim Management Agreement

### **INTERIM MANAGEMENT AGREEMENT**

[Note: This IMA is drafted for 3 Purchaser parties: Purchaser/Parent Company, Hospital Newco and Manager. This IMA is drafted assuming Purchaser is buying hospital's accounts receivable. This structure is not required.]

This Interim Management Agreement (the "Agreement") is made and entered into as of \_\_\_\_\_\_\_, 2020 (the "Signing Date"), by and among St. Francis Medical Center, a California nonprofit public benefit corporation (the "Corporation") on the one hand, and \_\_\_\_\_\_\_ ("Parent Company") and \_\_\_\_\_\_\_ (the "Manager," and collectively with the Corporation and Parent Company, the "Parties" and each individually a "Party")) on the other hand.

### RECITALS

**A.** On August 31, 2018, the Corporation and certain of its affiliates each filed a voluntary petition for relief (collectively, the "**Bankruptcy Cases**") under chapter 11 of title 11 of the United State Code (the "**Bankruptcy Code**").

**B.** The Bankruptcy Cases are jointly administered under Lead Case No. 18-20151 and are currently pending in the Bankruptcy Court for the Central District of California in Los Angeles (the "**Bankruptcy Court**").

C. Parent Company, as purchaser, and the Corporation and certain of its affiliates (collectively, the "Sellers" as defined in the APA), as seller, entered into that certain Asset Purchase Agreement, dated of even date herewith (the "APA"), which provides for the sale of certain assets of the Sellers (collectively, the "Assets"). All terms not otherwise defined herein shall have the meaning ascribed to them in the APA.

**D.** The Corporation operates St. Francis Medical Center as a 384-bed general acute care hospital (the "**Hospital**"), with its primary location at 3630 East Hospital Newco Highway, Lynwood, California 90262, and such other locations where the Hospital's services are provided (collectively, the "**Hospital Premises**")

**E.** Pursuant to the APA, Parent Company agreed to purchase the Assets, which include certain assets of the Corporation (the "Corporation's Assets").

F. Parent Company designated its affiliate, \_\_\_\_\_ ("Hospital Newco") as the owner of the Corporation's Assets, and the Manager as the operator of the Hospital.

**G.** Parent Company and the Manager have requested this Agreement to afford the Manager additional time to obtain its general acute care hospital license from the California Department of Public Health ("**CDPH**"), and its hospital pharmacy permit from the California Board of Pharmacy (together, the "**New Licenses**," and the date on which the New Licenses are issued is the "**Licensure Date**").

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 64 of 147

**H.** Pursuant to the APA, that portion of the Assets constituting drugs, dangerous devices, pharmacy systems, or other pharmacy assets (the "**Pharmacy Assets**"), the Medicare/Medi-Cal Agreements, the Lockboxes, the Transferred Managed Care Agreements and the Transferred Private Payor Agreements (collectively, the "**Licensure Date Assets**") shall transfer to Hospital Newco as of the Licensure Date.

**I.** The Corporation shall maintain a possessory interest in the Hospital and the Hospital Premises, and Parent Company and Hospital Newco on the one hand as lessor, and the Corporation, on the other hand as lessee, are entering into that certain Sale Leaseback Agreement of even date herewith, pursuant to which certain of the Corporation's Assets will be leased back to the Corporation (the "Leaseback Agreement").

J. Until Hospital Newco obtains the New Licenses, the Manager desires to assume the management of the Hospital, including its pharmacy, on behalf of the Corporation, and the Corporation desires to avail itself of such management services, upon the terms and conditions set forth in this Agreement.

### **TERMS OF AGREEMENT**

**NOW, THEREFORE,** in consideration of the promises and the mutual covenants contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby agree as follows:

### 1. Term.

**1.1** The term of this Agreement (the "**Management Period**") shall commence as of the Effective Time (as defined in the APA) and shall continue until the earlier of the Licensure Date or [June 30, 2021]. The Parties acknowledge that, during the Management Period, the Corporation shall remain the licensee of the Hospital, and in that capacity, and during such period, shall retain statutory and regulatory authority and responsibility for the Hospital and for oversight of the Manager.

### 2. Acknowledgements and Covenants.

**2.1** As of the Effective Time, Parent Company shall have acquired all of the Corporation's Assets as set forth in the APA (other than the Licensure Date Assets), which shall be leased back to the Corporation pursuant to the terms of the Leaseback Agreement during the term of this Agreement.

**2.2** Pursuant to Section 1.3 of the APA, contemporaneously with the date the New Licenses are issued and the termination of the Leaseback Agreement, ownership of the Licensure Date Assets shall be transferred to Hospital Newco. Parent Company and the Manager acknowledge that this Agreement and the foregoing subsequent transfer of the Licensure Date Assets are made at the request of Parent Company and the Manager, to provide more time for the Manager to obtain its New Licenses. As such, Parent Company acknowledges, covenants, and agrees that the total Purchase Price (as defined in the APA) shall be paid to the Sellers under the APA as if the Licensure Date Assets were transferred as of the Effective Time.

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 65 of 147

**2.3** As of the Effective Time, the Corporation shall have ended the employment or engagement of all employees and contractors and, to the extent they received offers of employment or engagement from the Manager (or its affiliates) and accepted such offers, shall have been transferred to the Manager (or its affiliate) as employees or contractors of the Manager (collectively, the "**Hired Employees**").

**2.4** During the Management Period, the Corporation will (a) assign an individual to oversee the operation of the Hospital and serve as the president of the Corporation, as required by the Centers for Medicare & Medicaid Services, CDPH, and applicable California laws and regulations (the "**Responsible Officer**"), and (b) maintain proper oversight by a board of directors.

### **3.** Appointment of Manager.

**3.1** During the Management Period, the Corporation hereby appoints the Manager as the sole and exclusive provider of the Services (defined below) and hereby grants to the Manager the exclusive right to manage the Hospital under the Corporation's Licenses (as defined in the APA) as a general acute care hospital, including without limitation, the right to undertake those certain management responsibilities and permitted activities described in <u>Section 4</u> below. The Manager hereby accepts such appointment for all purposes with respect to the Corporation's rights, duties, and responsibilities under the Licenses for the Hospital, to the fullest extent permitted by law, and agrees, to the fullest extent permitted by law, to provide management services to the Hospital on behalf of the Corporation (the "**Services**").

**3.2** The Manager's Services hereunder shall include management and operation of the Hospital's pharmacy on behalf of the Hospital, even though the Pharmacy Assets will not be transferred to Hospital Newco pursuant to the APA until the Licensure Date.

**3.3** Upon the Licensure Date, the Manager's Services provided to the Hospital under the Corporation's Licenses shall terminate and, thereafter, Hospital Newco (and its affiliates) will be operating the Hospital as the licensee holding its own New Licenses.

**3.4** During the Management Period, the Manager shall submit claims for services rendered by the Hospital to various governmental and non-governmental entities, patients, and other third parties pursuant to the Corporation's provider agreements, payor contracts, and NPI numbers as set forth in <u>Schedule 3.4</u>, attached hereto (collectively, the "Corporation's Billing Credentials"). Because all billing and collecting shall be under the Corporation's Billing Credentials, payments shall be made in the Corporation's name and deposited in the Corporation's bank accounts. The Parties acknowledge and agree that during the Management Period, the Corporation's control. Consistent with <u>Section 4.5(b)</u>, the Manager may transfer the Manager Compensation to its accounts. To the extent the Manager is unable to make such transfers to its own accounts, upon request by the Manager, the Corporation will, at the Manager's cost, forward to the Manager any payment or reimbursement received during the Management Period.

**3.5** To the extent the Corporation's lenders continue to have liens, security interestsclaims, charges, mortgages, security interests, rights of setoff, recoupment, netting and deductions, rights of first offer, first refusal and any other similar contractual property, legal or

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 66 of 147

<u>equitable rights, and any successor or successor-in-interest liability theories</u>, or any other encumbrances whatsoever (collectively, "**Encumbrances**") on the Corporation's bank accounts or lockboxes as of the Effective Time, the Corporation, Parent Company, and the Manager shall work cooperatively to facilitate termination and release of such Encumbrances.

**3.6** The Corporation shall retain and, upon the Licensure Date, the Manager shall assume, any contracts necessary for the Corporation to continue to be the holder of the Licenses and to bill for Hospital services during the Management Period, in accordance with the APA. Such contracts shall be as set forth in <u>Schedule 3.6</u>, attached hereto (the "**Retained Contracts**"). Parent Company hereby designates each Retained Contract as an Assigned Contract under Section 1.11 of the APA.

### 4. Management Responsibility.

**4.1** During the Management Period, the Manager shall, subject to all applicable legal and regulatory requirements and the Corporation's ultimate oversight and control, have responsibility for the management of the Hospital, and agrees to assume and discharge all responsibilities, duties, liabilities, payments, and obligations in connection with properly maintaining the Hospital in full compliance with all regulations and standards required of a general acute care hospital facility so licensed. In furtherance thereof, the Manager's Services shall include, but not be limited to, the following duties, which duties shall be performed at the Manager's sole cost and expense:

(a) Managing the operations of the Hospital as a general acute care hospital in compliance with all applicable laws, regulations, provider agreements, payor contracts, CDPH requirements for maintenance of the Licenses in good standing, Medicare conditions of participation and requirements for payment with respect to governmental programs, and the requirements for maintenance of the Hospital's accreditations;

(b) Employing and managing the Hired Employees and any other non-clinical and clinical personnel (i) deemed necessary by the Corporation for the operation of the Hospital as a general acute care hospital, or (ii) required by law so as to meet all applicable labor laws and regulations, and consistent with orders of the Bankruptcy Court;

(c) Ensuring that the Corporation is able to pay itself out of the Hospital Revenues for the costs and expenses set forth in <u>Sections 4.5(b)</u>;

(d) Maintaining and repairing, as needed, the Hospital Premises so as to ensure material compliance with all applicable local, state and federal law, and construction timelines imposed by OSHPD.

(e) Providing security services reasonably necessary to prevent unlawful entry or damage to the Hospital Premises;

(f) Affording the Responsible Officer or his or her designee access, during normal business hours, to the Hospital Premises, the books and records at the Hospital Premises or in Manager's possession, the Hired Employees and any other personnel of Manager or otherwise

### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 67 of 147

who are providing services associated with the operation of the Hospital, and such other access and assistance as reasonably requested by the Responsible Officer;

(g) Upon at least one (1) day's prior written notice to the Manager, providing access, during normal business hours, to the Hospital Premises to lessors of equipment at the Hospital Premises, if any, who have been authorized by order of the Bankruptcy Court to remove their equipment from the Hospital Premises, provided that the Manager shall have full power and authority to require that the removal of such equipment by such lessors does not damage the Hospital Premises;

(h) Providing access, during normal business hours, to the Hospital, the Hospital Premises, the Hospital's books and records, electronic health records, financial information systems, operating systems, laboratory systems, the Hired Employees and any other personnel of Manager or otherwise who are providing services associated with the operation of the Hospital, to the Corporation, the Corporation's directors, officers and representatives, and the Corporation's successors in interest, including, but not limited to, any plan administrator, liquidating trustee or similar representatives appointed or approved by the Bankruptcy Court, as well as the Patient Care Ombudsman appointed in the Bankruptcy Cases under Section 333 of the Bankruptcy Code (the "PCO") [see Docket No. 430], for the purpose of winding down the Corporation's affairs, pursuing litigation and adversary proceedings, and to effectuate a plan of liquidation, as approved by the Bankruptcy Court (the "Plan"), provided, however, that such access does not unreasonably disrupt the Hospital's operations;

(i) Maintaining, all licenses, permits consents, approvals, accreditations, and certifications currently held by the Corporation in good standing, in active status, and in compliance with all applicable local, state, and federal laws, including the timely payment of all applicable fees to support or renew these approvals;

(j) Maintaining and obtaining all insurance coverages, from and after the Effective Time, for the Hospital that a prudent hospital operator or owner would maintain, including directors and officers insurance with no less coverage than was maintained for directors and officers just prior to the Effective Time;

(k) Maintaining and obtaining those insurance coverages required under the Leaseback Agreement, for its own account, with the Corporation included as a named insured, and paying all amounts required under the Leaseback Agreement in a timely manner, including rent, utilities, taxes, and insurance premiums;

(1) Opening and forwarding all mail relating to the financial or business affairs of the Corporation to the notice address below;

(m) Periodically reporting to the Corporation (or its designee), either in person or telephonically, the condition of the Hospital and the Hospital Premises;

(n) Coordinating with the governing board and the organized medical staff (each, as established by the Corporation and the Sellers) on the appropriateness and quality of medical care and all medical staff issues requiring governing board oversight;

### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 68 of 147

(o) Paying all costs and expenses in connection with and incidental to ownership of the Corporation's Assets and management and operation of the Hospital hereunder, including but not limited to, all the Hospital operating costs, employee-related costs, and taxes, whether or not identified, described, or referenced in this Agreement;

(p) Cooperating with the Corporation in facilitating termination and release of any Encumbrances on the Corporation's bank accounts and lockboxes; and

(q) Performing such other duties and activities as are reasonably necessary for the Manager to fulfill its responsibilities under this Agreement and the APA.

**4.2 Permitted Manager Activities**. During the Management Period, the Manager may do any of the following, in consultation with the Corporation and subject to the requirements of applicable local, state, and federal law, which activities may be performed by the Manager at the Manager's sole cost and expense:

(a) Make alterations, improvements, and repairs to the interior or exterior of the Hospital Premises, including structural alterations, improvements, and repairs;

(b) Remove and dispose of furniture, fixtures, equipment (other than equipment owned by equipment lessors), and supplies at the Hospital Premises;

(c) Move into and install furniture, fixtures, equipment, and supplies at the Hospital Premises;

(d) Prepare the Hospital for a name change, except that no such name change may take effect, and no signage reflecting such change shall be installed, during the Management Period; and

(e) Perform, or permit to be performed, any other activities at the Hospital Premises that are not inconsistent with operating the Hospital under the Licenses, and receiving and retaining for the Manager's own account all revenues and proceeds of any such activities, to the extent they comprise the Manager Compensation.

### 4.3 **Prohibited Manager Activities.**

(a) Notwithstanding anything to the contrary in this Agreement, the Manager shall have no authority to take and shall not take any action with respect to any Excluded Assets or Excluded Liabilities (as such terms are defined in the APA) of the Corporation.

(b) The Manager's authority to manage and operate the Hospital is limited to those actions that Manager is expressly required or permitted to do hereunder.

(c) The Manager shall not (i) take any action that interferes with the Corporation's transfer of funds to pay itself out of the Hospital Revenues as set forth in <u>Section</u> 4.5(b); or (ii) remove, withdraw, or authorize removal or withdrawal of funds from the Corporation's bank accounts or lockboxes to the extent that the Corporation would be unable to fully pay itself as set forth in <u>Section 4.5(b)</u>.

### 4.4 APA Provisions; Transition Services Agreement.

(a) Nothing herein shall modify the prorations of expenses and utilities set forth in the APA, including specifically as set forth in <u>Section 1.6</u> of the APA.

(b) None of the information accessed, learned or obtained by Parent Company or the Manager or any of their affiliates in the course of performing their duties hereunder may serve as the basis for payment of less than the full Purchase Price or to otherwise assert a claim against the Sellers.

(c) Nothing herein shall modify the transfer of the Assets from the Sellers to Parent Company as contemplated in the APA, including specifically as set forth in <u>Section 1.7</u> of the APA.

(d) Nothing herein shall modify the APA in respect of the exclusion from purchase by Parent Company of the Excluded Assets, including specifically as set forth in Section 1.8 of the APA.

(e) Pursuant to Section 1.4(f) and Section 1.5(d) of the APA, the Sellers and Parent Company will be entering into a Transition Services Agreement to facilitate the winding down of the Sellers' businesses, the completion of the Bankruptcy Cases, and the dissolution of the Sellers. The Parties shall cooperate with each other to enable the Corporation and Parent Company to carry out their obligations under, and give effect to the terms of, the Transition Services Agreement.

### 4.5 Manager Compensation; Distribution of Revenue.

(a) As used in this Agreement, the term "**Hospital Revenues**" shall mean all revenues and reimbursement received by the Corporation after the Effective Time, including, but not limited to, cash, accounts, notes, or other accounts receivable, disproportionate share payments, quality assurance fee payments, Seller Cost Report (as defined in the APA) settlements, and capitation premiums, whether payable by Medicare, Medi-Cal or any other commercial or governmental payor, or any health maintenance organization or any other managed care program or any private pay patients, but only to the extent they are not Excluded Assets (as defined in the APA).

(b) As full and complete payment for the Manager's Services, the Manager shall be entitled to receive an amount equal to the Hospital Revenues less the following amounts that will be paid to the Corporation by Manager (the "**Manager Compensation**"):

(i) All costs and expenses incurred by the Corporation for the Corporation's purchase of drugs and dangerous devices that the Manager determines are necessary for the operation of the Hospital;

(ii) All salaries, stipends, costs and expenses associated with the employment or engagement of the board of directors of the Corporation and its parent organization, VHS, the Responsible Officer and any other retained employees or contractors of the

### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 70 of 147

Corporation primarily or exclusively providing services necessary for the operation of the Hospital and compliance with applicable laws, if any; and

(iii) All other costs and expenses associated with keeping the Corporation in good standing, maintaining its Licenses and maintaining the Retained Contracts, including, but not limited to, filings with the secretary of state, filing tax returns, board of directors expenses, directors and officers insurance, employment practices liability insurance, compensation, benefits, and such other costs and expenses incurred by a hospital corporation in the normal course of business.

(c) Corporation shall instruct the financial institutions that service its bank accounts or lockboxes to, immediately following the Effective Time and during the Management Period, sweep all Hospital Revenues from the Corporation's bank accounts or lockboxes on a daily basis into an account designated in writing by Manager, and Corporation shall not revise such written instructions without obtaining the prior written consent of Manager.

(d) Under no circumstance shall the Manager seek payment for the Manager's Services from the liquidating trustee, the Corporation, any of the Corporation's officers, directors, agents, contractors, personnel, affiliates or subsidiaries, Verity Health System of California, Inc. ("VHS"), or any of VHS' officers, directors, agents, contractors, personnel, affiliates or subsidiaries.

(e) The Corporation shall issue invoices to the Manager on a weekly basis for the items described in Section 4.5(b) (i), (ii) and (iii), with reasonable supporting detail therefor. The Manager shall pay such invoices within ten (10) business days of receipt of such invoices. If the Manager does not remit payment in respect of such invoices in accordance with the immediately preceding sentence, the unpaid amount of such invoices shall bear interest at the Prime Rate in effect on the calendar day upon which such payment was required to be made to the Corporation (the "**Invoice Payment Due Date**") plus five percent (5%) (or the maximum rate allowed by law, whichever is less), such interest accruing on each calendar day after the Invoice Payment Due Date until payment of such invoices and all interest thereon is made to the Corporation.

### 4.6 Liabilities and Losses.

(a) Parent Company shall be responsible for all the Hospital liabilities and losses incurred or accrued during the Management Period.

(b) Nothing hereunder shall, or is intended to, modify or supersede Parent Company's responsibility for the Transferred Obligations (as defined in the APA) or the Corporation's responsibility for the Excluded Liabilities (as defined in the APA).

(c) Specifically, with respect to any and all quality assurance fee ("QAF") liabilities owed by the Corporation, Parent Company shall pay all QAF fees that become due and owing or are overdue during the Management Period, even though such fees may have been calculated based on data prior to the Effective Time.

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 71 of 147

(d) If Parent Company (or its affiliate) has designated any health plan capitation agreements to be assigned to Hospital Newco, then Parent Company hereby assumes all liabilities and responsibilities related to managing the members covered by such health plan capitation agreements, including arranging for services from and payment under all related risk pool arrangements, all related arrangements with independent physician associations (IPAs), all downstream provider agreements, and all out-of-network provider arrangements or services, without regard to whether the member services were rendered prior to or after the Effective Time.

**4.7** The Corporation's Ultimate Control. Notwithstanding anything to the contrary in this <u>Section 4</u> or in this Agreement more generally, the Corporation, as holder of the Licenses, shall remain ultimately responsible for the operation of the Hospital, and may, at any time and from time-to-time during the Management Period, take any action (at Parent Company's or the Manager's sole cost and expense) necessary to ensure the Corporation's compliance with applicable laws and regulations, even if such action requires the Corporation to intervene in the Manager's performance of the Manager's duties or permitted activities pursuant to this <u>Section 4</u>.

### 5. Continued Responsibility of the Corporation.

**5.1** During the Management Period, the Corporation shall maintain (at Parent Company's or the Manager's sole cost and expense), and shall not take or voluntarily permit any actions which may adversely affect, the Corporation's corporate existence and its full rights as the licensee under the Licenses. In addition, during the Management Period, the Corporation and its officers shall reasonably cooperate with the Manager (at Parent Company's or the Manager's sole cost and expense) in the Manager's provision of the Services.

5.2 Notwithstanding the statutory and regulatory authority and responsibility of the Corporation for the continued management of the Hospital during the Management Period, the Parties recognize and acknowledge that under this Agreement, the Manager shall, subject to the ultimate oversight by and approval of the Corporation, be responsible for the day-to-day operation and maintenance of the Hospital as a general acute care hospital. In the event that any violation or alleged violation of or non-compliance with any statute or regulation applicable to the operation or maintenance of the Hospital as a general acute care hospital certified by the Medicare and Medi-Cal programs occurs during the Management Period, then without regard to legal or statutory fault on the part of the Manager or of the Corporation, the Manager shall immediately notify the Corporation of such violation or alleged violation or non-compliance and take reasonable efforts to avoid or minimize any related adverse consequences. Parent Company and the Manager shall be responsible for the costs of any penalty, fine or remediation identified during the Management Period arising out of or relating to any act, omission, event or occurrence connected with the operation of the Hospital after the Effective Time, including, without limitation, the cost of engaging third party consultants or experts to help address or resolve the violation, alleged violation or non-compliance, and shall indemnify and hold the Corporation harmless for the same in accordance with Section 9.2. The Corporation retains the right to join the Manager in contesting said violations upon providing the Manager with notice of its intent to do so.

### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 72 of 147

**5.3** The Corporation shall be responsible for purchasing drugs and dangerous devices that the Manager determines are necessary for the operation of the Hospital at the Manager's sole cost and expense, as set forth in Section 4.5(b)(i).

**5.4** The Corporation agrees to execute and deliver to the Manager such documents as the Manager may reasonably request to maintain the hospital license active and in good standing with CDPH and the other Licenses necessary or appropriate to maintain the Hospital as a general acute care hospital and to facilitate the Manager's obtaining of the New Licenses.

6. The Parties' Cooperation with Regulatory Agencies. The Manager shall use its best efforts to obtain the New Licenses as expeditiously as possible. The Manager shall provide updates to the Corporation weekly, and as requested by the Corporation, on the status of the Manager's efforts to obtain the New Licenses. To the extent not already submitted prior to the Effective Time, the Manager shall provide a copy of all such initial and supplementary or amended license applications for the New Licenses to Corporation at least two (2) business days prior to submission. The Manager and the Corporation shall cooperate with each other in the event the Corporation requests revisions to a license application for a New License. The Corporation shall, at Parent Company's or the Manager's cost, reasonably cooperate with the Manager's efforts to obtain the New Licenses, and may communicate and coordinate with licensing agencies as necessary in connection with obtaining the New Licenses. Notwithstanding the foregoing, obtaining all governmental consents, approvals, assignments, authorizations, and clearances necessary to obtain the New Licenses shall be solely Parent Company's and the Manager's (and not the Corporation's) responsibility, including payment of any fees, expenses, filing costs or other amounts related thereto.

### 7. Risk of Loss.

7.1 The Corporation assumes no risks or liability for damage to or injury occurring to the Hospital Premises, Assets or the Hospital during the term of this Agreement by any means whatsoever, including fire, storm, earthquake, vandalism, strike, cyberattack, accident or any other casualty (collectively, "**Casualty**"), and Parent Company shall have all right, title, and interest in and to the proceeds of any insurance it obtained and paid for covering such Casualty.

7.2 If, during the term of this Agreement, action is initiated to take the Hospital Premises or any portion thereof by eminent domain proceedings or by deed in lieu thereof (collectively, "Condemnation"), Parent Company, and not the Corporation, shall have all right, title, and interest in and to the award from the Condemnation.

**7.3** In the event of a Casualty or Condemnation, neither Parent Company nor the Corporation may terminate this Agreement.

8. Continued Access. Following the expiration or termination of this Agreement and until the entry of final decrees closing the Bankruptcy Cases, if expressly authorized or directed by the Bankruptcy Court, and only to the extent so authorized or directed, the PCO shall have continuing access to the Corporation's Assets and related personnel during normal business hours and upon at least one (1) business day's prior written notice to Parent Company or the Manager, for the purpose of winding down the Corporation's affairs, in connection with any litigation or adversary

proceedings, and to effectuate the Plan.

### 9. Exculpation; Indemnification.

**9.1** The Corporation and the Corporation's affiliates, members, officers, directors, employees, attorneys, accountants, consultants, agents, representatives, successors and assigns, including the liquidating trustee and responsible officer (collectively the "Corporation Indemnified Parties") shall have no liability in contract, tort or otherwise unless and until a Chosen Court finds in a final, non-appealable judgment that any Damages result solely from a Corporation Indemnified Party's gross negligence or willful misconduct.

9.2 Parent Company and the Manager shall promptly and fully keep and hold the Corporation Indemnified Parties forever harmless from, and shall indemnify and defend the Corporation Indemnified Parties from and against, without regard to materiality, any and all obligations, judgments, fines, civil money penalties, sanctions, awards, liabilities, losses, penalties, claims, costs, demands, damages, expenses, liens, and encumbrances, including investigation costs, time spent in depositions and reasonable attorneys' fees and expenses (collectively, "Damages"), whether civil or criminal, direct, indirect or consequential and no matter how arising, in any way related to, connected with, arising or resulting from, or under this Agreement, the APA, the Hired Employees (as defined in the APA), the Manager's performance of the Services, or the operation or management of the Hospital or the Corporation's Assets, in each instance after the Effective Time. Notwithstanding the foregoing, the Parties understand that except as otherwise specifically provided for in the APA, Parent Company and the Manager are not, by virtue of this Agreement or any term or provision herein, assuming any claim, liability, expense, debt or other obligation of the Corporation that both relates to the operation or management of the Hospital or the Corporation's Assets prior to the Effective Time and constitutes an Excluded Liability under the APA.

10. HIPAA Compliance. The Manager agrees to take such steps as are necessary to ensure compliance with the Health Insurance Portability and Accountability Act of 1996 (HIPAA), the California Confidentiality of Medical Information Act, and other applicable federal and state privacy laws and regulations (collectively, the "Privacy Laws") with respect to the Hospital and its operations, and the Corporation agrees not to take or voluntarily permit any actions which violate Privacy Laws with respect to the Hospital or its operations. Toward this end, Parent Company, the Manager, and the Corporation agree to execute and deliver that certain Business Associate Agreement, attached hereto as Exhibit A and incorporated by reference herein, upon execution of this Agreement.

**11. Further Assurances**. Each of the Parties hereto agree to execute and deliver any and all further agreements, documents or instruments necessary to effectuate this Agreement and the transactions referred to herein or contemplated hereby or reasonably requested by any other Party hereto to perfect or evidence their rights hereunder.

**12. Relationship of Parties**. In performing their duties and permitted activities under this Agreement, Parent Company, the Manager, and the Corporation shall, at all times be acting and performing as independent contractors. Parent Company, the Manager, and the Corporation are not partners or joint venturers with each other and nothing herein shall be construed as making

them partners or joint venturers or imposing upon any of them any liability as partners or joint venturers.

13. Notices and Demands. All notices and demands, requests, consents, approvals, and other similar communications under this Agreement shall be in writing and shall be sent by personal delivery or by either (a) United States certified or registered mail, return receipt requested, postage prepaid, or (b) Federal Express or similar generally recognized overnight carrier regularly providing proof of delivery, addressed as follows:

If to the Corporation:	<ul> <li>St. Francis Medical Center</li> <li>c/o Verity Health System of California, Inc.</li> <li>601 South Figueroa Street, Suite 4050</li> <li>Los Angeles, CA 90017-5704</li> <li>Attention: Chief Executive Officer</li> </ul>
With copies to: (which copy shall not constitute notice)	Tania Moyron, Esq. Dentons US LLP 601 South Figueroa St., Suite 2500 Los Angeles, CA 90017-5704

and

Hope Levy-Biehl, Esq. Davis Wright Tremaine LLP 865 S Figueroa St, Los Angeles, CA 90017

If to Parent Company and/or the Manager:

With copies to: (which copies shall not constitute notice)

Any notice so given by mail shall be deemed to have been given as of the date of delivery (whether accepted or refused) established by U.S. Post Office return receipt or the overnight carrier's proof of delivery, as the case may be, whether accepted or refused. Any such notice not so given shall be deemed given upon receipt of the same by the Party to whom the same is to be given. Any Party hereto may designate a different address for itself by notice to the other Party in accordance with this <u>Section 13</u>.

### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 75 of 147

14. Expenses. Except for Parent Company's and the Manager's obligations to be responsible for certain costs, fees, and expenses as set forth elsewhere in this Agreement, each Party to this Agreement shall pay its own expenses in connection with the preparation of this Agreement and the consummation of the transactions contemplated hereby, including the fees of any attorneys, accountants, financial advisors, investment bankers or other professionals engaged by such Party.

**15.** Entire Agreement. This Agreement, the Leaseback Agreement, and those provisions of the APA expressly identified in this Agreement, contain the entire agreement among the Parties with respect to the subject matter hereof and supersede all prior agreements, arrangements, and understandings relating to the subject matter hereof and thereof. There are no written or oral agreements, understandings, representations, or warranties among the Parties other than those set forth in this Agreement, the Leaseback Agreement, and those provisions of the APA expressly identified in this Agreement. Nothing in this Agreement modifies or shall be construed as modifying any orders entered by the Bankruptcy Court.

**16. Amendment**. This Agreement may not be modified, amended, altered or supplemented except by a written agreement executed by all the Parties.

17. Waiver. Waiver by any Party of any breach or failure to comply with any provision of this Agreement by any other Party shall not be construed as or constitute, a continuing waiver of such provision, or a waiver of any other breach of, or failure to comply with, any other provision of this Agreement. No waiver of any such breach or failure or of any term or condition of this Agreement shall be effective unless in a written notice signed by the waiving Party und delivered, in the manner required for notices generally, to each affected Party.

18. Severability. In case any provision of this Agreement shall be found by a court of competent jurisdiction to be invalid, illegal or unenforceable, such provision shall be construed and enforced as if it had been narrowly drawn so as not to be invalid, illegal or unenforceable, and the validity, legality and enforceability of the remaining provisions of this Agreement shall not in any way be affected or impaired thereby.

**19.** Successors and Assigns. This Agreement shall be binding upon, and shall inure to the benefit of, the Parties and their respective permitted successors and assigns. The Corporation shall not be permitted to assign its rights or its obligations under this Agreement without the prior consent of Parent Company and the Manager. The Parties further acknowledge and agree that the Manager may subcontract for any of the goods or services required to be provided by the Manager pursuant to this Agreement, and the Manger may assign any of its rights hereunder and/or delegate any of its obligations hereunder, so long as in each case the Manager remains responsible for such subcontracted goods or services and for any of such Manager obligations hereunder.

**20.** Attorneys' Fees. In the event of any litigation or arbitration between the Parties arising out of this Agreement, the prevailing Party therein shall be allowed to recover from the other Party all court costs and reasonable attorneys' fees which shall be fixed by the court or arbitrator.

**21. Headings**. The descriptive headings of sections and subsections of this Agreement are inserted for convenience only and do not constitute a part of this Agreement and shall not affect in any way the meaning or interpretation of this Agreement.

**22. Governing Law**. This Agreement shall be governed by and construed in accordance with the laws of the State of California without regard to its conflicts of laws principles or decisions, except to the extent that the laws of such State are superseded by the Bankruptcy Code or other applicable federal law.

23. Jurisdiction. The Parties agree that the Bankruptcy Court shall have exclusive personal and subject matter jurisdiction over any action or proceeding with respect to any claim arising out of or related to this Agreement, and any of the documents or transactions contained in or contemplated by this Agreement; provided, however, that the Parties agree that the United States District Court for the Central District of California (together with the Bankruptcy Court, the "Chosen Courts") shall have exclusive jurisdiction over such claim if (i) the Bankruptcy Cases are closed and if the Bankruptcy Cases are not reopened to adjudicate such claim after request by the Party bringing such claim or (ii) the Bankruptcy Court determines that it does not have jurisdiction over such claim. Solely in connection with claims arising under this Agreement, or any of the documents or transactions contemplated hereby, the Parties (a) irrevocably submit to the exclusive jurisdiction of the Chosen Courts, (b) waive any objection to laying venue in any such action or proceeding in the Chosen Courts, (c) waive any objection that the Chosen Courts are an inconvenient forum or do not have jurisdiction over any Party hereto, and (d) agree that service of process upon such Party in any such action or proceeding shall be effective if notice is given in accordance with Section 13 hereof.

**24. Commencement**. The Parties acknowledge that commencement of the Manager's Services under this Agreement is subject to and contingent upon the occurrence of the Closing (as defined in the APA). The Management Period shall not commence, and the Parties' obligations during the Management Period shall not commence, unless and until the Closing has occurred.

25. Cooperation on Regulatory Compliance Matters. Parent Company and the Manager understand and acknowledge that the Corporation intends to comply with applicable federal and state laws, regulations, and guidance. In the event the terms of this Agreement need to be amended or supplemented based on guidance from or at the request or direction of a regulator made during the term of this Agreement, Parent Company and the Manager shall cooperate with such amendment and/or supplement to ensure the Corporation's ability to comply with such guidance, request or directive.

26. Counterparts. This Agreement may be executed by one or more of the Parties on any number of separate counterparts (including by telecopy), and all of said counterparts taken together shall be deemed to constitute one and the same instrument. It may be delivered by facsimile or electronic transmission, including by e-mail as a PDF, and facsimile or PDF copies of executed signature pages, which shall be binding as originals.

**27. WAIVER OF JURY TRIAL**. TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT,

OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (B) EACH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (C) EACH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (D) EACH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 27. THIS IS AN AFFIRMATIVE WAIVER OF THE PARTIES' RIGHTS TO A JURY TRIAL UNDER CALIFORNIA LAW, Cal. C. Civ. Pro. Sec 631. BY SIGNING BELOW ON THE SIGNATURE LINES, EACH PARTY IS EXPLICITLY WAIVING JURY TRIAL AND AUTHORIZING ANY AND ALL PARTIES TO FILE THIS WAIVER WITH ANY COURT AS THE WAIVER REQUIRED UNDER Cal. C. Civ. Proc. Sec. 631(f)(2).

[Signature page follows.]

### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 78 of 147

**IN WITNESS WHEREOF,** the Parties agree to the foregoing terms of agreement through the execution below by their respective, duly authorized representatives as of the Signing Date.

### MANAGER:

## **CORPORATION:**

NAME

	ST. FRANCIS MEDICAL CENTER,
	a California non-profit public benefit
_	corporation

By:	
Name:	
Title:	

By:			
Name:			
Title:			

### **PARENT COMPANY:**

NAME

By:	
Name:	
Title:	

### **HOSPITAL NEWCO:**

NAME

By:	
Name:	
Title:	

Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 79 of 147

# **Exhibit A to IMA**

**Business Associate Agreement** 

# Schedule 3.4 to IMA

**Corporation's Billing Credentials** 

# Schedule 3.6 to IMA

**Retained Contracts** 

Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 82 of 147

# Exhibit 1.4(a) to APA

### Form of Bill of Sale

# Exhibit 1.4(b) to APA

# Form of Real Estate Assignment Agreements

Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 84 of 147

# Exhibit 1.4(c) to APA

## Form of Quitclaim Deed

# Exhibit 1.4(d) to APA

# Form of Transfer Agreement

# Exhibit 1.4(f) to APA

# Form of Transition Services Agreements

See attached.

## TRANSITION SERVICES AGREEMENT

This Transition Services Agreement (the "Agreement"), is made and entered into as of ..., 2020 (the "Signing Date"), by and among Verity Health System of California, Inc., a California nonprofit public benefit corporation ("Verity"), Verity Holdings, LLC, a California limited liability company ("Verity Holdings"), St. Francis Medical Center, a California nonprofit public benefit corporation ("St. Francis" and together with Verity and Verity Holdings, collectively the "Sellers" and each individually a "Seller") and ["Purchaser"]). The Sellers the Liquidating Trust and Purchaser may

("**Purchaser**"). The Sellers, the Liquidating Trust, and Purchaser may each be referred to herein as a "**Party**" and collectively as the "**Parties**."

This Agreement is being entered into in connection with the Asset APA, dated of even date herewith among Sellers and Purchaser (the "APA"). Capitalized terms used in this Agreement but not defined herein shall have the meanings given to them in the APA. The Parties have agreed to provide or cause one or more of their respective direct or indirect subsidiaries or affiliates to provide certain services as set forth below. The Party providing services hereunder (together with those of its direct and indirect subsidiaries and affiliates providing services hereunder) is referred to herein as the "Service Provider" and the Party receiving services hereunder (together with those of its direct and indirect subsidiaries and affiliates receiving services hereunder) is referred to herein as the "Service Recipient". The Parties agree that this Agreement will be amended to include the Liquidating Trust (as defined below) after its creation, pursuant to the effective date set forth in the plan of liquidation (the "Plan") to be confirmed by the Bankruptcy Court.

# RECITALS

A. On August 31, 2018, the Sellers and certain of their affiliates each filed a voluntary petition for relief (collectively, the "**Bankruptcy Cases**") under chapter 11 of title 11 of the United State Code (the "**Bankruptcy Code**").

B. The Bankruptcy Cases are jointly administered under Lead Case No. 18-20151 and are currently pending in the Bankruptcy Court for the Central District of California in Los Angeles (the "**Bankruptcy Court**").

C. Purchaser and the Sellers entered into the APA, which provides for the sale of the assets of the Sellers (collectively, the "Assets").

D. Sellers will file a motion with the Bankruptcy Court to approve the APA and upon approval, the Bankruptcy Court will enter a sale order (the "**Sale Order**").

E. Sellers and certain of their affiliates will file a chapter 11 Plan, pursuant to which a liquidating trust (the "**Liquidating Trust**") will be created on the effective date of the Plan after confirmation thereof.

F. The primary purpose of the Liquidating Trust is to, among other things, liquidate and distribute assets, which includes assets of the Sellers not sold to Purchaser pursuant to the APA.

Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 88 of 147

G. Following the Closing Date, the Sellers and the Liquidating Trust need access to certain assets, systems, facilities, equipment, and personnel of Purchaser in connection with the Sellers' wind-down of the Businesses, the completion of the Bankruptcy Cases, and the dissolution of the Sellers, and Purchaser agrees to provide such access and services in connection therewith.

## AGREEMENT

**NOW, THEREFORE,** in consideration of the promises and the mutual covenants contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby agree as follows:

## ARTICLE I SERVICES

Section 1.1 <u>Services to be Provided.</u>

(a) <u>Services to be Provided by the Sellers and/or the Liquidating Trust</u>. Subject to the terms and conditions of this Agreement, the Sellers and/or the Liquidating Trust, as the Service Provider, shall provide or cause to be provided to Purchaser, as the Service Recipient, those services in support of the business of Purchaser on such terms (including payment) as may be mutually agreed upon in writing from time to time by the Sellers and/or the Liquidating Trust and Purchaser (the "Seller Services").

(b) <u>Services to be Provided by Purchaser</u>. Subject to the terms and conditions of this Agreement, Purchaser, as the Service Provider, shall provide or cause to be provided to the Sellers and the Liquidating Trust, as the Service Recipient, those services and personnel in support of the business of the Sellers and/or the wind-down and liquidation of the Sellers set forth on <u>Schedule A</u> attached hereto (the "**Purchaser Services**").

#### Section 1.2 Access to and Use of Assets, Systems, Facilities, and Equipment.

Access. Beginning on the Effective Time and continuing through the termination **(a)** or expiration of this Agreement in accordance with its terms, Purchaser hereby grants to the Sellers and the Liquidating Trust, reasonable access to and the right to use the hospital assets, systems, facilities, and equipment to the extent reasonably required in connection with the Sellers' wind-down of the Businesses, the completion of the Bankruptcy Cases, and the dissolution of the Sellers. If any third party to any lease, license or other agreement requires Sellers to sign any documents in order for Sellers to be granted access to, and the right to use, any hospital asset, system, facility or equipment under such lease, license or other agreement, Sellers agree to execute and deliver any such documents; provided, that such documents, leases, licenses or other agreements do not require Sellers to pay any money or take any action, not required by the Sale Order; and provided, further, that if Sellers do not sign such requisite documents, leases, licenses or other agreements, then Purchaser has no obligation to provide Sellers with access and the right to use the associated hospital assets, systems, facilities and equipment provided by such third party; and provided, further, that Purchaser agrees to indemnify, defend and hold harmless each of the Sellers in connection with any obligations under any such documents, leases, licenses or other agreements other than those obligations of Sellers thereunder which are required by the Sale Order.

## Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 89 of 147

Office Space. In order to enable the provision of the Seller Services, Purchaser **(b)** hereby grants to Sellers' and their affiliates' employees and any third-party service providers or subcontractors the right to use and occupy the portion of St. Francis Medical Center identified on Schedule B attached hereto (the "Premises") at no cost for use exclusively as office space pursuant to the terms and conditions hereof, and until the termination of this Agreement pursuant to Article IV herein. Sellers shall keep the Premises neat, clean and sanitary and dispose of all garbage. Sellers may terminate its right to occupy the Premises at any time upon five (5) days prior written notice to Purchaser. Upon termination of Sellers' right to occupy the Premises, Sellers shall vacate the Premises in broom-clean condition. Purchaser shall make the Premises available to Sellers 24 hours a day, 7 days a week. Purchaser may access the Premises for purposes of making repairs or improvement or other reasonable purposes upon (except in the case of an emergency) reasonable prior written notice to Sellers; however, Purchaser shall at all times use its commercially reasonable efforts to prevent interference with or interruption of the Sellers' business activities on the Premises. The Sellers use of the Premises shall additionally be subject to the following terms and conditions:

(i) Purchaser shall have the right, on reasonable advance notice (which shall not be less than three (3) business days) to Sellers, to relocate the space covered by the preceding license to similar space in the same area as the original designated Premises, provided the replacement space is reasonably suitable for the same uses as contemplated herein. In the event of any such relocation, references herein to the Premises shall constitute references to such relocated space.

(ii) Sellers shall not make any material alterations, decorations, additions, or improvements in or to the Premises or the Purchaser's other property without the prior written consent of Purchaser, which may be given or withheld in Purchaser's sole discretion.

(iii) Sellers shall not do, or permit their representatives to do, anything that would: (a) unreasonably obstruct or materially interfere with the lawful operation of the Premises or other portions of the Purchaser's property by Purchaser and other tenants of the Purchaser's property; or (b) use the Premises for any illegal or unlawful purpose, including without limitation any violations of applicable laws or regulations relating to or regulating hazardous substances.

(iv) The license, occupancy and access rights granted to Sellers under this Agreement shall be effective notwithstanding any ground lease, mortgage, deed of trust, or any other hypothecation or security placed at any time upon the Premises or any of Purchaser's property of which the Premises is a part, and notwithstanding any change in ownership (including any sale and/or leaseback involving any property of which the Premises is a part).

**Section 1.3** <u>Subcontracting</u>. The Service Provider may, in its sole discretion and without any written notice to the Service Recipient engage one or more parties (including third parties or affiliates of the Service Provider) to provide the Services, whether in whole or in part or subcomponent. Without limiting the preceding, Purchaser will not be required to use the Sellers existing IT and similar systems, but shall have the right to replace IT and other systems, used to provide any of the Purchaser Services provided that the new or replacement systems provide equivalent functionality. Sellers and Purchaser agree to fully cooperate with each other, at Purchaser's sole cost and expense, in connection with any data or other transition steps required in

## Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 90 of 147

connection moving any of the functions covered by the Purchaser Services to a new vendor or system.

**Section 1.4** <u>Cooperation; Access</u>. Each Party shall, at reasonable times under the circumstances, make available to any other Party properly authorized personnel for the purpose of consultation and decisions in connection with the Services.

# Section 1.5 <u>Certain Qualifications</u>.

(a) <u>Provided AS-IS</u>. By accepting the use to the Premises and other access rights provided herein, Sellers conclusively shall be deemed, subject to the representations, warranties and obligations expressly set forth in this Agreement, to have (i) accepted the Premises and access to other property of Purchaser "AS IS" and without express or implied warranty; and (ii) is fully aware of and has approved the Premises in its condition existing as of such date.

(b) <u>Limited Representation</u>. Except for the representations, warranties and obligations expressly set forth in this Agreement, Purchaser makes no representations and warranties of any kind, implied or expressed, with respect to the Purchaser Services, including, without limitation, no warranties of merchantability or fitness for a particular purpose, which are specifically disclaimed.

(c) <u>Privacy and Confidentiality Protections</u>. All access provided to Sellers, to the Premises or otherwise as provided herein, shall be permitted only in a manner which protects patient privacy and the confidentiality of patient information under applicable Privacy Laws.

(d) <u>Limitation on Liability</u>. In no event shall any Party have any liability under any provision of this Agreement for any punitive, incidental, consequential, special or indirect damages, whether based on statute, contract, tort or otherwise, and whether or not arising from any other Party's sole, joint, or concurrent negligence, strict liability, criminal liability or other fault.

## ARTICLE II LICENSE GRANT

**Section 2.1** License Grant. Beginning on the Effective Time and continuing through the later of the termination or expiration of this Agreement in accordance with its term or the dissolution of the Sellers and their Businesses, the wind-down of the Liquidating Trust and the completion of the Bankruptcy Cases, Purchaser grants to the Sellers and the Liquidating Trust a royalty free, license to use the name "St. Francis Medical Center," including any trademarks, service marks, trademark and service mark registrations and registration applications, trade names, trade name registrations, logos, domain names, trade dress, copyrights, copyright registrations, website content, know-how, trade secrets, related to the use of the preceding name, and the corporate or company names of the Sellers and the name of the Hospital, together with all rights to sue and recover damages for infringement, dilution, misappropriation or other violation or conflict associated with any of the Businesses, the completion of the Bankruptcy Cases, and the dissolution of the Sellers. Sellers and the Liquidating Trust shall not assign, transfer or hypothecate any of the rights subject to, and pursuant to, the preceding license and shall not use any such license rights in any manner which

could adversely affect Purchaser's intellectual property rights with respect to the items which are the subject of such license.

# ARTICLE III PAYMENT

**Section 3.1** Service Fees. The fee, rate or amount to be charged to the Service Recipient for the Services (the "Service Fees") shall be as set forth on <u>Schedule A</u> hereto for the Purchaser Services and as agreed upon in writing from time to time by Sellers and/or the Liquidating Trust and Purchaser for the Purchaser Services. The Service Provider shall issue invoices on a monthly basis to the Service Recipient for the Service Recipient shall pay the undisputed portion of such invoices within thirty (30) days of receipt of such invoices. Any dispute relating to the Services shall be resolved in accordance with the dispute resolution procedures set forth in this Agreement.

## ARTICLE IV TERM AND TERMINATION

Section 4.1 Unless a Service is earlier terminated by the Service Recipient in Term. accordance with Section 4.2(b), the Service Provider shall provide each Service to the Service Recipient for the period beginning at the Effective Time (as defined under the APA) and continuing through and including the applicable dates set forth in: (i) Schedule A (with respect to the Purchaser Services); and (ii) any written agreement between Sellers and/or the Liquidating Trust and Purchaser (with respect to the Seller Services) (the "Applicable Term"); provided, that in the event a particular Service provided by Purchaser hereunder is required by one or more Sellers to allow such Seller(s) to comply with applicable laws, rules or regulations, the Applicable Term shall be extended for so long as such Service is required to allow such Seller(s) to comply with such laws, rules and regulations; provided, however, that notwithstanding any other term herein any Party may, in any case, terminate this Agreement by written notice to any other Party at any time on and after the end of the 18<sup>th</sup> calendar month following the Licensure Date (as defined in the APA). In addition, upon termination or expiration of this Agreement, the Parties may mutually agree in writing to a tail period for any portion of the Services, and the terms of this Agreement shall continue to apply to such ongoing Services until performed in full.

## Section 4.2 <u>Termination</u>.

(a) <u>Termination for Material Breach</u>. This Agreement may be terminated by the Sellers and/or the Liquidating Trust on the one hand, or Purchaser on the other hand, upon written notice to any other Party in the event that any other Party materially breaches any provision of this Agreement and fails to remedy such breach within thirty (30) days of its receipt of such written notice, except to the extent such remedy is prevented or not permitted by the Bankruptcy Court; and provided, further, however, that this Agreement may only be terminated under this <u>Section</u> <u>4.2(a)</u> following approval of such termination by the Bankruptcy Court.

(b) <u>Early Termination</u>. The Service Recipient may terminate any particular Service early by providing not less than thirty (30) days prior written notice to the Service Provider of its intent to terminate early such particular Service. Any such partial termination notice delivered

## Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 92 of 147

shall specify in detail (i) the Service or Services to be terminated and (ii) the effective date(s) of such termination (subject to the foregoing sentence). If any such termination of a Service prior to the expiration of its Applicable Term requires any out-of-pocket fee, cost or expense of a similar nature payable to a third party (a "**Termination Fee**"), the Service Provider shall have no obligation to terminate such Service prior to the expiration of its Applicable Term unless the Service Recipient approves in advance such Termination Fee in writing (a "**Termination Fee Approval**"). If the Termination Fee Approval is granted, the Termination Fee shall be billed to and payable by the Service Recipient.

(c) <u>Continued Access</u>. Notwithstanding the termination of any particular Services hereunder, Service Recipient shall continue to have the right, upon reasonable advance notice and consistent with the other provisions herein, to reasonable access data and Service Provider, personnel relating to such Services for reasonable purposes (including compliance with applicable law).

**Section 4.3** Effect of Termination. The termination of this Agreement pursuant to Section 4.2 shall not relieve any Party from liability for any willful and material breach of this Agreement prior to such termination. Further, the provisions of Article II, Article III, this Section 4.3, Article V and Article VI shall survive any termination of this Agreement pursuant to Section 4.2.

# ARTICLE V COORDINATION AND OTHER COVENANTS

**Section 5.1** <u>Coordination</u>. The Parties hereby appoint as of the Effective Time the following representatives responsible for coordinating and managing the provision and receipt of the Services, which representatives shall have authority to act on the applicable Party's behalf with respect to matters relating to this Agreement:

Sellers' Representative:

Address:

Telephone:

Facsimile:

Email:

Liquidating Trust Representative:
Address:
Telephone:
Facsimile:
Email:
Purchaser's Representative:
Address:
Telephone:
Facsimile:
Email:

In the event that the Sellers, the Liquidating Trust or Purchaser elect, in their sole discretion, to change their respective representative appointed for purposes of this <u>Section 5.1</u>, the Sellers, the Liquidating Trust or Purchaser (as the case may be) shall deliver written notice thereof to the other promptly following the effective date of appointment of the name and contact information of such new representative. The representatives shall work in good faith to address and attempt to resolve any dispute arising out of or relating to this Agreement or the breach, termination or validity thereof, whether sounding in contract or tort and whether arising during or after termination of this Agreement, including any pricing and other Service related matters.

Section 5.2 General Cooperation and Turnover Obligations. The Parties shall cooperate to ensure that any and all payments that constitute Excluded Assets (as defined in the APA), shall be paid to and received by Sellers, with any payments that constitute Assets (as defined in the APA) transferred to Purchaser transferred to Purchaser pursuant to Section 1.7 of the APA or that otherwise arise from services rendered by Purchaser on or after the Licensure Date be paid to and received by Purchaser. In this regard, for a period of two (2) years after the Effective Time ("Turnover Period"), the Parties shall, within ten (10) business days, copy and send to any other Party copies (either in hard copy or via electronic file) of all remittance advices for all deposits to all Lockboxes or other bank accounts for the Accounts Receivables (as defined in the APA), from whatever payor or source of funds, that are received on and after the Effective Time. In the event that Assets are deposited to a bank account of Sellers which is not automatically swept or transferred to Purchaser, then Sellers, within ten (10) business days of notice of the receipt of such Assets, shall turnover and pay Purchaser said funds. In the event that a deposit representing payment of any Excluded Assets is received by Purchaser, then Purchaser, within five (5) business days of notice of the receipt of funds representing any Excluded Assets, shall turnover and pay Sellers such funds. Each Party, moreover, shall have the right, within three (3) months after the expiration of the Turnover Period, to audit by an independent and competent auditor, at the

requesting Party's sole expense, of the bank records and remittance advices of any other Party. Thereafter, upon the findings of the auditor that there has either been an overpayment or an underpayment of funds due, the Party owing funds shall, within five (5) business days, make a payment of such funds to the Party to whom they are owed.

**Section 5.3** <u>HIPAA Compliance</u>. The Parties agrees to take such steps as are necessary to ensure compliance with the Health Insurance Portability and Accountability Act of 1996 ("**HIPAA**"), the California Confidentiality of Medical Information Act, and other applicable federal and state privacy laws and regulations (collectively, the "**Privacy Laws**") with respect to the Hospital and its operations, and the performance by the Parties of the services and obligations provide herein.

## Section 5.4 <u>Confidentiality</u>.

During the term of this Agreement and thereafter, the Parties shall, and shall (a)instruct their respective Representatives (as defined below) to, maintain in confidence and not disclose any other Party's financial, technical, sales, marketing, development, personnel, and other information, records, or data, including, without limitation, customer lists, supplier lists, trade secrets, designs, product formulations, product specifications or any other proprietary or confidential information, however recorded or preserved, whether written or oral (any such information, "Confidential Information"). Each Party hereto shall use the same degree of care, but no less than reasonable care, to protect any other Party's Confidential Information as it uses to protect its own Confidential Information of like nature. Unless otherwise authorized in any other agreement between the Parties, any Party receiving any Confidential Information of any other Party (the "Receiving Party") may use Confidential Information only for the purposes of fulfilling its obligations under this Agreement (the "Permitted Purpose"). Any Receiving Party may disclose such Confidential Information only to its directors/managing members, managers, officers, employees, consultants, financial advisors, prospective or existing lenders, counsel, accountants and other agents of such Receiving Party (and, in addition, Sellers may disclose such Confidential Information to the creditors committee in the Bankruptcy Cases and its consultants, financial advisors, counsel, accountants and other agents, the Liquidating Trust and its trustee, consultants, financial advisors, counsel, accountants and other agents and any advisory board members and their consultants, financial advisors, counsel, accountants and other agents) ("Representatives") who have a need to know such information for the Permitted Purpose and who have been advised of the terms of this Section 5.5 and the Receiving Party shall be liable for any breach of these confidentiality provisions by such persons; provided, however, that any Receiving Party may disclose such Confidential Information to the extent such Confidential Information is required to be disclosed by a governmental agency or court order ("Governmental Order"), in which case the Receiving Party shall promptly notify, to the extent possible, the disclosing Party (the "Disclosing Party"), and take reasonable steps to assist in contesting such Governmental Order or in protecting the Disclosing Party's rights prior to disclosure, and in which case the Receiving Party shall only disclose such Confidential Information that it is advised by its counsel in writing that it is legally bound to disclose under such Governmental Order.

(b) Notwithstanding the foregoing, "Confidential Information" shall not include any information that the Receiving Party can demonstrate: (i) was publicly known at the time of disclosure to it, or has become publicly known through no act of the Receiving Party or its

## Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 95 of 147

Representatives in breach of this <u>Section 5.5</u>; (ii) was rightfully received from a third party without a duty of confidentiality; or (iii) was developed by it independently without any reliance on the Confidential Information.

(c) Upon demand by the Disclosing Party at any time, or upon expiration or termination of this Agreement with respect to any Service, the Receiving Party agrees promptly to return or destroy, at the Receiving Party's option, all Confidential Information. If such Confidential Information is destroyed, an authorized officer of the Receiving Party shall certify to such destruction in writing.

## ARTICLE VI MISCELLANEOUS

# Section 6.1 <u>Relationship of the Parties</u>.

(a) The Parties and their respective representatives shall be deemed independent contractors for all purposes under this Agreement.

(b) This Agreement shall not be deemed or construed to create the relationship of employer or employee, partnership or any type of joint venture relationship among the Parties.

(c) The Parties acknowledge and agree that the Parties are not providing legal, accounting or tax advice under this Agreement. The Parties further acknowledge and agree that no fiduciary or other similar relationship is being created among the Parties relating to the Services or otherwise under this Agreement.

(d) Except as expressly set forth herein, neither the Sellers, the Liquidating Trust, nor any of their respective representatives or affiliates, on the one hand, nor Purchaser or any of its representatives or affiliates, on the other hand, shall have the authority to contract for or assume obligations of any nature in the name of any other Party, as the case may be, without the other's prior written consent.

**Section 6.2** <u>No Set-off</u>. Neither the Sellers, the Liquidating Trust, nor any of their respective representatives or affiliates, on the one hand, nor Purchaser nor any of its representatives or affiliates, on the other hand, shall have any set-off or other similar rights pursuant to this Agreement.

**Section 6.3** <u>Notices</u>. All notices, requests, claims, demands or other communications hereunder shall be deemed to have been duly given and made if in writing and (a) at the time personally delivered if served by personal delivery upon the Party for whom it is intended, (b) at the time received if delivered by registered or certified mail (postage prepaid, return receipt requested) or by a national courier service (delivery of which is confirmed), or (c) upon confirmation if sent by facsimile; in each case to the Party at the address set forth in <u>Section 5.1</u>, or such other address as may be designated in writing hereafter, in the same manner, by such Party.

**Section 6.4** <u>Amendment, Modification and Waiver</u>. Any provision of this Agreement may be amended, modified or waived if, and only if, such amendment, modification or waiver is in writing and signed, in the case of an amendment, by the Parties, or in the case of a waiver, by the Party

against whom the waiver is to be effective. No failure or delay by any Party in exercising any right, power or privilege hereunder shall operate as a waiver thereof nor shall any single or partial exercise thereof preclude any other or further exercise thereof or the exercise of any other right, power or privilege.

**Section 6.5** <u>Assignment</u>. Neither this Agreement nor any of the rights, interests or obligations under it may be directly or indirectly assigned, delegated, hypothecated, sublicensed, subleased or transferred by any of the Parties, in whole or in part, to any other person without the prior written consent of the non-assigning Party, and any attempted or purported assignment in violation of this <u>Section 6.5</u> will be null and void; provided, that Sellers may assign their rights hereunder to the Liquidating Trust. Subject to the preceding sentence, this Agreement will be binding upon, inure to the benefit of, and be enforceable by the Parties and their respective successors and permitted assigns. Notwithstanding the preceding, however, Purchaser may assign its rights or obligations hereunder to its any of its affiliates and may subcontract with third parties to provide any of the Purchaser Services as provided elsewhere in this Agreement, provided that any such assignment or subcontracting shall not relieve Purchaser of or reduce Purchaser's obligations under this Agreement.

**Section 6.6** <u>Entire Agreement</u>. This Agreement constitutes the entire agreement among the Parties with respect to the subject matter hereof and supersedes all prior agreements and understandings, oral or written, with respect to such matters.

**Section 6.7** <u>Third Party Beneficiaries</u>. Nothing expressed or implied in this Agreement is intended to confer any rights, benefits, remedies, obligations or liabilities upon any person other than the Parties and their respective successors and permitted assigns.

**Section 6.8** <u>Expenses</u>. In the event of litigation arising from or relating to this Agreement, the non-prevailing Party as determined by a court of competent jurisdiction in a final non-appealable order shall be liable and pay to the prevailing Party the reasonable attorney's fees and expenses incurred in connection with such litigation.

**Section 6.9** <u>Governing Law</u>. This Agreement shall be construed, performed, and enforced in accordance with, and governed by, the laws of the State of California (without giving effect to the principles of conflicts of laws thereof), except to the extent that the laws of such State are superseded by the Bankruptcy Code or other applicable federal law.

Section 6.10 WAIVER OF JURY TRIAL. TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT, OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (B) EACH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (C) EACH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (D) EACH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 6.10. THIS IS AN AFFIRMATIVE WAIVER OF THE PARTIES' RIGHTS TO A JURY TRIAL UNDER CALIFORNIA LAW, Cal. C. Civ. Pro. Sec 631. BY SIGNING BELOW ON THE SIGNATURE LINES, EACH PARTY IS EXPLICITLY WAIVING JURY TRIAL AND AUTHORIZING ANY AND ALL PARTIES TO FILE THIS WAIVER WITH ANY COURT AS THE WAIVER REQUIRED UNDER Cal. C. Civ. Proc. Sec. 631(f)(2).

**Section 6.11** Jurisdiction; Mediation. The Parties agree that the Bankruptcy Court shall have exclusive personal and subject matter jurisdiction over any action or proceeding with respect to any claim arising out of or related to this Agreement, and any of the documents or transactions contained in or contemplated by this Agreement; *provided, however*, that the Parties agree that the United States District Court for the Central District of California (together with the Bankruptcy Court, the "Chosen Courts") shall have exclusive jurisdiction over such claim if (i) the Bankruptcy Cases are closed and if the Bankruptcy Cases are not reopened to adjudicate such claim after request by the Party bringing such claim or (ii) the Bankruptcy Court determines that it does not have jurisdiction over such claim. Solely in connection with claims arising under this Agreement, or any of the documents or transactions contemplated hereby, the Parties (a) irrevocably submit to the exclusive jurisdiction of the Chosen Courts, (b) waive any objection that the Chosen Courts are an inconvenient forum or do not have jurisdiction over any Party hereto, and (d) agree that service of process upon such Party in any such action or proceeding shall be effective if notice is given in accordance with <u>Section 6.3</u> hereof.

**Section 6.12** <u>Mediation</u>. If a dispute arises out of or relates to this contract, or the breach thereof, and if the dispute cannot be settled through negotiation, the Parties agree first to try in good faith to settle the dispute by mediation administered by the American Arbitration Association under its Commercial Mediation Procedures before resorting to litigation in accordance with the terms of this Agreement. Any such mediation shall take place in Los Angeles, California and be subject to the governing law as set forth in this Agreement.

**Section 6.13** <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, each of which will be deemed to constitute an original, but all of which shall constitute one and the same agreement, and may be delivered by facsimile or other electronic means intended to preserve the original graphic or pictorial appearance of a document.

**Section 6.14** <u>Ownership of Intellectual Property</u>. Subject to the provisions of the APA, each of the Parties shall retain all right, title and interest in and to their respective intellectual property and data and other proprietary and other assets.

**Section 6.15** <u>Service Standards</u>. Service Provider shall use commercially reasonable efforts to provide the Services to be provided by it in accordance with all applicable laws, rules and regulations and in a professional, competent, workman-like and timely manner.

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 98 of 147

**Section 6.16** <u>Severability</u>. The provisions of this Agreement shall be deemed severable and the invalidity or unenforceability of any provision shall not affect the validity or enforceability of the other provisions hereof. If any provision of this Agreement, or the application thereof to any circumstance, is found by a court or other governmental authority of competent jurisdiction to be invalid or unenforceable, the remainder of this Agreement and the application of such provision to other circumstances shall not be affected by such invalidity or unenforceability, nor shall such invalidity or unenforceability affect the validity or enforceability of such provision, or the application thereof, in any other jurisdiction. If any provision of this Agreement is so broad as to be unenforceable, the provision shall be interpreted to be only so broad as would be enforceable.

**Section 6.17** Force Majeure. No Party shall be liable for nonperformance or late performance of any of its obligations under this Agreement to the extent and for such periods of time as such nonperformance, or late performance is due to reasons outside such Party's control (excluding any arising from the SARS-CoV-2 virus or mutations therefrom or in connection with the disease COVID-19), including acts of God, war (declared or undeclared), riots, revolutions, fire, floods, explosions, sabotage, nuclear incidents, lightning, weather, earthquakes, storms, sinkholes, epidemics, failure of utilities, (or similar nonperformance or late performance of suppliers or subcontractors). In the event of any such nonperformance or late performance, the Service Provider shall use commercially reasonable efforts to restore or resume the Services within a reasonable period of time.

**Section 6.18** <u>Fair Meaning</u>. This Agreement shall be construed according to its fair meaning and as if prepared by all Parties.

NAME	VERITY HEALTH SYSTEM OF
By:	CALIFORNIA, INC.
By:	Ву:
Name:	Name:
Title:	
NAME	Title:
INAME	VERITY HOLDINGS, LLC
Ву:	
Name:	Ву:
	Name:
Title:	
NAME	Title:
	ST. FRANCIS MEDICAL CENTER
By:	Byr
	By:

**IN WITNESS WHEREOF,** the Parties have executed this Agreement as of the Signing Date.

# Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 99 of 147

Name:	Name:
Title:	Title:

## Schedule A to TSA

## Services to be Provided by Purchaser to Sellers

I. Phase One: [from closing date to 6-9 months thereafter]

A. Description of Services: Purchaser will assist in activities related to the Debtors' efforts in support of its winding down activities, including accounting for monthly transactions, assisting in materials management and purchasing, facilitating voucher and payment of accounts payable and financial reporting. Purchaser will assist Debtors' in additional winding down activities, including facilitating the Debtors' efforts in claims reconciliation, contracts rejections/assumptions, preparing W2s and 1099s, and preparing tax returns.

B. Accounting - TBD

C. Accounts Payable / Supply Chain - TBD

D. Human Resources / Payroll - TBD

E. Information Technology: Provide application and infrastructure support; maintain licensing; maintain user access and perform required application and operating system upgrades as needed to perform services for all applications. List of applications: - TBD

II. Phase Two: [from end of Phase I to two years thereafter]

A. Description of Services: Purchaser will continue to archive and provide access to select data.

B. Accounting / Finance - TBD

C. Human Resources / Payroll - TBD

D. Information Technology: Provide application and infrastructure support; maintain licensing; maintain user access and perform required application and operating system upgrades as needed to perform services for all applications. List of applications: - TBD

Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 101 of 147

## Schedule B to TSA

**Office Space for Use by Sellers** 

# Exhibit 5.9(c) to APA

# Accepted Conditions (Attorney General)

See attached.

# Conditions to the Sale of St. Francis Medical Center<sup>21</sup> and Approval of the Asset Purchase Agreement by and among Verity Health System of California, Inc., Verity Holdings, LLC, St. Francis Medical Center and Prime Healthcare Services, Inc. ("Purchaser")

These Conditions shall be legally binding on Verity Health System of California, Inc., a California nonprofit public benefit corporation, Verity Holdings, LLC, a California limited liability company, St. Francis Medical Center, a California nonprofit public benefit corporation, and St. Francis Medical Center Foundation, a nonprofit public benefit corporation and Purchaser, a Delaware corporation, any other subsidiary, parent, general partner, limited partner, member, affiliate, successor, successor in interest, assignee, or person or entity serving in a similar capacity of any of the above-listed entities including, but not limited to, any entity succeeding thereto as a result of consolidation, affiliation, merger, or acquisition of all or substantially all of the real property or operating assets of St. Francis Medical Center, or the real property on which St. Francis Medical Center is located, any and all current and future owners, lessees, licensees, or operators of St. Francis Medical Center, and any and all current and future lessees and owners of the real property on which St. Francis Medical Center is located.

The transaction conditionally approved by the Attorney General consists of \_\_\_\_\_\_.

All the entities listed in Condition I, and any other parties referenced in the above agreements shall fulfill the terms of these agreements or documents and shall notify and obtain the Attorney General's approval in writing of any proposed modification or rescission of any of the terms of these agreements or documents. Such notifications shall be provided at least sixty days prior to their effective date in order to allow the Attorney General to consider whether they affect the factors set forth in Corporations Code section 5917 and obtain the Attorney General's approval.

For ten years from the closing date of the Asset Purchase Agreement, Verity Health System of California, Inc., Verity Holdings, LLC, Purchaser, and all future owners, managers, lessees, licensees, or operators of St. Francis Medical Center shall be required to provide written notice to the Attorney General sixty days prior to entering into any agreement or transaction to do any of the following:

Sell, transfer, lease, exchange, option, convey, manage, or otherwise dispose of St. Francis Medical Center;

Transfer control, responsibility, management, or governance of St. Francis Medical Center. The substitution, merger or addition of a new member or members of the governing body of Purchaser that transfers the control of, responsibility for or governance of St. Francis Medical Center, shall be deemed a transfer for purposes of this Condition. The substitution or addition of one or more members of the governing body of Purchaser, or any arrangement, written or oral, that would transfer voting control of the members of the governing body of Purchaser shall also be deemed a transfer for purposes of this Condition.

<sup>&</sup>lt;sup>21</sup> Throughout this document, the term "St. Francis Medical Center" shall mean the general acute care hospital located at 3630 East Imperial Highway, Lynwood, CA 90262, and any other clinics, laboratories, units, services, or beds included on the license issued to St. Francis Medical Center by the California Department of Public Health, effective January 1, 2019, unless otherwise indicated.

For ten years from the closing date of the Asset Purchase Agreement, St. Francis Medical Center shall be operated and maintained as a licensed general acute care hospital (as defined in California Health and Safety Code Section 1250) and shall maintain and provide 24-hour emergency and trauma medical services at no less than current<sup>3</sup> licensure and designation with the same types and/or levels of services, including the following:

46 emergency treatment stations at a minimum;
Designation as a Level II Trauma Center;
Designation as a 5150 Receiving Facility, as defined by the Welfare and Institutions Code, section 5150, for behavioral health patients under involuntary evaluation;
Psychiatric evaluation team;
Designation as an Emergency Department Approved for Pediatrics (EDAP);
Designation as a Paramedic Base Station; and Certification as a Primary Stroke Center.

Purchaser must give one-year advance written notice to the Los Angeles County Emergency Medical Services Agency and the California Department of Public Health if St. Francis Medical Center seeks to reduce trauma or trauma-related care services or stop operating the Level II Trauma Center after ten years from the closing date of the Asset Purchase Agreement.

For at least ten years from the closing date of the Asset Purchase Agreement, St. Francis Medical shall maintain on-call coverage contracts and/or comparable coverage arrangements with physicians that are necessary to retain its qualification as a Level II trauma center. The following on-call coverage contracts and/or comparable coverage arrangements are required to retain St. Francis Medical Center's status as a Level II trauma center:

Neurology; Obstetrics/gynecology; Ophthalmology; Oral or maxillofacial or head and neck; Orthopedic Plastic surgery; Reimplantation/microsurgery capability (this surgical service may be provided through a written transfer agreement); and Urology.

For at least ten years from the closing date of the Asset Purchase Agreement, St. Francis Medical Center shall maintain the following services at current licensure, types, and/or levels of services:

- Cardiac services, including three cardiac catheterization labs and the designation as a STEMI Receiving Center;
- Critical care services, including a minimum of 36 intensive care unit beds or 24 intensive care beds and 12 definitive observation beds;

Neonatal intensive care services, including a minimum of 29 neonatal intensive care beds, and at minimum, maintaining a Level II NICU;

 $<sup>\</sup>frac{32}{2}$  The term "current" or "currently" throughout this document means as of January 1, 2019.

Women's health services, including women's imaging services;Pediatric services, including a designated area with at least five general acute care beds for pediatric services;Orthopedic and rehabilitation services;Wound care services;Behavioral health services, including a minimum of 40 distinct part inpatient acute psychiatric beds; andPerinatal services, including a minimum of 50 perinatal beds.

Purchaser shall not place all or any portion of its above-listed licensed-bed capacity or services in voluntary suspension or surrender its license for any of these beds or services.

For at least ten years from the closing date of the Asset Purchase Agreement, St. Francis Medical Center shall maintain the same types and/or levels of women's healthcare services currently provided at the location below or a location within three miles of St. Francis Medical Center:

Family Life Center at St. Francis Medical Center, located at 3630 E Imperial Highway, Lynwood, California.

For at least five years from the closing date of the Asset Purchase Agreement, Purchaser shall either: (1) operate clinics (listed below) with the same number of physicians and mid-level provider full-time equivalents in the same or similar alignment structures, or (2) sell the clinics (listed below) with the same number of physician and mid-level provider full-time equivalents and require the purchaser(s) to maintain such services for 5 years from the closing date of the Asset Purchase Agreement and to participate in the Medi-Cal and Medicare programs as required in the conditions herein, or (3) ensure that a third party is operating the clinics (listed below) with the same number of physician and mid-level provider full-time equivalents and require the third party to maintain such services for 5 years from the closing date of the Asset Purchase Agreement and to participate in the closing date of the Asset Purchase Agreement and mid-level provider full-time equivalents and require the third party to maintain such services for 5 years from the closing date of the Asset Purchase Agreement and to participate in the losing date of the Asset Purchase Agreement and to participate in the Medi-Cal and Medicare programs as required in the conditions herein. For any of these options, each clinic can be moved to a different location within a three-mile radius of each clinic's current location, and St. Francis Medical Center can utilize an alternative structure in providing such services. The following clinic is subject to this condition:

Orthopedic services at 3628 E. Imperial Highway, Suite 300, Lynwood, California.

For ten years from the closing date of the Asset Purchase Agreement, Purchaser shall:

Be certified to participate in the Medi-Cal program at St. Francis Medical Center;

Maintain and have Medi-Cal Managed Care contracts with the below listed Medi-Cal Managed Care Plans to provide the same types and levels of emergency and non-emergency services at St. Francis Medical Center to Medi-Cal beneficiaries (both Traditional Medi-Cal and Medi-Cal Managed Care) as required in these Conditions, on the same terms and conditions as other similarly situated hospitals offering substantially the same services, without any loss, interruption of service or diminution in quality, or gap in contracted hospital coverage, unless the contract is terminated for cause or not extended or renewed by the Medi-Cal Managed Care Plan:

Local Initiative: L.A. Care Health Plan or its successor; and

Commercial Plan: Health Net Community Solutions, Inc. or its successor.

If Purchaser questions whether it is being reimbursed on the same terms and conditions as other similarly situated hospitals offering substantially the same services, it shall notify the Attorney General's Office with at least 120 days' notice prior to taking any action that would effectuate any loss, interruption of service or diminution in quality, or gap in contracted hospital coverage or prior to giving any required notice of taking such action.

Be certified to participate in the Medicare program by maintaining a Medicare Provider Number to provide the same types and levels of emergency and non-emergency services at St. Francis Medical Center to Medicare beneficiaries (both Traditional Medicare and Medicare Managed Care), on the same terms and conditions as other similarly situated hospitals, as required in these Conditions.

For six fiscal years from the closing date of the Asset Purchase Agreement, Purchaser shall provide an annual amount of Charity Care (as defined below) at St. Francis Medical Center equal to or greater than \$8,000,000 (the Minimum Charity Care Amount). For purposes hereof, the term "charity care" shall mean the amount of charity care costs (not charges) incurred by Purchaser in connection with the operation and provision of services at St. Francis Medical Center. The definition and methodology for calculating "charity care" and the methodology for calculating "costs" shall be the same as that used by Office of Statewide Health Planning Development (OSHPD) for annual hospital reporting purposes.<sup>43</sup>

Purchaser's obligation under this Condition shall be prorated on a daily basis if the closing date of the Asset Purchase Agreement is a date other than the first day of Verity Health System of California, Inc.'s fiscal year.

For the second fiscal year and each subsequent fiscal year, the Minimum Charity Care Amount shall be increased (but not decreased) by an amount equal to the Annual Percent increase, if any, in the 12 Months Percent Change: All Items Consumer Price Index for All Urban Consumers in the Los Angeles-Long Beach-Anaheim Average Base Period: 1982-84=100 (CPI-LA, as published by the U.S. Bureau of Labor Statistics).

If the actual amount of charity care provided at St. Francis Medical Center for any fiscal year is less than the Minimum Charity Care Amount (as adjusted pursuant to the above-referenced Consumer Price Index) required for such fiscal year, Purchaser shall pay an amount equal to the deficiency to one or more tax-exempt entities that provide direct healthcare services to residents in the St. Francis Medical Center's service area (31 ZIP codes), as described on page 54 in the Healthcare Impact Report authored by JD Healthcare dated August 16, 2019. (Exhibit 1.) Such payment(s) shall be made within six months following the end of such fiscal year.

For six fiscal years from the closing date of the Asset Purchase Agreement Purchaser shall provide an annual amount of Community Benefit Services at St. Francis Medical Center equal to or greater

 $<sup>4\</sup>frac{3}{2}$  OSHPD defines charity care by contrasting charity care and bad debt. According to OSHPD, "the determination of what is classified as . . . charity care can be made by establishing whether or not the patient has the ability to pay. The patient's accounts receivable must be written off as bad debt if the patient has the ability but is unwilling to pay off the account."

than \$1,139,301 (the "Minimum Community Benefit Services Amount") exclusive of any funds from grants. For six fiscal years, the following community benefit programs and services shall continue to be offered at its current or equivalent location:

Southern California Crossroads Program; Health Benefit Resource Center; Welcome Baby Program; Healthy Community Initiatives; American Career College access for onsite training; Paramedic Training and Education; and Patient Transportation support.

The planning of, and any subsequent changes to, the community benefit services provided at St. Francis Medical Center shall be decided after consultation with the Local Governing Board of Directors.

Purchaser's obligation under this Condition shall be prorated on a daily basis if the effective date of the Asset Purchase Agreement is a date other than the first day of Verity Health System of California, Inc.'s fiscal year.

For the second fiscal year and each subsequent fiscal year, the Minimum Community Benefit Services Amount shall be increased (but not decreased) by an amount equal to the Annual Percent increase, if any, in the 12 Months Percent Change: All Items Consumer Price Index for All Urban Consumers in the Los Angeles-Long Beach-Anaheim Average Base Period: 198284=100 (CPI-LA, as published by the U.S. Bureau of Labor Statistics).

If the actual amount of community benefit services provided at St. Francis Medical Center for any fiscal year is less than the Minimum Community Benefit Services Amount (as adjusted pursuant to the above-referenced Consumer Price Index) required for such fiscal year, Purchaser shall pay an amount equal to the deficiency to one or more tax-exempt entities that provide community benefit services for residents in St. Francis Medical Center's service area (31 ZIP codes), as defined on as described on page 54 in the Healthcare Impact

Report authored by JD Healthcare dated August 16, 2019. (Exhibit 1.) Such payment(s) shall be made within six months following the end of such fiscal year.

For at least ten years from the closing date of the Asset Purchase Agreement unless otherwise indicated, Purchaser shall maintain its contracts and any amendments and exhibits thereto with the City and/or County of Los Angeles for services, including the following:

- Participation in the Hospital Preparedness Program between the Hospital and Los Angeles County;
- Department of Mental Health Legal Entity Contract between the Hospital and Los Angeles County;

Paramedic Base Hospital Services between the Hospital and Los Angeles County; Radiation Therapy Services between the Hospital and Los Angeles County;

## Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 108 of 147

- Designation Agreement between the County of Los Angeles Department of Mental Health (LAC-DMH) and the Hospital and approved as a 72-hour Evaluation and Intensive Treatment facility;
- Trauma Center Service Agreement between the Hospital and Los Angeles County; and Paramedic Training Institute Students between the Hospital and Los Angeles County.

For at least ten years from the closing date of the Asset Purchase Agreement, Purchaser shall provide to the Los Angeles County Department of Health Services and Los Angeles County of Department of Mental Health information and documents related to staffing assessments, clinical guidelines, services provided, and technology needs for St. Francis Medical Center. The goal is to ensure that Purchaser's decisions or changes in these areas will not be motivated by a desire to move away from serving the Medi-Cal population. Such information and documents will also be provided to the Local Governing Board.

For ten years from the closing date of the Asset Purchase Agreement, Purchaser shall have at St. Francis Medical Center a Local Governing Board of Directors. Purchaser shall consult with the Local Governing Board of Directors prior to making changes to medical services, community benefit programs, making capital expenditures, including making changes to the charity care and collection policies, and making changes to charity care services provided at St. Francis Medical Center. The members of the Local Governing Board of Supervisors, and community representatives from St. Francis Medical Center's medical staff, St. Francis Medical Center's Chief of Staff, one member designated by the Los Angeles County Board of Supervisors, and community representatives from St. Francis Medical Center's primary service area (31 ZIP codes), as described on page 54 in the Healthcare Impact Report authored by JD Healthcare dated August 16, 2019 attached hereto as Exhibit 1, including at least one member from a local healthcare advocacy group. Such consultation shall occur at least sixty days prior to the effective date of such changes or actions unless done so on an emergency basis. The Local Governing Board's approval is required of all reports submitted to the Attorney General regarding compliance with these Conditions.

Purchaser shall commit to reserve or expend for St. Francis Medical Center for capital improvements to the hospital over the five-year period from the closing of the Asset Purchase Agreement, the amount of capital that remains unexpended from the \$180 million commitment required of Blue Mountain Capital Management, LLC as part of the Attorney General Conditions approved on December 3, 2015 but this amount can be no less than \$5.8 million among St. Francis Medical Center, Seton Medical Center and St. Vincent Medical Center. Purchaser understands that such condition has previously been satisfied but will still commit to expend approximately \$35 million on St. Francis Medical Center over the five-year period from the closing of the Asset Purchase Agreement.

Purchaser shall commit the necessary investments required to maintain OSHPD seismic compliance requirements at the Hospital through 2030 under the Alfred E. Alquist Hospital Facilities Seismic Safety Act of 1983, as amended by the California Hospital Facilities Seismic Safety Act, (Health & Safety. Code, § 129675-130070).

Purchaser shall maintain privileges for current medical staff who are in good standing as of the closing date of the Asset Purchase Agreement. Further, the closing of the Asset Purchase Agreement shall not change the independence of the medical staff.

## Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 109 of 147

There shall be no discrimination against lesbian, gay, bisexual, or transgender individuals at St. Francis Medical Center, and no restriction or limitation on providing or making reproductive health services available at St. Francis Medical Center, its medical office buildings, or at any of its facilities. Both of these prohibitions shall be set forth in Purchaser's written policies, adhered to, and strictly enforced.

For eleven fiscal years from the closing date of the Asset Purchase Agreement Purchaser shall submit to the Attorney General, no later than four months after the conclusion of each fiscal year, a report describing in detail compliance with each Condition set forth herein. The Chairman of the Board of Directors of Purchaser shall certify that the report is true, accurate, and complete and provide documentation of the review and approval of the report by the Local Governing Board.

At the request of the Attorney General, all parties listed in Condition I, Verity Health System of California, Inc., Verity Holdings, LLC, Purchaser, and any other parties referenced in the agreements listed in Condition II shall provide such information as is reasonably necessary for the Attorney General to monitor compliance with these Conditions and the terms of the transaction as set forth herein. The Attorney General shall, at the request of a Party and to the extent provided by law, keep confidential any information so produced to the extent that such information is a trade secret or is privileged under state or federal law, or if the private interest in maintaining confidentiality clearly outweighs the public interest in disclosure.

Once the Asset Purchase Agreement is closed, and so long as such actions are consistent with the Asset Purchase Agreement, all parties listed in Condition I, and any other parties referenced in the agreements listed in Condition II are deemed to have explicitly and implicitly consented to the applicability and compliance with each and every Condition and to have waived any right to seek judicial relief with respect to each and every Condition.

The Attorney General reserves the right to enforce each and every Condition set forth herein to the fullest extent provided by law. In addition to any legal remedies the Attorney General may have, the Attorney General shall be entitled to specific performance, injunctive relief, and such other equitable remedies as a court may deem appropriate for breach of any of these Conditions. Pursuant to Government Code section 12598, the Attorney General's office shall also be entitled to recover its attorney fees and costs incurred in remedying each and every violation.

## Analysis of the Hospital's Service Area

## Service Area Definition

Based upon the Hospital's CY 2017 inpatient discharges, the Hospital's service area is comprised of 31 ZIP Codes from which 75% of the Hospital's inpatient discharges came from. Approximately 51% of the Hospital's discharges originated from the top eight ZIP Codes, located in Lynwood, South Gate, Los Angeles, Bell, Compton, Bell Gardens, and Huntington Park. In CY 2017, the Hospital's market share in the service area was approximately 11% based on total area discharges.

		PATIE	NT ORIGIN, CY 2	017		
			Percentage of	Cumulative	Market	Total
ZIP Code	Community	Discharges	Discharges	Percentage	Share	Discharges
90262	Lynwood	2,490	11.1%	11.1%	38.1%	6,538
90280	South Gate	2,187	9.8%	20.9%	29.0%	7,554
90221	Compton	1,400	6.3%	27.2%	24.1%	5,812
90201	Bell	1,359	6.1%	33.3%	16.3%	8,363
90002	Los Angeles	1,066	4.8%	38.0%	18.4%	5,797
90255	Huntington Park	956	4.3%	42.3%	15.5%	6,172
90059	Los Angeles	948	4.2%	46.6%	17.2%	5,527
90001	Los Angeles	922	4.1%	50.7%	15.6%	5,901
90220	Compton	708	3.2%	53.9%	12.7%	5,554
90222	Compton	700	3.1%	57.0%	18.1%	3.868
90003	Los Angeles	625	2.8%	59.8%	7.6%	8,209
90044	Los Angeles	542	2.4%	62.2%	4.5%	11,994
90723	Paramount	525	2.3%	64.6%	11.7%	4,483
90061	Los Angeles	358	1.6%	66.2%	9.5%	3,764
90650	Norwalk	344	1.5%	67.7%	3.3%	10,373
90270	Maywood	282	1.3%	69.0%	12.2%	2,309
90805	Long Beach	267	1.2%	70.2%	2.7%	9,940
90706	Bellflower	263	1.2%	71.3%	3.6%	7,223
90242	Downey	252	1.1%	72.5%	6.2%	4,038
90241	Downey	224	1.0%	73.5%	6.0%	3,726
90660	Pico Rivera	91	0.4%	73.9%	1.4%	6,608
90240	Downey	69	0.3%	74.2%	3.3%	2,073
90670	Santa Fe Springs	46	0.2%	74.4%	2.7%	1,703
90605	Whittier	44	0.2%	74.6%	1.1%	4,082
90606	Whittier	44	0.2%	74.8%	1.4%	3,244
90703	Cerritos	37	0.2%	74.9%	0.9%	4,026
90604	Whittier	32	0.1%	75.1%	0.9%	3,698
90701	Artesia	31	0.1%	75.2%	1.7%	1,813
90638	La Mirada	30	0.1%	75.4%	0.7%	4,274
90603	Whittier	3	0.0%	75.4%	0.1%	2,152
90639	La Mirada	0	0.0%	75.4%	0.0%	10
Sub-Total		16,845	75.4%	75.4%	10.5%	160,828
All Other		5,504	24.6%	100%		
Grand Tota		22,349	100%			

Source: OSHPD Discharge Database, CY 2017

4"="1" "DM\_US 165807051-6.079742.0063" "" US\_Active\114483166\V-4 DM\_US 165807051-13.079742.0063

# **DISCLOSURE SCHEDULES**

<u>To</u>

# ASSET PURCHASE AGREEMENT

**By and Among** 

## ST. FRANCIS MEDICAL CENTER, VERITY HOLDINGS, LLC,

## and VERITY HEALTH SYSTEM OF CALIFORNIA, INC.

(as "Sellers")

<u>and</u>

(as "Purchaser")

Dated [ , 2020]

9

# Schedule 1.4.3 - Owned Real Property

<b>Description</b>	<u>Owner</u>	<u>Address</u>	APN
Hospital Campus	St. Francis Medical	<u>3630 East Imperial</u>	<u>6173-021-008</u>
(including Hospital	Center	<u>Highway, Lynwood,</u>	
Patient Tower, Health		<u>CA 90262</u>	
Services Pavilion,			
Progressive Care			
Unit, Family Life			
Center, Power Plant			
and Parking Structure			
<u>#1)</u>			
Huntington Park	St. Francis Medical	2700 East Slauson	<u>6320-006-069</u>
Medical Office	<u>Center</u>	Avenue, Huntington	
Building		<u>Park, CA 90255</u>	
Maywood Medical	St. Francis Medical	5953 South Atlantic	<u>6313-013-028</u>
Office Building	<u>Center</u>	Boulevard 5,	
		<u>Maywood, CA 90270</u>	
Parking lot (St.	Verity Holdings, LLC	3633 Martin Luther	<u>6173-015-047</u>
Francis Medical		King Jr., Boulevard,	
<u>Center</u> )		Lynwood, CA 90262	
Ministry Services	Verity Holdings, LLC	<u>3663 Martin Luther</u>	<u>6173-019-022</u>
Building (St. Francis		King Jr., Boulevard,	
Medical Center)		Lynwood, CA 90262	
Parking Lot (St.	Verity Holdings, LLC	<u>3621 Martin Luther</u>	<u>6173-015-048</u>
Francis Medical		King Jr., Boulevard,	
<u>Center</u> )		Lynwood, CA 90262	
Five Unit Apartment	Verity Holdings, LLC	3570 Brenton	<u>6191-016-008</u>
(St. Francis Medical		Avenue, Lynwood,	
Center		<u>CA 90057</u>	

## St. Francis Medical Center

# Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 113 of 147

# Schedule 1.7(b) – Licenses

St Francis Medical Center - A	correditation/Certifications	/Licenses
<u>St Plancis Medical Center - A</u>		
License/Permit/Certificate	License/Permit Number	Expiration date <sup>4</sup>
AABB - Certificate of Accreditation for		9/30/2021
Transfusion Activities		
ACR - for Adult Patients - Revolution EVO		7/7/2020
<u>-5</u>		
ACR - for Adult Patients - Revolution EVO		<u>7/7/2020</u>
<u>- 4</u>		
ACR - ADAC Laboratories TRANSCAM		<u>10/12/2021</u>
<u>1994-Accreditation</u>		
ACR - ADAC Laboratories VERTEX 1998		<u>10/12/2021</u>
for Planer - Accreditation		
ACR - MRI Siemens AVANTO 2005		<u>5/19/2021</u>
<u>-Accreditation</u>		
ACR - Siemens SYMBIA T 2011 for		<u>10/12/2021</u>
Planar, SPECT, Nuclear Cardiology		
ACR - Mammography - Lorad Medical	<u>MAP# 07026-05</u>	<u>8/4/2022</u>
Systems Dimensions 2012		
ACR - Mammography - Lorad Medical	<u>MAP# 07026-06</u>	<u>8/4/2022</u>
Systems Dimensions 2012 DBT		
Hospital Pharmacy Permit	<u>HSP 45706</u>	<u>2/1/2021</u>
<u>California Department of Health –</u>	<u>FAC00007446</u>	<u>6/30/2020</u>
Processing Reportable Sources of Radiation		
Sterile Compounding License	<u>LSC 100267</u>	<u>2/1/2021</u>
<u>City of Lynwood - Business License - 4</u>	<u>2301342</u>	<u>12/31/2020</u>
<u>Units</u>	0001756	10/21/2020
City of Lynwood - Business License	<u>2301756</u>	<u>12/31/2020</u>
City of South Gate - Business License	<u>3201</u>	<u>6/30/2020</u>
CLIA Certificate of Waiver	<u>05D0548351</u>	<u>2/27/2021</u>
CLIA Certificate of Waiver	<u>05D0949548</u>	<u>11/1/2021</u>
County of LA Public Health - Public Health	<u>PR0038667</u>	<u>6/30/2020</u>
Permit - Cafeteria	120560	0/4/2022
Department of Health & Human Services	<u>138560</u>	<u>8/4/2022</u>
-Mammography Facility	CDE0000001	12/20/2020
Clinical Laboratory Certificate of Deemed	<u>CDF00000291</u>	<u>12/28/2020</u>
Status	00((0	0/4/2022
Department of Public Health –	<u>20668</u>	<u>8/4/2022</u>
Mammography X-Ray Equipment and		
Facility 2 Dimension	20((0	0/4/2022
Department of Public Health –	<u>20669</u>	<u>8/4/2022</u>

<sup>&</sup>lt;sup>4</sup> Expiration date is subject to being updated.

# Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 114 of 147

Mammography X-Ray Equipment and		
Facility 3 Dimension	0.007054	4/16/2020
Department of Public Health - Medical	<u>SQG37254</u>	<u>4/16/2020</u>
Waste Generator Small Quantity		
Department of Public Health - Medical	<u>No. 289</u>	<u>11/16/2020</u>
Waste Large Quantity Management		
<u>Program</u>		
Hospital License	<u>930000157</u>	<u>12/31/2020</u>
Tissue Bank License - FLC	<u>CTB 00080570</u>	<u>12/12/2020</u>
Tissue Bank License - Hospital OR	<u>CTB 00080568</u>	<u>12/12/2020</u>
Department of Transportation Division	<u>LA-179(H)</u>	
Aeronautics - Heliport(TLOF) Permit		
DEA Controlled Substance Registration	<u>BS7805609</u>	2/28/2023
Institute for Medical Quality – Continuing	<u>N/A</u>	12/31/2021
Medical Education Certificate		
South Coast Air Quality Management	14924	7/15/2020
District - Rule 2202-On Road Motor		
Vehicle Mitigation Options		
Conveyance Permit	107436	2/7/2020
		Renewal in process,
		inspection occurred
		February 17-19, 2020.
		New permits expected
		in April 2020.
Conveyance Permit	107437	2/7/2020
		Renewal in process,
		inspection occurred
		February 17-19, 2020.
		New permits expected
		in April 2020.
Conveyance Permit	107438	2/7/2020
		Renewal in process,
		inspection occurred
		February 17-19, 2020.
		New permits expected
		in April 2020.
Conveyance Permit	113252	6/21/2020
Conveyance Permit	113372	2/8/2020
		Renewal in process,
		inspection occurred
		February 17-19, 2020.
		New permits expected
		in April 2020.
Conveyance Permit	<u>113373</u>	2/8/2020
	<u>115575</u>	Renewal in process,
		inspection occurred
		<u>inspection occurred</u>

# Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 115 of 147

		Eshmany 17 10 2020
		<u>February 17-19, 2020.</u>
		New permits expected
~	110071	<u>in April 2020.</u>
Conveyance Permit	<u>113374</u>	<u>6/21/2020</u>
Conveyance Permit	<u>113375</u>	<u>2/8/2020</u>
		Renewal in process,
		inspection occurred
		<u>February 17-19, 2020.</u>
		New permits expected
		<u>in April 2020.</u>
Conveyance Permit	<u>113401</u>	2/8/2020
		Renewal in process,
		inspection occurred
		February 17-19, 2020.
		New permits expected
		in April 2020.
Conveyance Permit	121533	2/8/2020
	<u>121355</u>	Renewal in process,
		inspection occurred
		February 17-19, 2020.
		· · · · · · · · · · · · · · · · · · ·
		New permits expected
	101505	<u>in April 2020.</u>
Conveyance Permit	<u>121535</u>	<u>2/8/2020</u>
		Renewal in process,
		inspection occurred
		<u>February 17-19, 2020.</u>
		New permits expected
		<u>in April 2020.</u>
Conveyance Permit	<u>121557</u>	<u>2/8/202</u>
		Renewal in process,
		inspection occurred
		February 17-19, 2020.
		New permits expected
		in April 2020.
Conveyance Permit	36304	2/8/2020
		Renewal in process,
		inspection occurred
		February 17-19, 2020.
		New permits expected
		in April 2020.
Conveyance Permit	36305	2/8/2020
	<u>30303</u>	Renewal in process,
		inspection occurred
		<u>February 17-19, 2020.</u>
		New permits expected
		<u>in April 2020.</u>

# Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 116 of 147

Conveyance Permit	<u>90171</u>	<u>2/7/2020</u>
		Renewal in process,
		inspection occurred
		February 17-19, 2020.
		New permits expected
		<u>in April 2020.</u>
Conveyance Permit	<u>99368</u>	2/8/2020
		Renewal in process,
		inspection occurred
		February 17-19, 2020.
		New permits expected
		in April 2020.
Conveyance Permit	99562	2/8/2020
	<u> <del>293</del>02</u>	
		<u>Renewal in process</u> ,
		inspection occurred
		<u>February 17-19, 2020.</u>
		New permits expected
		<u>in April 2020.</u>
Conveyance Permit	<u>99563</u>	2/8/2020
		Renewal in process,
		inspection occurred
		<u>February 17-19, 2020.</u>
		New permits expected
		<u>in April 2020.</u>
Conveyance Permit	99564	2/8/2020
		Renewal in process,
		inspection occurred
		February 17-19, 2020.
		New permits expected
		in April 2020.
Conveyance Permit	99565	2/8/2020
		Renewal in process,
		inspection occurred
		<u>February 17-19, 2020.</u>
		New permits expected
Dadiagative Material Lissues	0499.10	<u>in April 2020.</u>
Radioactive Material License	0488-19	<u>6/28/2024</u>
The College of America Pathologists	<u>2273801</u>	<u>8/22/2021</u>
Laboratory Certification	0.0.00	
The Joint Commission - Hospital	<u>9953</u>	<u>10/27/2021</u>
Accreditation Program		
The Joint Commission - Primary Stroke	<u>9953</u>	<u>6/21/2020</u>
Federal Communications Commission	<u>KUX246</u>	<u>2/9/2024</u>
Private Radio Licenses		
Federal Communications Commission	<u>KWK570</u>	3/21/2024
Private Radio Licenses		

# Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 117 of 147

Federal Communications Commission	<u>WQHA434</u>	<u>6/11/2027</u>
Private Radio Licenses		
Federal Communications Commission	<u>WQIG242</u>	<u>1/30/2028</u>
Private Radio Licenses		
California Department of Tax and Fee	097962355-10000	
Administration Seller's Permit		
Los Angeles County Department of Mental	<u>N/A</u>	<u>N/A</u>
Health Consolidated Medi-Cal		
<b><u>Re-Certification Approval for St. Francis</u></b>		
Medical Center Children's Center		
Baby-Friendly USA Designation as	<u>N/A</u>	12/8/2024
Baby-Friendly Hospital		
American College of Surgeons VRC	<u>N/A</u>	11/20/2021
Committee on Trauma Verification as		
Level II Trauma Center		

Provider Number Listing		
<u>Medicare</u>	<u>Hospital</u>	
<u>05-0104</u>	St Francis Medical Center	
<u>55-5238</u>	St Francis Medical Center - SNF	

Medi-Cal	<u>Hospital</u>
	<u>St Francis Medical Center</u>
<u>HSC30104G</u>	Acute Inpatient Contract
ZZT30104G	Acute Inpatient Non-Contract
ZZT40104G	Outpatient Contract
LTC55238G	SNF Inpatient

<u>OSHPD</u>	Hospital	
<u>106190754</u>	St Francis Medical Center	

# Schedule 1.7(c)(ii) - Leased Real Property

	St. Francis Medical Center as Tenant						
1.	Lease Agreement, dated May 1, 2018 between St. Francis Medical and Hopevale						
	Properties, L.P. c/o Reliable Properties for the premises located at 4390 Tweedy Blvd.,						
	South Gate, CA 90280 (Expires May 31, 2020).						
2.	St. Francis Medical Plaza Office Space Lease, dated January 1, 2006, between St. Francis						
	Medical Center (as Tenant) and St. Francis-Lynwood Medical Plaza L.P. (as Landlord) for						
	the premises located at 3628 E. Imperial Highway, Lynwood, CA 90262, as amended by						
	that certain First Amendment to Lease, dated May 1, 2006, the Second Amendment to						
	Lease, dated February 7, 2007, the Third Amendment to Lease, dated November 29, 2007,						
	the Fourth Amendment to Lease, dated January 1, 2010, between St. Francis Medical						
	Center (as Tenant) and NHP/PMB St. Francis-Lynwood Medical Plaza LLC (as Landlord),						
	the Fifth Amendment to Lease, dated December 31, 2010, the Sixth Amendment to Office						
	Lease and Agreement Regarding Property Tax Protection, dated May 27, 2011, the Seventh						
	Amendment to Lease, dated March 27, 2012, the Eighth Amendment to Lease, dated May						
	1, 2013, the Ninth Amendment to Lease, dated September 9, 2013, the Consent of Prime						
	Landlord to Sublease, dated April 1, 2014 by NHP/PMB St. Francis Lynwood Medical						
	Plaza, LLC, the Tenth Amendment to Lease, dated November 20, 2015, the Consent of						
	Prime Landlord to Sublease, dated January 15, 2016, by NHP/PMB St. Francis Lynwood						
	Medical Plaza, LLC, the Eleventh Amendment to Lease, dated September 27, 2016, and th						
	Consent of Prime Landlord to Sublease, dated December 1, 2016, by NHP/PMB St. Francis						
	Lynwood Medical Plaza, LLC (Expires December 31, 2021).						
<u>3.</u>	Standard Multi-Tenant Office Lease-Gross, dated March 1, 2006, between St. Francis						
	Medical Center (as Lessee) and Sunshine Capital Group, LLC (as Lessor) for the premises						
	located at 3680 E. Imperial Hwy Suites 450, 525 and 550, Lynwood, CA 90262, as						
	amended by that certain First Amendment to Office Lease, dated October 23, 2007, the						
	Second Amendment to Lease, dated December 5, 2012, the Third Amendment to Lease,						
	dated June 25, 2013, and the Fourth Amendment to Lease, dated September 17, 2013						
	(Expired September 30, 2018). <sup>5</sup>						

<sup>&</sup>lt;sup>5</sup> Within the Lease Schedules "\*" indicates the lease is currently month-to-month.

# Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 119 of 147

# Schedule 1.7(c)(iii) - Tenant Leases

	St. Francis Medical Center as Landlord					
<u>1.</u>	Parking Space Lease, dated on or about January 7, 2018 between St. Francis Medical					
	Center (as Landlord) and Davita Medical Management (as Tenant) for the premises located					
	at 2700 E. Slauson Avenue, Huntington Park, CA 90255 as amended by that certain First					
	Amendment, dated January 10, 2019 (Expires October 28, 2019).*					
<u>2.</u>	Medical Office Building Ground Lease, dated January 1, 1993 between St. Francis Medical					
	Center (as Landlord) and Pacific Medical Buildings, L.P. (as Tenant) for the premises					
	located in Lynwood, CA, as amended by that certain First Amendment to Medical Office					
	Building Ground Lease, dated February 1, 1993, the Second Amendment to Medical Office					
	Building Ground Lease, dated May 1, 1993, the Assignment and Assumption of Leases and					
	Subleases dated, June 4, 1993 between Pacific Medical Buildings, L.P. (as Assignor) and					
	St. Francis-Lynwood Medical Plaza, L.P. (as Assignee), and the Grant and Reservation of					
	Easements and Parking Covenants by St. Francis Medical Center, dated January 1, 1993					
	(Expires January 1, 2048).					
<u>3.</u>	Parking Garage Ground Lease, dated January 1, 1993 between St. Francis Medical Center					
	(as Landlord) and Pacific Medical Buildings, L.P. (as Tenant) for the premises located in					
	Lynwood, CA, as amended by that certain Assignment and Assumption of Leases and					
	Subleases, dated June 4, 1993 between Pacific Medical Buildings, L.P. (as Assignor) and					
	St. Francis-Lynwood Medical Plaza, L.P. (as Assignee) (Expires January 1, 2048).					
<u>4.</u>	Sublease, dated December 1, 2016, between St. Francis Medical Center (as Sublessor) and					
	Verity Medical Foundation (as Sublessee) for the premises located at 3628 East Imperial					
	Highway Suite 408, Lynwood, CA 90262 (Expires December 31, 2020).					
<u>5.</u>	Sublease, dated January 15, 2016, between St. Francis Medical Center (as Sublessor) and					
	St. Francis Multispecialty Medical Group, Inc. (as Sublessee) for the premises located at					
	<u>3628 East Imperial Highway Suite 303, Lynwood, CA 90262, as amended by that certain</u>					
	First Amendment to Sublease between St. Francis Medical Center (Sublessor) and St.					
	Francis Multispecialty Medical Group, Inc. (Sublessee), dated December 24, 2019.					
	(Expires December 31, 2020).					

## Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 120 of 147

Verity	Holdings,	LLC as	Landlord

1.	<u>Apartment Lease, dated July 10, 2015, between Verity Holdings, LLC,</u>				
	successor-in-interest to St. Francis Medical Center (as Landlord) and Diane Clark (as				
	Tenant) for the premises located at 3570 Brenton Avenue, Apt. B, Lynwood, CA 90262				

 <u>Apartment Lease, dated July 10, 2015, between Verity Holdings, LLC,</u> <u>successor-in-interest to St. Francis Medical Center (as Landlord) and Gregory Slack (as</u> <u>Tenant) for the premises located at 3570 Brenton Avenue, Apt. A, Lynwood, CA 90262 \*</u>

 <u>Apartment Lease, dated September 1, 2015, between Verity Holdings, LLC,</u> <u>successor-in-interest to St. Francis Medical Center (as Landlord) and Marco Monteon (as</u> Tenant) for the premises located at 3570 Brenton Avenue, Apt. D, Lynwood, CA 90262 \*

Schedule 1.7(d	- Assigned	<b>Contracts and</b>	Assigned Leases
Neme and Arrive	1 10 0 1 1 0 01	C 0 11 01 00 00 00 10 10	

Ref	<b>Debtor</b>	<u>Contract Counterparty</u>	<b>Nature of Contact / Lease</b>	<b><u>Termination</u></b>	Cure
<u>#</u>				<u>Date</u>	<u>Amount</u>
<u>57</u>	<u>St. Francis</u> Medical Center	<u>AMEDA INC</u>	BREAST PUMP AND ACCESSORIES	<u>10/31/2020</u>	<u>\$1,965.25</u>
<u>65</u>	<u>St. Francis</u> Medical Center	AMERICAN RED CROSS	BLOOD PRODUCTS	<u>11/11/2018</u>	<u>\$247,033.00</u>
<u>94</u>	<u>St. Francis</u> Medical Center	<u>ARJO INC</u>	PATIENT LIFTS	<u>7/31/2021</u>	<u>\$20,516.56</u>
<u>95</u>	<u>St. Francis</u> Medical Center	<u>ARROW</u> INTERNATIONAL INC	HEMODYNAMIC MONITORING PRODUCT	<u>6/30/2020</u>	<u>\$24,438.28</u>
<u>100</u>	<u>St. Francis</u> Medical Center	ASAHI INTECC USA INC	DIC PRODUCTS	<u>6/30/2019</u>	<u>\$3,417.77</u>
<u>136</u>	<u>St. Francis</u> Medical Center	BAUSCH AND LOMB	OPHTHALMOLOGY PRODUCTS	<u>6/30/2020</u>	<u>\$3,316.97</u>
<u>142</u>	St. Francis Medical Center	BAYER MEDICAL CARE	CM INJECTORS AND DISPOSABLES	<u>12/31/2020</u>	<u>\$46,138.51</u>
<u>143</u>	<u>St. Francis</u> Medical Center	BAYER MEDICAL CARE	CONTRAST MEDIA MR	<u>12/31/2020</u>	
<u>157</u>	<u>St. Francis</u> Medical Center	BIO RAD LABORATORIES DIAGNOSTICS	BLOOD BANK ANALYZERS REAGENTS	<u>1/31/2019</u>	<u>\$9,387.74</u>
<u>163</u>	<u>St. Francis</u> Medical Center	BIOVENTUS LLC	BONE AND BONE SUBSTITUTE PRODUCTS	<u>6/30/2021</u>	¢15 100 00
<u>164</u>	<u>St. Francis</u> Medical Center	BIOVENTUS LLC	<u>SYNTHETIC</u> <u>IMPLANTABLE</u> PRODUCTS	<u>6/30/2021</u>	<u>\$15,189.80</u>
<u>190</u>	<u>St. Francis</u> Medical Center	BRASSELER USA MEDICAL LLC	ORTHOPEDIC POWER TOOLS	<u>4/30/2019</u>	<u>\$1,224.86</u>
<u>338</u>	<u>St. Francis</u> Medical Center	CAREFUSION	CHLOROHEXADINE GLUCONATE (CHG)	7/31/2020	<u>\$730.96</u>
<u>396</u>	<u>St. Francis</u> Medical Center	COOK MEDICAL INC	SPECIALTY UROLOGY PRODUCTS	<u>12/31/2020</u>	
<u>397</u>	<u>St. Francis</u> Medical Center	COOK MEDICAL INC	GASTROINTESTINAL ENDOSCOPY	<u>1/31/2019</u>	
<u>398</u>	<u>St. Francis</u> Medical Center	COOK MEDICAL INC	PERIPHERAL AND BILIARY STENTS	<u>1/31/2020</u>	
<u>399</u>	<u>St. Francis</u> Medical Center	COOK MEDICAL INC	DIR PRODUCTS	<u>1/31/2020</u>	\$60,694.99
<u>400</u>	<u>St. Francis</u> Medical Center	COOK MEDICAL INC	CARDIAC RHYTHM MANAGEMENT	<u>3/31/2020</u>	<u>,</u>
<u>401</u>	<u>St. Francis</u> Medical Center	COOK MEDICAL INC	VENA CAVA FILTERS	<u>7/31/2020</u>	
<u>402</u>	<u>St. Francis</u> Medical Center	COOK MEDICAL INC	HIGH RISK OBGYN PRODS MED/SURG	<u>12/31/2019</u>	
<u>403</u>	<u>St. Francis</u> Medical Center	COOK MEDICAL INC	HIGH RISK OBGYN PROD SPECIALTY	<u>12/31/2019</u>	

<u>4" = "1" "DM\_US 165807051-6.079742.0063" ""</u> <u>US\_Active\114483166\V-4</u>

# Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 122 of 147

449     St. Francis Medical Center     DAVITA MEDICAL MANGEMENT LIC     LEASE-AS LANDLORD     228/2019     \$0.00       519     St. Francis Medical Center     GRACL MEDICAL INT IMPLANTS AND Medical Center     10/31/2019     \$180.86       525     St. Francis Medical Center     HEARDY DIAGNOSTICS     MANUAL MICROBIOLOGY     12/31/2019     \$7,043.31       539     St. Francis Medical Center     HELMER SCIENTIFIC     GENERAL LABORATORY PRODUCTS     3/31/2021     \$131.75       550     St. Francis Medical Center     IOPEVALE PROPERTIES LP (RELIABLE PROPERTY MANAGEMENT     LEASE-AS TENANT     4/30/2019     \$0.00       551     St. Francis Medical Center     HUFFBURT PROPERTY MANAGEMENT     LEASE-AS TENANT     4/30/2019     \$0.00       552     St. Francis Medical Center     HUFFBURT PROPERTY MANAGEMENT     LEASE-AS TENANT     4/30/2019     \$445.41       553     St. Francis Medical Center     LIC (FKA JAMES H. DEWALD TRUST)     LEASE-AS TENANT     4/30/2019     \$445.41       554     St. Francis Medical Center     LIC (FKA JAMES H. DEWALD TRUST)     LEASE-AS TENANT     4/30/2019     \$2.00.00       557     St. Francis Medical Center     LIC (FKA JAMES H. DEWALD TRUST)     LICA MICROSYSTEMS     HISTOLOGY AND     3/31/2011     \$445.41       560     St. Francis     LICA MICROSYSTEMS     HISTOLOGY AND     3/31/2021						
519     St. Francis Medical Center     GRACE MEDICAL     ENTIMPLANTS AND INSTRUMENTS     10/31/2019     \$180.86       522     St. Francis Medical Center     HARDY DIAGNOSTICS     MANUAL MURCOBIOLOGY     12/31/2012     \$2.043.31       539     St. Francis Medical Center     HELMER SCIENTIFIC     GENERAL LABORATORY PRODUCTS     3/31/2021     \$131.75       540     St. Francis Medical Center     HOPEVALE PROPERTIES IP (RELIABLE PROPERTIY)     LEASE-AS TENANT     4/30/2019     \$0.00       556     St. Francis Medical Center     HOPEVALE PROPERTY PROPERTY PROPERTY     LEASE-AS TENANT     4/30/2019     \$0.00       557     St. Francis Medical Center     HUEFBURT PROPERTY HUEFBURT PROPERTY DEWALD TRUST     LEASE-AS TENANT     4/30/2019     \$0.00       552     St. Francis Medical Center     HUEFBURT PROPERTY HUEFBURT PROPERTY HUEFBURT PROPERTY     LEASE-AS TENANT     4/30/2019     \$0.00       552     St. Francis Medical Center     HUEFBURT PROPERTY HUEFBURT PROPERTY     LEASE-AS TENANT     4/30/2019     \$0.00       553     St. Francis Medical Center     HUEFBURT PROPERTY HUEFBURT PROPERTY     LEASE-AS TENANT     4/30/2019     \$0.00       554     St. Francis Medical Center     HUEFBURT PROPERTY HUEAD TRUSTO     LEASE-AS TENANT     4/30/2019     \$2.075.00       647     St. Francis Medical Center     Maintex Inc     HUEXALE P	<u>449</u>	St. Francis	DAVITA MEDICAL	LEASE-AS LANDLORD	<u>2/28/2019</u>	<u>\$0.00</u>
Medical Center         INSTRUMENTS         Image: Context Markov Markav Markov Markov Markov Markov Markav Markov Markov M		Medical Center	MANAGEMENT LLC			
525         St. Francis Medical Center         HARDY DIAGNOSTICS         MANUAL MICROBIOLOGY         12/31/2019         \$7.043.31           539         St. Francis Medical Center         HELMER SCIENTIFIC         GENERAL LABORATORY PRODUCTS         3/31/2021         \$131.75           550         St. Francis Medical Center         HOPEVALE PROPERTIES IP (RELIABLE PROPERTIES) FKA CHUNG SUK PU TKA WIN PROPERTY MANAGEMENT         LEASE-AS TENANT         4/30/2019         \$0.00           555         St. Francis Medical Center         HUFEPURT PROPERTY ILC (FKA JAMFS H. DEWALD TRUST)         LEASE-AS TENANT         4/30/2019         \$0.00           557         St. Francis Medical Center         HUFEPURT PROPERTY ILC (FKA JAMFS H. DEWALD TRUST)         LEASE-AS TENANT         4/30/2019         \$0.00           557         St. Francis Medical Center         HUCROUNT RUST)         LEASE-AS TENANT         4/30/2012         \$445.41           518         St. Francis Medical Center         HUCROUNT RUST)         LEASE-AS TENANT         4/30/2012         \$245.00           647         St. Francis Medical Center         Mintex Inc         HISTOLOGY AND CYTOLOGY         1/31/2021         \$24.00.0           830         St. Francis Medical Center         MECCOMP         DIALYSIS PRODUCTS         1/31/2021         \$24.00.0           846.30         MERCEDES MEDICAL Medical C	<u>519</u>	St. Francis	GRACE MEDICAL	ENT IMPLANTS AND	<u>10/31/2019</u>	<u>\$180.86</u>
Medical Center         Image: control of the cont		Medical Center		<b>INSTRUMENTS</b>		
Medical Center         Image: control of the cont	525	St. Francis	HARDY DIAGNOSTICS	MANUAL MICROBIOLOGY	12/31/2019	\$7,043.31
Image: market in the second		Medical Center				
Image: market in the second	539	St. Francis	HELMER SCIENTIFIC	GENERAL LABORATORY	3/31/2021	\$131.75
550     St. Francis     HOPEVALE PROPERTIES LP (RELIABLE PROPERTIES) FKA CHUNG SUK PU FKA MIN PROPERTIES) FKA CHUNG SUK PU FKA MIN MANAGEMENT     4/30/2019     \$0.00       556     St. Francis Medical Center     HUFEBURT PROPERTY LLC (FKA JAMES H. DEWALD TRUST)     LEASE-AS TENANT     4/30/2019     \$0.00       557     St. Francis Medical Center     HUFEBURT PROPERTY LLC (FKA JAMES H. DEWALD TRUST)     LEASE-AS TENANT     4/30/2019     \$0.00       625     St. Francis Medical Center     HUFEBURT PROPERTY HUC (CKA JAMES H. DEWALD TRUST)     LEIASE-AS TENANT     4/30/2019     \$0.00       625     St. Francis Medical Center     HUFEBURT PROPERTY NC     HISTOLOGY AND CYTOLOGY     3/31/2021     \$445.41       626     St. Francis Medical Center     Maintex Inc     HOUSEKEEPING SUPPLIES     1/31/2021     \$2,075.00       627     St. Francis Medical Center     MERCEDES MEDICAL     HISTOLOGY AND CYTOLOGY     1/31/2021     \$24,062.00       830     St. Francis Medical Center     MICROTFK MEDICAL     HISTOLOGY AND CYTOLOGY     1/31/2021     \$21,008.98       821     St. Francis Medical Center     MICROTFK MEDICAL     PROUDY SUPPLIES     3/31/2019     \$21,008.98       823     St. Francis Medical Center     MICROTFK MEDICAL	<u> </u>					<u>\$101170</u>
Medical CenterIP.(RELIABLE PROPERTIES) FKA CRUNG SUK PU FKA WIN PROPERTY MANAGEMENTIE.ASE-AS TENANT4'30/2019556S. Francis Medical CenterHUFFBURT PROPERTY ILC (TKA JAMES II, DEWALD TRUST)LEASE-AS TENANT4'30/201950.00557S. Francis Medical CenterHUFFBURT PROPERTY LLC (TKA JAMES II, DEWALD TRUST)LEASE-AS TENANT4'30/201950.00558S. Francis Medical CenterHUFFBURT PROPERTY LLC (TKA JAMES II, DEWALD TRUST)LEASE-AS TENANT4'30/201954.63.00523S. Francis Medical CenterLEICA MICROSYSTEMS ILC (TKA JAMES II, DEWALD TRUST)HISTOLOGY AND CYTOLOGY3/31/202154.63.00531S. Francis Medical CenterMaintex IncPERFUSION PRODUCTSJ/31/2021\$2.000.00647S. Francis Medical CenterMEDCOMPDIALYSIS PRODUCTS10/31/2020\$4.623.006483S. Francis MERCEDES MEDICAL Medical CenterMERCEDES MEDICAL INCHISTOLOGY AND CYTOLOGY1/31/2021\$21.008.98849S. Francis MERCEDES MEDICAL Medical CenterMERCEDES MEDICAL MCORTEK MEDICAL PRODUCTS0/31/2021\$2.200.90849S. Francis MERCEDES MEDICAL Medical CenterMERCEDES SEDICAL PRODUCTS1/31/2021\$2.200.90849S. Francis MERCEDES MEDICAL Medical CenterMERCEDES MEDICAL PRODUCTS1/31/2021\$2.200.90849S. Francis MERCEDES MEDICAL MEDICALPLAY, LLC)PRODUCTS1/31/2021\$2.200.90906S. Francis Medi	550		HOPEVALE PROPERTIES		4/30/2019	\$0.00
ReoperationPROPERTIES) FKA CHUNG SUK PU FKA WIN PROPERTY MANAGEMENTLease-As TENANT4/30/2019556S. Francis Medical CenterHUFPBURT PROPERTY LLC (FKA JAMES H. DEWALD TRUST)LEASE-AS TENANT4/30/2019527S. Francis Medical CenterHUFFBURT PROPERTY LLC (FKA JAMES H. DEWALD TRUST)LEASE-AS TENANT4/30/2019628S. Francis Medical CenterLEICA MICROSYSTEMS INCHISTOLOGY AND CYTOLOGY3/31/2021\$445.41629S. Francis Medical CenterMaintex IncPERFUSION PRODUCTS1/31/2021\$2,075.00620S. Francis Medical CenterMaintex IncPERFUSION PRODUCTS1/31/2021\$2,000620S. Francis Medical CenterMECEDES MEDICAL INCHISTOLOGY AND CYTOLOGY1/31/2021\$2,000620S. Francis Medical CenterMERCEDES SMEDICAL INCHISTOLOGY AND CYTOLOGY1/31/2021\$2,100.898630S. Francis MERCEDES MEDICAL Medical CenterMICROTEK MEDICAL INCHISTOLOGY AND CYTOLOGY1/31/2021\$2,203.00838S. Francis Medical CenterORASURE FECHNOLOGES INC HOROSTICUTURES INC RENDUCTSRAPID DIAGNOSTIC TEST RANCIS LYNWOOD MEDICAL PLAZA, LLC)1/31/2021\$2,203.00901S. Francis Medical CenterPACIFIC MEDICAL FECHNOLOGIES INC RANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD RADUCTS1/31/2021\$2,203.00902S. Francis Medical CenterPACIFIC MEDICAL RANCIS LYNWOOD MEDICAL PLAZA, LLC)DIR PRODUCTS<	<u>550</u>			LLASE-AS TENANT	4/30/2017	<u>\$0.00</u>
Image: CHUNG SUK PU FKA WIN PROPERTY MANAGEMENTImage: CHUNG SUK PU FKA WIN PROPERTY MANAGEMENTImage: CHUNG SUK PU FKA WIN PROPERTY LLC (FKA JAMES H. DEWALD TRUST)Image: CHUNG SUK PU FKA WIN LLC (FKA JAMES H. DEWALD TRUST)Image: CHUNG SUK PU FKA WIN LLC (FKA JAMES H. DEWALD TRUST)Image: CHUNG SUK PU FKA WIN LLC (FKA JAMES H. DEWALD TRUST)Image: CHUNG SUK PU FKA WIN LLC (FKA JAMES H. DEWALD TRUST)Image: CHUNG SUK PU FKA WIN LC (FKA JAMES H. DEWALD TRUST)Image: CHUNG SUK PU FKA WIN LC (FKA JAMES H. DEWALD TRUST)Image: CHUNG SUK PU FKA WIN LC (FKA JAMES H. DEWALD TRUST)Image: CHUNG SUK PU FKA WIN LC (FKA JAMES H. DEWALD TRUST)Image: CHUNG SUK PU FKA WIN LC (FKA JAMES H. DEWALD TRUST)Image: CHUNG SUK PU FKA WIN LC (FKA JAMES H. DEWALD TRUST)Image: CHUNG SUK PU FKA WIN LC (FKA JAMES H. DEWALD TRUST)Image: CHUNG SUK PU FKA WIN LC (FKA JAMES H. DEWALD TRUST)Image: CHUNG SUK PU FKA WIN LC (FKA JAMES H. DEWALD TRUST)Image: CHUNG SUK PU FKA WIN LC (FKA JAMES H. DEWALD TRUST)Image: CHUNG SUK PU FKA WIN LC (FKA JAMES H. DEWALD TRUST)Image: CHUNG SUK PU FKA WIN LC (FKA JAMES H. DEWALD TRUST)Image: CHUNG SUK PU FKA WIN LC (FKA JAMES H. DEWALD TRUST)Image: CHUNG SUK PU FKA WIN LC (FKA JAMES H. DEWALD TRUST)Image: CHUNG SUK PU FKA WIN LC (FKA WIN WINC)Image: CHUNG SUK PU FKA WINC) <th< td=""><td></td><td></td><td></td><td></td><td></td><td></td></th<>						
Image: state in the state in						
Image: constraint of the sector of the sec						
556St. Francis Medical CenterHUFFBURT PROPERTY LLC(FKA JAMES H, DEWALD TRUST)LEASE-AS TENANT4/30/201950.00557St. Francis Medical CenterHUFFBURT PROPERTY LLC(FKA JAMES H, DEWALD TRUST)LEASE-AS TENANT4/30/201950.00523St. Francis Medical CenterHUFFBURT PROPERTY LLC(FKA JAMES H, DEWALD TRUST)LEASE-AS TENANT4/30/20195445.41524St. Francis Medical CenterLEICA MICROSYSTEMS INCHISTOLOGY AND CYTOLOGY3/31/20215445.41631St. Francis Medical CenterMaintex IncPERFUSION PRODUCTS1/31/202152.000.00647St. Francis Medical CenterMEDCOMPDIALYSIS PRODUCTS10/31/202054.623.00643St. Francis Medical CenterMEDCOMPDIALYSIS PRODUCTS10/31/20215910.00643St. Francis Medical CenterMERCEDES MEDICAL INCHISTOLOGY AND CYTOLOGY1/31/20215910.00830St. Francis Medical CenterMICROTEK MEDICAL INCOR EQUIPMENT DRAPES3/31/2019521.008.98871St. Francis Medical CenterNEUROSTRUCTURES INC TECHNOLOGIES INCSPINAL IMPLANTS AND PRODUCTS6/22/201953.307.50906St. Francis Medical CenterPACIFIC MEDICAL TECHNOLOGIES INC HEDICAL PLAZA, LLC)LEASE-AS LANDLORD1/2/31/202157.270.80907St. Francis Medical CenterPACIFIC MEDICAL FRANCIS LYNWOOD MEDICAL PLAZA, LLC)DIR PRODUCTS1/31/202050.00908St. Franci						
Medical CenterLLC (FKA JAMES H. DEWALD TRUST)Image: Constraint of the second						
Image: constraint of the image	<u>556</u>	St. Francis		LEASE-AS TENANT	<u>4/30/2019</u>	
557       St. Francis Medical Center       HUFFBURT PROPERTY LLC (FKA JAMES H. DEWALD TRUST)       LEASE-AS TENANT       4/30/2019         625       St. Francis Medical Center       LEICA MICROSYSTEMS INC       HISTOLOGY AND CYTOLOGY       3/31/2021       \$445.41         626       St. Francis Medical Center       LEICA MICROSYSTEMS INC       HISTOLOGY AND CYTOLOGY       1/31/2021       \$245.41         631       St. Francis Medical Center       Maintex Inc       HOUSEKEEPING SUPPLIES       1/31/2021       \$2.075.00         647       St. Francis Medical Center       Maintex Inc       HOUSEKEEPING SUPPLIES       1/31/2021       \$2.00.00         670       St. Francis Medical Center       MEDCOMP       DIALYSIS PRODUCTS       1/31/2021       \$910.00         830       St. Francis Medical Center       MERCEDES MEDICAL (ECOLAB)       OR EQUIPMENT DRAPES       3/31/2019       \$21.008.98         871       St. Francis Medical Center       NEUROSTRUCTURES INC PRODUCTS       OR EQUIPMENT DRAPES       1/31/2021       \$7.270.80         896       St. Francis Medical Center       DACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)       LEASE-AS LANDLORD       1/231/2047       \$0.00         906       St. Francis Medical Center       PACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)       LEASE-AS LANDLORD		Medical Center	LLC (FKA JAMES H.			
Medical CenterILC (FKA JAMES H. DEWALD TRUST)IISTOLOGY AND CYTOLOGY3/31/2021\$445.41625St. Francis Medical CenterLEICA MICROSYSTEMS NCHISTOLOGY AND CYTOLOGY3/31/2021\$445.41631St. Francis Medical CenterLIVANOVA USA INCPERFUSION PRODUCTS1/31/2021\$2.075.00647St. Francis Medical CenterMaintex IncHOUSEKEEPING SUPPLIES1/31/2020\$2.600.00647St. Francis Medical CenterMEDCOMPDIALYSIS PRODUCTS10/31/2020\$4.623.00630St. Francis Medical CenterMERCEDES MEDICAL INCHISTOLOGY AND CYTOLOGY1/31/2021\$910.00838St. Francis Medical CenterMICROTEK MEDICAL (ECOLAB)OR EQUIPMENT DRAPES3/31/2019\$21,008.98871St. Francis Medical CenterNEUROSTRUCTURES INC TECINOLOGIES INC Medical CenterSPINAL IMPLANTS AND PRODUCTS6/22/2019\$3.307.50896St. Francis Medical CenterORASURE TECINOLOGIES INC RANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD1/2/31/2047907St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD1/2/31/2047919St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)DIR PRODUCTS1/31/2020920St. Francis Medical CenterPENUMBRA INC NEUROVASCULAR INT RADIOLOGY8/31/2020\$7.985.48920St. Francis Med			DEWALD TRUST)			<u>\$0.00</u>
Medical CenterILC (FKA JAMES H. DEWALD TRUST)IISTOLOGY AND CYTOLOGY3/31/2021\$445.41625St. Francis Medical CenterLEICA MICROSYSTEMS NCHISTOLOGY AND CYTOLOGY3/31/2021\$445.41631St. Francis Medical CenterLIVANOVA USA INCPERFUSION PRODUCTS1/31/2021\$2.075.00647St. Francis Medical CenterMaintex IncHOUSEKEEPING SUPPLIES1/31/2020\$2.600.00647St. Francis Medical CenterMEDCOMPDIALYSIS PRODUCTS10/31/2020\$4.623.00630St. Francis Medical CenterMERCEDES MEDICAL INCHISTOLOGY AND CYTOLOGY1/31/2021\$910.00838St. Francis Medical CenterMICROTEK MEDICAL (ECOLAB)OR EQUIPMENT DRAPES3/31/2019\$21,008.98871St. Francis Medical CenterNEUROSTRUCTURES INC TECINOLOGIES INC Medical CenterSPINAL IMPLANTS AND PRODUCTS6/22/2019\$3.307.50896St. Francis Medical CenterORASURE TECINOLOGIES INC RANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD1/2/31/2047907St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD1/2/31/2047919St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)DIR PRODUCTS1/31/2020920St. Francis Medical CenterPENUMBRA INC NEUROVASCULAR INT RADIOLOGY8/31/2020\$7.985.48920St. Francis Med	557	St. Francis	HUFFBURT PROPERTY	LEASE-AS TENANT	4/30/2019	
Image: Constraint of the section of						
625St. Francis Medical CenterLEICA MICROSYSTEMS INCHISTOLOGY AND CYTOLOGY3/31/2021\$445.41631St. Francis Medical CenterLIVANOVA USA INCPERFUSION PRODUCTS1/31/2021\$2.075.00647St. Francis Medical CenterMaintex IncHOUSEKEEPING SUPPLIES1/31/2021\$2.075.00670St. Francis Medical CenterMEDCOMPDIALYSIS PRODUCTS10/31/2020\$4.623.00830St. Francis Medical CenterMERCEDES MEDICAL INCHISTOLOGY AND CYTOLOGY1/31/2021\$910.00838St. Francis Medical CenterMICROTEK MEDICAL (ECOLAB)OR EQUIPMENT DRAPES3/31/2019\$21.008.98871St. Francis Medical CenterMEUROSTRUCTURES INC PRODUCTSSPINAL IMPLANTS AND PRODUCTS6/22/2019\$3.307.50896St. Francis Medical CenterORASURE TECHNOLOGIES INCRAPID DIAGNOSTIC TEST KITS10/31/2021\$7.270.80906St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD12/31/2047907St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)DIR PRODUCTS1/31/2020907St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)DIR PRODUCTS1/31/2047909St. Francis Medical CenterPENUMBRA INCDIR PRODUCTS1/31/2020909St. Francis Medical CenterSENUMBR						
Medical CenterINCCYTOLOGYImage and the second seco	625	St Francis		HISTOLOGY AND	3/31/2021	\$445.41
631         St. Francis Medical Center         LIVANOVA USA INC         PERFUSION PRODUCTS         1/31/2021         \$2,075.00           647         St. Francis Medical Center         Maintex Inc         HOUSEKEEPING SUPPLIES         12/31/2019         \$2,00.00           670         St. Francis Medical Center         MEDCOMP         DIALYSIS PRODUCTS         10/31/2020         \$4,623.00           670         St. Francis Medical Center         MERCEDES MEDICAL INC         HISTOLOGY AND CYTOLOGY         1/31/2021         \$910.00           830         St. Francis Medical Center         MICROTEK MEDICAL (ECOLAB)         OR EQUIPMENT DRAPES         3/31/2019         \$21,008.98           871         St. Francis Medical Center         NEUROSTRUCTURES INC Medical Center         ORASURE TECHNOLOGIES INC         RAPID DIAGNOSTIC TEST NECHNOLOGIES INC         10/31/2021         \$7,270.80           896         St. Francis Medical Center         PACIFIC MEDICAL BUILINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)         LEASE-AS LANDLORD         12/31/2047         \$0.00           907         St. Francis Medical Center         PACIFIC MEDICAL BUILINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)         LEASE-AS LANDLORD         12/31/2047         \$0.00           907         St. Francis Medical Center         PACIFIC MEDICAL BUILINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)         DIR PRODUCTS	025				<u>5/51/2021</u>	<u>\[\[\ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \</u>
Image: market in the image in the image. The image is the image in the image. The image is the image in th	621				1/21/2021	\$2,075,00
647 Medical Center Medical CenterMaintex IncHOUSEKEEPING SUPPLIES PILALYSIS PRODUCTS12/31/2019\$2,500.00670St. Francis Medical CenterMEDCOMPDIALYSIS PRODUCTS10/31/2020\$4,623.00830St. Francis Medical CenterMERCEDES MEDICAL INCHISTOLOGY AND CYTOLOGY1/31/2021\$910.00838St. Francis Medical CenterMICROTEK MEDICAL (ECOLAB)HISTOLOGY AND CYTOLOGY1/31/2021\$21,008.98871St. Francis Medical CenterNEUROSTRUCTURES INC TECHNOLOGIES INCSPINAL IMPLANTS AND PRODUCTS6/22/2019\$3,307.50896St. Francis Medical CenterORASURE TECHNOLOGIES INC BUILDINGS L.P. (ST, FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD12/31/2047\$7,270.80907St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST, FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD12/31/2047907St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST, FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD12/31/2047919St. Francis Medical CenterPENUMBRA INC BUILDINGS L.P. (ST, FRANCIS LYNWOOD MEDICAL PLAZA, LLC)DIR PRODUCTS1/31/2020919St. Francis Medical CenterPENUMBRA INC BUILDINGS L.P. (ST, FRANCIS LYNWOOD MEDICAL PLAZA, LLC)SI/31/2020\$7,985.48920St. Francis Medical CenterPENUMBRA INC ENDOSCOPY DIVDIR PRODUCTS\$4/30/2019921St. Francis Medical CenterSMITH & NEPHEW INC	031		LIVANOVA USA INC	<u>FERFUSION FRODUCTS</u>	<u>1/31/2021</u>	<u>\$2,075.00</u>
Medical CenterMedical CenterMedical CenterMedical CenterMEDCOMPDIALYSIS PRODUCTS10/31/2020\$4.623.00830St. Francis Medical CenterMERCEDES MEDICAL INCHISTOLOGY AND CYTOLOGY1/31/2021\$910.00838St. Francis Medical CenterMERCEDES MEDICAL (ECOLAB)OR EQUIPMENT DRAPES PRODUCTS3/31/2019\$21.008.98871St. Francis Medical CenterNEUROSTRUCTURES INC (ECOLAB)SPINAL IMPLANTS AND PRODUCTS6/22/2019\$3.307.50896St. Francis Medical CenterORASURE TECHNOLOGIES INCRAPID DIAGNOSTIC TEST KITS10/31/2021\$7.270.80906St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD HEDICAL PLAZA, LLC)12/31/2047907St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD LEASE-AS LANDLORD12/31/2047919St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)DIR PRODUCTS1/31/2020919St. Francis Medical CenterPENUMBRA INC ENDUGSCOPY DIVDIR PRODUCTS1/31/2020920St. Francis Medical CenterPENUMBRA INC ENDUSCOPY DIVMEUROVASCULAR INT RADIOLOGY\$/30/2019921St. Francis Medical CenterSMITH & NEPHEW INCSKIN INTEGRITY: WOUND3/31/2020923St. Francis Medical CenterSMITH & NEPHEW INCSKIN INTEGRITY: WOUND	(17				10/01/0010	<b>\$2,500,00</b>
670St. Francis Medical CenterMEDCOMPDIALYSIS PRODUCTS10/31/2020\$4,623.00830St. Francis Medical CenterMERCEDES MEDICAL INCHISTOLOGY AND CYTOLOGY1/31/2021\$910.00838St. Francis Medical CenterMICROTEK MEDICAL (ECOLAB)OR EQUIPMENT DRAPES PRODUCTS3/31/2019\$21,008.98871St. Francis Medical CenterNEUROSTRUCTURES INC TECHNOLOGIES INCSPINAL IMPLANTS AND PRODUCTS6/22/2019\$3,307.50896St. Francis Medical CenterORASURE TECHNOLOGIES INC TECHNOLOGIES INC Hedical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD12/31/2047907St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD1/2/31/2047919St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)DIR PRODUCTS1/31/2020919St. Francis Medical CenterPENUMBRA INC ENDOSCOPY DIVDIR PRODUCTS1/31/2020919St. Francis Medical CenterPENUMBRA INC ENDOSCOPY DIVNEUROVASCULAR INT RADIOLOGY8/30/2019923St. Francis SMITH & NEPHEW INCSKIN INTEGRITY: WOUND3/31/2020	<u>64 /</u>		Maintex Inc	HOUSEKEEPING SUPPLIES	<u>12/31/2019</u>	<u>\$2,500.00</u>
Medical CenterMedical CenterMERCEDES MEDICALHISTOLOGY AND (YTOLOGY1/31/2021\$910.00838St. Francis Medical CenterMICROTEK MEDICAL (ECOLAB)OR EQUIPMENT DRAPES3/31/2019\$21.008.98871St. Francis Medical CenterMEUROSTRUCTURES INC (ECOLAB)SPINAL IMPLANTS AND PRODUCTS6/22/2019\$3.307.50896St. Francis Medical CenterORASURE TECHNOLOGIES INCRAPID DIAGNOSTIC TEST KITS10/31/2021\$7.270.80906St. Francis Medical CenterPACIFIC MEDICAL TECHNOLOGIES INCLEASE-AS LANDLORD12/31/2047\$0.00907St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD12/31/2047919St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD1/31/2020919St. Francis Medical CenterPENUMBRA INC ENONGRADIR PRODUCTS1/31/2020919St. Francis Medical CenterPENUMBRA INC ENDOSCOPY DIVNEUROVASCULAR INT RADIOLOGY\$3/3/2020919St. Francis Medical CenterPENUMBRA INC ENDOSCOPY DIVARTHROSCOPY SUPPLIES RADIOLOGY4/30/2019923St. Francis Medical CenterSMITH & NEPHEW INC ENDOSCOPY DIVSKIN INTEGRITY: WOUND SKIN INTEGRITY: WOUND3/31/2020						
830St. Francis Medical CenterMERCEDES MEDICAL INCHISTOLOGY AND CYTOLOGY1/31/2021\$910.00838St. Francis Medical CenterMICROTEK MEDICAL (ECOLAB)OR EQUIPMENT DRAPES3/31/2019\$21,008.98871St. Francis Medical CenterNEUROSTRUCTURES INC TECHNOLOGIES INCSPINAL IMPLANTS AND PRODUCTS6/22/2019\$3,307.50896St. Francis Medical CenterORASURE TECHNOLOGIES INCRAPID DIAGNOSTIC TEST KITS10/31/2021\$7,270.80906St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOODD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD12/31/2047907St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOODD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD12/31/2047919St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOODD MEDICAL PLAZA, LLC)DIR PRODUCTS1/31/2020919St. Francis Medical CenterPENUMBRA INC ENDOSCOPY DIVDIR PRODUCTS1/31/2020919St. Francis Medical CenterPENUMBRA INC ENDOSCOPY DIVNEUROVASCULAR INT RADIOLOGY8/31/2020920St. Francis Medical CenterSMITH & NEPHEW INC ENDOSCOPY DIVARTHROSCOPY SUPPLIES ARTHROSCOPY SUPPLIES4/30/2019923St. Francis Medical CenterSMITH & NEPHEW INC ENDOSCOPY DIVSXII NITEGRITY: WOUND3/31/2020	<u>670</u>		MEDCOMP	DIALYSIS PRODUCTS	<u>10/31/2020</u>	<u>\$4,623.00</u>
Medical CenterINCCYTOLOGYImage: Medical Center838St. Francis Medical CenterMICROTEK MEDICAL (ECOLAB)OR EQUIPMENT DRAPES PRODUCTS3/31/2019\$21,008.98871St. Francis Medical CenterNEUROSTRUCTURES INC TECHNOLOGIES INCSPINAL IMPLANTS AND PRODUCTS6/22/2019\$3,307.50896St. Francis Medical CenterORASURE TECHNOLOGIES INCRAPID DIAGNOSTIC TEST (KITS10/31/2021\$7,270.80906St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD12/31/2047907St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD12/31/2047919St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)DIR PRODUCTS1/31/2020919St. Francis Medical CenterPENUMBRA INC PENUMBRA INCDIR PRODUCTS\$/31/2020919St. Francis Medical CenterPENUMBRA INC ENDOSCOPY DIVNEUROVASCULAR INT RADIOLOGY\$/31/2020920St. Francis Medical CenterSMITH & NEPHEW INC ENDOSCOPY DIVARTHROSCOPY SUPPLIES RATHROSCOPY SUPPLIES4/30/2019921St. Francis Medical CenterSMITH & NEPHEW INC ENDOSCOPY DIVSKIN INTEGRITY: WOUND3/31/2020		Medical Center				
838St. Francis Medical CenterMICROTEK MEDICAL (ECOLAB)OR EQUIPMENT DRAPES PRODUCTS3/31/2019\$21,008.98871St. Francis Medical CenterNEUROSTRUCTURES INC TECHNOLOGIES INCSPINAL IMPLANTS AND PRODUCTS6/22/2019\$3,307.50896St. Francis Medical CenterORASURE TECHNOLOGIES INCRAPID DIAGNOSTIC TEST KITS10/31/2021\$7,270.80906St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD12/31/2047907St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD12/31/2047907St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)DIR PRODUCTS1/31/2020907St. Francis Medical CenterPENUMBRA INC ENONGER LEASE-AS LANDLORD1/31/2020\$7,985.48920St. Francis Medical CenterPENUMBRA INC ENDOSCOPY DIVDIR PRODUCTS\$/31/2020\$7,985.48921St. Francis Medical CenterSMITH & NEPHEW INC ENDOSCOPY DIVARTHROSCOPY SUPPLIES ARTHROSCOPY SUPPLIES4/30/2019 4/30/2019\$223,752.45923St. Francis Medical CenterSMITH & NEPHEW INC ENDOSCOPY DIVSKIN INTEGRITY: WOUND3/31/2020	<u>830</u>	St. Francis	MERCEDES MEDICAL	HISTOLOGY AND	<u>1/31/2021</u>	<u>\$910.00</u>
Medical Center(ECOLAB)Image: Center (ECOLAB)Image: Center (ECOLAB)871St. Francis Medical CenterNEUROSTRUCTURES INC TECHNOLOGIES INCSPINAL IMPLANTS AND PRODUCTS6/22/2019 (S1,270,80)\$3,307.50896St. Francis Medical CenterORASURE TECHNOLOGIES INCRAPID DIAGNOSTIC TEST KITS10/31/2021 (J2/31/2047)\$7,270.80906St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD PRODUCTS12/31/2047\$0,00907St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD PRODUCTS1/31/2020\$0,00907St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)DIR PRODUCTS1/31/2020919St. Francis Medical CenterPENUMBRA INC PENUMBRA INCDIR PRODUCTS1/31/2020920St. Francis Medical CenterPENUMBRA INC FRANCIS LYNWOOD MEDICAL PLAZA, LIC)NEUROVASCULAR INT RADIOLOGY8/31/2020921St. Francis Medical CenterSMITH & NEPHEW INC ENDOSCOPY DIVARTHROSCOPY SUPPLIES ATHROSCOPY SUPPLIES4/30/2019 AJ31/2020923St. FrancisSMITH & NEPHEW INC ENDOSCOPY DIVSKIN INTEGRITY: WOUND3/31/2020		Medical Center	INC	CYTOLOGY		
Medical Center(ECOLAB)Image: Context and the	<u>838</u>	St. Francis	MICROTEK MEDICAL	OR EQUIPMENT DRAPES	3/31/2019	\$21,008.98
871St. Francis Medical CenterNEUROSTRUCTURES INC PRODUCTSSPINAL IMPLANTS AND PRODUCTS6/22/2019 S1. Francis Medical Center\$3,307.50 S1. Francis PACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)RAPID DIAGNOSTIC TEST KITS10/31/2021 12/31/2047\$7,270.80906St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD FRANCIS LYNWOOD MEDICAL PLAZA, LLC)12/31/2047\$0,00907St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD FRANCIS LYNWOOD MEDICAL PLAZA, LLC)12/31/2047919St. Francis Medical CenterPENUMBRA INC PENUMBRA INCDIR PRODUCTS RADIOLOGY1/31/2020\$7,985.48920St. Francis Medical CenterPENUMBRA INC ENDOSCOPY DIVNEUROVASCULAR INT RADIOLOGY\$3/3/2019\$223,752.45923St. Francis Medical CenterSMITH & NEPHEW INC ENDOSCOPY DIVARTHROSCOPY SUPPLIES SKIN INTEGRITY: WOUND\$3/3/2020		Medical Center	(ECOLAB)			
Medical CenterMedical CenterMedical CenterPRODUCTSImage: Content of the sector of the sect	871			SPINAL IMPLANTS AND	6/22/2019	\$3,307,50
896St. Francis Medical CenterORASURE TECHNOLOGIES INCRAPID DIAGNOSTIC TEST KITS10/31/2021\$7,270.80906St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD12/31/2047\$0,00907St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD12/31/2047\$0,00907St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD12/31/2047919St. Francis Medical CenterPENUMBRA INC PENUMBRA INCDIR PRODUCTS1/31/2020\$7,985.48920St. Francis Medical CenterPENUMBRA INC ENDOSCOPY DIVNEUROVASCULAR INT RADIOLOGY\$/31/2020\$7,985.48921St. Francis Medical CenterSMITH & NEPHEW INC ENDOSCOPY DIVARTHROSCOPY SUPPLIES SKIN INTEGRITY: WOUND3/31/2020\$223,752.45	<u> </u>				0.11010	<u></u>
Medical CenterTECHNOLOGIES INCKITSImage: mail of the system906St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD12/31/2047907St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD12/31/2047907St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD12/31/2047919St. Francis Medical CenterPENUMBRA INC Medical CenterDIR PRODUCTS1/31/2020 RADIOLOGY3/31/2020920St. Francis Medical CenterPENUMBRA INC SMITH & NEPHEW INC ENDOSCOPY DIVNEUROVASCULAR INT RADIOLOGY8/31/2020923St. Francis Medical CenterSMITH & NEPHEW INC ENDOSCOPY DIVARTHROSCOPY SUPPLIES SKIN INTEGRITY: WOUND3/31/2020	896		ORASURE		10/31/2021	\$7 270 80
906St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD12/31/2047907St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD12/31/2047907St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD12/31/2047919St. Francis Medical CenterPENUMBRA INC PENUMBRA INCDIR PRODUCTS NEUROVASCULAR INT RADIOLOGY1/31/2020920St. Francis Medical CenterSMITH & NEPHEW INC ENDOSCOPY DIVARTHROSCOPY SUPPLIES SKIN INTEGRITY: WOUND3/31/2020923St. Francis Medical CenterSMITH & NEPHEW INC ENDOSCOPY DIVSKIN INTEGRITY: WOUND3/31/2020	070				10/31/2021	<u>\$7,270.00</u>
Medical CenterBUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)Image: Constraint of the sector	000				12/21/2047	
Image: St. Francis Medical CenterFRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD12/31/2047907St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD12/31/2047919St. Francis Medical CenterPENUMBRA INC PENUMBRA INCDIR PRODUCTS1/31/2020 RADIOLOGY\$7,985.48920St. Francis Medical CenterPENUMBRA INC RADIOLOGYNEUROVASCULAR INT RADIOLOGY\$31/2020\$7,985.48921St. Francis Medical CenterSMITH & NEPHEW INC ENDOSCOPY DIVARTHROSCOPY SUPPLIES SKIN INTEGRITY: WOUND\$3/31/2020\$223,752.45	900			LEASE-AS LANDLOKD	12/31/204/	
Image: section of the section of th		Medical Center				
907St. Francis Medical CenterPACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)LEASE-AS LANDLORD12/31/2047919St. Francis Medical CenterPENUMBRA INCDIR PRODUCTS1/31/2020920St. Francis Medical CenterPENUMBRA INCNEUROVASCULAR INT RADIOLOGY8/31/2020921St. Francis Medical CenterPENUMBRA INCNEUROVASCULAR INT RADIOLOGY8/31/2020922St. Francis Medical CenterSMITH & NEPHEW INC ENDOSCOPY DIVARTHROSCOPY SUPPLIES SKIN INTEGRITY: WOUND4/30/2019923St. Francis Medical CenterSMITH & NEPHEW INC ENDOSCOPY DIVSKIN INTEGRITY: WOUND SX1720203/31/2020						<b>\$0.00</b>
Medical CenterBUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)Image: Constant of the sector o						<u>\$0.00</u>
FRANCIS LYNWOOD MEDICAL PLAZA, LLC)FRANCIS LYNWOOD MEDICAL PLAZA, LLC)Image: Constraint of the state of th	<u>907</u>			LEASE-AS LANDLORD	<u>12/31/2047</u>	
Image: Medical CenterMEDICAL PLAZA, LIC)Image: Medical CenterMEDICAL PLAZA, LIC)Image: Medical CenterMedical Center <t< td=""><td></td><td>Medical Center</td><td>BUILDINGS L.P. (ST.</td><td></td><td></td><td></td></t<>		Medical Center	BUILDINGS L.P. (ST.			
919St. Francis Medical CenterPENUMBRA INC Medical CenterDIR PRODUCTS1/31/2020 ST,985.48920St. Francis Medical CenterPENUMBRA INC Medical CenterNEUROVASCULAR INT RADIOLOGY8/31/2020992St. Francis Medical CenterSMITH & NEPHEW INC ENDOSCOPY DIVARTHROSCOPY SUPPLIES Francis4/30/2019 SKIN INTEGRITY: WOUND993St. Francis SMITH & NEPHEW INCSKIN INTEGRITY: WOUND3/31/2020			FRANCIS LYNWOOD			
Medical CenterMedical CenterST,985.48920St. Francis Medical CenterPENUMBRA INC Medical CenterNEUROVASCULAR INT RADIOLOGY8/31/2020992St. Francis Medical CenterSMITH & NEPHEW INC ENDOSCOPY DIVARTHROSCOPY SUPPLIES ARTHROSCOPY SUPPLIES4/30/2019 SUPPLIES993St. Francis SMITH & NEPHEW INCSKIN INTEGRITY: WOUND SKIN INTEGRITY: WOUND3/31/2020			MEDICAL PLAZA, LLC)			
Medical CenterMedical CenterST,985.48920St. Francis Medical CenterPENUMBRA INC Medical CenterNEUROVASCULAR INT RADIOLOGY8/31/2020992St. Francis Medical CenterSMITH & NEPHEW INC ENDOSCOPY DIVARTHROSCOPY SUPPLIES ARTHROSCOPY SUPPLIES4/30/2019 St. Francis993St. Francis SMITH & NEPHEW INCSKIN INTEGRITY: WOUND SKIN INTEGRITY: WOUND3/31/2020	<u>919</u>	St. Francis		DIR PRODUCTS	1/31/2020	
920St. Francis Medical CenterPENUMBRA INC PENUMBRA INCNEUROVASCULAR INT RADIOLOGY8/31/2020992St. Francis Medical CenterSMITH & NEPHEW INC ENDOSCOPY DIVARTHROSCOPY SUPPLIES ARTHROSCOPY SUPPLIES4/30/2019993St. FrancisSMITH & NEPHEW INC ENDOSCOPY DIVSKIN INTEGRITY: WOUND3/31/2020						\$7,985,48
Medical CenterRADIOLOGY992St. FrancisSMITH & NEPHEW INCARTHROSCOPY SUPPLIES4/30/2019993St. FrancisSMITH & NEPHEW INCSKIN INTEGRITY: WOUND3/31/2020	920		PENUMBRA INC	NEUROVASCUI AR INT	8/31/2020	
992St. Francis Medical CenterSMITH & NEPHEW INC ENDOSCOPY DIVARTHROSCOPY SUPPLIES ARTHROSCOPY SUPPLIES4/30/2019 \$223,752.45993St. FrancisSMITH & NEPHEW INC SMITH & NEPHEW INCSKIN INTEGRITY: WOUND3/31/2020	<u>720</u>				010112020	
Medical Center         ENDOSCOPY DIV         Strancis         SMITH & NEPHEW INC         SKIN INTEGRITY: WOUND         3/31/2020	002		SMITH & NEDLEW INC		4/20/2010	
993         St. Francis         SMITH & NEPHEW INC         SKIN INTEGRITY: WOUND         3/31/2020	<u>992</u>			ARTIKOSCOFT SUFFLIES	<u>+/30/2019</u>	
<u>993</u> <u>St. Francis</u> <u>SMITH &amp; NEPHEW INC</u> <u>SKIN INTEGRITY: WOUND</u> <u>3/31/2020</u>	000				2/21/2022	\$223,752.45
Medical Center WOUND MANAGEMENT CARE	<u>993</u>				<u>3/31/2020</u>	
		Medical Center	WOUND MANAGEMENT	CARE		

<u>4" = "1" "DM\_US 165807051-6.079742.0063" ""</u> <u>US\_Active\114483166\V-4</u>

# Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 123 of 147

		5.0.1			
		DIV			
<u>994</u>	St. Francis	SMITH & NEPHEW INC	SKIN INTEGRITY: SKIN	<u>3/31/2020</u>	
	Medical Center	WOUND MANAGEMENT	CLENSER		
		DIV			
<u>995</u>	St. Francis	SMITH & NEPHEW INC	SURGICAL WOUND	<u>5/31/2020</u>	
	Medical Center	WOUND MANAGEMENT	DEBRIDEMENT		
		DIV			
<u>996</u>	St. Francis	SMITH & NEPHEW	ORTHO TOTAL JOINT	<u>9/30/2019</u>	
	Medical Center	<u>ORTHOPEDIC</u>	<u>IMPLANTS</u>		
<u>997</u>	St. Francis	SMITHS MEDICAL ASD	ARTERIAL BLOOD GAS	<u>11/30/2020</u>	
	Medical Center	INC	<u>KITS</u>		
<u>998</u>	St. Francis	SMITHS MEDICAL ASD	SAFETY HUBER NEEDLES	2/28/2021	
	Medical Center	INC			
<u>999</u>	St. Francis	SMITHS MEDICAL ASD	INFUSION DEVICES	1/31/2020	
	Medical Center	INC			
<u>1000</u>	St. Francis	SMITHS MEDICAL ASD	PULSE OXIMETRY AND	4/30/2021	
	Medical Center	INC	CAPNOGRAPHY		
1001	St. Francis	SMITHS MEDICAL ASD	SAFETY HYPODERMIC	9/30/2019	\$6,818.23
	Medical Center	INC	PRODUCTS		
1002	St. Francis	SMITHS MEDICAL ASD	SAFETY IV CATHETERS	9/30/2019	
	Medical Center	INC			
1003	St. Francis	SMITHS MEDICAL ASD	TRACHEOSTOMY TUBES	9/30/2021	
	Medical Center	INC			
1004	St. Francis	SMITHS MEDICAL ASD	TEMPERATURE	12/31/2018	
	Medical Center	INC	MONITORING PRODUCTS		
1005	St. Francis	SMITHS MEDICAL ASD	<b>REGIONAL ANESTHESIA</b>	1/31/2019	
	Medical Center	INC	TRAYS		
1018	St. Francis	ST JUDE MEDICAL	DIC PRODUCTS	6/30/2019	
	Medical Center				ф <i>сс</i> а до
1019	St. Francis	ST JUDE MEDICAL	CARDIAC RHYTHEM	11/14/2020	<u>\$567.79</u>
	Medical Center		MANAGEMENT		
1021	St. Francis	ST. FRANCIS LYNWOOD	LEASE-AS TENANT	12/31/2021	\$97,282.52
	Medical Center	MEDICAL PLAZA, LLC			
1022	St. Francis	St. Francis Multispecialty	SUBLEASE AGREEMENT	1/31/2018	\$0.00
	Medical Center	Clinic			
1059	St. Francis	STRYKER ORTHOPEDICS	ORTHOPEDIC TOTAL	5/31/2020	
	Medical Center		JOINT PRODUCTS		\$209.744.59
1060	St. Francis	STRYKER ORTHOPEDICS	GENERAL ORTHO	10/31/2019	<u>\$398,744.58</u>
	Medical Center		TRAUMA PRODUCTS		
1129	St. Francis	Verity Medical Foundation	SUBLEASE AGREEMENT	12/31/2020	\$7,700.00
	Medical Center				

# Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 124 of 147

<u>Ref #</u>	<u>Debtor</u>	<u>Contract</u> <u>Counterparty</u>	<u>Nature of</u> <u>Contact / Lease</u>	<u>Termination Date</u>	<u>Seton</u> <u>Medical</u> <u>Center &amp;</u> <u>Coastside</u>	<u>St.</u> <u>Francis</u> <u>Medical</u> <u>Center</u>	<u>Verity</u> <u>Health</u> <u>System</u>
<u>93</u>	<u>V Hold.</u>	<u>Campus Physical</u> <u>Therapy</u>	Lease-as Landlord	7/31/2027	<u>\$0.00</u>	<u>\$0.00</u>	<u>\$0.00</u>
<u>138</u>	<u>V Hold.</u>	Dave, Rajan, MD	Lease-as Tenant	<u>9/30/2017</u>	<u>\$0.00</u>	<u>\$0.00</u>	<u>\$0.00</u>
<u>201</u>	<u>V Hold.</u>	Hsiao, Michael	Lease-other	<u>5/10/2017</u>	<u>\$0.00</u>	<u>\$0.00</u>	<u>\$0.00</u>
<u>244</u>	<u>V Hold.</u>	Lee, Shu May MD, Inc.	Lease-as Landlord	<u>2/28/2022</u>	<u>\$0.00</u>	<u>\$0.00</u>	<u>\$0.00</u>
<u>257</u>	<u>V Hold.</u>	Longar, Susan <u>MD</u>	Lease-as Landlord	<u>3/31/2022</u>	<u>\$0.00</u>	<u>\$0.00</u>	<u>\$0.00</u>
<u>266</u>	<u>VHS</u>	<u>Mariposa 2040</u> <u>LLC</u>	Lease-as Tenant	<u>6/30/2047</u>	<u>\$0.00</u>	<u>\$0.00</u>	<u>\$0.00</u>
<u>289</u>	<u>V Hold.</u>	Medicus Integrated Health Services, Inc.	Lease-as Landlord	<u>2/28/2022</u>	<u>\$0.00</u>	<u>\$0.00</u>	<u>\$0.00</u>
<u>323</u>	<u>V Hold.</u>	<u>Mulligan,</u> <u>Timothy, MD</u>	Lease-as Landlord	<u>4/30/2022</u>	<u>\$0.00</u>	<u>\$0.00</u>	<u>\$0.00</u>
<u>329</u>	<u>V Hold.</u>	Nantworks, LLC	Lease-office space	<u>3/31/2019</u>	<u>\$0.00</u>	<u>\$0.00</u>	<u>\$34,722.48</u>
<u>362</u>	<u>V Hold.</u>	Order of Malta Los Angeles, Free Clinic	Lease-as Landlord	<u>6/30/2022</u>	<u>\$0.00</u>	<u>\$0.00</u>	<u>\$0.00</u>
<u>419</u>	<u>V Hold.</u>	<u>Rajan Davie,</u> <u>M.D.</u>	Lease-as Landlord	<u>9/30/2017</u>	<u>\$0.00</u>	<u>\$0.00</u>	<u>\$0.00</u>
<u>420</u>	<u>V Hold.</u>	<u>Ranavat, Amritlal</u> <u>MD</u>	Lease-as Landlord	<u>12/31/2017</u>	<u>\$0.00</u>	<u>\$0.00</u>	<u>\$0.00</u>
<u>421</u>	<u>V Hold.</u>	Raymond Dugan Velasco, MD, Inc.	Lease-as Landlord	<u>4/30/2022</u>	<u>\$0.00</u>	<u>\$0.00</u>	<u>\$0.00</u>
<u>472</u>	<u>V Hold.</u>	<u>Stefan, Michael</u> <u>R., M. D.</u>	Lease-as Landlord	4/14/2022	<u>\$0.00</u>	<u>\$0.00</u>	<u>\$0.00</u>
<u>527</u>	<u>V Hold.</u>	Uy, Santos MD	Lease-as Landlord	<u>11/30/2021</u>	<u>\$0.00</u>	<u>\$0.00</u>	<u>\$0.00</u>

<u>4" = "1" "DM\_US 165807051-6.079742.0063" ""</u> <u>US\_Active\114483166\V-4</u>

DM\_US 165807051-13.079742.0063

# Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 125 of 147

<u>538</u>	<u>V Hold.</u>	<u>Verity Medical</u> <u>Foundation</u>	Lease-as Tenant	7/31/2022	<u>\$0.00</u>	<u>\$0.00</u>	<u>\$0.00</u>
<u>561</u>	<u>V Hold.</u>	Yamamoto, Kenneth MD	Lease-as Landlord	<u>5/31/2019</u>	<u>\$0.00</u>	<u>\$0.00</u>	<u>\$0.00</u>
Added	<u>V Hold.</u>	<u>House Ear</u> <u>Institute</u>	Tenant Lease	<u>2/28/2021</u>	<u>\$0.00</u>	<u>\$0.00</u>	<u>\$0.00</u>
<u>Added</u>	<u>V Hold.</u>	<u>Rolando</u> <u>Mercader, M.D.</u>	<u>Tenant Lease</u>	7/31/2020	<u>\$0.00</u>	<u>\$0.00</u>	<u>\$0.00</u>
Added	<u>V Hold.</u>	Diane Clark	Tenant Lease	<u>Monthly</u>	<u>\$0.00</u>	<u>\$0.00</u>	<u>\$0.00</u>
Added	<u>V Hold.</u>	Gregory Slack	Tenant Lease	<u>Monthly</u>	<u>\$0.00</u>	<u>\$0.00</u>	<u>\$0.00</u>
Added	<u>V Hold.</u>	Marco Monteon	Tenant Lease	<u>Monthly</u>	<u>\$0.00</u>	<u>\$0.00</u>	<u>\$0.00</u>

**Schedule 1.7(I) - Permitted Exceptions** 

None.

<u>4" = "1" "DM\_US 165807051-6.079742.0063" ""</u> <u>US\_Active\114483166\V-4</u>

# Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 127 of 147

Schedule 1.8(bb) - Other Excluded Assets

None.

<u>4" = "1" "DM\_US 165807051-6.079742.0063" ""</u> <u>US\_Active\114483166\V-4</u> <u>10</u>

# Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 128 of 147

Schedule 1.9(i) - Other Obligations

None.

<u>4" = "1" "DM\_US 165807051-6.079742.0063" ""</u> <u>US\_Active\114483166\V-4</u> <u>11</u>

# Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 129 of 147

Schedule 2.4 – Seller's Knowledge of Breach or Default

None.

<u>4" = "1" "DM\_US 165807051-6.079742.0063" ""</u> <u>US\_Active\114483166\V-4</u> <u>12</u>

# Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 130 of 147

# **Schedule 2.5 - Brokers and Finders (Sellers)**

Cain Brothers, a division of KeyBanc Capital Markets.

<u>4" = "1" "DM\_US 165807051-6.079742.0063" ""</u> <u>US\_Active\114483166\V-4</u> 13

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 131 of 147

# Schedule 2.7 - Compliance (Sellers)

See Confidential Side Disclosure Letter, dated of even date herewith.

<u>4" = "1" "DM\_US\_165807051-6.079742.0063" ""</u> <u>US\_Active\114483166\V-4</u> <u>14</u>

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 132 of 147

# Schedule 2.8 - Required Consents (Sellers)

None, other than those consents required under §365 of the Bankruptcy Code.

<u>4" = "1" "DM\_US\_165807051-6.079742.0063" ""</u> <u>US\_Active\114483166\V-4</u> 15

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 133 of 147

#### Schedule 2.9(b) - Environmental Matters (Sellers)

<u>Underground storage tank non-compliance notice issued by Los Angeles Department of Water and</u> <u>Power (LADWP) in September 2016 to Saint Francis Medical Center. Seller removed tank per</u> <u>closure permit issued to Saint Francis Medical Center by LADWP. There was no soil</u> <u>contamination or clean up necessary.</u>

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 134 of 147

# Schedule 2.11 - Missing Material Licenses (Sellers)

<u>SFMC timely submitted the laboratory license for renewal and has received confirmation that</u> <u>CDPH Laboratory Field Services has received the renewal, is processing it, and the hospital may</u> <u>continue testing without penalty.</u>

<u>4" = "1" "DM\_US 165807051-6.079742.0063" ""</u> <u>US\_Active\114483166\V-4</u> 17

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 135 of 147

# Schedule 2.13 - Non-Bankruptcy Court Legal Proceedings (Sellers)

See attached folder "Schedule 2.13 - Non-Bankruptcy Court Legal Proceedings"

<u>4" = "1" "DM\_US\_165807051-6.079742.0063" ""</u> <u>US\_Active\114483166\V-4</u>

# Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 136 of 147

Schedule 2.19 - List of Excluded Individuals (Sellers)

None.

<u>4" = "1" "DM\_US 165807051-6.079742.0063" ""</u> <u>US\_Active\114483166\V-4</u> <u>19</u>

# Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 137 of 147

Schedule 3.4 - No Violation (Purchaser)

None.

<u>4" = "1" "DM\_US 165807051-6.079742.0063" ""</u> <u>US\_Active\114483166\V-4</u> <u>20</u>

# Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 138 of 147

Schedule 3.7 - Legal Proceedings (Purchaser)

None.

<u>4" = "1" "DM\_US 165807051-6.079742.0063" ""</u> <u>US\_Active\114483166\V-4</u> <u>21</u>

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 139 of 147

# Schedule 5.3 - Hospital Employees

See attached folder "Schedule 5.3 Hospital Employees"

<u>4" = "1" "DM\_US 165807051-6.079742.0063" ""</u> <u>US\_Active\114483166\V-4</u>

#### Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 140 of 147

#### Schedule 11.3(b) - Tax Allocation Schedule

The Parties agree that within sixty (60) days subsequent to the Closing Date (the "Allocation Date"), the Purchase Price shall be allocated to the Assets using the residual method and based upon the agreed fair market value of the various Classes of the Assets set forth below, as reasonably determined by the Purchaser and Seller, consistent with Section 1060 of the Internal Revenue Code. In particular, the Parties shall mutually agree upon the tax allocation associated with the Assets of Verity Holdings.

Class of Assets
Class I Assets (cash and general deposit accounts)
Class II Assets (Publicly traded stock; U.S. Government Securities)
Class III Assets (Accounts Receivable)
Class IV Assets (Inventory)
Class V Assets (Property, Plant and Equipment)
<u>Class v Assets (Hoperty, Hant and Equipment)</u>
Class VI Assets (Intangibles other than Goodwill)
<u>Class VI Assets (Intalgioles other than Goodwing</u>
Class VII Assets (Goodwill)

Em	Employment Practices Liability Litigation					
Plaintiff/Grievant	Venue	Case Number(s)				
Ahinasi, Sonnia	San Bernardino County Superior Court	<u>CIVDS1822881</u>				
Aliazis, Tricia	<u>EEOC Charge</u>	<u>37A-2019-03433-C;</u>				
	DFEH Charge	<u>201903-05587626</u>				
Aragon, Jasmine	Los Angeles County Superior Court	<u>BC717447</u>				
Brown, Star		Complaint Not Yet Filed				
Capizzi, Renee		Mediation				
Carcamo, Rosa	Los Angeles County Superior Court	<u>TC029195</u>				
Carter, Margena		Right To Sue Letter				
Dolmo, Norma	EEOC Charge DFEH Charge	480-2019-01625				
Engelman, Ivonne *settlement in process	Los Angeles County Superior Court	<u>BC704524</u>				
Harris, Leonard	Los Angeles County Superior Court	<u>19STVC38595</u>				
<u>Martin, Joan</u>	EEOC Charge	<u>37A-2018-02446</u>				
<u>Nguyen, Diane</u>	Los Angeles County Superior Court	<u>BC713041</u>				
<u>Nguyen, Natalie</u>	Los Angeles County Superior Court: EEOC Charge	<u>Small Claims</u> <u>19CMSC01398;</u> 480-2019-00050				
<u>Raj, Aaron</u>		Notice Of Intent				
Rapp, Dwight		Lawsuit Not Yet Filed - SOL of 4/17/20				
Sorto, Cynthia	Los Angeles County Superior Court	<u>20STCV04718</u>				
<u>Waheed, Wahidi,</u> <u>Ernesto Madrgil, et</u> <u>al</u>	San Mateo Superior Court	<u>18-CIV-03214</u>				
Xue, Baoru		Demand Letter				
General Liability Litigation						
<u>Plaintiff</u>	Venue	Case Number(s)				
Arreola, Maria		Notice Of Intent				
Burell, Lisa		Request For Payment (auto-related)				
Cortes, Synthia	Los Angeles County Superior Court	<u>BC659504</u>				
	21					

<u>4" = "1" "DM\_US 165807051-6.079742.0063" ""</u> <u>US\_Active\114483166\V-4</u> 24

# Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 142 of 147

TT 1 T '		DC(01220
Hernandez, Iris	Los Angeles County Superior	<u>BC681228</u>
771 ' 1 1 1'	<u>Court</u>	DC712((2
Khajekarrameddin,	Los Angeles County Superior	<u>BC712663</u>
<u>Frangis</u>	Court	
Woolum, Dale		Notice Of Intent
	Ialpractice (Professional Liability	
<u>Plaintiff</u>	Venue	Case Number(s)
Aguirre Huerta, Rafael	Los Angeles County Superior	<u>19STCV29922</u>
	Court	
Allen, III, LeRoy &		Notice Of Intent
Perkins, LaTrice		
Ancira, Mateo, by and	Los Angeles County Superior	<u>19STCV23682</u>
through guardian ad	Court	
litem Yolanda Frias		
Arencibia, Vivian	Los Angeles County Superior	<u>BC673423</u>
	Court	
Birdsong, Erica		Notice Of Intent
Braun, Brian	Los Angeles County Superior	19STCV13074
	Court	
Bustamante, Marvela	Los Angeles County Superior	<u>19STCV22710</u>
	Court	
Childress, Alberta	Los Angeles County Superior	20STCV00666
	Court	
Correa, Maximino	Los Angeles County Superior	BC625271
<u> </u>	Court	
De Los Angeles, Maria		Notice Of Intent
Durham, Breta	Los Angeles County Superior	BC680199
	Court	
Ferguson, William	Los Angeles County Superior	<u>19STCV08618</u>
	Court	<u>1)510700010</u>
Flores, Luz	Los Angeles County Superior	<u>19STCV38243</u>
<u>110103, Luz</u>	Court	<u>17510750245</u>
Fuentes, Federico	Los Angeles County Superior	<u>19STCV08306</u>
	Court	<u></u>
Hernandez, Daniel	Los Angeles County Superior	BC687878
<u>Tiomanuez, Damer</u>	Court	<u><u><u></u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u></u>
Iniguez, Aida	Los Angeles County Superior	<u>BC616038</u>
miguez, Alua	Court	<u>BC010030</u>
Lopez, Monique	Potentially Compensable	No Complaint, Demand
Lopez, monique	Incident	Letter, Or Notice Of
	mordont	Intent At This Time
Mitchell, Alice		Letter To Preserve
Mitchen, Ance		Record
	25	<u>ICCOIU</u>

<u>4" = "1" "DM\_US 165807051-6.079742.0063" ""</u> <u>US\_Active\114483166\V-4</u>

# Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 143 of 147

Manatava Jaggiaa	Los Angolos County Sunorion	BC712998
Morataya, Jessica	Los Angeles County Superior	<u>BC/12998</u>
	Court	
Navarro, Richard	Los Angeles County Superior	<u>BC669695</u>
	Court	
Ramirez, Guadalupe		Notice Of Intent
Robles, Josefina	Los Angeles County Superior	<u>8C697012</u>
	Court	
Rodriguez, Yolanda by	Los Angeles County Superior	Notice Of Intent And
and through,	Court	Unserved Complaint
Successor-in-Interest		Filed, Case No.
Corinna Jimenez, et al		<u>19STCV29707</u>
<u>Rojas, John</u>	Los Angeles County Superior	Notice Of Intent And
<u>1(0]45, 50111</u>	Court	Unserved Complaint
		· · · · ·
		Filed, Case No.
		<u>19STCV29868</u>
		(SFMC not named or
		<u>served)</u>
Rubio, Alicia, by and	Los Angeles County Superior	<u>18STCV04410</u>
through, guardian ad	Court	
litem Cariola Rubio		
Saleh, Ebtissam	Los Angeles County Superior	BC718838
	Court	
Sanchez, Isabel		Notice Of Intent
Sprownson, Tammy	Los Angeles County Superior	5C724114
	Court	
Vazquez, Eduardo, by	Los Angeles County Superior	BC609128
and through biological	Court	
mother, Denia Carolina		
Alvardo		
Zavala, Adamari, by	Los Angeles County Superior	BC691431
and through, guardian	Court	
ad litem, A11 Adanesne		
Quiñones		

Workers' Compensation Cases					
Grievant	Date of Loss	Case Number(s)			
Acosta, Leslie	<u>5/24/2018</u>	<u>ADJ12476569</u>			
Navarro Gutierrez,	<u>10/23/2018</u>	<u>ADJ12149342</u>			
Giselle					

<u>4" = "1" "DM\_US 165807051-6.079742.0063" ""</u> <u>US\_Active\114483166\V-4</u> <u>26</u>

Sanders, Valencia	<u>11/17/2016</u>	ADJ10754107
Haith, Leilani	11/17/2016	ADJ1066160
Sutton, Tina	3/5/2013	
Jean, Marlene	3/2/2010	ADJ7610324
Ponce, Diana	10/19/2015	
Balderas-St. Pierre, Luz	10/17/2015	
Balderas-St. Pierre, Luz	11/18/2016	
Monteon, Marco	9/20/2016	ADJ12088746
Padilla, Rebecca	1/25/2009	
Werner, Wendie	10/17/2010	ADJ7620832
Monteon, Angie	9/5/2012	ADJ9223401
Gardea, Richard	4/26/2004	ADJ11072161
Casey, Rochelle	11/7/2015	ADJ11332260
Bombay, Maura	9/16/2006	ADJ5815856
Anaya, Gloria	3/21/2009	ADJ7228730
Jones, Lisa	1/6/2004	ADJ772382
Herrera, Martha	1/31/2012	ADJ8485301
Gardea, Richard	5/8/2014	ADJ11073773
Silva, Marcelina	1/14/2015	
Silva, Marcelina	4/6/2016	
Anaya, Gloria	8/14/2009	ADJ7228730
Lee, Belinda	1/13/2015	ADJ12991056
Mendez, Maria	<u>4/20/2015</u>	ADJ12209*530
Quiñonez, Maria	7/10/2015	
Hernandez, Lorena	<u>4/8/2013</u>	<u>ADJ8942433</u>
Reed, Charotta	<u>2/18/2010</u>	<u>ADJ7309641</u>
Wang, Xiaoling	<u>1/11/2012</u>	<u>ADJ8180221</u>
Ganzon, Nancy	<u>3/20/2012</u>	<u>ADJ8391108</u>
Martinez, Marina C	<u>2/6/2015</u>	<u>ADJ11955687</u>
Mendez, Maria	<u>2/8/2016</u>	<u>ADJ12209530</u>
Infante, Elida	<u>5/14/2006</u>	<u>ADJ4439918</u>
Harris, Gloria	<u>7/27/1984</u>	<u>ADJ1323480</u>
<u>Wills, Ora</u>	<u>8/17/1989</u>	<u>ADJ2755470</u>
McCoy, Virginia	<u>1/24/2017</u>	<u>ADJ10802194</u>
Maravilla, Fernando	<u>2/15/2017</u>	<u>ADJ10772424</u>
Sears, Andrea	<u>2/28/2017</u>	
Giannelli, Michelle	<u>3/2/2017</u>	
Jornacion, Wilfredo	<u>5/18/2017</u>	<u>ADJ12475946</u>
Martirosyan, Alik	<u>6/6/2017</u>	
Epps, Cleveland	<u>6/23/2017</u>	<u>ADJ11035557</u>
<u>Plaza, Caroline</u>	<u>6/27/2017</u>	<u>ADJ12089432</u>

<u>4" = "1" "DM\_US 165807051-6.079742.0063" ""</u> <u>US\_Active\114483166\V-4</u>

Elizalde Maria	6/27/2017	ADJ11186591
Harris, Stanley	11/22/2017	ADJ11498006
Espinoza, Ramon	10/10/2017	
Harris, Stanley	5/15/2018	ADJ11498006
Diaz, Gabriela	6/9/2018	
Malone, Veronica	12/21/2017	ADJ12557870
Medal, Sonia	1/9/2018	ADJ11664975
Gardea, Richard	6/30/2016	ADJ11072161
Torres-Olivares, Maria	1/11/2018	ADJ11799845
Dacio, Joicelyne	<u>12/31/2017</u>	
Roots Chandra	<u>1/26/2018</u>	ADJ12964974
King, LaVonda	2/1/2018	
Monteon Marco	2/8/2018	ADJ12088798
Zavala, Mayra	2/28/2018	
Jones, Leatrice	<u>4/5/2018</u>	
Casey, Rochelle	<u>7/18/2018</u>	<u>ADJ11395500</u>
Ramirez, Teresita	<u>8/22/2018</u>	
Smith, Rita	<u>10/1/2018</u>	
Meza, Maria	<u>10/4/2018</u>	<u>ADJ11906361</u>
Scrubb, Jacqueline	<u>11/16/2018</u>	
Chavez, Sandra	<u>12/4/2018</u>	
Boub, Theodore	<u>2/13/2018</u>	<u>ADJ11965517</u>
Armstrong, Gary	<u>2/1/2019</u>	<u>ADJ2488567</u>
Rodriguez, Isaac	<u>2/27/2019</u>	<u>ADJ2050556</u>
Ricketts, Chanel	<u>3/9/2019</u>	<u>ADJ12129607</u>
Cariaso, Sylvester	<u>7/3/2014</u>	
Monteon, Marco Jr	<u>2/25/2019</u>	<u>ADJ12092140</u>
Duffy, Earl	<u>5/23/2019</u>	
Jadraque-Olmos, Anna	<u>6/10/2019</u>	<u>ADJ1271881</u>
Dominguez, Cassidy	<u>5/21/2019</u>	<u>ADJ12411106</u>
<u>Williams, Linda</u>	<u>7/15/2019</u>	
<u>Hill, Larry</u>	<u>5/17/2019</u>	
Palomares, Reyna	<u>7/31/2019</u>	
Pham, Patty	<u>7/26/2019</u>	<u>ADJ12566853</u>
Duran, Monica	<u>8/2/2019</u>	<u>ADJ12795212</u>
Giannelli, Michelle	<u>8/27/2019</u>	
Santos, Grace Mary	<u>9/6/2019</u>	
Morales Silva, Angelita	<u>8/14/2019</u>	<u>ADJ12504328</u>
Morales Silva, Anqelita	<u>8/21/2019</u>	<u>ADJ2505618</u>
Duronslet, Devine	<u>9/21/2019</u>	
<u>Alonza, Kathryn</u>	<u>9/20/2019</u>	

<u>4" = "1" "DM\_US 165807051-6.079742.0063" ""</u> <u>US\_Active\114483166\V-4</u>

# Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 146 of 147

Malana Vananiaa	4/10/2010	AD1125570(0
Malone, Veronica	<u>4/19/2019</u>	<u>ADJ12557869</u>
Partida, Aurelio	<u>10/1/2019</u>	
Washington, Shirlee	<u>10/6/2019</u>	ADJ12640087
Plaza, Caroline	<u>5/30/2019</u>	<u>ADJ12748038</u>
Harris, Stanley	<u>11/8/2019</u>	<u>ADJ12983668</u>
Yepez, Consuelo	<u>11/12/2019</u>	
<u>Acosta, Amber</u>	<u>11/1/2019</u>	
Pinks Jeannene	<u>9/11/2019</u>	<u>ADJ12675128</u>
Marin, Solita	<u>12/18/2019</u>	
Marinoble, Tawny	<u>12/31/2019</u>	
Roots, Chandra	<u>10/15/2019</u>	<u>ADJ12964974</u>
Roots, Chandra	<u>2/26/2018</u>	<u>ADJ13012642</u>
	Appeals	
Certain Medicare and	Medicaid Group Appeals Handl	ed By Two Law Firms:
Stephenson,	Aquisto & Coleman; And Hoope	er, Lundy and
	Bookman	
Union	Grievant/Case Name	Grievance Number
United Nurses	Carcamo, Rosa	SM 04-18
Associations of		
CA/Union of Health		
Care Professionals		
	Labor Grievances	
Union	Grievant/Case Name	Grievant Number
SEIU	Class Action - for SEIU	
	Bargaining Unit	
SEIU	Class Action - for SEIU	
	Bargaining Unit	
UNAC	Employee given Corrective	SFMC SB 01-20
	Action Plan (CAP) without	
	union representative	
UNAC	Union grieved and	<u>SM 04-20</u>
	written/verbal warning given	
	to employee	
UNAC	Union grieved disciplinary	<u>SM 07-20</u>
	action and employee received	
	written warning. Union	
	position is that written	
	warning should have been a	
	CAP	
UNAC v. SFMC	National Labor Relations	<u>ULP-21-CA-231856</u>
	Board - Region 21	

<u>4" = "1" "DM\_US 165807051-6.079742.0063" ""</u> <u>US\_Active\114483166\V-4</u> <u>29</u>

# Case 2:18-bk-20151-ER Doc 6669-3 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 10 - 4/1/2020 Redline Page 147 of 147

UNAC Union grieved and CAP SM 06-20

<u>4" = "1" "DM\_US 165807051-6.079742.0063" ""</u> <u>US\_Active\114483166\V-4</u> <u>30</u>

#### Case 2:18-bk-20151-ER Doc 6669-4 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 11 - Revised APA Email Page 1 of 2

## **Esposito**, Zach

To: Subject: Joel Richlin (PHMI) RE: EXTERNAL: FW: Prime/St. Francis- Revised APA

From: Joel Richlin (PHMI) <<u>JRichlin@primehealthcare.com</u>>

Sent: Wednesday, April 1, 2020 2:05 PM

To: Moyron, Tania M. <<u>tania.moyron@dentons.com</u>>; Paul, Elspeth <<u>ElspethPaul@verity.org</u>>; David Galfus <<u>dgalfus@thinkbrg.com</u>>; Chadwick, Peter (VHS) <<u>pchadwick@thinkbrg.com</u>>; Garms, R. Matthew <<u>matthew.garms@dentons.com</u>>; Steve Aleman (PHMI) <<u>SAleman2@primehealthcare.com</u>>; Moloney, Jim (<u>JMoloney@cainbrothers.com</u>) <<u>JMoloney@cainbrothers.com</u>>; John Schlant [ThinkBRG] <<u>jschlant@thinkbrg.com</u>>; Koffroth, Nick <<u>nick.koffroth@dentons.com</u>>; Montgomery, Claude D. <<u>claude.montgomery@dentons.com</u>>; Maizel, Samuel R. <<u>samuel.maizel@dentons.com</u>>; Adcock, Rich <<u>RichAdcock@verity.org</u>>; Beith, Carsten <<u>cbeith@cainbrothers.com</u>>; Project Phoenix (<u>project\_phoenix@keybank.com</u>) <<u>project\_phoenix@keybank.com</u>> Cc: Gertler, Gary <<u>ggertler@mwe.com</u>>; Olson, Christopher <<u>Colson@mwe.com</u>>; Reisner, Jeffrey <<u>Jreisner@mwe.com</u>> Subject: Prime/St. Francis- Revised APA Importance: High

Hi All,

Attached is Prime's revised APA with a redline off of Verity's most recent draft. We apologize for the delay but as you may be aware, we were monitoring the hearing this morning and wanted to get that insight in crafting certain provisions, and we did consider the Court's rulings today in making our edits. The key changes I will call to your attention are: (1) no longer pre-paying for QAF (Seller still gets to keep QAF so no change in net value to Seller, but reduced cash consideration at close); (2) revised AR formula to address expected COVID losses and supplemental payments (as discussed on our call); (3) compromise language on Material Adverse Effect; and (4) certain bid protections such as minimum overbid and bid increments. Also attached is an updated valuation for the AR, payroll, and PTO based on most recent schedules in the data room.

We think that the most efficient next steps would be for us to have a quick call to walk you through these edits and our thinking. Please let us know if you would like to jump on a call and a good time. Once you understand the edits, we can set up a follow up call with principals.

Joel

A. Joel Richlin, Esq. Vice President, Deputy General Counsel, & Chief Litigation Counsel 3480 E. Guasti Road, Ontario, CA 91761 Direct: 909-235-4235 | Facsimile: 909-235-4316 Email: JRichlin@primehealthcare.com

Prime Healthcare Saving hospitals. Saving jobs. Saving lives.



The information contained in this email may be confidential and/or legally privileged, including HIPAA Protected Health Information or information that is proprietary or trade secret. It has been sent for the sole use of the intended recipient(s). If you are not the intended recipient, you are hereby notified that any unauthorized review, use, disclosure, dissemination, distribution, or copying of this communication is strictly prohibited. If you received this communication in error, please

#### Case 2:18-bk-20151-ER Doc 6669-4 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 11 - Revised APA Email Page 2 of 2

inform the sender and destroy all copies of the message. The organization accepts no liability for any damage caused by transmittance of this email. Thank you.

#### 

This message is a PRIVATE communication. This message and all attachments are a private communication sent by a law firm and may be confidential or protected by privilege. If you are not the intended recipient, you are hereby notified that any disclosure, copying, distribution or use of the information contained in or attached to this message is strictly prohibited. Please notify the sender of the delivery error by replying to this message, and then delete it from your system. Our <u>Privacy Policy</u> explains how we may use your personal information or data and any personal information or data provided or made available to us. Thank you.

Please visit <u>http://www.mwe.com/</u> for more information about our Firm.

Case 2:18-bk-20151-ER Doc 6669-5 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 12 - APA Conference Email Page 1 of 4

#### **Esposito, Zach**

From:	Paul, Elspeth <elspethpaul@verity.org></elspethpaul@verity.org>
Sent:	Thursday, April 2, 2020 1:33 PM
То:	Joel Richlin (PHMI); Moloney, Jim; Beith, Carsten
Cc:	Steve Aleman (PHMI); Chadwick, Peter (VHS)
Subject:	RE: [External Email] St. Francis APA

#### WARNING: This email originated outside the Prime Healthcare email system! Do Not Click links if this user is unknown.

1pm is great. I'll send dial in to everyone on this email. Thanks!

Elspeth Paul General Counsel Verity Health System *NEW ADDRESS:* 601 S. Figueroa Street, Suite 4050 Los Angeles, CA 90017 Office: (424) 367-0733 Cell: (310) 387-3433

From: Joel Richlin (PHMI) <JRichlin@primehealthcare.com>
Sent: Thursday, April 02, 2020 11:30 AM
To: Paul, Elspeth <ElspethPaul@verity.org>; Moloney, Jim <JMoloney@cainbrothers.com>; Beith, Carsten
<cbeith@cainbrothers.com>
Cc: Steve Aleman (PHMI) <SAleman2@primehealthcare.com>; Chadwick, Peter (VHS) <pchadwick@thinkbrg.com>
Subject: RE: [External Email] St. Francis APA

Hi All,

Haven't heard back. I thought the urgency was coming from Cain so we were just trying to accommodate here. Did we want to just all talk at 1pm when Elspeth and Peter were going to call Steve and I?

#### A. Joel Richlin, Esq.

Vice President, Deputy General Counsel, & Chief Litigation Counsel 3480 E. Guasti Road, Ontario, CA 91761 Direct: 909-235-4235 | Facsimile: 909-235-4316 Email: JRichlin@primehealthcare.com





From: Joel Richlin (PHMI)
Sent: Thursday, April 2, 2020 10:34 AM
To: Paul, Elspeth <<u>ElspethPaul@verity.org</u>>; Moloney, Jim <<u>JMoloney@cainbrothers.com</u>>; Beith, Carsten
<<u>cbeith@cainbrothers.com</u>>

#### Case 2:18-bk-20151-ER Doc 6669-5 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 12 - APA Conference Email Page 2 of 4 teve Aleman (PHMI) <SAleman2@primebealthcare.com>: Chadwick Peter (VHS) spcbadwick@thinkbrg.com>:

Cc: Steve Aleman (PHMI) <<u>SAleman2@primehealthcare.com</u>>; Chadwick, Peter (VHS) <<u>pchadwick@thinkbrg.com</u>> Subject: RE: [External Email] St. Francis APA

Elspeth, Jim, Carsten,

I understand that Jim and Carsten have just talked to Dr. Reddy. Dr. Reddy has asked me to get on the phone with you all immediately to discuss final deal points. Are you available now?

Joel

#### A. Joel Richlin, Esq.

Vice President, Deputy General Counsel, & Chief Litigation Counsel 3480 E. Guasti Road, Ontario, CA 91761 Direct: 909-235-4235 | Facsimile: 909-235-4316 Email: JRichlin@primehealthcare.com





From: Paul, Elspeth <<u>ElspethPaul@verity.org</u>>
Sent: Thursday, April 2, 2020 9:58 AM
To: Joel Richlin (PHMI) <<u>JRichlin@primehealthcare.com</u>>
Cc: Steve Aleman (PHMI) <<u>SAleman2@primehealthcare.com</u>>; Chadwick, Peter (VHS) <<u>pchadwick@thinkbrg.com</u>>
Subject: RE: [External Email] St. Francis APA

WARNING: This email originated outside the Prime Healthcare email system! Do Not Click links if this user is unknown.

Thanks for your flexibility.

From: Joel Richlin (PHMI) <<u>JRichlin@primehealthcare.com</u>>
Sent: Thursday, April 02, 2020 9:56 AM
To: Paul, Elspeth <<u>ElspethPaul@verity.org</u>>
Cc: Steve Aleman (PHMI) <<u>SAleman2@primehealthcare.com</u>>; Chadwick, Peter (VHS) <<u>pchadwick@thinkbrg.com</u>>
Subject: RE: [External Email] St. Francis APA

Sure, I understand that Jim is talking to Dr. Reddy now.

A. Joel Richlin, Esq. Vice President, Deputy General Counsel, & Chief Litigation Counsel 3480 E. Guasti Road, Ontario, CA 91761 Direct: 909-235-4235 | Facsimile: 909-235-4316 Email: JRichlin@primehealthcare.com





#### Case 2:18-bk-20151-ER Doc 6669-5 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 12 - APA Conference Email Page 3 of 4

From: Paul, Elspeth <<u>ElspethPaul@verity.org</u>>
Sent: Thursday, April 2, 2020 9:55 AM
To: Joel Richlin (PHMI) <<u>JRichlin@primehealthcare.com</u>>
Cc: Steve Aleman (PHMI) <<u>SAleman2@primehealthcare.com</u>>; Chadwick, Peter (VHS) <<u>pchadwick@thinkbrg.com</u>>
Subject: RE: [External Email] St. Francis APA

#### WARNING: This email originated outside the Prime Healthcare email system! Do Not Click links if this user is unknown.

Joel & Steve:

Is it possible to reschedule our time for this afternoon? Cain is looking to talk to Dr. Reddy this morning. Would 1 pm work for you? Peter and I can be available later, too, if that is better for you.

Thanks, Elspeth

Elspeth Paul General Counsel **Verity Health System** *NEW ADDRESS:* 601 S. Figueroa Street, Suite 4050 Los Angeles, CA 90017 Office: (424) 367-0733 Cell: (310) 387-3433

From: Joel Richlin (PHMI) <<u>JRichlin@primehealthcare.com</u>>
Sent: Wednesday, April 01, 2020 7:25 PM
To: Paul, Elspeth <<u>ElspethPaul@verity.org</u>>
Cc: Steve Aleman (PHMI) <<u>SAleman2@primehealthcare.com</u>>
Subject: [External Email] St. Francis APA

Please use caution when opening attachments or following links. Do not enter your credentials when prompted by external links. Please notify the Help Desk immediately if you believe you received a spam or phishing email and the attachment was opened, a link was clicked or credentials were provided.

Elspeth,

I spoke to management and have feedback on our discussion. Let me know once you have feedback on your end and can talk. I think we need to have a discussion about AR where I need Steve to join and we probably need BRG on your end, or whoever you deem best.

Thank you,

Joel

#### A. Joel Richlin, Esq.

Vice President, Deputy General Counsel, & Chief Litigation Counsel 3480 E. Guasti Road, Ontario, CA 91761 Direct: 909-235-4235 | Facsimile: 909-235-4316 Email: JRichlin@primehealthcare.com Case 2:18-bk-20151-ER Doc 6669-5 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 12 - APA Conference Email Page 4 of 4



#### 15 TOP HEALTH SYSTEMS 100 TOP

The information contained in this email may be confidential and/or legally privileged, including HIPAA Protected Health Information or information that is proprietary or trade secret. It has been sent for the sole use of the intended recipient(s). If you are not the intended recipient, you are hereby notified that any unauthorized review, use, disclosure, dissemination, distribution, or copying of this communication is strictly prohibited. If you received this communication in error, please inform the sender and destroy all copies of the message. The organization accepts no liability for any damage caused by transmittance of this email. Thank you.

#### Case 2:18-bk-20151-ER Doc 6669-6 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 13 - Accounts Receivable Email Page 1 of 2

# Esposito, Zach

From:	Paul, Elspeth <elspethpaul@verity.org></elspethpaul@verity.org>
Sent:	Thursday, April 2, 2020 7:15 PM
То:	Joel Richlin (PHMI)
Subject:	EXTERNAL:RE: [External Email] FW: EXTERNAL:COVID \$

WARNING: This email originated outside the Prime Healthcare email system! Do Not Click links if this user is unknown.

Looks good. Adding it to APA. Do you have the MAE language?

From: Joel Richlin (PHMI) <JRichlin@primehealthcare.com>
Sent: Thursday, April 02, 2020 4:25 PM
To: Paul, Elspeth <ElspethPaul@verity.org>
Subject: [External Email] FW: EXTERNAL:COVID \$

Please use caution when opening attachments or following links. Do not enter your credentials when prompted by external links. Please notify the Help Desk immediately if you believe you received a spam or phishing email and the attachment was opened, a link was clicked or credentials were provided.

Elspeth,

I like your edits, I propose only one slight addition for clarity, see below in yellow highlight.

#### A. Joel Richlin, Esq.

Vice President, Deputy General Counsel, & Chief Litigation Counsel 3480 E. Guasti Road, Ontario, CA 91761 Direct: 909-235-4235 | Facsimile: 909-235-4316 Email: JRichlin@primehealthcare.com





From: Paul, Elspeth <<u>ElspethPaul@verity.org</u>>
Sent: Thursday, April 2, 2020 3:29 PM
To: Joel Richlin (PHMI) <<u>JRichlin@primehealthcare.com</u>>
Subject: EXTERNAL:COVID \$

WARNING: This email originated outside the Prime Healthcare email system! Do Not Click links if this user is unknown.

Joel: With your new definition of accounts receivable and receivables, I'm thinking the edit is simply to move where the COVID \$ is referenced:

(p) (i) all accounts and interest thereupon, notes and interest thereupon and other receivables of Sellers, including, without limitation, accounts, notes or other amounts receivable, and all claims, rights, interests and proceeds related thereto, including all accounts and other receivables, , in each case arising from the rendering of services or provision of goods,

#### Case 2:18-bk-20151-ER Doc 6669-6 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 13 - Accounts Receivable Email Page 2 of 2

products or supplies to inpatients and outpatients at the Hospital, billed and unbilled, recorded and unrecorded (including any such amounts that were written-off by Sellers for any reason), for services, goods, products and supplies provided by Sellers prior to the Licensure Date whether payable by Medicare, Medi-Cal, Medicaid, or any other payor (including an insurance company), or any health care provider or network (such as a health maintenance organization, preferred provider organization or any other managed care program) or any fiscal intermediary of the foregoing, private pay patients, private insurance or by any other source <u>and all claims, rights, interests and proceeds relating to any grant or</u> <u>governmental awards directly or indirectly related to COVID-19</u> (collectively, "Accounts Receivable"); and (ii) <del>all claims, rights, interests and proceeds relating to any grant or governmental awards directly or indirectly related to COVID-19</del>, trauma payments, disproportionate share payments, cost report, claim, EHR or other similar appeals and Seller Cost Report settlements in each case arising from the rendering of services or provision of goods, products or supplies to inpatients and outpatients at the Hospital (the "Other Receivables" and together with the Accounts Receivable, the "Receivables");

Cash payment of Sixty-One Million Dollars (\$61,000,000) (the "A/R Target Amount") as consideration for the Accounts Receivable transferred at Closing (subject to adjustment in Section 1.12);

1.12(d)(i): if the Final A/R Collected is more than the A/R Target Amount, then such excess amount shall be paid by Purchaser to Sellers within ten (10) business days of Purchaser's delivery of the accounting of the Final A/R Collected to Sellers, provided that, in no event, shall amounts of any governmental grants or awards directly or indirectly related to COVID-19 be paid to Sellers under this Section 1.12(d)(i); For the avoidance of doubt, any amounts of any governmental grants or awards directly or indirectly related to COVID-19 must be removed from the calculation of the Final A/R Collected to determine whether Purchaser owes to Sellers any amount above the A/R Target Amount;

If this looks ok to you, I'll have Dentons add it to the APA we are working on. If you want to wait to get everything at once, that is fine, too. Just trying to move in real time.

LMK, Elspeth

Elspeth D. Paul General Counsel Verity Health System *NEW ADDRESS:* 601 S. Figueroa Street, Suite 4050 Los Angeles, CA 90017 Office: (424) 367-0733 Efax: (310) 878-0254 ElspethPaul@Verity.org



Confidentiality Notice: This transmission and any attached documents may be confidential and contain information protected by State and Federal Medical Privacy statutes and is legally privileged. They are intended for the use only by the addressee. If you are not the intended recipient of this transmission, or an agent of the intended recipient, you are prohibited from reading, disclosing, printing, saying, copying, using or otherwise disseminating any information contained in this transmission. If you received this transmission in error, please accept my apologies and notify me via replay e-mail. Please delete the entire message and its attachments. Thank you.

The information contained in this email may be confidential and/or legally privileged, including HIPAA Protected Health Information or information that is proprietary or trade secret. It has been sent for the sole use of the intended recipient(s). If you are not the intended recipient, you are hereby notified that any unauthorized review, use, disclosure, dissemination, distribution, or copying of this communication is strictly prohibited. If you received this communication in error, please inform the sender and destroy all copies of the message. The organization accepts no liability for any damage caused by transmittance of this email. Thank you.

## **Esposito**, Zach

From:	Moyron, Tania M. <tania.moyron@dentons.com></tania.moyron@dentons.com>
Sent:	Friday, April 3, 2020 11:31 AM
То:	Joel Richlin (PHMI); Paul, Elspeth; dgalfus; Chadwick, Peter (VHS); Garms, R. Matthew;
	Moloney, Jim (JMoloney@cainbrothers.com); John Schlant [ThinkBRG]; Koffroth, Nick; Montgomery, Claude D.; Maizel, Samuel R.; Adcock, Rich; Beith, Carsten; Project Phoenix
	(project_phoenix@keybank.com)
Cc:	Gertler, Gary; Olson, Christopher; Reisner, Jeffrey; Steve Aleman (PHMI)
Subject:	EXTERNAL:RE: Prime/St. Francis- Redline of Sale Order

# WARNING: This email originated outside the Prime Healthcare email system! Do Not Click links if this user is unknown.

Thanks, Joel. Dentons has been brought up to speed on the discussions that occurred with you and others through late yesterday. We are working with our team and Elspeth to incorporate those points into the APA. Best.



Tania M. Moyron Partner

D +1 213 243 6101 | M +1 310 402 4284 | US Internal 36101 tania.moyron@dentons.com Bio | Website

Dentons US LLP

Bingham Greenebaum > Cohen & Grigsby > Sayarh & Menjra > Larraín Rencoret > Hamilton Harrison & Mathews > Mardemootoo Balgobin > HPRP > Zain & Co. > Delany Law > Dinner Martin > For more on the firms that have joined Dentons, go to <u>dentons.com/legacyfirms</u>

Dentons is a global legal practice providing client services worldwide through its member firms and affiliates. This email may be confidential and protected by legal privilege. If you are not the intended recipient, disclosure, copying, distribution and use are prohibited; please notify us immediately and delete this copy from your system. Please see <u>dentons.com</u> for Legal Notices.

From: Joel Richlin (PHMI) <JRichlin@primehealthcare.com>

Sent: Friday, April 03, 2020 9:26 AM

**To:** Moyron, Tania M. <tania.moyron@dentons.com>; Paul, Elspeth <ElspethPaul@verity.org>; David Galfus

- <dgalfus@thinkbrg.com>; Chadwick, Peter (VHS) <pchadwick@thinkbrg.com>; Garms, R. Matthew
- <matthew.garms@dentons.com>; Moloney, Jim (JMoloney@cainbrothers.com) <JMoloney@cainbrothers.com>; John
- Schlant [ThinkBRG] <jschlant@thinkbrg.com>; Koffroth, Nick <nick.koffroth@dentons.com>; Montgomery, Claude D.

<claude.montgomery@dentons.com>; Maizel, Samuel R. <samuel.maizel@dentons.com>; Adcock, Rich

<RichAdcock@verity.org>; Beith, Carsten <cbeith@cainbrothers.com>; Project Phoenix

(project\_phoenix@keybank.com) <project\_phoenix@keybank.com>

Cc: ggertler@mwe.com; Colson@mwe.com; Jreisner@mwe.com; Steve Aleman (PHMI)

<SAleman2@primehealthcare.com>

Subject: RE: Prime/St. Francis- Redline of Sale Order

Importance: High

#### [External Sender]

Hi All,

Prime stands ready to quickly review and finalize the redline APA, but we don't have it yet, so I think this goes without saying that we are running a serious risk at this point of not having time to meet the deadline. What is also clear is we will not have time for further back and forth with any deviation from the agreements we reached yesterday.

I'm at my desk if anyone wants to talk and otherwise I'm waiting for the redline.

Joel

## A. Joel Richlin, Esq.

Vice President, Deputy General Counsel, & Chief Litigation Counsel 3480 E. Guasti Road, Ontario, CA 91761 Direct: 909-235-4235 | Facsimile: 909-235-4316 Email: JRichlin@primehealthcare.com





From: Joel Richlin (PHMI)

Sent: Thursday, April 2, 2020 10:01 PM

To: Moyron, Tania M. <<u>tania.moyron@dentons.com</u>>; Paul, Elspeth <<u>ElspethPaul@verity.org</u>>; David Galfus <<u>dgalfus@thinkbrg.com</u>>; Chadwick, Peter (VHS) <<u>pchadwick@thinkbrg.com</u>>; Garms, R. Matthew <<u>matthew.garms@dentons.com</u>>; Moloney, Jim (<u>JMoloney@cainbrothers.com</u>) <<u>JMoloney@cainbrothers.com</u>>; John Schlant [ThinkBRG] <<u>jschlant@thinkbrg.com</u>>; Koffroth, Nick <<u>nick.koffroth@dentons.com</u>>; Montgomery, Claude D. <<u>claude.montgomery@dentons.com</u>>; Maizel, Samuel R. <<u>samuel.maizel@dentons.com</u>>; Adcock, Rich <<u>RichAdcock@verity.org</u>>; Beith, Carsten <<u>cbeith@cainbrothers.com</u>>; Project Phoenix (<u>project\_phoenix@keybank.com</u>) <<u>project\_phoenix@keybank.com</u>>
Cc: ggertler@mwe.com; Colson@mwe.com; Jreisner@mwe.com; Steve Aleman (PHMI) <<u>SAleman2@primehealthcare.com</u>>
Subject: Prime/St. Francis- Redline of Sale Order

Hi All,

Attached is a clean and redline of the template Sale Order. My understanding is that Dentons is working on a redline that reflects the parties' negotiations and agreements today. I reviewed the AR language with Elspeth and that should be good to go. I also sent Elspeth the language for the structure of the MAE provision and my understanding is that Cain has agreed on the calculation methodology with Steve and Cain is drafting that. So we are waiting in Cain's draft of that calculation and otherwise waiting on Dentons to turn a redline of the rest.

Please let me know if anything is in our court. We are otherwise prepared for the wire transfer and will be ready to review and finalize the redline in the morning. We can also review tonight if available.

Joel

## A. Joel Richlin, Esq.

Vice President, Deputy General Counsel, & Chief Litigation Counsel

Case 2:18-bk-20151-ER Doc 6669-7 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 14 - APA Negotiations Email Page 3 of 3 3480 E. Guasti Road, Ontario, CA 91761 Direct: 909-235-4235 | Facsimile: 909-235-4316 Email: JRichlin@primehealthcare.com





The information contained in this email may be confidential and/or legally privileged, including HIPAA Protected Health Information or information that is proprietary or trade secret. It has been sent for the sole use of the intended recipient(s). If you are not the intended recipient, you are hereby notified that any unauthorized review, use, disclosure, dissemination, distribution, or copying of this communication is strictly prohibited. If you received this communication in error, please inform the sender and destroy all copies of the message. The organization accepts no liability for any damage caused by transmittance of this email. Thank you.

# ASSET PURCHASE AGREEMENT

# By and Among

# ST. FRANCIS MEDICAL CENTER, VERITY HOLDINGS, LLC, and VERITY HEALTH SYSTEM OF CALIFORNIA, INC.

(as "Sellers")

and

# PRIME HEALTHCARE SERVICES, INC.

(as "Purchaser")

Dated \_\_\_\_\_, 2020

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 2 of 147

# **INDEX OF SCHEDULES & EXHIBITS**

# Schedules

Schedule 1.1(a)(i)	Base Price Adjustment
Schedule 1.4(c)	Owned Real Property
Schedule 1.7(b)	Licenses
Schedule 1.7(c)(ii)	Leased Real Property
Schedule 1.7(c)(iii)	Tenant Leases
Schedule 1.7(d)	Assigned Contracts and Assigned Leases
Schedule 1.8(bb)	Certain Other Excluded Assets
Schedule 1.9(i)	Other Obligations
Schedule 2.4	Seller's Knowledge of Breach or Default
Schedule 2.5	Brokers and Finders (Sellers)
Schedule 2.6	Compliance (Sellers)
Schedule 2.7	Required Consents (Sellers)
Schedule 2.8	Legal Proceedings (Sellers)
Schedule 2.9(b)	Environmental Matters (Sellers)
Schedule 2.11	Missing Material Licenses (Sellers)
Schedule 2.13	Non-Bankruptcy Court Legal Proceedings (Sellers)
Schedule 2.19	List of Excluded Individuals (Seller)
Schedule 3.4	No Violation (Purchaser)
Schedule 3.7	Legal Proceedings (Purchaser)
Schedule 5.3	"Hospital Employees"
Schedule 11.3(b)	Tax Allocation Schedule

# **Exhibits**

No.	Description	Status
Exhibit 1.2	Escrow Agreement	to come
Exhibit 1.3(a)	Leaseback Agreement	Attached
Exhibit 1.3(b)	Interim Management Agreement	Attached
Exhibit 1.4(a)	Bill of Sale	to come
Exhibit 1.4(b)	Real Estate Assignment Agreements	to come
Exhibit 1.4(c)	Quitclaim Deeds	to come
Exhibit 1.4(d)	Transfer Agreement	to come
Exhibit 1.4(f)	Transition Services Agreement	Attached
Exhibit	Accepted Conditions (Attorney General Conditions)	to come
<u>5.95.8</u> (c)		

# ASSET PURCHASE AGREEMENT

This Asset Purchase Agreement (the "Agreement") is made and entered into as of \_\_\_\_\_\_\_, 2020 (the "Signing Date") by and among St. Francis Medical Center, a California nonprofit public benefit corporation ("SFMC"), Verity Holdings, LLC ("Verity Holdings"), a California limited liability company and Verity Health System of California, Inc., a California nonprofit public benefit corporation ("VHS" and, together with SFMC and Verity Holdings, the "Sellers" and each individually a "Seller") and Prime Healthcare Services, Inc., a Delaware corporation ("Purchaser" and, together with the Sellers, the "Parties" and each individually a "Party").

# RECITALS

A. SFMC engages in the business of operating a hospital known as St. Francis Medical Center ("St. Francis") in the City of Lynwood, California, including the hospital pharmacy, laboratory and emergency department as well as through the medical office buildings and clinics owned or operated by SFMC (with St. Francis, collectively referred to herein, as the "Hospital").

B. VHS is the sole member of SFMC and Verity Holdings. Verity Holdings owns and rents certain properties on or near the Hospital campus.

C. Purchaser desires to purchase from Sellers, and Sellers desire to sell to Purchaser, the assets described in <u>Section 1.7</u> below for the consideration and upon the terms and conditions contained in this Agreement.

D. On August 28, 2018, Sellers filed voluntary petitions for relief under Chapter 11 of Title 11 of the United States Code (the "**Bankruptcy Code**") in the United States Bankruptcy Court for the Central District of California, Los Angeles Division, before the Honorable Ernest M. Robles (the "**Bankruptcy Court**"), which are jointly administered with their affiliates under Case No. 18-20151 (the "**Bankruptcy Cases**").

E. On February 10, 2020, Sellers filed the Debtors' Notice of Motion and Motion for the Entry of (I) an Order (1) Approving Form of Asset Purchase Agreement; (2) Approving Auction Sale Format, Bidding Procedures and Stalking Horse Bid Protections; (3) Approving Form of Notice to be Provided to Interested Parties; (4) Scheduling a Court Hearing to Consider Approval of the Sale to the Highest Bidder; and (5) Approving Procedures Related to the Assumption of Certain Executory Contracts and Unexpired Leases; and (II) an Order (A) Authorizing the Sale of Property Free and Clear of All Claims, Liens and Encumbrances; Memorandum Of Points And Authorities In Support Thereof [Docket No. 4069] (the "Bidding Procedures Motion") which was approved by an order of the Bankruptcy Court (the "Bidding Procedures Order") [Docket No. 4165] and pursuant thereto the Parties have entered into this Agreement.

F. The Parties intend to effectuate the transactions contemplated by this Agreement through a sale of the Assets pursuant to the Bidding Procedures and as approved by the Bankruptcy Court pursuant to Sections 363 and 365 of the Bankruptcy Code.

### AGREEMENT

NOW, THEREFORE, in consideration of the foregoing premises and the mutual promises and covenants contained in this Agreement, and for their mutual reliance and incorporating into this Agreement the above recitals, the Parties hereto agree as follows:

# ARTICLE 1 SALE AND TRANSFER OF ASSETS; CONSIDERATION; CLOSING

#### 1.1 <u>Purchase Price</u>.

(a) Subject to the terms and conditions of this Agreement, the aggregate consideration shall be as follows:

(i) Cash payment to Sellers of Two Hundred Million Dollars-(\$200,000,000) ("Base Price"); Payment to Sellers of a cash base purchase price ("Base Price") as follows: Two Hundred Million Dollars (\$200,000,000): provided, that in the event at Closing the Annualized Normalized EBITDA (as defined in Schedule 1.1(a)(i)) is more than Fifty Million Dollars (\$50,000,000) less than the Prior Period Annualized Normalized EBITDA (as defined in Schedule 1.1(a)(i)), the Purchaser shall offset from the portion of the remaining QAF V net receipts collected after Closing (but not by more than an aggregate of Thirty Million Dollars (\$30,000,000)) one dollar for every dollar of difference between Prior Period Annualized Normalized EBITDA and Annualized Normalized EBITDA over Fifty Million Dollars (\$50,000,000) up to Eighty Million Dollars (\$80,000,000). **INEED TO DETERMINE MECHANISM FOR** DELIVERY AND REVIEW OF ANNUALIZED NORMALIZED **EBITDAI**:

(ii) Sellers shall retain, as an Excluded Asset, an estimated amountof the QAF V Payments (defined below) and the QAF VI Seller Net Payments (defined below) as described in Section 1.8(b), which are currently estimated at Twenty-Nine Million Dollars (\$29,000,000) in connection with the QAF V payments Payments and Eighty-Three Million Dollars (\$83,000,000) in connection with QAF VI payments (subject toadjustment as described in Section 1.8(b)) the QAF VI Seller Net Payments;

(iii) Cash payment of Sixty-One Million Dollars (\$61,000,000) (the "A/R Target Amount") as consideration for the Accounts Receivable transferred at Closing (subject to adjustment in Section 1.12);

(iv) Cash payment for Sellers' payroll liabilities at Closing (the "**Payroll Amount**") (which as of October 31, 2019, had an aggregate value of approximately Five Million Dollars (\$5,000,000));

(v) Cash payment for accrued vacation and other paid time-off of Sellers' employees at Closing (the "**PTO Amount**") (which as of October 31, 2019, had an aggregate value of approximately Ten Million Dollars (\$10,000,000));

(vi) An amount equal to the Cure Costs (defined below) associated with outstanding liabilities of Sellers under any Assigned Leases and/or Assigned Contracts; and

(vii) An amount determined in accordance with <u>Section 1.6</u>.

(b) At the Closing, Purchaser shall pay to Sellers, by wire transfer of immediately available funds to the accounts specified by Sellers to Purchaser in writing, an aggregate amount equal to the Base Price, *plus* the A/R Target Amount, *plus* the Cure Cost Amount, *plus* the PTO Amount, *plus* the Payroll Amount (collectively, the "**Purchase Price**"), *minus* the Deposit (defined below).

(c) For purposes of this Agreement, the "QAF Program" means the California Department of Health Care Services ("DHCS") Hospital Quality Assurance Fee Programs V ("QAF V") and VI ("QAF VI"). During the period prior to the Closing, Sellers shall pay or permit DHCS to offset from amounts owed to St. Francis any fees due and owing prior to the Closing under QAF V and QAF VI (such payments or offsets shall be included in the formula described in Section 1.8(b)), and Sellers shall be entitled to retain all payments received under QAF V and QAF VI.

(d) Separate from, and in addition to, the Purchase Price, Purchaser commits to invest Forty-Seven Million Dollars (\$47,000,000) in capital expenditures for St. Francis (including NPC-3 seismic compliance responsibilities).

1.2 Deposit. Purchaser has deposited an amount equal to ten percent (10%) of the Purchase Price §27,725,342.48 (the "Initial Deposit") by wire transfer to Chicago Title Insurance Company ("Escrow Agent") pursuant to that certain Escrow Agreement attached hereto as Exhibit 1.2. Within two (2) business day after the date of the conclusion of the auction for the Assets, if any, if Purchaser is the Prevailing Bidder (as defined below) and as a result of such auction has opted to increase the Purchase Price (the "Auction Purchase Price"), Purchaser shall deposit an additional amount with the Escrow Agent such that the total amount deposited with the Escrow Agent is equal to ten percent (10%) of the Auction Purchase Price (the Initial Deposit and any additional amount deposited with the Escrow Agent (if any) referred to collectively as the "Deposit") by wire transfer to the Escrow Agent. All fees of the Escrow Agent shall be paid by Purchaser. The Deposit shall be non-refundable in all events, except in the event the Closing does not occur due to Purchaser's termination of the Agreement pursuant to Sections 9.1 (a), (c)(i), (d), (f), (g), (h) or (i) hereof. Upon Closing, the Deposit will be credited against the Purchase Price.

1.3 <u>Closing Date</u>. The consummation of the transactions contemplated by this Agreement (the "**Closing**") shall take place at the offices of Dentons US LLP, 601 South Figueroa St., Suite 2500, Los Angeles, CA 90017-5704 (the day on which Closing actually

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 6 of 147

occurs, the "Closing Date") within three (3) business days following the satisfaction or waiver of the conditions set forth in ARTICLE 7 and ARTICLE 8, and other than those conditions that by their nature are to be satisfied at Closing but subject to fulfillment or waiver of those conditions. The Closing shall be deemed to occur and to be effective as of 12:00 a.m. pacific time on the day immediately after the Closing Date (the "Effective Time"). Purchaser and Sellers agree that because the change of ownership and regulatory approval process may take an extended period of time, at the Effective Time, the Assets (less any Assets constituting drugs or pharmacy assets, the Medicare/Medi-Cal Agreements, the Lockboxes, the Transferred Managed Care Agreements and the Transferred Private Payor Agreements (as such terms are defined below)) will be sold to Purchaser and immediately leased back to Sellers (substantially in the form of the Leaseback Agreement attached hereto as Exhibit 1.3(a), the "Leaseback Agreement"), with a concurrent management arrangement (substantially in the form of the Interim Management Agreement attached hereto as Exhibit 1.3(b), the "IMA"). On the effective date that Purchaser obtains a general acute care hospital license from the California Department of Public Health ("CDPH") and a hospital pharmacy permit from the California State Board of Pharmacy ("BOP") (i) the Leaseback Agreement and IMA will terminate and (ii) the drugs and pharmacy assets, the Medicare/Medi-Cal Agreements, the Lockboxes, the Transferred Managed Care Agreements and the Transferred Private Payor Agreements will be transferred to Purchaser (without payment of any additional Purchase Price) (the "Licensure Date"). For the avoidance of any doubt, the Licensure Date shall be the date the Purchaser's hospital license and pharmacy permit are effective, even if they are not actually issued until a later date. The Licensure Date may be determined based on oral assurances from CDPH and the BOP.

1.4 <u>Items to be Delivered by Sellers at Closing</u>. At or before the Closing, Sellers shall deliver, or cause to be delivered, to Purchaser the following:

(a) a Bill of Sale substantially in the form of <u>Exhibit 1.4(a)</u> attached hereto (the "**Bill of Sale**"), duly executed by each Seller;

(b) Real Estate Assignment Agreements (the "**Real Estate Assignments**") in the form of <u>Exhibit 1.4(b)</u> attached hereto with respect to (i) the Leased Real Property, and (ii) the Tenant Leases, each duly executed by the applicable Sellers;

(c) Quitclaim Deeds in the form of <u>Exhibit 1.4(c)</u> attached hereto with respect to the real property listed in <u>Schedule 1.4(c)</u>, together with all plant, buildings, structures, installments, improvements, fixtures, betterments, additions and constructions in progress situated thereon (the "**Owned Real Property**") duly executed by SFMC and Verity Holdings as applicable;

(d) an Assigned Contract Transfer Agreement (the "**Transfer Agreement**") in the form of Exhibit 1.4(d) attached hereto, duly executed by the applicable Sellers;

(e) evidence of payment of all Cure Costs;

(f) the Transition Services Agreement (the "**Transition Services Agreement**") in form attached hereto as <u>Exhibit 1.4(f)</u>, duly executed by the applicable Sellers;

(g) the Leaseback Agreement and IMA, duly executed by SFMC;

(h) favorable certificates of good standing, of each Seller, issued by the State of California, dated no earlier than a date which is fifteen (15) calendar days prior to the Closing Date;

(i) a duly executed certificate of an officer of each Seller certifying to Purchaser (i) the incumbency of the officers of such Seller on the Signing Date and on the Closing Date and bearing the authentic signatures of all such officers who shall execute this Agreement and any additional documents contemplated by this Agreement and (ii) the due adoption and text of the resolutions or consents of the Board of Directors of such Seller authorizing (I) the transfer of the Assets and transfer of the Transferred Obligations by such Seller to Purchaser and (II) the due execution, delivery and performance of this Agreement and all additional documents contemplated by this Agreement, and that such resolutions have not been amended or rescinded and remain in full force and effect on the Closing Date;

(j) a certified copy of the Sale Order (as defined below);

(k) non-foreign affidavits of SFMC and Verity Holdings, as applicable, dated as of the Closing Date, in the form of Exhibit 1.4(k); and

(1) Any such other instruments, certificates, consents or other documents which Purchaser and Sellers mutually deem reasonably necessary to carry out the transactions contemplated by this Agreement and to comply with the terms hereof.

1.5 <u>Items to be Delivered by Purchaser at Closing</u>. At or before the Closing, Purchaser shall deliver or cause to be delivered to Sellers the following:

(a) payment of the Purchase Price, minus the Deposit;

(b) a duly executed certificate of the Secretary of Purchaser certifying to Sellers (a) the incumbency of the officers of Purchaser on the Signing Date and on the Closing Date and bearing the authentic signatures of all such officers who shall execute this Agreement and any additional documents contemplated by this Agreement and (b) the due adoption and text of the resolutions of the Board of Directors of Purchaser authorizing the execution, delivery and performance of this Agreement and all additional documents contemplated by this Agreement, and that such resolutions have not been amended or rescinded and remain in full force and effect on the Closing Date;

(c) favorable certificate of good standing, of Purchaser, issued by the California Secretary of State dated no earlier than a date which is fifteen (15) calendar days prior to the Closing Date;

- (d) the Transition Services Agreement, duly executed by Purchaser;
- (e) the Leaseback Agreement and IMA, duly executed by Purchaser;
- (f) the Bill of Sale, duly executed by Purchaser;

- (g) the Real Estate Assignment(s), duly executed by Purchaser;
- (h) the Transfer Agreement, duly executed by Purchaser;

(i) Preliminary Change of Ownership Report(s) (BOE-502-A) with respect to the Owned Real Property, duly executed by Purchaser; and

(j) any such other instruments, certificates, consents or other documents which Purchaser and Sellers mutually deem reasonably necessary to carry out the transactions contemplated by this Agreement and to comply with the terms hereof.

1.6 <u>Prorations and Utilities</u>. All items of income and expense listed below with respect to the Assets shall be prorated in accordance with the principles and the rules for the specific items set forth hereafter:

(a) All transfer, conveyance, sales, use, stamp, similar state and local taxes arising from the sale of the Assets hereunder shall be the responsibility of, and allocated to, Purchaser.

Other than the Utility Deposits (defined below), which are governed by (b)Section 1.8(n), and other than with respect to Cure Costs, the following costs and expenses shall be prorated based upon the payment period (*i.e.*, calendar or other tax fiscal year) to which the same are attributable: (i) all real estate and personal property lease payments, real estate and personal property taxes, real estate assessments and other similar charges against real estate, (ii) power and utility charges, (iii) payments made by Sellers for information technology software or services for periods after the Effective Time, (iv) payments made by Sellers in association with Licenses or dues paid to government and non-governmental agencies for calendar year 2020, and (v) other similar costs for items or services which continue past the Effective Time (collectively, the "Prorated Charges"). Sellers shall pay at or prior to the Closing (or Purchaser shall receive credit for) any unpaid Prorated Charges attributable to periods or portions thereof occurring prior to the Effective Time, and Purchaser shall be responsible for or, to the extent previously paid by Sellers, pay to Sellers at the Closing all Prorated Charges attributable to periods or portions thereof occurring from and after the Effective Time. In the event that as of the Closing Date the actual tax bills for the tax year or years in question are not available and the amount of taxes to be prorated as aforesaid cannot be ascertained, then rates, millages and assessed valuation of the previous year, with known changes, shall be used. The Parties agree that if the real estate and personal property tax prorations are made based upon the taxes for the preceding tax period, the prorations shall be re-prorated after the Closing. As to power and utility charges, "final readings" as of the Closing Date shall be ordered from the utilities; the cost of obtaining such "final readings," if any, shall be paid by Purchaser.

(c) Sellers shall be entitled to all rents and other payments under Tenant Leases accruing for the period prior to the Effective Time ("**Pre Effective Time Lease Amounts**"), and Purchaser shall be entitled to all rents and other payments under tenant leases accruing for the period after the Effective Time ("**Post Effective Time Lease Amounts**" and together with the Pre Effective Time Lease Amounts, the "Lease Amounts"). All Lease Amounts that are collected prior to the Closing shall be prorated as of the Closing in accordance

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 9 of 147

with the immediately preceding sentence. All Lease Amounts that are accrued but unpaid as of the Closing (including, without limitation, rents and other payments accrued prior to the Closing but payable in arrears after the Closing) (collectively, the "Unpaid Amounts") shall belong to Sellers, and Purchaser shall, upon receipt of said rents and other payments, receive the same in trust for Sellers and shall promptly remit any of such amounts to Sellers within ten (10) days after Purchaser's receipt of same.

(d) If Purchaser requests that Sellers transfer electronic medical records in a specific electronic format due to the Sellers' electronic medical record ("EMR") system and Purchaser's EMR system not being interoperable, then Purchaser shall reimburse, on the Closing Date, all amounts paid or to be paid by Sellers to transfer electronic medical records to Purchaser in such different electronic format.

(e) All prorations and payments to be made under the foregoing provisions shall be agreed upon by Purchaser and Sellers prior to the Closing and shall be binding upon the Parties; *provided, however*, with respect to the Unpaid Amounts, in the event any proration, apportionment or computation shall prove to be incorrect for any reason, then either Sellers or Purchaser shall be entitled to an adjustment to correct the same, provided that said Party makes written demand on the Party from whom it is entitled to such adjustment within fifteen (15) calendar days after the erroneous payment or computation was made, or such later time as may be required, in the exercise of due diligence, to obtain the necessary information for proration.

(f) This <u>Section 1.6</u> shall survive Closing.

1.7 <u>Transfer of Sellers Assets</u>. On the Closing Date, subject to the terms and conditions of this Agreement and except as otherwise contemplated in <u>Section 1.3</u>, Sellers shall sell, assign, transfer, convey and deliver to Purchaser, free and clear of all interests, including but not limited to all liens, privileges, pledges, security interests, rights of first refusal, options, defects in title and encumbrances <u>("Encumbrances")</u> other than the Permitted Exceptions (defined below), and Purchaser shall acquire, all of Sellers' right, title and interest in and to only the following assets and properties, as such assets shall exist on the Closing Date, to the extent not included among the Excluded Assets, such transfer being deemed to be effective at the Effective Time (the "Assets"):

(a) all of the tangible personal property owned by Sellers, or to the extent assignable or transferable by Sellers, and used by Sellers in the operation of the Hospital, including equipment, furniture, fixtures, machinery, vehicles, office furnishings and leasehold improvements (the "**Personal Property**");

(b) subject to <u>Section 1.7(e)</u>, all of Sellers' rights, to the extent assignable or transferable, to all licenses, permits, approvals, certificates of exemption, franchises, accreditations and registrations and other governmental licenses, permits or approvals issued to Sellers for use in the operation of the Hospital (the "**Licenses**"), including, without limitation, the Licenses set forth on <u>Schedule 1.7(b)</u>, except to the extent Purchaser elects, in its discretion, not to take assignment of any such Licenses;

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 10 of 147

(c) all of Sellers' interest in and to the Owned Real Property and all of Sellers' interest, to the extent assignable or transferable and that have been designated by Purchaser, in and to all of the following (the "Assigned Leases"): (i) the real property leases for all real property leased by Sellers and set forth on <u>Schedule 1.7(c)(i)</u> (the "Leased Real Property"), and (ii) the real property leased or subleased by Sellers to a third party and set forth on <u>Schedule 1.7(c)(ii)</u> (the "Tenant Leases");

(d) all of Sellers' interest in, and all of Sellers' obligations due under, from and after the Effective Time, to the extent assignable or transferable, all contracts and agreements (including, but not limited to, purchase orders) that have been designated by Purchaser as an Assigned Contract, pursuant to <u>Section 1.11</u> and appearing on <u>Schedule 1.7(d)</u>;

(e) all of Sellers' interest in, from and after the Licensure Date, to the extent assignable or transferable, the Hospital's Medicare Provider Agreement (and provider number) and the Hospital's Medi-Cal Provider Agreement (and provider number) (collectively, the "Medicare/Medi-Cal Agreements");

(f) all of Sellers' interest in, and all of Sellers' obligations due under, from and after the Licensure Date, to the extent assignable or transferable, in and to any of the Hospital's managed care, pre-paid, capitated or other full-risk health plan agreements (collectively, the "Managed Care Agreements") that have been designated by Purchaser as an Assigned Contract pursuant to <u>Section 1.11</u> (to the extent so designated, the "Transferred Managed Care Agreements");

(g) subject to <u>Section 4.6</u>, all of Sellers' interest in, and all of Sellers' obligations due under, from and after the Licensure Date, to the extent assignable or transferable, in and to any of the Hospital's services, participation or provider agreements with private health plans, insurers or other third party payors (collectively, the "**Private Payor Agreements**") that have been designated by Purchaser as an Assigned Contract pursuant to <u>Section 1.11</u> (to the extent so designated, the "**Transferred Private Payor Agreements**"), *provided that*, Private Payor Agreements shall not include any "risk-sharing" agreements with independent physician associations;

(h) to the extent assignable or transferable, all inventories of supplies, drugs, food, janitorial and office supplies and other disposables and consumables (i) located at the Hospital or (ii) used in the operation of the Hospital (the "**Inventory**") except as set forth in <u>Sections 1.3</u> and <u>1.8(i)</u>;

(i) other than Utility Deposits, all prepaid rentals, deposits, prepayments and similar amounts relating to the Assigned Contracts and/or the Assigned Leases, which were made with respect to the operation of the Hospital (the "**Prepaids**");

(j) to the extent assignable or transferable, all of the following that are not proprietary to Sellers and/or owned by or proprietary to Sellers' affiliates: operating manuals, files and computer software with respect to the operation of the Hospital, including, without limitation, all patient records, medical records, employee records, billing records, financial records, equipment records, construction plans and specifications, and medical and

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 11 of 147

administrative libraries; *provided, however*, that any electronic medical records may be transferred in paper or "pdf" if Sellers' EMR system and Purchaser's EMR system are not interoperable;

(k) to the extent assignable or transferable, all rights in all warranties of any manufacturer or vendor in connection with the Personal Property;

(1) all right, title and interest in and to the name "St. Francis Medical Center," including any associated Hospital trademarks, service marks, trade names, logos and domain names but excluding the domain <u>https://stfrancis.verity.org</u> and content therein;

(m) all goodwill of the Hospital evidenced by the Assets;

(n) to the extent transferable or assignable, Sellers' right or interest in the telephone and facsimile numbers used with respect to the operation of the Hospital;

(o) to the extent assignable or transferable, Sellers' lock box account(s) associated with Medicare or Medi-Cal fee-for-service receivables (the "Lockboxes") on or after the Licensure Date;

(p) (i) all accounts and interest thereupon, notes and interest thereupon and other receivables of Sellers, including, without limitation, accounts, notes or other amounts receivable, and all claims, rights, interests and proceeds related thereto, including all accounts and other receivables, , in each case arising from the rendering of services or provision of goods, products or supplies to inpatients and outpatients at the Hospital, billed and unbilled, recorded and unrecorded (including any such amounts that were written-off by Sellers for any reason), for services, goods, products and supplies provided by Sellers prior to the Licensure Date whether payable by Medicare, Medi-Cal, Medicaid, or any other payor (including an insurance company), or any health care provider or network (such as a health maintenance organization, preferred provider organization or any other managed care program) or any fiscal intermediary of the foregoing, private pay patients, private insurance or by any other source (collectively, "Accounts-Receivable"); and (ii)and all claims, rights, interests and proceeds relating to any grant or governmental awards directly or indirectly related to COVID-19,19 (collectively, "Accounts Receivable"); and (ii) trauma payments, disproportionate share payments (subject to Section 1.8(c)), California Health Foundation & Trust payments, cost report, claim, EHR or other similar appeals and Seller Cost Report settlements in each case arising from the rendering of services or provision of goods, products or supplies to inpatients and outpatients at the Hospital (the "Other Receivables" and together with the Accounts Receivable, the "Receivables");

(q) all documents, records, correspondence, work papers and other documents, other than patient records, primarily relating to the Receivables;

(r) all QAF payments due to the Hospital from the State of California or any of its administrative entities or other entitles, including without limitations, Medi-Cal managed care plans, <u>that are both</u> received on and after the Licensure Date and attributable to QAF VI and any subsequent QAF Program, regardless of the state fiscal year for which the payments are made in reference to and regardless of the state fiscal year for which the data was derived to

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 12 of 147

calculate eligibility for such payments, other than the QAF payments received prior to Closing or specifically excluded pursuant to <u>Section 1.8(b)</u> as QAF VI Seller Net Payments;

(s) except as set forth in <u>Section 1.8(1)</u> (*i.e.*, certain causes of action), all claims, causes of action, choses in action, rights of recovery, rights of set off and rights of recoupment of Sellers against third parties related to or associated with the physical condition of any of the Assets; and

(t) to the extent assignable or transferable, any other assets owned by Sellers (which are not otherwise specifically described above in this <u>Section 1.7</u>) that are used primarily in the operation of the Hospital.

As used herein, the term "**Permitted Exceptions**" means (i) the Transferred Obligations; (ii) liens for taxes not yet due and payable; (iii) easements, rights of way, zoning ordinances and other similar encumbrances affecting real property; (iv) imperfections of title or encumbrances identified in the Title Commitments (defined below) other than those specifically identified in <u>Schedule 1.7(I)</u>; and (iv) other imperfections of title or encumbrances, if any, which are not monetary in nature and that are not, individually or in the aggregate, material to the business of the Hospital.

1.8 <u>Excluded Assets</u>. Notwithstanding anything to the contrary in <u>Section 1.7</u>, Sellers shall retain all interests, rights and other assets owned directly or indirectly by it (or any of Sellers' affiliates) which are not among the Assets, including, without limitation, the following interests, rights and other assets of Sellers (collectively, the "**Excluded Assets**"):

(a) cash, cash equivalents and short-term investments;

(b) all QAF payments received prior to Closing under any QAF programProgram and, in addition, with respect to the QAF VI Program,([]) the Excluded Assets shall include the "all payments received or to be received by the Hospital with respect to QAF V (the "QAF V Payments"), and (II) with respect to QAF VI, the Excluded Assets shall include the "QAF VI Seller Net Payments" which shall be, with respect to the Hospital, an amount equal to the product of: (A) all payments received or to be received by the Hospital under the QAF VI Program\_in respect of QAF VI minus all payments already made, required to be made in the future or to be offset by the government with respect to the QAF VI Program, including-payments to the California Health Foundation & TrustQAF VI, multiplied by (B) the Closing Date Percentage. The "Closing Date Percentage" shall mean the percentage derived by dividing the total number of days between July 1, 2019 and the Effective Time, by 915;

(c) all Disproportionate Share Hospital Payments ("**DSH**") received on or after the Effective Time but calculated based on data from periods prior to the Effective Time (whether received before or after the Effective Time and whether paid to Sellers or Purchaser);

(d) all Sellers Plans (defined below) and the assets of all Sellers Plans and any asset that would revert to the employer upon the termination of any Sellers Plan, including, without limitation, any assets representing a surplus or overfunding of any Sellers Plan;

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 13 of 147

(e) all contracts that are not Assigned Contracts and all risk sharing agreements with independent physician associations ("**IPAs**");

(f) any Private Payor Agreement that is not a Transferred Private Payor Agreement and any Managed Care Agreement that is not a Transferred Managed Care Agreement;

(g) all collective bargaining agreements or other arrangements with unions representing Sellers' employees;

(h) all leases that have not been designated as Assigned Leases, and all Contracts that have not been designated as Assigned Contracts and all rents, deposits, prepayments, and similar amounts relating thereto;

(i) the portions of Inventory, Prepaids, and other assets disposed of, expended or canceled, as the case may be, by Sellers after the Signing Date and prior to the Effective Time in the ordinary course of business;

(j) assets owned by vendors of services or goods to the Hospital;

(k) all of Sellers' organizational or corporate record books, minute books and

tax records;

Except as set forth in Section 1.7(k) or 1.7(s) hereof (i.e., rights under (1)warranties and physical condition claims), all claims, counterclaims and causes of action of Sellers or Sellers' bankruptcy estate (including parties acting for or on behalf of Sellers' bankruptcy estate, including, but not limited to, the official committee of unsecured creditors appointed in the Bankruptcy Case), including, without limitation, (A) causes of action arising out of any claims and causes of action under chapter 5 of the Bankruptcy Code ("Avoidance Claims"), (B) any claims, counterclaims and causes of action under applicable non-bankruptcy law (including claims, counterclaims and causes of action against any health plan or other third party payors related to services provided prior to the Effective Time), and (C) any rights to challenge liens asserted against property of the Sellers' bankruptcy estate (including, but not limited to, liens attaching to the Purchase Price paid to the Sellers), and the proceeds from any of the foregoing; provided, however, that Purchaser shall acquire and be deemed to release and waive as of the Effective Time all Avoidance Claims against counterparties to Assigned Contracts and Assigned Leases solely to the extent such avoidance claims arise from, or are in connection with, executory contracts and unexpired leases assigned by the Sellers to Purchaser pursuant to Section 1.11 hereof;

(m) Except as set forth in <u>Section 1.7(s)</u>, all insurance policies and contracts and coverages obtained by Sellers or listing Sellers as insured party, a beneficiary or loss payee, including prepaid insurance premiums, and all rights to insurance proceeds under any of the foregoing, and all subrogation proceeds related to any insurance benefits arising from or relating to (i) Assets prior to the Licensure Date and/or (ii) Excluded Assets whether prior to or after the Licensure Date;

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 14 of 147

(n) all deposits made with any entity that provides utilities to the Hospital (the "Utility Deposits");

(o) all unclaimed property of any third party as of the Effective Time, including, without limitation, property which is subject to applicable escheat laws;

(p) all bank accounts of Sellers, other than the Lockboxes as set forth in <u>Section 1.7(o);</u>

(q) all writings and other items that are protected from discovery by the attorney-client privilege, the attorney work product doctrine or any other cognizable privilege or protection;

(r) the rights of Sellers to receive mail and other communications with respect to Excluded Assets or Excluded Liabilities;

(s) all tax refunds and tax assets of Sellers;

(t) all documents, records, operating manuals and film pertaining to the Hospital that the Parties agree that Sellers is required by law to retain;

(u) all patient records and medical records which are not part of any electronic medical record software transferred to Purchaser and are not required by law (including Section 351 of the Bankruptcy Code) to be maintained by Purchaser as of the Effective Time;

(v) any rights or documents relating to any Excluded Liability or other Excluded Asset;

(w) all deposits or other prepaid charges and expenses paid in connection with or relating to any other Excluded Assets and/or Excluded Liabilities;

(x) any rights or remedies provided to Sellers under this Agreement and each other document executed in connection with the Closing and the actions necessary to complete the sale of the Hospital pursuant to this Agreement;

(y) any rights or remedies, including deposits, against any individual or entity arising pursuant to (including in connection with Sellers' termination of) or relating to that certain Asset Purchase Agreement dated January 8, 2019 between, *inter alia*, Sellers and Strategic Global Management, Inc.;

(z) any (i) personnel files for employees of Sellers who are not hired by Purchaser; (ii) all documents, records, correspondence (including with respect to any employees), work papers, patient records or other books and records that Sellers is required by Law to retain; *provided, however*, that except as prohibited by Law and subject to <u>ARTICLE 5</u>, Purchaser shall have the right to make copies of any portions of such retained books and records that relate to the business of the Hospital as conducted before the Closing or that relate to any of the Assets; (iii) documents which Sellers is not permitted to transfer pursuant to any contractual obligation owed to any third party; (iv) documents primarily related to any Excluded Assets; and (v) documents

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 15 of 147

necessary to prepare tax returns (Purchaser shall be entitled to a copy of such documents). With respect to documents necessary to prepare cost reports, Purchaser shall receive the original document and Sellers shall be entitled to retain or request a copy of such documents;

(aa) any and all rights and entitlements of Sellers in respect of that certain Settlement Agreement, executed as of April 29, 2019, by and between, on the one hand, Premier, Inc., Premier Services, LLC ("**Premier GP**"), Premier Healthcare Alliance, L.P. ("**Premier LP**"), Premier Healthcare Solutions, Inc. ("**PHSI**") and each of Premier, Inc.'s other subsidiaries (collectively and including Premier GP, Premier LP and PHSI, "**Premier**"), and on the other hand, VHS, as approved by the Bankruptcy Court by order entered on May 29, 2019 [Docket No. 2461], including but not limited to the right to convert and exchange partnership interests arising under that certain Amended and Restated Limited Partnership Agreement, effective as of October 1, 2013, as amended, by and among Premier LP, Premier GP and the limited partners of Premier LP party thereto (including VHS); and

(bb) any assets identified in <u>Schedule 1.8(bb)</u>.

For the avoidance of doubt, Purchaser is not acquiring any asset owned by any affiliate of Sellers.

1.9 <u>Transferred Obligations</u>. Purchaser is not assuming any liabilities of Sellers. Instead, on and after the Closing Date, Purchaser shall be responsible for and agrees to discharge, perform and satisfy fully, on and after the Effective Time, the following liabilities and obligations (collectively, the "**Transferred Obligations**"):

(a) the Assigned Contracts, after Sellers pay the Cure Costs from the proceeds of the Purchase Price;

(b) the Assigned Leases, after Sellers pay the Cure Costs from the proceeds of the Purchase Price;

(c) all liabilities and obligations arising out of or relating to any act, omission, event or occurrence connected with the use, ownership or operation by Purchaser of the Hospital or any of the Assets on or after the Effective Time;

(d) all liabilities and obligations related to the Hired Employees arising on or following the Effective Time (which for avoidance of doubt, does not include any duties, obligations or liabilities arising from or related to employment-related documentation required to be maintained by such Seller prior to the Effective Time, including but not limited to, documentation of I-9 compliance for Sellers' employees and any alternative work schedule compliance duties, obligations or liabilities that relate to Sellers' employees);

(e) all unpaid real and personal property taxes, if any, that are attributable to the Assets after the Effective Time, subject to the prorations provided in <u>Section 1.6;</u>

(f) all liabilities and obligations relating to utilities being furnished to the Assets, subject to the prorations provided in <u>Section 1.6;</u>

(g) any documentary, sales and transfer tax liabilities of Sellers incurred as a result of the consummation of the transaction contemplated by this Agreement;

- (h) all liabilities or obligations provided for in <u>Section 5.3</u>; and
- (i) any other obligations and liabilities identified in <u>Schedule 1.9(i)</u>.

1.10 <u>Excluded Liabilities</u>. Purchaser shall have those duties, obligations and liabilities set forth in this Agreement, the IMA, the Leaseback Agreement, the Transition Services Agreement, the Bill of Sale, the Transfer Agreement and the Real Estate Assignment(s) and shall be responsible for the Transferred Obligations. However, except as expressly set forth herein, Purchaser is not assuming any liabilities of Sellers related to the Assets, Sellers' employees or the Hospitals, is purchasing the Assets free and clear of the Encumbrances except the Permitted Exceptions (defined herein) and shall not be deemed a successor to Sellers or their estates by reason of any theory of law or equity with respect to any claims or liens against Sellers or the Assets (the "Excluded Liabilities").

1.11 Designation of Assigned Contracts and Assigned Leases. Each Seller will assign to Purchaser the Assigned Leases and such other contracts and leases as are subject to evaluation by Purchaser for assumption or rejection (collectively "Evaluated Contracts"). At the later of (i) 5:00 p.m. pacific time on the day that is seven (7) calendar days prior to the date of the auction for the Assets and (ii) the date the Purchaser submits its bid for the Assets: (A) Purchaser shall notify Seller in writing signed and dated by Purchaser of which Evaluated Contracts are to be assigned to Purchaser (the "Assigned Contracts") and (B) Purchaser shall notify Seller in writing signed and dated by Purchaser of which Evaluated Contracts are to be rejected by Seller (the "Rejected Contracts"); provided, that Purchaser shall have the right to designate any contracts on the Assigned Contract list as a Rejected Contract on or prior to 5:00 p.m. pacific time on the day that is thirty (30) days prior to Closing Date, and Sellers shall have the absolute right to remove any Evaluated Contract from the list of Assigned Contracts in order to preserve avoidance claims; provided, however, that notwithstanding anything to the contrary, and to enable Sellers to comply with the terms of the Bidding Procedures Order with respect to UnitedHealthcare Insurance Company ("UnitedHealthcare"), Purchaser shall irrevocably designate all UnitedHealthcare agreements as Assigned Contracts or Rejected Contracts by the date that is two (2) calendars days after the Bankruptcy Court enters the Sale Order. The final list of Assigned Contracts at Closing will appear on Schedule 1.7(d).

# 1.12 Accounts Receivable Reconciliation.

(a) Sellers and Purchaser have mutually agreed that the target Accounts Receivable amount that will be transferred from Sellers to Purchaser as of the Effective Time is the A/R Target Amount. In connection with the Account Receivable reconciliation process, Purchaser shall collect the Accounts Receivable during the one hundred thirty-five (135) day period immediately following the Closing Date (collectively, the "Final A/R Collected"), and within the thirty (30) day period thereafter, Purchaser shall provide Sellers, in good faith, a schedule which provides an accounting of the Final A/R Collected (the "A/R Accounting Schedule"), together with reasonably detailed schedules and data supporting such accounting.

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 17 of 147

(b) After receipt of the A/R Accounting Schedule, Sellers shall have sixty (60) days to review the A/R Accounting Schedule and the Final A/R Collected as proposed by Purchaser, together with the work papers used in the preparation thereof, and have their representatives and advisors review such A/R Accounting Schedule and proposed Final A/R Collected. In connection with the review of the A/R Accounting Schedule and proposed Final A/R Collected, Purchaser shall give, and shall cause its representatives and advisors to give, to Sellers and their representatives and advisors reasonable access, upon reasonable prior notice, to the books, records and other materials and the personnel of, and work papers prepared by or for, Purchaser and its representatives and advisors, including to such historical financial information relating to the Sellers and the Accounts Receivable as Sellers or their representatives or advisors may request, in each case, in order to permit the timely and complete review of the A/R Accounting Schedule and proposed Final A/R Collected in accordance with this Section 1.12(b) and so long as such access does not unreasonably interfere with the operations of the Purchaser.

(c) If Sellers disagree with the A/R Accounting Schedule and/or proposed Final A/R Collected, they shall notify Purchaser in writing within sixty (60) days after the date on which Purchaser delivers such A/R Accounting Schedule and proposed Final A/R Collected to Sellers, which shall include the items as to which they disagree and their calculation of such disputed amounts with reasonable supporting detail (the "<u>Statement of Objections</u>"). Purchaser and Sellers shall reasonably cooperate to resolve any such disagreements. If Purchaser and Seller are unable to resolve all such disagreements on or before the date which is thirty (30) days following notification by Sellers of any such disagreements, such disagreements shall be submitted to the Bankruptcy Court for resolution. The A/R Accounting Schedule and Final A/R Collected shall become final on the earlier of (i) failure by Sellers to deliver a Statement of Objections within the time period required by this section; (ii) mutual written agreement by Sellers and Purchaser; or (iii) a determination by the Bankruptcy Court.

(d) Once the A/R Accounting Schedule and Final A/R Collected have become final in accordance with Section 1.12(c), the following shall occur:

(i) if the Final A/R Collected is more than the A/R Target Amount, then such excess amount shall be paid by Purchaser to Sellers within ten (10) business days of Purchaser's delivery of the accounting of the Final A/R Collected to Sellers, provided that, in no event shall amounts of any governmental grants or awards directly or indirectly related to COVID-19 be paid to Sellers under this Section 1.12(d)(i) (for the avoidance of doubt, any amounts of any governmental grants or awards directly related to COVID-19 must be removed from the calculation of the Final A/R Collected to determine whether Purchaser owes to Sellers any amount above the A/R Target Amount);

(ii) if the Final A/R Collected is less than the A/R Target Amount, then such deficit amount shall be paid by Sellers to Purchaser within ten (10) business days of Purchaser's delivery of the accounting of the Final A/R Collected to Sellers (and if not paid, Purchaser shall have collateral rights in the right to offset such amounts against Seller's QAF VI Seller Net Payments); or

(iii) if the Final A/R Collected is equal to the A/R Target Amount, then no adjusting payments in respect of the Accounts Receivable shall be required by either

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 18 of 147

Purchaser or Sellers.

(e) During the one hundred thirty-five (135) day period immediately following the Closing Date Purchaser shall (i) use good faith, commercially reasonable best efforts to collect the Accounts Receivable (including at least the efforts used by Purchaser to collect its other receivables) within the one hundred thirty-five (135) day period immediately following the Closing Date; (ii) not take any action for the purpose of or which would be reasonably likely to result in any of the Accounts Receivable not being collected in a timely manner and within the one hundred thirty-five (135) day period immediately following the Closing Date; and (iii) provide Sellers with weekly written updates on its collection of the Accounts Receivable.

# 1.13 Disclaimer of Warranties; Release.

THE ASSETS TRANSFERRED TO PURCHASER WILL BE SOLD BY (a) SELLERS AND PURCHASED BY PURCHASER IN THEIR PHYSICAL CONDITION AT THE EFFECTIVE TIME, "AS IS, WHERE IS AND WITH ALL FAULTS AND NONCOMPLIANCE WITH LAWS" WITH NO WARRANTIES, INCLUDING, WITHOUT LIMITATION, THE WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE, SUITABILITY, USAGE, WORKMANSHIP, QUALITY, PHYSICAL CONDITION, OR VALUE, AND ANY AND ALL SUCH OTHER REPRESENTATIONS AND WARRANTIES ARE HEREBY EXPRESSLY DISCLAIMED, AND WITH RESPECT TO THE LEASED REAL PROPERTY WITH NO WARRANTY OF HABITABILITY OR FITNESS FOR HABITATION, INCLUDING, WITHOUT LIMITATION, THE LAND, THE BUILDINGS AND THE IMPROVEMENTS. ALL OF THE PROPERTIES. ASSETS, RIGHTS, LICENSES, PERMITS, PRIVILEGES. LIABILITIES, AND OBLIGATIONS OF SELLERS INCLUDED IN THE ASSETS AND THE TRANSFERRED OBLIGATIONS ARE BEING ACQUIRED OR RECEIVED "AS IS, WHERE IS" ON THE CLOSING DATE AND IN THEIR PRESENT CONDITION, WITH ALL FAULTS. ALL OF THE TANGIBLE ASSETS SHALL BE FURTHER SUBJECT TO NORMAL WEAR AND TEAR AND NORMAL AND CUSTOMARY USE OF THE INVENTORY AND SUPPLIES IN THE ORDINARY COURSE OF BUSINESS UP TO THE EFFECTIVE TIME.

(b) Purchaser acknowledges that Purchaser has examined, reviewed and inspected all matters which in Purchaser's judgment bear upon the Purchase Price, the Assets, the Sellers, the Hospital, the businesses of the Hospital and their value and suitability for Purchaser's purposes and, except for Sellers representations and warranties in <u>ARTICLE 2</u>, is relying solely on Purchaser's own examination, review and inspection of the Assets and Transferred Obligations. Purchaser hereby releases Sellers and their affiliates from all responsibility and liability regarding the condition, valuation, salability or utility of the businesses of the Hospital or the Assets, or their suitability for any purpose whatsoever. Purchaser further acknowledges that the representations and warranties of Sellers contained in <u>ARTICLE 2</u> of this Agreement are the sole and exclusive representations and warranties made by Sellers to Purchaser (including with respect to the Hospital, the Assets and the Transferred Obligations) and shall expire, and be of no further force or effect at the Closing.

# ARTICLE 2 REPRESENTATIONS AND WARRANTIES OF SELLERS

Each Seller hereby represents, warrants and covenants to Purchaser, severally (and not jointly) with respect to such Seller that the following matters are true and correct as of the Signing Date, except as would not have a Material Adverse Effect (as defined below) upon the Assets, and except as disclosed in the disclosure schedule, as may be amended pursuant to the terms of this Agreement (the "**Disclosure Schedule**"), provided that the representations and warranties set forth in <u>Sections 2.1</u> (Authorization), <u>2.2</u> (Binding Agreement), <u>2.3</u> (Organization and Good Standing; No Violation), <u>2.7</u> (Compliance with Legal Requirements), <u>2.8</u> (Required Consents), <u>2.10</u> (Title) and <u>2.13</u> (Legal Proceedings) (the "**Sale Order Date Representations**") shall also be made as of immediately prior to the entry of the Sale Order (the "**Sale Order Date"**):

The term "**Material Adverse Effect**" means any event, change or occurrence that, individually or in the aggregate with other events, changes or occurrences, has had or would reasonably be expected to have, a material adverse effect on the Sellers' financial condition, operation or Assets in excess of Fifty Million Dollars (\$50,000,000); provided, that if such threshold is exceeded, then in lieu of Purchaser's right of termination, the Parties shall negotiate in good faith to reduce the Purchase Price as may be appropriate and any dispute will be resolved by the Bankruptcy Court as to the determination of any reasonable reduction provided, that in determining whether a Material Adverse Effect has occurred, any degradation in earnings and/or Receivables shall be excluded.

2.1 <u>Authorization</u>. Such Seller has all necessary corporate power and authority to enter into this Agreement and, subject to Bankruptcy Court approval, to carry out the transactions contemplated hereby.

2.2 <u>Binding Agreement</u>. This Agreement has been duly and validly executed and delivered by Sellers and, assuming due and valid execution by Purchaser, this Agreement constitutes a valid and binding obligation of Sellers enforceable in accordance with its terms subject to (a) applicable bankruptcy, reorganization, insolvency, moratorium and other laws affecting creditors' rights generally from time to time in effect (including, without limitation approval of the Bankruptcy Court) and (b) limitations on the enforcement of equitable remedies.

2.3 Organization and Good Standing; No Violation.

(a) Each of SFMC and VHS is a non-profit corporation duly organized, validly existing and in good standing under the laws of the State of California. Verity Holdings is a limited liability company duly organized, validly existing and in good standing under the laws of the State of California. Each Seller has all necessary power and authority to own, operate and lease its properties and to carry on its businesses as now conducted.

(b) Neither the execution and delivery by such Seller of this Agreement nor the consummation of the transactions contemplated hereby by such Seller nor compliance with any of the material provisions hereof by such Seller, will violate, conflict with or result in a

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 20 of 147

breach of any material provision of such Seller's articles of incorporation or bylaws or any other organizational documents of such Seller.

2.4 <u>Contracts</u>. Except as set forth in <u>Schedule 2.4</u>, upon entry of the Sale Order and payment of the Cure Costs, to Sellers' knowledge, Sellers are not in breach or default of the Assigned Contracts or Assigned Leases. No provision of this <u>Section 2.4</u> shall apply to any failure to obtain consents to the assignment of the Assigned Contracts and/or Assigned Leases from third parties in connection with the assignment of such Assigned Contracts and/or Assigned Leases to Purchaser (the "**Contract and Lease Consents**").

2.5 <u>Brokers and Finders</u>. Except as set forth on <u>Schedule 2.5</u>, neither such Seller nor any affiliate thereof, nor any officer or director thereof, have engaged or incurred any liability to any finder, broker or agent in connection with the transactions contemplated hereunder.

2.6 <u>Non-Contravention</u>. Neither the execution and delivery by Sellers of this Agreement and each Ancillary Agreement nor performance of any of the material provisions hereof by Sellers, will violate, conflict with or result in a breach of any material provisions of the articles of incorporation or bylaws of Sellers.

2.7 <u>Compliance with Legal Requirements</u>. Except as set forth in <u>Schedule 2.7</u>, to the knowledge of Sellers: each Seller, with respect to the operation of the Hospital, is in material compliance with all applicable laws, statutes, ordinances, orders, rules, regulations, licenses, certificates, judgments or decrees of all judicial or governmental authorities (federal, state, local, foreign or otherwise) (collectively, "Legal Requirements"). Except as set forth in <u>Schedule 2.7</u>, to the knowledge of Sellers, none of the Sellers, with respect to the operation of the Hospital, has been charged in writing with or been given written notice of any material violation or any obligation to take material remedial action under, any applicable Legal Requirements.

2.8 <u>Required Consents</u>. Except as set forth in <u>Schedule 2.8</u>, and other than in connection with any Licenses, any provider agreements (including any such agreements with a governmental authority) and the Attorney General, Sellers are not a party to or bound by, nor are any of the Assets subject to, any mortgage, or any material lien, deed of trust, material lease, or material contract or any material order, judgment or decree which, after giving effect to the Sale Order (a) will require the consent of any third party to the execution of this Agreement or (b) will require the consent of any third party to consummate the transactions contemplated by this Agreement.

# 2.9 Environmental Matters

(a) Sellers have provided Purchaser with the Phase I Environmental Site Assessments in Seller's electronic data room.

(b) Except as disclosed in <u>Schedule 2.9(b)</u>, to the knowledge of Sellers, the operations of the Hospital are not in material violation of any applicable limitations, restrictions, conditions, standards, prohibitions, requirements and obligations of Environmental Laws and related orders of any court or any other governmental authority.

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 21 of 147

(c) For the purposes of this Section, the term "Environmental Laws" shall mean all state, federal or local laws, ordinances, codes or regulations relating to Hazardous Substances or to the protection of the environment, including, without limitation, laws and regulations relating to the storage, treatment and disposal of medical and biological waste. For purposes of this Agreement, the term "Hazardous Substances" shall mean (i) any hazardous or toxic waste, substance, or material defined as such in (or for the purposes of) any Environmental Laws, (ii) asbestos-containing material, (iii) medical and biological waste, (iv) polychlorinated biphenyls, (v) petroleum products, including gasoline, fuel oil, crude oil and other various constituents of such products, and (vi) any other chemicals, materials or substances, exposure to which is prohibited, limited or regulated by any Environmental Laws.

2.10 <u>Title</u>. Sellers have delivered at their own expense for all Owned Real Property (i) preliminary title reports issued by Chicago Title Insurance Company (the "**Title Commitments**") and (ii) copies of or access to all material underlying title documents listed on the Title Commitments.

2.11 <u>Certain Other Representations with Respect to the Hospitals</u>.

(a) Except as set forth in <u>Schedule 2.11</u>, all Licenses which are material and necessary to the operation of the Hospital by SFMC are valid and in good standing and SFMC is in compliance with the terms and conditions of all such Licenses in all material respects, in each case except where the failure to be valid and in good standing or in compliance would not have a Material Adverse Effect on the Assets. Except as set forth in <u>Schedule 2.11</u>, as of the Closing Date, SFMC will have all material Licenses required under Legal Requirements to operate the Hospital as presently operated by SFMC, except where the failure to have any such License would not have a Material Adverse Effect on the Assets. To the knowledge of Sellers, no loss of any License is pending or threatened.

(b) SFMC is certified for participation in the Medicare, Medi-Cal and TRICARE programs and any other federal or state health care reimbursement programs in which SFMC participates, and has current and valid provider agreements with each such program, except where the failure to be so certified or have such provider agreements would not have a Material Adverse Effect.

(c) SFMC has not been excluded from Medicare, Medi-Cal or any federal or state health care reimbursement program, and, to the knowledge of Sellers, there is no pending or threatened exclusion action by a governmental authority against SFMC.

# 2.12 Financial Statements.

(a) Sellers have provided to Purchaser: (i) the unaudited balance sheets of the Sellers as of June 30, 2018; (ii) unaudited income statements of the Sellers for the twelve-month periods ended June 30, 2018; (iii) the audited consolidated income statements of Sellers for the years ended 2016 and 2017; and (iv) the unaudited consolidated balance sheet of Sellers as of June 30, 2018 (collectively, the "Historical Financial Statements").

(b) the income statements contained in the Historical Financial Statements

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 22 of 147

present, fairly in all material respects the results of the operations of the Sellers as of and for the periods covered therein and the balance sheets contained in the Historical Financial Statements (i) are true, complete and correct in all material respects; (ii) present, fairly in all material respects the financial condition of the Sellers as of the dates indicated thereon; and (iii) to the extent prepared by an independent certified public accounting firm, have been prepared in accordance with generally accepted accounting principles consistently applied throughout the periods covered, except as disclosed therein.

2.13 Legal Proceedings. Except as set forth on Schedule 2.13, and except for any and all cases and/or pleadings filed or to be filed in the Bankruptcy Court, which shall be available Sellers' claims noticing through and agent's website at http://www.kccllc.com/VERITYHEALTH/, to the knowledge of Sellers, there are no material claims, proceedings or investigations pending or threatened with respect to the ownership of the Assets or the operation of the Hospital or the Hospital by Sellers before any governmental authority. Except as set forth on <u>Schedule 2.13</u>, and other than any action or proceeding brought in the Bankruptcy Court, to the knowledge of Sellers, Sellers are not subject to any government order with respect to the ownership or operation by SFMC of the Hospital and SFMC is in substantial compliance with respect to any such government order.

2.14 <u>Employee Benefits</u>. Sellers have provided Purchaser with a list of (i) each pension, profit sharing, bonus, deferred compensation, or other retirement plan or arrangement of Seller with respect to the operation of the Hospital, whether oral or written, which constitutes an "employee pension benefit plan" as defined in Section 3(2) of ERISA, (ii) each medical, health, disability, insurance or other plan or arrangement of Seller with respect to the operation of the Hospital, whether oral or written, which constitutes an "employee welfare benefit plan" as defined in Section 3(1) of ERISA, and (iii) each other employee benefit or perquisite provided by Seller with respect to the operation of the Hospital, in which any employee of Seller participates in his/her capacity as such (collectively, the "Seller Plans").

2.15 <u>Personnel</u>. Sellers have provided Purchaser with a complete list (as of the date set forth therein) of names, positions and current annual salaries or wage rates and scheduled bonus, and the accrued paid time off pay of all employees of Sellers (including employees of the Hospital and employees of VHS and Verity Holdings) immediately prior to March 31, 2020, whether such employees are full time employees, part-time employees, on short-term or long-term disability or on leave of absence pursuant to Sellers's policies, the Family and Medical Leave Act of 1993 or other similar Legal Requirements (the "Hospital Employees") and indicating whether the Hospital Employee is full- time or part-time. Sellers shall provide an updated list to Purchaser no later than five (5) Business Days before the date scheduled for the Closing.

2.16 <u>Insurance</u>. Sellers have provided Purchaser with a list of all material insurance maintained by Sellers with respect to the Assets, as of the Signing Date.

2.17 <u>Receivables</u>. To the knowledge of Sellers, all Receivables included in the Assets at Closing result from the bona fide provision of products or services in the ordinary course of business. All proceeds of Sellers' Receivables are currently deposited, either electronically or manually, into those bank accounts provided to Purchaser in Seller's electronic data room.

## Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 23 of 147

2.18 <u>Payor Contracts</u>. To the knowledge of Sellers, and subject to Section 365 of the Bankruptcy Code, Sellers have provided Purchaser with a complete list of all written contracts with private third-party payors including insurance companies and HMOs. Sellers have provided Purchaser with a true and correct copy of all material Payor Contracts.

2.19 Excluded Individuals. Except as set forth on Schedule 2.19, to the knowledge of Sellers: neither Sellers, Hospital nor any director, officer or employee of Sellers or Hospital (a) was, is or is proposed to be, suspended, excluded from participation in, or sanctioned under, any federal or state health care program (including, without limitation, Medicare and Medicaid) (an "**Excluded Individual**"); (b) has been convicted of any criminal offense related to the delivery of any medical or health care services or supplies, or related to the neglect or abuse of patients; (c) has failed to maintain its current License to provide the services required to be provided by it to or on behalf of Sellers and Hospital; or (d) is unable to obtain or maintain liability insurance consistent with commercially reasonable industry practices.

2.20 <u>Seller Knowledge</u>. References in this Agreement to "Sellers' knowledge" or "the knowledge of Sellers" means the actual knowledge of the Chief Executive Officer and Chief Financial Officer of the applicable Seller, without independent research. No constructive or imputed knowledge shall be attributed to any such individual by virtue of any position held, relationship to any other Person or for any other reason.\_\_

# ARTICLE 3 REPRESENTATIONS AND WARRANTIES OF PURCHASER

As an inducement to Sellers to enter into this Agreement and to consummate the transactions contemplated by this Agreement, Purchaser hereby represents and warrants to Sellers as to the following matters as of the Signing Date and as of the Closing Date:

3.1 <u>Authorization</u>. Purchaser has full power and authority to enter into this Agreement and has full power and authority to perform its obligations hereunder and to carry out the transactions contemplated hereby. No additional internal consents are required in order for Purchaser to perform its obligations and agreements hereunder.

3.2 <u>Binding Agreement</u>. This Agreement has been duly and validly executed and delivered by Purchaser and, assuming due and valid execution by Sellers, this Agreement constitutes a valid and binding obligation of Purchaser enforceable in accordance with its terms subject to (a) applicable bankruptcy, reorganization, insolvency, moratorium and other laws affecting creditors' rights generally from time to time in effect, and (b) limitations on the enforcement of equitable remedies.

3.3 <u>Organization and Good Standing</u>. Purchaser is a corporation duly organized, validly existing and in good standing under the laws of the State of California, is or will be duly authorized to transact business in the State of California, and has full power and authority to own, operate and lease its properties and to carry on its business as now conducted.

3.4 <u>No Violation</u>. Except as set forth in <u>Schedule 3.4</u>, neither the execution and delivery by Purchaser of this Agreement nor the consummation of the transactions contemplated

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 24 of 147

hereby nor compliance with any of the material provisions hereof by Purchaser will (a) violate, conflict with or result in a breach of any material provision of the Articles of Incorporation, Bylaws or other organizational documents of Purchaser or any contract, lease or other instrument by which Purchaser is bound; (b) require any approval or consent of, or filing with, any governmental agency or authority, (c) violate any law, rule, regulation, or ordinance to which Purchaser is or may be subject, or (d) violate any judgment, order or decree of any court or other governmental agency or authority to which Purchaser is subject.

3.5 <u>Brokers and Finders</u>. Neither Purchaser nor any affiliate thereof nor any officer or director thereof has engaged any finder or broker in connection with the transactions contemplated hereunder.

3.6 <u>Representations of Sellers</u>. Purchaser acknowledges that it is purchasing the Assets on an "AS IS, WHERE IS" basis (as more particularly described in <u>Section 1.12</u>), and that Purchaser is not relying on any covenant, representation, warranty (expressed or implied, oral or otherwise) made by or on behalf of Sellers, or any other provision of this Agreement, in entering into and performing under this Agreement. Purchaser further acknowledges that Sellers are not making any covenants, representations or warranties herein relating to the Assets or the operation of the Hospital on or after the Effective Time.

3.7 <u>Legal Proceedings</u>. Except as described on <u>Schedule 3.7</u>, there are no claims, proceedings or investigations pending or, to the best knowledge of Purchaser, threatened relating to or affecting Purchaser or any affiliate of Purchaser before any court or governmental body (whether judicial, executive or administrative) in which an adverse determination would adversely affect Purchaser's ability to consummate the transactions contemplated hereby. Neither Purchaser nor any affiliate of Purchaser is subject to any judgment, order, decree or other governmental restriction specifically (as distinct from generically) applicable to Purchaser or any affiliate of Purchaser's ability to consummate the transactions contemplated hereby.

No Knowledge of Sellers' Breach. Neither Purchaser nor any of its affiliates has 3.8 knowledge of any breach of any covenant, representation or warranty by Sellers or of any condition or circumstance that would give Purchaser a right to terminate this Agreement pursuant to Section 9.1(c). If information comes to Purchaser's attention on or before the Closing Date (whether through Sellers or otherwise and whether before or after the Signing Date) which indicates that Sellers have breached any of their covenants, representations, warranties or any other provision or condition under this Agreement, then the effect shall be as if the covenants, representations and warranties or any other provision or condition of this Agreement had been modified in accordance with the actual state of facts existing prior to the Effective Time such that there will be no breach under Sellers' covenants, representations and warranties or any other provision or condition of this Agreement in relation to such information; provided, further, that Purchaser must immediately notify Sellers if any such breach comes to its attention on or before the Closing Date, and Purchaser's failure to so notify Sellers shall constitute a waiver by Purchaser of Sellers' breach, if any, of this Agreement or any ancillary agreements entered into pursuant to this Agreement. Upon written request of Sellers, Purchaser shall promptly confirm and remake this representation in writing.

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 25 of 147

3.9 <u>Ability to Perform.</u> Purchaser has the ability to obtain funds and at the Closing shall have cash in amounts necessary to consummate the transactions contemplated by this Agreement by means of cash, credit facilities or otherwise. Upon the earlier of ten (10) days after satisfaction of <u>Section 8.3</u> or five (5) days prior to the Closing Date, Purchaser shall have delivered to Sellers true, correct and complete copies of (i) executed commitment letter(s) from one or more banks or other lending institutions or sources (the "**Debt Commitment Letter**"), pursuant to which, and subject to the terms and conditions thereof, the lender parties thereto have committed to lend the amounts set forth therein to Purchaser for the purpose of funding the transactions contemplated by this Agreement and (ii) any executed equity commitment letter(s) (the "**Equity Commitment Letter**" and, together with the Debt Commitment Letter, the "**Financing Commitments**"). The Financing Commitments provide sufficient funds to consummate the transactions contemplated by this Agreement.

3.10 Investigation. Purchaser has been afforded reasonable access to, and has been provided adequate time to review, the books, records, information, operations, facilities and personnel of Sellers and the Hospital for purposes of conducting a due diligence investigation of Sellers and the Hospital. Purchaser has conducted a reasonable due diligence investigation of Sellers and the Hospital and has received satisfactory answers to all inquiries it has made respecting Sellers and the Hospital and has received all information it considers necessary to make an informed business evaluation of Sellers and the Hospital. In connection with its due diligence investigation of Sellers and the Hospital, purchaser has not relied upon any books, records, information, operations, facilities and personnel provided by Sellers, including in making its determination to enter into this Agreement and/or consummate the transactions contemplated hereby. Purchaser has completed all of its due diligence of Sellers and the Hospital by Purchaser.

3.11 <u>Purchaser Knowledge</u>. References in this Agreement to "Purchaser's knowledge" or "the knowledge of Purchaser" means the actual knowledge of the Chief Executive Officer and Chief Financial Officer of Purchaser, without independent research. No constructive or imputed knowledge shall be attributed to any such individual by virtue of any position held, relationship to any other Person or for any other reason.

# ARTICLE 4 COVENANTS OF SELLERS

# 4.1 Access and Information; Inspections.

(a) From the Signing Date through the Effective Time, (a) Sellers shall afford to the officers and agents of Purchaser (which shall include accountants, attorneys, bankers and other consultants and authorized agents of Purchaser) reasonable access during normal business hours at Sellers' corporate headquarters in Los Angeles, California to, and the right to inspect, the books, accounts, records and all other relevant documents and information with respect to the assets, liabilities and business of the Hospital and the plant and property of the Hospital at the Hospital and (b) Sellers shall furnish Purchaser with such additional financial and operating data and other information in Sellers' possession as to the businesses and properties of the Hospital as Purchaser or its representatives may from time to time reasonably request; *provided, however*,

## Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 26 of 147

that Sellers are not obligated to disclose information which is proprietary to Sellers and would not be essential to the ongoing operation of the Hospital by Purchaser; *provided*, *further*, that all disclosures of information shall be consistent with the confidentiality agreements and any other non-disclosure agreements entered into (or to be entered into) among Purchaser, its representatives and Sellers or their representatives. Purchaser's right of access and inspection shall be exercised in such a manner as not to interfere unreasonably with the operations of Sellers or the Hospital.

(b) Notwithstanding anything contained herein, Sellers shall not be required to provide Purchaser or its representatives or agents access to or disclose information where such access or disclosure would violate the rights of its patients, jeopardize the attorney-client or similar privilege with respect to such information or contravene any law, judgment, fiduciary duty or contract entered into prior to or on the date of this Agreement with respect to such information.

# 4.2 <u>Cooperation</u>.

(a) The Parties shall reasonably cooperate with each other and their respective authorized representatives and attorneys in: (a) all efforts to obtain all consents, approvals, authorizations, clearances and licenses required to carry out the transactions contemplated by this Agreement (including, without limitation, those of governmental and regulatory authorities), (b) the preparation of any document or other material which may be required by any governmental agency as a predicate to or result of the transactions contemplated in this Agreement, and (c) Purchaser's efforts to effectuate the assignment of Assigned Contracts to Purchaser as of the Closing Date.

(b) Except as may be otherwise requested by Sellers in order to comply with applicable law or regulatory guidance, notwithstanding anything contained herein, other than Bankruptcy Court orders and authorizations, it shall be Purchaser's sole responsibility (including payment of any fees, expenses, filing costs or other amounts) to obtain the Contract and Lease Consents, as well as all governmental consents, approvals, assignments, authorizations, clearances and licenses required to (x) carry out the transactions contemplated by this Agreement and/or (y) transfer any of the Assets, including any Licenses, Transferred Managed Care Agreements and/or Transferred Private Payor Agreements, not including the application required to be submitted to the California Attorney General and the transfer of the Medicare Provider Agreement and the Medi-Cal Provider Agreement, which are dealt with in Sections 5.95.8(c) and 4.5 hereto respectively. Upon request, Sellers shall provide Seller-specific information that may be needed by Purchaser to obtain such Contract and Lease Consents and such governmental consents, approvals, assignments, authorizations, clearances and licenses.

(c) Notwithstanding any provision to the contrary contained in this Agreement, Sellers shall not be obligated to obtain the approval or consent to the assignment, to Purchaser, of any Assigned Contracts or Assigned Leases, from any party to any of the Assigned Contracts or Assigned Leases even if any such contract or lease states that it is not assignable without such party's consent.

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 27 of 147

4.3 <u>Sellers' Efforts to Close</u>. Sellers shall use their reasonable commercial efforts to satisfy all of the conditions precedent set forth in <u>ARTICLE 7</u> and <u>ARTICLE 8</u> to their or Purchaser's obligations under this Agreement to the extent that Sellers' action or inaction can control or materially influence the satisfaction of such conditions; *provided, however*, that Sellers shall not be required to pay or commit to pay any amount to (or incur any obligation in favor of) any person (other than any filing or application fees to the extent Sellers are responsible therefor under this Agreement).

4.4 <u>Termination Cost Reports</u>. Sellers shall file all Medicare, Medi-Cal and any other termination cost reports required to be filed as a result of the consummation of (a) the transfer of the Assets to Purchaser and (b) the transactions contemplated by this Agreement, provided that Purchaser shall fund reasonable costs and expenses of preparation, filing and audit of such reports. Purchaser shall permit Sellers access to all Hospital books and records to prepare such reports and shall assist Sellers in the process of preparing, filing, and reviewing the termination cost reports. All such termination cost reports shall be filed by Sellers in a manner that is consistent with current laws, rules and regulations. Sellers shall be responsible for filing governmental cost reports for all periods through the day immediately preceding the Licensure Date. Purchaser shall be responsible for their own cost report filings relating to the Hospital beginning on the Licensure Date.

4.5 <u>Medicare and Medi-Cal Enrollment</u>. Sellers shall make commercially reasonable efforts to expeditiously enter into settlement agreements with the Centers for Medicare and Medicaid Services ("**CMS**"), with respect to the Medicare Provider Agreement, and the California Department of Health Care Services ("**DHCS**"), with respect to the Medi-Cal Provider Agreement or, alternatively, obtain Bankruptcy Court rulings and Final Orders that the Medicare Provider Agreement and/or the Medi-Cal Provider Agreement may be transferred without the consent of CMS or DHCS, as applicable, free and clear of Encumbrances and Excluded Liabilities and without successor liability, to enable such agreements to be assigned to Purchaser. Between the Closing Date and the Licensure Date, Purchaser may bill and collect for patient services under Seller's health plan agreements, pursuant to the terms of the IMA and Leaseback Agreement.

4.6 Transferred Private Payor Agreements. For purposes of this Section 4.6, the following terms shall have the meanings set forth herein: (a) "Payor" means the non-debtor counter-party to a Transferred Private Payor Agreement; (b) "Overpayment" means an overpayment made by a Payor to SFMC on account of a Contracted Payment; (c) "Contracted Pavment" means the contractual reimbursement due from a Payor to SFMC under a Transferred Private Payor Agreement for covered services rendered by Hospital under such agreement with a date of service on or prior to the Effective Time. Purchaser agrees that, as a condition to the assignment of any Transferred Private Payor Agreement on the Licensure Date, it shall honor any defenses to the payment of, and shall permit recoupment against, an Account Receivable due from Payor to Purchaser based on an Overpayment under such agreement, provided that, the determination and allowance of any Overpayment shall remain subject to the terms and conditions of the relevant Transferred Private Payor Agreement. Without limiting the scope of the foregoing sentence, Purchaser acknowledges and agrees that (a) the Sale Order shall authorize a Payor to continue to exercise its defenses to the payment to Purchaser of an Account Receivable based on an Overpayment, and (b) Overpayments do not constitute Cure Costs under

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 28 of 147

this Agreement. This <u>Section 4.6</u> shall satisfy the requirements under Sections 365(b) and (f) of the Bankruptcy Code. Following the Licensure Date, pursuant to Section 365(k) of the Bankruptcy Code, the SFMC shall be relieved and released from any obligation to any Payor on account of any Overpayment or otherwise.

4.7 <u>Hospital Operations</u>. From the date of the Sale Order until the Closing, Sellers shall, with respect to the operation of the Hospital, use commercially reasonable efforts (in each case subject to actions relating to and impacts arising from the SARS-CoV-2 virus or mutations therefrom or in connection with the disease COVID-19) to:

(a) without regard to negative financial impacts or any Material Adverse Effect, carry on Sellers' operation of the Hospital consistent with past practice, but subject to the Bankruptcy Cases and Sellers' obligations and actions in connection therewith;

(b) without regard to Material Adverse Effect, maintain in effect the insurance coverages with respect to the Assets;

(c) without regard to Material Adverse Effect, perform Sellers' material obligations under all Assigned Contracts with respect to the Assets in compliance with the Bankruptcy Code;

(d) with respect to material deficiencies, if any, cited by any governmental authority or accreditation body in the most recent surveys conducted by each, develop and implement a plan of correction (without regard to Material Adverse Effect) that is reasonably acceptable to such governmental authority or such accreditation body, but excluding any deficiencies cited (i) by the Attorney General or (ii) with respect to seismic or OSHPD-related requirements; *provided, however*, that Sellers may appeal any material deficiency citations under applicable laws if a plan of correction is not accepted by the applicable governmental authority or accreditation body which appeal shall satisfy Sellers obligations under this subsection;

(e) following entry of the Sale Order, permit and allow reasonable access by Purchaser and its representatives to make offers of post-Closing employment to any of Sellers' personnel and to establish relationships with physicians, medical staff and others having business relations with Sellers, provided, that any written materials shall be approved by Sellers prior to being sent, and provided further that such actions by Purchaser do not unreasonably interfere with Sellers' operation of the Hospital;

(f) timely file or cause to be filed all material reports, notices and tax returns required to be filed and pay all required taxes as they come due, unless such failure would not have a Material Adverse Effect; and

(g) without regard to Material Adverse Effect, maintain all existing material approvals, permits and environmental permits relating to the Hospital.

4.8 <u>Interim Collections</u>. Purchaser may bill and collect for patient services rendered by Hospital between the Closing Date and the Licensure Date under Seller's Private Payor

Agreements and Transferred Managed Care Agreements, pursuant to the terms of the IMA and Leaseback Agreement.

# 4.9 <u>Contract With Unions</u>. <del>Representatives</del>

Promptly following the Signing Date, representatives of Sellers who are parties to St. Francis related collective bargaining agreements and of the Purchaser, respectively, shall meet and confer from time to time as reasonably requested by either Party to discuss strategic business options and alternative approaches in negotiating each including terms contained under all operative collective bargaining agreementagreements. The applicable Sellers and Purchaser shall each participate in all union negotiations related to anythe potential modification and assignment of specific Seller's collective bargaining agreement. Promptly following the Signing Date, agreements to Purchaser. The applicable Sellers shall use commercially reasonable efforts to initiate discussions with Purchaser and unions and conduct discussions to renegotiate each collective bargaining agreement currently in effect with each applicable union. The applicable Sellers will not unreasonably withhold, condition or delay Bankruptcy Court approval or implementation of any successfully renegotiated collective bargaining agreement. The Parties recognize that Seller's failure to conclude a successor collective bargaining agreement shall not be a breach of Sellers' obligation under this Agreement, provided that if the unions refuse to negotiate, or otherwise are not timely, reasonable or realistic in negotiating modified or new collective bargaining agreements during the period between the Signing Date and the Closing Date, Sellers and Purchaser will jointly consider, and negotiate mutually in good faith, alternative approaches that may be availableand/or necessary to reduce Sellers' labor cost structure, including, but not limited to, seeking toreject the or otherwise excuse Purchaser's obligations under this Agreement.

(b) On or before the date that is thirty (30) days after the Sale Order Date, the negotiations pursuant to Section 4.9(a) shall have resulted in each, such labor unions, agreeing to either (i) either modification of the St. Francis related collective bargaining agreements under terms that are to be substantially consistent with the Purchaser's existing and most current collective bargaining agreements with each such respective labor union, and that settle all liabilities under the existing Seller collective bargaining agreements that shall be assigned to Purchaser, provided that there are shall be no cure obligations to the Sellers or (ii) enter into new collective bargaining agreement(s) agreements with each such respective labor union; provided, that if Purchaser and each labor union have not entered into such agreements described in (i) or (ii) above, or have entered into an agreement under (ii), then Sellers shall have the absolute right to file or take any other action to reject and terminate any such collective bargaining agreement and, in such event, the Bankruptcy Court shall have entered an order granting Sellers' requested rejection of such collective bargaining agreement prior to the Closing Date.

4.10 <u>Consulting Services</u>. Subject to compliance with applicable Legal Requirements and after entry of the Sale Order if the Purchaser is the Prevailing Bidder:

(a) Purchaser shall provide consulting services to the Hospital for the period prior to the Closing Date;

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 30 of 147

(b) the consulting services provided by Purchaser may include, but not be limited to, reviewing and advising SFMC regarding accounting and financial records, contracting, billing and collection activities, compliance with law, any of the plans or actions proposed in the following sentence, and other functions;

(c) Sellers shall consider and implement any mutually reasonably agreed upon plans and actions proposed by Purchaser to (a) assist the Hospital in meeting any seismic compliance deadlines, (b) stabilize and improve the operations of the Hospital or (c) develop and implement turnaround plans for the long-term viability of the Hospital; and

(d) Purchaser shall have reasonable access to the business office and records of the Hospital, which shall include reasonable access to the Hospital's chief executive officer, chief financial officer, chief operations officer or other equivalent personnel of the Hospital reasonably necessary for Purchaser to perform the aforementioned consulting services.

Notwithstanding the foregoing, VHS, SFMC and their boards of directors shall at all times retain ultimate control and governance over the assets and operation of the Hospital. Neither VHS nor SFMC shall delegate to Purchaser any of the powers, duties and responsibilities required to own or operate the Hospital or those that are retained by VHS and/or SFMC under law (including all certificates and licenses issued under authority of law for ownership or operation of the Hospital). For the period of time during which such consulting services are provided, Hospital and VHS shall have the non-exclusive, royalty-free, unlimited license to use any implemented Purchaser-owned intellectual property. Purchaser shall not charge Sellers any consulting fee for providing the services described in this <u>Section 4.11</u>.

4.11 <u>Cure Costs</u>. On or about the Closing Date, Sellers (from the proceeds of the Purchase Price) shall pay an amount equal to the Cure Costs to each counter party to an Assigned Contract and Assigned Lease so that each such Assigned Contract and Assigned Lease may be assumed by Sellers and assigned to Purchaser in accordance with the provisions of Section 365 of the Bankruptcy Code. For purposes of this Agreement, "**Cure Costs**", means all amounts that must be paid and all obligations that otherwise must be satisfied, including pursuant to Sections 365(b)(1)(A) and (B) of the Bankruptcy Code in connection with the assumption and/or assignment of the Assigned Contracts and Assigned Leases to Purchaser as provided herein.

# ARTICLE 5 COVENANTS OF PURCHASER

5.1 <u>Purchaser's Efforts to Close</u>. Purchaser shall use its reasonable commercial efforts to satisfy all of the conditions precedent set forth in <u>ARTICLE 7</u> and <u>ARTICLE 8</u> to its or Sellers' obligations under this Agreement to the extent that Purchaser's action or inaction can control or materially influence the satisfaction of such conditions.

5.2 <u>Consents</u>. Purchaser shall be entitled, but not obligated, to seek to obtain the Contract and Lease Consents. Purchaser shall be entitled, but not obligated, to solicit and seek to obtain estoppel certificates from any third party to any Leased Real Property. Purchaser's failure to obtain any or all of the Contract and Lease Consents or estoppel certificates as of the Closing

Date shall not be a condition precedent to either Party's obligation to consummate the Closing and perform all transactions contemplated by this Agreement.

# 5.3 <u>Certain Employee Matters</u>.

(a) Purchaser agrees to make offers of employment, effective as of the Effective Time, to substantially all persons (whether such persons are full time employees, part-time employees, on short-term or long-term disability or on leave of absence, military leave or workers compensation leave) who, immediately prior to the Effective Time are: (i) employees of SFMC; or (ii) employed by another Seller or affiliate and are listed on <u>Schedule 5.3</u> (collectively, the "**Hospital Employees**"). For the avoidance of doubt, the Hospital Employees shall not include any employees of VHS or any other affiliate of Sellers unless such individual is listed on <u>Schedule 5.3</u>. Any of the Hospital Employees who accept an offer of employment with Purchaser as of or after the Effective Time shall be referred to in this Agreement as the "**Hired Employees**." All employees who are Hired Employees shall cease to be employees of Sellers or its affiliates as of the Closing Date.

(b) After the Closing Date, Purchaser's human resources department will give reasonable assistance to Sellers and their affiliates with respect to Sellers' and Sellers' affiliates' post-Closing administration of Sellers' and Sellers' affiliates' pre-Closing employee benefit plans for the Hospital Employees. Within five (5) days after the Closing Date, Purchaser shall provide to Sellers a list of all the Hospital Employees who were offered employment by Purchaser but refused such employment along with a list of all Hired Employees (which such list Purchaser shall periodically update).

(c) With respect to any collective bargaining agreements or labor contract with respect to any union employees, Purchaser shall comply with the applicable laws, or to the extent applicable, Bankruptcy Court orders relating to collective bargaining agreements or labor contracts.

(d) The provisions of this <u>Section 5.3</u> are solely for the benefit of the Parties, and no employee or former employee or any other individual associated therewith or any employee benefit plan or trustee thereof shall be regarded for any purpose as a third party beneficiary of this Agreement, and nothing herein shall be construed as an amendment to any employee benefit plan for any purpose.

5.4 <u>Waiver of Bulk Sales Law Compliance</u>. Purchaser hereby waives compliance by Sellers with the requirements, if any, of Article 6 of the Uniform Commercial Code as in force in any state in which the Assets are located and all other laws applicable to bulk sales and transfers.

5.5 <u>Conduct Pending Closing</u>. Prior to consummation of the transactions contemplated hereby or the termination or expiration of this Agreement pursuant to its terms, unless Sellers shall otherwise consent in writing, Purchaser shall not take any action or fail or omit to take any action which would cause any of Purchaser's representations and warranties set forth in <u>ARTICLE 4</u> to be inaccurate or untrue during the term of this Agreement and as of the Closing.

## Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 32 of 147

5.6 <u>Resale Certificate</u>. Purchaser agrees to furnish to Sellers any resale certificate or certificates or other similar documents reasonably requested by Sellers to comply with or obtain an exemption from pertinent excise, sales and use tax laws.

5.7 <u>Operating Covenant</u>. Purchaser shall act in good faith in fulfilling its obligations under this Agreement.

# 5.8 <u>Governmental Approvals</u>.

(a) <u>Best Efforts.</u> Purchaser (a) shall use its best efforts to secure, as promptly as possible after the Signing Date, all consents, approvals (or exemptions therefrom), authorizations, clearances and licenses required to be obtained from governmental and regulatory authorities in order to carry out the transactions contemplated by this Agreement and to cause all of its covenants and agreements to be performed, satisfied and fulfilled, and (b) will provide such other information and communications to governmental and regulatory authorities as Sellers or such authorities may reasonably request. Purchaser is responsible for all filings with and requests to governmental authorities necessary to enable Purchaser to operate the Hospital at and after the Licensure Date. Purchaser acknowledges that Sellers may independently contact governmental and regulatory authorities as part of this process.

(b) <u>Change of Ownership Applications</u>. Purchaser shall, promptly, but no later than ten (10) business days after the entry of the Sale Order, or sooner if required by applicable governmental or regulatory authorities, file all applications, licensing packages and other documents with all applicable governmental and regulatory authorities which are necessary for the operation of a hospital and the consummation of the transactions hereunder, including the hospital license change of ownership application with CDPH, the hospital pharmacy change of ownership applications.

Attorney General Application. Purchaser and Sellers shall, promptly, but (c) no later than five (5) business days after the date of the Sale Order, file with the California Attorney General (the "Attorney General"), the application and report forms required for the transactions contemplated hereby and any supplemental information that may be reasonably requested in connection therewith pursuant to Title 11, Division 1, Chapter 15, Section 999.5 of the California Code of Regulations ("Section 999.5"), which application and report forms and supplemental information will comply in all material respects with the requirements of such regulations and shall state that Purchaser agrees to close the transactions contemplated by this Agreement so long as any conditions imposed by the Attorney General are not materially more burdensome than the conditions attached hereto as Exhibit 5.95.8(c) (the "Accepted **Conditions**"). Purchaser shall pay all fees (including, without limitation, attorneys' fees) required of Purchaser or Sellers with respect to the preparation and submission of the application, hearings, expert reports, Attorney General attorney review time, reports and other requirements of the Attorney General under Section 999.5. Each of Purchaser and Sellers shall furnish to the other such information and assistance as the other shall reasonably requires in connection with the preparation and submission to, or agency proceedings by, the Attorney General, and each of Purchaser and Sellers shall keep the other promptly apprised of any communications with, and inquires or requests for information from, the Attorney General. Purchaser shall take such action

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 33 of 147

as may be required by the Attorney General in order to resolve with the minimum practicable delay any objections the Attorney General may have to the transactions contemplated by this Agreement under Section 999.5. For any late submission of the Attorney General application, at Seller's election, either the Purchase Price shall be increased by an amount equal to One Thousand Dollars (\$1,000) per day or Sellers may terminate this Agreement.

HSR Filing. Purchaser and Sellers shall, promptly, but no later than five (d) (5) business days after the date of the Sale Order, file with the Federal Trade Commission and the Department of Justice the notification and report forms required for the transactions contemplated hereby and any supplemental information that may be reasonably requested in connection therewith pursuant to the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended (the "HSR Act"), which notification and report forms and supplemental information will comply in all material respects with the requirements of the HSR Act. Purchaser shall pay all filing fees required of Purchaser or Sellers with respect to the notification, report and other requirements of the HSR Act. Each of Purchaser and Sellers shall furnish to the other such information and assistance as the other shall reasonably requires in connection with the preparation and submission to, or agency proceedings by, any governmental authority under the HSR Act, and each of Purchaser and Sellers shall keep the other promptly apprised of any communications with, and inquires or requests for information from, such governmental Purchaser shall take such action (including divestitures or hold separate authorities. arrangements) as may be required by any governmental authority in order to resolve with the minimum practicable delay any objections such governmental authorities may have to the transactions contemplated by this Agreement under the HSR Act. For any late submission of the application under the HSR Act, at Seller's election, either the Purchase Price shall be increased by an amount equal to One Thousand Dollars (\$1,000) per day or Sellers may terminate this Agreement.

# ARTICLE 6 BANKRUPTCY COURT APPROVAL

# 6.1 Bankruptcy Court Approval.

Sellers and Purchaser acknowledge that this Agreement has been solicited (a) in conformity with the Bidding Procedures approved by the Bankruptcy Court on February 26, 2020 [Docket No. 4165] (the "Bid Procedures Order"), and that the sale of the Assets and the assignment of the Assigned Contracts and Assigned Leases remain subject to Bankruptcy Court approval. Purchaser further acknowledges that this Agreement and the transactions contemplated hereby are subject to Sellers' right and ability to consider higher or better competing bids with respect to the Assets subject to the terms of the Bid Procedures Order. Pursuant to the Bid Procedures Order, Purchaser shall, if its bid is determined to be the second highest bid, serve as a back-up bidder (the "Back-up Bidder") and, subject to the provisions of the Bid Procedures Order with respect to the retention of the Deposit, keep Purchaser's bid to consummate the transactions contemplated by this Agreement on the terms and conditions set forth in this Agreement (as the same may be modified with the written consent of Purchaser in any auction under the Bid Procedures Order) open and irrevocable until the earlier of (i) (a) 5:00 p.m. pacific time on the date which is thirty (30) days after the entry by the Bankruptcy Court of the Sale Order; or (b) 5:00 p.m. pacific time on May 31, 2020 (the "Outside Back-up Date"), or (ii) the

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 34 of 147

date of closing of an alternative transaction with the bidder who prevails at the auction (the "**Prevailing Bidder**"). Following the entry of the Sale Order and prior to the Outside Back-up Date, if the Prevailing Bidder breaches or fails to perform its obligations under the terms and conditions of its respective transaction documents and Sellers terminate such agreement with the Prevailing Bidder, the Back-up Bidder will be deemed to have the new prevailing bid, and the Bankruptcy Court order approving the sale to the Prevailing Bidder shall provide that Sellers will be authorized, without further order of the Bankruptcy Court, to consummate the transactions contemplated by this Agreement on the terms and conditions set forth in this Agreement (as the same may be modified with the written consent of Purchaser in the auction) with the Back-up Bidder.

(b) Purchaser further acknowledges that this Agreement is subject to termination, up to the time of entry of the Sale Order, except to the extent otherwise provided in the Bid Procedures Order, in the event Sellers in the reasonable exercise of their fiduciary duties determine that they have received a higher and better offer for Sellers' Assets and the Bankruptcy Court has authorized the Sellers to accept and implement the terms of such offer in accordance with the Bankruptcy Code.

Sellers shall, at any hearing to consider approval of this Agreement (the (c) "Sale Hearing"), exercise reasonable efforts to expeditiously obtain a Final "Sale Order" approving this Agreement subject to the reasonable exercise of their fiduciary duties to consider and accept a higher and better offer for Sellers' assets in accordance with the Bankruptcy Code and this Agreement. For purposes of this Agreement, the term "Sale Order" shall mean an order of the Bankruptcy Court in form and substance set forth in Exhibit [ ] and reasonably acceptable to Purchaser, authorizing the sale of the Assets pursuant to Section 363(b) of the Bankruptcy Code (including the Sellers' assumption and assignment to Purchaser of the Assigned Contracts and Assigned Leases pursuant to Section 365 of the Bankruptcy Code) on the terms and conditions set forth herein, free and clear of all Encumbrances (other than Permitted Exceptions) and Excluded Liabilities, including, for the avoidance of doubt, any successor liability, to the maximum extent permitted by the Bankruptcy Code.(d) For purposes of this Agreement, "Final Order" means an order, ruling, or judgment of the Bankruptcy Court or any other court of competent jurisdiction as to which the time to appeal, petition for certiorari, or move for reargument or rehearing has expired and as to which no appeal, petition for certiorari, or other proceedings for reargument or rehearing shall then be pending, or as to which any right to appeal, petition for certiorari, reargue, or rehear shall have been waived in writing, in form and substance satisfactory to Purchaser or, in the event that an appeal, writ of certiorari, or reargument or rehearing thereof has been sought, such order of the Bankruptcy Court of othercourt of competent jurisdiction shall have been determined by the highest court to which such order was appealed, or certiorari, reargument or rehearing shall have been denied and the time totake any further appeal, petition for certiorari, or move for reargument or rehearing shall have expired; provided, however, that the possibility that a motion under Rule 59 or Rule 60 of the Federal Rules of Civil Procedure, or any analogous Bankruptcy Rule or applicable state court rules of civil procedure, may be filed with respect to such order shall not cause such order not to be a Final Order. For the avoidance of doubt, in the event the Sale Order is not in form and substance reasonably acceptable to Purchaser, Purchaser may, at its sole election, terminate the transaction proposed hereby.

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 35 of 147

(d) (e)-Sellers agree, subject to the reasonable exercise of their fiduciary duties, to expeditiously seek a Bankruptcy Court determination that Purchaser is a good faith purchaser within the meaning of Section 363(m) of the Bankruptcy Code and in good faith to file such declarations and other evidence as may be required to support a determination.

(e) (f) Sellers shall seek expeditiously an order from the Bankruptcy Court retaining jurisdiction over all matters relating to claims against Sellers, whether or not arising in connection with this Agreement, solely in the Bankruptcy Court.

6.2 <u>Appeal of Sale Order</u>. In the event an appeal is taken or a stay pending appeal is requested from the Sale Order, Sellers shall immediately notify Purchaser of such appeal or stay request and shall provide to Purchaser promptly a copy of the related notice of appeal or order of stay. Sellers shall also provide Purchaser with written notice of any motion or application filed in connection with any appeal from either of such orders. In the event of an appeal of the Sale Order, Sellers shall be primarily responsible for drafting pleadings and attending hearings as necessary to defend against the appeal.

6.3 <u>Bidding Procedures</u>. Sellers and Purchaser shall comply with the terms of the Bid Procedures Order. Sellers shall sign this Agreement as and when permitted pursuant to the Bid Procedures Order and the Sale Order. Notwithstanding anything to the contrary herein or in the Bid Procedures Order, once approved the Sale Order shall not be materially modified absent the written consent of Purchaser, unless such modification has no substantive effect on Purchaser. In the event that the Sale Order is modified in contravention of this provision, Purchaser may, at its sole election, terminate the transaction proposed hereby.

(a) Provided that there is a Competing Bidder and a competitive sale of the Assets is conducted rather than a sale outright to Purchaser without a further sale process, the initial overbid over the Purchase Price shall be Twenty Million Dollars (\$20,000,000.00) and each subsequent bid shall be in increments of Ten Million Dollars (\$10,000,000.00). Any Competing Bidders must be a Qualified Bidder under the conditions set forth in the Bid Procedures without waiver thereof. Purchaser is irrevocably deemed to be a Qualified Bidder.

(b) The Sellers shall provide to Purchaser, immediately upon receipt/determination, each of the following: (i) notice of the determination of which others have been determined to be Qualified Bidders (as defined in the Bid Procedures Order), (ii) notice of which bidders have been determined to not have submitted Qualified Bids, and (iii) provide to Purchaser copies of all bids.

# ARTICLE 7 CONDITIONS PRECEDENT TO OBLIGATIONS OF SELLERS

Sellers' obligation to sell the Assets and to close the transactions as contemplated by this Agreement shall be subject to the satisfaction of each of the following conditions on or prior to the Closing Date unless specifically waived in writing by Sellers in whole or in part at or prior to the Closing:

## Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 36 of 147

7.1 <u>Signing and Delivery of Instruments</u>. Purchaser shall have executed and delivered all documents, instruments and certificates required to be executed and delivered pursuant to the provisions of this Agreement.

7.2 <u>No Restraints</u>. No temporary restraining order, preliminary or permanent injunction or other order preventing the consummation of the transactions contemplated in this Agreement shall have been issued by any court of competent jurisdiction or any other governmental body and remain in effect on the Closing Date.

7.3 <u>Governmental Submissions</u>. Purchaser shall have submitted the hospital license change of ownership application to CDPH and the hospital pharmacy change of ownership application to the BOP.

7.4 <u>Attorney General Approval</u>. The Attorney General shall have approved the transactions contemplated by this Agreement, or the Debtors shall have obtained a Bankruptcy Court order or orders providing for the transfer of the Debtors' assets on conditions which are not materially more burdensome than those forth in <u>Exhibit 5.95.8(c)</u>.

7.5 <u>Bankruptcy Court Approval</u>. The Bankruptcy Court shall have entered the Sale Order.

7.6 <u>HSR Act</u>. The applicable waiting period under the HSR Act shall have expired or been earlier terminated.

# ARTICLE 8 CONDITIONS PRECEDENT TO OBLIGATIONS OF PURCHASER

Purchaser's obligation to purchase the Assets and to close the transactions contemplated by this Agreement shall be subject to the satisfaction of each of the following conditions on or prior to the Closing Date unless specifically waived in writing by Purchaser in whole or in part at or prior to the Closing.

8.1 <u>Signing and Delivery of Instruments</u>. Sellers shall have executed and delivered all documents, instruments and certificates required to be executed and delivered pursuant to the provisions of this Agreement.

8.2 <u>No Restraints</u>. No temporary restraining order, preliminary or permanent injunction or other order preventing the consummation of the transactions contemplated in this Agreement shall have been issued by any court of competent jurisdiction or any other governmental body and remain in effect on the Closing Date.

8.3 <u>Attorney General Approval</u>. The Attorney General shall have approved the transactions contemplated by this Agreement, or the Debtors shall have obtained a Final-OrderBankruptcy Court order or orders providing for the transfer of the Debtors' assets on conditions which are not materially more burdensome than those forth in Exhibit 5.95.8(c). In the event the Attorney General imposes conditions on the transactions contemplated by this Agreement, or on Purchaser in connection therewith which are materially more burdensome than the Purchaser Approved Conditions set forth on Exhibit 5.95.8(c) (the "Additional

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 37 of 147

**Conditions**"), Sellers shall file a motion with the Bankruptcy Court seeking the entry of an order ("Supplemental Sale Order") finding that the Additional Conditions are an "interest in property" for purposes of 11 U.S.C. § 363(f), and that the Assets can be sold free and clear of the Additional Conditions without the imposition of any other conditions, which would adversely affect the Purchaser, and such Order shall have become Final. For purposes of this Section 8.3, Additional Conditions which individually or collectively impose a direct or indirect cost to Purchaser of Five Million Dollars (\$5,000,000), or more, shall be conclusively deemed to be "materially more burdensome." If Sellers fail to obtain such Final Supplemental Sale Order within sixty (60) days of the Attorney General's imposition of Additional Conditions, Purchaser (at its sole discretion) shall be entitled to terminate this Agreement and receive the return of its Deposit or elect to extend such sixty (60) day period. Upon the entry of such an order becominga Final Orderfrom the Bankruptcy Court, and so long as such order remains in full force and effectis not stayed, Purchaser shall be required to consummate the transactions contemplated by this Agreement. If Sellers do not obtain such an order, or such order does not relieve the Additional Conditions to the satisfaction of Purchaser or such order does not become Final within the applicable time period, Purchaser shall not be required to consummate the transactions contemplated by this Agreement.

8.4 <u>Bankruptcy Court Approval</u>. The Bankruptcy Court shall have entered the Sale Order and made a finding that Purchaser is a "good faith" purchaser, and such order shall be infull force and effect and not subject to any stay requested or obtained within the period in whichsuch stay may be so requested or obtained.

8.5 <u>HSR Act</u>. The applicable waiting period under the HSR Act shall have expired or been earlier terminated.

Medicare and Medi-Cal Provider Agreement. 8.6 Following consultation with Purchaser, Sellers shall have obtained agreements with CMS and DHCS or a Final Order by a court having proper jurisdiction shall have been enteredan order from the Bankruptcy Court that has not been stayed with respect to the transfer of the Medicare Provider Agreement and/or the Medi-Cal Provider Agreement, such that (a) all liabilities, obligations and Encumbrances under the Medicare/Medi-Cal Agreements are fully satisfied, discharged, and released with regard to any claims under the Medicare/Medi-Cal Agreements, whether known or unknown, that CMS or DHCS has against the Sellers or Purchaser for liabilities and obligations arising under the Medicare/Medi-Cal Agreements before the Effective Time, and (b) the Medicare/Medi-Cal Agreements will be transferred to Purchaser as of the Effective Time free and clear of such pre-Closing liabilities, obligations and Encumbrances; provided, however, that Purchaser acknowledges and agrees that it may be treated by CMS and DHCS as the successor to the quality history associated with the relevant Medicare/Medi-Cal Agreements assigned and, for purposes of survey and certification issues associated with such quality history, Purchaser may be treated as if it is the relevant Seller and no change of ownership occurred.

8.7 <u>Collective Bargaining Agreements and Labor Contracts</u>. Purchaser agrees to close the transactions contemplated by this Agreement, so long as (a) after good faith negotiations with the applicable St. Francis labor unions, such labor unions agree to either (i) modify the St. Francis collective bargaining agreements to be substantially consistent with the Purchaser's existing new collective bargaining agreements with each such respective labor union, and settle-

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 38 of 147

all liabilities under the existing agreements, or (ii) enter into new collective bargaining agreements that are substantially consistent with the Purchaser's existing collective bargaining agreements with each such respective labor union, and settle all liabilities under the existing agreements, or (b) the Bankruptey Court takes action to reject any such collective bargaining agreement that is not substantially consistent or where the union has not agreed to settle all liabilities under the agreement. In the event Purchaser and the respective St. Francis labor union are unable to agree to a modified or new collective bargaining agreement with the Purchaser's existing collective bargaining agreement with the Purchaser's existing collective bargaining agreement with the Bankruptey Court does not reject such collective bargaining agreement pursuant to a final order, then Purchaser shall not be required to consummate the transactions contemplated by this Agreement. Sellers shall have satisfied, in all material respects, their obligations set forth in Section 4.9(b).

## ARTICLE 9 TERMINATION

## 9.1 <u>Termination</u>. This Agreement may be terminated at any time prior to Closing:

(a) by the mutual written consent of the Parties;

(b) by Sellers if a material breach of this Agreement has been committed by Purchaser and such breach has not been (i) waived in writing by Sellers or (ii) cured by Purchaser to the reasonable satisfaction of Sellers within fifteen (15) business days after Sellers provide Purchaser of a written notice which describes the nature of such breach and acceptable cure actions; *provided*, *however*, Sellers shall not be permitted to terminate this Agreement pursuant to this Section 9.1(b) if Sellers are also in material breach of this Agreement;

(c) by Purchaser if Purchaser has complied with <u>Section 3.8</u> and a material breach of this Agreement has been committed by Sellers, which material breach has resulted, or would more likely than not result, in a Material Adverse Effect on the Assets taken as a whole, and such breach has not been (i) waived in writing by Purchaser or (ii) cured by Sellers to the reasonable satisfaction of Purchaser within fifteen (15) business days after Purchaser provides Sellers of a written notice which describes the nature of such breach and acceptable cure actions; *provided, however*, Purchaser shall not be permitted to terminate this Agreement pursuant to this <u>Section 9.1(c)</u> if Purchaser is also in material breach of this Agreement;

(d) by Purchaser if satisfaction of any condition in <u>ARTICLE 8</u> is or becomes impossible and Purchaser has not waived such condition in writing (provided that the failure to satisfy the applicable condition or conditions has occurred by reason other than (i) through the failure of Purchaser to comply with its obligations under this Agreement or (ii) Sellers' failure to provide their closing deliveries on the Closing Date as a result of Purchaser not being ready, willing and able to close the transaction on the Closing Date);

(e) by Sellers if satisfaction of any such condition in <u>ARTICLE 7</u> is or becomes impossible and Sellers have not waived such condition in writing (provided that the failure to satisfy the applicable condition or conditions has occurred by reason other than (i)

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 39 of 147

through the failure of Sellers to comply with their obligations under this Agreement or (ii) Purchaser's failure to provide its closing deliveries on the Closing Date as a result of Sellers not being ready, willing and able to close the transaction on the Closing Date);

(f) by either Purchaser or Sellers if the Bankruptcy Court enters an order dismissing the Bankruptcy Case prior to the sale closing or fails to approve the sale of the Assets to Purchaser;

(g) by Sellers in the event Sellers in the exercise of their fiduciary duties determine that they have received a higher and better offer for Sellers' assets and the Bankruptcy Court has authorized the Sellers to accept and implement the terms of such offer in accordance with the Bankruptcy Code;

(h) by Purchaser in the event that its Back-Up Bidder status has expired in accordance with the Bid Procedures Order and/or the terms of this Agreement; and

(i) by either Purchaser or Sellers if the Closing has not occurred (other than through the failure of any Party seeking to terminate this Agreement to comply fully with its obligations under this Agreement) on or before September 1, 2020 (the "**Termination Date**"); <u>provided</u>, that the Termination Date shall be extended in the event the conditions set forth in Section 7.4 and Section 8.3 have not been satisfied (but the conditions to Closing in Article 7 and Article 8 have otherwise been satisfied other than such conditions that are to be satisfied by payments and deliveries to be made at the Closing) to the earlier of (A) ten (10) business days after the satisfaction of the conditions set forth in Section 7.4 and Section 8.3 or (B) December 31, 2020.

9.2 <u>Termination Consequences</u>. If this Agreement is terminated pursuant to <u>Section</u> 9.1: (a) all further obligations of the Parties under this Agreement shall terminate, other than Purchaser's right to receive a return on the Deposit in accordance with <u>Section 1.2</u> and a break-up fee in accordance with the Bidding Procedures, and provided that the provisions of <u>ARTICLE 11</u> shall survive, and (b) each Party shall pay the costs and expenses incurred by it in connection with this Agreement; *provided, however*, that in the case of any termination based on <u>Section 9.1(b)</u> or <u>Section 9.1(c)</u>, the consequences of such termination shall be determined in accordance with <u>ARTICLE 11</u> hereof. Each Party acknowledges that the agreements contained in this <u>Section 9.2</u> are an integral part of the transactions contemplated by this Agreement, that without these agreements such Party would not have entered into this Agreement.

9.3 <u>Additional Termination Consequences</u>. <u>In the event that Purchaser is not the</u> winning bidder as defined in the Bidding Procedures order, Purchaser shall be paid the Stalking Horse Bidder Protections in accordance with the Bidding Procedures Order.

(a) In the event that (i) a bid from a purchaser other than Purchaser for all or any part of the Assets in accordance with the Bid Procedures Order (a "**Competing Bid**") is consummated, or (ii) this Agreement is terminated for any reason other than a material breach of Purchaser, in consideration for the Purchaser having expended considerable time and expense in connection with this Agreement and the negotiation thereof and the identification and quantification of assets of Sellers, and without the requirement of any notice or demand from Purchaser or any other application to or order of the Bankruptcy Court, (A) the Deposit shall be returned to Purchaser in accordance with the Bid Procedures Order (if not previously returnedpursuant to the Bid Procedures Order because Purchaser is neither the Winning Bidder (asdefined in the Bid Procedures Order) nor the Back-Up Bidder (as defined in the Bid Procedures-Order)) and (B) Sellers shall jointly and severally, pay (or cause to be paid to) Purchaser, inaccordance with the terms hereof, a termination payment solely in the event of a successful-Competing Bid in an amount equal to two and one-half percent (2.5%) of the Purchase Price (the-"Stalking Horse Bidder Protections"). Sellers shall pay such Stalking Horse Bidder Protectionsin immediately available funds in the event of termination pursuant to <u>Section 9.3(a)(ii)</u>, or uponconsummation of a Competing Bid pursuant to Section 9.3(a)(i), to such account or accounts as may be specified in writing by Purchaser. The Stalking Horse Bidder Protections shall constitute an allowed administrative expense claim of Sellers' estates under sections 503(b) and 507 of the Bankruptcy Code and notwithstanding the prior entry of any order of the Bankruptcy Court relating to the use of cash collateral, be paid on the second (2nd) business day following the dateof consummation of a Competing Bid or termination of this Agreement under Section 9.1(a), (c), (d), (f), (g), (h) or (i) if no material breach by Purchaser of this Agreement has occurred. The Sale Order shall provide, although not required for the enforceability hereof, for payment by Sellers of the Stalking Horse Bidder Protections as and when such amounts are due and payable hereunder. Nothing in this <u>Section 9.3(a)</u> shall relieve Purchaser or Sellers of any liability for a breach of this Agreement prior to the date of termination.

(b) Each of the Parties acknowledges and agrees that the agreements contained in this Section 9.3 are an integral part of the transactions contemplated by this Agreement and that, without these agreements, the other Parties would not enter into this Agreement. Each of the Parties further acknowledges that the payment by Sellers of the Stalking Horse Bidder-Protections is not a penalty, but rather liquidated damages in a reasonable amount that willcompensate Purchaser, in the circumstances in which such Stalking Horse Bidder Protection ispayable, for the efforts and resources expended and the opportunities foregone while negotiatingthis Agreement and in reliance on this Agreement and on the expectation of the consummation of the transactions contemplated by this Agreement, which amount would otherwise be impossibleto calculate with precision. The obligation to pay the Stalking Horse Bidder Protections inaccordance with the provisions of this Agreement will (i) be binding upon and enforceable jointlyand severally against each Seller and (ii) survive the subsequent termination of this Agreement, solely to the extent permitted by applicable law. The obligation to pay the Stalking Horse Bidder-Protections as and when required under this Agreement, are intended to be, and are, binding upon-(A) each Seller, (B) any successors or assigns of any Seller and (C) any trustee, examiner or otherrepresentative of a Seller's estate (each of (A) through (C), a "Successor") as if such Successor were a Seller hereunder.

## ARTICLE 10 POST-CLOSING MATTERS

10.1 <u>Excluded Assets</u>. Subject to <u>Section 10.2</u> hereof, any Excluded Asset (or proceeds thereof) (a) pursuant to the terms of this Agreement, (b) as otherwise determined by the Parties' mutual written agreement or (c) absent such agreement, as determined by adjudication by the Bankruptcy Court, which comes into the possession, custody or control of Purchaser (or its respective successors-in-interest, assigns or affiliates) shall, within <u>thirtyten</u> (3010) business days

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 41 of 147

following receipt, be transferred, assigned or conveyed by Purchaser (and its respective successors-in-interest, assigns and affiliates) to Sellers without imposing any charge to Sellers for Purchaser's transfer, storage, handling or holding of same on and after the Effective Time. Purchaser (and its respective successors-in-interest, assigns and affiliates) shall have neither the right to offset amounts payable to Sellers under this <u>Section 10.1</u> against, nor the right to contest its obligation to transfer, assign and convey to Sellers because of, outstanding claims, liabilities or obligations asserted by Purchaser against Sellers. If Purchaser does not remit any monies included in the Excluded Assets (or proceeds thereof) to Sellers in accordance with the first sentence of this <u>Section 10.1</u>, such withheld funds shall bear interest at the Prime Rate in effect on the calendar day upon which such payment was required to be made to Sellers (the "**Excluded Asset Due Date**") plus five percent (5%) (or the maximum rate allowed by law, whichever is less), such interest accruing on each calendar day after the Excluded Asset Due Date until payment of the Excluded Assets and all interest thereon is made to Sellers.

### 10.2 Preservation and Access to Records After the Closing.

From the Licensure Date until seven (7) years after the Licensure Date or (a) such longer period as required by law (the "Document Retention Period"), Purchaser shall keep and preserve all medical records, patient records, medical staff records and other books and records which are among the Assets as of the Effective Time, but excluding any records which are among the Excluded Assets. Purchaser will afford to the representatives of Sellers, any of their affiliates, the Official Committee of the Unsecured Creditors of the Sellers, Sellers' estate representative or any liquidating trustee of the Sellers' bankruptcy estate ("Seller Parties"), including their counsel and accountants, full and complete access to, and copies (including, without limitation, color laser copies) of, such records with respect to time periods prior to the Licensure Date (including, without limitation, access to records of patients treated at the Hospital prior to the Licensure Date) during normal business hours after the Licensure Date, to the extent reasonably needed by any Seller Party for any lawful purpose. Purchaser acknowledges that, as a result of entering into this Agreement and operating the Hospital, it will gain access to patient records and other information which are subject to rules and regulations concerning Purchaser shall abide by any such rules and regulations relating to the confidentiality. confidential information it acquires. Purchaser shall maintain the patient and medical staff records at the Hospital in accordance with applicable law and the requirements of relevant insurance carriers. After the expiration of the Document Retention Period, if Purchaser intends to destroy or otherwise dispose of any of the documents described in this Section 10.2(a), Purchaser shall provide written notice to Sellers of Purchaser's intention no later than forty-five (45) calendar days prior to the date of such intended destruction or disposal. Any of the Seller Parties shall have the right, at its sole cost, to take possession of such documents during such forty-five (45) calendar day period. If any of the Seller Parties do not take possession of such documents during such forty-five (45) calendar day period, Purchaser shall be free to destroy or otherwise dispose of such documentation upon the expiration of such forty-five (45) calendar day period.

(b) Provided that Purchaser shall not incur any out of pocket costs, Purchaser shall give full cooperation to the Seller Parties and their insurance carriers in connection with the administration of Sellers' estate, including, without limitation, in connection with all claims, actions, causes of action or audits relating to the Excluded Assets, Excluded Liabilities or

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 42 of 147

pre-Closing operation of the Sellers or the Hospital that any Seller Party may elect to pursue, dispute or defend, in respect of events occurring prior to the Licensure Date with respect to the operation of the Hospital. Such cooperation shall include, without limitation, making the Hired Employees available for interviews, depositions, hearings and trials and other assistance in connection with the administration of Sellers' estate and such cooperation shall also include making all of its employees available to assist in the securing and giving of evidence and in obtaining the presence and cooperation of witnesses (all of which shall be done without payment of any fees or expenses to Purchaser or to such employees. In addition, Sellers and Sellers' affiliates shall be entitled to remove from the Hospital originals of any such records, but only for purposes of pending litigation involving the persons to whom such records refer, as certified in writing prior to removal by counsel retained by Sellers or any of Sellers' affiliates in connection with such litigation. Any records so removed from the Hospital shall be promptly returned to Purchaser following Sellers' or its applicable affiliate's use of such records.

(c) In connection with (i) the transition of the Hospital pursuant to the transaction contemplated by this Agreement, (ii) Sellers' rights to the Excluded Assets, (iii) any claim, audit, or proceeding, including, without limitation, any tax claim, audit, or proceeding and (iv) the Sellers' obligations under the Excluded Liabilities, Purchaser shall after the Effective Time give Sellers access during normal business hours to Purchaser's books, personnel, accounts and records and all other relevant documents and information with respect to the assets, liabilities and business of the Hospital as representatives of Sellers and Sellers' affiliates may from time to time reasonably request, all in such manner as not to unreasonably interfere with the operations of the Hospital.

(d) Purchaser and its representatives shall be given access by Sellers during normal business hours to the extent reasonably needed by Purchaser for business purposes to all documents, records, correspondence, work papers and other documents retained by Sellers pertaining to any of the Assets prior to the Effective Time (excluding confidential employee information, privileged materials and patient records), all in such manner as to not interfere unreasonably with Sellers. Such documents and other materials shall be, at Sellers' option, either (i) copied by Sellers for Purchaser at Purchaser's expense, or (ii) removed by Purchaser from the premises, copied by Purchaser and promptly returned to Sellers.

(e) Purchaser shall comply with, and be solely responsible for, all obligations under the Standards for Privacy of Individually Identifiable Health Information (45 CFR Parts 160 and 164) promulgated pursuant to the Health Insurance Portability and Accountability Act of 1996 and any and all corresponding California state law requirements with respect to the operation of the Hospital on and after the Licensure Date.

(f) Purchaser shall cooperate with Sellers, on a timely basis and as reasonably requested by Sellers, in connection with the provision of all data of the Hospital and other information required by Sellers for reporting purposes, including without limitation QAF or DSH reporting.

(g) To the maximum extent permitted by law, if any Person requests or demands, by subpoena or otherwise, any documents relating to the Excluded Liabilities or Excluded Assets, including without limitation, documents relating to the operations of any of the

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 43 of 147

Hospital or any of the Hospital's committees prior to the Licensure Date, prior to any disclosure of such documents, Purchaser shall notify Sellers and shall provide Sellers with the opportunity to object to, and otherwise coordinate with respect to, such request or demand.

10.3 Provision of Benefits of Certain Contracts. Notwithstanding anything contained herein to the contrary, this Agreement shall not constitute an agreement to assign any Assigned Contract or Assigned Lease, if, notwithstanding the provisions of Sections 363 and 365 of the Bankruptcy Code, an attempted assignment thereof, without the consent of the third party thereto, would constitute a breach thereof or in any way negatively affect the rights of Sellers or Purchaser, as the assignee of such Assigned Contract or Assigned Lease, as the case may be, thereunder. If, notwithstanding the provisions of Sections 363 and 365 of the Bankruptcy Code, such consent or approval is required but not obtained, Sellers will cooperate with Purchaser in any reasonable arrangement designed to both (a) provide Purchaser with the benefits of or under any such Assigned Contract or Assigned Lease. Further, notwithstanding anything contained in this Agreement to the contrary, this Agreement shall not constitute an agreement to assign any Account Receivable the assignment of which is either prohibited by law or by the terms of any contract with a payor without the consent of such payor.

General Cooperation and Turnover Obligations. The Parties shall cooperate to 10.4 ensure that any and all payments that constitute "Excluded Assets" shall be paid to and received by Sellers, with any payments that constitute "Assets" transferred to Purchaser pursuant to Section 1.7 or that otherwise arise from services rendered by Purchaser on or after the Licensure Date be paid to and received by Purchaser. In this regard, for a period of two (2) years after the Effective Date ("Turnover Period"), the Parties shall, within ten (10) business days of receipt, copy and send to the other Party copies (either in hard copy or via electronic file) of all remittance advices for all deposits to all Lockboxes or other bank accounts for the Receivables, from whatever payor or source of funds, that are received on and after the Effective Time. In the event that payments that constitute a transferred Asset are deposited to a bank account of Sellers which is not automatically swept or transferred to Purchaser, then Sellers, within ten (10) days of notice of the receipt of such payments shall turnover and pay Purchaser said funds. In the event that a deposit representing payment of any Excluded Assets is received by Purchaser, then Purchaser, within ten (10) days of notice of the receipt of funds representing any Excluded Assets, shall turnover and pay Sellers such funds. Each Party shall have the right, within three (3) months after the expiration of the Turnover Period, to audit by an independent and competent auditor, at the requesting Party's sole expense, of the bank records and remittance advices of the Thereafter, upon the findings of the auditor that there has either been an other Party. overpayment or and underpayment of funds due, the Party owning funds shall, within ten (10) business days, make a payment of such funds to whom they are owed.

10.5 <u>Closing of Financials</u>. Provided that Purchaser shall not incur any out of pocket costs, Purchaser shall cause the individual acting as the chief financial officer of the Hospital after the Effective Time (the "**Post-Effective Time CFO**") to cooperate with Sellers' representatives in order to complete the standardized closing of Sellers' financial records through the Licensure Date including, without limitation, the closing of general ledger account reconciliations (collectively, the "**Closing of Financials**"). Purchaser shall cause the Post-Effective Time CFO to use their good faith efforts to cooperate with Sellers' representatives

### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 44 of 147

in order to complete the Closing of Financials by no later than the date which is sixty (60) calendar days after the Closing Date. The Post-Effective Time CFO and other appropriate personnel shall be reasonably available to Sellers for a period of no less than one hundred eighty (180) calendar days after the Closing Date to assist Sellers in the completion of Sellers' post-Closing audit, such assistance not to interfere unreasonably with such Post-Effective Time CFO's other duties.

10.6 <u>Medical Staff</u>. To ensure continuity of care in the community, Purchaser agrees that the Hospital's medical staff members in good standing as of the Licensure Date shall maintain medical staff privileges at the Hospital as of the Licensure Date. On and after the Licensure Date, the medical staff will be subject to the Hospital's Medical Staff Bylaws then currently in effect, provided that such Bylaws are in compliance with all applicable laws and regulations and contain customary obligations.

10.7 <u>Shared Intangible Assets</u>. In the event and to the extent that certain intangible Assets transferred by Sellers have been used to operate businesses of Verity or Verity Holdings or their affiliates ("**Shared Intangible Assets**") and such Shared Intangible Assets continue to be used by Verity or Verity Holdings or their affiliates to operate such businesses after Closing, Verity and Verity Holdings retain the rights to continue to use such Assets notwithstanding their sale to Purchaser. Purchaser shall reasonably cooperate with Verity and Verity Holdings and their affiliates to give effect to such rights and shall provide Verity and Verity Holdings and their affiliates such documentation, records and information and reasonable access to such systems as necessary for Verity and Verity Holdings and their affiliates to continue to operate such businesses; all in such manner as not to reasonably interfere with the operations of the Hospitals; *provided, however*, Purchaser shall not be required to incur any out-of-pocket costs in association therewith unless reimbursed by Verity and Verity Holdings and their affiliates.

# ARTICLE 11 DEFAULT, TAXES AND COST REPORTS

11.1 <u>Purchaser Default</u>. If the Agreement is terminated pursuant to <u>Section 9.1(b)</u>, Sellers shall be entitled to retain the Deposit, and Sellers may, in addition thereto, pursue any rights or remedies that Sellers may have under this Agreement or applicable law, including the right to sue for damages or specific performance.

11.2 <u>Sellers Default</u>. If the Agreement is terminated pursuant to <u>Section 9.1(c)</u>, Purchaser shall have the right to demand and receive a refund of the Deposit, and Purchaser may, in addition thereto, pursue any rights or remedies that Purchaser may have under applicable law, including the right to sue for damages or specific performance.

# 11.3 <u>Tax Matters; Allocation of Purchase Price</u>.

(a) After the Licensure Date, the Parties shall cooperate fully with each other and shall make available to each other, as reasonably requested, all information, records or documents relating to tax liabilities or potential tax liabilities attributable to Sellers with respect to the operation of the Hospital for all periods prior to the Licensure Date and shall preserve all such information, records and documents at least until the expiration of any applicable statute of

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 45 of 147

limitations or extensions thereof. The Parties shall also make available to each other to the extent reasonably required, and at the reasonable cost of the requesting Party (for out-of-pocket costs and expenses only), personnel responsible for preparing or maintaining information, records and documents in connection with tax matters and as Sellers reasonably may request in connection with the completion of any post-Closing audits of the Hospital.

(b) Solely for purposes of tax reporting, <u>Schedule 11.3</u> sets forth the allocation of the Purchase Price (including any liabilities that are considered to be an increase to the Purchase Price for United States federal income tax purposes) among the Assets in accordance with Section 1060 of the Code and the Treasury Regulations promulgated thereunder (the "Allocation Schedule"). The Allocation Schedule shall be final and binding upon Sellers and Purchaser with respect to matters relating to required tax reporting by each such Party. The Parties shall refrain from taking any position that is inconsistent with the Allocation Schedule with respect to tax reporting.

## 11.4 <u>Cost Report Matters</u>.

(a) Consistent with <u>Section 4.4</u>, Sellers shall, at Purchaser's expense, prepare and timely file all cost reports relating to the periods ending prior to the Licensure Date or required as a result of the consummation of the transactions described in this Agreement, including, without limitation, those relating to Medicare, Medicaid, and other third party payors which settle on a cost report basis (the "Seller Cost Reports").

(b) Upon reasonable notice and during normal business office hours, Purchaser will cooperate reasonably with Sellers in regard to Sellers' preparation and filing of the Seller Cost Reports. Such cooperation shall include, at no cost to Sellers, obtaining access to files at the Hospital and Purchaser's provision to Sellers of data and statistics, and the coordination with Sellers pursuant to reasonable notice of Medicare and Medicaid exit conferences or meetings. Sellers shall have no obligations after the Licensure Date with respect to Seller Cost Reports except for preparation and filing thereof.

## ARTICLE 12 MISCELLANEOUS PROVISIONS

12.1 <u>Further Assurances and Cooperation</u>. Each Party shall execute, acknowledge and deliver to the other Party any and all other assignments, consents, approvals, conveyances, assurances, documents and instruments reasonably requested by such Party at any time and shall take any and all other actions reasonably requested by such Party at any time for the purpose of consummating the transactions hereunder and fulfilling such Party's obligations hereunder. After consummation of the transactions contemplated in this Agreement, the Parties agree to cooperate with each other and take such further actions as may be necessary or appropriate to effectuate, carry out and comply with all of the terms of this Agreement, the documents referred to in this Agreement and the transactions contemplated hereby.

12.2 <u>Successors and Assigns</u>. All of the terms and provisions of this Agreement shall be binding upon and shall inure to the benefit of and be enforceable by the respective successors and assigns of the Parties hereto; *provided, however,* that no Party hereto may assign any of its

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 46 of 147

rights or delegate any of its duties under this Agreement without the prior written consent of the other Parties which consent shall not be unreasonably withheld or delayed, except that Purchaser may, without the prior written consent of Sellers, assign all or any portion of its rights under this Agreement to one or more of its affiliates prior to the Closing Date so long as such affiliate was described and was part of the Attorney General application under Section 5.95.8(c) and provided that any such assignment shall not relieve Purchaser of or reduce Purchaser's obligations under this Agreement.

12.3 <u>Governing Law; Venue</u>. This Agreement shall be construed, performed, and enforced in accordance with, and governed by, the laws of the State of California (without giving effect to the principles of conflicts of laws thereof), except to the extent that the laws of such State are superseded by the Bankruptcy Code or other applicable federal law. For so long as Sellers are subject to the jurisdiction of the Bankruptcy Court, the Parties irrevocably elect, as the sole judicial forum for the adjudication of any matters arising under or in connection with the Agreement, and consent to the exclusive jurisdiction of, the Bankruptcy Court. The Parties hereby consent to the jurisdiction of such court and waive their right to challenge any proceeding involving or relating to this Agreement on the basis of lack of jurisdiction over the Person or forum non conveniens.

12.4 <u>Amendments</u>. This Agreement may not be amended other than by written instrument signed by the Parties.

Exhibits, Schedules and Disclosure Schedule. The Disclosure Schedule and all 12.5 exhibits and schedules referred to in this Agreement shall be attached hereto and are incorporated by reference herein. From the Signing Date until the Closing, the Parties agree that Sellers may update the Disclosure Schedule as necessary upon written notice to Purchaser, and the applicable representation and warranty shall thereafter be deemed amended for all purposes by such updated Disclosure Schedule. Notwithstanding the foregoing, should any exhibit or schedule not be completed and attached hereto as of the Signing Date, Sellers and Purchaser shall promptly negotiate in good faith any such exhibit or schedule, which exhibit or schedule must be acceptable to each of Sellers and Purchaser in their reasonable discretion prior to being attached hereto. Any matter disclosed in this Agreement or in the Disclosure Schedule with reference to any Section of this Agreement shall be deemed a disclosure in respect of all sections to which such disclosure may apply. The headings, if any, of the individual sections of the Disclosure Schedule are provided for convenience only and are not intended to affect the construction or interpretation of this Agreement. The Disclosure Schedule is arranged in sections and paragraphs corresponding to the numbered and lettered sections and paragraphs of this Agreement merely for convenience, and the disclosure of an item in one section of the Disclosure Schedule as an exception to a particular obligation, representation or warranty shall be deemed adequately disclosed as an exception with respect to all other obligations, representations or warranties, notwithstanding the presence or absence of an appropriate section of the Disclosure Schedule with respect to such other obligation, representations or warranties or an appropriate cross reference thereto.

12.6 <u>Notices</u>. Any notice, demand, letter or other communication required, permitted, or desired to be given hereunder shall be deemed effectively given when either personally delivered, or when received by telegraphic or other electronic means (including facsimile) or

overnight courier, or five (5) calendar days after being deposited in the United States mail, with postage prepaid thereon, certified or registered mail, return receipt requested, addressed as follows:

If to Sellers:	Verity Health System of California, Inc. 601 South Figueroa St., Suite 4050 Los Angeles, CA 90017-5704 Attention: Chief Executive Officer
With a copies to:	Dentons US LLP
(which copies shall	601 South Figueroa St., Suite 2500
not constitute notice)	8
	Attention: Tania Moyron, Esq. Telephone: 213-243-6101
	Telephone. 215-245-0101
If to Purchaser:	Prime Healthcare Services, Inc.
If to Purchaser:	Prime Healthcare Services, Inc. 3480 East Guasti Road, 2 <sup>nd</sup> Floor
If to Purchaser:	3480 East Guasti Road, 2 <sup>nd</sup> Floor Ontario, California 91761
If to Purchaser:	3480 East Guasti Road, 2 <sup>nd</sup> Floor Ontario, California 91761 Attention: General Counsel
If to Purchaser:	3480 East Guasti Road, 2 <sup>nd</sup> Floor Ontario, California 91761
	3480 East Guasti Road, 2 <sup>nd</sup> Floor Ontario, California 91761 Attention: General Counsel Facsimile: 909-235-4316
If to Purchaser: With a copy to: (which copies shall	3480 East Guasti Road, 2 <sup>nd</sup> Floor Ontario, California 91761 Attention: General Counsel
With a copy to:	<ul> <li>3480 East Guasti Road, 2<sup>nd</sup> Floor</li> <li>Ontario, California 91761</li> <li>Attention: General Counsel</li> <li>Facsimile: 909-235-4316</li> <li>McDermott Will &amp; Emery LLP</li> <li>2049 Century Park East, Suite 3200</li> </ul>
With a copy to: (which copies shall	<ul> <li>3480 East Guasti Road, 2<sup>nd</sup> Floor</li> <li>Ontario, California 91761</li> <li>Attention: General Counsel</li> <li>Facsimile: 909-235-4316</li> <li>McDermott Will &amp; Emery LLP</li> <li>2049 Century Park East, Suite 3200</li> </ul>

or at such other address as one Party may designate by notice hereunder to the other Parties.

12.7 <u>Headings</u>. The section and other headings contained in this Agreement and in the Disclosure Schedule, exhibits and schedules to this Agreement are included for the purpose of convenient reference only and shall not restrict, amplify, modify or otherwise affect in any way the meaning or interpretation of this Agreement or the Disclosure Schedule, exhibits and schedules hereto.

12.8 <u>Publicity</u>. Prior to the Closing Date, Sellers and Purchaser shall consult with each other as to the form and substance of any press release or other public disclosure materially related to this Agreement or any other transaction contemplated hereby and each shall have the right to review and comment on the other's press releases at least forty-eight (48) hours prior to issuance; *provided, however*, that nothing in this <u>Section 12.8</u> shall be deemed to prohibit either Sellers or Purchaser from making any disclosure that its counsel deems necessary or advisable in order to satisfy either Party's disclosure obligations imposed by law subject to reasonable prior notice to the other Party thereof.

12.9 <u>Fair Meaning</u>. This Agreement shall be construed according to its fair meaning and as if prepared by all Parties.

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 48 of 147

12.10 <u>Gender and Number; Construction; Affiliates</u>. All references to the neuter gender shall include the feminine or masculine gender and vice versa, where applicable, and all references to the singular shall include the plural and vice versa, where applicable. Unless otherwise expressly provided, the word "including" followed by a listing does not limit the preceding words or terms and shall mean "including, without limitation." Any reference in this Agreement to an "affiliate" shall mean any Person directly or indirectly controlling, controlled by or under common control with a second Person. The term "control" (including the terms "controlled by" and "under common control with") means the possession, direct or indirect, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by contract or otherwise. A "Person" shall mean any natural person, partnership, corporation, limited liability company, association, trust or other legal entity.

12.11 <u>Third Party Beneficiary</u>. None of the provisions contained in this Agreement are intended by the Parties, nor shall they be deemed, to confer any benefit on any person not a party to this Agreement, except for the Parties' successors and permitted assigns, and except for any liquidating trustee or plan administrator for Sellers' estate.

12.12 Expenses and Attorneys' Fees. Except as otherwise provided in this Agreement, each Party shall bear and pay its own costs and expenses relating to the preparation of this Agreement and to the transactions contemplated by, or the performance of or compliance with any condition or covenant set forth in, this Agreement, including without limitation, the disbursements and fees of their respective attorneys, accountants, advisors, agents and other representatives, incidental to the preparation and carrying out of this Agreement, whether or not the transactions contemplated hereby are consummated. The Parties expressly agree that all sales, transfer, documentary transfer and similar taxes, fees, surcharges and the like in connection with the sale of the Assets shall be borne by Purchaser. If any action is brought by any Party to enforce any provision of this Agreement, the prevailing Party shall be entitled to recover its court costs and reasonable attorneys' fees.

12.13 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same Agreement, binding on all of the Parties hereto. The Parties agree that facsimile copies of signatures shall be deemed originals for all purposes hereof and that a Party may produce such copies, without the need to produce original signatures, to prove the existence of this Agreement in any proceeding brought hereunder.

12.14 <u>Entire Agreement</u>. This Agreement, the Disclosure Schedule, the exhibits and schedules, and the documents referred to in this Agreement contain the entire understanding between the Parties with respect to the transactions contemplated hereby and supersede all prior or contemporaneous agreements, understandings, representations and statements, oral or written, between the Parties on the subject matter hereof (the "**Superseded Agreements**"), which Superseded Agreements shall be of no further force or effect.

12.15 <u>No Waiver</u>. Any term, covenant or condition of this Agreement may be waived at any time by the Party which is entitled to the benefit thereof but only by a written notice signed by the Party expressly waiving such term or condition. The subsequent acceptance of

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 49 of 147

performance hereunder by a Party shall not be deemed to be a waiver of any preceding breach by any other Party of any term, covenant or condition of this Agreement, other than the failure of such other Party to perform the particular duties so accepted, regardless of the accepting Party's knowledge of such preceding breach at the time of acceptance of such performance. The waiver of any term, covenant or condition shall not be construed as a waiver of any other term, covenant or condition of this Agreement.

12.16 <u>Severability</u>. If any term, provision, condition or covenant of this Agreement or the application thereof to any Party or circumstance shall be held to be invalid or unenforceable to any extent in any jurisdiction, then the remainder of this Agreement and the application of such term, provision, condition or covenant in any other jurisdiction or to persons or circumstances other than those as to whom or which it is held to be invalid or unenforceable, shall not be affected thereby, and each term, provision, condition and covenant of this Agreement shall be valid and enforceable to the fullest extent permitted by law.

12.17 <u>Time is of the Essence</u>. Time is of the essence for all dates and time periods set forth in this Agreement and each performance called for in this Agreement.

12.18 WAIVER OF JURY TRIAL. TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT, OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (B) EACH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (C) EACH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (D) EACH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 12.18. THIS IS AN AFFIRMATIVE WAIVER OF THE PARTIES' RIGHTS TO A JURY TRIAL UNDER CALIFORNIA LAW, Cal. C. Civ. Pro. Sec 631. BY SIGNING BELOW ON THE SIGNATURE LINES, EACH PARTY IS EXPLICITLY WAIVING JURY TRIAL AND AUTHORIZING ANY AND ALL PARTIES TO FILE THIS WAIVER WITH ANY COURT AS THE WAIVER REQUIRED UNDER Cal. C. Civ. Proc. Sec. 631(f)(2).

[REMAINDER OF PAGE IS BLANK]

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 50 of 147

**IN WITNESS WHEREOF,** this Agreement has been entered into as of the day and year first above written.

# **PURCHASER:**

# Prime Healthcare Services, Inc.

Signature:	
Print Name:_	
Title:	
Date:	

## **SELLERS:**

# Verity Health System of California, Inc.

Signature:	
Print Name:	
Title:	
Date:	

## Verity Holdings, LLC

Signature:	
Print Name:_	
Title:	
Date:	

# **St. Francis Medical Center**

Signature:	
Print Name:	
Title:	
Date:	

Description	Owner	Address	APN
Hospital Campus (including Health Services Pavilion, Progressive Care Unit, Family Life Center, Power Plant and Parking Structure #1)	St. Francis Medical Center	3630 East Imperial Highway, Lynwood, CA 90262	6173-021-008
Huntington Park MOB	St. Francis Medical Center	2700 East Slauson Avenue, Huntington Park, CA 90255	6320-006-069
Maywood MOB	St. Francis Medical Center	5953 South Atlantic Boulevard 5, Maywood, CA 90270	6313-013-028
Parking Lot (St. Francis Medical Center)	Verity Holdings, LLC	3633 Martin Luther King Jr., Boulevard, Lynwood, CA 90262	6173-015-047
Ministry Services Building (St. Francis Medical Center)	Verity Holdings, LLC	3663 Martin Luther King Jr., Boulevard, Lynwood, CA 90262	6173-019-022
Parking Lot (St. Francis Medical Center)	Verity Holdings, LLC	3621 Martin Luther King Jr., Boulevard, Lynwood, CA 90262	6173-015-048
Five Unit Apartment (St. Francis Medical Center)	Verity Holdings, LLC	3570 Brenton Avenue, Lynwood, CA 90057	6191-016-008

# <u>Schedule 1.4(c)</u> St. Francis Medical Center

# Exhibit 1.3(a) to APA

## Form of Leaseback Agreement

See attached.

## SALE LEASEBACK AGREEMENT

[Note: This agreement is drafted for 3 Purchaser parties: Purchaser/Parent Company, Hospital Newco and Manager. This structure is not required.]

This Sale Leaseback Agreement (the "Leaseback Agreement") is made and entered into as of \_\_\_\_\_\_, 2020 (the "Signing Date"), by and among St. Francis Medical Center, a California nonprofit public benefit corporation (the "Corporation") on the one hand, and ("Parent Company") and

("Hospital Newco," and collectively with Parent Company, the "Parent Company Parties") on the other hand. The Corporation, Parent Company, and Hospital Newco may be referred to herein individually as a "Party," and collectively as the "Parties."

## RECITALS

**A.** On August 31, 2018, the Corporation and certain of its affiliates each filed a voluntary petition for relief (collectively, the "**Bankruptcy Cases**") under chapter 11 of title 11 of the United State Code (the "**Bankruptcy Code**").

**B.** The Bankruptcy Cases are jointly administered under Lead Case No. 18-20151 and are currently pending in the Bankruptcy Court for the Central District of California in Los Angeles (the "**Bankruptcy Court**").

C. Parent Company and the Corporation have entered into that certain Asset Purchase Agreement, dated of even date herewith (the "APA"), which provides for the sale of the assets of the Corporation and related assets (collectively, the "Assets"). Capitalized terms not defined in this Leaseback Agreement shall have the meanings given to such terms in the APA.

**D.** The Corporation operates St. Francis Medical Center as a 384-bed general acute care hospital (the "**Hospital**"), with its primary location at 3630 East Hospital Newco Highway, Lynwood, California 90262, and such other locations where the Hospital's services are provided (collectively, the "**Hospital Premises**").

E. Parent Company designated its affiliate, Hospital Newco, as the owner of the Corporation's Assets purchased by Parent Company under the APA (the "Designation"), and its affiliate (the "Manager") as the operator of the Hospital.

**F.** Pursuant to Section 1.3 of the APA, the Corporation, Parent Company, and the Manager, are entering into that certain Interim Management Agreement ("IMA") of even date herewith, and commencing at the Effective Time, to enable the Manager to manage the day-to-day operations of the Hospital following the Closing until the Manager is issued the Licenses necessary to operate the Hospital (for the avoidance of doubt, that date will occur when the Manager is issued both a license to operate the Hospital as an acute care hospital by the California Department of Public Health, and a permit to operate a hospital-based pharmacy by the California Board of Pharmacy (collectively, the "New Licenses")).

A. Pursuant to the APA, at the Effective Time, Parent Company will purchase the Corporation's Assets, except for the Excluded Assets (as set forth in the APA). Also pursuant to the APA, that portion of the Assets constituting drugs, dangerous devices, pharmacy systems, or other pharmacy assets (the "**Pharmacy Assets**"), the Medicare/Medi-Cal Agreements, the

### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 54 of 147

Lockboxes, the Transferred Managed Care Agreements and the Transferred Private Payor Agreements (collectively, the "Licensure Date Assets") shall transfer to Parent Company (or its affiliate) as of the Licensure Date.

**B.** Immediately following the Closing, and until the Manager obtains the New Licenses, the Parent Company Parties desire to lease back or license to the Corporation all of the then-acquired Corporation's Assets used in the operation of the Hospital, and the Corporation desires to so lease or license such Corporation's Assets from the Parent Company Parties on the terms and conditions set forth herein. The Corporation's Assets shall exclude the Licensure Date Assets, which the Corporation shall own until the New Licenses are issued.

## AGREEMENT

NOW, THEREFORE, in consideration of the covenants and conditions contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby agree as follows:

1. <u>Definitions</u>. Any capitalized term appearing herein that is not defined shall have the same definition ascribed to it under the IMA or the APA (as designated when such term first appears herein).

## 2. <u>Description of The Leased or Licensed Assets</u>.

**2.1** Leased or Licensed Assets. The Parent Company Parties hereby lease or license to the Corporation, and the Corporation leases or licenses from the Parent Company Parties, all of the Corporation's Assets used in the operation of the Hospital, except for the Licensure Date Assets, which the Corporation shall continue to own until the Licensure Date (as defined in the APA) (collectively, the "Leased or Licensed Assets"). Specifically, the Leased or Licensed Assets shall include: (a) the Hospital Premises (which shall be leased); (b) tangible personal property, including, but not limited to, fixtures, furnishings, hard copy medical and financial records and equipment (including, but not limited to, hardware to operate and run the electronic health record systems, hospital operating systems, laboratory information systems, and financial reporting systems) (which shall be leased); and (c) intangible intellectual property saved or embodied in the electronic health record systems, hospital operating systems, laboratory information systems, laboratory information systems, and financial reporting systems (which shall be licensed for use by the Corporation).

## 2.2 <u>Management of Pharmacy and Use of Pharmacy Assets</u>.

(a) The Corporation shall at all times during the term of this Leaseback Agreement be the owner of the Pharmacy Assets. The Parties acknowledge, however, that under the IMA, the Manager will be managing the Corporation's pharmacy during the Management Period (as defined in the IMA), and the Corporation therefore grants the Manager (i) access to and authority to use the drugs, dangerous drug delivery devices, or other tangible pharmacy assets, and (ii) a license to use the intangible intellectual property saved or embodied in the pharmacy systems, in each case to the extent necessary for the Manager to fulfill its obligations under the IMA.

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 55 of 147

(b) Pursuant to the IMA, the Corporation shall be responsible for purchasing drugs and dangerous devices identified by the Manager as necessary for the operations of the Hospital, and the Manager shall reimburse the Corporation for all costs and expenses incurred for such purchases.

**3.** <u>Term; Termination</u>. This Leaseback Agreement shall have a term coextensive with the Management Period (as defined in the IMA) and shall automatically terminate upon the termination of the IMA.

4. <u>Payments by the Parent Company Parties</u>.

**4.1** <u>Utilities</u>. The Parent Company Parties shall pay all utilities and services supplied to the Hospital during the term hereof, including but not limited to water, gas, air conditioning, heat, light, power, telephone service, and waste removal services.

**4.2** <u>Taxes</u>. The Parent Company Parties shall pay all taxes, assessments, and levies of any kind or nature whatsoever, including real property taxes, personal property taxes, income taxes, employment taxes, and sales or use taxes, that are taxed, assessed, levied, invoiced or imposed upon or against the Leased or Licensed Assets, the Hospital, and/or the Corporation, after the Effective Time.

**4.3** Insurance. The Parent Company Parties shall pay for all insurance coverages, including premiums, deductibles, stop-loss, and any other insurance covering the Leased or Licensed Assets, the Licensure Date Assets, the Hospital, and the Corporation during the term hereof. The Parent Company Parties covenant and agree that the Leased or Licensed Assets, the Licensure Date Assets, the Hospital, and the Corporation are covered as of the date hereof and will be covered at all times by general liability, fire, theft, business interruption, cyber, professional liability, directors and officers insurance, employment practices liability, terrorism, workers' compensation & employers' liability, directors and officers, fiduciary, crime, punitive damages excess liability, physical damage, property liability, automobile, storage tank, helipad and non-owned aviation, sexual misconduct and molestation, medical provider professional liability, and provider capitation stop loss (managed care excess loss) insurance. All such insurance shall name the Parent Company Parties and the Corporation as insureds as their respective interests may appear.

4.4 <u>Repairs and Maintenance: Alterations</u>. During the term of this Leaseback Agreement, the Parent Company Parties shall pay all costs of repairing (including replacement of) and maintaining the Leased or Licensed Assets and Hospital and every part thereof in good and sanitary order, condition and repair during the term hereof, reasonable wear and tear excepted, including, without limitation, all costs of all repairs, replacements and maintenance required by any applicable governmental law, statute, ordinance, rule or regulation, including the California Office of Statewide Health Planning and Development ("**OSHPD**"). The Corporation shall not make any alterations or changes to the Leased or Licensed Assets without prior written approval of the Parent Company Parties, which may be given or withheld in the Parent Company Parties' sole discretion.

### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 56 of 147

**4.5** <u>Payment</u>. Nothing in this <u>Section 4</u> shall in any way limit, reduce, or otherwise affect Parent Company's payment obligations under the IMA or the APA.

5. <u>Use</u>. The Leased or Licensed Assets shall be used for the operation of the Hospital, subject to the terms of the APA and the IMA.

6. <u>Risk of Loss</u>.

**6.1** The Corporation assumes no risks or liability for damage to or injury occurring to the Leased or Licensed Assets or Hospital during the term of this Leaseback Agreement by any means whatsoever, including fire, storm, earthquake, vandalism, strike, accident or any other casualty (collectively, "**Casualty**"), and the Parent Company Parties shall have all right, title, and interest in and to the proceeds of any insurance it obtained and paid for covering such Casualty.

**6.2** If, during the term of this Leaseback Agreement, action is initiated to take the Hospital Premises or any portion thereof by eminent domain or condemnation proceedings, exercise of state authority under an executive order or by deed in lieu thereof (collectively, "**Condemnation**"), the Parent Company Parties, and not the Corporation, shall have all right, title, and interest in and to the award from the Condemnation.

**6.3** In the event of a Casualty or Condemnation, neither the Parent Company Parties nor the Corporation may terminate this Leaseback Agreement.

7. <u>Continued Access</u>. Following termination of this Leaseback Agreement, and until the entry of final decrees closing the Bankruptcy Cases, the Patient Care Ombudsman, appointed by the United States Trustee pursuant to Bankruptcy Code § 333 and approved by the Bankruptcy Court [*see* Docket No. 430], shall have continuing access to the Leased or Licensed Assets and related personnel during normal business hours and upon at least one (1) business day's prior written notice to the Parent Company Parties, for the purpose of winding down the Corporation's affairs, in connection with any litigation or adversary proceedings, and to effectuate the chapter 11 plan as approved by the Bankruptcy Court.

8. <u>Miscellaneous</u>.

**8.1** <u>Further Assurances</u>. Each of the Parties hereto agrees to execute and deliver any and all further agreements, documents or instruments necessary to effectuate this Leaseback Agreement and the transactions referred to herein or contemplated hereby or reasonably requested by the other Party to perfect or evidence their rights hereunder.

**8.2** <u>Notices and Demands</u>. All notices and demands, requests, consents, approvals, and other similar communications under this Leaseback Agreement shall be in writing and shall be sent by personal delivery or by either (a) United States certified or registered mail, return receipt requested, postage prepaid, or (b) Federal Express or similar generally recognized overnight carrier regularly providing proof of delivery, addressed as follows:

If to the Corporation: St. Francis Medical Center

### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 57 of 147

c/o Verity Health System of California, Inc.601 South Figueroa Street, Suite 4050Los Angeles, CA 90017-5704Attention: Chief Executive Officer

With copies to:Tania Moyron, Esq.(which copy shall not<br/>constitute notice)Dentons US LLP<br/>601 South Figueroa St., Suite 2500<br/>Los Angeles, CA 90017-5704

and

Hope Levy-Biehl, Esq. Davis Wright Tremaine LLP 865 S Figueroa St, Los Angeles, CA 90017

If to the Parent Company Parties:

With copies to: (which copies shall not constitute notice)

Any notice so given by mail shall be deemed to have been given as of the date of delivery (whether accepted or refused) established by U.S. Post Office return receipt or the overnight carrier's proof of delivery, as the case may be, whether accepted or refused. Any such notice not so given shall be deemed given upon receipt of the same by the Party to whom the same is to be given. Any Party hereto may designate a different address for itself by notice to the other Parties in accordance with this Section 8.2.

**8.3** <u>Payment of Expenses</u>. Except for the Parent Company Parties' obligation to be responsible for certain costs, fees, and expenses as set forth elsewhere in this Leaseback Agreement, each Party hereto shall bear its own legal, accounting, and other expenses incurred in connection with the preparation and negotiation of this Leaseback Agreement and the consummation of the transactions contemplated hereby, whether or not the transaction is consummated.

**8.4** <u>Rent</u>. The Corporation has prepaid the sum of One Thousand Dollars (\$1,000.00), the receipt of which is hereby acknowledged by the Parent Company Parties, and the Corporation shall not be required to pay the Parent Company Parties any additional rent under this Leaseback Agreement.

**8.5** <u>Entire Agreement; Amendment; Waiver</u>. This Leaseback Agreement, the IMA, and those provisions of the APA expressly identified in this Leaseback Agreement contain the entire agreement among the Parties with respect to the subject matter hereof and supersede all prior agreements, arrangements, and understandings relating to the subject matter hereof and

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 58 of 147

thereof. There are no written or oral agreements, understandings, representations, or warranties among the Parties other than those set forth in this Leaseback Agreement, the IMA, and those provisions of the APA expressly identified in this Leaseback Agreement. Nothing in this Leaseback Agreement modifies or shall be construed as modifying any orders entered by the Bankruptcy Court. This Leaseback Agreement may not be modified or amended except in writing signed by the Parties. No waiver of any term, provision or condition of this Leaseback Agreement in any one or more instances, shall be deemed to be or be construed as a further or continuing waiver of any such term, provision or condition of this Leaseback Agreement. No failure to act shall be construed as a waiver of any term, provision, condition or rights granted hereunder.

**8.6** <u>Assignment</u>. Neither this Leaseback Agreement nor the rights, duties or obligations arising hereunder shall be assignable or delegable by the Corporation or the Parent Company Parties without the prior written consent of the other Parties, which may be granted, denied or conditioned in such Party's absolute discretion except that the Parent Company Parties may assign this Leaseback Agreement in connection with any permitted assignment under the IMA. Subject to the foregoing, this Leaseback Agreement shall be binding upon, and inure to the benefit of, the respective successors and assigns of the Parties.

**8.7** <u>Joint Venture; No Third Party Beneficiaries</u>. Nothing contained herein shall be construed as forming a joint venture or partnership among the Parties with respect to the subject matter hereof. The Parties do not intend that any third party shall have any rights under this Leaseback Agreement.

**8.8** <u>Captions</u>. The section headings contained herein are for convenience only and shall not be considered or referred to in resolving questions of interpretation.

**8.9** <u>Governing Law</u>. This Leaseback Agreement shall be governed by and construed in accordance with the laws of the State of California without regard to its conflicts of laws principles or decisions, except to the extent that the laws of such State are superseded by the Bankruptcy Code or other applicable federal law.

The Parties agree that the Bankruptcy Court shall have 8.10 Jurisdiction. exclusive personal and subject matter jurisdiction over any action or proceeding with respect to any claim arising out of or related to this Leaseback Agreement, and any of the documents or transactions contained in or contemplated by this Leaseback Agreement; provided, however, that the Parties agree that the United States District Court for the Central District of California (together with the Bankruptcy Court, the "Chosen Courts") shall have exclusive jurisdiction over such claim if (i) the Bankruptcy Cases are closed and if the Bankruptcy Cases are not reopened to adjudicate such claim after request by the Party bringing such claim or (ii) the Bankruptcy Court determines that it does not have jurisdiction over such claim. Solely in connection with claims arising under this Leaseback Agreement, or any of the documents or transactions contemplated hereby, the Parties (a) irrevocably submit to the exclusive jurisdiction of the Chosen Courts, (b) waive any objection to laying venue in any such action or proceeding in the Chosen Courts, (c) waive any objection that the Chosen Courts are an inconvenient forum or do not have jurisdiction over any Party hereto, and (d) agree that service of process upon such

### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 59 of 147

Party in any such action or proceeding shall be effective if notice is given in accordance with <u>Section 8.2</u> hereof

**8.11** <u>Conditions to Effectiveness</u>. The Parties acknowledge that this Leaseback Agreement is subject to and contingent upon the occurrence of the Closing under the APA. This Leaseback Agreement shall not be effective, nor shall any Party have any obligations hereunder, unless and until the Closing under the APA has occurred.

**8.12** <u>Cooperation on Regulatory Compliance Matters</u>. The Parent Company Parties understand and acknowledges that the Corporation intends to comply with applicable federal and state laws, regulations, and guidance, as well as the requirements or recommendations of any accrediting agencies. In the event the terms of this Leaseback Agreement need to be amended or supplemented based on guidance from or at the request or direction of a regulator made during the term of this Leaseback Agreement, the Parent Company Parties shall cooperate with such amendment and/or supplement to ensure the Corporation's ability to comply with such guidance, request, recommendation or directive.

**8.13** <u>Transition Services Agreement</u>. Pursuant to Section 1.4(f) and Section 1.5(d) of the APA, the Sellers and Parent Company will enter into a Transition Services Agreement to facilitate the winding down of the Sellers' businesses, the completion of the Bankruptcy Cases, and the dissolution of the Sellers. The Parties shall cooperate with each other to enable the Corporation and Parent Company to carry out their obligations under, and give effect to the terms of, the Transition Services Agreement.

**8.14** Fair Meaning. This Leaseback Agreement shall be construed according to its fair meaning and as if prepared by all Parties.

**8.15** <u>Counterparts</u>. This Leaseback Agreement may be executed by one or more of the Parties on any number of separate counterparts (including by telecopy), and all of said counterparts taken together shall be deemed to constitute one and the same instrument. This Leaseback Agreement may be delivered by facsimile or electronic transmission, including by e-mail as a PDF, and facsimile or PDF copies of executed signature pages, which shall be binding as originals.

WAIVER OF JURY TRIAL. 8.16 TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES. AND THEREFORE EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY OUT OF OR RELATING ARISING TO THIS AGREEMENT, OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (B) EACH PARTY

UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (C) EACH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (D) EACH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 8.16. THIS IS AN AFFIRMATIVE WAIVER OF THE PARTIES' RIGHTS TO A JURY TRIAL UNDER CALIFORNIA LAW, Cal. C. Civ. Pro. Sec 631. BY SIGNING BELOW ON THE SIGNATURE LINES, EACH PARTY IS EXPLICITLY WAIVING JURY TRIAL AND AUTHORIZING ANY AND ALL PARTIES TO FILE THIS WAIVER WITH ANY COURT AS THE WAIVER REQUIRED UNDER Cal. C. Civ. Proc. Sec. 631(f)(2).

[Signature Page Follows.]

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 61 of 147

**IN WITNESS WHEREOF,** the Parties agree to the foregoing terms of agreement through the execution below by their respective, duly authorized representatives as of the Signing Date.

Parent Company Parties:	Corporation:
NAME	ST. FRANCIS MEDICAL CENTER, a California non-profit public benefit corporation
By:	
Name:	By:
Title:	Name: Title:
By:	
Name:	
Title:	
By:	
Name:	
Title:	

# Exhibit 1.3(b) to APA

# Form of Interim Management Agreement

See attached.

## **INTERIM MANAGEMENT AGREEMENT**

[Note: This IMA is drafted for 3 Purchaser parties: Purchaser/Parent Company, Hospital Newco and Manager. This IMA is drafted assuming Purchaser is buying hospital's accounts receivable. This structure is not required.]

This Interim Management Agreement (the "Agreement") is made and entered into as of \_\_\_\_\_\_, 2020 (the "Signing Date"), by and among St. Francis Medical Center, a California nonprofit public benefit corporation (the "Corporation") on the one hand, and \_\_\_\_\_\_ ("Parent Company") and \_\_\_\_\_\_ (the "Manager," and collectively with the Corporation and Parent Company, the "Parties" and each individually a "Party")) on the other hand.

## RECITALS

**A.** On August 31, 2018, the Corporation and certain of its affiliates each filed a voluntary petition for relief (collectively, the "**Bankruptcy Cases**") under chapter 11 of title 11 of the United State Code (the "**Bankruptcy Code**").

**B.** The Bankruptcy Cases are jointly administered under Lead Case No. 18-20151 and are currently pending in the Bankruptcy Court for the Central District of California in Los Angeles (the "**Bankruptcy Court**").

C. Parent Company, as purchaser, and the Corporation and certain of its affiliates (collectively, the "Sellers" as defined in the APA), as seller, entered into that certain Asset Purchase Agreement, dated of even date herewith (the "APA"), which provides for the sale of certain assets of the Sellers (collectively, the "Assets"). All terms not otherwise defined herein shall have the meaning ascribed to them in the APA.

**D.** The Corporation operates St. Francis Medical Center as a 384-bed general acute care hospital (the "**Hospital**"), with its primary location at 3630 East Hospital Newco Highway, Lynwood, California 90262, and such other locations where the Hospital's services are provided (collectively, the "**Hospital Premises**")

**E.** Pursuant to the APA, Parent Company agreed to purchase the Assets, which include certain assets of the Corporation (the "Corporation's Assets").

F. Parent Company designated its affiliate, \_\_\_\_\_ ("Hospital Newco") as the owner of the Corporation's Assets, and the Manager as the operator of the Hospital.

**G.** Parent Company and the Manager have requested this Agreement to afford the Manager additional time to obtain its general acute care hospital license from the California Department of Public Health ("**CDPH**"), and its hospital pharmacy permit from the California

## Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 64 of 147

Board of Pharmacy (together, the "New Licenses," and the date on which the New Licenses are issued is the "Licensure Date").

**H.** Pursuant to the APA, that portion of the Assets constituting drugs, dangerous devices, pharmacy systems, or other pharmacy assets (the "**Pharmacy Assets**"), the Medicare/Medi-Cal Agreements, the Lockboxes, the Transferred Managed Care Agreements and the Transferred Private Payor Agreements (collectively, the "**Licensure Date Assets**") shall transfer to Hospital Newco as of the Licensure Date.

**I.** The Corporation shall maintain a possessory interest in the Hospital and the Hospital Premises, and Parent Company and Hospital Newco on the one hand as lessor, and the Corporation, on the other hand as lessee, are entering into that certain Sale Leaseback Agreement of even date herewith, pursuant to which certain of the Corporation's Assets will be leased back to the Corporation (the "Leaseback Agreement").

J. Until Hospital Newco obtains the New Licenses, the Manager desires to assume the management of the Hospital, including its pharmacy, on behalf of the Corporation, and the Corporation desires to avail itself of such management services, upon the terms and conditions set forth in this Agreement.

## **TERMS OF AGREEMENT**

**NOW, THEREFORE,** in consideration of the promises and the mutual covenants contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby agree as follows:

## 1. Term.

1.1 The term of this Agreement (the "**Management Period**") shall commence as of the Effective Time (as defined in the APA) and shall continue until the earlier of the Licensure Date or [June 30, 2021]. The Parties acknowledge that, during the Management Period, the Corporation shall remain the licensee of the Hospital, and in that capacity, and during such period, shall retain statutory and regulatory authority and responsibility for the Hospital and for oversight of the Manager.

## 2. Acknowledgements and Covenants.

**2.1** As of the Effective Time, Parent Company shall have acquired all of the Corporation's Assets as set forth in the APA (other than the Licensure Date Assets), which shall be leased back to the Corporation pursuant to the terms of the Leaseback Agreement during the term of this Agreement.

**2.2** Pursuant to Section 1.3 of the APA, contemporaneously with the date the New Licenses are issued and the termination of the Leaseback Agreement, ownership of the Licensure Date Assets shall be transferred to Hospital Newco. Parent Company and the Manager acknowledge that this Agreement and the foregoing subsequent transfer of the Licensure Date Assets are made at the request of Parent Company and the Manager, to provide more time for the

### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 65 of 147

Manager to obtain its New Licenses. As such, Parent Company acknowledges, covenants, and agrees that the total Purchase Price (as defined in the APA) shall be paid to the Sellers under the APA as if the Licensure Date Assets were transferred as of the Effective Time.

**2.3** As of the Effective Time, the Corporation shall have ended the employment or engagement of all employees and contractors and, to the extent they received offers of employment or engagement from the Manager (or its affiliates) and accepted such offers, shall have been transferred to the Manager (or its affiliate) as employees or contractors of the Manager (collectively, the "**Hired Employees**").

**2.4** During the Management Period, the Corporation will (a) assign an individual to oversee the operation of the Hospital and serve as the president of the Corporation, as required by the Centers for Medicare & Medicaid Services, CDPH, and applicable California laws and regulations (the "**Responsible Officer**"), and (b) maintain proper oversight by a board of directors.

# **3.** Appointment of Manager.

**3.1** During the Management Period, the Corporation hereby appoints the Manager as the sole and exclusive provider of the Services (defined below) and hereby grants to the Manager the exclusive right to manage the Hospital under the Corporation's Licenses (as defined in the APA) as a general acute care hospital, including without limitation, the right to undertake those certain management responsibilities and permitted activities described in <u>Section 4</u> below. The Manager hereby accepts such appointment for all purposes with respect to the Corporation's rights, duties, and responsibilities under the Licenses for the Hospital, to the fullest extent permitted by law, and agrees, to the fullest extent permitted by law, to provide management services to the Hospital on behalf of the Corporation (the "Services").

**3.2** The Manager's Services hereunder shall include management and operation of the Hospital's pharmacy on behalf of the Hospital, even though the Pharmacy Assets will not be transferred to Hospital Newco pursuant to the APA until the Licensure Date.

**3.3** Upon the Licensure Date, the Manager's Services provided to the Hospital under the Corporation's Licenses shall terminate and, thereafter, Hospital Newco (and its affiliates) will be operating the Hospital as the licensee holding its own New Licenses.

**3.4** During the Management Period, the Manager shall submit claims for services rendered by the Hospital to various governmental and non-governmental entities, patients, and other third parties pursuant to the Corporation's provider agreements, payor contracts, and NPI numbers as set forth in <u>Schedule 3.4</u>, attached hereto (collectively, the "**Corporation's Billing Credentials**"). Because all billing and collecting shall be under the Corporation's Billing Credentials, payments shall be made in the Corporation's name and deposited in the Corporation's bank accounts. The Parties acknowledge and agree that during the Management Period, the Corporation's bank accounts and lockboxes shall remain under the Corporation's name and in the Corporation to its accounts. To the extent the Manager is unable to make such transfers to its own accounts, upon request by the Manager, the Corporation will, at the

Manager's cost, forward to the Manager any payment or reimbursement received during the Management Period.

**3.5** To the extent the Corporation's lenders continue to have liens, claims, charges, mortgages, security interests, rights of setoff, recoupment, netting and deductions, rights of first offer, first refusal and any other similar contractual property, legal or equitable rights, and any successor or successor in interest liability theories, or any other encumbrances whatsoever (collectively, "Encumbrances")Encumbrances on the Corporation's bank accounts or lockboxes as of the Effective Time, the Corporation, Parent Company, and the Manager shall work cooperatively to facilitate termination and release of such Encumbrances.

**3.6** The Corporation shall retain and, upon the Licensure Date, the Manager shall assume, any contracts necessary for the Corporation to continue to be the holder of the Licenses and to bill for Hospital services during the Management Period, in accordance with the APA. Such contracts shall be as set forth in <u>Schedule 3.6</u>, attached hereto (the "**Retained Contracts**"). Parent Company hereby designates each Retained Contract as an Assigned Contract under Section 1.11 of the APA.

## 4. Management Responsibility.

**4.1** During the Management Period, the Manager shall, subject to all applicable legal and regulatory requirements and the Corporation's ultimate oversight and control, have responsibility for the management of the Hospital, and agrees to assume and discharge all responsibilities, duties, liabilities, payments, and obligations in connection with properly maintaining the Hospital in full compliance with all regulations and standards required of a general acute care hospital facility so licensed. In furtherance thereof, the Manager's Services shall include, but not be limited to, the following duties, which duties shall be performed at the Manager's sole cost and expense:

(a) Managing the operations of the Hospital as a general acute care hospital in compliance with all applicable laws, regulations, provider agreements, payor contracts, CDPH requirements for maintenance of the Licenses in good standing, Medicare conditions of participation and requirements for payment with respect to governmental programs, and the requirements for maintenance of the Hospital's accreditations;

(b) Employing and managing the Hired Employees and any other non-clinical and clinical personnel (i) deemed necessary by the Corporation for the operation of the Hospital as a general acute care hospital, or (ii) required by law so as to meet all applicable labor laws and regulations, and consistent with orders of the Bankruptcy Court;

(c) Ensuring that the Corporation is able to pay itself out of the Hospital Revenues for the costs and expenses set forth in <u>Sections 4.5(b)</u>;

(d) Maintaining and repairing, as needed, the Hospital Premises so as to ensure material compliance with all applicable local, state and federal law, and construction timelines imposed by OSHPD.

### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 67 of 147

(e) Providing security services reasonably necessary to prevent unlawful entry or damage to the Hospital Premises;

(f) Affording the Responsible Officer or his or her designee access, during normal business hours, to the Hospital Premises, the books and records at the Hospital Premises or in Manager's possession, the Hired Employees and any other personnel of Manager or otherwise who are providing services associated with the operation of the Hospital, and such other access and assistance as reasonably requested by the Responsible Officer;

(g) Upon at least one (1) day's prior written notice to the Manager, providing access, during normal business hours, to the Hospital Premises to lessors of equipment at the Hospital Premises, if any, who have been authorized by order of the Bankruptcy Court to remove their equipment from the Hospital Premises, provided that the Manager shall have full power and authority to require that the removal of such equipment by such lessors does not damage the Hospital Premises;

(h) Providing access, during normal business hours, to the Hospital, the Hospital Premises, the Hospital's books and records, electronic health records, financial information systems, operating systems, laboratory systems, the Hired Employees and any other personnel of Manager or otherwise who are providing services associated with the operation of the Hospital, to the Corporation, the Corporation's directors, officers and representatives, and the Corporation's successors in interest, including, but not limited to, any plan administrator, liquidating trustee or similar representatives appointed or approved by the Bankruptcy Court, as well as the Patient Care Ombudsman appointed in the Bankruptcy Cases under Section 333 of the Bankruptcy Code (the "PCO") [see Docket No. 430], for the purpose of winding down the Corporation's affairs, pursuing litigation and adversary proceedings, and to effectuate a plan of liquidation, as approved by the Bankruptcy Court (the "Plan"), provided, however, that such access does not unreasonably disrupt the Hospital's operations;

(i) Maintaining, all licenses, permits consents, approvals, accreditations, and certifications currently held by the Corporation in good standing, in active status, and in compliance with all applicable local, state, and federal laws, including the timely payment of all applicable fees to support or renew these approvals;

(j) Maintaining and obtaining all insurance coverages, from and after the Effective Time, for the Hospital that a prudent hospital operator or owner would maintain, including directors and officers insurance with no less coverage than was maintained for directors and officers just prior to the Effective Time;

(k) Maintaining and obtaining those insurance coverages required under the Leaseback Agreement, for its own account, with the Corporation included as a named insured, and paying all amounts required under the Leaseback Agreement in a timely manner, including rent, utilities, taxes, and insurance premiums;

(1) Opening and forwarding all mail relating to the financial or business affairs of the Corporation to the notice address below;

## Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 68 of 147

(m) Periodically reporting to the Corporation (or its designee), either in person or telephonically, the condition of the Hospital and the Hospital Premises;

(n) Coordinating with the governing board and the organized medical staff (each, as established by the Corporation and the Sellers) on the appropriateness and quality of medical care and all medical staff issues requiring governing board oversight;

(o) Paying all costs and expenses in connection with and incidental to ownership of the Corporation's Assets and management and operation of the Hospital hereunder, including but not limited to, all the Hospital operating costs, employee-related costs, and taxes, whether or not identified, described, or referenced in this Agreement;

(p) Cooperating with the Corporation in facilitating termination and release of any Encumbrances on the Corporation's bank accounts and lockboxes; and

(q) Performing such other duties and activities as are reasonably necessary for the Manager to fulfill its responsibilities under this Agreement and the APA.

**4.2 Permitted Manager Activities**. During the Management Period, the Manager may do any of the following, in consultation with the Corporation and subject to the requirements of applicable local, state, and federal law, which activities may be performed by the Manager at the Manager's sole cost and expense:

(a) Make alterations, improvements, and repairs to the interior or exterior of the Hospital Premises, including structural alterations, improvements, and repairs;

(b) Remove and dispose of furniture, fixtures, equipment (other than equipment owned by equipment lessors), and supplies at the Hospital Premises;

(c) Move into and install furniture, fixtures, equipment, and supplies at the Hospital Premises;

(d) Prepare the Hospital for a name change, except that no such name change may take effect, and no signage reflecting such change shall be installed, during the Management Period; and

(e) Perform, or permit to be performed, any other activities at the Hospital Premises that are not inconsistent with operating the Hospital under the Licenses, and receiving and retaining for the Manager's own account all revenues and proceeds of any such activities, to the extent they comprise the Manager Compensation.

# 4.3 **Prohibited Manager Activities.**

(a) Notwithstanding anything to the contrary in this Agreement, the Manager shall have no authority to take and shall not take any action with respect to any Excluded Assets or Excluded Liabilities (as such terms are defined in the APA) of the Corporation.

### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 69 of 147

(b) The Manager's authority to manage and operate the Hospital is limited to those actions that Manager is expressly required or permitted to do hereunder.

(c) The Manager shall not (i) take any action that interferes with the Corporation's transfer of funds to pay itself out of the Hospital Revenues as set forth in <u>Section</u> 4.5(b); or (ii) remove, withdraw, or authorize removal or withdrawal of funds from the Corporation's bank accounts or lockboxes to the extent that the Corporation would be unable to fully pay itself as set forth in <u>Section 4.5(b)</u>.

## 4.4 APA Provisions; Transition Services Agreement.

(a) Nothing herein shall modify the prorations of expenses and utilities set forth in the APA, including specifically as set forth in <u>Section 1.6</u> of the APA.

(b) None of the information accessed, learned or obtained by Parent Company or the Manager or any of their affiliates in the course of performing their duties hereunder may serve as the basis for payment of less than the full Purchase Price or to otherwise assert a claim against the Sellers.

(c) Nothing herein shall modify the transfer of the Assets from the Sellers to Parent Company as contemplated in the APA, including specifically as set forth in <u>Section 1.7</u> of the APA.

(d) Nothing herein shall modify the APA in respect of the exclusion from purchase by Parent Company of the Excluded Assets, including specifically as set forth in <u>Section 1.8</u> of the APA.

(e) Pursuant to Section 1.4(f) and Section 1.5(d) of the APA, the Sellers and Parent Company will be entering into a Transition Services Agreement to facilitate the winding down of the Sellers' businesses, the completion of the Bankruptcy Cases, and the dissolution of the Sellers. The Parties shall cooperate with each other to enable the Corporation and Parent Company to carry out their obligations under, and give effect to the terms of, the Transition Services Agreement.

# 4.5 Manager Compensation; Distribution of Revenue.

(a) As used in this Agreement, the term "**Hospital Revenues**" shall mean all revenues and reimbursement received by the Corporation after the Effective Time, including, but not limited to, cash, accounts, notes, or other accounts receivable, disproportionate share payments, quality assurance fee payments, Seller Cost Report (as defined in the APA) settlements, and capitation premiums, whether payable by Medicare, Medi-Cal or any other commercial or governmental payor, or any health maintenance organization or any other managed care program or any private pay patients, but only to the extent they are not Excluded Assets (as defined in the APA).

### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 70 of 147

(b) As full and complete payment for the Manager's Services, the Manager shall be entitled to receive an amount equal to the Hospital Revenues less the following amounts that will be paid to the Corporation by Manager (the "Manager Compensation"):

(i) All costs and expenses incurred by the Corporation for the Corporation's purchase of drugs and dangerous devices that the Manager determines are necessary for the operation of the Hospital;

(ii) All salaries, stipends, costs and expenses associated with the employment or engagement of the board of directors of the Corporation and its parent organization, VHS, the Responsible Officer and any other retained employees or contractors of the Corporation primarily or exclusively providing services necessary for the operation of the Hospital and compliance with applicable laws, if any; and

(iii) All other costs and expenses associated with keeping the Corporation in good standing, maintaining its Licenses and maintaining the Retained Contracts, including, but not limited to, filings with the secretary of state, filing tax returns, board of directors expenses, directors and officers insurance, employment practices liability insurance, compensation, benefits, and such other costs and expenses incurred by a hospital corporation in the normal course of business.

(c) Corporation shall instruct the financial institutions that service its bank accounts or lockboxes to, immediately following the Effective Time and during the Management Period, sweep all Hospital Revenues from the Corporation's bank accounts or lockboxes on a daily basis into an account designated in writing by Manager, and Corporation shall not revise such written instructions without obtaining the prior written consent of Manager.

(d) Under no circumstance shall the Manager seek payment for the Manager's Services from the liquidating trustee, the Corporation, any of the Corporation's officers, directors, agents, contractors, personnel, affiliates or subsidiaries, Verity Health System of California, Inc. ("VHS"), or any of VHS' officers, directors, agents, contractors, personnel, affiliates or subsidiaries.

(e) The Corporation shall issue invoices to the Manager on a weekly basis for the items described in Section 4.5(b) (i), (ii) and (iii), with reasonable supporting detail therefor. The Manager shall pay such invoices within ten (10) business days of receipt of such invoices. If the Manager does not remit payment in respect of such invoices in accordance with the immediately preceding sentence, the unpaid amount of such invoices shall bear interest at the Prime Rate in effect on the calendar day upon which such payment was required to be made to the Corporation (the "**Invoice Payment Due Date**") plus five percent (5%) (or the maximum rate allowed by law, whichever is less), such interest accruing on each calendar day after the Invoice Payment Due Date until payment of such invoices and all interest thereon is made to the Corporation.

## 4.6 Liabilities and Losses.

(a) Parent Company shall be responsible for all the Hospital liabilities and losses incurred or accrued during the Management Period.

(b) Nothing hereunder shall, or is intended to, modify or supersede Parent Company's responsibility for the Transferred Obligations (as defined in the APA) or the Corporation's responsibility for the Excluded Liabilities (as defined in the APA).

(c) Specifically, with respect to any and all quality assurance fee ("QAF") liabilities owed by the Corporation, Parent Company shall pay all QAF fees that become due and owing or are overdue during the Management Period, even though such fees may have been calculated based on data prior to the Effective Time.

(d) If Parent Company (or its affiliate) has designated any health plan capitation agreements to be assigned to Hospital Newco, then Parent Company hereby assumes all liabilities and responsibilities related to managing the members covered by such health plan capitation agreements, including arranging for services from and payment under all related risk pool arrangements, all related arrangements with independent physician associations (IPAs), all downstream provider agreements, and all out-of-network provider arrangements or services, without regard to whether the member services were rendered prior to or after the Effective Time.

**4.7** The Corporation's Ultimate Control. Notwithstanding anything to the contrary in this <u>Section 4</u> or in this Agreement more generally, the Corporation, as holder of the Licenses, shall remain ultimately responsible for the operation of the Hospital, and may, at any time and from time-to-time during the Management Period, take any action (at Parent Company's or the Manager's sole cost and expense) necessary to ensure the Corporation's compliance with applicable laws and regulations, even if such action requires the Corporation to intervene in the Manager's performance of the Manager's duties or permitted activities pursuant to this <u>Section 4</u>.

# 5. Continued Responsibility of the Corporation.

**5.1** During the Management Period, the Corporation shall maintain (at Parent Company's or the Manager's sole cost and expense), and shall not take or voluntarily permit any actions which may adversely affect, the Corporation's corporate existence and its full rights as the licensee under the Licenses. In addition, during the Management Period, the Corporation and its officers shall reasonably cooperate with the Manager (at Parent Company's or the Manager's sole cost and expense) in the Manager's provision of the Services.

**5.2** Notwithstanding the statutory and regulatory authority and responsibility of the Corporation for the continued management of the Hospital during the Management Period, the Parties recognize and acknowledge that under this Agreement, the Manager shall, subject to the ultimate oversight by and approval of the Corporation, be responsible for the day-to-day operation and maintenance of the Hospital as a general acute care hospital. In the event that any violation or alleged violation of or non-compliance with any statute or regulation applicable to the operation or maintenance of the Hospital as a general acute care hospital certified by the

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 72 of 147

Medicare and Medi-Cal programs occurs during the Management Period, then without regard to legal or statutory fault on the part of the Manager or of the Corporation, the Manager shall immediately notify the Corporation of such violation or alleged violation or non-compliance and take reasonable efforts to avoid or minimize any related adverse consequences. Parent Company and the Manager shall be responsible for the costs of any penalty, fine or remediation identified during the Management Period arising out of or relating to any act, omission, event or occurrence connected with the operation of the Hospital after the Effective Time, including, without limitation, the cost of engaging third party consultants or experts to help address or resolve the violation, alleged violation or non-compliance, and shall indemnify and hold the Corporation harmless for the same in accordance with <u>Section 9.2</u>. The Corporation retains the right to join the Manager in contesting said violations upon providing the Manager with notice of its intent to do so.

**5.3** The Corporation shall be responsible for purchasing drugs and dangerous devices that the Manager determines are necessary for the operation of the Hospital at the Manager's sole cost and expense, as set forth in Section 4.5(b)(i).

**5.4** The Corporation agrees to execute and deliver to the Manager such documents as the Manager may reasonably request to maintain the hospital license active and in good standing with CDPH and the other Licenses necessary or appropriate to maintain the Hospital as a general acute care hospital and to facilitate the Manager's obtaining of the New Licenses.

6. The Parties' Cooperation with Regulatory Agencies. The Manager shall use its best efforts to obtain the New Licenses as expeditiously as possible. The Manager shall provide updates to the Corporation weekly, and as requested by the Corporation, on the status of the Manager's efforts to obtain the New Licenses. To the extent not already submitted prior to the Effective Time, the Manager shall provide a copy of all such initial and supplementary or amended license applications for the New Licenses to Corporation at least two (2) business days prior to submission. The Manager and the Corporation shall cooperate with each other in the event the Corporation requests revisions to a license application for a New License. The Corporation shall, at Parent Company's or the Manager's cost, reasonably cooperate with the Manager's efforts to obtain the New Licenses, and may communicate and coordinate with licensing agencies as necessary in connection with obtaining the New Licenses. Notwithstanding the foregoing, obtaining all governmental consents, approvals, assignments, authorizations, and clearances necessary to obtain the New Licenses shall be solely Parent Company's and the Manager's (and not the Corporation's) responsibility, including payment of any fees, expenses, filing costs or other amounts related thereto.

## 7. Risk of Loss.

7.1 The Corporation assumes no risks or liability for damage to or injury occurring to the Hospital Premises, Assets or the Hospital during the term of this Agreement by any means whatsoever, including fire, storm, earthquake, vandalism, strike, cyberattack, accident or any other casualty (collectively, "Casualty"), and Parent Company shall have all right, title, and interest in and to the proceeds of any insurance it obtained and paid for covering such Casualty.

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 73 of 147

7.2 If, during the term of this Agreement, action is initiated to take the Hospital Premises or any portion thereof by eminent domain proceedings or by deed in lieu thereof (collectively, "Condemnation"), Parent Company, and not the Corporation, shall have all right, title, and interest in and to the award from the Condemnation.

7.3 In the event of a Casualty or Condemnation, neither Parent Company nor the Corporation may terminate this Agreement.

8. Continued Access. Following the expiration or termination of this Agreement and until the entry of final decrees closing the Bankruptcy Cases, if expressly authorized or directed by the Bankruptcy Court, and only to the extent so authorized or directed, the PCO shall have continuing access to the Corporation's Assets and related personnel during normal business hours and upon at least one (1) business day's prior written notice to Parent Company or the Manager, for the purpose of winding down the Corporation's affairs, in connection with any litigation or adversary proceedings, and to effectuate the Plan.

#### 9. Exculpation; Indemnification.

**9.1** The Corporation and the Corporation's affiliates, members, officers, directors, employees, attorneys, accountants, consultants, agents, representatives, successors and assigns, including the liquidating trustee and responsible officer (collectively the "Corporation Indemnified Parties") shall have no liability in contract, tort or otherwise unless and until a Chosen Court finds in a final, non-appealable judgment that any Damages result solely from a Corporation Indemnified Party's gross negligence or willful misconduct.

9.2 Parent Company and the Manager shall promptly and fully keep and hold the Corporation Indemnified Parties forever harmless from, and shall indemnify and defend the Corporation Indemnified Parties from and against, without regard to materiality, any and all obligations, judgments, fines, civil money penalties, sanctions, awards, liabilities, losses, penalties, claims, costs, demands, damages, expenses, liens, and encumbrances, including investigation costs, time spent in depositions and reasonable attorneys' fees and expenses (collectively, "Damages"), whether civil or criminal, direct, indirect or consequential and no matter how arising, in any way related to, connected with, arising or resulting from, or under this Agreement, the APA, the Hired Employees (as defined in the APA), the Manager's performance of the Services, or the operation or management of the Hospital or the Corporation's Assets, in each instance after the Effective Time. Notwithstanding the foregoing, the Parties understand that except as otherwise specifically provided for in the APA, Parent Company and the Manager are not, by virtue of this Agreement or any term or provision herein, assuming any claim, liability, expense, debt or other obligation of the Corporation that both relates to the operation or management of the Hospital or the Corporation's Assets prior to the Effective Time and constitutes an Excluded Liability under the APA.

10. HIPAA Compliance. The Manager agrees to take such steps as are necessary to ensure compliance with the Health Insurance Portability and Accountability Act of 1996 (HIPAA), the California Confidentiality of Medical Information Act, and other applicable federal and state privacy laws and regulations (collectively, the "**Privacy Laws**") with respect to the Hospital and its operations, and the Corporation agrees not to take or voluntarily permit any actions which

violate Privacy Laws with respect to the Hospital or its operations. Toward this end, Parent Company, the Manager, and the Corporation agree to execute and deliver that certain Business Associate Agreement, attached hereto as <u>Exhibit A</u> and incorporated by reference herein, upon execution of this Agreement.

**11. Further Assurances**. Each of the Parties hereto agree to execute and deliver any and all further agreements, documents or instruments necessary to effectuate this Agreement and the transactions referred to herein or contemplated hereby or reasonably requested by any other Party hereto to perfect or evidence their rights hereunder.

12. Relationship of Parties. In performing their duties and permitted activities under this Agreement, Parent Company, the Manager, and the Corporation shall, at all times be acting and performing as independent contractors. Parent Company, the Manager, and the Corporation are not partners or joint venturers with each other and nothing herein shall be construed as making them partners or joint venturers or imposing upon any of them any liability as partners or joint venturers.

**13.** Notices and Demands. All notices and demands, requests, consents, approvals, and other similar communications under this Agreement shall be in writing and shall be sent by personal delivery or by either (a) United States certified or registered mail, return receipt requested, postage prepaid, or (b) Federal Express or similar generally recognized overnight carrier regularly providing proof of delivery, addressed as follows:

If to the Corporation:	St. Francis Medical Center c/o Verity Health System of California, Inc. 601 South Figueroa Street, Suite 4050 Los Angeles, CA 90017-5704 Attention: Chief Executive Officer
With copies to: (which copy shall not constitute notice)	Tania Moyron, Esq. Dentons US LLP 601 South Figueroa St., Suite 2500 Los Angeles, CA 90017-5704 and
	Hope Levy-Biehl, Esq. Davis Wright Tremaine LLP 865 S Figueroa St, Los Angeles, CA 90017
If to Parent Company	

If to Parent Company and/or the Manager:

I

With copies to: (which copies shall not constitute notice)

Any notice so given by mail shall be deemed to have been given as of the date of delivery (whether accepted or refused) established by U.S. Post Office return receipt or the overnight carrier's proof of delivery, as the case may be, whether accepted or refused. Any such notice not so given shall be deemed given upon receipt of the same by the Party to whom the same is to be given. Any Party hereto may designate a different address for itself by notice to the other Party in accordance with this <u>Section 13</u>.

14. Expenses. Except for Parent Company's and the Manager's obligations to be responsible for certain costs, fees, and expenses as set forth elsewhere in this Agreement, each Party to this Agreement shall pay its own expenses in connection with the preparation of this Agreement and the consummation of the transactions contemplated hereby, including the fees of any attorneys, accountants, financial advisors, investment bankers or other professionals engaged by such Party.

**15.** Entire Agreement. This Agreement, the Leaseback Agreement, and those provisions of the APA expressly identified in this Agreement, contain the entire agreement among the Parties with respect to the subject matter hereof and supersede all prior agreements, arrangements, and understandings relating to the subject matter hereof and thereof. There are no written or oral agreements, understandings, representations, or warranties among the Parties other than those set forth in this Agreement, the Leaseback Agreement, and those provisions of the APA expressly identified in this Agreement. Nothing in this Agreement modifies or shall be construed as modifying any orders entered by the Bankruptcy Court.

**16. Amendment**. This Agreement may not be modified, amended, altered or supplemented except by a written agreement executed by all the Parties.

17. Waiver. Waiver by any Party of any breach or failure to comply with any provision of this Agreement by any other Party shall not be construed as or constitute, a continuing waiver of such provision, or a waiver of any other breach of, or failure to comply with, any other provision of this Agreement. No waiver of any such breach or failure or of any term or condition of this Agreement shall be effective unless in a written notice signed by the waiving Party und delivered, in the manner required for notices generally, to each affected Party.

18. Severability. In case any provision of this Agreement shall be found by a court of competent jurisdiction to be invalid, illegal or unenforceable, such provision shall be construed and enforced as if it had been narrowly drawn so as not to be invalid, illegal or unenforceable, and the validity, legality and enforceability of the remaining provisions of this Agreement shall not in any way be affected or impaired thereby.

**19.** Successors and Assigns. This Agreement shall be binding upon, and shall inure to the benefit of, the Parties and their respective permitted successors and assigns. The Corporation shall not be permitted to assign its rights or its obligations under this Agreement without the

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 76 of 147

prior consent of Parent Company and the Manager. The Parties further acknowledge and agree that the Manager may subcontract for any of the goods or services required to be provided by the Manager pursuant to this Agreement, and the Manger may assign any of its rights hereunder and/or delegate any of its obligations hereunder, so long as in each case the Manager remains responsible for such subcontracted goods or services and for any of such Manager obligations hereunder.

**20.** Attorneys' Fees. In the event of any litigation or arbitration between the Parties arising out of this Agreement, the prevailing Party therein shall be allowed to recover from the other Party all court costs and reasonable attorneys' fees which shall be fixed by the court or arbitrator.

**21. Headings**. The descriptive headings of sections and subsections of this Agreement are inserted for convenience only and do not constitute a part of this Agreement and shall not affect in any way the meaning or interpretation of this Agreement.

**22. Governing Law**. This Agreement shall be governed by and construed in accordance with the laws of the State of California without regard to its conflicts of laws principles or decisions, except to the extent that the laws of such State are superseded by the Bankruptcy Code or other applicable federal law.

23. Jurisdiction. The Parties agree that the Bankruptcy Court shall have exclusive personal and subject matter jurisdiction over any action or proceeding with respect to any claim arising out of or related to this Agreement, and any of the documents or transactions contained in or contemplated by this Agreement; provided, however, that the Parties agree that the United States District Court for the Central District of California (together with the Bankruptcy Court, the "Chosen Courts") shall have exclusive jurisdiction over such claim if (i) the Bankruptcy Cases are closed and if the Bankruptcy Cases are not reopened to adjudicate such claim after request by the Party bringing such claim or (ii) the Bankruptcy Court determines that it does not have jurisdiction over such claim. Solely in connection with claims arising under this Agreement, or any of the documents or transactions contemplated hereby, the Parties (a) irrevocably submit to the exclusive jurisdiction of the Chosen Courts, (b) waive any objection to laying venue in any such action or proceeding in the Chosen Courts, (c) waive any objection that the Chosen Courts are an inconvenient forum or do not have jurisdiction over any Party hereto, and (d) agree that service of process upon such Party in any such action or proceeding shall be effective if notice is given in accordance with Section 13 hereof.

**24. Commencement**. The Parties acknowledge that commencement of the Manager's Services under this Agreement is subject to and contingent upon the occurrence of the Closing (as defined in the APA). The Management Period shall not commence, and the Parties' obligations during the Management Period shall not commence, unless and until the Closing has occurred.

**25.** Cooperation on Regulatory Compliance Matters. Parent Company and the Manager understand and acknowledge that the Corporation intends to comply with applicable federal and state laws, regulations, and guidance. In the event the terms of this Agreement need to be amended or supplemented based on guidance from or at the request or direction of a regulator made during the term of this Agreement, Parent Company and the Manager shall cooperate with

such amendment and/or supplement to ensure the Corporation's ability to comply with such guidance, request or directive.

26. Counterparts. This Agreement may be executed by one or more of the Parties on any number of separate counterparts (including by telecopy), and all of said counterparts taken together shall be deemed to constitute one and the same instrument. It may be delivered by facsimile or electronic transmission, including by e-mail as a PDF, and facsimile or PDF copies of executed signature pages, which shall be binding as originals.

27. WAIVER OF JURY TRIAL. TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT, OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (B) EACH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (C) EACH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (D) EACH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 27. THIS IS AN AFFIRMATIVE WAIVER OF THE PARTIES' RIGHTS TO A JURY TRIAL UNDER CALIFORNIA LAW, Cal. C. Civ. Pro. Sec 631. BY SIGNING BELOW ON THE SIGNATURE LINES, EACH PARTY IS EXPLICITLY WAIVING JURY TRIAL AND AUTHORIZING ANY AND ALL PARTIES TO FILE THIS WAIVER WITH ANY COURT AS THE WAIVER REQUIRED UNDER Cal. C. Civ. Proc. Sec. 631(f)(2).

[Signature page follows.]

**IN WITNESS WHEREOF,** the Parties agree to the foregoing terms of agreement through the execution below by their respective, duly authorized representatives as of the Signing Date.

#### **MANAGER:**

## **CORPORATION:**

NAME

By:	
Name:	
Title:	

ST. FR	ANCIS MEDICAL CENTER,
a Calif	ornia non-profit public benefit
corpora	ation

By:	
Name:	
Title:	

#### **PARENT COMPANY:**

NAME

By:	
Name:	
Title: _	

## **HOSPITAL NEWCO:**

NAME

By:	
Name:	
Title:	

## **Exhibit A to IMA**

**Business Associate Agreement** 

## Schedule 3.4 to IMA

# **Corporation's Billing Credentials**

Schedule 3.6 to IMA

**Retained Contracts** 

# Exhibit 1.4(a) to APA

## Form of Bill of Sale

See attached.

# Exhibit 1.4(b) to APA

## Form of Real Estate Assignment Agreements

See attached.

# Exhibit 1.4(c) to APA

## Form of Quitclaim Deed

See attached.

# Exhibit 1.4(d) to APA

## Form of Transfer Agreement

See attached.

# Exhibit 1.4(f) to APA

## Form of Transition Services Agreements

See attached.

4" = "1" "DM\_US 165807051-6.079742.0063" "" US\_Active\114483166\V-4 DM\_US 165807051-13.079742.0063 US\_Active\114563955\V-1

I

#### **TRANSITION SERVICES AGREEMENT**

This Transition Services Agreement (the "Agreement"), is made and entered into as of \_\_\_\_\_\_, 2020 (the "Signing Date"), by and among Verity Health System of California, Inc., a California nonprofit public benefit corporation ("Verity"), Verity Holdings, LLC, a California limited liability company ("Verity Holdings"), St. Francis Medical Center, a California nonprofit public benefit corporation ("St. Francis" and together with Verity and Verity Holdings, collectively the "Sellers" and each individually a "Seller") and \_\_\_\_\_\_

("**Purchaser**"). The Sellers, the Liquidating Trust, and Purchaser may each be referred to herein as a "**Party**" and collectively as the "**Parties**."

This Agreement is being entered into in connection with the Asset APA, dated of even date herewith among Sellers and Purchaser (the "APA"). Capitalized terms used in this Agreement but not defined herein shall have the meanings given to them in the APA. The Parties have agreed to provide or cause one or more of their respective direct or indirect subsidiaries or affiliates to provide certain services as set forth below. The Party providing services hereunder (together with those of its direct and indirect subsidiaries and affiliates providing services hereunder) is referred to herein as the "Service Provider" and the Party receiving services hereunder (together with those of its direct and indirect subsidiaries and affiliates receiving services hereunder) is referred to herein as the "Service Recipient". The Parties agree that this Agreement will be amended to include the Liquidating Trust (as defined below) after its creation, pursuant to the effective date set forth in the plan of liquidation (the "Plan") to be confirmed by the Bankruptcy Court.

#### RECITALS

A. On August 31, 2018, the Sellers and certain of their affiliates each filed a voluntary petition for relief (collectively, the "**Bankruptcy Cases**") under chapter 11 of title 11 of the United State Code (the "**Bankruptcy Code**").

B. The Bankruptcy Cases are jointly administered under Lead Case No. 18-20151 and are currently pending in the Bankruptcy Court for the Central District of California in Los Angeles (the "**Bankruptcy Court**").

C. Purchaser and the Sellers entered into the APA, which provides for the sale of the assets of the Sellers (collectively, the "Assets").

D. Sellers will file a motion with the Bankruptcy Court to approve the APA and upon approval, the Bankruptcy Court will enter a sale order (the "**Sale Order**").

E. Sellers and certain of their affiliates will file a chapter 11 Plan, pursuant to which a liquidating trust (the "**Liquidating Trust**") will be created on the effective date of the Plan after confirmation thereof.

F. The primary purpose of the Liquidating Trust is to, among other things, liquidate and distribute assets, which includes assets of the Sellers not sold to Purchaser pursuant to the APA.

G. Following the Closing Date, the Sellers and the Liquidating Trust need access to certain assets, systems, facilities, equipment, and personnel of Purchaser in connection with the Sellers' wind-down of the Businesses, the completion of the Bankruptcy Cases, and the dissolution of the Sellers, and Purchaser agrees to provide such access and services in connection therewith.

#### AGREEMENT

**NOW, THEREFORE,** in consideration of the promises and the mutual covenants contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby agree as follows:

#### ARTICLE I SERVICES

Section 1.1 <u>Services to be Provided</u>.

(a) <u>Services to be Provided by the Sellers and/or the Liquidating Trust</u>. Subject to the terms and conditions of this Agreement, the Sellers and/or the Liquidating Trust, as the Service Provider, shall provide or cause to be provided to Purchaser, as the Service Recipient, those services in support of the business of Purchaser on such terms (including payment) as may be mutually agreed upon in writing from time to time by the Sellers and/or the Liquidating Trust and Purchaser (the "Seller Services").

(b) <u>Services to be Provided by Purchaser</u>. Subject to the terms and conditions of this Agreement, Purchaser, as the Service Provider, shall provide or cause to be provided to the Sellers and the Liquidating Trust, as the Service Recipient, those services and personnel in support of the business of the Sellers and/or the wind-down and liquidation of the Sellers set forth on <u>Schedule A</u> attached hereto (the "**Purchaser Services**").

Section 1.2 Access to and Use of Assets, Systems, Facilities, and Equipment.

(a) Access. Beginning on the Effective Time and continuing through the termination or expiration of this Agreement in accordance with its terms, Purchaser hereby grants to the Sellers and the Liquidating Trust, reasonable access to and the right to use the hospital assets, systems, facilities, and equipment to the extent reasonably required in connection with the Sellers' wind-down of the Businesses, the completion of the Bankruptcy Cases, and the dissolution of the Sellers. If any third party to any lease, license or other agreement requires Sellers to sign any documents in order for Sellers to be granted access to, and the right to use, any hospital asset, system, facility or equipment under such lease, license or other agreement, Sellers agree to execute and deliver any such documents; provided, that such documents, leases, licenses or other agreements do not require Sellers to pay any money or take any action, not required by the Sale Order; and provided, further, that if Sellers do not sign such requisite documents, leases, licenses or other agreements, then Purchaser has no obligation to provide Sellers with access and

the right to use the associated hospital assets, systems, facilities and equipment provided by such third party; and provided, further, that Purchaser agrees to indemnify, defend and hold harmless each of the Sellers in connection with any obligations under any such documents, leases, licenses or other agreements other than those obligations of Sellers thereunder which are required by the Sale Order.

**(b)** Office Space. In order to enable the provision of the Seller Services, Purchaser hereby grants to Sellers' and their affiliates' employees and any third-party service providers or subcontractors the right to use and occupy the portion of St. Francis Medical Center identified on Schedule B attached hereto (the "Premises") at no cost for use exclusively as office space pursuant to the terms and conditions hereof, and until the termination of this Agreement pursuant to Article IV herein. Sellers shall keep the Premises neat, clean and sanitary and dispose of all garbage. Sellers may terminate its right to occupy the Premises at any time upon five (5) days prior written notice to Purchaser. Upon termination of Sellers' right to occupy the Premises, Sellers shall vacate the Premises in broom-clean condition. Purchaser shall make the Premises available to Sellers 24 hours a day, 7 days a week. Purchaser may access the Premises for purposes of making repairs or improvement or other reasonable purposes upon (except in the case of an emergency) reasonable prior written notice to Sellers; however, Purchaser shall at all times use its commercially reasonable efforts to prevent interference with or interruption of the Sellers' business activities on the Premises. The Sellers use of the Premises shall additionally be subject to the following terms and conditions:

(i) Purchaser shall have the right, on reasonable advance notice (which shall not be less than three (3) business days) to Sellers, to relocate the space covered by the preceding license to similar space in the same area as the original designated Premises, provided the replacement space is reasonably suitable for the same uses as contemplated herein. In the event of any such relocation, references herein to the Premises shall constitute references to such relocated space.

(ii) Sellers shall not make any material alterations, decorations, additions, or improvements in or to the Premises or the Purchaser's other property without the prior written consent of Purchaser, which may be given or withheld in Purchaser's sole discretion.

(iii) Sellers shall not do, or permit their representatives to do, anything that would: (a) unreasonably obstruct or materially interfere with the lawful operation of the Premises or other portions of the Purchaser's property by Purchaser and other tenants of the Purchaser's property; or (b) use the Premises for any illegal or unlawful purpose, including without limitation any violations of applicable laws or regulations relating to or regulating hazardous substances.

(iv) The license, occupancy and access rights granted to Sellers under this Agreement shall be effective notwithstanding any ground lease, mortgage, deed of trust, or any other hypothecation or security placed at any time upon the Premises or any of Purchaser's property of which the Premises is a part, and notwithstanding any change in ownership (including any sale and/or leaseback involving any property of which the Premises is a part).

**Section 1.3** <u>Subcontracting</u>. The Service Provider may, in its sole discretion and without any written notice to the Service Recipient engage one or more parties (including third parties or

affiliates of the Service Provider) to provide the Services, whether in whole or in part or subcomponent. Without limiting the preceding, Purchaser will not be required to use the Sellers existing IT and similar systems, but shall have the right to replace IT and other systems, used to provide any of the Purchaser Services provided that the new or replacement systems provide equivalent functionality. Sellers and Purchaser agree to fully cooperate with each other, at Purchaser's sole cost and expense, in connection with any data or other transition steps required in connection moving any of the functions covered by the Purchaser Services to a new vendor or system.

**Section 1.4** <u>Cooperation; Access</u>. Each Party shall, at reasonable times under the circumstances, make available to any other Party properly authorized personnel for the purpose of consultation and decisions in connection with the Services.

#### Section 1.5 <u>Certain Qualifications</u>.

(a) <u>Provided AS-IS</u>. By accepting the use to the Premises and other access rights provided herein, Sellers conclusively shall be deemed, subject to the representations, warranties and obligations expressly set forth in this Agreement, to have (i) accepted the Premises and access to other property of Purchaser "AS IS" and without express or implied warranty; and (ii) is fully aware of and has approved the Premises in its condition existing as of such date.

(b) <u>Limited Representation</u>. Except for the representations, warranties and obligations expressly set forth in this Agreement, Purchaser makes no representations and warranties of any kind, implied or expressed, with respect to the Purchaser Services, including, without limitation, no warranties of merchantability or fitness for a particular purpose, which are specifically disclaimed.

(c) <u>Privacy and Confidentiality Protections</u>. All access provided to Sellers, to the Premises or otherwise as provided herein, shall be permitted only in a manner which protects patient privacy and the confidentiality of patient information under applicable Privacy Laws.

(d) <u>Limitation on Liability</u>. In no event shall any Party have any liability under any provision of this Agreement for any punitive, incidental, consequential, special or indirect damages, whether based on statute, contract, tort or otherwise, and whether or not arising from any other Party's sole, joint, or concurrent negligence, strict liability, criminal liability or other fault.

#### ARTICLE II LICENSE GRANT

**Section 2.1** <u>License Grant</u>. Beginning on the Effective Time and continuing through the later of the termination or expiration of this Agreement in accordance with its term or the dissolution of the Sellers and their Businesses, the wind-down of the Liquidating Trust and the completion of the Bankruptcy Cases, Purchaser grants to the Sellers and the Liquidating Trust a royalty free, license to use the name "St. Francis Medical Center," including any trademarks, service marks, trademark and service mark registrations and registration applications, trade name registrations, logos, domain names, trade dress, copyrights, copyright registrations, website

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 91 of 147

content, know-how, trade secrets, related to the use of the preceding name, and the corporate or company names of the Sellers and the name of the Hospital, together with all rights to sue and recover damages for infringement, dilution, misappropriation or other violation or conflict associated with any of the foregoing, in all respects solely as reasonably needed in connection with the wind-down of the Businesses, the completion of the Bankruptcy Cases, and the dissolution of the Sellers. Sellers and the Liquidating Trust shall not assign, transfer or hypothecate any of the rights subject to, and pursuant to, the preceding license and shall not use any such license rights in any manner which could adversely affect Purchaser's intellectual property rights with respect to the items which are the subject of such license.

## ARTICLE III PAYMENT

**Section 3.1** <u>Service Fees</u>. The fee, rate or amount to be charged to the Service Recipient for the Services (the "Service Fees") shall be as set forth on <u>Schedule A</u> hereto for the Purchaser Services and as agreed upon in writing from time to time by Sellers and/or the Liquidating Trust and Purchaser for the Purchaser Services. The Service Provider shall issue invoices on a monthly basis to the Service Recipient for the Services, with a statement of such Services and reasonable supporting detail therefor. The Service Recipient shall pay the undisputed portion of such invoices within thirty (30) days of receipt of such invoices. Any dispute relating to the Services shall be resolved in accordance with the dispute resolution procedures set forth in this Agreement.

## ARTICLE IV TERM AND TERMINATION

Section 4.1 Term. Unless a Service is earlier terminated by the Service Recipient in accordance with Section 4.2(b), the Service Provider shall provide each Service to the Service Recipient for the period beginning at the Effective Time (as defined under the APA) and continuing through and including the applicable dates set forth in: (i) Schedule A (with respect to the Purchaser Services); and (ii) any written agreement between Sellers and/or the Liquidating Trust and Purchaser (with respect to the Seller Services) (the "Applicable Term"); provided, that in the event a particular Service provided by Purchaser hereunder is required by one or more Sellers to allow such Seller(s) to comply with applicable laws, rules or regulations, the Applicable Term shall be extended for so long as such Service is required to allow such Seller(s) to comply with such laws, rules and regulations; provided, however, that notwithstanding any other term herein any Party may, in any case, terminate this Agreement by written notice to any other Party at any time on and after the end of the 18<sup>th</sup> calendar month following the Licensure Date (as defined in the APA). In addition, upon termination or expiration of this Agreement, the Parties may mutually agree in writing to a tail period for any portion of the Services, and the terms of this Agreement shall continue to apply to such ongoing Services until performed in full.

## Section 4.2 <u>Termination</u>.

(a) <u>Termination for Material Breach</u>. This Agreement may be terminated by the Sellers and/or the Liquidating Trust on the one hand, or Purchaser on the other hand, upon written notice to any other Party in the event that any other Party materially breaches any

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 92 of 147

provision of this Agreement and fails to remedy such breach within thirty (30) days of its receipt of such written notice, except to the extent such remedy is prevented or not permitted by the Bankruptcy Court; and provided, further, however, that this Agreement may only be terminated under this Section 4.2(a) following approval of such termination by the Bankruptcy Court.

(b) <u>Early Termination</u>. The Service Recipient may terminate any particular Service early by providing not less than thirty (30) days prior written notice to the Service Provider of its intent to terminate early such particular Service. Any such partial termination notice delivered shall specify in detail (i) the Service or Services to be terminated and (ii) the effective date(s) of such termination (subject to the foregoing sentence). If any such termination of a Service prior to the expiration of its Applicable Term requires any out-of-pocket fee, cost or expense of a similar nature payable to a third party (a "**Termination Fee**"), the Service Provider shall have no obligation to terminate such Service prior to the expiration of its Applicable Term unless the Service Recipient approves in advance such Termination Fee in writing (a "**Termination Fee Approval**"). If the Termination Fee Approval is granted, the Termination Fee shall be billed to and payable by the Service Recipient.

(c) <u>Continued Access</u>. Notwithstanding the termination of any particular Services hereunder, Service Recipient shall continue to have the right, upon reasonable advance notice and consistent with the other provisions herein, to reasonable access data and Service Provider, personnel relating to such Services for reasonable purposes (including compliance with applicable law).

**Section 4.3** Effect of Termination. The termination of this Agreement pursuant to <u>Section 4.2</u> shall not relieve any Party from liability for any willful and material breach of this Agreement prior to such termination. Further, the provisions of <u>Article II</u>, <u>Article III</u>, this <u>Section 4.3</u>, <u>Article V</u> and <u>Article VI</u> shall survive any termination of this Agreement pursuant to <u>Section 4.2</u>.

## ARTICLE V COORDINATION AND OTHER COVENANTS

**Section 5.1** <u>Coordination</u>. The Parties hereby appoint as of the Effective Time the following representatives responsible for coordinating and managing the provision and receipt of the Services, which representatives shall have authority to act on the applicable Party's behalf with respect to matters relating to this Agreement:

Sellers' Representative:

Address:

Telephone:

Facsimile:

Email:

Liquidating Trust Representative:
Address:
Telephone:
Facsimile:
Email:
Purchaser's Representative:
Address:
Telephone:
Facsimile:
Email:

In the event that the Sellers, the Liquidating Trust or Purchaser elect, in their sole discretion, to change their respective representative appointed for purposes of this <u>Section 5.1</u>, the Sellers, the Liquidating Trust or Purchaser (as the case may be) shall deliver written notice thereof to the other promptly following the effective date of appointment of the name and contact information of such new representative. The representatives shall work in good faith to address and attempt to resolve any dispute arising out of or relating to this Agreement or the breach, termination or validity thereof, whether sounding in contract or tort and whether arising during or after termination of this Agreement, including any pricing and other Service related matters.

General Cooperation and Turnover Obligations. The Parties shall cooperate to Section 5.2 ensure that any and all payments that constitute Excluded Assets (as defined in the APA), shall be paid to and received by Sellers, with any payments that constitute Assets (as defined in the APA) transferred to Purchaser transferred to Purchaser pursuant to Section 1.7 of the APA or that otherwise arise from services rendered by Purchaser on or after the Licensure Date be paid to and received by Purchaser. In this regard, for a period of two (2) years after the Effective Time ("Turnover Period"), the Parties shall, within ten (10) business days, copy and send to any other Party copies (either in hard copy or via electronic file) of all remittance advices for all deposits to all Lockboxes or other bank accounts for the Receivables (as defined in the APA), from whatever payor or source of funds, that are received on and after the Effective Time. In the event that Assets are deposited to a bank account of Sellers which is not automatically swept or transferred to Purchaser, then Sellers, within ten (10) business days of notice of the receipt of such Assets, shall turnover and pay Purchaser said funds. In the event that a deposit representing payment of any Excluded Assets is received by Purchaser, then Purchaser, within five (5) business days of notice of the receipt of funds representing any Excluded Assets, shall turnover and pay Sellers such funds. Each Party, moreover, shall have the right, within three (3) months after the expiration of the Turnover Period, to audit by an independent and competent auditor, at the

requesting Party's sole expense, of the bank records and remittance advices of any other Party. Thereafter, upon the findings of the auditor that there has either been an overpayment or an underpayment of funds due, the Party owing funds shall, within five (5) business days, make a payment of such funds to the Party to whom they are owed.

**Section 5.3** <u>HIPAA Compliance</u>. The Parties agrees to take such steps as are necessary to ensure compliance with the Health Insurance Portability and Accountability Act of 1996 ("**HIPAA**"), the California Confidentiality of Medical Information Act, and other applicable federal and state privacy laws and regulations (collectively, the "**Privacy Laws**") with respect to the Hospital and its operations, and the performance by the Parties of the services and obligations provide herein.

#### Section 5.4 Confidentiality.

During the term of this Agreement and thereafter, the Parties shall, and shall (a) instruct their respective Representatives (as defined below) to, maintain in confidence and not disclose any other Party's financial, technical, sales, marketing, development, personnel, and other information, records, or data, including, without limitation, customer lists, supplier lists, trade secrets, designs, product formulations, product specifications or any other proprietary or confidential information, however recorded or preserved, whether written or oral (any such information, "Confidential Information"). Each Party hereto shall use the same degree of care, but no less than reasonable care, to protect any other Party's Confidential Information as it uses to protect its own Confidential Information of like nature. Unless otherwise authorized in any other agreement between the Parties, any Party receiving any Confidential Information of any other Party (the "Receiving Party") may use Confidential Information only for the purposes of fulfilling its obligations under this Agreement (the "Permitted Purpose"). Any Receiving Party may disclose such Confidential Information only to its directors/managing members, managers, officers, employees, consultants, financial advisors, prospective or existing lenders, counsel, accountants and other agents of such Receiving Party (and, in addition, Sellers may disclose such Confidential Information to the creditors committee in the Bankruptcy Cases and its consultants, financial advisors, counsel, accountants and other agents, the Liquidating Trust and its trustee, consultants, financial advisors, counsel, accountants and other agents and any advisory board members and their consultants, financial advisors, counsel, accountants and other agents) ("Representatives") who have a need to know such information for the Permitted Purpose and who have been advised of the terms of this Section 5.5 and the Receiving Party shall be liable for any breach of these confidentiality provisions by such persons; provided, however, that any Receiving Party may disclose such Confidential Information to the extent such Confidential Information is required to be disclosed by a governmental agency or court order ("Governmental Order"), in which case the Receiving Party shall promptly notify, to the extent possible, the disclosing Party (the "Disclosing Party"), and take reasonable steps to assist in contesting such Governmental Order or in protecting the Disclosing Party's rights prior to disclosure, and in which case the Receiving Party shall only disclose such Confidential Information that it is advised by its counsel in writing that it is legally bound to disclose under such Governmental Order.

(b) Notwithstanding the foregoing, "Confidential Information" shall not include any information that the Receiving Party can demonstrate: (i) was publicly known at the time of

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 95 of 147

disclosure to it, or has become publicly known through no act of the Receiving Party or its Representatives in breach of this <u>Section 5.5</u>; (ii) was rightfully received from a third party without a duty of confidentiality; or (iii) was developed by it independently without any reliance on the Confidential Information.

(c) Upon demand by the Disclosing Party at any time, or upon expiration or termination of this Agreement with respect to any Service, the Receiving Party agrees promptly to return or destroy, at the Receiving Party's option, all Confidential Information. If such Confidential Information is destroyed, an authorized officer of the Receiving Party shall certify to such destruction in writing.

#### ARTICLE VI MISCELLANEOUS

Section 6.1 <u>Relationship of the Parties</u>.

(a) The Parties and their respective representatives shall be deemed independent contractors for all purposes under this Agreement.

(b) This Agreement shall not be deemed or construed to create the relationship of employer or employee, partnership or any type of joint venture relationship among the Parties.

(c) The Parties acknowledge and agree that the Parties are not providing legal, accounting or tax advice under this Agreement. The Parties further acknowledge and agree that no fiduciary or other similar relationship is being created among the Parties relating to the Services or otherwise under this Agreement.

(d) Except as expressly set forth herein, neither the Sellers, the Liquidating Trust, nor any of their respective representatives or affiliates, on the one hand, nor Purchaser or any of its representatives or affiliates, on the other hand, shall have the authority to contract for or assume obligations of any nature in the name of any other Party, as the case may be, without the other's prior written consent.

**Section 6.2** <u>No Set-off</u>. Neither the Sellers, the Liquidating Trust, nor any of their respective representatives or affiliates, on the one hand, nor Purchaser nor any of its representatives or affiliates, on the other hand, shall have any set-off or other similar rights pursuant to this Agreement.

**Section 6.3** <u>Notices</u>. All notices, requests, claims, demands or other communications hereunder shall be deemed to have been duly given and made if in writing and (a) at the time personally delivered if served by personal delivery upon the Party for whom it is intended, (b) at the time received if delivered by registered or certified mail (postage prepaid, return receipt requested) or by a national courier service (delivery of which is confirmed), or (c) upon confirmation if sent by facsimile; in each case to the Party at the address set forth in <u>Section 5.1</u>, or such other address as may be designated in writing hereafter, in the same manner, by such Party.

**Section 6.4** <u>Amendment, Modification and Waiver</u>. Any provision of this Agreement may be amended, modified or waived if, and only if, such amendment, modification or waiver is in writing and signed, in the case of an amendment, by the Parties, or in the case of a waiver, by the Party against whom the waiver is to be effective. No failure or delay by any Party in exercising any right, power or privilege hereunder shall operate as a waiver thereof nor shall any single or partial exercise thereof preclude any other or further exercise thereof or the exercise of any other right, power or privilege.

**Section 6.5** <u>Assignment</u>. Neither this Agreement nor any of the rights, interests or obligations under it may be directly or indirectly assigned, delegated, hypothecated, sublicensed, subleased or transferred by any of the Parties, in whole or in part, to any other person without the prior written consent of the non-assigning Party, and any attempted or purported assignment in violation of this <u>Section 6.5</u> will be null and void; provided, that Sellers may assign their rights hereunder to the Liquidating Trust. Subject to the preceding sentence, this Agreement will be binding upon, inure to the benefit of, and be enforceable by the Parties and their respective successors and permitted assigns. Notwithstanding the preceding, however, Purchaser may assign its rights or obligations hereunder to its any of its affiliates and may subcontract with third parties to provide any of the Purchaser Services as provided elsewhere in this Agreement, provided that any such assignment or subcontracting shall not relieve Purchaser of or reduce Purchaser's obligations under this Agreement.

**Section 6.6** <u>Entire Agreement</u>. This Agreement constitutes the entire agreement among the Parties with respect to the subject matter hereof and supersedes all prior agreements and understandings, oral or written, with respect to such matters.

**Section 6.7** <u>Third Party Beneficiaries</u>. Nothing expressed or implied in this Agreement is intended to confer any rights, benefits, remedies, obligations or liabilities upon any person other than the Parties and their respective successors and permitted assigns.

**Section 6.8** <u>Expenses</u>. In the event of litigation arising from or relating to this Agreement, the non-prevailing Party as determined by a court of competent jurisdiction in a final non-appealable order shall be liable and pay to the prevailing Party the reasonable attorney's fees and expenses incurred in connection with such litigation.

**Section 6.9** <u>Governing Law</u>. This Agreement shall be construed, performed, and enforced in accordance with, and governed by, the laws of the State of California (without giving effect to the principles of conflicts of laws thereof), except to the extent that the laws of such State are superseded by the Bankruptcy Code or other applicable federal law.

Section 6.10 WAIVER OF JURY TRIAL. TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT, OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (B) EACH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (C) EACH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (D) EACH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 6.10. THIS IS AN AFFIRMATIVE WAIVER OF THE PARTIES' RIGHTS TO A JURY TRIAL UNDER CALIFORNIA LAW, Cal. C. Civ. Pro. Sec 631. BY SIGNING BELOW ON THE SIGNATURE LINES, EACH PARTY IS EXPLICITLY WAIVING JURY TRIAL AND AUTHORIZING ANY AND ALL PARTIES TO FILE THIS WAIVER WITH ANY COURT AS THE WAIVER REQUIRED UNDER Cal. C. Civ. Proc. Sec. 631(f)(2).

**Section 6.11** Jurisdiction; Mediation. The Parties agree that the Bankruptcy Court shall have exclusive personal and subject matter jurisdiction over any action or proceeding with respect to any claim arising out of or related to this Agreement, and any of the documents or transactions contained in or contemplated by this Agreement; *provided, however*, that the Parties agree that the United States District Court for the Central District of California (together with the Bankruptcy Court, the "Chosen Courts") shall have exclusive jurisdiction over such claim if (i) the Bankruptcy Cases are closed and if the Bankruptcy Cases are not reopened to adjudicate such claim after request by the Party bringing such claim or (ii) the Bankruptcy Court determines that it does not have jurisdiction over such claim. Solely in connection with claims arising under this Agreement, or any of the documents or transactions contemplated hereby, the Parties (a) irrevocably submit to the exclusive jurisdiction of the Chosen Courts, (b) waive any objection to laying venue in any such action or proceeding in the Chosen Courts, (c) waive any objection that the Chosen Courts are an inconvenient forum or do not have jurisdiction over any Party hereto, and (d) agree that service of process upon such Party in any such action or proceeding shall be effective if notice is given in accordance with <u>Section 6.3</u> hereof.

**Section 6.12** <u>Mediation</u>. If a dispute arises out of or relates to this contract, or the breach thereof, and if the dispute cannot be settled through negotiation, the Parties agree first to try in good faith to settle the dispute by mediation administered by the American Arbitration Association under its Commercial Mediation Procedures before resorting to litigation in accordance with the terms of this Agreement. Any such mediation shall take place in Los Angeles, California and be subject to the governing law as set forth in this Agreement.

**Section 6.13** <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, each of which will be deemed to constitute an original, but all of which shall constitute one and the same agreement, and may be delivered by facsimile or other electronic means intended to preserve the original graphic or pictorial appearance of a document.

**Section 6.14** <u>Ownership of Intellectual Property</u>. Subject to the provisions of the APA, each of the Parties shall retain all right, title and interest in and to their respective intellectual property and data and other proprietary and other assets.

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 98 of 147

**Section 6.15** <u>Service Standards</u>. Service Provider shall use commercially reasonable efforts to provide the Services to be provided by it in accordance with all applicable laws, rules and regulations and in a professional, competent, workman-like and timely manner.

**Section 6.16** Severability. The provisions of this Agreement shall be deemed severable and the invalidity or unenforceability of any provision shall not affect the validity or enforceability of the other provisions hereof. If any provision of this Agreement, or the application thereof to any circumstance, is found by a court or other governmental authority of competent jurisdiction to be invalid or unenforceable, the remainder of this Agreement and the application of such provision to other circumstances shall not be affected by such invalidity or unenforceability, nor shall such invalidity or unenforceability affect the validity or enforceability of such provision, or the application thereof, in any other jurisdiction. If any provision of this Agreement is so broad as to be unenforceable, the provision shall be interpreted to be only so broad as would be enforceable.

**Section 6.17** Force Majeure. No Party shall be liable for nonperformance or late performance of any of its obligations under this Agreement to the extent and for such periods of time as such nonperformance, or late performance is due to reasons outside such Party's control (excludingincluding any arising from the SARS-CoV-2 virus or mutations therefrom or in connection with the disease COVID-19), including acts of God, war (declared or undeclared), riots, revolutions, fire, floods, explosions, sabotage, nuclear incidents, lightning, weather, earthquakes, storms, sinkholes, epidemics, failure of utilities, (or similar nonperformance or late performance or late performance, the Service Provider shall use commercially reasonable efforts to restore or resume the Services within a reasonable period of time.

**Section 6.18** <u>Fair Meaning</u>. This Agreement shall be construed according to its fair meaning and as if prepared by all Parties.

**IN WITNESS WHEREOF**, the Parties have executed this Agreement as of the Signing Date.

NAME	VERITY HEALTH SYSTEM OF
	CALIFORNIA, INC.
By:	
	By:
Name:	
	Name:
Title:	
	Title:
NAME	
D	VERITY HOLDINGS, LLC
By:	P
	By:
Name:	
	Name:
Title:	<b>T</b> '41
	Title:

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 99 of 147

NAME	ST. FRANCIS MEDICAL CENTER
By:	Ву:
Name:	Name:
Title:	Title:

#### Schedule A to TSA

#### Services to be Provided by Purchaser to Sellers

I. Phase One: [from closing date to 6-9 months thereafter]

A. Description of Services: Purchaser will assist in activities related to the Debtors' efforts in support of its winding down activities, including accounting for monthly transactions, assisting in materials management and purchasing, facilitating voucher and payment of accounts payable and financial reporting. Purchaser will assist Debtors' in additional winding down activities, including facilitating the Debtors' efforts in claims reconciliation, contracts rejections/assumptions, preparing W2s and 1099s, and preparing tax returns.

- B. Accounting TBD
- C. Accounts Payable / Supply Chain TBD
- D. Human Resources / Payroll TBD

E. Information Technology: Provide application and infrastructure support; maintain licensing; maintain user access and perform required application and operating system upgrades as needed to perform services for all applications. List of applications: - TBD

II. Phase Two: [from end of Phase I to two years thereafter]

A. Description of Services: Purchaser will continue to archive and provide access to select data.

- B. Accounting / Finance TBD
- C. Human Resources / Payroll TBD

D. Information Technology: Provide application and infrastructure support; maintain licensing; maintain user access and perform required application and operating system upgrades as needed to perform services for all applications. List of applications: - TBD

## Schedule B to TSA

Office Space for Use by Sellers

## Exhibit 5.95.8(c) to APA

# Accepted Conditions (Attorney General)

See attached.

I

# Conditions to the Sale of St. Francis Medical Center<sup>1</sup> and Approval of the Asset Purchase Agreement by and among Verity Health System of California, Inc., Verity Holdings, LLC, St. Francis Medical Center and Prime Healthcare Services, Inc. ("Purchaser")

These Conditions shall be legally binding on Verity Health System of California, Inc., a California nonprofit public benefit corporation, Verity Holdings, LLC, a California limited liability company, St. Francis Medical Center, a California nonprofit public benefit corporation, and St. Francis Medical Center Foundation, a nonprofit public benefit corporation and Purchaser, a Delaware corporation, any other subsidiary, parent, general partner, limited partner, member, affiliate, successor, successor in interest, assignee, or person or entity serving in a similar capacity of any of the above-listed entities including, but not limited to, any entity succeeding thereto as a result of consolidation, affiliation, merger, or acquisition of all or substantially all of the real property or operating assets of St. Francis Medical Center, lessees, licensees, or operators of St. Francis Medical Center, and any and all current and future lessees and owners of the real property on which St. Francis Medical Center is located, Center is located.

The transaction conditionally approved by the Attorney General consists of \_\_\_\_\_\_.

All the entities listed in Condition I, and any other parties referenced in the above agreements shall fulfill the terms of these agreements or documents and shall notify and obtain the Attorney General's approval in writing of any proposed modification or rescission of any of the terms of these agreements or documents. Such notifications shall be provided at least sixty days prior to their effective date in order to allow the Attorney General to consider whether they affect the factors set forth in Corporations Code section 5917 and obtain the Attorney General's approval.

For ten years from the closing date of the Asset Purchase Agreement, Verity Health System of California, Inc., Verity Holdings, LLC, Purchaser, and all future owners, managers, lessees, licensees, or operators of St. Francis Medical Center shall be required to provide written notice to the Attorney General sixty days prior to entering into any agreement or transaction to do any of the following:

Sell, transfer, lease, exchange, option, convey, manage, or otherwise dispose of St. Francis Medical Center;

Transfer control, responsibility, management, or governance of St. Francis Medical Center. The substitution, merger or addition of a new member or members of the governing body of Purchaser that transfers the control of, responsibility for or governance of St. Francis Medical Center, shall be deemed a transfer for purposes of this Condition. The substitution or addition of one or more members of the governing body of Purchaser, or any arrangement, written or oral, that would transfer voting control of the members of the governing body of Purchaser shall also be deemed a transfer for purposes of this Condition.

<sup>&</sup>lt;sup>1</sup> Throughout this document, the term "St. Francis Medical Center" shall mean the general acute care hospital located at 3630 East Imperial Highway, Lynwood, CA 90262, and any other clinics, laboratories, units, services, or beds included on the license issued to St. Francis Medical Center by the California Department of Public Health, effective January 1, 2019, unless otherwise indicated.

For ten years from the closing date of the Asset Purchase Agreement, St. Francis Medical Center shall be operated and maintained as a licensed general acute care hospital (as defined in California Health and Safety Code Section 1250) and shall maintain and provide 24-hour emergency and trauma medical services at no less than current<sup>2</sup> licensure and designation with the same types and/or levels of services, including the following:

46 emergency treatment stations at a minimum;
Designation as a Level II Trauma Center;
Designation as a 5150 Receiving Facility, as defined by the Welfare and Institutions Code, section 5150, for behavioral health patients under involuntary evaluation;
Psychiatric evaluation team;
Designation as an Emergency Department Approved for Pediatrics (EDAP);
Designation as a Paramedic Base Station; and Certification as a Primary Stroke Center.

Purchaser must give one-year advance written notice to the Los Angeles County Emergency Medical Services Agency and the California Department of Public Health if St. Francis Medical Center seeks to reduce trauma or trauma-related care services or stop operating the Level II Trauma Center after ten years from the closing date of the Asset Purchase Agreement.

For at least ten years from the closing date of the Asset Purchase Agreement, St. Francis Medical shall maintain on-call coverage contracts and/or comparable coverage arrangements with physicians that are necessary to retain its qualification as a Level II trauma center. The following on-call coverage contracts and/or comparable coverage arrangements are required to retain St. Francis Medical Center's status as a Level II trauma center:

Neurology; Obstetrics/gynecology; Ophthalmology; Oral or maxillofacial or head and neck; Orthopedic Plastic surgery; Reimplantation/microsurgery capability (this surgical service may be provided through a written transfer agreement); and Urology.

For at least ten years from the closing date of the Asset Purchase Agreement, St. Francis Medical Center shall maintain the following services at current licensure, types, and/or levels of services:

- Cardiac services, including three cardiac catheterization labs and the designation as a STEMI Receiving Center;
- Critical care services, including a minimum of 36 intensive care unit beds or 24 intensive care beds and 12 definitive observation beds;
- Neonatal intensive care services, including a minimum of 29 neonatal intensive care beds, and at minimum, maintaining a Level II NICU;
- Women's health services, including women's imaging services;

<sup>&</sup>lt;sup>2</sup> The term "current" or "currently" throughout this document means as of January 1, 2019.

- Pediatric services, including a designated area with at least five general acute care beds for pediatric services;
- Orthopedic and rehabilitation services;
- Wound care services;
- Behavioral health services, including a minimum of 40 distinct part inpatient acute psychiatric beds; and
- Perinatal services, including a minimum of 50 perinatal beds.

Purchaser shall not place all or any portion of its above-listed licensed-bed capacity or services in voluntary suspension or surrender its license for any of these beds or services.

For at least ten years from the closing date of the Asset Purchase Agreement, St. Francis Medical Center shall maintain the same types and/or levels of women's healthcare services currently provided at the location below or a location within three miles of St. Francis Medical Center:

Family Life Center at St. Francis Medical Center, located at 3630 E Imperial Highway, Lynwood, California.

For at least five years from the closing date of the Asset Purchase Agreement, Purchaser shall either: (1) operate clinics (listed below) with the same number of physicians and mid-level provider full-time equivalents in the same or similar alignment structures, or (2) sell the clinics (listed below) with the same number of physician and mid-level provider full-time equivalents and require the purchaser(s) to maintain such services for 5 years from the closing date of the Asset Purchase Agreement and to participate in the Medi-Cal and Medicare programs as required in the conditions herein, or (3) ensure that a third party is operating the clinics (listed below) with the same number of physician and mid-level provider full-time equivalents and require the third party to maintain such services for 5 years from the closing date of the Asset Purchase Agreement and to participate in the Medi-Cal and Medicare programs as required to maintain such services for 5 years from the closing date of the Asset Purchase Agreement and to participate in the Medi-Cal and Medicare programs as required in the conditions herein. For any of these options, each clinic can be moved to a different location within a three-mile radius of each clinic's current location, and St. Francis Medical Center can utilize an alternative structure in providing such services. The following clinic is subject to this condition:

Orthopedic services at 3628 E. Imperial Highway, Suite 300, Lynwood, California.

For ten years from the closing date of the Asset Purchase Agreement, Purchaser shall:

Be certified to participate in the Medi-Cal program at St. Francis Medical Center;

Maintain and have Medi-Cal Managed Care contracts with the below listed Medi-Cal Managed Care Plans to provide the same types and levels of emergency and non-emergency services at St. Francis Medical Center to Medi-Cal beneficiaries (both Traditional Medi-Cal and Medi-Cal Managed Care) as required in these Conditions, on the same terms and conditions as other similarly situated hospitals offering substantially the same services, without any loss, interruption of service or diminution in quality, or gap in contracted hospital coverage, unless the contract is terminated for cause or not extended or renewed by the Medi-Cal Managed Care Plan:

Local Initiative: L.A. Care Health Plan or its successor; and

Commercial Plan: Health Net Community Solutions, Inc. or its successor.

If Purchaser questions whether it is being reimbursed on the same terms and conditions as other similarly situated hospitals offering substantially the same services, it shall notify the Attorney General's Office with at least 120 days' notice prior to taking any action that would effectuate any loss, interruption of service or diminution in quality, or gap in contracted hospital coverage or prior to giving any required notice of taking such action.

Be certified to participate in the Medicare program by maintaining a Medicare Provider Number to provide the same types and levels of emergency and non-emergency services at St. Francis Medical Center to Medicare beneficiaries (both Traditional Medicare and Medicare Managed Care), on the same terms and conditions as other similarly situated hospitals, as required in these Conditions.

For six fiscal years from the closing date of the Asset Purchase Agreement, Purchaser shall provide an annual amount of Charity Care (as defined below) at St. Francis Medical Center equal to or greater than \$8,000,000 (the Minimum Charity Care Amount). For purposes hereof, the term "charity care" shall mean the amount of charity care costs (not charges) incurred by Purchaser in connection with the operation and provision of services at St. Francis Medical Center. The definition and methodology for calculating "charity care" and the methodology for calculating "costs" shall be the same as that used by Office of Statewide Health Planning Development (OSHPD) for annual hospital reporting purposes.<sup>3</sup>

Purchaser's obligation under this Condition shall be prorated on a daily basis if the closing date of the Asset Purchase Agreement is a date other than the first day of Verity Health System of California, Inc.'s fiscal year.

For the second fiscal year and each subsequent fiscal year, the Minimum Charity Care Amount shall be increased (but not decreased) by an amount equal to the Annual Percent increase, if any, in the 12 Months Percent Change: All Items Consumer Price Index for All Urban Consumers in the Los Angeles-Long Beach-Anaheim Average Base Period: 1982-84=100 (CPI-LA, as published by the U.S. Bureau of Labor Statistics).

If the actual amount of charity care provided at St. Francis Medical Center for any fiscal year is less than the Minimum Charity Care Amount (as adjusted pursuant to the above-referenced Consumer Price Index) required for such fiscal year, Purchaser shall pay an amount equal to the deficiency to one or more tax-exempt entities that provide direct healthcare services to residents in the St. Francis Medical Center's service area (31 ZIP codes), as described on page 54 in the Healthcare Impact Report authored by JD Healthcare dated August 16, 2019. (Exhibit 1.) Such payment(s) shall be made within six months following the end of such fiscal year.

For six fiscal years from the closing date of the Asset Purchase Agreement Purchaser shall provide an annual amount of Community Benefit Services at St. Francis Medical Center equal to or greater than \$1,139,301 (the "Minimum Community Benefit Services Amount") exclusive of

<sup>&</sup>lt;sup>3</sup> OSHPD defines charity care by contrasting charity care and bad debt. According to OSHPD, "the determination of what is classified as . . . charity care can be made by establishing whether or not the patient has the ability to pay. The patient's accounts receivable must be written off as bad debt if the patient has the ability but is unwilling t o pay off the account."

any funds from grants. For six fiscal years, the following community benefit programs and services shall continue to be offered at its current or equivalent location:

Southern California Crossroads Program; Health Benefit Resource Center; Welcome Baby Program; Healthy Community Initiatives; American Career College access for onsite training; Paramedic Training and Education; and Patient Transportation support.

The planning of, and any subsequent changes to, the community benefit services provided at St. Francis Medical Center shall be decided after consultation with the Local Governing Board of Directors.

Purchaser's obligation under this Condition shall be prorated on a daily basis if the effective date of the Asset Purchase Agreement is a date other than the first day of Verity Health System of California, Inc.'s fiscal year.

For the second fiscal year and each subsequent fiscal year, the Minimum Community Benefit Services Amount shall be increased (but not decreased) by an amount equal to the Annual Percent increase, if any, in the 12 Months Percent Change: All Items Consumer Price Index for All Urban Consumers in the Los Angeles-Long Beach-Anaheim Average Base Period: 198284=100 (CPI-LA, as published by the U.S. Bureau of Labor Statistics).

If the actual amount of community benefit services provided at St. Francis Medical Center for any fiscal year is less than the Minimum Community Benefit Services Amount (as adjusted pursuant to the above-referenced Consumer Price Index) required for such fiscal year, Purchaser shall pay an amount equal to the deficiency to one or more tax-exempt entities that provide community benefit services for residents in St. Francis Medical Center's service area (31 ZIP codes), as defined on as described on page 54 in the Healthcare Impact

Report authored by JD Healthcare dated August 16, 2019. (Exhibit 1.) Such payment(s) shall be made within six months following the end of such fiscal year.

For at least ten years from the closing date of the Asset Purchase Agreement unless otherwise indicated, Purchaser shall maintain its contracts and any amendments and exhibits thereto with the City and/or County of Los Angeles for services, including the following:

- Participation in the Hospital Preparedness Program between the Hospital and Los Angeles County;
- Department of Mental Health Legal Entity Contract between the Hospital and Los Angeles County;

Paramedic Base Hospital Services between the Hospital and Los Angeles County; Radiation Therapy Services between the Hospital and Los Angeles County;

- Designation Agreement between the County of Los Angeles Department of Mental Health (LAC-DMH) and the Hospital and approved as a 72-hour Evaluation and Intensive Treatment facility;
- Trauma Center Service Agreement between the Hospital and Los Angeles County; and
- Paramedic Training Institute Students between the Hospital and Los Angeles County.

For at least ten years from the closing date of the Asset Purchase Agreement, Purchaser shall provide to the Los Angeles County Department of Health Services and Los Angeles County of Department of Mental Health information and documents related to staffing assessments, clinical guidelines, services provided, and technology needs for St. Francis Medical Center. The goal is to ensure that Purchaser's decisions or changes in these areas will not be motivated by a desire to move away from serving the Medi-Cal population. Such information and documents will also be provided to the Local Governing Board.

For ten years from the closing date of the Asset Purchase Agreement, Purchaser shall have at St. Francis Medical Center a Local Governing Board of Directors. Purchaser shall consult with the Local Governing Board of Directors prior to making changes to medical services, community benefit programs, making capital expenditures, including making changes to the charity care and collection policies, and making changes to charity care services provided at St. Francis Medical Center. The members of the Local Governing Board of Supervisors, and community representatives from St. Francis Medical center's medical staff, St. Francis Medical Center's Chief of Staff, one member designated by the Los Angeles County Board of Supervisors, and community representatives from St. Francis Medical Center's primary service area (31 ZIP codes), as described on page 54 in the Healthcare Impact Report authored by JD Healthcare dated August 16, 2019 attached hereto as Exhibit 1, including at least one member from a local healthcare advocacy group. Such consultation shall occur at least sixty days prior to the effective date of such changes or actions unless done so on an emergency basis. The Local Governing Board's approval is required of all reports submitted to the Attorney General regarding compliance with these Conditions.

Purchaser shall commit to reserve or expend for St. Francis Medical Center for capital improvements to the hospital over the five-year period from the closing of the Asset Purchase Agreement, the amount of capital that remains unexpended from the \$180 million commitment required of Blue Mountain Capital Management, LLC as part of the Attorney General Conditions approved on December 3, 2015 but this amount can be no less than \$5.8 million among St. Francis Medical Center, Seton Medical Center and St. Vincent Medical Center. Purchaser understands that such condition has previously been satisfied but will still commit to expend approximately \$35 million on St. Francis Medical Center over the five-year period from the closing of the Asset Purchase Agreement.

Purchaser shall commit the necessary investments required to maintain OSHPD seismic compliance requirements at the Hospital through 2030 under the Alfred E. Alquist Hospital Facilities Seismic Safety Act of 1983, as amended by the California Hospital Facilities Seismic Safety Act, (Health & Safety. Code, § 129675-130070).

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 109 of 147

Purchaser shall maintain privileges for current medical staff who are in good standing as of the closing date of the Asset Purchase Agreement. Further, the closing of the Asset Purchase Agreement shall not change the independence of the medical staff.

There shall be no discrimination against lesbian, gay, bisexual, or transgender individuals at St. Francis Medical Center, and no restriction or limitation on providing or making reproductive health services available at St. Francis Medical Center, its medical office buildings, or at any of its facilities. Both of these prohibitions shall be set forth in Purchaser's written policies, adhered to, and strictly enforced.

For eleven fiscal years from the closing date of the Asset Purchase Agreement Purchaser shall submit to the Attorney General, no later than four months after the conclusion of each fiscal year, a report describing in detail compliance with each Condition set forth herein. The Chairman of the Board of Directors of Purchaser shall certify that the report is true, accurate, and complete and provide documentation of the review and approval of the report by the Local Governing Board.

At the request of the Attorney General, all parties listed in Condition I, Verity Health System of California, Inc., Verity Holdings, LLC, Purchaser, and any other parties referenced in the agreements listed in Condition II shall provide such information as is reasonably necessary for the Attorney General to monitor compliance with these Conditions and the terms of the transaction as set forth herein. The Attorney General shall, at the request of a Party and to the extent provided by law, keep confidential any information so produced to the extent that such information is a trade secret or is privileged under state or federal law, or if the private interest in maintaining confidentiality clearly outweighs the public interest in disclosure.

Once the Asset Purchase Agreement is closed, and so long as such actions are consistent with the Asset Purchase Agreement, all parties listed in Condition I, and any other parties referenced in the agreements listed in Condition II are deemed to have explicitly and implicitly consented to the applicability and compliance with each and every Condition and to have waived any right to seek judicial relief with respect to each and every Condition.

The Attorney General reserves the right to enforce each and every Condition set forth herein to the fullest extent provided by law. In addition to any legal remedies the Attorney General may have, the Attorney General shall be entitled to specific performance, injunctive relief, and such other equitable remedies as a court may deem appropriate for breach of any of these Conditions. Pursuant to Government Code section 12598, the Attorney General's office shall also be entitled to recover its attorney fees and costs incurred in remedying each and every violation.

#### Analysis of the Hospital's Service Area

#### Service Area Definition

Based upon the Hospital's CY 2017 inpatient discharges, the Hospital's service area is comprised of 31 ZIP Codes from which 75% of the Hospital's inpatient discharges came from. Approximately 51% of the Hospital's discharges originated from the top eight ZIP Codes, located in Lynwood, South Gate, Los Angeles, Bell, Compton, Bell Gardens, and Huntington Park. In CY 2017, the Hospital's market share in the service area was approximately 11% based on total area discharges.

PATIENT ORIGIN, CY 2017						
			Percentage	Cumulative	Market	Total
ZIP Code	Community	Discharges	of Discharges	Percentage	Share	Discharges
90262	Lynwood	2,490	11.1%	11.1%	38.1%	6,538
90280	South Gate	2,187	9.8%	20.9%	29.0%	7,554
90221	Compton	1,400	6.3%	27.2%	24.1%	5,812
90201	Bell	1,359	6.1%	33.3%	16.3%	8,363
90002	Los Angeles	1,066	4.8%	38.0%	18.4%	5,797
90255	Huntington Park	956	4.3%	42.3%	15.5%	6,172
90059	Los Angeles	948	4.2%	46.6%	17.2%	5,527
90001	Los Angeles	922	4.1%	50.7%	15.6%	5,901
90220	Compton	708	3.2%	53.9%	12.7%	5,554
90222	Compton	700	3.1%	57.0%	18.1%	3.868
90003	Los Angeles	625	2.8%	59.8%	7.6%	8,209
90044	Los Angeles	542	2.4%	62.2%	4.5%	11,994
90723	Paramount	525	2.3%	64.6%	11.7%	4,483
90061	Los Angeles	358	1.6%	66.2%	9.5%	3,764
90650	Norwalk	344	1.5%	67.7%	3.3%	10,373
90270	Maywood	282	1.3%	69.0%	12.2%	2,309
90805	Long Beach	267	1.2%	70.2%	2.7%	9,940
90706	Bellflower	263	1.2%	71.3%	3.6%	7,223
90242	Downey	252	1.1%	72.5%	6.2%	4,038
90241	Downey	224	1.0%	73.5%	6.0%	3,726
90660	Pico Rivera	91	0.4%	73.9%	1.4%	6,608
90240	Downey	69	0.3%	74.2%	3.3%	2,073
90670	Santa Fe Springs	46	0.2%	74.4%	2.7%	1,703
90605	Whittier	44	0.2%	74.6%	1.1%	4,082
90606	Whittier	44	0.2%	74.8%	1.4%	3,244
90703	Cerritos	37	0.2%	74.9%	0.9%	4,026
90604	Whittier	32	0.1%	75.1%	0.9%	3,698
90701	Artesia	31	0.1%	75.2%	1.7%	1,813
90638	La Mirada	30	0.1%	75.4%	0.7%	4,274
90603	Whittier	3	0.0%	75.4%	0.1%	2,152
90639	La Mirada	0	0.0%	75.4%	0.0%	10
Sub-Tot al		16,845	75.4%	75.4%	10.5%	160,828
All Other		5,504	24.6%	100%		
Grand Total		22,349	100%	100/0		
	PD Discharge Database		100/0			

Source: OSHPD Discharge Database, CY 2017

#### **DISCLOSURE SCHEDULES**

To

#### ASSET PURCHASE AGREEMENT

By and Among

#### ST. FRANCIS MEDICAL CENTER, VERITY HOLDINGS, LLC,

# and VERITY HEALTH SYSTEM OF CALIFORNIA, INC.

(as "Sellers")

and

(as "Purchaser")

Dated [\_\_\_\_\_, 2020]

# **Schedule 1.4.3 - Owned Real Property**

Description	Owner	Address	APN
Hospital Campus	St. Francis Medical	3630 East Imperial	6173-021-008
(including Hospital	Center	Highway, Lynwood,	
Patient Tower, Health		CA 90262	
Services Pavilion,			
Progressive Care			
Unit, Family Life			
Center, Power Plant			
and Parking Structure			
#1)			
Huntington Park	St. Francis Medical	2700 East Slauson	6320-006-069
Medical Office	Center	Avenue, Huntington	
Building		Park, CA 90255	
Maywood Medical	St. Francis Medical	5953 South Atlantic	6313-013-028
Office Building	Center	Boulevard 5,	
		Maywood, CA 90270	
Parking lot (St.	Verity Holdings, LLC	3633 Martin Luther	6173-015-047
Francis Medical		King Jr., Boulevard,	
Center)		Lynwood, CA 90262	
Ministry Services	Verity Holdings, LLC	3663 Martin Luther	6173-019-022
Building (St. Francis		King Jr., Boulevard,	
Medical Center)		Lynwood, CA 90262	
Parking Lot (St.	Verity Holdings, LLC	3621 Martin Luther	6173-015-048
Francis Medical		King Jr., Boulevard,	
Center)		Lynwood, CA 90262	
Five Unit Apartment	Verity Holdings, LLC	3570 Brenton	6191-016-008
(St. Francis Medical		Avenue, Lynwood,	
Center		CA 90057	

St. Francis Medical Center

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 113 of 147

St Francis Medical Center - A	ccreditation/Certifications	/Licenses
License/Permit/Certificate	License/Permit Number	Expiration date <sup>4</sup>
AABB - Certificate of Accreditation for		9/30/2021
Transfusion Activities		
ACR - for Adult Patients - Revolution		7/7/2020
EVO - 5		
ACR - for Adult Patients - Revolution		7/7/2020
EVO - 4		
ACR - ADAC Laboratories TRANSCAM		10/12/2021
1994-Accreditation		
ACR - ADAC Laboratories VERTEX 1998		10/12/2021
for Planer - Accreditation		
ACR - MRI Siemens AVANTO 2005		5/19/2021
-Accreditation		
ACR - Siemens SYMBIA T 2011 for		10/12/2021
Planar, SPECT, Nuclear Cardiology		
ACR - Mammography - Lorad Medical	MAP# 07026-05	8/4/2022
Systems Dimensions 2012		
ACR - Mammography - Lorad Medical	MAP# 07026-06	8/4/2022
Systems Dimensions 2012 DBT		
Hospital Pharmacy Permit	HSP 45706	2/1/2021
California Department of Health –	FAC00007446	6/30/2020
Processing Reportable Sources of Radiation		
Sterile Compounding License	LSC 100267	2/1/2021
City of Lynwood - Business License - 4	2301342	12/31/2020
Units		
City of Lynwood - Business License	2301756	12/31/2020
City of South Gate - Business License	3201	6/30/2020
CLIA Certificate of Waiver	05D0548351	2/27/2021
CLIA Certificate of Waiver	05D0949548	11/1/2021
County of LA Public Health - Public	PR0038667	6/30/2020
Health Permit - Cafeteria		
Department of Health & Human Services	138560	8/4/2022
-Mammography Facility		
Clinical Laboratory Certificate of Deemed	CDF00000291	12/28/2020
Status		
Department of Public Health –	20668	8/4/2022
Mammography X-Ray Equipment and		
Facility 2 Dimension		
Department of Public Health –	20669	8/4/2022
Mammography X-Ray Equipment and		

# <u>Schedule 1.7(b) – Licenses</u>

<sup>4</sup> Expiration date is subject to being updated.

SQG37254	4/16/2020
	11/16/2020
	12/31/2020
CTB 00080570	12/12/2020
CTB 00080568	12/12/2020
LA-179(H)	
BS7805609	2/28/2023
N/A	12/31/2021
14924	7/15/2020
107436	2/7/2020
	Renewal in process,
	inspection occurred
	February 17-19, 2020.
	New permits expected
	in April 2020.
107437	2/7/2020
	Renewal in process,
	inspection occurred
	February 17-19, 2020.
	New permits expected
	in April 2020.
107438	2/7/2020
	Renewal in process,
	inspection occurred
	February 17-19, 2020.
	New permits expected
	in April 2020.
113252	6/21/2020
113372	2/8/2020
	Renewal in process,
	inspection occurred
	February 17-19, 2020.
	New permits expected
	in April 2020.
113373	2/8/2020
	Renewal in process,
	inspection occurred
	February 17-19, 2020.
	New permits expected
	930000157 CTB 00080570 CTB 00080568 LA-179(H) BS7805609 N/A 14924 107436 107437 107438

|

		in April 2020.
Conveyance Permit	113374	6/21/2020
Conveyance Permit	113375	2/8/2020
5		Renewal in process,
		inspection occurred
		February 17-19, 2020.
		New permits expected
		in April 2020.
Conveyance Permit	113401	2/8/2020
,		Renewal in process,
		inspection occurred
		February 17-19, 2020.
		New permits expected
		in April 2020.
Conveyance Permit	121533	2/8/2020
5		Renewal in process,
		inspection occurred
		February 17-19, 2020.
		New permits expected
		in April 2020.
Conveyance Permit	121535	2/8/2020
5		Renewal in process,
		inspection occurred
		February 17-19, 2020.
		New permits expected
		in April 2020.
Conveyance Permit	121557	2/8/202
2		Renewal in process,
		inspection occurred
		February 17-19, 2020.
		New permits expected
		in April 2020.
Conveyance Permit	36304	2/8/2020
·		Renewal in process,
		inspection occurred
		February 17-19, 2020.
		New permits expected
		in April 2020.
Conveyance Permit	36305	2/8/2020
-		Renewal in process,
		inspection occurred
		February 17-19, 2020.
		New permits expected
		in April 2020.
Conveyance Permit	90171	2/7/2020
-		Renewal in process,

		• • •
		inspection occurred
		February 17-19, 2020.
		New permits expected
		in April 2020.
Conveyance Permit	99368	2/8/2020
		Renewal in process,
		inspection occurred
		February 17-19, 2020.
		New permits expected
		in April 2020.
Conveyance Permit	99562	2/8/2020
	· · · · · · · · · · · · · · · · · · ·	Renewal in process,
		inspection occurred
		February 17-19, 2020.
		New permits expected
		in April 2020.
Conveyance Permit	99563	2/8/2020
Conveyance Permit	99303	
		Renewal in process,
		inspection occurred
		February 17-19, 2020.
		New permits expected
		in April 2020.
Conveyance Permit	99564	2/8/2020
		Renewal in process,
		inspection occurred
		February 17-19, 2020.
		New permits expected
		in April 2020.
Conveyance Permit	99565	2/8/2020
		Renewal in process,
		inspection occurred
		February 17-19, 2020.
		New permits expected
		in April 2020.
Radioactive Material License	0488-19	6/28/2024
The College of America Pathologists	2273801	8/22/2021
Laboratory Certification	2273001	0,22,2021
The Joint Commission - Hospital	9953	10/27/2021
Accreditation Program	<i>))))</i>	10/2//2021
The Joint Commission - Primary Stroke	9953	6/21/2020
Federal Communications Commission	KUX246	2/9/2024
	KUA240	2/9/2024
Private Radio Licenses		2/21/2024
Federal Communications Commission	KWK570	3/21/2024
Private Radio Licenses		
Federal Communications Commission	WQHA434	6/11/2027
Private Radio Licenses		

|

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 117 of 147

Federal Communications Commission	WQIG242	1/30/2028
Private Radio Licenses		
California Department of Tax and Fee	097962355-10000	
Administration Seller's Permit		
Los Angeles County Department of Mental	N/A	N/A
Health Consolidated Medi-Cal		
Re-Certification Approval for St. Francis		
Medical Center Children's Center		
Baby-Friendly USA Designation as	N/A	12/8/2024
Baby-Friendly Hospital		
American College of Surgeons VRC	N/A	11/20/2021
Committee on Trauma Verification as		
Level II Trauma Center		

Provider Number Listing		
Medicare Hospital		
05-0104 St Francis Medical Center		
55-5238	St Francis Medical Center - SNF	

Medi-Cal	Hospital	
	St Francis Medical Center	
HSC30104G	Acute Inpatient Contract	
ZZT30104G Acute Inpatient Non-Contract		
ZZT40104G	Outpatient Contract	
LTC55238G	SNF Inpatient	

OSHPD	Hospital
106190754	St Francis Medical Center

# Schedule 1.7(c)(ii) - Leased Real Property

	St. Francis Medical Center as Tenant
1.	Lease Agreement, dated May 1, 2018 between St. Francis Medical and Hopevale
	Properties, L.P. c/o Reliable Properties for the premises located at 4390 Tweedy Blvd.,
	South Gate, CA 90280 (Expires May 31, 2020).
2.	St. Francis Medical Plaza Office Space Lease, dated January 1, 2006, between St. Francis
	Medical Center (as Tenant) and St. Francis-Lynwood Medical Plaza L.P. (as Landlord) for
	the premises located at 3628 E. Imperial Highway, Lynwood, CA 90262, as amended by
	that certain First Amendment to Lease, dated May 1, 2006, the Second Amendment to
	Lease, dated February 7, 2007, the Third Amendment to Lease, dated November 29, 2007,
	the Fourth Amendment to Lease, dated January 1, 2010, between St. Francis Medical
	Center (as Tenant) and NHP/PMB St. Francis-Lynwood Medical Plaza LLC (as
	Landlord), the Fifth Amendment to Lease, dated December 31, 2010, the Sixth
	Amendment to Office Lease and Agreement Regarding Property Tax Protection, dated
	May 27, 2011, the Seventh Amendment to Lease, dated March 27, 2012, the Eighth
	Amendment to Lease, dated May 1, 2013, the Ninth Amendment to Lease, dated
	September 9, 2013, the Consent of Prime Landlord to Sublease, dated April 1, 2014 by
	NHP/PMB St. Francis Lynwood Medical Plaza, LLC, the Tenth Amendment to Lease,
	dated November 20, 2015, the Consent of Prime Landlord to Sublease, dated January 15,
	2016, by NHP/PMB St. Francis Lynwood Medical Plaza, LLC, the Eleventh Amendment
	to Lease, dated September 27, 2016, and the Consent of Prime Landlord to Sublease,
	dated December 1, 2016, by NHP/PMB St. Francis Lynwood Medical Plaza, LLC
	(Expires December 31, 2021).
3.	Standard Multi-Tenant Office Lease-Gross, dated March 1, 2006, between St. Francis
	Medical Center (as Lessee) and Sunshine Capital Group, LLC (as Lessor) for the premises
	located at 3680 E. Imperial Hwy Suites 450, 525 and 550, Lynwood, CA 90262, as
	amended by that certain First Amendment to Office Lease, dated October 23, 2007, the
	Second Amendment to Lease, dated December 5, 2012, the Third Amendment to Lease,
	dated June 25, 2013, and the Fourth Amendment to Lease, dated September 17, 2013
	(Expired September 30, 2018). <sup>5</sup>

I

<sup>&</sup>lt;sup>5</sup> Within the Lease Schedules "\*" indicates the lease is currently month-to-month.

# Schedule 1.7(c)(iii) - Tenant Leases

	St. Francis Medical Center as Landlord
1.	Parking Space Lease, dated on or about January 7, 2018 between St. Francis Medical
	Center (as Landlord) and Davita Medical Management (as Tenant) for the premises
	located at 2700 E. Slauson Avenue, Huntington Park, CA 90255 as amended by that
	certain First Amendment, dated January 10, 2019 (Expires October 28, 2019).*
2.	Medical Office Building Ground Lease, dated January 1, 1993 between St. Francis
	Medical Center (as Landlord) and Pacific Medical Buildings, L.P. (as Tenant) for the
	premises located in Lynwood, CA, as amended by that certain First Amendment to
	Medical Office Building Ground Lease, dated February 1, 1993, the Second Amendment
	to Medical Office Building Ground Lease, dated May 1, 1993, the Assignment and
	Assumption of Leases and Subleases dated, June 4, 1993 between Pacific Medical
	Buildings, L.P. (as Assignor) and St. Francis-Lynwood Medical Plaza, L.P. (as Assignee),
	and the Grant and Reservation of Easements and Parking Covenants by St. Francis
	Medical Center, dated January 1, 1993 (Expires January 1, 2048).
3.	Parking Garage Ground Lease, dated January 1, 1993 between St. Francis Medical Center
	(as Landlord) and Pacific Medical Buildings, L.P. (as Tenant) for the premises located in
	Lynwood, CA, as amended by that certain Assignment and Assumption of Leases and
	Subleases, dated June 4, 1993 between Pacific Medical Buildings, L.P. (as Assignor) and
	St. Francis-Lynwood Medical Plaza, L.P. (as Assignee) (Expires January 1, 2048).
4.	Sublease, dated December 1, 2016, between St. Francis Medical Center (as Sublessor) and
	Verity Medical Foundation (as Sublessee) for the premises located at 3628 East Imperial
	Highway Suite 408, Lynwood, CA 90262 (Expires December 31, 2020).
5.	Sublease, dated January 15, 2016, between St. Francis Medical Center (as Sublessor) and
	St. Francis Multispecialty Medical Group, Inc. (as Sublessee) for the premises located at
	3628 East Imperial Highway Suite 303, Lynwood, CA 90262, as amended by that certain
	First Amendment to Sublease between St. Francis Medical Center (Sublessor) and St.
	Francis Multispecialty Medical Group, Inc. (Sublessee), dated December 24, 2019.
	(Expires December 31, 2020).

 $\label{eq:states} \begin{array}{l} 4" = "1" "DM_{US} \ 165807051 \ -6.079742.0063" "" \\ US_Active \ 114483166 \ V-4 \ \underline{6} \end{array}$ 

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 120 of 147

	Verity Holdings, LLC as Landlord				
1.	Apartment Lease, dated July 10, 2015, between Verity Holdings, LLC,				
	successor-in-interest to St. Francis Medical Center (as Landlord) and Diane Clark (as				
	Tenant) for the premises located at 3570 Brenton Avenue, Apt. B, Lynwood, CA 90262 *				
2.	Apartment Lease, dated July 10, 2015, between Verity Holdings, LLC,				
	successor-in-interest to St. Francis Medical Center (as Landlord) and Gregory Slack (as				
	Tenant) for the premises located at 3570 Brenton Avenue, Apt. A, Lynwood, CA 90262 *				
3.	Apartment Lease, dated September 1, 2015, between Verity Holdings, LLC,				
	successor-in-interest to St. Francis Medical Center (as Landlord) and Marco Monteon (as				
	Tenant) for the premises located at 3570 Brenton Avenue, Apt. D, Lynwood, CA 90262 *				

Ref #	Debtor	Contract Counterparty	Nature of Contact / Lease	Termination Date	Cure Amount	
57	St. Francis Medical Center	AMEDA INC	BREAST PUMP AND ACCESSORIES	10/31/2020	\$1,965.25	
65	St. Francis Medical Center	AMERICAN RED CROSS	BLOOD PRODUCTS	11/11/2018	\$247,033.00	
94	St. Francis Medical Center	ARJO INC	PATIENT LIFTS	7/31/2021	\$20,516.56	
95	St. Francis Medical Center	ARROW INTERNATIONAL INC	HEMODYNAMIC MONITORING PRODUCT	6/30/2020	\$24,438.28	
100	St. Francis Medical Center	ASAHI INTECC USA INC	DIC PRODUCTS	6/30/2019	\$3,417.77	
136	St. Francis Medical Center	BAUSCH AND LOMB	OPHTHALMOLOGY PRODUCTS	6/30/2020	\$3,316.97	
142	St. Francis Medical Center	BAYER MEDICAL CARE LLC	CM INJECTORS AND DISPOSABLES	12/31/2020	\$46,138.51	
143	St. Francis Medical Center	BAYER MEDICAL CARE LLC	CONTRAST MEDIA MR	12/31/2020		
157	St. Francis Medical Center	BIO RAD LABORATORIES DIAGNOSTICS	BLOOD BANK ANALYZERS REAGENTS	1/31/2019	\$9,387.74	
163	St. Francis Medical Center	BIOVENTUS LLC	BONE AND BONE SUBSTITUTE PRODUCTS	6/30/2021	\$15,189.80	
164	St. Francis Medical Center	BIOVENTUS LLC	SYNTHETIC IMPLANTABLE PRODUCTS	6/30/2021		
190	St. Francis Medical Center	BRASSELER USA MEDICAL LLC	ORTHOPEDIC POWER TOOLS	4/30/2019	\$1,224.86	
338	St. Francis Medical Center	CAREFUSION	CHLOROHEXADINE GLUCONATE (CHG)	7/31/2020	\$730.96	
396	St. Francis Medical Center	COOK MEDICAL INC	SPECIALTY UROLOGY PRODUCTS	12/31/2020		
397	St. Francis Medical Center	COOK MEDICAL INC	GASTROINTESTINAL ENDOSCOPY	1/31/2019		
398	St. Francis Medical Center	COOK MEDICAL INC	PERIPHERAL AND BILIARY STENTS	1/31/2020		
399	St. Francis Medical Center	COOK MEDICAL INC	DIR PRODUCTS	1/31/2020	\$60,694.99	
400	St. Francis Medical Center	COOK MEDICAL INC	CARDIAC RHYTHM MANAGEMENT	3/31/2020		
401	St. Francis Medical Center	COOK MEDICAL INC	VENA CAVA FILTERS	7/31/2020		
402	St. Francis Medical Center	COOK MEDICAL INC	HIGH RISK OBGYN PRODS MED/SURG	12/31/2019		
403	St. Francis	COOK MEDICAL INC	HIGH RISK OBGYN PROD	12/31/2019		

# Schedule 1.7(d) - Assigned Contracts and Assigned Leases

4" = "1" "DM\_US 165807051-6.079742.0063" "" US\_Active\114483166\V-4<u>6</u>

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 122 of 147

	Medical Center		SPECIALTY			
449	St. Francis Medical Center	DAVITA MEDICAL MANAGEMENT LLC	LEASE-AS LANDLORD	2/28/2019	\$0.00	
519	St. Francis Medical Center	GRACE MEDICAL	ENT IMPLANTS AND INSTRUMENTS	10/31/2019	\$180.86	
525			MANUAL MICROBIOLOGY	12/31/2019	\$7,043.31	
539	St. Francis HELMER SCIENTIFIC		GENERAL LABORATORY PRODUCTS	3/31/2021	\$131.75	
550	St. Francis Medical Center	HOPEVALE PROPERTIES LP (RELIABLE PROPERTIES) FKA CHUNG SUK PU FKA WIN PROPERTY MANAGEMENT	LEASE-AS TENANT	4/30/2019	\$0.00	
556	St. Francis Medical Center	HUFFBURT PROPERTY LLC (FKA JAMES H. DEWALD TRUST)	LEASE-AS TENANT	4/30/2019	\$0.00	
557	St. Francis Medical Center	HUFFBURT PROPERTY LLC (FKA JAMES H. DEWALD TRUST)	LEASE-AS TENANT	4/30/2019		
625	St. Francis Medical Center	LEICA MICROSYSTEMS INC	HISTOLOGY AND CYTOLOGY	3/31/2021	\$445.41	
631	St. Francis Medical Center	LIVANOVA USA INC	PERFUSION PRODUCTS	1/31/2021	\$2,075.00	
647	St. Francis Medical Center	Maintex Inc	HOUSEKEEPING SUPPLIES	12/31/2019	\$2,500.00	
670	St. Francis Medical Center	MEDCOMP	DIALYSIS PRODUCTS	10/31/2020	\$4,623.00	
830	St. Francis Medical Center	MERCEDES MEDICAL INC	HISTOLOGY AND CYTOLOGY	1/31/2021	\$910.00	
838	St. Francis Medical Center	MICROTEK MEDICAL (ECOLAB)	OR EQUIPMENT DRAPES	3/31/2019	\$21,008.98	
871	St. Francis Medical Center	NEUROSTRUCTURES INC	SPINAL IMPLANTS AND PRODUCTS	6/22/2019	\$3,307.50	
896	St. Francis Medical Center	ORASURE TECHNOLOGIES INC	RAPID DIAGNOSTIC TEST KITS	10/31/2021	\$7,270.80	
906	St. Francis Medical Center	PACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)	LEASE-AS LANDLORD	12/31/2047	\$0.00	
907	St. Francis Medical Center	PACIFIC MEDICAL BUILDINGS L.P. (ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC)	LEASE-AS LANDLORD	12/31/2047		
919	St. Francis Medical Center	PENUMBRA INC	DIR PRODUCTS	1/31/2020	\$7,985.48	
920	St. Francis Medical Center	PENUMBRA INC	NEUROVASCULAR INT RADIOLOGY	8/31/2020		
992	St. Francis Medical Center	SMITH & NEPHEW INC ENDOSCOPY DIV	ARTHROSCOPY SUPPLIES	4/30/2019	\$223,752.45	

 $\begin{array}{l} 4" = "1" "DM_{US} \ 165807051 - 6.079742.0063" "" \\ US_Active \ 114483166 \ V-4 \underline{6} \end{array}$ 

DM\_US 165807051-13.079742.0063 US\_Active\114563955\V-1

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 123 of 147

993	St. Francis Medical Center	SMITH & NEPHEW INC WOUND MANAGEMENT	SKIN INTEGRITY: WOUND CARE	3/31/2020		
	Wiedledr Center	DIV	WOULD CHILL			
994	St. Francis Medical Center	SMITH & NEPHEW INC WOUND MANAGEMENT DIV	SKIN INTEGRITY: SKIN CLENSER	3/31/2020		
995	St. Francis Medical Center	SMITH & NEPHEW INC WOUND MANAGEMENT DIV	SURGICAL WOUND DEBRIDEMENT	5/31/2020		
996	St. Francis Medical Center	SMITH & NEPHEW ORTHOPEDIC	ORTHO TOTAL JOINT IMPLANTS	9/30/2019		
997	St. Francis Medical Center	SMITHS MEDICAL ASD INC	ARTERIAL BLOOD GAS KITS	11/30/2020		
998	St. Francis Medical Center	SMITHS MEDICAL ASD INC	SAFETY HUBER NEEDLES	2/28/2021		
999	St. Francis Medical Center	SMITHS MEDICAL ASD INC	INFUSION DEVICES	1/31/2020		
1000	St. Francis Medical Center	SMITHS MEDICAL ASD INC	PULSE OXIMETRY AND CAPNOGRAPHY	4/30/2021		
1001	St. Francis Medical Center	SMITHS MEDICAL ASD INC	SAFETY HYPODERMIC PRODUCTS	9/30/2019	\$6,818.23	
1002	St. Francis Medical Center	SMITHS MEDICAL ASD INC	SAFETY IV CATHETERS	9/30/2019		
1003	St. Francis Medical Center	SMITHS MEDICAL ASD INC	TRACHEOSTOMY TUBES	9/30/2021		
1004	St. Francis Medical Center	SMITHS MEDICAL ASD INC	TEMPERATURE MONITORING PRODUCTS	12/31/2018		
1005	St. Francis Medical Center	SMITHS MEDICAL ASD INC	REGIONAL ANESTHESIA TRAYS	1/31/2019		
1018	St. Francis Medical Center	ST JUDE MEDICAL	DIC PRODUCTS	6/30/2019	¢5(7.70	
1019	St. Francis Medical Center	ST JUDE MEDICAL	CARDIAC RHYTHEM MANAGEMENT	11/14/2020	\$567.79	
1021	St. Francis Medical Center	ST. FRANCIS LYNWOOD MEDICAL PLAZA, LLC	LEASE-AS TENANT	12/31/2021	\$97,282.52	
1022	St. Francis Medical Center	St. Francis Multispecialty Clinic	SUBLEASE AGREEMENT	1/31/2018	\$0.00	
1059	St. Francis Medical Center	STRYKER ORTHOPEDICS	ORTHOPEDIC TOTAL JOINT PRODUCTS	5/31/2020	\$200 744 50	
1060	St. Francis Medical Center	STRYKER ORTHOPEDICS	GENERAL ORTHO TRAUMA PRODUCTS	10/31/2019	\$398,744.58	
1129	St. Francis Medical Center	Verity Medical Foundation	SUBLEASE AGREEMENT	12/31/2020	\$7,700.00	

4" = "1" "DM\_US 165807051-6.079742.0063" "" US\_Active\114483166\V-4<u>6</u>

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 124 of 147

Ref #	Debtor	Contract Counterparty	Nature of Contact / Lease	Termination Date	Seton Medical Center & Coastside	St. Francis Medical Center	Verity Health System
93	V Hold.	Campus Physical Therapy	Lease-as Landlord	7/31/2027	\$0.00	\$0.00	\$0.00
138	V Hold.	Dave, Rajan, MD	Lease-as Tenant	9/30/2017	\$0.00	\$0.00	\$0.00
201	V Hold.	Hsiao, Michael	Lease-other	5/10/2017	\$0.00	\$0.00	\$0.00
244	V Hold.	Lee, Shu May MD, Inc.	Lease-as Landlord	2/28/2022	\$0.00	\$0.00	\$0.00
257	V Hold.	Longar, Susan MD	Lease-as Landlord	3/31/2022	\$0.00	\$0.00	\$0.00
266	VHS	Mariposa 2040 LLC	Lease-as Tenant	6/30/2047	\$0.00	\$0.00	\$0.00
289	V Hold.	Medicus Integrated Health Services, Inc.	Lease-as Landlord	2/28/2022	\$0.00	\$0.00	\$0.00
323	V Hold.	Mulligan, Timothy, MD	Lease-as Landlord	4/30/2022	\$0.00	\$0.00	\$0.00
329	V Hold.	Nantworks, LLC	Lease-office space	3/31/2019	\$0.00	\$0.00	\$34,722.48
362	V Hold.	Order of Malta Los Angeles, Free Clinic	Lease-as Landlord	6/30/2022	\$0.00	\$0.00	\$0.00
419	V Hold.	Rajan Davie, M.D.	Lease-as Landlord	9/30/2017	\$0.00	\$0.00	\$0.00
420	V Hold.	Ranavat, Amritlal MD	Lease-as Landlord	12/31/2017	\$0.00	\$0.00	\$0.00
421	V Hold.	Raymond Dugan Velasco, MD, Inc.	Lease-as Landlord	4/30/2022	\$0.00	\$0.00	\$0.00
472	V Hold.	Stefan, Michael R., M. D.	Lease-as Landlord	4/14/2022	\$0.00	\$0.00	\$0.00

 $\label{eq:states} \begin{array}{l} 4" = "1" "DM_US \ 165807051 - 6.079742.0063" "" \\ US_Active \ 114483166 \ V-4 \underline{6} \end{array}$ 

DM\_US 165807051-13.079742.0063 US\_Active\114563955\V-1

I

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 125 of 147

527	V Hold.	Uy, Santos MD	Lease-as Landlord	11/30/2021	\$0.00	\$0.00	\$0.00
538	V Hold.	Verity Medical Foundation	Lease-as Tenant	7/31/2022	\$0.00	\$0.00	\$0.00
561	V Hold.	Yamamoto, Kenneth MD	Lease-as Landlord	5/31/2019	\$0.00	\$0.00	\$0.00
Added	V Hold.	House Ear Institute	Tenant Lease	2/28/2021	\$0.00	\$0.00	\$0.00
Added	V Hold.	Rolando Mercader, M.D.	Tenant Lease	7/31/2020	\$0.00	\$0.00	\$0.00
Added	V Hold.	Diane Clark	Tenant Lease	Monthly	\$0.00	\$0.00	\$0.00
Added	V Hold.	Gregory Slack	Tenant Lease	Monthly	\$0.00	\$0.00	\$0.00
Added	V Hold.	Marco Monteon	Tenant Lease	Monthly	\$0.00	\$0.00	\$0.00

 $\label{eq:states} \begin{array}{l} 4" = "1" "DM_{US} \ 165807051 \ -6.079742.0063" "" \\ US_Active \ 114483166 \ V-4 \ \underline{6} \end{array}$ 

I

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 126 of 147

# **Schedule 1.7(I) - Permitted Exceptions**

None.

 $\begin{array}{l} 4" = "1" "DM\_US \ 165807051 \ 6.079742.0063" "" \\ US\_Active \ 114483166 \ V-4 \underline{6} \end{array}$ 

DM\_US 165807051-13.079742.0063 US\_Active\114563955\V-1

126 4/3/2020 Redline

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 127 of 147

# Schedule 1.8(bb) - Other Excluded Assets

None.

 $\begin{array}{l} 4" = "1" "DM\_US \ 165807051 \ 6.079742.0063" "" \\ US\_Active \ 114483166 \ V-4 \underline{6} \end{array}$ 

DM\_US 165807051-13.079742.0063 US\_Active\114563955\V-1

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 128 of 147

# Schedule 1.9(i) - Other Obligations

None.

 $\begin{array}{l} 4" = "1" "DM\_US \ 165807051 \ 6.079742.0063" "" \\ US\_Active \ 114483166 \ V-4 \underline{6} \end{array}$ 

DM\_US 165807051-13.079742.0063 US\_Active\114563955\V-1

I

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 129 of 147

# Schedule 2.4 – Seller's Knowledge of Breach or Default

None.

 $\begin{array}{l} 4" = "1" "DM\_US \ 165807051 \ 6.079742.0063" "" \\ US\_Active \ 114483166 \ V-4 \underline{6} \end{array}$ 

DM\_US 165807051-13.079742.0063 US\_Active\114563955\V-1

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 130 of 147

# Schedule 2.5 - Brokers and Finders (Sellers)

Cain Brothers, a division of KeyBanc Capital Markets.

 $\begin{array}{l} 4" = "1" "DM\_US \ 165807051 \ 6.079742.0063" "" \\ US\_Active \ 114483166 \ V-4 \underline{6} \end{array}$ 

DM\_US 165807051-13.079742.0063 US\_Active\114563955\V-1

13

130 4/3/2020 Redline

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 131 of 147

# **Schedule 2.7 - Compliance (Sellers)**

See Confidential Side Disclosure Letter, dated of even date herewith.

 $\label{eq:states} \begin{array}{l} 4" = "1" "DM_{US} \ 165807051 \ -6.079742.0063" "" \\ US_Active \ 114483166 \ V-4 \ \underline{6} \end{array}$ 

DM\_US 165807051-13.079742.0063 US\_Active\114563955\V-1

14

131 4/3/2020 Redline

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 132 of 147

# Schedule 2.8 - Required Consents (Sellers)

None, other than those consents required under §365 of the Bankruptcy Code.

 $\label{eq:states} \begin{array}{l} 4" = "1" "DM_{US} \ 165807051 \ -6.079742.0063" "" \\ US_Active \ 114483166 \ V-4 \ \underline{6} \end{array}$ 

DM\_US 165807051-13.079742.0063 US\_Active\114563955\V-1

#### Schedule 2.9(b) - Environmental Matters (Sellers)

Underground storage tank non-compliance notice issued by Los Angeles Department of Water and Power (LADWP) in September 2016 to Saint Francis Medical Center. Seller removed tank per closure permit issued to Saint Francis Medical Center by LADWP. There was no soil contamination or clean up necessary.

 $\begin{array}{l} 4" = "1" "DM\_US \ 165807051 \ -6.079742.0063" "" \\ US\_Active \ 114483166 \ V-4 \ \underline{6} \end{array}$ 

DM\_US 165807051-13.079742.0063 US\_Active\114563955\V-1

I

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 134 of 147

#### Schedule 2.11 - Missing Material Licenses (Sellers)

SFMC timely submitted the laboratory license for renewal and has received confirmation that CDPH Laboratory Field Services has received the renewal, is processing it, and the hospital may continue testing without penalty.

 $\label{eq:states} \begin{array}{l} 4" = "1" "DM_US \ 165807051 \ -6.079742.0063" "" \\ US_Active \ 114483166 \ V-4 \ \underline{6} \end{array}$ 

DM\_US 165807051-13.079742.0063 US\_Active\114563955\V-1

Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 135 of 147

#### **Schedule 2.13 - Non-Bankruptcy Court Legal Proceedings (Sellers)**

See attached folder "Schedule 2.13 - Non-Bankruptcy Court Legal Proceedings"

 $\label{eq:states} \begin{array}{l} 4" = "1" "DM_{US} \ 165807051 \ -6.079742.0063" "" \\ US_Active \ 114483166 \ V-4 \ \underline{6} \end{array}$ 

DM\_US 165807051-13.079742.0063 US\_Active\114563955\V-1

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 136 of 147

# Schedule 2.19 - List of Excluded Individuals (Sellers)

None.

 $\begin{array}{l} 4" = "1" "DM\_US \ 165807051 \ 6.079742.0063" "" \\ US\_Active \ 114483166 \ V-4 \underline{6} \end{array}$ 

DM\_US 165807051-13.079742.0063 US\_Active\114563955\V-1

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 137 of 147

# Schedule 3.4 - No Violation (Purchaser)

None.

 $\begin{array}{l} 4" = "1" "DM\_US \ 165807051 \ 6.079742.0063" "" \\ US\_Active \ 114483166 \ V-4 \underline{6} \end{array}$ 

DM\_US 165807051-13.079742.0063 US\_Active\114563955\V-1

I

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 138 of 147

# Schedule 3.7 - Legal Proceedings (Purchaser)

None.

 $\begin{array}{l} 4" = "1" "DM\_US \ 165807051 \ 6.079742.0063" "" \\ US\_Active \ 114483166 \ V-4 \underline{6} \end{array}$ 

DM\_US 165807051-13.079742.0063 US\_Active\114563955\V-1

21

138 4/3/2020 Redline

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 139 of 147

# **Schedule 5.3 - Hospital Employees**

See attached folder "Schedule 5.3 Hospital Employees"

 $\label{eq:states} \begin{array}{l} 4" = "1" "DM_{US} \ 165807051 \ -6.079742.0063" "" \\ US_Active \ 114483166 \ V-4 \ \underline{6} \end{array}$ 

DM\_US 165807051-13.079742.0063 US\_Active\114563955\V-1

22

139 4/3/2020 Redline

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 140 of 147

#### Schedule 11.3(b) - Tax Allocation Schedule

The Parties agree that within sixty (60) days subsequent to the Closing Date (the "Allocation **Date**"), the Purchase Price shall be allocated to the Assets using the residual method and based upon the agreed fair market value of the various Classes of the Assets set forth below, as reasonably determined by the Purchaser and Seller, consistent with Section 1060 of the Internal Revenue Code. In particular, the Parties shall mutually agree upon the tax allocation associated with the Assets of Verity Holdings.

Class of Assets
Class I Assets (cash and general deposit accounts)
Class II Assets (Publicly traded stock; U.S. Government Securities)
Class III Assets (Accounts Receivable)
Class IV Assets (Inventory)
Class V Assets (Property, Plant and Equipment)
Class VI Assets (Intangibles other than Goodwill)
Class VII Assets (Goodwill)

4" = "1" "DM\_US 165807051-6.079742.0063" "" US\_Active\114483166\V-4<u>6</u>

Em	ployment Practices Liability Litig	gation
Plaintiff/Grievant	Venue	Case Number(s)
Ahinasi, Sonnia	San Bernardino County	CIVDS1822881
	Superior Court	
Aliazis, Tricia	EEOC Charge	37A-2019-03433-C;
	DFEH Charge	201903-05587626
Aragon, Jasmine	Los Angeles County Superior Court	BC717447
Brown, Star		Complaint Not Yet Filed
Capizzi, Renee		Mediation
Carcamo, Rosa	Los Angeles County Superior Court	TC029195
Carter, Margena		Right To Sue Letter
Dolmo, Norma	EEOC Charge DFEH Charge	480-2019-01625
Engelman, Ivonne *settlement in process	Los Angeles County Superior Court	BC704524
Harris, Leonard	Los Angeles County Superior Court	19STVC38595
Martin, Joan	EEOC Charge	37A-2018-02446
Nguyen, Diane	Los Angeles County Superior Court	BC713041
Nguyen, Natalie	Los Angeles County Superior Court; EEOC Charge	Small Claims 19CMSC01398; 480-2019-00050
Raj, Aaron		Notice Of Intent
Rapp, Dwight		Lawsuit Not Yet Filed - SOL of 4/17/20
Sorto, Cynthia	Los Angeles County Superior Court	20STCV04718
Waheed, Wahidi, Ernesto Madrgil, et al	San Mateo Superior Court	18-CIV-03214
Xue, Baoru		Demand Letter
	General Liability Litigation	
Plaintiff	Venue	Case Number(s)
Arreola, Maria		Notice Of Intent
Burell, Lisa		Request For Payment (auto-related)
Cortes, Synthia	Los Angeles County Superior	BC659504

4"="1" "DM\_US 165807051-6.079742.0063" "" US\_Active\114483166\V-4<u>6</u>

DM\_US 165807051-13.079742.0063 US\_Active\114563955\V-1

	Court	
Hernandez, Iris	Los Angeles County Superior Court	BC681228
Khajekarrameddin,	Los Angeles County Superior	BC712663
Frangis	Court	
Woolum, Dale		Notice Of Intent
	Ialpractice (Professional Liability	
Plaintiff	Venue	Case Number(s)
Aguirre Huerta, Rafael	Los Angeles County Superior Court	19STCV29922
Allen, III, LeRoy & Perkins, LaTrice		Notice Of Intent
Ancira, Mateo, by and through guardian ad litem Yolanda Frias	Los Angeles County Superior Court	19STCV23682
Arencibia, Vivian	Los Angeles County Superior Court	BC673423
Birdsong, Erica		Notice Of Intent
Braun, Brian	Los Angeles County Superior Court	19STCV13074
Bustamante, Marvela	Los Angeles County Superior Court	19STCV22710
Childress, Alberta	Los Angeles County Superior Court	20STCV00666
Correa, Maximino	Los Angeles County Superior Court	BC625271
De Los Angeles, Maria		Notice Of Intent
Durham, Breta	Los Angeles County Superior Court	BC680199
Ferguson, William	Los Angeles County Superior Court	19STCV08618
Flores, Luz	Los Angeles County Superior Court	19STCV38243
Fuentes, Federico	Los Angeles County Superior Court	19STCV08306
Hernandez, Daniel	Los Angeles County Superior Court	BC687878
Iniguez, Aida	Los Angeles County Superior Court	BC616038
Lopez, Monique	Potentially Compensable Incident	No Complaint, Demand Letter, Or Notice Of Intent At This Time

 $\begin{array}{l} 4"="1" "DM_{US} 165807051 - 6.079742.0063" "" \\ US_Active 114483166 \ V-4 \underline{6} \end{array}$ 

# Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 143 of 147

Mitchell, Alice		Letter To Preserve Record
Morataya, Jessica	Los Angeles County Superior Court	BC712998
Navarro, Richard	Los Angeles County Superior Court	BC669695
Ramirez, Guadalupe		Notice Of Intent
Robles, Josefina	Los Angeles County Superior Court	8C697012
Rodriguez, Yolanda by and through, Successor-in-Interest Corinna Jimenez, et al	Los Angeles County Superior Court	Notice Of Intent And Unserved Complaint Filed, Case No. 19STCV29707
Rojas, John	Los Angeles County Superior Court	Notice Of Intent And Unserved Complaint Filed, Case No. 19STCV29868 (SFMC not named or served)
Rubio, Alicia, by and through, guardian ad litem Cariola Rubio	Los Angeles County Superior Court	18STCV04410
Saleh, Ebtissam	Los Angeles County Superior Court	BC718838
Sanchez, Isabel		Notice Of Intent
Sprownson, Tammy	Los Angeles County Superior Court	5C724114
Vazquez, Eduardo, by and through biological mother, Denia Carolina Alvardo	Los Angeles County Superior Court	BC609128
77 1 4 1 1		BC691431
Zavala, Adamari, by and through, guardian ad litem, A11 Adanesne Quiñones	Los Angeles County Superior Court	
and through, guardian ad litem, A11 Adanesne		

Workers' Compensation Cases			
Grievant	Date of Loss	Case Number(s)	

4"="1" "DM\_US 165807051-6.079742.0063" "" US\_Active\114483166\V-4<u>6</u>

DM\_US 165807051-13.079742.0063 US\_Active\114563955\V-1

Acosta, Leslie	5/24/2018	ADJ12476569
Navarro Gutierrez,	10/23/2018	ADJ12149342
Giselle	10/25/2010	110312119312
Sanders, Valencia	11/17/2016	ADJ10754107
Haith, Leilani	11/17/2016	ADJ1066160
Sutton, Tina	3/5/2013	ADJ11216966
Jean, Marlene	3/2/2010	ADJ7610324
Ponce, Diana	10/19/2015	
Balderas-St. Pierre, Luz	10/17/2015	
Balderas-St. Pierre, Luz	11/18/2016	
Monteon, Marco	9/20/2016	ADJ12088746
Padilla, Rebecca	1/25/2009	
Werner, Wendie	10/17/2010	ADJ7620832
Monteon, Angie	9/5/2012	ADJ9223401
Gardea, Richard	4/26/2004	ADJ11072161
Casey, Rochelle	11/7/2015	ADJ11332260
Bombay, Maura	9/16/2006	ADJ5815856
Anaya, Gloria	3/21/2009	ADJ7228730
Jones, Lisa	1/6/2004	ADJ772382
Herrera, Martha	1/31/2012	ADJ8485301
Gardea, Richard	5/8/2014	ADJ11073773
Silva, Marcelina	1/14/2015	
Silva, Marcelina	4/6/2016	
Anaya, Gloria	8/14/2009	ADJ7228730
Lee, Belinda	1/13/2015	ADJ12991056
Mendez, Maria	4/20/2015	ADJ12209*530
Quiñonez, Maria	7/10/2015	
Hernandez, Lorena	4/8/2013	ADJ8942433
Reed, Charotta	2/18/2010	ADJ7309641
Wang, Xiaoling	1/11/2012	ADJ8180221
Ganzon, Nancy	3/20/2012	ADJ8391108
Martinez, Marina C	2/6/2015	ADJ11955687
Mendez, Maria	2/8/2016	ADJ12209530
Infante, Elida	5/14/2006	ADJ4439918
Harris, Gloria	7/27/1984	ADJ1323480
Wills, Ora	8/17/1989	ADJ2755470
McCoy, Virginia	1/24/2017	ADJ10802194
Maravilla, Fernando		AD110772424
	2/15/2017	ADJ10772424
Sears, Andrea	2/15/2017 2/28/2017	ADJ10//2424
Sears, Andrea Giannelli, Michelle		ADJ10772424

4" = "1" "DM\_US 165807051-6.079742.0063" ""

US\_Active\114483166\V-46

I

27

Martirosyan, Alik	6/6/2017	
Epps, Cleveland	6/23/2017	ADJ11035557
Plaza, Caroline	6/27/2017	ADJ12089432
Elizalde Maria	6/27/2017	ADJ12009492 ADJ11186591
Harris, Stanley	11/22/2017	ADJ11498006
Espinoza, Ramon	10/10/2017	ADJ11498000
Harris, Stanley	5/15/2018	ADJ11498006
Diaz, Gabriela	6/9/2018	ADJ11498000
Malone, Veronica	12/21/2017	ADJ12557870
Medal, Sonia	1/9/2018	ADJ12557870 ADJ11664975
Gardea, Richard	6/30/2016	
Torres-Olivares, Maria	1/11/2018	
1		ADJ11799845
Dacio, Joicelyne Roots Chandra	<u>12/31/2017</u> 1/26/2018	ADJ12964974
	2/1/2018	ADJ12904974
King, LaVonda Monteon Marco	2/1/2018	AD112000700
	2/8/2018	ADJ12088798
Zavala, Mayra		
Jones, Leatrice	4/5/2018	AD111205500
Casey, Rochelle	7/18/2018	ADJ11395500
Ramirez, Teresita	8/22/2018	
Smith, Rita	10/1/2018	A D II 100/2/1
Meza, Maria	10/4/2018	ADJ11906361
Scrubb, Jacqueline	11/16/2018	
Chavez, Sandra	12/4/2018	
Boub, Theodore	2/13/2018	ADJ11965517
Armstrong, Gary	2/1/2019	ADJ2488567
Rodriguez, Isaac	2/27/2019	ADJ2050556
Ricketts, Chanel	3/9/2019	ADJ12129607
Cariaso, Sylvester	7/3/2014	
Monteon, Marco Jr		ADJ12092140
Duffy, Earl	5/23/2019	
Jadraque-Olmos, Anna	6/10/2019	ADJ1271881
Dominguez, Cassidy	5/21/2019	ADJ12411106
Williams, Linda	7/15/2019	
Hill, Larry	5/17/2019	
Palomares, Reyna	7/31/2019	
Pham, Patty	7/26/2019	ADJ12566853
Duran, Monica	8/2/2019	ADJ12795212
Giannelli, Michelle	8/27/2019	
Santos, Grace Mary	9/6/2019	
Morales Silva, Angelita	8/14/2019	ADJ12504328

4" = "1" "DM\_US 165807051-6.079742.0063" ""

US\_Active\114483166\V-46

I

28

Maralas Silva Angelita	8/21/2010	1012505619
Morales Silva, Anqelita	8/21/2019	ADJ2505618
Duronslet, Devine	9/21/2019	
Alonza, Kathryn	9/20/2019	AD1125570(0
		ADJ12557869
Partida, Aurelio 10/1/2019		AD112(40007
Washington, Shirlee	10/6/2019	ADJ12640087
Plaza, Caroline	5/30/2019	ADJ12748038
Harris, Stanley	11/8/2019	ADJ12983668
Yepez, Consuelo	11/12/2019	
Acosta, Amber	11/1/2019	
Pinks Jeannene	9/11/2019	ADJ12675128
Marin, Solita	12/18/2019	
Marinoble, Tawny	12/31/2019	
Roots, Chandra	10/15/2019	ADJ12964974
Roots, Chandra	2/26/2018	ADJ13012642
	Appeals	
Stephenson,	Aquisto & Coleman; And Hoope Bookman	er, Lundy and
TT .		
Union	Grievant/Case Name	Grievance Number
United Nurses	Grievant/Case Name Carcamo, Rosa	Grievance Number SM 04-18
United Nurses Associations of		
United Nurses Associations of CA/Union of Health		
United Nurses Associations of	Carcamo, Rosa	
United Nurses Associations of CA/Union of Health Care Professionals	Carcamo, Rosa Labor Grievances	SM 04-18
United Nurses Associations of CA/Union of Health Care Professionals Union	Carcamo, Rosa Labor Grievances Grievant/Case Name	
United Nurses Associations of CA/Union of Health Care Professionals	Carcamo, Rosa Labor Grievances	SM 04-18
United Nurses Associations of CA/Union of Health Care Professionals Union	Carcamo, Rosa Labor Grievances Grievant/Case Name Class Action - for SEIU Bargaining Unit Class Action - for SEIU	SM 04-18
United Nurses Associations of CA/Union of Health Care Professionals Union SEIU	Carcamo, Rosa Labor Grievances Grievant/Case Name Class Action - for SEIU Bargaining Unit	SM 04-18
United Nurses Associations of CA/Union of Health Care Professionals Union SEIU	Carcamo, Rosa Labor Grievances Grievant/Case Name Class Action - for SEIU Bargaining Unit Class Action - for SEIU	SM 04-18
United Nurses Associations of CA/Union of Health Care Professionals Union SEIU SEIU	Carcamo, Rosa Labor Grievances Grievant/Case Name Class Action - for SEIU Bargaining Unit Class Action - for SEIU Bargaining Unit	SM 04-18 Grievant Number
United Nurses Associations of CA/Union of Health Care Professionals Union SEIU SEIU	Carcamo, Rosa Labor Grievances Grievant/Case Name Class Action - for SEIU Bargaining Unit Class Action - for SEIU Bargaining Unit Employee given Corrective Action Plan (CAP) without union representative	SM 04-18 Grievant Number
United Nurses Associations of CA/Union of Health Care Professionals Union SEIU SEIU	Carcamo, Rosa Labor Grievances Grievant/Case Name Class Action - for SEIU Bargaining Unit Class Action - for SEIU Bargaining Unit Employee given Corrective Action Plan (CAP) without	SM 04-18 Grievant Number
United Nurses Associations of CA/Union of Health Care Professionals Union SEIU SEIU UNAC	Carcamo, Rosa Labor Grievances Grievant/Case Name Class Action - for SEIU Bargaining Unit Class Action - for SEIU Bargaining Unit Employee given Corrective Action Plan (CAP) without union representative Union grieved and written/verbal warning given	SM 04-18 Grievant Number SFMC SB 01-20
United Nurses Associations of CA/Union of Health Care Professionals Union SEIU SEIU UNAC	Carcamo, Rosa Labor Grievances Grievant/Case Name Class Action - for SEIU Bargaining Unit Class Action - for SEIU Bargaining Unit Employee given Corrective Action Plan (CAP) without union representative Union grieved and	SM 04-18 Grievant Number SFMC SB 01-20
United Nurses Associations of CA/Union of Health Care Professionals Union SEIU SEIU UNAC	Carcamo, Rosa Labor Grievances Grievant/Case Name Class Action - for SEIU Bargaining Unit Class Action - for SEIU Bargaining Unit Employee given Corrective Action Plan (CAP) without union representative Union grieved and written/verbal warning given to employee Union grieved disciplinary	SM 04-18 Grievant Number SFMC SB 01-20
United Nurses Associations of CA/Union of Health Care Professionals Union SEIU SEIU UNAC	Carcamo, Rosa Labor Grievances Grievant/Case Name Class Action - for SEIU Bargaining Unit Class Action - for SEIU Bargaining Unit Employee given Corrective Action Plan (CAP) without union representative Union grieved and written/verbal warning given to employee Union grieved disciplinary action and employee received	SM 04-18 Grievant Number SFMC SB 01-20 SM 04-20
United Nurses Associations of CA/Union of Health Care Professionals Union SEIU SEIU UNAC	Carcamo, Rosa Labor Grievances Grievant/Case Name Class Action - for SEIU Bargaining Unit Class Action - for SEIU Bargaining Unit Employee given Corrective Action Plan (CAP) without union representative Union grieved and written/verbal warning given to employee Union grieved disciplinary action and employee received written warning. Union	SM 04-18 Grievant Number SFMC SB 01-20 SM 04-20
United Nurses Associations of CA/Union of Health Care Professionals Union SEIU SEIU UNAC	Carcamo, Rosa Labor Grievances Grievant/Case Name Class Action - for SEIU Bargaining Unit Class Action - for SEIU Bargaining Unit Employee given Corrective Action Plan (CAP) without union representative Union grieved and written/verbal warning given to employee Union grieved disciplinary action and employee received	SM 04-18 Grievant Number SFMC SB 01-20 SM 04-20

4" = "1" "DM\_US\_165807051-6.079742.0063" "" US\_Active\114483166\V-4<u>6</u>

I

#### Case 2:18-bk-20151-ER Doc 6669-8 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 15 - 4/3/2020 Redline Page 147 of 147

	CAP	
UNAC v. SFMC	National Labor Relations	ULP-21-CA-231856
	Board - Region 21	
UNAC	Union grieved and CAP	SM 06-20

4" = "1" "DM\_US 165807051-6.079742.0063" "" US\_Active\114483166\V-4<u>6</u>

I

30

Case 2:18-bk-20151-ER Doc 6669-9 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 16 - Final APA Email Page 1 of 2

#### Esposito, Zach

Subject:

RE: EXTERNAL: FW: Prime/St. Francis- Revised APA

From: Garms, R. Matthew <<u>matthew.garms@dentons.com</u>>
Sent: Friday, April 3, 2020 12:23 PM
To: Joel Richlin (PHMI) <<u>JRichlin@primehealthcare.com</u>>; Moyron, Tania M. <<u>tania.moyron@dentons.com</u>>; Paul,
Elspeth <<u>ElspethPaul@verity.org</u>>; David Galfus <<u>dgalfus@thinkbrg.com</u>>; Chadwick, Peter (VHS)
<<u>pchadwick@thinkbrg.com</u>>; Moloney, Jim (<u>JMoloney@cainbrothers.com</u>) <<u>JMoloney@cainbrothers.com</u>>; John
Schlant [ThinkBRG] <<u>ischlant@thinkbrg.com</u>>; Koffroth, Nick <<u>nick.koffroth@dentons.com</u>>; Montgomery, Claude D.
<<u>claude.montgomery@dentons.com</u>>; Maizel, Samuel R. <<u>samuel.maizel@dentons.com</u>>; Adcock, Rich
<<u>RichAdcock@verity.org</u>>; Beith, Carsten <<u>cbeith@cainbrothers.com</u>>; Project Phoenix
(<u>project\_phoenix@keybank.com</u>) <<u>project\_phoenix@keybank.com</u>>
Cc: Gertler, Gary <<u>ggertler@mwe.com</u>>; Olson, Christopher <<u>Colson@mwe.com</u>>; Reisner, Jeffrey
<<u>Jreisner@mwe.com</u>>; Steve Aleman (PHMI) <<u>SAleman2@primehealthcare.com</u>>

#### [External Email]

All, please see attached revised clean and redlined drafts of the Asset Purchase Agreement, along with a draft of the purchase price adjustment schedule.

大成DENTONS R. Matthew Garms

D +1 816 460 2649 | US Internal 22649 matthew.garms@dentons.com Bio | Website

Dentons US LLP

Larraín Rencoret > Hamilton Harrison & Mathews > Mardemootoo Balgobin > HPRP > Zain & Co. > Delany Law > Dinner Martin > Maclay Murray & Spens > Gallo Barrios Pickmann > Muñoz > Cardenas & Cardenas > Lopez Velarde > Rodyk > Boekel > OPF Partners

Dentons is a global legal practice providing client services worldwide through its member firms and affiliates. This email may be confidential and protected by legal privilege. If you are not the intended recipient, disclosure, copying, distribution and use are prohibited; please notify us immediately and delete this copy from your system. Please see <u>dentons.com</u> for Legal Notices.

From: Joel Richlin (PHMI)
Sent: Friday, April 03, 2020 12:01 AM
To: Moyron, Tania M. ; Paul, Elspeth ; David Galfus ; Chadwick, Peter (VHS) ; Garms, R. Matthew ; Moloney, Jim (<u>JMoloney@cainbrothers.com</u>) ; John Schlant [ThinkBRG] ; Koffroth, Nick ; Montgomery, Claude D. ; Maizel, Samuel R. ; Adcock, Rich ; Beith, Carsten ; Project Phoenix (<u>project\_phoenix@keybank.com</u>)
Cc: ggertler@mwe.com; Colson@mwe.com; Jreisner@mwe.com; Steve Aleman (PHMI)
Subject: Prime/St. Francis- Redline of Sale Order

#### Case 2:18-bk-20151-ER Doc 6669-9 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Exhibit 16 - Final APA Email Page 2 of 2

# [External Sender]

Hi All,

Attached is a clean and redline of the template Sale Order. My understanding is that Dentons is working on a redline that reflects the parties' negotiations and agreements today. I reviewed the AR language with Elspeth and that should be good to go. I also sent Elspeth the language for the structure of the MAE provision and my understanding is that Cain has agreed on the calculation methodology with Steve and Cain is drafting that. So we are waiting in Cain's draft of that calculation and otherwise waiting on Dentons to turn a redline of the rest.

Please let me know if anything is in our court. We are otherwise prepared for the wire transfer and will be ready to review and finalize the redline in the morning. We can also review tonight if available.

Joel

#### A. Joel Richlin, Esq.

Vice President, Deputy General Counsel, & Chief Litigation Counsel 3480 E. Guasti Road, Ontario, CA 91761 Direct: 909-235-4235 | Facsimile: 909-235-4316 Email: JRichlin@primehealthcare.com





The information contained in this email may be confidential and/or legally privileged, including HIPAA Protected Health Information or information that is proprietary or trade secret. It has been sent for the sole use of the intended recipient(s). If you are not the intended recipient, you are hereby notified that any unauthorized review, use, disclosure, dissemination, distribution, or copying of this communication is strictly prohibited. If you received this communication in error, please inform the sender and destroy all copies of the message. The organization accepts no liability for any damage caused by transmittance of this email. Thank you.

This message is a PRIVATE communication. This message and all attachments are a private communication sent by a law firm and may be confidential or protected by privilege. If you are not the intended recipient, you are hereby notified that any disclosure, copying, distribution or use of the information contained in or attached to this message is strictly prohibited. Please notify the sender of the delivery error by replying to this message, and then delete it from your system. Our <u>Privacy Policy</u> explains how we may use your personal information or data and any personal information or data provided or made available to us. Thank you.

Please visit http://www.mwe.com/ for more information about our Firm.

	Case 2:18-bk-20151-ER Doc 6669-10 Fi Desc Prime Healthcare Services Inc.s	
1 2 3 4	Jason D. Strabo (SBN 246426) MCDERMOTT WILL & EMERY LLP 2049 Century Park East, Suite 3200 Los Angeles, California 90067-3206 Telephone: (310) 788-4125 Facsimile: (310) 277-4730 Email: jstrabo@mwe.com	
5 6 7 8	James W. Kapp, III (admitted <i>pro hac vice</i> ) MCDERMOTT WILL & EMERY LLP 444 West Lake Street, Suite 4000 Chicago, IL 60606-0029 Telephone: (312) 372-2000 Facsimile: (312) 984-7700 E-mail: jkapp@mwe.com	
9	Attorneys for Prime Healthcare Services, Inc.	
10		BANKRUPTCY COURT FORNIA – LOS ANGELES DIVISION
11	In re:	Lead Case No. 2:18-bk-20151-ER
12		
13	VERITY HEALTH SYSTEM OF, CALIFORNIA, INC., <i>et al.</i> ,	Jointly Administered With: Case No. 2:18-bk-20162-ER
14	Debtors and Debtors In Possession.	Case No. 2:18-bk-20163-ER Case No. 2:18-bk-20164-ER
15	Possession.	Case No. 2:18-bk-20165-ER Case No. 2:18-bk-20167-ER
16	<ul> <li>Affects All Debtors</li> <li>Affects Verity Health System of</li> </ul>	Case No. 2:18-bk-20168-ER Case No. 2:18-bk-20169-ER
	California, Inc.	Case No. 2:18-bk-20171-ER
17	□ Affects O'Connor Hospital	Case No. 2:18-bk-20172-ER Case No. 2:18-bk-20173-ER
18	<ul> <li>Affects Saint Louise Regional Hospital</li> <li>Affects St. Francis Medical Center</li> </ul>	Case No. 2:18-bk-20175-ER
19	$\square$ Affects St. Vincent Medical Center	Case No. 2:18-bk-20176-ER Case No. 2:18-bk-20178-ER
20	□ Affects Seton Medical Center	Case No. 2:18-bk-20179-ER
20	<ul> <li>Affects O'Connor Hospital Foundation</li> <li>Affects Saint Louise Regional Hospital</li> </ul>	Case No. 2:18-bk-20180-ER Case No. 2:18-bk-20181-ER
21	Foundation	
22	□ Affects St. Francis Medical Center of	Hon. Ernest M. Robles
23	Lynwood Foundation Affects St. Vincent Foundation	PRIME HEALTHCARE SERVICES, INC.'S
	$\Box$ Affects St. Vincent Dialysis Center, Inc.	EVIDENTIARY OBJECTIONS
24	□ Affects Seton Medical Center	Relates to Docket Nos. 6645 and 6662
25	Foundation <ul> <li>Affects Verity Business Services</li> </ul>	
26	□ Affects Verity Medical Foundation	Hearing Date and Time:
	□ Affects Verity Holdings, LLC	Date: October 6 2021
27	□ Affects De Paul Ventures, LLC	Time: 10:00 a.m.
28		1

Possession.	

#### Case 2:18-bk-20151-ER Doc 6669-10 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Prime Healthcare Services Inc.s Evidentiary Objections Page 3 of 11

1

11

2 Declaration of Peter Chadwick (the "Chadwick Decl.") or submitted in support of the Post-3 Effective Date Debtors and Liquidating Trustee's Memorandum in Opposition to Prime Healthcare 4 Services, Inc.'s Motion to Enforce Provisions of the Asset Purchase Agreement Pertaining to 5 Accounts Receivable Adjustment; Declaration of Peter Chadwick [Docket No. 6662] (the "Objection" filed by the "Debtors") filed in response to Prime Healthcare Services, Inc.'s Motion 6 7 to Enforce Provisions of the Asset Purchase Agreement Pertaining to Accounts Receivable 8 Adjustment; Declaration of A. Joel Richlin in Support Thereof; Declaration of Ken Wheeler in 9 Support Thereof; Declaration of Steve Aleman in Support Thereof [Docket No. 6645] (the 10 "Motion"), as follows:

12	Material Objected To:	Specific Objections:
13	1. <u>Chadwick Decl., P11, Ins. 22:23-26.</u>	1. <u>Best Evidence Rule</u> (Fed. R. Evid.
14	"[T]he schedule exchanged during that review	1008 ("[a]n original writing, recording, or
15	plainly shows that 'S-9 COUNTY TRAMA'	photograph is required in order to prove its
16	was considered and explicitly included in	content unless these rules or a federal
17	calculating the A/R Target Amount."	statute provides otherwise"); Lack of
18		Personal Knowledge (Fed. R. Evid. 602)
19		("A witness may testify to a matter only if
20		evidence is introduced sufficient to support
21		a finding that the witness has personal
22		knowledge of the matter"); <u>Hearsay</u> (Fed.
23		R. Evid. 801); <u>Relevance ((Fed. R. Evid.</u>
24		402) ("Irrelevant evidence is not
25		admissible"); <u>Prejudice</u> ((Fed. R. Evid. 403)
26		("[t]he court may exclude relevant evidence
27		if its probative value is substantially
28		

## Case 2:18-bk-20151-ER Doc 6669-10 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Prime Healthcare Services Inc.s Evidentiary Objections Page 4 of 11

1	Material Objected To:	Specific Objections:
2		outweighed by a danger of one or more of
3		outweighed by a danger of one or more of
4		the following: unfair prejudice, confusing
5		the issues, misleading the jury, undue delay,
6		wasting time, or needlessly presenting
7		cumulative evidence"); <u>Improper Opinion</u>
8		Testimony (Fed. R. Evid. 702); Improper
9		Testimony Regarding Legal Matter (United
10		<i>States v. Boulware</i> , 558 F.3d 971, 975 (9 <sup>th</sup>
11		Cir. 2009); Incomplete / Misstates Contents
12		of Referenced Document; Failure to
12		Authenticate (Fed. R. Evid. 901; Orr v.
14		Bank of America, NT & SA, 285 F.3d 764,
15		773 ("[a]uthentication is a condition
16		precedent to admissibility").
10		
		Through this testimony, Mr. Chadwick
18		seeks to describe the schedule exchanged
19		during due diligence review, but he provides
20		no insight as to whether he personally
21		prepared the schedule, the timing of such
22		preparation, how the Parties allegedly
23		considered the schedule, or whether he
24		personally negotiated with Prime pertaining
25		to such schedule. Nonetheless, Mr.
26		Chadwick purports to testify as to the
27		contents and legal effect of a document that
28		

## Case 2:18-bk-20151-ER Doc 6669-10 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Prime Healthcare Services Inc.s Evidentiary Objections Page 5 of 11

1		Material Objected To:	Specific Objections:
2			proceeded on the relevant contracts
3			preceded entry into the relevant contracts,
4			was superseded by multiple subsequent
5			drafts, and which was superseded by a fully
6			integrated written APA subject to an express
7			merger clause. Mr. Chadwick's
8			characterization of this document, which is
9			nowhere incorporated into the APA, is
10			therefore irrelevant to the apparent purposes
11			for which it is offered – namely, the
12			interpretation of section 1.7(p) of the
13			integrated APA, whether the parties
14			intended to remove "trauma payments" from
15			the APA. Further, even if the schedule at
16			issue were somehow relevant, Mr.
17			Chadwick utterly fails to authenticate the
18			document attached as Exhibit C, and even if
19			he had, the document speaks for itself, and
20			Mr. Chadwick's attempts to characterize the
20			documents contents or legal effect are
			unnecessary and improper.
22	2.	Exhibit C to Debtor's Objection.	2. <u>Failure to Authenticate (Fed. R.</u>
23			Evid. 901; Orr v. Bank of America, NT &
24			SA, 285 F.3d 764, 773 ("[a]uthentication is
25			a condition precedent to admissibility");
26			Relevance ((Fed. R. Evid. 402) ("Irrelevant
27			evidence is not admissible"); Prejudice
28			

## Case 2:18-bk-20151-ER Doc 6669-10 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Prime Healthcare Services Inc.s Evidentiary Objections Page 6 of 11

1	Material Objected To:	Specific Objections:
2		((E-1 D E-:: 1 402) (%[4]].
3		((Fed. R. Evid. 403) ("[t]he court may
4		exclude relevant evidence if its probative
5		value is substantially outweighed by a
6		danger of one or more of the following:
7		unfair prejudice, confusing the issues,
8		misleading the jury, undue delay, wasting
9		time, or needlessly presenting cumulative
10		evidence").
11		The sole manner in which the Debtors
12		purport to authenticate the schedule attached
13		as Exhibit C to their Opposition is through
14		the testimony of Peter Chadwick referenced
15		in Objection No. 1 set forth herein. As
16		noted in that objection, Mr. Chadwick
17		provides no insight as to whether he
18		personally prepared the schedule, the timing
19		of such preparation, how the Parties
20		allegedly considered the schedule, or
21		whether he personally negotiated with Prime
22		pertaining to such schedule. For that reason
23		and the reasons set forth more fully in
24		Objection No. 1, the Debtors therefore fail
25		to properly authenticate Exhibit C. Because
26		Exhibit C is not properly authenticated, it is
27		legally irrelevant and references to it are
28		

## Case 2:18-bk-20151-ER Doc 6669-10 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Prime Healthcare Services Inc.s Evidentiary Objections Page 7 of 11

	1
	1
	2
	3
	4
	5
	6
	7
	8
	9
1	0
1	1
1	2
1	3
1	4
1	5
1	6
1	7
1	8
1	9
2	0
2	1
2	2
2	3
2	4
2	5
2	6
2	7
2	8

Material Objected To:	Specific Objections:
	unduly prejudicial. Exhibit C should be
	stricken.
3. <u>Chadwick Decl., ₱ 17, lns. 24:24-25:4.</u>	3. <u>Lack of Personal Knowledge</u> (Fed.
"But conversations with Prime's head of	R. Evid. 602) ("A witness may testify to a
collections affirmed the deliberate reduction in	matter only if evidence is introduced
collection efforts. Prime's PFS Director, Ana	sufficient to support a finding that the
Goff, stated on November 17, 2020 that there	witness has personal knowledge of the
were systems issues preventing her team's	matter"); <u>Relevance ((Fed. R. Evid. 402)</u>
ordinary-course follow up, without explaining	("Irrelevant evidence is not admissible");
what type of issues. In addition, Ms. Goff	Prejudice ((Fed. R. Evid. 403) ("[t]he cou
indicated that collections people had been	may exclude relevant evidence if its
pulled away from Verity A/R to focus on a	probative value is substantially outweight
Prime audit. She stated that the impact was	by a danger of one or more of the
effectively two collectors working five days a	following: unfair prejudice, confusing the
week. Ms. Goff also stated that she had staffing	issues, misleading the jury, undue delay,
issues and had insufficient staff to assist her	wasting time, or needlessly presenting
with the legacy accounts."	cumulative evidence").
	Through this testimony, Mr. Chadwick
	purports to characterize the contents of
	certain alleged statements by Ana Goff.
	Mr. Chadwick does not provide the date
	upon which such comments occurred,
	through what medium, who was present,
	nor any other details from which to
	conclude that Mr. Chadwick was actually

## Case 2:18-bk-20151-ER Doc 6669-10 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Prime Healthcare Services Inc.s Evidentiary Objections Page 8 of 11

1	Material Objected To:	Specific Objections:
2		
3		present with or personally witnessed any
4		such conversations. This is for good
5		reason, as described in the Declaration of
6		Ana Goff being filed concurrently with
7		these objections, Ms. Goff has never
8		spoken with Mr. Chadwick. Mr.
9		Chadwick likewise does not even describe
10		where he heard about these alleged
11		statements, under what circumstances, or
11		from who. The declaration is utterly
12		devoid of any basis from which to
		conclude that he is in any respect
14		competent to testify to alleged statements
15		as to which he was plainly nowhere in the
16		room. This evidence is therefore utterly
17		unreliable. Accordingly, Mr. Chadwick
18		lacks personal knowledge of the purported
19		statements described in this testimony. In
20		addition, because Mr. Chadwick's
21		testimony is without personal knowledge,
22		it is irrelevant and the admission of such
23		testimony is unduly prejudicial.
24		Accordingly, this testimony should be
25		stricken and disregarded. Finally, Mr.
26		Chadwick's statements here are
27		Chauwick's statements here are
28		

## Case 2:18-bk-20151-ER Doc 6669-10 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Prime Healthcare Services Inc.s Evidentiary Objections Page 9 of 11

	1	
	2	
	3	
	4	
	5	
	6	
	7	
	8	
	9	
1	0	
1	1	
1	2	
1	3	
1	4	
1	5	
1	6	
1	7	
1	8	
1	9	
2	0	
2	1	
2	2	
2	3	
2	4	
2	5	
2	6	
2	7	
2	8	

Material Objected To:	Specific Objections:
	impermissible double hearsay, as to whic
	no exception applies.
4. <u>Chadwick Decl., ¶ 14, lns: 23:20-25.</u>	4. <u>Settlement Privilege</u> ((Fed. R. Evid
"In negotiations between the parties to	408) ("[e]vidence of (1)offeringor (2)
resolve this dispute, a telephone conversation	acceptinga valuable consideration in
that Prime indicates occurred on February 18,	compromising or attempting to compromi
2021, Steve Aleman, Prime's CFO conceded	a claim which was disputed as to either
that if LA County Trauma payments were	validity or amount, is not admissible to
included in the A/R Target Amount then that	prove liability for or invalidity of the clai
should be dispositive of whether they should	or its amount. Evidence of conduct or
be included in the Final A/R Collected.	statements made in compromise
Afterwards, Prime and its counsel have	negotiations is likewise not admissible.")
denied that Aleman made such a representation, but I stand by what I heard him say in that telephone conversation."	By its express terms, the entirety of this testimony relates to statements allegedly made "[i]n negotiations between the parties to resolve this dispute" While flatly wrong, this testimony is therefore Mr. Chadwick's own express admissions "evidence of conduct or a statement made during compromise negotiations and squarely within the scope of Section 408's protections. <sup>1</sup> This testimony shou therefore be stricken, and the Debtors

<sup>&</sup>lt;sup>1</sup>The Debtors concede that the statements were made "in settlement discussions. . ." *See* Objection p.10, fn 32.

Material Objected To:	Specific Objections:
	reference to or attempts to introduce
	settlement privileged material.
	settlement privileged material.

	Case 2:18-bk-20151-ER Doc 6669-10 Filed 09/28/21 Entered 09/28/21 18:34:40 Desc Prime Healthcare Services Inc.s Evidentiary Objections Page 11 of 11		
1	Dated: September 28, 2021		
2	Los Angeles, California		
3	McDermott Will & Emery, LLP		
4	By: <u>/s/ Jason D. Strabo</u> Jason D. Strabo (SBN 246426)		
5	2049 Century Park East, Suite 3200 Los Angeles, California 90067-3206		
6	Telephone: (310) 788-4125 Facsimile: (310) 277-4730		
7	E-mail: jstrabo@mwe.com		
8	- and —		
9	James W. Kapp, III (admitted <i>pro hac vice</i> ) 444 West Lake Street, Suite 4000		
10	Chicago, IL 60606-0029 Telephone: (312) 372-2000		
11	Facsimile: (312) 984-7700 E-mail: jkapp@mwe.com		
12			
13	Attorneys for Prime Healthcare Services, Inc.		
14			
15			
16			
17			
18			
19			
20			
21			
22			
23			
24			
25			
26			
27			
28	_		
	9		